Game Theory in Wireless and Communication Networks

This unified treatment of game theory focuses on finding state-of-the-art solutions to issues surrounding the next generation of wireless and communication networks. Future networks will rely on autonomous and distributed architectures to improve the efficiency and flexibility of mobile applications, and game theory provides the ideal framework for designing efficient and robust distributed algorithms. This book enables readers to develop a solid understanding of game theory, its applications, and its use as an effective tool for addressing various problems in wireless communication and networking.

The key results and tools of game theory are covered, as are various real-world technologies including 3G/4G networks, wireless LANs, sensor networks, cognitive networks, and Internet networks. The book also covers a wide range of techniques for modeling, designing, and analyzing communication networks using game theory, as well as state-of-the-art distributed design techniques. This is an ideal resource for communications engineers, researchers, and graduate and undergraduate students.

Zhu Han is an Assistant Professor of Electrical and Computer Engineering at the University of Houston. He was awarded his Ph.D. in Electrical Engineering from the University of Maryland, College Park, in 2003 and worked for two years in industry as an R&D Engineer for JDSD.

Dusit Niyato is an Assistant Professor in the School of Computer Engineering at the Nanyang Technological University (NTU), Singapore. He received his Ph.D. in Electrical and Computer Engineering from the University of Manitoba, Canada, in 2008.

Walid Saad is an Assistant Professor at the Electrical and Computer Engineering Department at the University of Miami. His research interests include applications of game theory in wireless networks, small cell networks, cognitive radio, wireless communication systems (UMTS, WiMAX, LTE, etc), and smart grids.

Tamer Başar is a Swanlund Chair holder and CAS Professor of Electrical and Computer Engineering at the University of Illinois at Urbana-Champaign. He is a member of the US National Academy of Engineering, a Fellow of the IEEE and the IFAC, founding president of the ISDG, and current president of the AACC.

Are Hjørungnes was a Professor in the Faculty of Mathematics and Natural Sciences at the University of Oslo, Norway. He was a Senior Member of the IEEE and received his Ph.D. from the Norwegian University of Science and Technology in 2000.

Game Theory in Wireless and Communication Networks

Theory, Models, and Applications

ZHU HAN University of Houston

DUSIT NIYATO Nanyang Technological University, Singapore

WALID SAAD Princeton University

TAMER BAŞAR University of Illinois at Urbana-Champaign

ARE HJØRUNGNES University of Oslo, Norway



CAMBRIDGE UNIVERSITY PRESS Cambridge, New York, Melbourne, Madrid, Cape Town, Singapore, São Paulo, Delhi, Tokyo, Mexico City

Cambridge University Press The Edinburgh Building, Cambridge CB2 8RU, UK

Published in the United States of America by Cambridge University Press, New York

www.cambridge.org Information on this title: www.cambridge.org/9780521196963

© Cambridge University Press 2012

This publication is in copyright. Subject to statutory exception and to the provisions of relevant collective licensing agreements, no reproduction of any part may take place without the written permission of Cambridge University Press.

First published 2012

Printed in the United Kingdom at the University Press, Cambridge

A catalog record for this publication is available from the British Library

Library of Congress Cataloging in Publication Data
Game theory in wireless and communication networks: theory, models, and applications / Zhu Han... [et al.].
p. cm.
Includes bibliographical references and index.
ISBN 978-0-521-19696-3 (hardback)
1. Wireless communication systems. 2. Mobile communication systems. 3. Computer networks. 4. Telecommunication systems. 5. Game theory. I. Han, Zhu, 1974– II. Title. TK5103.2.G35 2011
621.38401'5193-dc23 2011014906

ISBN 978-0-521-19696-3 Hardback

Cambridge University Press has no responsibility for the persistence or accuracy of URLs for external or third-party internet websites referred to in this publication, and does not guarantee that any content on such websites is, or will remain, accurate or appropriate.

While on a sabbatical at the University of Hawaii, our colleague and co-author, Dr. Are Hjørungnes, went missing and passed away during a mountain run on the island of Oahu. Words fail to express our sadness and sorrow in losing our dear friend. Are, you will remain forever engraved in our hearts and memories, as the Viking who was stronger than life itself. We will always remember your openness, great spirit, and technical brilliance. We would like to dedicate this book to you, as your efforts and perseverance were instrumental in the completion of this work.

May your soul rest in peace. ZH, DN, WS, TB

To my daughter, Melody Han — Zhu Han To my family — Dusit Niyato To my wife Mary and my son Karim — Walid Saad To my wife, Tangül — Tamer Başar To my grandmother, Margit — Are Hjørungnes

Contents

2

	Prefc	исе		page xv
1	Intro	duction		1
	1.1	Brief i	ntroduction to the history of game theory	1
	1.2	Game	theory in wireless and communication networks	3
	1.3	Organ	ization and targeted audience	4
		1.3.1	Timeliness of the book	6
		1.3.2	Outline of the book	9
2	Wirel	ess netw	vorks: an introduction	14
	2.1	Wirele	ess channel models	15
		2.1.1	Radio propagation	15
		2.1.2	Interference channel	20
	2.2	Catego	prization of wireless networks	21
		2.2.1	3G cellular networks and beyond	21
		2.2.2	WiMAX networks	25
		2.2.3	WiFi networks	27
		2.2.4	Wireless personal area networks	31
		2.2.5	Wireless ad hoc networks	37
		2.2.6	Wireless sensor networks	40
	2.3	Advan	ced wireless technology	45
		2.3.1	OFDM technology	45
		2.3.2	Multiple-antenna systems	47
		2.3.3	Cognitive radio	49
Part I	Fundame	entals of	f game theory	

Non-	cooperat	ive games	55
3.1	Non-c	ooperative games: preliminaries	55
	3.1.1	Introduction	55
	3.1.2	Basics of non-cooperative games	56

3.2	Non-co	ooperative games in strategic form	58
	3.2.1	Matrix games	58
	3.2.2	Dominating strategies	61
	3.2.3	Nash equilibrium	63
	3.2.4	Static continuous-kernel games	65
	3.2.5	Mixed strategies	69
	3.2.6	Efficiency and equilibrium selection	72
3.3	Dynan	nic non-cooperative games	74
	3.3.1	Non-cooperative games in extensive form	74
	3.3.2	Repeated games	80
	3.3.3	Stochastic games	84
3.4	Specia	al classes of non-cooperative games	85
	3.4.1	Potential games	85
	3.4.2	Stackelberg games	88
	3.4.3	Correlated equilibrium	91
	3.4.4	Supermodular games	94
	3.4.5	Wardrop equilibrium	96
3.5	Summ	hary	100
Baye	sian gam	ies	101
4.1	Overv	iew of Bayesian games	101
	4.1.1	Simple example	101
	4.1.2	Static Bayesian game	102
	4.1.3	Bayesian dynamic games in extensive form	104
	4.1.4	Cournot duopoly model with incomplete information	105
	4.1.5	Auction with incomplete information	107
4.2	Applic	cations in wireless communications and networking	109
	4.2.1	Packet-forwarding game	109
	4.2.2	K-player Bayesian water-filling game	112
	4.2.3	Channel-access game	116
	4.2.4	Bandwidth-auction game	119
	4.2.5	Bandwidth-allocation game	121
4.3	Summ	ary	122
Diffe	rential ga	ames	124
5.1	Optim	al-control theory	125
	5.1.1	Dynamic programming	125
	5.1.2	The maximum principle	126
5.2	Differe	ential games	128
	5.2.1	Main ingredients and general results	128
	5.2.2	Stackelberg differential game	130
5.3	Applic	cations of differential games in wireless communications	
	and ne	etworking	136
5.4	Summ	ary	137

4

6	Evolu	utionary games	138
	6.1	The evolutionary process	
		6.1.1 Evolutionarily stable strategies	139
		6.1.2 Replicator dynamics	141
		6.1.3 The evolutionary game and reinforcement learning	143
	6.2	Applications of evolutionary games in wireless communications and	
		networking	144
		6.2.1 Congestion control	144
		6.2.2 Evolutionary game for the Aloha protocol	146
		6.2.3 Evolutionary game for WCDMA access	148
		6.2.4 Routing-potential game	149
		6.2.5 Cooperative sensing in cognitive radio	151
		6.2.6 TCP throughput adaptation	154
		6.2.7 User churning behavior	158
		6.2.8 Dynamic bandwidth allocation with evolutionary network	
		selection	163
	6.3	Summary	170
7	Соор	perative games	171
	7.1	Bargaining theory	171
		7.1.1 Introduction	171
		7.1.2 The Nash bargaining solution	172
		7.1.3 Sample applications in wireless and communication networks	178
	7.2	Coalitional game theory: basics	185
		7.2.1 Introduction	185
		7.2.2 Coalitional-game theory: preliminaries	185
	7.3	Class I: canonical coalitional games	189
		7.3.1 Main properties of canonical coalitional games	189
		7.3.2 The core as a solution for canonical coalitional games	190
		7.3.3 The Shapley value	195
		7.3.4 The nucleolus	196
		7.3.5 Sample applications in wireless and communication networks	198
	7.4	Class II: coalition-formation games	203
		7.4.1 Main properties of coalition-formation games	203
		7.4.2 Impact of a coalitional structure on solution concepts for	
		canonical coalitional games	203
		7.4.3 Dynamic coalition-formation algorithms	205
		7.4.4 Sample applications in wireless and communication networks	209
	7.5	Class III: coalitional graph games	215
		7.5.1 Main properties of coalitional graph games	215
		7.5.2 Coalitional graph games and network-formation games	216
		7.5.3 Sample applications in wireless and communication networks	219
	7.6	Summary	220

Auction theory and mechanism design			2	221
8.1	Introduc	ction and auction basics	2	222
8.2	Mechanism design		2	226
	8.2.1	Equilibrium concepts	2	226
	8.2.2	Participation and incentive compatibility	2	227
	8.2.3	Revelation principle	2	228
	8.2.4	Budget balance and efficiency	2	28
	8.2.5	Groves mechanism	2	229
	8.2.6	Impossibility and possibility	2	29
8.3	Special	auctions	2	230
	8.3.1	VCG auction	2	230
	8.3.2	Share auction	2	232
	8.3.3	Double auction	2	233
8.4	Exampl	es of communication applications	2	235
	8.4.1	Cognitive radio	2	236
	8.4.2	Physical-layer security	2	248
8.5	Summa	ry	2	251
	Auctio 8.1 8.2 8.3 8.4 8.5	Auction theory 8.1 Introduce 8.2 Mechan 8.2.1 8.2.1 8.2.1 8.2.1 8.2.2 8.2.3 8.2.4 8.2.5 8.2.6 8.3 8.3.1 8.3.2 8.3.3 8.4 Example 8.4.1 8.4.2 8.5	 Auction theory and mechanism design 8.1 Introduction and auction basics 8.2 Mechanism design 8.2.1 Equilibrium concepts 8.2.2 Participation and incentive compatibility 8.2.3 Revelation principle 8.2.4 Budget balance and efficiency 8.2.5 Groves mechanism 8.2.6 Impossibility and possibility 8.3 Special auctions 8.3.1 VCG auction 8.3.2 Share auction 8.3.3 Double auction 8.4 Examples of communication applications 8.4.1 Cognitive radio 8.4.2 Physical-layer security 8.5 Summary 	Auction theory and mechanism design28.1Introduction and auction basics28.2Mechanism design28.2.1Equilibrium concepts28.2.2Participation and incentive compatibility28.2.3Revelation principle28.2.4Budget balance and efficiency28.2.5Groves mechanism28.2.6Impossibility and possibility28.3Special auctions28.3.1VCG auction28.3.3Double auction28.4Examples of communication applications28.4.1Cognitive radio28.5Summary2

Part II Applications of game theory in communications and networking

9	Cellul	ar and b	roadband wireless access networks	255
	9.1	Uplink	c power control in CDMA networks	257
		9.1.1	Single-cell CDMA networks	258
		9.1.2	Multi-cell wireless CDMA networks	263
	9.2	Resour	rce allocation in single-cell OFDMA networks	269
		9.2.1	OFDMA resource-allocation model	270
		9.2.2	Nash bargaining solution for subcarrier allocation	272
		9.2.3	Algorithms for reaching the Nash bargaining solution	274
	9.3	Power	allocation in femtocell networks	279
		9.3.1	Femtocell power control as a Stackelberg game	280
		9.3.2	Multi-leader multi-follower Stackelberg equilibrium	284
		9.3.3	Algorithm for reaching the Stackelberg equilibrium	286
	9.4	IEEE 8	802.16 broadband wireless access networks	287
		9.4.1	Resource allocation and admission control	287
		9.4.2	Relay-station deployment in IEEE 802.16j	299
	9.5	Netwo	Network selection in multi-technology wireless networks	
		9.5.1	Network selection as a non-cooperative game	309
		9.5.2	Network selection with incomplete information	311
	9.6	Summ	ary	320
10	Wirel	ess local	area networks	321
	10.1	MAC	protocol design	322
		10.1.1	Static game	323

х
_

11

	10.1.2	Dynamic game	324
	10.1.3	Deviation detection and penalization	325
	10.1.4	Related work	326
10.2	Randor	m-access control	326
	10.2.1	Choice of utility function	327
	10.2.2	Dynamics of a random-access game	328
	10.2.3	Extension with propagation delay and	
		estimation error	329
	10.2.4	Related work	329
10.3	Rate se	election for VoIP service on WLAN	330
	10.3.1	Game formulation	330
	10.3.2	Payoff function	331
10.4	Access	-point selection	332
	10.4.1	Formulation of a population game	333
	10.4.2	Price of anarchy	335
	10.4.3	Access pricing	335
	10.4.4	Related work	336
10.5	Admiss	sion control	337
	10.5.1	Two-player game formulation	337
	10.5.2	Interpretation of payoff	339
10.6	WiFi ad	ccess-point pricing	339
	10.6.1	Pricing scheme for direct payment	340
	10.6.2	User with Web browsing	341
	10.6.3	User with file transfer	342
	10.6.4	Model for uncertain application	343
10.7	Summa	ary	344
Multi-	hop netv	vorks	345
11 1	Routing	g-game basics	345
11.1	Cooper	ration enforcement and learning using a repeated game	349
11.2	11 2 1	System model and problem formulation	349
	11.2.1	Self-learning cooperation-enforcing framework	350
	11.2.2	Asynchronous network	352
	11.2.4	Case analysis and performance evaluations	353
11.3	Hierard	chical routing using a network-formation game	357
	11.3.1	System model and game formulation	358
	11.3.2	Hierarchical network-formation game solution	362
	11.3.3	Hierarchical network-formation algorithm	364
	11.3.4	Simulation results and analysis	366
11.4	Other t	vpical approaches	369
	11.4.1	Price-based solution	369
	11.4.2	Truthfulness and security using auction theory	370

11.4.2Truthfulness and security using auction theory37011.4.3Evolutionary-game approach37211.5Summary373

12	Coope	erative-transmission networks	375
	12.1	Basics of cooperative transmission	376
		12.1.1 Cooperative-transmission protocols	376
		12.1.2 State of the art and impact on different layers	380
	12.2	Non-cooperative game for relay selection and power control	380
		12.2.1 Relay-selection and power-control problem	381
		12.2.2 Stackelberg-game approach	382
	12.3	Auction-theory-based resource allocation	389
		12.3.1 Resource-allocation objectives	389
		12.3.2 Share-auction approach	392
	12.4 Cooperative transmission using a cooperative game in MANET		399
		12.4.1 Selfishness in packet-forwarding networks	400
		12.4.2 Cooperative transmission using a coalitional game	402
	12.5	Cooperative routing	411
		12.5.1 Cooperative-routing algorithms	412
		12.5.2 WiMAX IEEE 802.16j	413
	12.6	Summary	416
13	Cognit	tive-radio networks	418
	13.1	Cooperative spectrum sensing	421
		13.1.1 System model	421
		13.1.2 Coalitional-game formulation	423
		13.1.3 Centralized approach and performance comparison	426
	13.2	Power allocation as a non-cooperative game	426
		13.2.1 Underlay spectrum access and power allocation	426
		13.2.2 Properties of the Nash equilibrium for power allocation	428
		13.2.3 Distributed algorithm	429
		13.2.4 Pigouvian taxation and social optimality	431
		13.2.5 Related work	432
	13.3	Medium access control	432
		13.3.1 Channel allocation	433
		13.3.2 Channel access	434
		13.3.3 Distributed algorithms	435
	13.4	Decentralized dynamic spectrum access	436
		13.4.1 Overlay dynamic spectrum access	436
		13.4.2 Utility function	438
		13.4.3 Decentralized algorithm for channel access	439
		13.4.4 Alternative algorithms	440
	13.5	Radio resource competition based on a stochastic learning game	441
		13.5.1 System model of radio resource competition	441
		13.5.2 Auction mechanism	442
		13.5.3 Secondary-user strategy	443
		13.5.4 Learning algorithm	445

13.6	Cheat-p	proof strategies for open spectrum sharing	446
	13.6.1	One-shot non-cooperative game	446
	13.6.2	Cooperative strategy	447
	13.6.3	Repeated games	448
	13.6.4	Cheat-proof strategy	449
13.7	Spectru	m leasing and cooperation	450
	13.7.1	Game formulation with instantaneous CSI	451
	13.7.2	Game formulation with long-term CSI	454
13.8	Service	-provider competition for dynamic spectrum allocation	455
	13.8.1	User demand	455
	13.8.2	Optimal price	457
	13.8.3	Related work	458
13.9	Summa	ıry	458
Intern	et netwo	rks	460
14.1	Combin	ned flow control and routing in communication networks	462
	14.1.1	Single user with multiple links	463
	14.1.2	Multiple users with multiple parallel links	465
	14.1.3	Sample Nash equilibria	471
14.2	Conges	tion control in networks with a single service provider	473
	14.2.1	Pricing and congestion control	474
	14.2.2	Non-cooperative Nash game between followers	476
	14.2.3	Optimal pricing policy for the service provider	478
	14.2.4	Network with a large number of followers	479
14.3	Pricing	and revenue sharing for Internet service providers	481
	14.3.1	Pricing game among Internet service providers	482
	14.3.2	Revenue-sharing strategies	484
	14.3.3	Distributed algorithm for finding a Nash equilibrium	485
14.4	Coopera	ative file sharing in peer-to-peer networks	487
	14.4.1	Cooperative vs. non-cooperative file sharing	489
	14.4.2	File sharing as a coalitional game in partition form	491
	14.4.3	Distributed algorithm for coalition formation	493
	14.4.4	Coalition formation in two-peer and N-peer networks	495
14.5	Summa	ıry	499
Refere	ences		501
Index			530

Preface

With the recent advances in telecommunications technologies, wireless networking has become ubiquitous because of the great demand created by pervasive mobile applications. The convergence of computing, communications, and media will allow users to communicate with each other and access any content at any time and at any place. Future wireless networks are envisioned to support various services such as high-speed access, telecommuting, interactive media, video conferencing, real-time Internet games, e-business ecosystems, smart homes, automated highways, and disaster relief. Yet many technical challenges remain to be addressed in order to make this wireless vision a reality. A critical issue is devising *distributed* and *dynamic* algorithms for ensuring a robust network operation in time-varying and heterogeneous environments. Therefore, in order to support tomorrow's wireless services, it is essential to develop efficient mechanisms that provide an optimal cost-resource-performance tradeoff and that constitute the basis for next-generation ubiquitous and autonomic wireless networks.

Game theory is a formal framework with a set of mathematical tools to study the complex interactions among interdependent rational players. For more than half a century, game theory has led to revolutionary changes in economics, and it has found a number of important applications in politics, sociology, psychology, communication, control, computing, and transportation, to list only a few. During the past decade, there has been a surge in research activities that employ game theory to model and analyze modern communication systems. This is mainly due to (i) the emergence of the Internet as a global platform for computation and communication, which has sparked the development of largescale, distributed, and heterogeneous communication systems; (ii) the deregulation of the telecommunications industry, and the dramatic improvement in computation power, which has made it possible for various network entities to make independent and selfish decisions; and (iii) the need for robust designs against uncertainties, e.g., in security situations that can sometimes be modeled as games of users with malicious intent.

Consequently, combining game theory with the design of efficient distributed algorithms for wireless networks is desirable but at the same time challenging. On the one hand, wireless network users are generally selfish in nature. For instance, distributed mobile users tend to maximize their own performance, regardless of how this maximization affects the other users in the network, subsequently giving rise to competitive scenarios. On the other hand, in some scenarios, cooperation is required among wireless network users for performance enhancement. These situations recently motivated researchers and engineers to adopt game-theoretic techniques for characterizing competition and cooperation in wireless networks. As a result, game theory has been applied to solve many problems in wireless systems, e.g., those that arise in power control, network formation, admission control, cognitive radio, and traffic relaying. In fact, game theory provides solid mathematical tools for analyzing competition and cooperation in an ensemble of multiple players having individual self-interests. Various solution concepts from game theory are highly appropriate for communications and networking problems, such as equilibrium solutions that are desirable in competitive scenarios, since they lead to designs that are robust to the deviations made by any player. There are many popular wireless and communications applications that have recently explored game-theoretic techniques, including, but not limited to, cognitive radio, heterogeneous wireless networks, cellular networks, cooperative networks, and multi-hop networks. It is now commonly acknowledged that within the rich landscape of game theory, new aspects of network design (e.g., with cooperative and non-cooperative behaviors of the wireless entities) can be investigated using appropriate solution concepts.

Although game theory has been applied to wireless communications and networking for many years, there are only a few books that allow researchers, engineers, and graduate/undergraduate students to study game theory from an engineering perspective. On the one hand, most of the existing game theory books focus on the mathematical and economical aspects, which are considerably different from the engineering (and particularly the application-oriented) perspective. On the other hand, the wireless communications and networking books focus mainly on system optimization or control techniques while overlooking distributed algorithms. In addition, the cooperative and non-cooperative behaviors of the network entities (e.g., users or service providers) cannot be modeled and analyzed effectively using the techniques presented in these books. Therefore, there is a need to develop a comprehensive and useful reference source that can provide complete coverage on how to adequately apply game theory to the design of wireless communications and networking.

In this regard, this book not only focuses on the description of the main aspects of game theory in the context of wireless communications, but also provides an extensive review of the applications of game theory in wireless communications and networking problems. In a nutshell, it provides a comprehensive treatment of game theory in wireless communications and networking. The topics range from the basic concepts of game theory to the state of the art of analysis, design, and optimization of game-theoretic techniques for wireless and communication networks. The three main objectives of this book are as follows:

- This book introduces the basics of game theory from an engineering perspective. In particular, the basics of game theory are explained and discussed in the context of wireless communications and networking. For example, the book provides a clear description of the main game-theoretic entities in a communication environment (e.g., the players, their strategies, utilities and payoffs, and the physical meaning, in a wireless network environment, of the different game-theoretic concepts such as equilibria).
- This book provides an extensive review/survey of the applications of game theory to wireless communications and networking. With this review/survey of applications,

readers can understand how game theory can be applied in different wireless systems and can acquire an in-depth knowledge of the recent developments in this area. In this context, this book presents tutorial-like chapters that explain, clearly and concisely, how game-theoretic techniques can be applied to solving state-of-the-art wireless communications problems. In particular, the benefits of using game theory in wireless communications environments are emphasized. The target audience of this book are researchers, engineers, and undergraduate and graduate students who are looking for a self-contained book from which to learn game theory and its application to multi-player decision-making problems in wireless and other engineering systems.

• Most of the research in this field has been focused on applying standard game-theoretic models and techniques to several limited topics, such as power control in wireless networks and routing in wire-line networks. However, game theory is a very powerful tool and can help us better understand many other aspects of communication networks. The goals of this book are to provide the fundamental concepts of game theory and also to bring together the state-of-the-art research contributions that address the major opportunities and challenges of applying game theory in wireless engineering problems. The applications presented here are varied and cover a significant part of the most recent challenges and problems in wireless communications and networking systems. In this respect, we believe that this book will be useful to a variety of readers from the wireless communications and networking fields. The material from this book can be used to design and develop more efficient, scalable, and robust communication protocols.

To summarize, the key features of this book are

- · a unified view of game-theoretic approaches to wireless networks
- comprehensive treatment of state-of-the-art distributed techniques for wireless communications problems
- coverage of a wide range of techniques for modeling, designing, and analyzing of wireless networks using game theory
- an outline of the key research issues related to wireless applications of game theory.

We would like to thank Dr. K. J. Ray Liu, Dr. Vincent Poor, Dr. John M. Cioffi, Dr. Luiz DaSilva, Dr. Allen MacKenzie, Dr. Mérouane Debbah, Dr. Ekram Hossain, Dr. Jianwei Huang, Dr. Ninoslav Marina, Dr. Guan-Ming Su, Dr. Yan Sun, Dr. Husheng Li, Dr. Beibei Wang, Dr. Charles Pandana, Dr. Zhu Ji, Dr. Rong Zheng, Dr. Xinbing Wang, Dr. Amir Leshem, Dr. Tansu Alpcan, Dr. Eduard Jorswieck, Mr. Quanyan Zhu, Dr. Eitan Altman, Dr. Corinne Touati, and Dr. María Ángeles Vázquez-Castro for their support on the book. We also would like to acknowledge the support of Mr. Ray Hardesty for text-editing and Ms. Jessy Stephan for her book cover design.

We would also like to acknowledge various granting agencies that supported part of the work reported in this book. These agencies are the US NSF through grants CNS-0905556, CNS-0910461, CNS-0953377, and ECCS-1028782; NTU Start-Up Grant – Project "Radio Resource Management in Heterogeneous Wireless Networks"; Singapore Ministry of Education (MOE) AcRF Tier 1 – Project "Radio Resource Management

over Cognitive Radio Networks"; A*STAR – SERC (Science and Engineering Research Council) "Data Value Chain as a Service" – Project "Design and Analysis of Cloud Computing for Data Value Chain: Operation Research Approach"; the Research Council of Norway for their funding of the VERDIKT Project "Mobile-to-Mobile Communication Systems (M2M)" (project number 183311/S10) and the FRITEK Project "Theoretical Foundations of Mobile Flexible Networks (THEFONE)" (project number 197565/V30); and the US AFOSR and DOE through grants AF FA9550-09-1-0249 and DOE SC0003879 ARRA.

Zhu Han Dusit Niyato Walid Saad Tamer Başar Are Hjørungnes

1.1 Brief introduction to the history of game theory

Game theory can be viewed as a branch of applied mathematics as well as of applied sciences. It has been used in the social sciences, most notably in economics, but has also penetrated into a variety of other disciplines such as political science, biology, computer science, philosophy, and, recently, wireless and communication networks. Even though game theory is a relatively young discipline, the ideas underlying it have appeared in various forms throughout history and in numerous sources, including the Bible, the Talmud, the works of Descartes and Sun Tzu, and the writings of Charles Darwin, and in the 1802 work *Considérations sur la Théorie Mathématique du Jeu* of André-Marie Ampère, who was influenced by the 1777 *Essai d'Arithmétique Morale* of Georges Louis Buffon. Nonetheless, the main basis of modern-day game theory can be considered an outgrowth of three seminal works:

- Augustin Cournot's *Mathematical Principles of the Theory of Wealth* in 1838, which gives an intuitive explanation of what would, over a century later, be formalized as the celebrated Nash equilibrium solution to non-cooperative games. Furthermore, Cournot's work provides an evolutionary or dynamic notion of the idea of a "best response," i.e., situations in which a player chooses the best action given the actions of other players, this being so for all players.
- Francis Ysidro Edgeworth's *Mathematical Physics* (1881), which demonstrated the notion of competitive equilibria in a two-person (as well as two-type) economy, and Emile Borel's *Algebre et Calcul des Probabilites* (*Comptes Rendus Academie des Sciences*, volume 184, 1927), which provided the first insight into mixed strategies, i.e., that randomization may support a stable outcome.
- While many other contributors hold places in the history of game theory, it is widely accepted that modern analysis started with John von Neumann and Oskar Morgenstern's 1944 book, *Theory of Games and Economic Behavior*, and was given its modern methodological framework by John Nash's seminal work on non-cooperative games and bargaining, which had von Neumann and Morgenstern's results as a first building block. It is worth mentioning that some two decades prior to this, in 1928, John von Neumann himself had resolved completely an open fundamental problem in zero-sum games, that *every finite two-player zero-sum game admits a saddle point in mixed strategies*, which is known as the *Minimax Theorem* [492]—a result which Emile Borel had conjectured to be false eight years earlier.

Following the publication of von Neumann and Morgenstern's book, and the seminal work of John Nash, game theory has enjoyed over 65 years of scientific development, and has experienced incessant growth in both the number of theoretical results and the scope and variety of applications. As a recognition of the vitality of the field, a total of three Nobel Prizes have been given in the economic sciences for work primarily in game theory, with the first such recognition given in 1994 to John Harsanyi, John Nash, and Reinhard Selten "for their pioneering analysis of equilibria in the theory of non-cooperative games." The second Nobel Prize went to Robert Aumann and Thomas Schelling in 2005, "for having enhanced our understanding of conflict and cooperation through game-theory analysis." And the most recent one was in 2007, recognizing Leonid Hurwicz, Eric Maskin, and Roger Myerson, "for having laid the foundations of mechanism design theory." We should add to this list of highest-level awards in game theory the Crafoord Prize (the highest prize in the biological sciences), which went to John Maynard Smith (along with Ernst Mayr and G. Williams) in 1991 "for developing the concept of evolutionary biology;" Smith's recognized contributions had a strong game-theoretic underpinning, through his work on evolutionary games and evolutionarily stable equilibrium.

One classical example of game theory is the so-called "Prisoner's Dilemma." This game captures a scenario in which a conflict of interest arises because of the requirement of independent decision-making. The Prisoner's Dilemma pertains to analyzing the decision-making process in the following hypothetical setting. Two criminals are arrested after being suspected of a crime in unison, but the police do not have enough evidence to convict either. Thus, the police separate the two and offer them a deal: if one testifies against the other, he will get a reduced sentence or go free. Here, the prisoners do not have information about each other's "moves," as they would in some social games such as chess. The payoff if they both say nothing (and thus cooperate with each other) is somewhat favorable, since neither can be convicted of the real crime without further proof (though they will be convicted of a lesser crime). If one of them betrays and the other one does not, then the betrayer benefits because he goes free while the other one is imprisoned, since there is now sufficient evidence to convict the silent one. If they both confess, they both get reduced sentences, which can be viewed as a null result. The obvious dilemma is the choice between two options, where a favorable decision, acceptable to both, cannot be made without cooperation.

A representative Prisoner's Dilemma is depicted in Table 1.1. One player acts as the row player and the other the column player, and both have the action options of cooperating (*C*) or defecting (*D*). Thus, there are four possible outcomes to the game: $\{(C, C), (D, D), (C, D), (D, C)\}$. Under mutual cooperation, $\{(C, C)\}$, both players will receive a reward payoff of 3. Under mutual defection, $\{(D, D)\}$, both players receive the punishment of defection, 1. When one player cooperates while the other one defects, $\{(C, D), (D, C)\}$, the cooperating player receives a payoff of, 0, and the defecting player receives the temptation to defect, 5.

In The Prisoner's Dilemma example, if one player cooperates, the other player will have a better payoff (5 instead of 3) if he or she defects; if one player defects, the other player will still have a better payoff (1 instead of 0) if he or she also defects. Regardless of the other player's strategy, a player in The Prisoner's Dilemma has an incentive to

	Cooperate	Defect
Cooperate	(3,3)	(0,5)
Defect	(5,0)	(1,1)

always select defection, and $\{(D, D)\}$ is an equilibrium. Although cooperation will give each player a better payoff of 3, greediness and lack of trust leads to an inefficient outcome. This simple example shows how the game-theoretic concept of an equilibrium can provide a lot of insight into the outcome of decision-making in an adversarial or conflicting situation.

1.2 Game theory in wireless and communication networks

Recent advances in technology and the ever-growing need for pervasive computing and communication have led to an incessant need for novel analytical frameworks that can be suited to tackle the numerous technical challenges accompanying current and future wireless and communication networks. As a result, in recent years game theory has emerged as a central tool for the design of future wireless and communication networks. This is mainly due to the need for incorporating decision-making rules and techniques into next-generation wireless and communication nodes, to enable them to operate efficiently and meet the users' needs in terms of communication services (e.g., video streaming over mobile networks, ubiquitous Internet access, simultaneous use of multiple technologies, peer-to-peer file sharing, etc.).

One of the most popular examples of game theory in wireless networks pertains to modeling the problem of power control in cellular networks using non-cooperative games. For example, in the uplink of a cellular system, researchers and engineers have been concerned with the problem of designing a mechanism that allows the users (utilizing a common frequency such as in a CDMA system) to regulate their transmit power, given the interference that they cause (or that is caused by the other users) in the network. In doing so, wireless researchers were able to draw a striking similarity between the problems of power control and non-cooperative game theory. In a non-cooperative game, a number of players are involved in a competitive situation in which, whenever a player makes a move (or chooses a strategy), this move has an impact (positive or negative) on the utility (e.g., a measure of benefit or gain) of the other players. Similarly, in a power control game, we have a competitive situation in which the transmit power level (strategy) of a wireless user can impact positively or negatively (because of interference) on the transmission rate and quality of service (QoS) of the other users. As a result, solving a power control game has been shown to be equivalent to solving a non-cooperative game, e.g., by finding a Nash equilibrium. Power control is only one example in which game theory can be used to design next-generation wireless and communication networks. In fact, following the early work on non-cooperative games in power control, a plethora of novel application areas for game theory have emerged in the wireless, communications, and signal processing communities.

The key challenge in applying game theory in a communications context lies in the fact that game theory was essentially developed as a tool to be used in economics and the social sciences. Hence, leveraging game theory for use in engineering applications is accompanied by many difficulties. For instance, researchers interested in applying game-theoretic models to problems in wireless and communication networks face many hurdles in finding accurate models and solutions. This is due to the fact that existing game-theoretic models are not tailored to cope with engineering-specific issues such as modeling time-varying wireless channels, developing performance functions (i.e., utilities) that depend on restrictive communication metrics (e.g., transmission rate, queueing delay, signal-to-noise ratio), and conforming to certain standards (e.g., IEEE 802.16, LTE). This has necessitated a timely, comprehensive reference source that can guide researchers and communications engineers in their quest to find effective analytical models from game theory that can be applied to the design of future wireless and communication networks.

1.3 Organization and targeted audience

Our aim with this book is to provide researchers and engineers working in communications and networking with a comprehensive and detailed introduction to game theory, as relevant to wireless and communication networks. After introducing some fundamentals of wireless networks, the book starts, in Part I, with an in-depth study of important game-theoretic frameworks. In this part of the book, we mainly focus on presenting important classes of games that admit potential applications in wireless and communication networks. In essence, Part I provides a detailed study of a variety of games ranging from classical non-cooperative games to more advanced games such as dynamic games, coalitional games, network-formation games, Bayesian games, evolutionary games, and auction theory. For each type of game, we focus on the fundamental notions, possible solutions, key objectives, and important properties, while highlighting potential application scenarios in a game-theoretic as well as a communications and networking environment. Thus, in each chapter of Part I we start with an overview of the studied class of games, and then delve into key elements such as game components, solution concepts, and mathematical properties of the studied game. In each chapter we provide carefully selected examples from game theory and wireless networks to enable readers to grasp the presented ideas and to start drawing some links between the problems solved in game theory and their counterparts in the communications world. The objective of Part I is, thus, to provide a thorough treatment of the key branches of game theory, while starting to show that such game-theoretic concepts, originally rooted in economics, have a lot to offer in addressing the problems that face researchers and engineers working in wireless and communication networks.

After laying the foundations of game-theoretic techniques and drawing their connections to the wireless and communication worlds, in Part II of the book we start developing

5

game-theoretic models in a wide range of wireless and communication applications such as cellular and broadband networks, wireless local area networks, multi-hop networks, cooperative networks, cognitive-radio networks, and Internet networks. Each chapter in Part II constitutes a didactic study that explains how game theory can be applied to solve key problems in a state-of-the-art field within wireless and communication networks. In Part II, within every application area we enable readers to understand how, using the game-theoretic techniques studied in Part I, one can solve challenging problems such as resource allocation, MAC (medium access control) protocol design, randomaccess control, network selection, cooperative routing and packet forwarding, spectrum sensing in cognitive networks, dynamic spectrum access, flow control and routing in Internet networks, a peer-to-peer incentive mechanisms. Within each chapter of Part II, we start by identifying the main technical challenges and problems of the studied application area. Then, after clearly determining the system model of interest, we highlight the problem that needs to be treated, and we map it to a relevant, sufficiently rich class of games as described in Part I. Once the game is formulated by identifying its components, we apply suitable solution concepts and discuss the insights that they yield within the context of the studied problem. We also shed light on potential extensions and future uses of the developed game-theoretic techniques and communication models. In particular, Part II shows how concepts such as the Nash equilibrium, the Stackelberg equilibrium, and evolutionarily stable strategies, can yield meaningful outcomes and implications within a wireless and communication problem. Hence, the objective of Part II is to demonstrate the usefulness of game theory in the design of future wireless and communication networks as well as to provide readers with exhaustive guidelines to enable them to develop networking-oriented game-theoretic approaches using Part I as a basis.

In a nutshell, the main goal of the book is to formalize the use of game theory in wireless and communication networks, by providing not only an introduction to the fundamental branches of game theory but also a thorough and instructive treatment on developing game-theoretic techniques for analyzing state-of-the-art and emerging communications and networking applications. The main goal of the book can, thus, be summarized in the following three objectives:

- The first objective is to provide a general introduction to wireless communications and networking while pinpointing the most recent developments and challenges. These aspects are discussed, in detail, throughout the book.
- The second objective is to introduce different game-theoretic techniques and their applications for designing distributed and efficient solutions for a diverse number of wireless communications and networking problems. This is mainly dealt with in Part I of the book.
- The third objective is to provide a didactic study of how game theory can be leveraged for use in state-of-the-art and emerging applications in wireless and communication networks. This includes identifying key problems in a variety of communications applications, linking them to game-theoretic frameworks, and studying the properties and implications of the solutions and outcomes.

By achieving these objectives, the book enables the reader to clearly identify the links and connections between the technical challenges looming in future wireless communication networks and the classical economics-oriented applications of game theory. In particular, in recent years, engineers and researchers in the wireless communication community have been seeking a reference source, such as this book, that integrates the notions of game theory and of wireless engineering, while emphasizing how game theory can be applied in wireless networks from an engineering perspective. This book serves this purpose, and is intended, primarily, for the following audience:

- communications engineers interested in studying the new tools of distributed optimization and management in wireless networking systems
- researchers interested in state-of-the-art research on distributed algorithm design, cooperation, and networking for a wide range of wireless communication applications
- graduate and undergraduate students interested in obtaining comprehensive information on the design and evaluation of game-theoretic approaches to find suitable topics for their dissertations.

1.3.1 Timeliness of the book

Because of the rapid growth in communication networks and its projected evolution, a broad range of novel technical challenges are emerging daily. This requires solid and robust analytical frameworks, such as game theory, that can enable researchers in the wireless and communications industry to overcome these challenges. Hence, this book constitutes a timely contribution, for the following reasons:

Promising distributed game-theoretic approaches for future wireless networks. In recent years, there has been an unprecedented increase in consumer demand for wireless services. This growing demand has led to the emergence of large-scale wireless networks that cover huge areas and that are expected to meet stringent quality-of-service (QoS) requirements. In this regard, wireless network entities such as base stations are unable to cope with this growth, which requires such entities to gather a large amount of information from the network (e.g., channel conditions, users' actions, etc.), which in turn yields extensive complexity, overhead, and signaling. Consequently, devising distributed solutions and algorithms constitutes a promising direction for the efficient design of future wireless networks. Nonetheless, deriving distributed algorithms for wireless networks is accompanied by several challenging issues. On the one hand, wireless network users are generally selfish. For instance, distributed mobile users tend to maximize their own performance, regardless of how this maximization affects the other users in the network, giving rise to competing scenarios. On the other hand, in some scenarios, cooperation is required among wireless network users in order to achieve the best performance. These situations recently motivated researchers and engineers to adopt game-theoretic techniques for characterizing competition and cooperation in wireless networks. As an example, distributed resource allocation can be modeled as a game that deals largely with how rational and intelligent individuals interact with each other in an effort to achieve

their own goals in terms of sharing the network resources. In this game, each mobile user is self-interested and will attempt to optimize its own benefit. In brief, applying game theory in future wireless networks presents many advantages:

- Local information-based decisions and distributed implementation: By using gametheoretic approaches, individual mobile users optimize their performance by taking individual decisions based on the local observation of a well-defined game's outcome. As a result, by adopting game-theoretic models, there is no need for collecting global information and conducting optimization in a centralized manner.
- More robust outcomes: In large-scale wireless networks, adopting centralized solutions for optimizing performance may yield inefficient results owing to errors occurring during the complex information-gathering phase. In contrast, local information is generally more reliable and accurate. Hence, in many situations, the outcome of distributed game approaches is more robust than that of centralized solutions.
- Convenient approaches for solving problems of a combinatorial nature: Traditional optimization techniques such as mathematical programming require handling combinatorial problems that are inherently hard to manipulate. In game theory, most problems are naturally studied in a discrete form, which is relatively easy to analyze. For example, in a cognitive-radio network, analyzing the spectrum access strategy of the unlicensed user using game theory is tractable, while solving this problem in a centralized manner with reasonable complexity is not feasible in many cases.
- Rich mathematical and analytical tools for optimization: Game theory provides a variety of analytical and mathematical tools for adequately analyzing the outcome of well-defined classes of games. For instance, in non-cooperative games, static games (i.e., games in which decisions are made only once) can be solved using well-defined notions such as the best-response function and the Nash equilibrium. Moreover, in dynamic games (i.e., games in which decisions can be applied (e.g., behavioral equilibria, repeated-game solutions). In addition, whenever cooperation between players is required, cooperative game theory provides a rich framework suitable for such an analysis. Finally, auction theory as well as other game-theoretic concepts can be applied for efficient and robust mechanism design in various situations (e.g., bidder/seller games).

Most existing game theory books are oriented toward economic aspects, and most existing network optimization books focus on centralized approaches. In the current market, most books dealing with game theory and its applications draw their applications from economics. As a result, such books are difficult for engineers to understand and use, because of unfamiliar terminology as well as a significant number of assumptions (e.g., demand/supply and transferable money) that are fundamentally different from engineering problems. In addition, most existing books dealing with wireless network optimization study centralized approaches such as constrained optimization. Consequently, there is a gap between understanding game theory and applying it to the design of next-generation wireless networks. Moreover, designing game-theoretic solutions for wireless networks requires interdisciplinary knowledge from multiple scientific and engineering disciplines to achieve the desired design objectives. Therefore, a unified treatment of this subject area is desirable. In this regard, this book aims to fill this void in the literature by closely combining game-theoretic approaches with wireless network design problems. Briefly, this book will provide a unified reference source on the application of game theory to wireless networks, tailored to the technical needs of engineers.

Emergence of new wireless applications and services. The emergence of a large class of wireless applications requiring distributed solutions is a motivation for the application of game theory. A few of these emerging wireless applications are as follows:

- Cognitive radios: The introduction of cognitive radios in future wireless networks faces several challenges that require a broad range of analytical tools from game theory such as non-cooperative games and mechanism design. For example, the spectrum can be accessed by non-cooperative multiple unlicensed users, or it can be traded among licensed and unlicensed users.
- Cooperative communication: Recently, there has been a growing interest in studying cooperative scenarios in wireless networks. It has been shown that, through cooperation, the wireless network performance can be significantly improved. Hence, cooperative communication is rapidly emerging as a pillar technology in next-generation wireless networks, and it has already been incorporated in various standards, such as the IEEE 802.16 WirelessMAN (WiMAX) family of broadband networks. The introduction of cooperative communication in wireless networks faces several challenges (deriving autonomous and distributed cooperative strategies, analyzing users' interactions, etc.) that can only be analyzed by solid and robust analytical tools such as game theory.
- Autonomic communication in heterogeneous networks: Currently, a broad range of wireless-network standards exists (UMTS, LTE, WiMAX, etc.), with each type of network having its own characteristics. Consequently, there is a need to produce wireless devices that can autonomously operate within heterogeneous environments, allowing for interoperability between these wireless standards. Autonomic communications aims to: (i) provide distributed algorithms that can ease the burden of managing complex and heterogeneous networks, and (ii) provide large-scale networks that are self-configuring, self-organizing, and able to learn and adapt to their environments (changes in topology, technologies, service requirements, etc.). Clearly, game theory is the natural tool for achieving these objectives of autonomic communications.
- Wireless intelligent transportation system: A wireless intelligent transportation system (ITS) refers to an integrated wireless communication and software system that facilitates information exchange and processing for improving the safety and the efficiency of vehicle transportation. Since mobility is a key feature in such a communication environment, a distributed and efficient wireless communication system can improve

system performance. For example, the vehicular node can relay safety-related data of other nodes, or the vehicular nodes can download data from a roadside unit. If the vehicular nodes have self-interests, radio resource management based on a game model would be required to obtain equilibrium solutions. Essentially, an equilibrium solution must be obtained as quickly as possible because the connection duration of vehicular nodes is very short, owing to the high mobility of the vehicles. In this case, speed of convergence will be crucial for the rational vehicular node to access the radio resources required for supporting wireless ITS services.

- Multi-hop communications: The service area and throughput of a wireless network can
 be improved by using multi-hop communication (e.g., ad hoc and mesh network). Various wireless technologies will support multi-hop communication (e.g., IEEE 802.16).
 In such a network, wireless nodes interact with one another to relay their data to the
 destination. If these wireless nodes have self-interests, the data relaying behavior of
 each node can be modeled using game theory. The equilibrium relaying strategy will
 provide a stable solution for each wireless node in a multi-hop network. Moreover,
 several other aspects of multi-hop communication can be modeled using game theory,
 including distributed topology design and distributed relaying.
- Mobile wireless multimedia network: With the need to support multimedia applications, wireless networks have to be designed to provide QoS guarantee and reliable multimedia communication. In this case, the multimedia users can have heterogeneous QoS requirements that the radio resource management algorithm is required to handle adequately. In this context, game theory can be applied to wireless multimedia networks to obtain a fair and efficient solution for radio resource sharing between the mobile multimedia users.

Applications of game-theoretic concepts in traditional wireless systems. Gametheoretic techniques can be readily applied to traditional wireless communication systems to achieve a better flexibility of radio resource usage so that system performance can be improved while the signaling overhead is reduced. For example, load balancing/dynamic channel selection in traditional cellular wireless systems and WLANs, distributed subcarrier allocation in orthogonal frequency-division multiplexing (OFDM) systems, transmit power control in ultra wideband (UWB) systems, and spectrum access for cognitive radios can be achieved by using distributed game-theoretic techniques.

1.3.2 Outline of the book

To achieve the aforementioned objectives, the book is organized as follows.

In Chapter 2, we first study the basic characteristics of wireless channels. Then we introduce different wireless access technologies (e.g., cellular wireless, WLAN, WMAN, WPAN, and WRAN technologies) and the related standards. Some typical wireless networks such as ad hoc/sensor networks will also be presented. This includes the basic components, features, and potential applications. Then, advanced wireless technologies such as OFDM, MIMO, and cognitive radio are discussed. For distributed

implementation, the research challenges in the different layers of the protocol stack are discussed.

Part I: Fundamentals of game theory

Before we discuss how to apply game theory in different wireless network problems, the choice of a design technique is crucial and must be emphasized. In this context, this part presents different game-theoretic techniques that can be applied to the design, analysis, and optimization of wireless networks.

- In Chapter 3, the best-known type of games (i.e., non-cooperative games) is discussed. Various non-cooperative static games, in which multiple users (or players) are selfish and engage in a non-cooperative competition, are presented. We define and discuss the celebrated Nash equilibrium concept. We also pursue our discussion by introducing and presenting dynamic and repeated games. Unlike static games in which players are involved in the decision process once, dynamic games study the evolution of the process of decision-making of the players, taking into account the presence or lack of information. For instance, when the players are allowed to act multiple times, the behavior of these players can be analyzed using various concepts from repeated or dynamic games. In addition, for repeated games, we present a number of different strategies (e.g., trigger and punishment) that can be adopted by the users. Some special game concepts are finally discussed, such as the potential game, the Stackelberg game, the correlated equilibrium, the supermodular game, and the Wardrop equilibrium.
- In Chapter 4, game models (i.e., Bayesian and learning games) with incomplete information are discussed. In general, Bayesian games are adequate for modeling scenarios in which the players lack some necessary information when making their strategic choices. Bayesian games can be used to capture this incompleteness of information. With Bayesian games, a player can develop a belief about the payoffs and strategies of other players. Alternatively, the player can implement learning algorithms to gain knowledge of the game and the environment so that a suitable equilibrium solution can be reached. Accordingly, we provide a clear introduction to Bayesian and learning games, while outlining their significance in wireless and engineering problems. Finally, we provide several examples of Bayesian game approaches such as the packetforwarding game, the *K*-player Bayesian water-filling game, the channel-access game, the bandwidth auction game, and the network game.
- Chapter 5 covers differential games which extend static non-cooperative game theory by adopting the methods and models developed in optimal control. Differential-game theory provides a means of obtaining the equilibrium solution for rational entities with time-varying objectives or payoff functions and evolving states as well as information. Two major approaches to optimal-control theory are the *dynamic programming* introduced by Bellman and the *maximum principle* introduced by Pontryagin. These approaches have been adopted in differential game theory in which the payoff for a player depends on (i.e., is constrained by) the state, which evolves over time. The

common solution concepts of a differential game are the Nash equilibrium and the Stackelberg solution for non-hierarchical and hierarchical decision-making structures, respectively. Using techniques from optimal-control theory, and beyond, not only can these solutions be obtained but their stability can also be analyzed. A study of two example games in ad hoc routing and dynamic spectrum allocation concludes the chapter.

- In Chapter 6, a special type of game, the evolutionary game, is presented. An evolutionary game can be used to analyze a situation in which the players gradually adapt their strategies (i.e., over time), which could be due to irrational behavior. The dynamics of the strategy adaptation can be modeled using a concept known as replicator dynamics. At a steady state, a special type of equilibrium, the evolutionary equilibrium, is considered to be the solution of the strategy adaptation process. Also, reinforcement learning is investigated in this chapter for achieving the equilibrium. Hence, we delve here into the details and applications of evolutionary games. Sample applications are studied, such as congestion control, the Aloha protocol, WCDMA access, the routing potential game, cooperative sensing for cognitive-radio networks, and user churning behavior.
- In Chapter 7, having covered static and dynamic non-cooperative games, we introduce cooperative game theory, which is used to analyze the situation in which players can negotiate agreements and cooperate among themselves. In this context, in a cooperative game scenario, the players are allowed to form agreements that can impact the strategic choices of the players as well as their utilities. Cooperative games encompass two main branches: bargaining theory and coalitional games. The former describes the bargaining process between a set of players who need to agree on the terms of cooperation, while the latter describes the formation of cooperating groups of players, referred to as coalitions, that can strengthen the players' positions in a game. Key characteristics, properties, and solution concepts are examined for both branches of cooperative games as well as sample applications within wireless and communication networks.
- In Chapter 8, the use of game theory for an auction process to determine the price of commodities and services is presented. Auction theory is widely used in trading if the price of a commodity is undetermined, e.g., the commodity or service is rare and has limited capacity. There are many possible designs (or sets of rules) for an auction, and typical issues studied by auction theorists include the efficiency of a given auction design, optimal and equilibrium bidding strategies, and revenue comparison. Mechanism design is a subfield of game theory, which studies solution concepts and designs for a class of private-information games. These games have two distinguishing features. First, a game "designer" chooses the game structure rather than inheriting one. Thus, the mechanism design is often called "reverse game theory." Second, the designer is interested in the game's outcome. Such a game is called a "game of mechanism design" and is usually solved by motivating players to disclose their private information. Some typical auctions such as the Vickrey–Clarke–Groves (VCG) auction, the share auction, and the double auction are investigated, followed by applications to cognitive-radio networks and physical-layer security.

Part II: Applications of game theory in communications and networking

This part of the book deals with the modeling, design, and analysis of game-theoretic schemes in communications and networking applications. Different game models that have been applied to solve a diverse set of problems in wireless and communication networks are discussed. The major research issues and challenges are also identified.

- In Chapter 9, we consider one of the most popular types of wireless networks, the mobile cellular system. In this context, we present game-theoretic formulations for various problems such as admission control, power control in a CDMA cellular network (e.g., 3G), and resource allocation for OFDMA-based wireless cellular networks (e.g., IEEE 802.16). The range of applications covered by cellular and broad-band wireless access networks is very wide and is evolving quickly. In this chapter, using a variety of game-theoretic tools, we tackle the following key technical challenges in cellular and broadband networks: uplink power control in CDMA networks, resource allocation in OFDMA networks, power control in femtocell networks, IEEE 802.16 broadband wireless access, and vertical handover in heterogeneous wireless networks.
- In Chapter 10, we review the game models developed to analyze the performance, with rational users and services providers, of wireless local area networks (WLANs), which have been widely deployed in many places for both residential and commercial usage. These models consider different aspects of WLAN, i.e., MAC protocol design, power and rate control, access point selection, admission control, service pricing, and heterogeneous wireless access.
- In Chapter 11, we review and discuss game models for multi-hop networks (e.g., ad hoc, mesh, sensor, and cooperative networks). In such networks, the optimization of routing is a critical problem that involves many aspects such as link qualities, energy efficiency, and security. First, we introduce important models and examples of routing games. Then, we provide two detailed examples (repeated routing game and hierarchical routing game) in which cooperation is enforced. Finally, we list some other typical approaches in the literature, such as price-based routing, VCG auctioning, and evolutionary-game approaches.
- In Chapter 12, we present the use of game theory in a cooperative network, which has attracted significant recent attention as a transmission strategy for future wireless networks. This efficiently takes advantage of the broadcast nature of wireless networks to allow network nodes to share their messages and transmit cooperatively as a virtual antenna array, thus providing diversity that can significantly improve the system performance. Several distributed resource-allocation examples for cooperative transmission are analyzed, including a non-cooperative game for relay selection and power control, auction theory-based resource allocation, cooperative transmission using a cooperative game in MANET, and routing problems in general multi-hop networks.
- In Chapter 13, game theory-based models are presented for a number of challenging problems in cognitive-radio networks, which is a paradigm for the design of wire-less communication systems. Cognitive radio aims to enhance the utilization of the radio-frequency spectrum. In this chapter, the following game models, developed to

analyze the performance of cognitive-radio networks with rational primary and secondary users, are covered: cooperative spectrum sensing, power allocation/control, medium access control, decentralized dynamic spectrum access, cheat-proof strategies for open spectrum sharing, spectrum leasing and cooperation, service provider competition for dynamic spectrum allocation, and price competition in spectrum trading.

• Finally, in Chapter 14, we investigate the impact of game theory on Internet-scale communication networks. To efficiently analyze and study such Internet-like networks, there is a need for rich analytical frameworks such as game theory that can provide models and algorithms to capture the numerous challenges arising in the current and emerging communication networks. This chapter will leverage the use of game theory to tackle important challenges in Internet networks, such as routing and flow control, congestion control and pricing, revenue sharing between Internet service providers, incentive mechanisms in peer-to-peer communication networks, and cooperative peer-to-peer file sharing.

In summary, the objective of this book is to provide a didactic approach to studying game theory which is tailored for use by researchers and engineers working in wireless and communication networks. Through the aforementioned organization, this book provides an easy-to-follow structure that can enable readers to grasp the fundamental concepts of game theory and their application, and constitutes a complete and comprehensive reference for game theory as it applies to problems in wireless communications and networking. A wireless network refers to a telecommunications network whose interconnections between nodes are implemented without the use of wires. Wireless networks have experienced unprecedented growth over the last few decades, and are expected to continue to evolve in the future. Seamless mobility and coverage ensure that various types of wireless connections can be made anytime, anywhere. In this chapter, we introduce some basic types of wireless networks and provide the reader with some necessary background on state-of-art development.

Wireless networks use electromagnetic waves, such as radio waves, for carrying information. Therefore, their performance is greatly affected by the randomly fluctuating wireless channels. To develop an understanding of channels, in Section 2.1 we will study the radio frequency band first, then the existing wireless channel models used for different network scenarios, and finally the interference channel.

There exist several wireless standards. We describe them according to the order of coverage area, starting with cellular wireless networks. In Section 2.2.1 we provide an overview of the key elements and technologies of the third-generation wireless cellular network standards. In particular, we discuss WCDMA, CDMA2000, TD/S CDMA, and 4G and beyond. WiMax, based on the IEEE 802.16 standard for wireless metropolitan area networks, is discussed in Section 2.2.2. A wireless local area network (WLAN) is a network in which a mobile user can connect to a local area network through a wireless connection. The IEEE 802.11 group of standards specify the technologies for WLAN. WiFi, based on IEEE 802.11, is a brand originally licensed by the WiFi Alliance to describe the WLAN technology. In Section 2.2.3, we study some specifications in IEEE 802.11 standards. A wireless personal area network (WPAN) is a personal area network for wireless interconnecting devices centered around an individual person's workspace. IEEE 802.15 standards specify some technologies used in Bluetooth, ZigBee, and Ultra Wideband. We describe these technologies in Section 2.2.4. Networks without any infrastructure, such as ad hoc and sensor networks, are discussed in Sections 2.2.5 and 2.2.6, respectively.

Finally, in Section 2.3 we discuss briefly various advanced wireless technologies such as OFDM, MIMO, space-time coding, beamforming, and cognitive radio. The motivations for deploying such techniques, the design challenges to maintain basic functionality, and recent developments in real implementation are explained in detail.

2.1 Wireless channel models

2.1.1 Radio propagation

Unlike wired channels that are stationary and predictable, wireless channels are extremely random and hard to analyze. Modeling wireless channels is one of the most challenging tasks encountered in wireless network design. Wireless channel models can be classified as large-scale propagation models and small-scale propagation models, relative to the wavelength.

Large-scale models predict behavior averaged over distances much longer than the wavelength. The models are usually functions of distance and significant environmental features, and roughly frequency-independent. The large-scale models are useful for modelling the range of a radio system and rough capacity planning. Some large-scale theoretical models (the first four) and large-scale experimental models (the rest) are as follows.

• Free-space model

Path loss is a measure of attenuation based only on the distance from the transmitter to the receiver. The free-space model is only valid in the far field and only if there is no interference or obstruction. The received power $P_r(d)$ of the free-space model as a function of distance *d* can be written as

$$P_r(d) = \frac{P_t G_t G_r \lambda^2}{(4\pi)^2 d^2 L},$$
 (2.1)

where P_t is the transmit power, G_t is the transmitter antenna gain, G_r is the receiver antenna gain, λ is the wavelength, and L is the system loss factor not related to propagation. Path-loss models typically define a "close-in" point d_0 and reference other points from this point. The received power in dB form can be written as

$$P_d(d) \, dBm = 10 \log_{10} \left[\frac{P_r(d_0)}{0.001 W} \right] + 20 \log_{10} \left(\frac{d_0}{d} \right).$$
 (2.2)

• Reflection model

Reflection is the change in the direction of a wavefront at an interface between two different media so that the wavefront returns to the medium from which it originated. In the large-scale reflection model, the radio propagation wave impinges on an object which is large compared to wavelength, e.g., the surface of the Earth, a building, or a wall.

The two-ray model is one of the most important reflection models for wireless channels. An example of a reflection in the two-ray model is shown in Fig. 2.1. In the two-ray model the receiving antenna sees a direct-path signal as well as a signal reflected off the ground. Specular reflection, much like light off a mirror, is assumed, and is the case to a very close approximation. The specular reflection arrives with



Fig. 2.2 Knife-edge diffraction model.

strength equal to that of the direct-path signal (i.e., without loss in strength by reflection). The reflected signal shows up with a delay relative to the direct-path signal and, as a consequence, may add constructively (in phase) or destructively (out of phase). The received power of the two-ray model can be written as

$$P_r = P_t G_t G_r \frac{h_t^2 h_r^2}{d^4}, \qquad (2.3)$$

where h_t and h_r are the transmitter height and receiver height, respectively, and *d* is the distance between the two antennas.

Diffraction model

Diffraction occurs when the radio path between transmitter and receiver is obstructed by a surface with sharp, irregular edges. Radio waves bend around the obstacle, even when a line of sight (LOS) does not exist. In Fig. 2.2, we show a knife-edge diffraction model, where the radio wave of the diffraction path from the knife edge and the LOS radio wave are combined at the receiver. As in the reflection model, the radio waves might add constructively or destructively.

Scattering model

Scattering is a general physical process whereby the radio waves are forced to deviate from a straight trajectory by one or more localized non-uniformities in the medium through which they pass. In conventional use, this also includes deviation of reflected radiation from the angle predicted by the law of reflection. The obstructing objects are smaller than the wavelength of the propagation wave, e.g., foliage, street signs, or lamp posts. One scattering example is shown in Fig. 2.3.



Fig. 2.3 Scattering.

• Log-scale propagation model and log-normal shadowing model From experimental measurement, the received signal power decreases logarithmically with distance. However, because of a variety of factors, the decrease in speed is very random. To characterize the mean and variance of this randomness, the log-scale propagation model and log-normal shadowing model are used, respectively.

The log-scale propagation model generalizes path loss to account for other environmental factors. The model chooses a distance d_0 in the far field and measures the path loss $PL(d_0)$. The propagation path-loss factor α indicates the rate at which the path loss increases with distance. The path-loss in the log-scale propagation model is given by

$$PL(d) (dB) = PL(d_0) + 10\alpha \log_{10} \left(\frac{d}{d_0}\right).$$
(2.4)

In the free-space propagation model, the path-loss factor α equals 2.

Shadowing occurs when objects block the LOS between transmitter and receiver. A simple statistical model can account for unpredictable "shadowing" as

$$PL(d) (dB) = PL(d) + X_0, \qquad (2.5)$$

where X_0 is a 0-mean Gaussian random variable with variance typically from 3 to 12. The propagation factor and the variance of log-normal shadowing are usually determined by experimental measurement.

Outdoor-propagation models

In the outdoor models, the terrain profile of a particular area needs to be taken into account in estimating the path loss. Most of the following models are based on a systematic interpretation of measurement data obtained in the service area. Some typical outdoor-propagation models are the Longley–Rice model, the ITU terrain model, the Durkin's model, the Okumura model, the Hata's model, the PCS extension of the Hata model, the Walfisch and Bertoni model, and the wideband PCS microcell model [397].

Indoor-propagation models

For indoor applications, the distances are much shorter than those in the outdoor models. The variability of the environment is much greater, and key variables are the layout of the building, construction materials, building type, and antenna location. In general, indoor channels may be classified either as LOS or obstruction with varying degrees of clutter. The losses between floors of a building are determined by the external dimensions and the materials of the building, as well as the type of construction used to create the floors and the external surroundings. Some available indoor propagation models are the Ericsson multiple breakpoint model, the ITU model for indoor attenuation, the log distance path-loss model, the attenuation factor model, and the Devasirvatham's model.

Small-scale (fading) models describe signal variability on a scale of wavelengths. In fading, multi-path and Doppler effects dominate. Fading is frequency-dependent and time-variant. The focus is on modelling fading, the rapid change in signal strength over a short distance or time.

Multi-path fading is caused by interference between two or more versions of the transmitted signal, which arrive at slightly different times. Multi-path fading causes rapid changes in signal strength over a small travel distance or time interval, random frequency modulation due to varying Doppler shifts on different multi-path signals, and time dispersion resulting from propagation delays.

To measure the time dispersion of multiple paths, the power delay profile and the root mean square (RMS) are the most important parameters. Power delay profiles are generally represented as plots of relative received power as a function of excess delay with respect to a fixed time delay reference. The mean excess delay is the first moment of the power delay profile and is defined as $\bar{\tau} = \frac{\sum_{k} a_{k}^{2} \tau_{k}}{\sum_{k} a_{k}^{2}}$, where τ_{k} is the delay of the *k*th multi-path and a_{k} is its corresponding amplitude. The RMS is the square root of the second central moment of the power delay profile, defined as $\sigma_{\tau} = \sqrt{\tau^{2} - (\tau)^{2}}$, where $\bar{\tau}^{2} = \frac{\sum_{k} a_{k}^{2} \tau_{k}^{2}}{\sum_{k} a_{k}^{2}}$. Typical values of RMS delay spread are on the order of microseconds in outdoor mobile radio channels and on the order of nanoseconds in indoor radio channels.

Analogous to the delay spread parameters in the time domain, coherent bandwidth is used to characterize the channel in the frequency domain. Coherent bandwidth is the range of frequencies over which two frequency components have a strong potential for amplitude correlation. If the frequency correlation between two multi-paths is above 0.9, then the coherent bandwidth is $B_c = \frac{1}{50\sigma}$ [397]. If the correlation is above 0.5, the coherent bandwidth is $B_c = \frac{1}{5\sigma}$. Coherent bandwidth is a statistical measure of the range of frequencies over which the channel can be considered flat.

Delay spread and coherent bandwidth describe the time-dispersive nature of the channel in a local area, but they do not offer information about the time-varying nature of the channel caused by relative motion of transmitter and receiver. Next, we define Doppler spread and coherence time, which describe the time-varying nature of the channel in a small-scale region.
Doppler frequency shift is caused by movement of the mobile users. The frequency shift is positive when a mobile user moves toward the source; otherwise, the frequency shift is negative. In a multi-path environment, the frequency shift for each ray may be different, leading to a spread of received frequencies. Doppler spread is defined as the maximum Doppler shift $f_m = \frac{v}{\lambda}$, where v is the mobile user's speed and λ is the wavelength. If we assume that signals arrive from all angles in the horizontal plane, the Doppler spectrum can be modelled as Clarke's model [397].

Coherence time is the time duration over which the channel impulse response is essentially invariant. Coherence time is defined as $T_c = \frac{C}{f_m}$, where *C* is a constant [397]. This definition of coherence time implies that two signals arriving with a time separation greater than T_c are affected differently by the channel. If the symbol period of the baseband signal (the reciprocal of the baseband signal bandwidth) is greater than the coherence time, then the signal will distort, since the channel will change during the transmission of the signal.

Based on the transmit signal's bandwidth and symbol period relative to the multipath RMS and coherent bandwidth, the small-scale fading can be classified as either flat fading or frequency-selective fading. This classification means that the band-limited transmit signal sees a flat-frequency channel or a frequency-selective channel. Based on coherence time due to Doppler spread, the small-scale fading can be classified as fast fading or slow fading. This classification is according to whether the channel changes during each signal symbol. The details are shown in Fig. 2.4.

Multi-path and Doppler effects describe the time and frequency characteristics of wireless channels. But further analysis is necessary for statistical characterization of the amplitudes. Rayleigh distributions describe the received signal envelope distribution for channels, where all the components are non-LOS. Ricean distributions describe the



Fig. 2.4 Classification of small-scale fading.

received signal envelope distribution for channels where one of the multi-path components is the LOS component. Nakagami distributions are used to model dense scatterers, and can be reduced to Rayleigh distributions. But they provide more control over the extent of the fading.

2.1.2 Interference channel

Since networks accommodate an increasing number of users and bandwidth is limited, radio frequencies are reused beyond a certain distance, which leads to co-channel interference. In this subsection, we study the interference channel. The system model for an interference channel is shown in Fig. 2.5. The received signal vector **y** can be written as

$$\mathbf{y} = \mathbf{G}\mathbf{x} + \mathbf{z},\tag{2.6}$$

where **x** is the transmitted signal vector, **z** is the noise vector, and **G** is the channel gain matrix with elements $G_{k,n}$. Here k is the transmitter index and n is the receiver index.

For an interference channel, the interference from other users is generally considered as noise. This assumption leads to optimal rates for weak and medium interference. So instead of simply using SNR (signal-to-noise ratio, given by $\frac{P_k G_{k,k}}{\sigma^2}$), we consider the SINR (signal-to-interference-and-noise ratio) to calculate the capacity of the network. Therefore R_k , the capacity of user k, is given by

$$R_{k} = \log_{2} \left(1 + \frac{P_{k}G_{k,k}}{\sum_{i \neq k}^{n} P_{i}G_{i,k} + \sigma^{2}} \right), \qquad (2.7)$$

where P_k is the transmit power of the *k*th user, $G_{i,k}$ is the channel gain from user *i* to base station *k*, and the term $\sum_{i \neq k} P_i G_{i,k}$ represents the interference caused by other



users to user k. Without loss of generality, we consider the variance of additive Gaussian noise as a constant σ^2 for all subcarriers. The spectrum management problem defines the objective of the network and the various constraints that are to be applied depending on the network capabilities. One sample spectrum management problem has the following objective and limitations:

- Objective: to maximize the overall rate of the network
- Constraints: limited transmit power to achieve the minimum data rate while causing least interference to other users.

Mathematically, by defining w_k as the weight factor, the problem can be expressed as

$$\max_{\substack{P_k^n \ge 0, \forall n, k}} \sum_{k=1}^K w_k \sum_{n=1}^N \log_2 \left(1 + \frac{P_k^n G_{k,k}^n}{\sum_{i \ne k} P_i^n G_{i,k}^n + \sigma^2} \right) \qquad \text{s.t.} \sum_n P_k^n \le P_k^{\max}.$$
(2.8)

The capacity region of an interference channel is still an open problem. Once the goals of the network have been tied down, there are various algorithms proposed in the literature (such as iterative water-filling [523], optimal spectral balancing [95], iterative spectral balancing [94], SCALE [381], autonomous spectral balancing [93], and band preference [509, 110]), which try to achieve the largest capacity region possible while adhering to the constraints of maximum transmitter power and minimum target rate of each user.

2.2 Categorization of wireless networks

We list various standards in Figs. 2.6 and 2.7 for different communication rates and different communication ranges. These standards will fit the different needs of various applications. We will discuss techniques that can utilize multiple standards in different situations, so that connections can be made anytime and anywhere. In the following, we categorize wireless networks and provide some specifics.

2.2.1 3G cellular networks and beyond

Third-generation (3G) mobile communication systems based on the wideband codedivision multiple-access (WCDMA) and CDMA2000 radio access technologies have seen widespread deployment around the world. The applications supported by these commercial systems range from circuit-switched services such as voice and video telephony to packet-switched services such as video streaming, email, and file transfer. As more packet-based applications are developed and put into service, the need increases for better support for different quality-of-service (QoS) level, higher spectral efficiency, and higher data rate for packet-switched services, in order to further enhance user experience while maintaining efficient use of system resources. This has resulted in the evolution of 3G standards, as shown in Fig. 2.8. For 3G cellular systems, there are



Fig. 2.6 Standards comparison.



Fig. 2.7 Comparison of wireless networks.



Fig. 2.8 Evolution of 2G to 3G cellular networks.

two camps: 3G Partnership Project (3GPP) [5] and 3G Partnership Project 2 (3GPP2) [2], which are based on different 2G technologies.

The development of 3G will follow a few key trends, and the evolution following these trends will continue as long as physical limitations or backward compatibility requirements do not force the development to move from evolution to revolution. The key trends include:

- Voice services will continue to be important in the foreseeable future, which means that capacity optimization for voice services will continue.
- Along with increasing use of IP-based applications, the importance of data as well as simultaneous voice and data will increase.
- Increased need for data means that the efficiency of data services needs to be improved.
- When more and more attractive multimedia terminals emerge on the market, the use of such terminals will spread from office, homes, and airports to roads, and finally everywhere. This means that high-quality, high-data-rate applications will be needed everywhere.
- When the volume of data increases, the cost per transmitted bit needs to decrease in order to make new services and applications affordable for everybody.
- The other current trend is that in the 3G evolution path very high data rates are achieved in hot spots with WLAN rather than cellular-based standards.

CDMA2000

The CDMA (code-divison multiple-access) family of cellular networks grew out of work undertaken by Qualcomm, a California-based company. Working on direct sequence spread spectrum (DSSS) techniques, by using different spreading codes, a large number of users could occupy the same channel at the same time, which could provide a multiple access scheme for cellular telecommunications. The first standard was the IS-95 and the first network was launched in Hong Kong in 1996 under the brand name CDMAOne.

The CDMA system has the following standards in its developmental stage, as shown in Fig. 2.8: IS-95, IS-95A, IS-95B, CDMA2000 (1x/EV-DO, 1xEV-DV, 1xRTT, and 3xRTT). The first version of the system standard IS-95 was never launched for commercial purposes because of its prematurity. IS-95A was applied for business since then and is still used widely nowadays. IS-95B was a short version since the CDMA2000 standard was announced six months after it came out. The original IS-95A standard only allowed for circuit-switched data at 14.4 kbit s⁻¹, and IS-95B provided up to 64 kbit s⁻¹ data rates as well as a number of additional services. A major step improvement came later with the development of 3G services. The first 3G standard was known as CDMA2000 1x, which initially provided data rates up to 144 kbit s⁻¹. With further development, the systems promise to allow a maximum data rate of 307 kbits s⁻¹.

WCDMA/UMTS

WCDMA was developed by NTT DoCoMo as the air interface for their 3G network, FOMA. Later NTT DoCoMo submitted the specification to the International Telecommunication Union (ITU) as a candidate for the international 3G standard known as IMT-2000. The ITU eventually accepted WCDMA as part of the IMT-2000 family of 3G standards, as an alternative to CDMA2000, EDGE, and the short-range DECT (digital enhanced cordless telecommunications) system. Later, WCDMA was selected as the air interface for Universal Mobile Telecommunications System (UMTS), the 3G successor to GSM.

TD-SCDMA

Transmit diversity (TD) is one of the key contributing technologies to the ITU-endorsed 3G systems WCDMA and CDMA2000. Spatial diversity is introduced into the signal by transmitting through multiple antennas. The antennas are spaced far enough apart¹ that the signals emanating from them can be assumed to undergo independent fading. In addition to diversity gain, antenna gain can also be incorporated through channel-state feedback. This leads to the categorization of TD methods into open-loop and closed-loop methods. Several methods of transmit diversity in the forward link have been either under consideration or adopted for the various 3G standards.

4G and beyond

Looking at development in the Internet and applications, it is clear that the complexity of transferred content is rapidly increasing and will increase further in the future. Generally,

¹ A typical spacing is half the wavelength.

	3G	4G
Major requirement driving architecture	Voice-driven data add-on	Data/voice over IP
Network architecture	Wide area cell–based	Hybrid with WiFi and WPAN
Speed	384 kbps–2 Mbps	20–10 Mbps
Frequency band	1.8–2.4 GHz	2–8 GHz
Bandwidth	1.25, 5, 20 MHz	100 MHz
Switching design	Circuit and packet	Packet
Access	DS-CDMA	OFDM/MC-CDMA
FEC	Convolution/turbo code	Concatenated coding
Component design	Antenna, multi-band adapter	Smart antennas, software radios

Table 2.1 Comparison of 3G and 4G.

it can be said that the more bandwidth is available, the more bandwidth applications will consume. In order to justify a new air interface, goals need to be set high enough to ensure that the system will be able to serve us long into the future. A reasonable approach would be to aim at 100 Mb s^{-1} full-mobility wide area coverage and 1 Gb s^{-1} low-mobility local area coverage with a next-generation cellular system. Also, future application and service requirements will bring new requirements to the air interface and new emphasis on air interface design. One such issue, which already strongly impacts the 3G revolution, is the need to support IP and IP-based multimedia. If both the technology and the spectrum to meet such requirements cannot be found, the whole discussion of 4G may become obsolete. In Table 2.1, we compare key parameters of 4G and 3G. There are also some pre-4G systems such as Long-term evolution (LTE), WiMAX, Ultra Mobile Broadband (UMB, formerly EV-DO rev. C) and Flash-OFDM, as shown in Fig. 2.8.

2.2.2 WiMAX networks

Wireless metropolitan area network (WMAN) technology is a relatively new field that was started in 1998. From that time a new standard has emerged to handle its implementation, IEEE 802.16. The equivalent of 802.16 in Europe is HIPERMAN. The WiMAX Forum is working to ensure that 802.16 and HIPERMAN interoperate seamlessly. This standard has helped to pave the way for WMAN technology globally and since its inception has received six expansions onto the standards. WMAN differs from other wireless technologies in that it is designed for a broader audience, such as a large corporation or an entire city.

The 802.16 MAC uses a scheduling algorithm for which the subscriber station need compete only once (for initial entry into the network). After the competition, the subscriber station is allocated an access slot by the base station. The time slot can enlarge or contract, but remains assigned to the subscriber station, which means that other subscribers cannot use it. The 802.16 scheduling algorithm is stable under overload and



Fig. 2.9 WiMAX protocol stacks.

Table 2.2	Comparison	of 802.16	standards
-----------	------------	-----------	-----------

	802.16	802.16a/802.16d	802.16e
Date	Dec. 2001	Jan. 2003/Q3, 2004	Q3, 2004
Spectrum	10-66 GHz	<11 GHz	<6 GHz
Channels	Line-of-sight only	Non-line-of-sight	Non-line-of-sight
Modulation	QPSK,16QAM, 64QAM	OFDM256, QPSK, 16QAM, 64QAM	Same as 802.11a
Mobility	Fixed	Fixed	Pedestrian mobility regional roaming
Bandwidth	20, 25, 28 MHz	1.25–20 MHz	Same as 802.16a
Throughput	Up to 75 Mbps	Up to 75 Mbps	Up to 30 Mbps
Cell radius	1–3 miles	3–5 miles	1–3 miles

over-subscription (unlike 802.11). It can also be more bandwidth-efficient. The scheduling algorithm also allows the base station to control QoS parameters by balancing the time-slot assignments among the application needs of the subscriber stations. Moreover, the MAC layer is also in charge of protocol data unit (PDU) assembly and de-assembly. A detailed illustration for different layers' protocols of 802.16 is shown in Fig. 2.9.

The operation standards for WMANs are regulated under IEEE Standard 802.16 [10]. WMANs are allowed the operating frequency range 10–66 GHz. With such a broad spectrum to work with, WMANs have the ability to transmit over previous wireless frequencies such as IEEE 802.11b/g, causing less interference with other wireless products. The only downside to using such high frequencies is that WMAN needs a line of sight between transmitters and receivers, much like a directional antenna. Using line of sight, however, will decrease multi-path distortion, allowing higher bandwidths to be achieved, and can attain up to 75 Mbps for both uplink and downlink on a single channel [487]. Some extensions of 802.16 standards are listed below and in Table 2.2.

• IEEE 802.16a: The IEEE has developed 802.16a for use at licensed and license-exempt frequencies from 2 to 11 GHz. Most commercial interest in IEEE 802.16 is in these

lower frequency ranges. At the lower ranges, the signals can penetrate barriers and thus do not require a line of sight between transceiver and antenna. This enables more flexible WiMax implementation while maintaining the technology's data rate and transmission range. IEEE 802.16a supports mesh deployment, in which transceivers can pass a single communication on to other transceivers, thereby extending the basic 802.16 transmission range.

- IEEE 802.16b: This extension increases the spectrum the technology can use in the 5 and 6 GHz frequency bands, and improves quality of service. WiMax provides QoS to ensure priority transmission for real-time voice and video and offers differentiated service levels for different traffic types.
- IEEE 802.16c: IEEE 802.16c represents a 10 to 66 GHz system profile that standardizes more details of the technology. This encourages more consistent implementation and, therefore, interoperability.
- IEEE 802.16d: IEEE 802.16d includes minor improvements and fixes for 802.16a. This extension also creates system profiles for compliance testing of 802.16a devices.
- IEEE 802.16e: This technology will standardize networking between carriers' base stations and mobile devices, rather than just between base stations and fixed recipients. IEEE 802.16e would enable the high-speed signal handoffs necessary for communication with users moving at vehicular speeds.

In addition to IEEE 802.16, the Mobile Broadband Wireless Access (MBWA) working group aims to prepare a formal specification for a packet-based air interface designed for IP-based services. The goal is to create an interface that will allow the creation of low-cost, always-on, and truly mobile broadband wireless networks, nicknamed "Mobile-Fi". IEEE 802.20 will be specified according to a layered architecture, which is consistent with other IEEE 802 specifications. The scope of the working group consists of the physical (PHY), medium access control (MAC), and logical link control (LLC) layers. The air interface will operate in bands below 3.5 GHz and with a peak data rate of over 1 Mbit s⁻¹. The goals of 802.20 and 802.16e, the so-called "mobile WiMAX," are similar.

WiMAX can be viewed as "last mile" connectivity at high data rates. This could result in lower pricing for both home and business customers as competition lowers prices. In areas without pre-existing physical cable or telephone networks, WiMAX may be a viable alternative for broadband access that has been economically unavailable. Prior to WiMAX, many operators used proprietary fixed wireless technologies for broadband services. For this reason, WiMAX has significant market in rural areas and developing countries.

2.2.3 WiFi networks

IEEE 802.11 denotes a set of Wireless Local Area Network (WLAN) standards developed by working group 11 of the IEEE LAN/MAN Standards Committee (IEEE 802). WiFi is a brand originally licensed by the WiFi Alliance to describe the underlying technology of WLAN based on IEEE 802.11 specifications. It was developed for mobile computing devices such as laptops in LANs, but is now increasingly used for other services, including Internet and voice over IP (VoIP) phone access, gaming, and basic connectivity of consumer electronics such as televisions, DVD players, and digital cameras.

In the physical layer, 802.11b operates within the 2.4 GHz industrial, scientific, and medical (ISM) band. The original 802.11b defines data rates of 1 Mbps and 2 Mbps via radio waves using frequency-hopping spread spectrum (FHSS) or direct sequence spread spectrum (DSSS). For FHSS, 2.4 GHz band is divided into 75 1-MHz subchannels. The sender and receiver agree on a hopping pattern, and data is sent over a sequence of the subchannels. Each conversation within the 802.11 network occurs over a different hopping pattern. Because of Federal Communications Commission (FCC) regulations that restrict subchannel bandwidth to 1 MHz, FHSS techniques are limited to speeds of no higher than 2 Mbps. DSSS divides the 2.4 GHz band into 14 22-MHz channels. Adjacent channels overlap one another partially, with 3 of the 14 being completely non-overlapping. The spreading code is an 11-bit Barker sequence. Binary phase shift keying (BPSK) and quadrature phase shift keying (QPSK) are used to provide different rates.

To increase the data rate to 5.5 Mbps and 11 Mpbs in the 802.11b standard, an advanced coding technique, complementary code keying (CCK) is employed. A complementary code contains a pair of finite bit sequences of equal length, such that the number of pairs of identical elements (1 or 0) with any given separation in one sequence is equal to the number of pairs of unlike elements having the same separation in the other sequence. A network using CCK can transfer more data per unit time for a given signal bandwidth than a network using the Barker code, because CCK makes more efficient use of the bit sequences. CCK consists of a set of 64 8-bit code words. The 5.5 Mbps rate uses CCK to encode 4 bits per carrier, while the 11 Mbps rate encodes 8 bits per carrier. Both speeds use QPSK as the modulation technique and signal at 1.375 MSps. Table 2.3 shows the differences rates for 802.11b.

Standard 802.11a adopts orthogonal frequency-division multiplexing (OFDM) at 5.15–5.25 GHz, 5.25–5.35 GHz, and 5.725–5.825 GHz to support multiple data rates up to 54 Mbps. 802.11g utilizes the 2.4 GHz band with OFDM modulation and is also backward-compatible with 802.11b. For OFDM, the FFT (fast Fourier transform) has 64 subcarriers. There are 48 data subcarriers and 4 carrier pilot subcarriers for a total of 52 nonzero subcarriers defined in IEEE 802.11a, plus 12 guard subcarriers. The IEEE 802.11a/g physical layer provides eight PHY modes with different modulation schemes and different convolutional coding rates, and can offer various data rates.

To achieve higher data rates in the PHY layer, in January 2004 IEEE announced that it had formed a new 802.11 task group (TGn) to develop an amendment to the 802.11 standards for wireless local area networks. 802.11n builds upon previous 802.11 standards by adding MIMO (multiple-input multiple-output). MIMO uses multiple transmitter and receiver antennas to allow for increased data throughput through spatial multiplexing and increased range by exploiting the spatial diversity. There are several proposal groups, referred to as TGnSync, WWiSE (short for "World-Wide Spectrum Efficiency"), and MITMOT ("Mac and mImo Technologies for MOre Throughput"). All proposals occupy the frequency band 2.5 GHz with 20 MHz or 40 MHz bandwidth so as to support

Data rate	Code length	Modulation	Symbol rate	Bits/symbol
1 Mbps	11 (DSSS)	BPSK	1 MSps	1
2 Mbps	11 (DSSS)	QPSK	1 MSps	2
5.5 Mbps	8 (CCK)	QPSK	1.375 MSps	4
11 Mbps	8 (CCK)	QPSK	1.375 MSps	8

Table 2.3 802.11b rates.

Table 2.4 Comparison of 802.11 standards.

	802.11b	802.11a/g	802.11n
Air rate	11 Mbps	54 Mbps	200+ Mbps
MAC SAP rate	5 Mbps	25 Mbps	100 Mbps
Range	30 m	30 m	50 m
Frequency	2.4G	5.25,5.6,5.8G/2.4G	2.4G
Bandwidth	20M	20M	20M or 40M
Modulation	DSSS/CCK	DSSS/CCK/OFDM	DSSS/CCK/OFDM with MIMO
Special streams	1	1	1,2,3,4

the communication speed, more than 200 Mbps. 802.11n is backward-compatible with 802.11b and 802.11g. In Table 2.4, we compare various parameters of the three 802.11 standards.

The IEEE 802.11 MAC protocol supports two kinds of access methods, namely, distributed coordination function (DCF) and point coordination function (PCF). In both mechanisms, only one user occupies all the bandwidth at each time slot. PCF is based on polling, controlled by a point coordinator like Access Point, to communicate with a node listening and to see if the airwaves are free. PCF seems to be implemented only in very few hardware devices as it is not part of the WiFi Alliance's interoperability standard.

In contrast, DCF is an access mechanism using carrier-sense multiple-access with collision avoidance (CSMA/CA). DCF mandates that a station wishing to transmit must listen for the channel status for a DCF interframe space (DIFS) interval. If the channel is found to be busy during the DIFS interval, the station defers its transmission or proceeds otherwise. In a network where a number of stations contend for the multi-access channel, if multiple stations sense that the channel is busy and defer their access, they will find that the channel is released virtually simultaneously and will then try to seize the channel again at the same time. As a result, collisions may occur. In order to avoid such collisions, DCF also specifies random backoff, which forces a station to defer its access to the channel for an extra period. DCF also has an optional virtual carrier-sense mechanism that exchanges short request-to-send (RTS) and clear-to-send (CTS) frames between the source and destination stations before the long data frame is transmitted. The details of RTS/CTS will be presented in later chapters.

In order to take full advantage of the future market opportunity for WiFi, several key challenges must be overcome. In the following, we list some near-future design topics and their possible solutions.

• *Security*. The greatest concentration of WiFi is on free public access. However, eavesdroppers and hackers can take full advantage of the WiFi system. Currently, all 802.11a, b, and g devices support WEP (Wired Equivalent privacy) encryption, which has had flaws.

IEEE 802.11i, also known as WiFi Protected Access 2 (WPA2), is an amendment to the 802.11 standard specifying security mechanisms for wireless networks. The 802.11i specification defines two classes of security algorithms: Robust Security Network Association (RSNA) and Pre-RSNA. Pre-RSNA security consists of WEP and 802.11 entity authentication. RSNA provides two data confidentiality protocols, called the Temporal Key Integrity Protocol (TKIP) and the Counter-mode/CBC-MAC Protocol (CCMP). The RSNA establishment procedure includes 802.1X authentication and key management protocols. Beyond IEEE 802.11i, it is worth mentioning that WAPI (WLAN Authentication and Privacy Infrastructure) is a Chinese National Standard for wireless LAN (GB 15629.11-2003).

- *Mobility*. Mobility is an important attribute of wireless networks. Current wireless LAN standards provide mobility through roaming capabilities. IEEE 802.11p, also referred to as Wireless Access for the Vehicular Environment (WAVE), defines enhancements to 802.11 required to support Intelligent Transportation Systems (ITS) applications. This includes data exchange between high-speed vehicles and between vehicles and roadside infrastructure in the licensed ITS band of 5.9 GHz (5.85–5.925 GHz). 802.11p will be used as the groundwork for DSRC (Dedicated Short Range Communications), a US Department of Transportation project which will be emulated elsewhere, looking at vehicle-based communication networks, particularly for applications such as toll collection, vehicle safety services, and commerce transactions via cars. The ultimate vision is a nationwide network that enables communications between vehicles and roadside access points or other vehicles.
- **QoS support.** 802.11e is the first wireless standard that spans home and business environments. It adds quality-of-service (QoS) features and multimedia support to the existing 802.11 wireless standards, while maintaining full backward compatibility with these standards. QoS and multimedia support are critical to wireless home networks where voice, video, and audio will be delivered. Broadband service providers view QoS and multimedia-capable home networks as essential ingredients in offering residential customers video on demand, audio on demand, VoIP, and high-speed Internet access.

802.11e introduces two enhancements, Enhanced DCF (EDCF) and Hybrid Coordination Function (HCF). In EDCF, a station with high-priority traffic waits a little less, on average, before it sends its packet than a station with low-priority traffic, so that high-priority traffic has a higher chance of being sent than low-priority traffic. In addition, each priority level is assigned a Transmit Opportunity (TXOP), which is a bounded time interval during which a station can send as many frames as possible. HCF works more like PCF. With the PCF, QoS can be configured with great precision. QoS-enabled stations have the ability to request specific transmission parameters (data rate, jitter, etc.) which should allow advanced applications like VoIP and video streaming to work more effectively on a WiFi network.

• Integration of 3G and WLAN. The third-generation cellular networks and 802.11 local area wireless networks possess complementary characteristics. 3G cellular networks promise to offer always-on, ubiquitous connectivity and mobility with relatively low data rates. 802.11 offers much higher data rates, comparable to the cellular networks, but can cover only smaller areas without mobility, suitable for hot-spot applications in hotels and airports. The performance and flexibility of wireless data services would be dramatically improved if users could seamlessly roam across the two networks. By offering integrated 802.11/3G services, 3G operators and wireless Internet service providers (WISPs) can attract a wider user base and ultimately facilitate the ubiquitous introduction of high-speed wireless services. Users can also benefit from the enhanced performance and lower overall cost of such a combined service. For a network node changing the type of connectivity between 3G cellular phone and WLAN, the concept of vertical handoff will be discussed in later chapters.

2.2.4 Wireless personal area networks

A wireless personal area network (WPAN) is a computer network used for wireless communication among devices (including telephones and personal digital assistants) close to a person. The reach of a WPAN is typically a few meters. WPANs can be used for communication among the personal devices themselves (intrapersonal communication), or for connecting to a higher-level network or the Internet (an uplink). 802.15 is a communications specification that was approved in early 2002 by the IEEE Standards Association (IEEE-SA) for WPANs. Specifically, we list the following three sub-standards:

- The IEEE Standard 802.15.1 was approved as a new standard for Bluetooth by the IEEE-SA Standards Board on 15 April 2002. The Bluetooth standard enables wireless communication between multiple electronic devices within 10 m of each other. Bluetooth devices are organized in piconets, which include one master device and up to seven slave devices. Bluetooth devices communicate in the 2.4 GHz radio frequency band, enabling devices to communicate without line-of-sight spacing, such as through walls or through a person's body. Bluetooth piconets utilize frequencyhopping spread spectrum in 79 1-MHz bands, reducing the likelihood of interference with other Bluetooth piconets.
- 802.15.3 is the IEEE standard for high-data-rate WPAN designed to provide quality of service (QoS) for real-time distribution of multimedia content, like video and music. It is ideally suited for a home multimedia wireless network. The original standard uses a "traditional" carrier-based 2.4 GHz radio as the physical layer (PHY). A follow-on standard, 802.15.3a, defines an alternative PHY, based on UWB, that will provide in excess of 110 Mbps at 10 m and 480 Mbps at 2 m. This will allow applications requiring streaming of high-definition video between media servers and flat-screen

HD monitors, and extremely fast transfer of media files between media servers and portable media devices.

• IEEE 802.15.4-2003 (Low-Rate WPAN) deals with low data rates but very long battery life (months or even years) and very low complexity. The first edition of the 802.15.4 standard was released in May 2003. In March 2004, after forming task group 4b, task group 4 put itself in hibernation. The ZigBee set of high-level communication protocols is based upon the specification produced by the IEEE 802.15.4 task group.

Bluetooth/ZigBee

Bluetooth is a standard for wireless communications that use short-range radio frequencies to enable communications among multiple electronic devices. Bluetooth technology is envisioned as a replacement for the interconnection cables between personal devices such as notebook computers, cellular phones, personal digital assistants, and digital cameras. If widely adopted, Bluetooth would enable a uniform interface for accessing data services. Thus, calendars, address books, and business cards stored in personal devices could be automatically synchronized using push-button synchronization and proximity operation.

The Bluetooth Special Interest Group (SIG) was founded by Ericsson, IBM, Intel, Nokia, and Toshiba in February 1998 (and later joined by many other companies as Associate or Adopter Members) to develop an open specification for short-range wireless connectivity. The SIG offered all of the intellectual property explicitly included in the Bluetooth specification royalty-free to adopter members, to facilitate the widespread acceptance of the technology. The SIG now includes thousands of companies. To use the intellectual property in the Bluetooth specification, Adopter Members must qualify any Bluetooth products they intend to bring to market through the Bluetooth qualification program. The Bluetooth qualification program includes radio and protocol conformance testing, profile conformance testing, and interoperability testing. Bluetooth is also known as IEEE 802.15.1.

Bluetooth is a radio frequency technology utilizing the unlicensed 2.4 GHz ISM band. Bluetooth enables wireless connections up to 10 m under standard transmitter power, and owing to the use of radio frequencies, devices need not be within line of sight of each other and may connect through walls or other non-metal objects. In active mode, Bluetooth devices typically consume 0.1 W of active power for class 1 with range of 100 m, 2.5 mW for class 2 with range 10 m, and 1 mW for class 3 with range 1 m. The modulation technique utilized in Bluetooth technology is binary Gaussian frequency shift keying, and the baud rate is 1 Msymbol s⁻¹. Thus, the bit time is 1 μ s and the raw transmission speed is 1 Mb s⁻¹.

The baseband signals used in Bluetooth devices, which are typically 1 MHz in bandwidth, cannot be transmitted directly on the wireless medium. Modulation of the 1 MHz baseband signals into the 2.4 GHz band is difficult to achieve in one step because CMOS transistors do not operate at these frequencies. Bluetooth radio devices solve this problem by modulating the baseband signal onto an intermediate frequency, such as 3 MHz, and then using a frequency mixer to increase the frequency of the signal to the 2.4 GHz band. Because the unlicensed ISM band in which Bluetooth operates is often cluttered with signals from other devices, such as garage-door openers, baby monitors, cordless phones, and microwave ovens, Bluetooth utilizes frequency-hopping spread spectrum for security and to avoid interference with the signals from other devices. Frequency hopping also allows multiple piconets to exist within range of each other with minimal interference. Frequency hopping typically involves generating a frequency shift keyed signal, and then shifting the frequency of the frequency shift keyed signal by an amount determined by a pseudonoise code. The pseudonoise code is random in that it appears to be unpredictable to an outsider, but it is generated by deterministic means. The pseudonoise code is unique to the piconet, and is determined by the master device.

Bluetooth utilizes a slow hopping scheme, hopping in a pseudo-random fashion through 79 1-MHz channels. The frequency channels are located at (2402+k) MHz, with k = 0, 1, ..., 78. A Bluetooth piconet hops through 1600 different frequencies per second. Each frequency hop corresponds to one slot, with each slot lasting $1/1600 \text{ s} = 625 \mu \text{s}$. Each packet may be one, three, or five slots long. A frame consists of two packets, one packet being a transmitted packet and the other a received packet.

A packet consists of an access code, a header, and a payload. The access code is 72 bits long and is used for clock synchronization, DC offset compensation, identification, and signaling. The header is 54 bits long and is used for addressing, identifying the packet type, controlling flow, sequencing to filter retransmitted packets, and verifying header integrity (ensuring that the header was not altered by another source). The payload is between zero and 2744 bits, depending on the type of packet. In packets that are one slot long, the payload is 240 bits long. In three-slot-long packets, the payload is 1500 bits long. In packets that are five slots long, the payload is 2744 bits long.

Each Bluetooth device includes a unique IEEE-type 48-bit address, called a Bluetooth device address, assigned to it at manufacture, and a 28-bit clock. The clock ticks once every 312.5 μ s, which corresponds to half the residence time in a frequency band when the radio hops at the rate of 1600 hops per second.

Bluetooth devices in communication with each other are organized into groups of two to eight devices called piconets, as shown in Fig. 2.10. A piconet consists of a single master device and between one and seven slave devices. A device may belong to more than one piconet, but may be the master in no more than one piconet; thus, a device may be a slave in two piconets or a master in one piconet and a slave in another piconet.

The slaves utilize the Bluetooth clock of the master to maintain time synchronization. The pseudo-random hopping sequence is determined by the 48-bit Bluetooth device address of the master. The Bluetooth clock of the master clock determines the phase in the hopping pattern, thereby determining the particular frequency to be used at a particular time slot. Thus, the communication channel in a particular piconet is fully identified by the master, and this communication channel serves to distinguish one piconet from another.

The master and the slaves alternate transmit opportunities according to a time-division duplexing scheme. According to this scheme, the master transmits on even-numbered time slots, as defined by the master's Bluetooth clock, while the slaves transmit on





odd-numbered slots. A given slave may transmit only if the master has just transmitted to this slave.

In order to determine the presence and identities of other Bluetooth devices, Bluetooth devices engage in inquiry and page processes. The inquiry process is performed without knowledge of the identity or presence of other Bluetooth devices, whereas the paging process is performed with knowledge of the identity and presence of other Bluetooth devices.

During an inquiry process, a prospective master device makes its presence known by transmitting inquiry messages. Devices that are searching for inquiry messages respond with inquiry messages that contain their Bluetooth device addresses. After the master has acquired knowledge of the Bluetooth device address and presence of other Bluetooth devices within range, the master explicitly pages the other Bluetooth devices to join its piconet. Devices responding to the page will provide additional information, such as their clock phases, to the master.

Bluetooth devices have three low-power modes in which they reside when they are not in active communication. In sniff mode, a slave agrees with its master to listen for master transmissions periodically. In hold mode, a device agrees with another device in the piconet to remain silent for a given amount of time. A device that has gone into hold mode does not relinquish its temporary member address within the piconet. In park mode, a slave agrees with its master to park until further notice. In park mode, the slave device relinquishes its temporary member address within the piconet, and periodically listens to transmissions from the master. The slave may be invited back to active communications by the master, or may send a request to the master to be unparked.

Bluetooth devices typically provide link layer security between any two Bluetooth radios. A challenge/response system, such as an E1 algorithm [1], is used for authentication. The authentication is based on a link key, which is a 128-bit shared secret between the two Bluetooth devices. The link key is generated by a challenge-and-response process

between the two Bluetooth devices. Data sent between two Bluetooth devices may also be encrypted, and may be ciphered with an E0 algorithm [1]. An encryption key may be between 8 and 128 bits long, and may be derived from the link key. The Bluetooth devices may use a configuration encryption key zero to 16 bytes in length for key management and usage. The authentication and encryption keys may be generated with E2-E3 algorithms [1].

For WPANs, besides Bluetooth technology, ZigBee is a specification for a suite of high-level communication protocols using small, low-power digital radios based on the IEEE 802.15.4 standard for WPANs. ZigBee operates in the ISM radio bands: 868 MHz in Europe, 915 MHz in the USA and 2.4 GHz in most jurisdictions worldwide. The technology is intended to be simpler and cheaper than other WPANs such as Bluetooth. The specification supports data transmission rates of up to 250 kbps at a range of up to 30 m. ZigBee's technology is slower than 802.11b (11 Mbps) and Bluetooth (1 Mbps) but it consumes significantly less power.

Ultra wideband

Ultra wideband (UWB) is a technology for transmitting information spread over a large bandwidth (>500 MHz) that is able to share spectrum with other users. In 2002, the FCC authorized the unlicensed use of UWB in the 3.1–10.6 GHz band. The intention is to provide an efficient use of scarce radio bandwidth while enabling high-data-rate personal area network (PAN) wireless connectivity. Deliberations in the International Telecommunication Union Radiocommunication Sector (ITU-R) resulted in a Report and Recommendation on UWB in November of 2005.

The FCC power spectral density emission limit for UWB emitters operating in the UWB band is $-41.3 \text{ dBm MHz}^{-1}$. This is the same limit that applies to unintentional emitters in the UWB band, the so-called Part 15 Limit [12]. However, the emission limit for UWB emitters can be significantly lower (as low as -75 dBm MHz^{-1}) in other segments of the spectrum, to prevent interference with other applications such as GPS. The FCC UWB spectrum mask is shown in Fig. 2.11.

The ability of UWB technology to provide significantly high data rates within short ranges has made it an excellent alternative to Bluetooth for the physical layer of the IEEE 802.15.3a standard for WPANs. However, like 802.11 standards, two opposing groups of UWB developers are competing over the IEEE standard. The two competing technologies are single-band UWB and multi-band UWB. The single-band technique, backed by Motorola/XtremeSpectrum, supports the idea of impulse radio that occupies a wide spectrum. The multi-band approach divides the available UWB frequency spectrum into multiple smaller and non-overlapping bands with bandwidths greater than 500 MHz to obey the FCC's definition of UWB signals. The multi-band approach is supported by several companies, including Staccato Communications, Intel, Texas Instruments, General Atomics, and Time Domain Corporation.

For single-band UWB, the most popular proposal, Direct Sequence (DS)-UWB, uses a combination of a single-carrier spread-spectrum design and wide coherent bandwidth. Unlike conventional wireless systems, which use narrow-band modulated carrier waves



Fig. 2.11 FCC UWB mask. f_c is the carrier frequency.



Fig. 2.12 Multi-band UWB system.

to transmit information, DS-UWB transmits data using pulses of energy generated at very high rates (in excess of 1 billion pulses per second) providing support for data rates of 28, 55, 110, 220, 500, 660, and 1320 Mbit s⁻¹. A fixed UWB chip rate in conjunction with variable-length spreading code words enables this scalable support.

For multi-band UWB, as shown in Fig. 2.12, the available UWB spectrum, from 3.1 GHz to 10.6 GHz, is divided into S = 14 subbands. Each subband occupies a bandwidth of at least 500 MHz, in compliance with FCC regulations. The UWB system employs OFDM with N = 128 subcarriers, which are modulated using quadrature phase shift keying (QPSK). At each OFDM symbol period, the modulated symbol is transmitted over one of the *S* subbands. These symbols are time-interleaved across subbands. Different bit rates are achieved by using different channel coding, frequency spreading,

or time spreading rates. Frequency-domain spreading is obtained by choosing conjugate symmetric inputs to the IFFT, while time-domain spreading is achieved by repeating the same information in an OFDM symbol on two different subbands [64]. The receiver combines the information transmitted via different times or frequencies to increase the signal-to-noise ratio (SNR) of received data.

Owing to the extremely low emission levels currently allowed by regulatory agencies, UWB systems tend to be short-range and high-speed. High-data-rate UWB can enable wireless monitors, the efficient transfer of data from digital camcorders, wireless printing of digital pictures from a camera without the need for an intervening personal computer, and the transfer of files among cellphone handsets and other handheld devices like personal digital audio and video players. UWB is also used in "see-through-thewall" precision radar imaging technology, precision positioning and tracking (using distance measurements between radios), and precision time-of-arrival-based localization approaches.

2.2.5 Wireless ad hoc networks

An ad hoc network is an autonomous collection of mobile users that communicate over bandwidth-constrained wireless links. The network is decentralized, so all network activity including discovering the topology and delivering messages must be executed by the nodes themselves. Ad hoc networks need efficient distributed algorithms to determine network organization, link scheduling, and routing. For a special case of an ad hoc network, Mobile Ad Hoc Networks (MANETs), because the nodes are mobile the network topology may change rapidly and unpredictably over time.

The first generation of ad hoc networks was initiated in the early 1970s, when Packet Radio Networks (PRNET) was proposed by the Defense Advanced Research Projects Agency (DARPA) for multi-hop networks in a combat environment, and Areal Locations of Hazardous Atmospheres (Aloha) was proposed in Hawaii for distributed channel access management. The second generation of ad hoc networks emerged in the 1980s, when the ad hoc network systems were further enhanced and implemented as a part of the Survivable Adaptive Radio Networks (SURAN) program. SURAN provided a packet-switched network for the mobile battlefield, in an environment without infrastructure, to improve the performance of radios by making them smaller, cheaper, and resilient to electronic attacks. In the 1990s, the concept of commercial ad hoc networks arrived with notebook computers and other viable communications equipment. At this time the IEEE 802.11 subcommittee adopted the term "ad hoc networks."

The advantages of ad hoc networks are their ease and speed of deployment, important requirements for military applications. For civil applications, ad hoc networks decrease dependence on expensive infrastructure. The set of applications for ad hoc networks is diverse, ranging from small, static networks that are constrained by power sources, to large-scale, mobile, highly dynamic networks. Some typical applications are in personal area networking, emergency operations such as policing and fire fighting, civilian environments such as taxi networks, and military use on the battlefield. In contrast to the traditional wireless network with infrastructure, an ad hoc network needs its own design requirements so as to be functional. We list some important aspects:

• *Distributed operation and self-organization.* No node in the ad hoc network can depend on a network in the background to support basic functions like routing. Instead, these functions must be implemented and operated efficiently in a distributed manner. Moreover, in events such as topology changes due to mobility, the network must be self-organized to adapt to the changes.

In addition, the ad hoc nodes might belong to different authorities, who might not be willing to cooperate to fulfil network functions. However, this non-cooperation can cause severe network breakdown. To motivate distributed autonomous users to cooperate is an important research and design topic. Traditionally, pricing anarchy is employed using the distributed control theory. Later in this book, we explore other methods such as game theory to motivate users' cooperative behavior.

• **Dynamic routing.** For MANET, the routing problem between any pair of nodes is challenging because of the mobility of nodes. The optimal source-to-destination route is time-variant. Moreover, compared to traditional networks in which the routing protocols are proactive, the ad hoc dynamic routing protocols are reactive. The routes are determined only when the source requests a transmission to the destination. There are two types of ad hoc dynamic routing protocols: table-driven routing protocols and source-initiated on-demand routing protocols.

Table-driven routing protocols require each node to maintain one or more tables to store routing information. The protocols rely on an underlying routing-table update mechanism that involves the constant propagation of routing information. Packets can be forwarded immediately since the routes are always available. However, this type of protocol causes substantial signaling traffic and power-consumption problems. Protocols include in the literature destination-sequenced distance-vector routing [387], clusterhead gateway switch routing [105], and wireless routing [343].

Source-initiated on-demand routing creates routing only when desired by the source node. A disadvantage is that the packet at the source node must wait until a route can be discovered. But an advantage is that periodic route updates are not required. Routing protocols available in the literature include ad hoc on-demand distance vector routing [388], dynamic source routing [236], temporally ordered routing [386], associativity-based routing [476], and signal stability-based adaptive routing [135].

• *Connectivity.* To achieve a connected ad hoc network, for any node there must be a multi-hop path to any other node. There are many types of connectivity definitions. In an undirected graph *G*, two vertices *u* and *v* are called connected if *G* contains a path from *u* to *v*. Otherwise, they are called disconnected. A graph is called connected if every pair of vertices in the graph is connected.

One of the most frequently adopted definitions is k-connectivity, which states that each node can still connect to the rest of its network if k - 1 of its neighbor nodes are destroyed.

DEFINITION 2.1 A graph G with edge set V(G) is said to be k-connected if $G \setminus Y$ is connected for all $Y \subseteq V(G)$ with |Y| < k. In other words, a graph is k-connected if the graph remains connected when fewer than k vertices are deleted from the graph.

If a graph *G* is *k*-connected, and k < |V(G)|, then $k \le \Delta(G)$, where $\Delta(G)$ is the minimum degree of any vertex $v \in V(G)$.

The following theorem due to Menger [329], a special case of the Max-Cut Min-Flow Theorem, indicates how to calculate k for k-connectivity:

THEOREM 2.1 Let G be a finite undirected graph and i and j be two non-adjacent vertices. Then the size of the minimum vertex cut for i and j (the minimum number of vertices whose removal disconnects i and j) is equal to the maximum number of pair-wise vertex-independent paths from i to j.

 Mobility. To test a new protocol for MANET, it is important to use a mobility model that accurately represents the mobile users using the protocol, so that practical implementation matches the simulation. There are two types of mobility models: traces and synthetic models. For traces, the mobility patterns are observed in real-life systems, so that accurate information is provided. However, traces are limited to existing environments. For unknown environments, accurate synthetic models are necessary.

A synthetic model tries to simulate the real movement of mobile users. In [90], several mobility models are discussed and explained. If the different mobile users are moving randomly relative to each other, mobility models include:

- Random walk: a simple model based on random directions and speeds.
- Random waypoint: a model including pause times between changes in direction and speed.
- Random direction: a model forcing mobile users to travel to the edge of the simulation area before changing direction or speed.
- Boundless simulation area: a model converting a 2D rectangular simulation area into a torus-shaped simulation area.
- Gauss–Markov: a model using a set of parameters to change the degree of randomness in mobility patterns.
- Probabilistic version of random walk: a model specifying probabilities of the next positions of mobile users.
- City section: a model in which movement is on the streets of a city.
- Random trip [79]: a model that contains, as special cases, random waypoint on convex or non-convex domains, random walk, billiards, city section, space graph, and other models.

If the mobile users are moving in groups, mobility models include:

• Exponential correlated random: a model using a motion function to create movement.

- Column: a model in which mobile users form a line and are uniformly moving forward in a certain direction.
- Nomadic community: a model in which mobile users move together from one position to another.
- Pursue: a model in which mobile users follow a given target.
- Reference point group: a model in which group movements are based on the path travelled by a logical center.
- Other issues such as lower power, security, and localization will be discussed in the next section, together with sensor networks.

Finally, we discuss two ad hoc standards in 802.11 for wireless local networks and in 802.15 for wireless personal area networks. One future design goal of ad hoc networks is to let mobile users form connections and perform basic functions. These mobile users belong to different types of networks such as cellular, WiFi, and WPAN.

Most installed wireless LANs today utilize an "infrastructure" mode that requires the use of one or more access points. With this configuration, the access point provides an interface to a distribution system (e.g., Ethernet), which enables wireless users to use corporate servers and Internet applications. As an option, the 802.11 standard specifies "ad hoc" mode, which allows the radio network interface card (NIC) to operate in what the standard refers to as an Independent Basic Service Set (IBSS) network configuration. With an IBSS, there are no access points. User devices communicate directly with each other in a peer-to-peer manner.

For WPAN applications such as Bluetooth, the ad hoc network is set up by forming piconets. Within each piconet, only one master device and possibly several slave devices form connections. A slave device can belong to different piconets and serve as a connection between piconets. The major functions of the piconets are piconet forming and maintenance, packet forwarding, and intra-piconet/inter-piconet scheduling.

2.2.6 Wireless sensor networks

A wireless sensor network (WSN) is a wireless network consisting of spatially distributed autonomous devices using sensors to cooperatively monitor physical or environmental conditions such as temperature, sound, vibration, pressure, motion, or pollutants, at different locations. The goals and tasks of sensor networks are to determine the value of some parameter at a given location, to detect the occurrence of events of interest and estimate parameters of the detected events, to classify a detected object, or to track an object. The development of wireless sensor networks was originally motivated by military applications such as battlefield surveillance. However, wireless sensor networks are now used in many civilian application areas, including environment and habitat monitoring, healthcare applications, home automation, and traffic control. Some examples are as follows:

• Military sensor networks to detect and gain as much information as possible about enemy movements, explosions, and other phenomena of interest

- Sensor networks to detect and monitor environmental changes such as in plains, forests, and oceans
- Wireless traffic sensor networks to monitor vehicle traffic on highways or in congested parts of a city
- Wireless surveillance sensor networks to provide security in shopping malls, parking garages, and other civil facilities
- Manufacturing sensors to facilitate the monitoring and control process, which can reduce cost, improve flexibility, and enhance accuracy
- Sensors for supermarkets to speed products to shelves and provide customers with better-quality products
- Medical sensors, especially implanted sensors, to constantly monitor patients; these must have sufficient battery life and be able to transmit the sensed information out of the body via wireless channels.

In Fig. 2.13 we show the typical structure of one unit of a sensor network. In addition to one or more sensors, each node in a sensor network is typically equipped with a radio transceiver or other wireless communications device, a small microprocessor, some memory, and an energy source, usually a battery. Sensing applications can include temperature, light, humidity, pressure, acceleration, magnetic fields, chemical properties, acoustics, and images/videos. The microprocessor has significant constraint on its computational power. Currently, devices typically have a component-based, embedded operating system. Available memory is also very limited. Current radio transceivers for sensor networks are low-rate and short-range. Some sensors can be powered by a wired power source, while most widely deployed sensors are powered by battery. Exchanging the energy-depleting sensors is a challenging job, so power saving is critical in the design of such wireless sensor networks.

The size of a single sensor node can vary from shoe-box size down to devices the size of a grain of dust. The cost of sensor nodes is similarly variable, ranging from



Fig. 2.13 Sensor network structure.

hundreds of dollars to a few cents, depending on the size of the sensor network and the complexity required of individual sensor nodes. Size and cost constraints on sensor nodes result in corresponding constraints on resources such as energy, memory, computational speed, and bandwidth. Basically, to design a wireless sensor network, the following requirements should be considered [6]:

- Large number of (mostly stationary) sensors
- Low energy use to extend network lifetime
- Self-organization
- Collaborative signal processing
- Querying ability.

In practice, there are other design issues [257], including how to deploy the sensor networks, how to locate a specific sensor, how to shut down and reactivate a sensor to save energy, how to route information back to data-collection points, how to reduce packet-forwarding loads by data fusion, and finally how to have secure sensor networks.

• Sensor network deployment. The problem is to select the locations to place the sensor networks, give a particular application context, an operational region, and a set of wireless sensor devices. The sensors can be deployed in a structured sense or in a randomly scattered manner. The density of sensors is determined by the robustness and cost of the networks.

Most sensor networks have two types of sensors: a low-capability sensor that is in charge of collecting data, and a cluster head or data sink that is more powerful in computation and data transmission. The network topology between these two types of sensors can be star-connected single-hop, multi-hop mesh/grid, or multiple-tier hierarchical cluster.

Because transmit power is bounded, a sensor can reach other sensors only within a limited distance, giving rise to connectivity problems for sensor networks. The sensing range is also restrained. The sensing area is a function of the density of the sensors. There are different types of coverage metrics, including *k*-coverage, minimum coverage, and maximal breach distance [326].

• *Localization*. This refers to the determination of the location of a sensed event. This information can be used to provide a location stamp over the event, track the monitored object, determine the coverage, form the cluster, facilitate routing, and perform efficient querying. Even though such information can be obtained using GPS, cost and indoor environments prohibit sensors from being equipped with GPS.

Nevertheless, the task of localization captures multiple aspects of sensor networks. Physical layer imposes measurement challenges, due to multi-path, shadowing, sensor imperfections, and changes in propagation properties. Extensive computation is necessary for many formulations of localization problems. Moreover, problems must sometimes be solved in a distributed manner or on a memory-constrained processor. Furthermore, for networking and coordination issues, sensor nodes have to collaborate and communicate with each other to know the topology of the whole network. Finally, it is challenging to integrate location services with other applications.

There are several types of localization mechanisms. An active localization system sends signals to localize a target. Examples include RADAR and LIDAR (LADAR). In cooperative localization, the target cooperates with the system. For example, the target emits a signal with known characteristics, and the system deduces its location by detecting a signal. A passive localization system deduces location from observation of signals that are "already present." An example is the use of geometric methods to calculate location by measuring signal strength at receivers in different locations. A blind localization system deduces the location of a target without a-priori knowledge of its characteristics.

• *Time synchronization.* While localization provides the sensor networks with spatial information, accurate time synchronization is also essential. Since the time delay for the information to the sink is unpredictable, each sensor needs to have a consistent time stamp for the message. This is very important for some types of data such as tsunami alarming, since the time information provides many scientific clues. For localization, the transmitter and receiver need to have synchronized time so that time-of-flight can be calculated. For multiple access, such as TDMA (time-division multiple-access)-based schemes, each sensor needs to transmit at exact time slots. For sleeping scheduling, energy is saved by turning sensors on and off at certain time.

Accurate time can be obtained from a GPS signal, but this approach is very expensive. Quartz-crystal oscillators can provide an accuracy of a few μ s. For better accuracy, techniques such as phase-locked loops need to be implemented to synchronize the clocks.

• *Sleeping mechanism.* In most sensors, the primary power consumption is by the radio used for transmitting, receiving, and listening. If the sensors only wake up when the radio is active and sleep during the remaining time, energy can be conserved and the lifetime of the sensor network can be prolonged.

However, the sleep-and-wake-up mechanism causes other design problems. First, there is a tradeoff between the delay of information and energy consumption. Moreover, the design of the MAC-layer multiple-access protocol needs to consider the wake-up time. Furthermore, the transmitter and receiver should be synchronized to wake up at the same time. Finally, the fairness issue needs to be considered so that some sensors are not overloaded, leading to early energy depletion.

• *Energy-efficient routing.* Since energy is a major concern in the design of wireless sensor networks, energy-efficient routes from the sensors to the data sink can significantly improve the network lifetime. In addition, when multiple routes are considered, individually energy-efficient routes are not optimal, in the sense that some sensors on critical paths might be depleted early. So joint optimization is necessary.

In addition to energy concerns, routing protocols also should take account of latency arising from the sleeping mechanism of sensors. Routing protocols should also consider data fusion/aggregation. Finally, for large sensor networks, scalability is an important issue. For situations in which the sensors are mobile or can join/leave the network frequently, adaptive ability is also a design challenge. • *Fusion/aggregation*. Fusion/aggregation is a process of association, correlation, and combination of data and information from single and multiple sources to achieve refined position and identity estimates for observed entities, and to achieve complete and timely assessments of situations and threats, and their significance. In wireless sensor networks, especially large ones, it can be difficult and energy-inefficient to gather the information to make such a decision. Instead, along the path to the data sink, a data-fusion node can collect the results from multiple nodes, fuse the results with its own based on a decision criterion, and then send the fused data to another node/base station. By doing this, the traffic load can be greatly reduced and energy can be conserved.

There are two types of data fusion/aggregation. In the first, data from different node measurements are combined to form larger packets. It is simple to implement, but requires a higher computational burden, higher communication burden, and larger training data. The second type is decision fusion, in which decisions (hard or soft) based on node measurements are combined. Decision fusion solves the problems of the first type of fusion/aggregation. The decision can be made by mechanisms such as voting. For example, a fusion node may arrive at a consensus by a voting scheme: majority voting, complete agreement, or weighted voting. Other fusion decision algorithms include a probability-based Bayesian model [101] and stacked generalization [378].

For sensor networks, there are various fusion architectures from the sensors to the data sink. In a centralized architecture, a central processor fuses the reports collected by all other sensing nodes. The centralized one has the advantages that erroneous reports can be easily detected and that it is simple to implement. On the other hand, it has the disadvantages that it is inflexible to sensor changes and that the workload is concentrated at a single point. In a decentralized architecture, data fusion occurs locally at each node on the basis of local observations and the information obtained from neighboring nodes. There is no central processor node. Advantages are that it is scalable and tolerant to the addition or loss of sensing nodes or dynamic changes in the network. In a hierarchical architecture, nodes are partitioned into hierarchical levels. The sensing nodes are at level 0 and the data sink at the highest level. Reports move from the lower levels to higher ones. This architecture has the advantage that workload is balanced among nodes.

• *Security.* Because sensor networks may interact with sensitive data and/or operate in hostile, unattended environments (e.g., military sensors), it is important for security to be addressed in system design. Moreover, because of inherent resource and computing constraints, security in sensor networks poses more challenges than traditional network/computer security. Possible security attacks include denial-of-service attack, Sybil attack, traffic-analysis attack, node-replica attack, privacy attack, physical attack, and collusion attack. In the literature, there are some defensive mechanisms such as key cryptography and trust management. The security issue is beyond the scope of this book. A good survey of security issues in wireless sensor networks can be found in [493].

2.3 Advanced wireless technology

2.3.1 OFDM technology

Orthogonal frequency-division multiplexing (OFDM) is a technique for transmitting multiple digital signals simultaneously over a large number of orthogonal subcarriers. Based on the fast Fourier transform algorithm to generate and detect the signal, data transmission can be performed over a large number of carriers that are spaced at precise frequencies. The frequencies (or tones) are orthogonal to each other. Therefore, the spacing between the subcarriers can be reduced and high spectral efficiency can be achieved. OFDM transmission is also resilient to interference and multi-path distortion, which causes inter-symbol interference (ISI).

OFDM transmitter and receiver block diagrams are shown in Figs. 2.14 and 2.15, respectively; s[n] is a serial stream of binary digits to transmit. After serial-to-parallel conversion, the data is split into *N* streams. Each stream is then coded to $X_0, ..., X_{N-1}$ with possible different modulation methods, such as PSK (phase shift keying) or QAM (quadrature amplitude modulation), depending on the subchannel condition. An inverse FFT is computed on each set of symbols, giving a set of complex time-domain



Fig. 2.14 Illustration of OFDM transmitter.



Fig. 2.15 Illustration of OFDM receiver.

samples. These samples are then quadrature-mixed to passband in the standard way: the real and imaginary components are first converted to the analogue domain using digital-to-analogue converters (DACs); the analogue signals are then used to modulate cosine and sine waves, respectively, at the carrier frequency, f_c . These signals are then summed to yield the transmission signal, s(t). The OFDM receiver picks up the signal r(t), which is s(t) transmitted through radio channels and contaminated by noise. Then r(t) is quadrature-mixed down to baseband using cosine and sine waves at the carrier frequency. The baseband signals are then sampled and digitized using analogue-to-digital converters (ADCs), and a forward FFT is used to convert back to the frequency domain. This returns N parallel streams, each of which is converted to a binary stream using an appropriate symbol detector. These streams are then recombined into a serial stream, $\hat{s}[n]$, which is an estimate of the original binary stream at the transmitter.

Mathematically, the low-pass equivalent OFDM signal is expressed as

$$V(t) = \sum_{k=0}^{N-1} X_k e^{j2\pi kt/T}, \ 0 \le t < T,$$
(2.9)

where X_k is the modulated data symbol for the *k*th data stream, and *T* is the OFDM symbol time.

To avoid inter-symbol interference in multi-path fading channels, a guard interval T_g is inserted prior to the OFDM block. During this interval, a cyclic prefix is transmitted such that the signal in the interval $-T_g \le t < 0$ equals the signal in the interval $(T - T_g) \le t < T$. A cyclic prefix is often used in conjunction with modulation in order to retain sinusoids' properties in multi-path channels. It is well known that sinusoidal signals are eigenfunctions of linear and time-invariant systems. Therefore, if the channel is assumed to be linear and time-invariant, then a sinusoid of infinite duration would be an eigenfunction. However, in practice, this cannot be achieved, as real signals are always time-limited. So, to mimic the infinite behavior, prefixing the end of the symbol to the beginning makes the linear convolution of the channel appear as though it were a circular convolution, and thus preserves this property in the part of the symbol after the cyclic prefix.

Orthogonal frequency-division multiple access (OFDMA) is a multiple-access scheme based on OFDM. In OFDMA, different users are allocated with different subcarriers, and hence multiple users can transmit their data simultaneously. QoS can be achieved in OFDMA by allocating different numbers of subcarriers to users with different QoS requirements. In OFDMA, different users can occupy different time-frequency slots so as to fully utilize the diversity. OFDM can be combined with the CDMA scheme (i.e., multi-carrier code-division multiple-access (MC-CDMA) or OFDM-CDMA). In this case, different codes are assigned to different users for concurrent transmissions.

Dynamic spectral management for OFDMA can be categorized into two groups based on the network control architecture.

• Centralized control: A spectral management center (SMC) monitors and controls the transmit spectra of all the users in the system. It needs a lot of coordination and

communication between the users and the SMC. This greatly increases the control signaling overhead but leads to better performance when compared to distributed control. Centralized control can be further categorized as follows [245, 509]:

- Level 1: The data rate and transmit power of the user are reported to a central controller and corresponding control signals are generated to control the rate and transmit power.
- Level 2: The noise spectra and the received signal spectra are monitored and the transmit power is controlled by the central controller.
- Level 3: This allows complete coordination in real-time control of transmit power while monitoring the signals and noise spectra.
- Distributed control: Users have the capability to sense the channel conditions and adjust their transmit spectra accordingly. This may be efficient from the perspective of signaling and coordination but has the drawback of converging to a suboptimal point. It is also referred to as Level 0 [509, 245], where there is no dynamic spectrum management (DSM) involved and control is fully distributed.

The control exhibited by the SMC over the users determines the computational complexity of the overall system. A centralized control is efficient but consumes a lot of bandwidth in control messaging between the users and the base station. On the other hand, distributed algorithms increase the complexity of the user's receiver. The resource allocation of OFDMA can be broadly categorized into three areas:

- Subcarrier assignment: The subcarriers with the best channel gains as seen by the user are allocated to the particular user.
- Rate allocation: The data rate is allocated depending on user application requirements.
- Power control: Optimal transmit power is to be allocated to the user in order to meet its rate requirements while not interfering with other users.

OFDM and OFDMA are used in emerging standards including IEEE 802.11a/WiFi, IEEE 802.15/WiPAN, IEEE 802.16/WiMAX, IEEE 802.20/MobileFi, IEEE 802.22/WiRAN, digital audio broadcasting (DAB), terrestrial broadcasting of digital television (DVB-T, DVB-H), Flash-OFDM (fast low-latency access with seamless handoff orthogonal frequency-division multiplexing), SDARS for satellite radio, G.DMT (ITU G.992.1) for ADSL, and ITU-T G.hn for power line communication.

2.3.2 Multiple-antenna systems

For spatial diversity, transceivers employ antenna arrays and adjust their beam patterns so that they have good channel gain in the desired directions, while aggregate interference power is minimized at their output. Antenna array processing techniques such as beamforming, MIMO technology, and space-time coding can be applied to receive and transmit multiple signals that are separated in space. Hence, multiple cochannel users can be supported in each cell to increase the capacity by exploring the spatial diversity. In this subsection, we briefly discuss various multiple antenna technologies.



Fig. 2.16 Illustration of beamforming.

Beamforming

Beamforming is a technique in signal processing used for directional data transmission and reception. Using beamforming, the signal is received/transmitted in a particular direction to improve the receive/transmit gain. When transmitting, a beamformer controls the phase and amplitude of the signal to create a pattern of constructive and destructive interference in the wavefront. When receiving, these signals are combined in such a way that the expected pattern of radiation is preferentially observed. Different weights can be assigned to the different signals so that the decoded information has the smallest error probability. In a conventional beamforming system, these weights are fixed and can be obtained according to the location of the receiving antenna and the direction of the signal. Alternatively, the weights can be adaptively adjusted by considering the characteristics of the received signal to mitigate the interference from unwanted sources. In Fig. 2.16, we show an illustration of beamforming technology.

MIMO technology and space-time coding

To enhance the performance of wireless transmission, multiple-input multiple-output (MIMO) or multiple antennas can be used to transmit and receive the radio signals. Data transmitted from multiple antennas will experience different multi-path fading, and at the receiver these different multi-path signals are received by multiple antennas. By using advanced signal processing techniques, multi-path signals at the receiver can be combined to reconstruct original data. MIMO systems take advantage of this spatial diversity to achieve higher data rates or lower bit error rates (BER). There are two basic types of MIMO systems, space-time coding MIMO (for diversity maximization) and spatial multiplexing MIMO (for data rate maximization).

In a space-time coding MIMO system, a single data stream is redundantly transmitted over multiple antennas by using suitably designed transmit signals. At the receiver, multiple copies of the signal are received and used to reconstruct the original data. Space-time coding can be categorized into space-time trellis coding (STTC), in which trellis code is transmitted over multiple antennas and multiple time-slots, and space-time block coding (STBC) in which a block of data is transmitted over multiple antennas. Both STTC and STBC can achieve diversity gain which improves the error performance. While STTC can also achieve coding gain (i.e., results with a lower error rate), STBC can be implemented with less complexity.

Instead of transmitting the same data over multiple antennas, a spatial multiplexing MIMO system transmits different data streams over multiple antennas. In this case, the number of transmitting antennas is equal to or larger than that of the receiving antennas and the data rate can increase by a factor of the number of transmitting antennas [479]. There is a fundamental tradeoff between diversity gain and multiplexing gain. In simple terms, the diversity gain improves the BER performance, while multiplexing gain will increase the rate but the BER performance may be reduced. Detailed analysis can be found in the literature [535].

One simple example of an STBC is the Alamouti code [19], which was designed for a two-transmitting antenna system and has the coding matrix:

$$C = \begin{bmatrix} s_1 & s_2 \\ -s_2^* & s_1^* \end{bmatrix}, \qquad (2.10)$$

where s_1 and s_2 are two symbols, * denotes complex conjugation, and components of the matrix are sent on two antennas and two time slots, respectively. The coding rate is 1, and using the optimal linear decoding scheme the BER of this STBC is equivalent to maximal ratio combining (MRC).

Multi-user MIMO (MU-MIMO) was proposed to support data transmission from multiple users simultaneously. In this case, data from different users can be transmitted over different antennas. In this MU-MIMO, MIMO broadcast channels and MIMO multiple-access channels are used for downlink and uplink transmissions, respectively. Alternatively, space-division multiple-access (SDMA) can exploit information about users' locations to adjust the transmission and reception parameters to achieve the best path gain in the direction of each user. Phased-array antenna techniques are generally used for SDMA. MIMO is an optional feature in the IEEE 802.16/WiMAX standard, while it is be part of the IEEE 802.11n standard.

2.3.3 Cognitive radio

The FCC Federal Radio Act allows predetermined users the right to transmit at given frequencies. Unlicensed users are regarded as "harmful interference," and in most cases sidebands were implemented to ensure that interference is not an issue. As technology advanced, higher frequency bands were sold at auction, bringing considerable revenue to the government. For example since the 1994 PCS auction, over US\$30 billion has been

generated. As demands for wireless communication become more and more pervasive, wireless devices must find a way to transmit at frequencies in the limited radio band. However, there exists a large number of frequency bands that have considerable, and sometimes periodic, dormant time intervals. For example, TV stations often do not work at night. There exist some spectrum holes at given times over different spectrum bands. So there is a dilemma, in that on the one hand mobile users have no spectrum to transmit, while on the other hand some spectra are not fully utilized.

In order to cope with the dilemma, cognitive radio is a paradigm for wireless communication in which either a network or a wireless node changes its transmission or reception parameters to communicate efficiently without interfering with licensed users. This alteration of parameters is based on active monitoring of several factors in the external and internal radio environment, such as radio-frequency spectra, user behaviors, and network states.

Depending on the parameters taken into account in deciding on transmission and reception changes, and for historical reasons, we can distinguish several types of cognitive radio:

- Full cognitive radio ("Mitola radio"), in which every possible parameter observable by a wireless node or network is taken into account.
- Spectrum-sensing cognitive radio, in which only the radio-frequency spectrum is considered.
- Licensed-band cognitive radio, which is capable of using bands assigned to licensed users.
- Unlicensed-band cognitive radio, which can only utilize unlicensed parts of the radiofrequency spectrum.

Cognitive radio can be designed as an enhancement layer on top of the softwaredefined radio (SDR) concept. An SDR system is a radio communication system which can tune to any frequency band and receive any modulation across a large frequency spectrum by means of programmable hardware which is controlled by software.

An SDR performs significant amounts of signal processing in a general-purpose computer, or a reconfigurable piece of digital electronics. The goal of this design is to produce a radio that can receive and transmit a new form of radio protocol just by running new software. The hardware of a software-defined radio typically consists of a superheterodyne RF front end which converts RF signals from (and to) analogue IF signals, and analogue-to-digital converter and digital-to-analogue converters which convert a digitized IF signal from and to analogue form, respectively.

Software radios have significant utility for the military and cellphone services, both of which must serve a wide variety of changing radio protocols in real time. Software-defined radio can currently be used to implement simple radio modem technologies. In the long run, software-defined radio is expected by its proponents to become the dominant technology in radio communications. It is the enabler of cognitive radio.

The cognitive transmitter and receiver can adapt to 3G, WiFi, and WPAN networks. By sensing the available spectrum, the cognitive radio can adapt to the most suitable available communication links. For example, a user at home can communicate via Bluetooth; if the user travels to the airport, WiFi communication can be available; if the user drives on the highway, the cellular phone system can provide reliable communication links. Another analogy for cognitive radio is as follows. Licensed users are legally used to spectrum, like high-occupancy vehicles (HOV) which can drive in HOV designated lanes. However, the HOV lanes are not always occupied. Aside from during rush hour, other vehicles can also drive in the HOV lanes. In this sense, cognitive radios are like non-HOV vehicles.

Two major objectives of cognitive radio are to reliably communicate anywhere and at any time, and to efficiently utilize the radio spectrum. To achieve these objectives, three fundamental cognitive tasks [201] for cognitive radio must be fulfilled:

- Radio-scene sensing, which analyzes interferences and detects spectrum holes.
- Spectrum analysis, such as channel-state estimation and predictions of channel capacity.
- Transmit-power control and dynamic spectrum management.

In Fig. 2.17 we show the cognitive cycle, in which these three tasks interact to handle the outside world so that the best strategy can be calculated and implemented.

The IEEE 802.22 Working Group on Wireless Regional Area Networks (WRAN) is a group of the IEEE 802 LAN/MAN Standards Committee. Standards for WRAN Part 22 (Cognitive Wireless RAN Medium Access Control (MAC) and Physical Layer (PHY) Specifications) regulate policies and procedures for operation in the TV bands. The standard focuses on constructing a consistent, fixed point-to-multipoint WRAN that will utilize UHF/VHF TV bands between 54 and 862 MHz. Specific TV channels as well as the guard bands of these channels are to be used for communication in IEEE 802.22.

The IEEE, together with the FCC, is pursuing a centralized approach to available spectrum discovery. Specifically, each access point (AP) would be armed with a GPS



RF channel bandwidth	6 MHz
Average spectrum efficiency	$3 \text{ bit s}^{-1} \text{ Hz}^{-1}$
Downlink user capacity	$1.5 { m Mbit s^{-1}}$
Uplink user capacity	$384 \rm kbit s^{-1}$
Over-subscription ratio	50
Number of users per downlink	600
Minimum number of users	90
Assumed early take-up rate	$3 \rm bit s^{-1} Hz^{-1}$
Potential number of users	1800
Number of users per household	2.5
Number of users per coverage area	4500
WRAN base station power	98.3 W
Coverage	30.7 km
Minimum population density	$1.5 {\rm km}^{-2}$

Table 2.5 WRAN coverage and capacity.

receiver which would allow its position to be reported. This information would be sent back to centralized servers (in the USA these would be managed by the FCC), which would respond with information about available free TV channels and guard bands in the area of the AP. Other proposals would allow local spectrum sensing only, in which the AP would decide by itself which channels are available for communication. A combination of these two approaches is also envisioned. Table 2.5 describes the coverage and capacity of IEEE 802.22 WRAN.

Overall, cognitive radio can bring in a variety of benefits. For a regulator, it can lead to a significant increase in spectrum availability for new and existing applications. For a license holder, it can reduce the complexity of frequency planning, facilitate secondary-spectrum market agreements, increase system capacity through access to more spectrum, and avoid interference. For the equipment manufacturer, it can increase demand for wireless devices. Finally, for the user, cognitive radio can provide a higher level of capacity per user, enhance interoperability and bandwidth-on-demand for public-safety and emergency-response operations, and provide ubiquitous mobility with a single user device across disparate spectrum-access environments.

Part I

Fundamentals of game theory
Non-cooperative game theory is one of the most important branches of game theory, focusing on the study and analysis of competitive decision-making involving several players. It provides an analytical framework suited for characterizing the interactions and decision-making process involving several players with partially or totally conflicting interests over the outcome of a decision process which is affected by their actions. Examples of non-cooperative games are ubiquitous. In economics, firms operating in the same market compete over pricing strategies, market control, trading of goods, and the like. In wireless and communication networks, wireless nodes are involved in numerous non-cooperative scenarios such as allocation of resources, choices of frequencies or transmit power, packet forwarding, and interference management. Beyond economics and networking, non-cooperative game theory has made its impact over a broad range of disciplines such as biology, political science, sociology, and military tactics. In this chapter, we introduce non-cooperative game theory along with different types of games, while presenting underlying fundamental notions and key solution concepts.

3.1 Non-cooperative games: preliminaries

In this section, we introduce some preliminary concepts and terminology that pertain to non-cooperative game theory.

3.1.1 Introduction

A non-cooperative game involves a number of players having totally or partially conflicting interests in the outcome of a decision process. For example, consider a number of wireless nodes attempting to control their transmit power, given the interference generated by other nodes. In such a situation, while all the nodes have an incentive to transmit, the presence of interference presents a conflict, coupling the decisions of the nodes: Every node wants to transmit at its maximal power level, to improve its performance; however, doing so increases the overall interference in the system, which, in turn, adversely impacts the performance of all the involved wireless nodes.

A non-cooperative game is a game reflecting a competitive situation where each player needs to take its decision independently of the other players, given the possible choices of the other players and their effect on the player's objectives or utilities. Note that the term

non-cooperative does not always imply that the players do not cooperate, but it means that any cooperation that might arise must be self-enforcing with no communication or coordination of strategic choices among the players.

A game is said to be *static* if the players take their actions only once, independently of each other. In some sense, a static game is a game without any notion of time, where no player has any knowledge of the decisions taken by the other players. Even though, in practice, the players may have made their strategic choices at different points in time, a game would still be considered static if no player has any information on the decisions of others. Thus, such a game can still be analyzed as if the decisions were made simultaneously. In contrast, a *dynamic* game is one where the players have some information about each others' choices and can act more than once, and where time has a central role in the decision-making. Although these definitions of static and dynamic games are the most widely accepted ones in the game-theory literature, static and dynamic games have no universally acknowledged definitions (see [58] for more details on this issue).

3.1.2 Basics of non-cooperative games

In describing a static or dynamic non-cooperative game, the notion of a strategic (or normal) form proves to be one of the most popular representations. In this regard, a non-cooperative game in *strategic (or normal) form* has three components: the set of players, their strategies, and the payoffs or utilities. More formally, a strategic game is defined as follows:

DEFINITION 3.1 A non-cooperative game in strategic (or normal) form is a triplet $G = (\mathcal{N}, (\mathcal{S}_i)_{i \in \mathcal{N}}, (u_i)_{i \in \mathcal{N}})$, where:

- \mathcal{N} is a finite set of players, i.e., $\mathcal{N} = \{1, \dots, N\}$.
- S_i is the set of available strategies for player *i*.
- $u_i : S \to \mathbb{R}$ is the utility (payoff) function for player *i*, with $S = S_1 \times \cdots \times S_i \times \cdots \times S_N$ (*Cartesian product of the strategy sets*).

Note that while different game forms can be defined for dynamic games (as will be seen in Section 3.3), for static games the strategic (or normal) form is essentially the *only* representation.

When dealing with *dynamic* games (e.g., in Section 3.3), the choices of each player are generally dependent on some available information. Thus, in dynamic games, one has to distinguish between the notion of an *action* and a *strategy*. For example, if an individual has to decide what to do in the evening and the options are camping in nature or staying at home, then a possible strategy would be "If the weather report predicts dry weather for the evening, then I will go out camping; otherwise, I will stay at home." After knowing about the weather, the individual would take an *action*, which is to go camping or to stay at home, depending on the available information. Thus, in essence, a strategy can be seen as a mapping from the information available to a player (e.g., the weather report) to the action set of this player (e.g., go camping or stay at home). A *constant*

strategy, e.g., a decision to stay at home regardless of any information, coincides with the notion of an action. As a result, in *static* games, the choice that a given player makes is independent of any information; thus, in such games, the concepts of an action and a strategy are identical. Throughout this book, whenever we deal with static games, i.e., games where information has no role and where the decisions are simultaneous, we do not distinguish between the notions of action and strategy, and so the terms will be used interchangeably.

Given the definition of a strategic game, for any player *i*, every element $s_i \in S_i$ is the strategy of *i*, $\mathbf{s}_{-i} = [s_j]_{j \in \mathcal{N}, j \neq i}$ denotes the vector of strategies of all players *except i*, and $\mathbf{s} = (s_i, \mathbf{s}_{-i}) \in S$ is referred to as a *strategy profile*. Whenever the sets of strategies S_i are finite for all $i \in \mathcal{N}$, the game is called *finite*.

For a game in strategic form, each player has to select a strategy so as to optimize its utility function.¹ Whenever each player $i \in \mathcal{N}$ selects a strategy $s_i \in S_i$ in a deterministic manner, i.e., with probability 1, then this strategy is known as a *pure strategy*. Furthermore, unless stated otherwise, in the remainder of this chapter we deal with games with *complete information*, a notion defined as follows:

DEFINITION 3.2 A game is said to be one with complete information if all elements of the game are common knowledge. Otherwise, the game is said to be one with incomplete information, or an incomplete information game.

Thus, in a game with complete information, each player is aware of the identities of all other players, their strategies, and the payoffs that would result from any combination of strategies. For an *incomplete information* game, the players may not know the identities of all other players, their payoffs, or their strategies; in this case the use of inference methods such as Bayes' rule can be useful. In this chapter, we deal mainly with games with complete information; games with incomplete information will be tackled under the heading of Bayesian games in Chapter 4.

One of the most common types of non-cooperative games is the *two-player zero-sum* game. A two-player zero-sum game is one involving two players where the gains of one player are the losses of the other player. In other words, it is a game such that $\mathcal{N} = \{1, 2\}$ and $\sum_{i=1}^{2} u_i(\mathbf{s}) = 0, \forall \mathbf{s} \in S$, where \mathbf{s} is a strategy profile. In such a game, one player is a maximizer, i.e., aims to maximize its gain, while the other player). A zero-sum game can also be thought of as a constant-sum game where the gains and losses of the players add up to a constant value, for any strategy profile. The popular board game Go is a typical example of a zero-sum game: It is impossible for both players to win. In wireless and communication networks, zero-sum games are especially popular when modeling security games involving an attacker and a defender (e.g., see [22] and references therein). In such games, the attacker's gains are most often equal to the defender's losses, yielding a zero-sum situation.

¹ The discussion in this chapter is also applicable to games where the players seek to minimize a cost function instead of maximizing a utility function.

Despite being one of the most well studied and analyzed classes of games, zero-sum games are restrictive. In fact, in many disciplines, especially wireless and communication networks, the majority of the studied problems are better modeled as *non-zero-sum* games. Non-zero-sum games describe situations in which (without any loss of generality) all players can be viewed as maximizers or minimizers, while having no constraint on the total sum of utilities. In some sense, non-zero-sum games describe scenarios where the participants can all gain or suffer together. Examples are prevalent in wireless and communication networks. For example, the popular power-control non-cooperative game is a non-zero-sum game: The sum of the rates (i.e., utilities) achieved by the wireless users (i.e., players) is different depending on the selection of transmit power (i.e., strategy profile) by the users.

3.2 Non-cooperative games in strategic form

In this section, we deal with non-cooperative games represented in strategic form. First, we define the concept of a matrix game, which provides an interesting approach for solving non-cooperative games. Then, we study various properties and concepts useful for treating non-cooperative games in strategic form, such as dominance, Nash equilibrium, Pareto optimality, and mixed strategies.

3.2.1 Matrix games

Non-cooperative games in strategic form with complete information are central building blocks in the understanding of game-theoretic decision-making. In these games, the objective is to determine whether a reasonable outcome or solution to the game exists. A solution implies a set of strategies that the players, when acting rationally, i.e., so as to optimize their own utility, would select.

In order to analyze a non-cooperative game in strategic form, one must first clearly specify the players, their strategies, and their potential payoffs. In this context, any two-player non-cooperative finite game can be represented in a matrix format whereby the strategies of the players constitute the rows and columns of the matrix, and each element is a pair of numbers that represent the payoffs for the two players when a certain combination of strategies is used. A game represented by a matrix is referred to as a *matrix game*. The matrix representation of a game is mainly composed of the following:

- Each row represents a strategy for the first player in the game, sometimes referred to as the *row player*. Thus, the number of rows is equal to the number of strategies for the row player.
- Each column represents the strategies of the second player in the game, sometimes referred to as the *column player*. Thus, the number of columns is equal to the number of strategies for the column player.

	Confess (C)	Not confess (NC)
Confess (<i>C</i>)	(-4,-4)	(0,-5)
Not confess (<i>NC</i>)	(-5,0)	(-2,-2)

Table 3.1 Prisoner's Dilemma (prisoner 1, prisoner 2).

• Each entry of the matrix is a pair (x, y), where x represents the payoff for the first player, i.e., the row player, and y represents the payoff for the second player, i.e., the column player.

In what follows, we provide some classical examples of non-cooperative matrix games:

Example 3.1 (The Prisoner's Dilemma) Two suspects are arrested for a crime and placed in two isolated rooms. Each one of the suspects has to decide whether or not to confess and implicate the other. The governing rules are the following. If none of the suspects confesses, then each will serve 2 years in jail. If both of them confess and implicate each other, they will both go to prison for 4 years. However, if one prisoner confesses and implicates the other while the other one does not confess, the one who has cooperated with the police, i.e., confessed, will be set free, while the other one will spend 5 years in prison.

In this situation, a non-cooperative game in strategic form can be formulated with the players being the two prisoners, with each prisoner having two strategies: to confess (strategy C) or not to confess (strategy NC). The utility for each prisoner is simply the number of years that will be spent in prison. A matrix representation of this game is given in Table 3.1. The payoffs shown in Table 3.1 are negative numbers since we deal with games where the players seek to maximize a utility. Finally, Table 3.1 clearly shows that this game is a non-zero-sum game.

Example 3.2 (Battle of the Sexes) A husband and wife have to make a choice out of two alternatives for an evening's entertainment: attend a boxing match or watch an opera. The husband prefers the boxing match while the wife prefers the opera, yet both prefer being together rather than being apart. They must decide, simultaneously and without communication, which event to attend. This situation can be modeled as a non-cooperative game in strategic form, where the players are the husband and the wife. The strategy of each player is to select either the boxing match (strategy B) or the opera (strategy O). The payoffs depend on the strategy combination. If each one goes to a different event, then the payoffs received by the wife and the husband are 0 since they both prefer to be together. In contrast, if the husband (wife) manages to convince the wife (husband) and they both choose the boxing match (opera) then the husband (wife) gets a higher payoff than the wife (husband) although both payoffs would be positive. This situation is seen in matrix representation in Table 3.2. For example, if both husband

	Boxing (B)	Opera (O)
Boxing (B)	(1,2)	(0,0)
Opera (O)	(0,0)	(2,1)

Table 3.2 Battle of the sexes (wife, husband).

Table 3.3 Chicken (driver 1, driver 2).

	Straight (ST)	Swerve (S)
Straight (<i>ST</i>)	(-100, -100)	(1,-1)
Swerve (<i>S</i>)	(-1, 1)	(0,0)

and wife select the opera, then the wife gets a payoff of 2, which is higher than that of the husband, which is just 1. By inspecting Table 3.2 one can easily see that this is a non-zero-sum game since the sum of payoffs is not equal for all strategy combinations.

Example 3.3 (The Game of Chicken) Consider two drivers who are driving towards each other on a collision course. One must swerve, or both may die in the crash. However, if one driver swerves and the other does not, the one who swerved will be called a "chicken," i.e., a coward. In this scenario, we can formulate a non-cooperative game in strategic form, with the players being the drivers and the strategies of each player are to swerve (strategy S) or to go straight (strategy ST). A player would win if he goes straight while the other player goes straight, and would crash if both players go straight. The payoffs are shown in the matrix representation in Table 3.3. These are chosen so as to reflect the preferences of the players, which can be described as follows. Each player would prefer to win (payoff of 1) over tying, to tie (payoff of 0) over losing, and to lose (payoff of -1) over crashing (payoff of -100). This game is a non-zero-sum game.

Example 3.4 (Matching Pennies) Consider a situation where two players, player 1 and player 2, must secretly and simultaneously choose to turn a penny to heads or tails. Once the players reveal their pennies, the outcome of the game is as follows:

- If the two pennies match (both are heads or both are tails), then player 2 wins a dollar from player 1.
- If the two pennies are different (one is heads and one is tails), then player 1 wins a dollar from player 2.

This game is a non-cooperative game where each player selects a strategy of either "heads" (strategy H) or "tails" (strategy T). The payoff for each player is either 1,

	Heads (H)	Tails (T)	
Heads (H)	(-1,1)	(1,-1)	
Tails (T)	(1,-1)	(-1,1)	

 Table 3.4 Matching pennies: (player 1 (wins when pennies mismatch), player 2 (wins when pennies match)).

i.e., win a dollar, or -1, i.e., lose a dollar. In Table 3.4 we provide a matrix representation of the game. The rows represent the strategies of player 1, i.e., the player who wins if there's a mismatch. Clearly, since one player loses a dollar to the other player, then for any combination of strategies the total sum of utilities is 0. Thus, Matching Pennies is a two-player zero-sum game. Furthermore, this game is similar to the famous "rock, paper, scissors" game, which can be seen as a three-strategy version of the Matching Pennies game.

In Examples 3.1–3.4, we note that the players have complete information, i.e., they are aware of all the elements of the matrix representation. Once a game is expressed in strategic or matrix form, the next step is to solve it. Solving a game implies predicting the strategies that might be adopted by each player and the possible outcomes of the game. In the remainder of this section, we discuss in detail how to solve non-cooperative games using a variety of concepts.

3.2.2 Dominating strategies

One useful notion for solving non-cooperative games in strategic form is the concept of dominating strategies. The use of dominating strategies simplifies the solution of a game by eliminating some strategies, i.e., rows or columns in a matrix game, which are known from the very beginning to have no effect on the outcome of the game. First, we look at the concept of a dominant strategy:

DEFINITION 3.3 A strategy $s_i \in S_i$ is said to be dominant for player *i* if

$$u_i(s_i, \mathbf{s}_{-i}) \ge u_i(s'_i, \mathbf{s}_{-i}), \ \forall s'_i \in \mathcal{S}_i \ and \ \forall \mathbf{s}_{-i} \in \mathcal{S}_{-i}, \tag{3.1}$$

where $S_{-i} = \prod_{j \neq i} S_j$ is the set of all strategy profiles for all players except *i*. Hence, a dominant strategy is a player's best strategy, i.e., the strategy that yields the highest utility for the player regardless of what strategies the other players choose.

Whenever a player has a dominant strategy, a rational player has no incentive to choose any other strategy. Consequently, if each player possesses a dominant strategy then all players will choose their dominant strategies. This intuitive choice gives rise to the following solution concept for a non-cooperative game:

DEFINITION 3.4 A strategy profile $\mathbf{s}^* \in S$ is the dominant-strategy equilibrium if every element \mathbf{s}_i^* of \mathbf{s}^* is a dominant strategy of player *i*.

The concept of a dominant-strategy equilibrium is a natural outcome for a given game. For example, in The Prisoner's Dilemma (Example 3.1) in Table 3.1, each player has a better payoff by confessing, i.e., choosing *C*, independent of the strategic choice of the other player. Thus, (C, C) is a dominant-strategy equilibrium which yields a payoff vector (-4, -4). Note that although this point is a solution for the game, the payoffs received are not the best for both players. We will revisit the issue of efficiency at the outcome of a game in Section 3.2.6. While the dominant-strategy equilibrium point is not guaranteed. In fact, there are many games in which no player has a dominant strategy. For instance, in Examples 3.2–3.4 one can clearly see that no dominant strategies exist.

Beyond the idea of a dominant strategy, it is useful to define the converse concept, a *strictly dominated strategy*, as follows:

DEFINITION 3.5 A strategy $s'_i \in S_i$ of a player *i* is said to be strictly dominated by a strategy $s_i \in S_i$ if

$$u_i(\mathbf{s}_i, \mathbf{s}_{-i}) > u_i(\mathbf{s}'_i, \mathbf{s}_{-i}), \ \forall \mathbf{s}_{-i} \in \mathcal{S}_{-i}.$$

$$(3.2)$$

Thus, a strategy is strictly dominated for a player if this player has another strategy that performs better, irrespective of what other players choose. Hence, given complete information, it is natural that a rational player eliminates all strictly dominated strategies before making a decision. This leads to the concept of *iterated strict dominance*, which can be used to assist in solving a matrix game. Iterated strict dominance implies eliminating all strictly dominated strategies in a given game. By doing so, we reduce the number of possibilities and, in some cases, we can arrive at a reasonable outcome for the game. In The Prisoner's Dilemma (Example 3.1) in Table 3.1, clearly *NC* is strictly dominated by *C* for both players; thus, by eliminating *NC*, (*C*, *C*) is found to be a reasonable outcome of the game. Iterated strict dominance cannot be used to solve Examples 3.2-3.4 since no player has a strictly dominated strategy. To further highlight the use of iterated strict dominance, we consider the following example:

Example 3.5 Consider the two-player matrix game of Table 3.5. In this game, player 1 has two strategies, either Left (L) or Right (R), while player 2 has four strategies: Left (L), Right (R), Up (U), and Down (D). In order to solve this game by iterated strict dominance, we first look at player 2. Clearly, for player 2, strategy R strictly dominates strategy L, and, hence, player 2 can eliminate strategy L. For player 1, strategy R is strictly dominated by strategies L and D, and, hence, it can be eliminated. Similarly, player 1 can eliminate strategy U since it is strictly dominated by strategy D. After these eliminations, the game reduces to the matrix in Table 3.6.

In the reduced game of Table 3.6, one can see that strategy D strictly dominates strategy L for player 1. Thus, the outcome of the game is (D, R), yielding the payoffs (5, 4).

	L	R
L	(5,1)	(4,2)
R	(3,1)	(3,2)
U	(2,1)	(4,2)
D	(4,3)	(5,4)

Table 3.5 Example of iterated strict dominance (player 1, player 2).

Table 3.6 Reduced game fromExample 3.5 (player 1, player 2).

	R
L D	(4,2)

Example 3.5 illustrates the use of iterated strict dominance for solving matrix games. Although, at first glance, one may think that the order of elimination of strictly dominated strategies can affect the outcome, it is of interest to note that in iterated *strict* dominance *the order of elimination does not affect the outcome of a game*. However, this is not true when using the notion of weak dominance, defined as follows:

DEFINITION 3.6 A strategy $s'_i \in S_i$ of player *i* is said to be weakly dominated by strategy $s_i \in S_i$ if

$$u_i(s_i, \mathbf{s}_{-i}) \ge u_i(s_i', \mathbf{s}_{-i}), \ \forall \mathbf{s}_{-i} \in \mathcal{S}_{-i},$$

$$(3.3)$$

with strict inequality for at least one \mathbf{s}_{-i} .

Using the idea of a weakly dominated strategy, one can also define an *iterated weak dominance* procedure for eliminating weakly dominant strategies. As already mentioned, although in iterated strict dominance the reduced game is unique (independent of the order of elimination), this is not so with iterated *weak* dominance, where the reduced game could be different depending on the order of elimination of the strategies (see [160] for detailed examples).

Although iterated-dominance techniques are compelling and interesting to study, in many cases they are not sufficient to predict the outcome of a game. Nonetheless, it is always of interest to consider the elimination of dominated strategies since, even if no unique outcome can be found through iterated dominance, it can still reduce the strategy space and make it easier to solve the game using other concepts.

3.2.3 Nash equilibrium

As already mentioned, the majority of non-cooperative games are not solvable by iterated dominance, so alternative solution concepts must investigated. In this regard, the most accepted solution concept for a non-cooperative game is that of a *Nash equilibrium*, introduced by John F. Nash in his seminal work [351]. Loosely speaking, a Nash equilibrium is a state of a non-cooperative game where no player can improve its utility by changing its strategy, if the other players maintain their current strategies. Formally, when dealing with pure strategies, i.e., deterministic choices by the players, the Nash equilibrium is defined as follows [58]:

DEFINITION 3.7 A pure-strategy Nash equilibrium of a non-cooperative game $G = (\mathcal{N}, (\mathcal{S}_i)_{i \in \mathcal{N}}, (u_i)_{i \in \mathcal{N}})$ is a strategy profile $\mathbf{s}^* \in \mathcal{S}$ such that $\forall i \in \mathcal{N}$ we have the following:

$$u_i(s_i^*, \mathbf{s}_{-i}^*) \ge u_i(s_i, \mathbf{s}_{-i}^*), \ \forall s_i \in \mathcal{S}_i.$$

$$(3.4)$$

In other words, a strategy profile is a pure-strategy Nash equilibrium if no player has an incentive to *unilaterally* deviate to another strategy, given that other players' strategies remain fixed. In the event where we have $u_i(s_i^*, \mathbf{s}_{-i}^*) > u_i(s_i, \mathbf{s}_{-i}^*)$, $\forall s_i \in S_i$, $s_i \neq s_i^*$, $\forall i \in \mathcal{N}$, the Nash equilibrium is said to be *strict*.

With this definition, we can check whether a Nash equilibrium solution can be found for Examples 3.1–3.4 by studying possible deviations by the players for each combination of strategies. For instance, in Example 3.1, by inspecting Table 3.1 we can easily find that (C, C) is the only Nash equilibrium of this game. Hence, in The Prisoner's Dilemma, as shown in the previous subsection, the Nash equilibrium can be found using iterated dominance or the dominant-strategy equilibrium. In Example 3.2, by looking at Table 3.2 it can be seen that attending either event, i.e., the opera (O, O) or the boxing match (B, B) is a pure-strategy Nash equilibrium. (O, O) is a Nash equilibrium because, from the husband's perspective, if the wife decides on the opera, the husband's payoff would move from 1 to 0 if he decides to unilaterally deviate and go to the boxing match. Similarly, from the wife's perspective, if the husband decides on going to the opera, the wife will also choose to go to the opera, since otherwise her payoff would drop from 2 to 0 if boxing is selected. By similarly checking the possible deviations at each outcome of Example 3.3, one can see that (S, ST) with payoffs (-1, 1) and (ST, S)with payoffs (1, -1) are pure-strategy Nash equilibria. Hence, in the Game of Chicken, the pure-strategy Nash equilibria dictate that one driver swerves while the other driver continues straight.

By closely looking at Example 3.4, one can see that, at any outcome, a player would have an incentive to deviate and change its strategy from heads to tails or vice versa. Consequently, the Matching Pennies game admits *no* pure-strategy Nash equilibrium. For example, if player 1 plays heads, player 2 has an incentive to play heads as well, since it can achieve a payoff of 1. However, when player 2 plays heads, player 1 has an incentive to play tails since (H, H) gives player 1 a payoff of -1 while (T, H) gives it a payoff of 1. Similarly, when player 1 plays tails, player 2 has an incentive to play tails. In the same manner, one can verify that for any combination of strategies, one of the players has an incentive to unilaterally deviate. Consequently, this game has no pure-strategy Nash equilibrium.

From these examples, we can deduce the following statements regarding the concept of a pure-strategy Nash equilibrium:

- Existence and multiplicity A non-cooperative game can admit zero, one, or multiple Nash equilibria.
- Efficiency A Nash equilibrium is not necessarily the best outcome, from the perspective of payoff.

Hence, when studying the Nash equilibria of a game, the key points of interest are existence, multiplicity, and efficiency. The existence and multiplicity of Nash equilibria have been previously shown in Examples 3.1–3.4. For instance, games such as The Prisoner's Dilemma admit a unique pure strategy Nash equilibrium, games such as the Game of chicken admit multiple Nash equilibria, while a game such as Matching Pennies has *no* pure-strategy Nash equilibrium.

With regard to efficiency, in The Prisoner's Dilemma, as mentioned in the previous subsection, the pure-strategy Nash equilibrium of the game (-4, -4) is inefficient. For instance, the two prisoners could do better, i.e., achieve (-2, -2), if both choose not to confess; however, this outcome is not stable in a non-cooperative setting, i.e., not an equilibrium point, since unilateral deviations are possible. This shows that, although cooperating by not confessing will give each player a better payoff of -2, the greediness of each prisoner leads to an inefficient outcome. This demonstrates that the pure-strategy Nash equilibrium solution of a non-cooperative game can be inefficient. In Section 3.2.6 we discuss further the issue of equilibrium efficiency.

3.2.4 Static continuous-kernel games

Heretofore, we mainly dealt with non-cooperative games where the strategy spaces S_i , $\forall i \in \mathcal{N}$ are discrete and finite sets. In many scenarios, notably in wireless and communication applications, the action space available to each player is a continuum and the utility functions are continuous. For example, in wireless networks, it is common to find action sets that represent the transmit-power intervals that a node must choose. In economics, firms must select a price from a continuous are of central interest in many applications. In this section, we deal with *static* games where the action (strategy) sets have uncountably many elements, such as subsets of a finite-dimensional Euclidean space, and the payoff functions are continuous on these sets. Such games are generally known as static *continuous-kernel games*.

In a static continuous-kernel game, each strategy (equivalently, action) set S_i is an element of a finite dimensional space. Most commonly, the strategies are intervals, or unions of subintervals, of the real line. As with the discrete finite strategies case, continuous-kernel games can be represented in strategic (normal) form. However, one cannot use a matrix representation for such games since each strategy space is a continuum. In solving any non-cooperative game, especially for continuous-kernel games, the concept of a *best-response function* is useful, and is defined as follows:

DEFINITION 3.8 The best-response function $b_i(\mathbf{s}_{-i})$ of a player *i* to the profile of strategies \mathbf{s}_{-i} is a set of strategies for that player such that

$$b_i(\mathbf{s}_{-i}) = \{ s_i \in \mathcal{S}_i \mid u_i(s_i, \mathbf{s}_{-i}) \ge u_i(s'_i, \mathbf{s}_{-i}), \ \forall s'_i \in \mathcal{S}_i \}.$$
(3.5)

Hence, for a player *i*, when the strategies of the other players are fixed as \mathbf{s}_{-i} , any strategy in $b_i(\mathbf{s}_{-i})$ is at least as good as every other available strategy in \mathcal{S}_i . The bestresponse function is *set-valued* as it associates, for a player *i*, a set of strategies with any strategy profile \mathbf{s}_{-i} of the other players. Every element of the best-response function $b_i(\mathbf{s}_{-i})$ is a best-response of player i to \mathbf{s}_{-i} . In other words, the best response of a player i implies that, if each of the other players adheres to \mathbf{s}_{-i} , then player *i* cannot do better than to choose a member of $b_i(\mathbf{s}_{-i})$. In Example 3.1, the best-response functions for the first prisoner if the second prisoner confesses or does not confess are, respectively, $b_1(C) =$ $\{C\}$ and $b_1(NC) = \{C\}$. The same best-response functions can be found for the second prisoner. Hence, in The Prisoner's Dilemma, each prisoner has a single best response to any action of the other prisoner. In contrast, if we consider Example 3.2, the bestresponse function for the wife if the husband chooses boxing is $b_1(B) = \{B\}$; in contrast, if the husband chooses opera, the best response of the wife would be $b_1(O) = \{O\}$. This highlights that, for a given player, the best-response function need not lead to a unique action for any strategy of the other players (that is, it need not be a constant function). Moreover, although in these two examples each best-response function consists of a single element, many games exist where a best-response function can have multiple elements (that is, be non-unique as a function).

The concept of a best-response function leads to an alternative characterization of a pure-strategy Nash equilibrium [160]:

PROPOSITION 3.1 A strategy profile $\mathbf{s}^* \in S$ is a Nash equilibrium of a non-cooperative game if and only if every player's strategy is a best response to the other players' strategies; that is,

$$s_i^* \in b_i(\mathbf{s}_{-i}^*)$$
 for every player i. (3.6)

Hence, a Nash equilibrium is a strategy profile for which every player's strategy is a best response to the other players' strategies. Note that this definition applies to any type of game and not just to continuous-kernel games.

If, for every strategy profile \mathbf{s}_{-i} , each player *i* has a *single* best response, then (3.6) can be written as the following *N* equations (in this case every $b_i(\mathbf{s}_{-i})$ is a singleton):

$$s_i^* = b_i^* (\mathbf{s}_{-i}^*)$$
, for every player $i = 1, ..., N$. (3.7)

To find the Nash equilibria of a non-cooperative game, one can use the following method, which is particularly useful for continuous-kernel games:

- 1. Find the best-response function of each player.
- 2. Find the strategy profiles satisfying (3.6), which reduces to the collection of equations (3.7) in the case where every player *i* has a single best response to each strategy profile \mathbf{s}_{-i} of the other players.

To illustrate this method, we consider the following example:

Example 3.6 (The Relationship Game) Two players are involved in a relationship whereby, if both players devote more effort to the relationship, they are both better off. For any given effort of player j, the return on individual i's effort starts by increasing, and then decreases. We represent an effort level by a non-negative number. For every effort level s_i , the utility function for any player i is defined as $u_i(s_i, s_j) = s_i(c + s_j - s_i)$, where s_j is the effort level of player j and c > 0 is a constant. Consequently, we formulate a non-cooperative game between the two players where each strategy set S_i is the interval of non-negative numbers and the payoffs are given by u_i , i = 1, 2.

To find the Nash equilibria we construct the best-response functions by setting, for every player *i*, the derivative of u_i with respect to s_i to 0, i.e., $\frac{\partial u_i(s_i,s_j)}{\partial s_i} = 0$, i = 1, 2, which yields

$$b_1(s_2) = \frac{1}{2}(c+s_2), \qquad (3.8)$$

$$b_2(s_1) = \frac{1}{2}(c+s_1). \tag{3.9}$$

In Fig. 3.1, we plot the best-response functions of the players in (3.8) and (3.9). The strategies of player 1 are shown on the x-axis while those of player 2 are shown on the y-axis. Consequently, the best-response function b_1 of player 1 associates a unique strategy for player 1 to each strategy of player 2. Similarly, the best-response function b_2 associates a unique strategy for player 2 to each strategy of player 1. In Fig. 3.1 we can see that the two best-response functions intersect at a unique point (c, c). This point, in fact, constitutes the unique pure-strategy Nash equilibrium $(s_1^*, s_2^*) = (c, c)$ of the game since, at this point, every player's strategy is a best response to the other player's strategies, i.e., $s_1^* = b_1(s_2^*)$ and $s_2^* = b_2(s_1^*)$. Although for the Relationship Game the best-response functions intersect at a single point, in general continuous-kernel games, they may intersect at more than one point, i.e., multiple Nash equilibrium.

Algebraically, by setting $s_1^* = b_1(s_2^*)$ and $s_2^* = b_2(s_1^*)$, from (3.8) and (3.9) one would obtain

$$s_1^* = \frac{1}{2}(c + s_2^*), \tag{3.10}$$

$$s_2^* = \frac{1}{2}(c + s_1^*).$$
 (3.11)



Fig. 3.1 The players' best-response functions in the Relationship Game of Example 3.6.

By substituting (3.11) into (3.10), we get $s_1^* = \frac{3}{4}c + \frac{1}{4}s_1^*$ which yields $s_1^* = c$. Substituting into (3.11), we get $s_2^* = c$. Thus, we can conclude that the game has a unique Nash equilibrium $(s_1^*, s_2^*) = (c, c)$, which corroborates the result seen graphically in Fig. 3.1.

As shown in Example 3.6, whenever closed-form expressions for the best-response functions can be found, the pure-strategy Nash equilibria (and their existence) of a non-cooperative game can be found by checking the intersection of these best-response functions. Moreover, whenever expressions for the best-response functions can be explicitly found, one can show the uniqueness of the pure-strategy Nash equilibrium through the concept of a standard function:

DEFINITION 3.9 A function $g : S \to \mathbb{R}^N_+$ is said to be standard if it has the following properties:

- Monotonicity: $\forall \mathbf{s}, \mathbf{s}' \in S$, $\mathbf{s} \leq \mathbf{s}' \Rightarrow g(\mathbf{s}) \leq g(\mathbf{s}')$ (component-wise).
- Scalability: $\forall \alpha > 0, \mathbf{s} \in S, g(\alpha \mathbf{s}) \leq \alpha g(\mathbf{s}).$

Applying the result that a standard function has a unique fixed point to the best response of a non-cooperative game, we present the following theorem (introduced and proved in [519]):

THEOREM 3.1 [519] If the best-response functions of a non-cooperative game G are standard functions for all players, i.e., $\forall i \in N$, then the game has a unique Nash equilibrium in pure strategies.

This result is mainly useful whenever the best responses can be written in closed form. However, in numerous problems, writing down these functions can be quite complex [58, 160, 283]. In fact, proving the existence of pure-strategy Nash equilibria and finding these equilibria in a generic non-cooperative game is a challenging task [160]. For this purpose, beyond the use of best-response functions, some theorems exist for characterizing whether a Nash equilibrium in pure strategies exists in a given game, notably for continuous-kernel games. One of the most popular of such theorems is the following [167, 141, 125]:

THEOREM 3.2 [167, 141, 125] Given a non-cooperative game in strategic form $(\mathcal{N}, (S_i)_{i \in \mathcal{N}}, (u_i)_{i \in \mathcal{N}})$, if $\forall i \in \mathcal{N}$, every strategy set S_i is compact and convex, $u_i(s_i, \mathbf{s}_{-i})$ is a continuous function in the profile of strategies $\mathbf{s} \in S$ and quasi-concave in s_i , then the game has at least one pure-strategy Nash equilibrium.

In other words, by proving certain properties on the strategy sets and the utility functions, one can show the existence of a pure-strategy Nash equilibrium. Recall that a function *f* is quasi-concave on a convex set *S* if, for all $\alpha \in \mathbb{R}$, the upper contour set $C = \{x \in S, f(x) \ge \alpha\}$ is convex. Although Theorem 3.2 is useful for showing the existence of pure-strategy Nash equilibria, it does not give any idea of the number of these equilibria. Interestingly enough, Rosen showed that, under certain conditions, a *unique* pure-strategy Nash equilibrium exists, as per the following theorem [402]:

THEOREM 3.3 [402] Consider a strategic game G where $\forall i \in N$, every strategy set S_i is compact and convex, $u_i(s_i, \mathbf{s}_{-i})$ is a continuous function in the profile of strategies $\mathbf{s} \in S$ and concave in s_i . Let $\mathbf{r} = (r_1, ..., r_N)$ be an arbitrary vector of fixed positive parameters. If the diagonal strict concavity (DSC) property holds true, i.e.,

$$\exists \mathbf{r} > 0 : (\mathbf{s} - \mathbf{s}')(g(\mathbf{s}, \mathbf{r}) - g(\mathbf{s}', \mathbf{r})) > 0, \ \forall \mathbf{s}, \mathbf{s}' \in \mathcal{S}, \ \mathbf{s} \neq \mathbf{s}',$$
(3.12)

with $g(\mathbf{s}, \mathbf{r}) \triangleq [r_1 \frac{\partial u_1(\mathbf{s}_1, \mathbf{s}_{-1})}{\partial \mathbf{s}_1}, \dots, r_N \frac{\partial u_N(\mathbf{s}_N, \mathbf{s}_{-N})}{\partial \mathbf{s}_N}]^T$, then the game has a unique pure-strategy Nash equilibrium.

Theorem 3.3 provides a powerful result on the uniqueness of the Nash equilibrium in pure strategies, which has been used in a variety of problems, notably in wireless and communication networks, such as the water-filling game of [277] and the power-control game in multiple-access channels [66]. Nonetheless, as discussed in [283], in complex problems, proving the DSC property in complicated scenarios can be restrictive because this condition needs to be satisfied for all strategy profiles in $\mathbf{s} \in S$. Some other conditions, based on the contraction properties of best-response maps, lead to the uniqueness of the Nash equilibrium as well as iterative computational algorithms [291].

3.2.5 Mixed strategies

So far, the main focus in the study of strategic games has been on *pure strategies* and *pure Nash equilibria*. As mentioned earlier, a pure strategy is a *deterministic* selection of

a strategy by a given player. In general, a player may be able to select each pure strategy with a certain probability, which is the basis of the concept of a *mixed strategy*. For a given player, a mixed strategy consists of a number of possible moves and a probability distribution (collection of weights) which corresponds to how frequently each move would be selected by the player.

Given a strategic game $(\mathcal{N}, (\mathcal{S}_i)_{i \in \mathcal{N}}, (u_i)_{i \in \mathcal{N}})$, for each player *i* we define Σ_i as the set of probability distributions over its set of strategies \mathcal{S}_i . A mixed strategy $\sigma_i(s_i) \in \Sigma_i$ of player *i* is a probability distribution over the pure strategies $s_i \in \mathcal{S}_i$. For example, whenever the set \mathcal{S}_i is finite, then σ_i is a probability mass function of the pure strategies. Given the profile of mixed strategies $\sigma \in \Sigma = \prod_{i=1}^{N} \Sigma_i$ and assuming that the pure strategy sets \mathcal{S}_i are finite, we let $\sup(\sigma_i) = \{s_i \in \mathcal{S}_i | \sigma_i(s_i) > 0\}$ denote the support of the set of strategies which are assigned positive probabilities. Consequently, the payoff for a mixed strategy corresponds to the expected value of the pure-strategy profiles in its support, i.e.,

$$u_i(\boldsymbol{\sigma}) = \sum_{\mathbf{s}\in\mathcal{S}} \left(\prod_{j=1}^N \sigma_j(s_j)\right) u_i(s_i, \mathbf{s}_{-i}), \qquad (3.13)$$

where $u_i(s_i, \mathbf{s}_{-i})$ is the pure-strategy payoff pure for a strategy *N*-tuple (s_i, \mathbf{s}_{-i}) . We can now define the concept of a *mixed-strategy Nash equilibrium* MSNE:

DEFINITION 3.10 A mixed-strategy profile $\sigma^* \in \Sigma$ is a mixed-strategy Nash equilibrium *if, for each player i* $\in N$, we have:

$$u_i(\sigma_i^*, \boldsymbol{\sigma}_{-i}^*) \ge u_i(\sigma_i, \boldsymbol{\sigma}_{-i}^*), \ \forall \sigma_i \in \Sigma_i$$
(3.14)

Note that a pure-strategy Nash equilibrium can also be considered as a mixed-strategy Nash equilibrium with a mixed-strategy profile in which each player selects one strategy with probability 1 (a pure strategy) while assigning zero probability to all other strategies. Because of this, a mixed-strategy equilibrium which assigns positive probability to at least two strategies of at least one player is given a name, a *proper mixed Nash equilibrium*.

Furthermore, given that $u_i(\sigma_i, \sigma_{-i}^*) = \sum_{s_i \in S_i} \sigma_i(s_i) u_i(s_i, \sigma_{-i}^*)$, we can see that, when determining whether a mixed-strategy profile is a Nash equilibrium or not, it is sufficient to check only *deviations in pure strategies*, as follows [160]:

PROPOSITION 3.2 A mixed-strategy profile $\sigma^* \in \Sigma$ is a mixed-strategy Nash equilibrium *if, for each player i* $\in N$ and for each $s_i \in S_i$, we have:

$$u_i(\sigma_i^*, \boldsymbol{\sigma}_{-i}^*) \ge u_i(s_i, \boldsymbol{\sigma}_{-i}^*). \tag{3.15}$$

To solve for the mixed-strategy Nash equilibrium, one can use the following lemma [160]:

LEMMA 3.1 A strategy profile $\sigma^* \in \Sigma$ is a mixed-strategy Nash equilibrium if, and only if, for each player $i \in N$, the following two conditions hold:

- 1. The expected payoff given σ_{-i}^* to every $s_i \in \sup(\sigma_i^*)$ is the same.
- 2. The expected payoff given σ_{-i}^* to the strategies which are not in the support of σ_i^* must be less than or equal to the expected payoff in (3.13).

In other words, this lemma implies that a mixed-strategy profile $\sigma^* \in \Sigma$ is a mixedstrategy Nash equilibrium if and only if, for each player $i \in N$, every pure strategy in the support of σ_i^* is a best response to σ_{-i}^* . In fact, if the strategies in the support have different payoffs, then one can just pick the pure strategy with the highest expected payoff, which would contradict the fact that σ^* is a Nash equilibrium. In some sense, this lemma means that, at a mixed-strategy Nash equilibrium, the players would be indifferent as to their pure strategies contributing to their support set, i.e., the expected payoffs at those pure strategies are equal.

To illustrate the idea of a mixed-strategy Nash equilibrium as well as the use of Lemma 3.1 to find the equilibrium, we reconsider Examples 3.2–3.4. In the Battle of Sexes game of Example 3.2, we showed that (B, B), with payoffs (1, 2), and (O, O), with payoffs (2, 1), are pure-strategy Nash equilibria. Now, assume that the wife picks the strategy *B* with probability *p* and *O* with probability 1 - p, while the husband picks the strategy *B* with probability *q* and *O* with probability 1 - q. Using Lemma 3.1 on the wife's strategies, we obtain the following:

$$1 \cdot q + 0 \cdot (1 - q) = 0 \cdot q + 2 \cdot (1 - q). \tag{3.16}$$

Next, applying Lemma 3.1 to the husband's strategies, we obtain:

$$2 \cdot p + 0 \cdot (1 - p) = 0 \cdot p + 1 \cdot (1 - p). \tag{3.17}$$

Consequently, we conclude that the only possible proper mixed-strategy Nash equilibrium for Example 3.2 is $p = \frac{1}{3}$ and $q = \frac{2}{3}$, which is the solution to (3.16) and (3.17). Using a similar approach for Example 3.3, we find that the only possible proper mixed-strategy equilibrium indicates that each driver plays *S* with probability 0.99 and *ST* with probability 0.01. In the Matching Pennies game of Example 3.4, using Lemma 3.1 we find that, although *no pure-strategy Nash equilibrium* exists, we have a single proper mixed-strategy Nash equilibrium where both players randomize with probability $\frac{1}{2}$ on heads and $\frac{1}{2}$ on tails.

The fact that, in Example 3.4, a mixed-strategy Nash equilibrium exists although no pure-strategy equilibrium exists highlights the most important and crucial result of game theory, which is the following theorem by Nash [351]:

THEOREM 3.4 [351] Every finite non-cooperative game in strategic form has a mixedstrategy Nash equilibrium. The proof of this result relies on the Brouwer–Kakutani Fixed-Point Theorem, and can be found in [58, 351]. The result of Theorem 3.4 deals primarily with *finite* strategy spaces. Nonetheless, by a generalization of the Brouwer–Kakutani Fixed-Point Theorem, Glicksberg extended the result to the case where the strategy spaces S_i are non-empty compact metric spaces while the utility functions are continuous functions, through the following theorem [160]:

THEOREM 3.5 [167] A strategic game $(\mathcal{N}, (\mathcal{S}_i)_{i \in \mathcal{N}}, (u_i)_{i \in \mathcal{N}})$, where \mathcal{S}_i are non-empty compact metric spaces and u_i , $\forall i \in \mathcal{N}$ are continuous functions, has a mixed-strategy Nash equilibrium.

To conclude, the Nash equilibrium in mixed or pure strategies provides a powerful solution concept for non-cooperative strategic games which has revolutionized game theory since the work of Nash [351]. As will be seen in the rest of this book, many applications, concepts, and classes of games deal with models and solutions that, in one way or another, rely on concepts inspired by the Nash equilibrium.

3.2.6 Efficiency and equilibrium selection

In the previous subsections, we mainly studied the existence and characterization of Nash equilibria, in both pure and mixed strategies. For instance, Theorem 3.4 states that, in a broad class of games, there always exists at least a mixed-strategy Nash equilibrium. Nonetheless, once we have verified the existence (for pure strategies) and number of equilibria, it is important to select an equilibrium that is desired in the game, e.g., optimal or efficient. One important measure of efficiency can be found in the concept of Pareto optimality, defined as follows:

DEFINITION 3.11 A strategy profile $\mathbf{s} \in S$ is Pareto-superior to another strategy profile $\mathbf{s}' \in S$ if, for every player $i \in N$, we have:

$$u_i(s_i, \mathbf{s}_{-i}) \ge u_i(s'_i, \mathbf{s}'_{-i}),$$
 (3.18)

with strict inequality for at least one player. Accordingly, a strategy profile $s^{\circ} \in S$ is Pareto-optimal if there exists no other strategy profile that is Pareto-superior to s° .

Thus, the outcome of a game is Pareto-optimal if there is no other outcome that makes every player at least as well off and at least one player strictly better off. Hence, a Pareto-optimal outcome cannot be improved upon without hurting at least one player. As a result, in games where a large number of Nash equilibria exist, it is desirable to select a Pareto-optimal equilibrium, if possible. Nonetheless, one would note that often a Nash equilibrium is not Pareto-optimal. For example, in Example 3.1, one Paretooptimal point is the point (*NC*, *NC*), i.e., (-2, -2), because one cannot improve the payoff for one prisoner without decreasing the payoff for the other (e.g., by moving from (-2, -2) to (-5, 0), prisoner 2 improves while prisoner 1 has a worse performance). Note that points (-5, 0) and (0, -5) are also Pareto-optimal points. However, none of the Pareto-optimal points of this game constitutes an equilibrium. Also, we note that the Nash equilibrium (-4, -4) of Example 3.1 is inefficient since we can improve the payoffs for both prisoners by moving to the point (-2, -2), which is Pareto-optimal and better for both prisoners when compared with (-4, -4).

Beyond Pareto optimality, one useful notion for evaluating the performance of a Nash equilibrium (or any non-cooperative equilibrium) is the *price of anarchy*. The price of anarchy is defined as the ratio of the maximum social welfare, i.e., the total utility, achieved by a centralized or genie-aided solution to the social welfare achieved at the worst-case equilibrium. Hence, let S^{NE} be the set of Nash equilibrium strategy profiles of a given game. Define

$$u_{\rm NE} = \min_{\mathbf{s}\in\mathcal{S}^{\rm NE}} \left(\sum_{i\in\mathcal{N}} u_i(\mathbf{s}) \right) \tag{3.19}$$

as the social welfare at the worst-case Nash-equilibrium and

$$u_{\rm CS} = \max_{\mathbf{s}\in\mathcal{S}} \left(\sum_{i\in\mathcal{N}} u_i(\mathbf{s}) \right) \tag{3.20}$$

as the maximum social welfare achieved, by a centralized solution, for example. Hence, we can define the price of anarchy η as [382, 360]

$$\eta = \frac{u_{\rm CS}}{u_{\rm NE}}.\tag{3.21}$$

The *price of stability* is defined similarly to the price of anarchy η but replacing the denominator with the best Nash equilibrium. The two concepts are equivalent if there exists a unique Nash equilibrium. Using the price of anarchy, one can evaluate the performance of the Nash equilibria with respect to the solution that maximizes the social welfare. In particular, in some games, one can find upper bounds on the price of anarchy using techniques such as variational inequalities [283, 360]. For example, in simple scenarios where each player has a cost function that is affine, the upper bound on the price of anarchy is $\frac{4}{3}$ [360]. In brief, the price of anarchy can be an interesting metric to evaluate the performance of equilibria in a given game, and this concept has been especially used in routing and flow-control games (see [360]).

Finally, we note that, in general, no formal rules exist for selecting an efficient equilibrium, although concepts such as Pareto optimality and the price of anarchy can be useful in this context. For particular games, namely in wireless and communication networks, some approaches such as pricing or the introduction of hierarchy can be used for efficient equilibrium selection [283]. For a detailed survey of these techniques, we refer the reader to [377]. We stress that the topic of equilibrium selection is of central interest in game theory [58, 377, 419] as well as in wireless and communication networks [283]. This topic will be further explored in Part III of this book through examples of games that arise in wireless and communication networks.

3.3 Dynamic non-cooperative games

Dynamic non-cooperative games are games where the sequence of strategic choices made by the players, as well as the information known (or gathered) by the players on the other players' decisions, strongly impact the outcome of the game. Unlike in static games, players in a dynamic game have at least some information about the actions chosen by the others; thus their play may be contingent on past moves. For example, in a dynamic non-cooperative game, based on the history of selected actions and the threats from other players one can encourage cooperation without the need for communication among the players themselves. Therefore, in dynamic games, as emphasized in Section 3.1.2, there is a clear distinction between the action of a player, i.e., the set of options that a player can select, and the strategy of the player, i.e., the mapping between the information available to the player and its action set. This distinction will be made explicit throughout this section.

In this section, we study three main classes of dynamic games. First, we discuss the framework of dynamic games represented in extensive form. Then, we study the main concepts of repeated games. We conclude this section with an overview of stochastic games.

3.3.1 Non-cooperative games in extensive form

Sequential games constitute a major class of dynamic games in which players take their decisions (select a strategy) in a certain predefined order. In a sequential game, some of the players are able to observe the moves of players who acted before them, and make strategic choices accordingly. Therefore, each player can devise its *strategy*, given information available on the *actions* of the other players. Note that a static game is a particular case of a sequential game where no player is able to observe the moves of the others.

Thus, in a sequential game, the role of information, i.e., the moves known by the players, is of central importance. In this regard, for dynamic sequential games, one can distinguish between two types of information knowledge: perfect information and imperfect information. A sequential game has *perfect information* if only one player moves at a time and if each player knows every action of the players that moved before it at every point of the game. Intuitively, if it is a player's turn to select an action, a perfect-information game assumes that this player is always aware of what every other player has done up to that point. In contrast, a sequential game has *imperfect information* whenever some of the players do not know all the previous choices of the other players. It is common in many scenarios that, whenever a player's turn to move is reached, this player may need to take a decision without full knowledge of every single action taken by the other players prior to its turn.

It must be stressed that the notion of perfect or imperfect information is *quite different* from the notions of complete and incomplete information discussed in Section 3.2.1. While the notion of complete information is concerned with the information that every

player has on the *elements of the game*, e.g., the strategy space, the possible payoffs, and so on, the notion of perfect information is concerned with the information that a player has on *the actions taken by the other players* or their sequence.

While dynamic games can be represented in strategic (normal) form, because of the presence of sequential decision-making it is of interest to utilize an alternative representation that can clearly highlight the notion of time or sequence of moves. In this context, one of the most useful representations of a dynamic sequential game is the extensive form (or game tree). A game in extensive form is a graphical representation of a non-cooperative dynamic game which provides not only a representation of the players, payoffs, and their actions, but also a representation of the order of moves (or sequence) and the information sets. A game tree consists of graph nodes, which are the points at which players can take actions, connected by edges, which represent the moves that may be taken by a player present at a certain node. An initial (or root) node represents the *first decision* to be made by one of the players (the root node is typically represented as the upper node in the game tree and play progresses from the upper node all the way down to the terminal nodes). For every player, a number of nodes might be enclosed by dotted lines which represent the *information sets*, i.e., the information available to the player at the time of play. In a game with perfect information, every information set contains exactly one node since each player knows *exactly* all the information on the actions of the previous player. In contrast, in a game with imperfect information, there exists at least one information set containing more than one node. A single level of the tree is referred to as a stage. Every set of vertices from the first node through the tree eventually arrives at a terminal node, representing an end to the game. Each terminal node is labeled with the payoffs earned by each player if the game ends at that node. At a given node in a given stage of the tree, *history* refers to the sequence of actions that were taken up to the considered stage. In other words, at a given node in a given stage, a history is the set of actions, i.e., the path in the tree, taken from the initial node up to the considered node.

We note that a game in extensive form can always be converted into the strategic form after finding the different strategies of the players, given their possible actions and the available information (recall that a strategy is a mapping from the information sets to the action sets of a player). Moreover, dynamic games where the players have finite action sets and act only a finite number of times are equivalent, in strategic form, to finite static games, and hence to matrix games.

An illustrative example of a game in extensive form is shown in Fig. 3.2(a) and (b). The two trees shown in Fig. 3.2 represent a two-player game: player 1 and player 2. The numbers noted at every terminal node represent the payoffs to the players (e.g., (2, 1) represents a payoff of 2 to player 1 and a payoff of 1 to player 2). The labels by every edge of the graph are the name (label) of the strategy that each edge represents. In this game, each player has a choice between two actions, Up (*U*) or Down (*D*). In Fig. 3.2(a), player 2 does not know at which node it is, so both nodes are in the *same* information set. Accordingly, Fig. 3.2(a) represents a game in extensive form with imperfect information. In contrast, in Fig. 3.2(b), player 2 knows exactly at which point in the game tree it will be acting, so this game is of perfect information.



Fig. 3.2 (a) Imperfect information; (b) perfect information.

Table 3.7 Matrix form of Fig. 3.2(a)(player 1, player 2).

	D	U	
D	(3,1)	(1,2)	
U	(2,1)	(0,0)	

Because player 2 is not aware of its position on the tree, Fig. 3.2(a) can be reduced into the matrix game of Table 3.7. By careful inspection of Table 3.7, we can see that this game admits a unique Nash equilibrium (D, U) with payoffs (1, 2). In this game, there is no role for information in the decision of player 2, owing to the lack of this information.

The game shown in Fig. 3.2(b) has perfect information and progresses as follows. First, player 1, which owns the initial node, selects its action first. Once player 1 makes a decision between action U and action D (at the first stage), player 2, acting second, would decide between U or D depending on the action chosen by player 1. Hence, while for player 1 we can simply define two strategies U or D, for player 2 the strategies need to be a function of the action taken by player 1, i.e., the information available on the moves of player 1. Hence, the set of possible strategies for player 2 are UU, UD, DU, and DD, where the first (second) element of the pair represents the strategy chosen by player 2 when player 1 chooses its first (second) action. For example, DU means player 2 selects D if player 1 plays U, and player 2 selects U if player 1 selects D. If we denote by a_1 the action taken by player 1 at the initial node, the strategy $s_2(a_1) = DU$ of player 2, given the action of player 1, can be written as

For each combination of strategies, the payoffs received for each sequence of play are as specified at the terminal nodes of the tree in Fig. 3.2(b). We note that the game in Fig. 3.2(b) can be converted into a matrix game (in strategic form) where player 1 has two strategies, U and D, while player 2 has four strategies, UU, UD, DU, and DD.

In dynamic games, one can distinguish between two types of games: single-act games and multi-act games. In a single-act game, each player makes a decision only *once*. In contrast, in a multi-act game, a player is allowed to act more than once. Both games of Fig. 3.2 are single-act games.

In order to solve dynamic games with perfect information, one can use the concepts defined in Section 3.2, mainly the Nash equilibrium. For example, in Fig. 3.2(b), (U, DU) and (D, UU) are pure-strategy Nash equilibria. To see how these strategies are equilibria, let us investigate the possible deviations if strategy (U, DU) is played. When player 1 plays U, player 2 plays D and the payoffs achieved are (2, 1). If player 1 deviates by choosing D, then player 2 plays U and the payoffs achieved are (1, 2); hence, player 1 has no incentive to deviate since it prefers to obtain 2 over 1. For player 2, given that player 1 selected U, player 2 can select either DU or DD since in both cases it will receive the payoff of 1 (since player 1's strategy is fixed to U). However, (U, DD) is not a Nash equilibrium, since, if player 2 plays DD, player 1 prefers to play D over U since (D, DD) yields a payoff of 3 for player 1 while (U, DD) yields a payoff of 2. In contrast, (U, DU) is a Nash equilibrium since, given that player 2 plays DU, player 1 prefers to play DU, player 1 prefers to play DI, player 1 prefers to play DI. Since it receives 2 while playing D, i.e., (D, DU) yields only a payoff of 1 for player 1. Similar reasoning can be used to show that (D, UU) is also a Nash equilibrium.

One useful method for finding equilibria in a dynamic game (in extensive form) with perfect information is through the use of *backward induction*. Backward induction is an iterative technique similar to dynamic programming, which can be useful for solving finite single-act extensive-form games [160]. In backward induction, one first determines the optimal choice of the player who makes the last move of the game. Then, the optimal action of the player moving next-to-last is determined, taking the last player's action as given. The process continues this way backwards in time until all players' actions have been determined. For example, in Fig. 3.2(b), we start by finding the optimal strategy of player 2. Clearly, if player 1 selects *D* then player 2 would select *U*, and if player 1 selects *U* then player 2 would select *D*. Hence, the optimal strategy of player 2 is *UD*. By anticipating this result, player 1 would then decide to select *U*, and the backward induction solution would be (U, D).

Using backward induction for extensive-form games with perfect information, Kuhn proved the following result [258, 160]:

THEOREM 3.6 [258] Every finite extensive-form game with perfect information has a pure-strategy Nash equilibrium.

While dynamic games with perfect information can be solved in extensive form by backward induction, this method cannot be used for games with imperfect information. In addition, for dynamic games (notably when the game is one with imperfect information) one would need an equilibrium solution which requires the strategy of each player to be

optimal not only at the start of the game (such as in the Nash equilibrium) but also after every history. This leads to the concept of a subgame-perfect equilibrium [160]:

DEFINITION 3.12 A subgame of a dynamic non-cooperative game consists of a single node in the extensive-form representation of the game, i.e., the game tree, and all of its successors down to the terminal nodes. The information sets and payoffs of a subgame are inherited from the original game. Moreover, the strategies of the players are restricted to the history of actions in the subgame.

DEFINITION 3.13 A strategy profile $\mathbf{s} \in S$ is a subgame-perfect equilibrium if players' strategies (restricted to a subgame) constitute a Nash equilibrium in every subgame of the original game.

To find a subgame-perfect equilibrium, it is generally useful to refer to the extensive form of the game, find the subgames, and then identify their Nash equilibria. Nonetheless, finding the subgame-perfect equilibrium requires checking every subgame of a dynamic game, which can be a tedious task. For this purpose, one can use the one-stage deviation principle, defined as follows [160]:

DEFINITION 3.14 The one-stage deviation principle requires that there must not exist any information set in which a player i can gain by deviating from its subgame-perfect equilibrium strategy (at this information set) while its strategy at other information sets as well as the strategies of the other players are fixed.

In other words, a strategy profile s^* is a subgame-perfect equilibrium for each player $i \in N$ and at each information set where player *i* moves if we:

- fix the other players' strategies as **s***.
- fix player *i*'s moves at other information sets as in **s**^{*}.

Then, player *i* cannot improve its payoff (at the information set) by deviating from s_i at the information set only. Note that for games with perfect information the above definitions reduce to backward induction.

We illustrate the idea of a subgame-perfect equilibrium through the dynamic game in extensive form of Fig. 3.3. At first glance, this game appears as multi-act, since player 1 follows itself at one information set. However, the actions of this player can be collapsed down to three, all at one stage: DL, DR, U. If player 1 plays DL or DR, then player 2 enters the game, otherwise player 2 does not play; hence each player acts only once (at most) and the game can be seen as a single-act game. Moreover, clearly this game has imperfect information, so backward induction cannot be used to obtain the outcome of the game. However, we can compute the subgame-perfect equilibrium of the game by considering the two subgames: the subgame that starts after player 1 plays D and the subgame which is the game itself. To compute the subgame-perfect equilibrium, we first consider the subgame. Then, fixing the equilibrium actions as they are in this subgame and taking the equilibrium payoffs in this subgame as the payoffs for entering the subgame, we compute a Nash equilibrium in the remaining game. The considered



Fig. 3.3 A dynamic game in extensive form.





subgame clearly has only one Nash equilibrium, which is (L, U), yielding a payoff of (3, 2) (this equilibrium can be found using strict dominance as the subgame is, in fact, a static game). Consequently, the remaining game is shown in Fig. 3.4. Clearly, in Fig. 3.4, player 1 would choose D which yields the subgame-perfect equilibrium in which player 1 picks D at stage 1 and L at stage 2, while player 2 would always pick U.

For instance, in a finite game, the following technique can be used for finding the subgame-perfect equilibrium:

- 1. Pick a subgame that does not contain any other subgame.
- 2. Compute a Nash equilibrium of this subgame.
- 3. Assign the payoff vector associated with this equilibrium to the starting node, and eliminate the subgame.
- 4. Iterate this procedure until a move is assigned at every contingency, when there remains no subgame to eliminate.

Note that, in dynamic games, a Nash equilibrium is not necessarily a subgame-perfect equilibrium.

Here we have restricted our attention to extensive-form games where the solution sought is a pure-strategy equilibrium. However, the approaches described readily extend to mixed-strategy spaces, and the interested reader is referred to [58] for further details.

3.3.2 Repeated games

Beyond single-act and multi-act sequential games, an important type of dynamic games is the *repeated game*.² In simple terms, a repeated game can be seen as a static noncooperative strategic game that is repeated over time. By repeating a game over time, the players may become aware of past behavior of the players and change their strategies accordingly. The motivation for this class of games is rooted in The Prisoner's Dilemma, Example 3.1. In this example, the non-cooperative behavior where both prisoners confess (or fink on each other) is the unique pure-strategy Nash equilibrium of the game. However, both prisoners are better off if they cooperate and do not confess. The main idea behind a repeated game is that, if a game such as The Prisoner's Dilemma is played repeatedly, then the mutually desired outcome in which both prisoners remain silent, i.e., do not confess, in every period is stable if each player believes that a defection, i.e., a confession, will terminate the cooperation, resulting in a subsequent loss for him that outweighs the short-term gain.

In the context of a repeated game, the strategic game will be referred to as a *stage* or *constituent game* of the repeated game. The decisions made by the players in a constituent game at any stage will be referred to as *actions* while the players' decisions in the repeated game itself will be the *strategies*.³ In a repeated game, at any stage t we define the *history* of the game h(t) as the set of past actions at all periods before t. Then, for t = 0, we have $h(0) = \emptyset$, and, for $t \ge 1$, we have $h(t) = \{a(0), ..., a(t-1)\}$, where $a(z) = [a_1(z), ..., a_N(z)]$ is the profile of actions chosen by the N players at time (stage) z. The strategy of a player i at a stage t is thus defined as a function of the history at time t, i.e., $s_i(h(t))$. Hence, for every history h(t) of the game, each player can define a strategy $s_i(h(t))$ which is a function that associates with each history h(t) an action $a_i(t)$ for player i at stage t, i.e., $a_i(t) = s_i(h(t))$. For example, since the initial history h(0) is empty, every player i needs to select an action $a_i(0)$ from its action space in the initial constituent game. Here, we mainly deal with repeated games with *observable actions* and *perfect monitoring*, implying that each player knows all the actions of others as well as its own previous actions at each stage in the repeated game.

For example, consider the *repeated Prisoner's Dilemma* game, i.e., a game where The Prisoner's Dilemma of Example 3.1 is repeated over a period T. For a two-stage

² Note that repeated games can also be modeled in extensive form and solved using the methods of Section 3.3.1. However, the main interest of this subsection is in repeated strategic games, since their analysis differs from that of extensive-form games.

³ This is similar to the distinction between actions and strategies in extensive-form games.

repeated Prisoner's Dilemma game, a possible strategy for player i is CCCCC, where:

- At the first stage, the initial action of player *i* at t = 0 is $a_i(0) = C$.
- At the second stage, t = 1, the strategy of player *i* is to confess, i.e., choose *C*, for every history h(1), i.e.,

$$s_{i}(h(1)) = \begin{cases} C, & \text{if } h(1) = \{(C, C)\}, \\ C, & \text{if } h(1) = \{(C, NC)\}, \\ C, & \text{if } h(1) = \{(NC, NC)\}, \\ C, & \text{if } h(1) = \{(NC, C)\}. \end{cases}$$
(3.23)

Clearly, the possible strategy spaces for a repeated game grow very fast with the number of stages. For example, in the two-stage repeated Prisoner's Dilemma, we have $2^5 = 32$ possible strategies, so finding a Nash equilibrium through an exhaustive search of best-response strategies is quite complicated. However, as we will see in the rest of this subsection, alternative approaches can be used to find equilibrium points of a repeated game.

Now that we have provided the motivation behind repeated games and the basic concepts, we can formally define a repeated game, as follows:

DEFINITION 3.15 Let $G = (\mathcal{N}, (\mathcal{A})_{i \in \mathcal{N}}, (g_i)_{i \in \mathcal{N}})$ be a strategic game and $\delta \in [0, 1)$ be a discount factor. The repeated game, denoted by $G(T, \delta)$, consists of game G repeated for T + 1 periods from t = 0 until t = T. For every player i, we define player i's strategy for the repeated game as $\mathbf{s}_i = [\mathbf{s}_i(h(0)), \dots, \mathbf{s}_i(h(T))]$. Thus the strategy profile of the opponents can be denoted by $\mathbf{s}_{-i} = [(\mathbf{s}_j)_{j \in \mathcal{N}, j \neq i}]$. Thus, the utility for a player i in the repeated game is given by

$$u_i(\mathbf{s}_i, \mathbf{s}_{-i}) = \sum_{t=0}^T \delta^t g_i(a_i(t), \mathbf{a}_{-i}(t)), \qquad (3.24)$$

where $\mathbf{a}_i(t) = \mathbf{s}_i(h(t))$ denotes the action taken by player *i* at stage *t*, with $g_i(a_i(t), \mathbf{a}_{-i}(t))$ being the payoff for player *i* from the constituent game at period *t*. If *T* goes to infinity, then $G(\infty, \delta)$ is referred to as a repeated game with infinite horizon; otherwise, we have a repeated game with finite horizon.

First, we note that the idea of an infinite-horizon game does not necessarily mean that the game will continue indefinitely. A model with an infinite horizon is appropriate if, after each period, the players believe that the game will continue for an additional period, while a model with a finite horizon is appropriate if the players clearly perceive a well-defined final period. In other words, if the players are unaware of the duration of the game and the stage at which it ends, the use of an infinite-horizon repeated game can be more appropriate.

Furthermore, in games with infinite horizon, the discounted utility in (3.24) is often normalized, and so it is replaced with the following normalized utility:

$$u_i(\mathbf{s}_i, \mathbf{s}_i) = (1 - \delta) \sum_{t=0}^T \delta^t g_i(a_i(t), \mathbf{a}_{-i}(t)).$$
(3.25)

The factor $1 - \delta$ simply ensures that the stage payoff of the repeated game is expressed in the same unit as the static strategic game. For example, if the stage payoffs $g_i(a_i(t), \mathbf{a}_{-i}(t)) = 1$, $\forall t = 0, 1, ...$, then the normalized utility in (3.25) is $u_i(\mathbf{s}_i, \mathbf{s}_i) = (1 - \delta) \sum_{t=0}^{\infty} \delta^t = (1 - \delta) \frac{1}{1 - \delta} = 1$.

Let us now revisit the repeated Prisoner's Dilemma game. In this game, if we consider a finite horizon T, then one can use backward induction to solve the game. Since the game is assumed to be of perfect information with observable actions and perfect monitoring, then the players know the end of the game. Hence, the two players can conclude that, in the last stage, the dominant strategy is to confess, i.e., C. Thus, by using this argument and going back through the stages of the game, we find that playing C in every stage is a subgame-perfect equilibrium of the repeated Prisoner's Dilemma game with finite horizon. When the game is played infinitely, however, it turns out that we have the following result [160]:

PROPOSITION 3.3 If $1 > \delta \ge \frac{1}{2}$, then the repeated Prisoner's Dilemma has a subgameperfect equilibrium in which (NC, NC) is played in every period, i.e., the prisoners do not confess, and thus cooperate, in every period.

In some sense, this proposition means that, if the players sufficiently value future payoffs compared to present ones, then (NC, NC) is a sustainable outcome. Indeed, if each player plays NC as long as the other one does so in the past, and plays C otherwise, both players have an incentive to always play NC. This is because the short-term gain obtained by playing C is more than offset by future losses entailed by the opponent playing C at all future stages.

Note that this proposition does not imply that (*NC*, *NC*) is the only subgame-perfect equilibrium. In fact, depending on the value of δ , other equilibria can exist. Furthermore, depending on the value of δ (notably when $0 \le \delta < \frac{1}{2}$), the outcome (*C*, *C*) can also be a subgame-perfect equilibrium, even when the horizon is infinite.

For repeated games with infinite horizon, the so-called Folk Theorem provides an interesting result which implies that a feasible outcome that gives each player a payoff that is better than the static game Nash equilibrium can be obtained through a repeated game. Before formally stating the Folk Theorem, we define the set of *feasible payoffs* \mathcal{U} as follows:

$$\mathcal{U} = \operatorname{Conv}\{\mathbf{u} \mid \exists \mathbf{a} \in \mathcal{A} \text{ such that } g(\mathbf{a}) = \mathbf{u}\},$$
(3.26)

where Conv represents the convex hull and $g(\mathbf{a})$ is the function which associates with every action profile \mathbf{a} an *N*-dimensional payoff vector (each element is the payoff for a player *i* as given by function g_i). In other words, \mathcal{U} is simply the convex hull of $g(\mathbf{a})$. For example, in The Prisoner's Dilemma of Example 3.1, the set \mathcal{U} of feasible payoffs is the convex hull of $\{(-4, -4), (-2, -2), (-5, 0), (0, -5)\}$.

Furthermore, we define, at any given stage of a repeated game, the notion of *min-max* value \underline{u}_i for a player *i*, as follows (the stage subscript *t* is dropped since we consider a single stage):

$$\underline{u}_{i} = \min_{\mathbf{a}_{-i}} \left[\max_{a_{i}} g_{i}(a_{i}, \mathbf{a}_{-i}) \right].$$
(3.27)

The min-max value is the lowest-stage payoff that the opponents can enforce on player *i*, provided that *i* plays its best response against them. For example, in The Prisoner's Dilemma of Example 3.1, the min-max value for both players is -4. We say that a payoff vector $\mathbf{u} \in \mathbb{R}^N$ is *strictly individually rational* if $u_i > \underline{u}_i$, $\forall i \in \mathcal{N}$. Note that in a static Nash equilibrium, the payoff for any player *i* is at least \underline{u}_i .

Consequently, the Folk Theorem is stated as follows:

THEOREM 3.7 (*Folk Theorem*) If $\mathbf{u} = (u_1, ..., u_N)$ is a feasible and strictly individually rational payoff vector, then there exists a discounting factor $0 \le \underline{\delta} < 1$ such that for all $\delta > \underline{\delta}$, the repeated game with infinite horizon $G(\infty, \delta)$ has a Nash equilibrium (which is also a subgame-perfect equilibrium) with payoff vector \mathbf{u} .

The main motivation behind the Folk Theorem is that, if the game duration is long enough (δ close to 1), the gain obtained by a player by deviating once is outweighed by the loss in every subsequent period, when loss is due to the min-max strategy of the other players. Hence, on the one hand, with enough patience, i.e., a large δ , a player's noncooperative behavior will be punished by the future actions of other cooperative players. On the other hand, a player's cooperation (through independent decision-making and with no communication) can be rewarded in the future by others' cooperation. Hence, in the long run, the players, although acting non-cooperatively, might choose a cooperative behavior so as to obtain a payoff that is better than the min-max value.

While the Folk Theorem states that payoffs that are better than the min-max values are possible through a repeated game, the next key challenge is to define a rule suited to achieving these better payoffs by enforcing cooperation among non-cooperative players. For this purpose, two approaches can be used: Tit-for-Tat and cartel maintenance.

Tit-for-Tat is a type of *trigger strategy*; this is a class of strategies that rely on punishment to enforce cooperation. In a trigger strategy, typically applied to the repeated Prisoner's Dilemma, a player begins by cooperating (i.e., not confessing) but defects (i.e., confesses) to cheating for a predefined period of time as a response to a defection by the opponent. Hence, Tit-for-Tat is a trigger strategy in which a player starts by cooperating but responds in one stage of the repeated game with the same strategy its opponent used in the previous period. Hence, once an opponent defects from the cooperative strategy, the player would respond by also defecting. In the repeated Prisoner's Dilemma, it has been shown that Tit-for-Tat strategies by both prisoners result in a Pareto-optimal Nash equilibrium [160]. The advantage of Tit-for-Tat is that it is an easy approach to implement, although it suffers from some drawbacks. First, in a given game and for a given player, choosing the same strategy as that of an opponent is not necessarily the best response of this player. Moreover, gathering information on all the strategies of the other players, as needed for Tit-for-Tat, is quite difficult in many scenarios (although this is a general drawback for games with observable actions and perfect monitoring). Hence, although Tit-for-Tat is useful in many research problems, the drawbacks limit its field of application.

As an alternative to Tit-for-Tat, one can use trigger strategies which can yield payoffs closer to optimal (than in Tit-for-Tat) and, in some cases, are harsher strategies. One such approach is cartel maintenance [390]. The basic idea for the cartel-maintenance repeated-game framework is to provide enough of a threat to greedy players to prevent them from deviating from a potential cooperation. In this context, one would first compute a cooperative point in such a way that all players have better payoffs than in Nash equilibrium points. However, if any player deviates from cooperation while the others are still playing cooperative strategies, this deviating player has a better utility, while others have relatively worse utilities. Without any enforcement rule, the other cooperative players will also have incentives to deviate. Consequently, the overall outcome of the game might revert back to an inefficient non-cooperative point such as the Nash equilibrium. The cartel-maintenance framework provides a mechanism so that the current defecting gains of any player will be outweighed by future punishment strategies from other players. Thus, the threat of future punishments prevents any player, acting rationally, from deviating. As a result, cooperation is enforced and the overall payoffs are better. Further details on the use of the cartel-maintenance framework, notably within a wireless environment, can be found in [187].

Finally, while we have focused here on repeated games with observable actions and perfect monitoring, some of these concepts can also be extended to games with imperfect monitoring, as shown in [160].

3.3.3 Stochastic games

A stochastic game, an idea introduced by Shapley [448], is a repeated game with stochastic (probabilistic) transitions between the different states of the game. A stochastic game is thus a dynamic game which is composed of a number of stages, and in which, at the beginning of each stage, the game is in a particular state. In this state, the players select their actions and each player receives a payoff that depends on the current state and the chosen actions. The game then moves to a new random state whose distribution depends on the previous state and the actions chosen by the players. The procedure is repeated at the new state and the game continues for a finite or infinite number of stages. The total payoff to a player is often taken to be the discounted sum of the stage payoffs (similar to the discounted sum of repeated games) or the limit inferior of the averages of the stage payoffs. Notice that a one-state stochastic game is equal to an (infinitely) repeated game, and a one-agent stochastic game is equal to a Markov decision process (MDP). The formal definition of a stochastic game is as follows:

DEFINITION 3.16 An N-player stochastic game G consists of a finite, non-empty set of states S, a set N of N players, a finite set A_i of actions for the players, a conditional probability distribution p on $S \times (A_1 \times A_2 \times \cdots \times A_N)$ known as the Law of Motion, and bounded real-valued payoff functions u_i defined on the history space $H = S \times A \times$ $S \times A \cdots$, where $A = \prod_{i=1}^{N} A_i$. The game is called an N-player deterministic game if, for each state $s \in S$ and each action choice $\mathbf{a} = (a_1, a_2, \dots, a_N)$, there is a unique state s'such that $\mathbf{p}(s'|s, \mathbf{a}) = 1$.

If the number of players, the action sets, and the set of states are finite, then a stochastic game with a finite number of strategies always has a Nash equilibrium. This result is also true for a stochastic game with infinitely many stages if the total payoff is of discounted sum. Vieille [488] has shown that all two-person stochastic games with finite state and action spaces have approximate Nash equilibria when the total payoff is the limit inferior of the averages of the stage payoffs. Whether such equilibria exist when there are more than two players is still a challenging open problem. Shapley [448] has proposed an algorithm for finding the equilibrium of a two-player zero-sum stochastic game using value iteration.

While the detailed treatment of stochastic games is outside the scope of this book, readers are referred to [150] for a unified and rigorous treatment of the theories of Markov decision processes and two-person stochastic games, and to [11] for a more advanced and comprehensive study of stochastic games. Finally, we note that stochastic games admit many applications in economics, evolutionary biology, and more recently wireless and communication networks. For instance, in wireless networking, stochastic games have been widely used to study problems in areas such as flow control, routing, and scheduling [25, 31].

3.4 Special classes of non-cooperative games

In this section, we investigate several important special games that are widely used to formulate problems in wireless and communication networks.

3.4.1 Potential games

Potential games are non-zero-sum games in which the determination of a Nash equilibrium can be equivalently posed as the maximization of a single function (called the potential function) collectively by all players. The concept was introduced by Monderer and Shapley [342], and is related to the notion of strategic equivalence of a game to a team problem [57] (see also [58]). The potential function is a useful tool for analyzing equilibrium properties of games, since the objectives and goals of all players are mapped into one function, and the set of pure Nash equilibria can be found by simply locating the person-by-person optima of the potential function. Note that the definitions and discussions hereinafter pertain to *static* potential games. We formally define the potential game as follows:

DEFINITION 3.17 A non-cooperative strategic game $(\mathcal{N}, (\mathcal{S}_i)_{i \in \mathcal{N}}, (u_i)_{i \in \mathcal{N}})$ is an exact (cardinal) potential game if there exists an exact potential function $\Phi : S \to \mathbb{R}$ such that $\forall i \in \mathcal{N}$

$$\Phi(x, \mathbf{s}_{-i}) - \Phi(z, \mathbf{s}_{-i}) = u_i(x, \mathbf{s}_{-i}) - u_i(z, \mathbf{s}_{-i}), \ \forall x, z \in \mathcal{S}_i, \forall \mathbf{s} \in \mathcal{S}.$$
(3.28)

A game is a general (ordinal) potential game *if there is an ordinal potential function* $\Phi: S \to \mathbb{R}$ such that

$$\operatorname{sgn}[\Phi(x, \mathbf{s}_{-i}) - \Phi(z, \mathbf{s}_{-i})] = \operatorname{sgn}[u_i(x, \mathbf{s}_{-i}) - u_i(z, \mathbf{s}_{-i})], \ \forall x, z \in \mathcal{S}_i, \forall \mathbf{s} \in \mathcal{S}, \quad (3.29)$$

where sgn denotes the sign function.

In other words, in *exact potential games*, the difference in individual utilities achieved by each player when changing *unilaterally* its strategy has the same value as the difference in values of the potential function. In *ordinal potential games*, only the signs of the differences have to be the same. We note that the potential function does not depend on the indices of the players. Hence, for every player, Φ quantifies the impact of unilateral deviations on all the players' utilities in exact potential games, while it gives a sign of the difference in ordinal potential games.

The interest in potential games stems from the following existence result [160]:

COROLLARY 3.1 Every finite potential game (exact or ordinal) has at least one purestrategy Nash equilibrium.

In fact, every strategy vector **s** which maximizes a potential function will be a purestrategy equilibrium for the game. However, other pure-strategy Nash equilibria might exist, which are the person-by-person maxima of the ordinal potential. For the case of infinite potential games, we note the following existence result [160]:

COROLLARY 3.2 For infinite potential games (with a finite number of players), a purestrategy Nash equilibrium exists if: (i) S_i are compact strategy sets and (ii) the potential function Φ is upper semi-continuous on S.

Recall that a function $f : \mathcal{X} \to \mathbb{R}$ is upper semi-continuous at a point $x_0 \in \mathcal{X}$ if, for any ϵ there exists a neighborhood $v(x_0)$ such that $x \in v(x_0)$ implies that $f(x) < f(x_0) + \epsilon$. The function f is upper semi-continuous if it is upper semi-continuous at each $x_0 \in \mathcal{X}$.

The interest in potential games stems from the guarantee of the existence of purestrategy Nash equilibria. Moreover, when the strategy set S is compact and convex and the potential function Φ is a continuously differentiable function on the interior of S and strictly concave on S, then the Nash equilibrium is *unique*.

Although potential games have very interesting properties, finding a potential function for a game is complex, and in many cases such a function does not exist. In [342], the

existence of a potential function for a given strategic game is shown for the case where the strategy sets are intervals of \mathbb{R} as per the following theorem:

THEOREM 3.8 Given a strategic game where the strategy sets S_i , $\forall i \in N$, are intervals of real number, and assuming the utilities are twice continuously differentiable, then this game is a potential game if and only if

$$\frac{\partial^2 (u_i - u_j)}{\partial s_i \partial s_j} = 0, \forall i \in \mathcal{N}, j \in \mathcal{N}.$$
(3.30)

To illustrate the idea of a potential game, we first discuss the following two-player game:

Example 3.7 In a two-player game, each player has two strategies and the utilities are given by the function $u_i(s_i, s_j) = b_i s_i + w s_i s_j$, where s_i is player i's strategy, s_j is the opponent's strategy, b_i is a constant, and w is a positive externality from choosing the same strategy. The strategy choice of each player is either +1 or -1, as seen in the payoff matrix of Table 3.8(a). This game has the following potential function:

$$\Phi(s_1, s_2) = b_1 s_1 + b_2 s_2 + w s_1 s_2. \tag{3.31}$$

If player 1 moves from -1 to +1, the utility difference is $\Delta u_1 = u_1(+1, s_2) - u_1(-1, s_2) = 2b_1 + 2ws_2$. The change in potential is $\Delta \Phi = \Phi(+1, s_2) - \Phi(-1, s_2) = (b_1 + b_2s_2 + ws_2) - (-b_1 + b_2s_2 - ws_2) = 2b_1 + 2ws_2 = \Delta u_1$. Similarly, one can check the utility differences for player 2.

Using numerical values $b_1 = 2, b_2 = -1, w = 3$, this example transforms into a simple Battle of the Sexes game, as shown in Table 3.8(b). The game has two pure Nash equilibria, (+1, +1) and (-1, -1). These are also the person-by-person maxima of the potential function (Table 3.8(c)). The only stochastically stable equilibrium is (+1, +1), the global maximum of the potential function. Note that one can also define a continuous-kernel game version of this example, where the utilities are given by the function $u_i(s_i, s_j) = -s_i^2 + b_i s_i + w s_i s_j$, where the strategies s_1 and s_2 take values on the real line. In this continuous-kernel game, the potential function would be given by

$$\Phi(s_1, s_2) = -s_1^2 - s_2^2 + b_1 s_1 + b_2 s_2 + w s_1 s_2. \tag{3.32}$$

The use of potential games is quite popular in wireless networking and resource allocation [308, 342, 356, 354, 355, 142, 26, 330, 209, 430]. The major applications are to power control and waveform adaptation. The design implication is the property of convergence, with shared and independent outcomes for each individual.

For example, the use of potential games in CDMA networks is discussed in [430]. Consider a single-cell CDMA network with $\mathcal{N} = \{1, \dots, N\}$ players and where the

Table 3.8 (a) Potential game example (upper); (b) Battle of the sexes (payoffs, lower left); (c) Battle of the sexes (potentials, lower right).

(a)	+1	l		-1		
+1	$(b_1 + w, b_1)$	$(b_2 + w)$	(b ₁ -	- w,-	$b_2 - 1$	N)
-1	$(-b_1 - w_1)$	$(b_2 - w)$	$(-b_1$	+ w,-	$-b_2 +$	W)
(<i>b</i>)	+1	-1		(c)	+1	-1
+1	(5,2)	(-1,-2))	+1	4	0
-1	(-5,-4)	(1,4)		-1	-6	2

received SINR of player $i \in \mathcal{N}$ is

$$\gamma_i(\mathbf{p}) = \frac{p_i h_i}{\sigma^2 + \sum_{j \neq i} p_j h_j}$$

where p_i is the transmit power of player *i*, h_i is the channel gain from player *i*'s transmitter to the base station, and σ^2 is the variance of the Gaussian noise. For each player $i \in \mathcal{M}$, we want to solve the following power-control problem:

$$\min_{p_i \in [0, P_i^{\max}]} \quad p_i, \quad \text{s.t.} \quad f_i(\gamma_i(\mathbf{p})) \ge \gamma_i^{\text{thresh}}. \tag{3.33}$$

Here, P_i^{max} is the maximum power, f_i is a function reflecting the quality of service (QoS), and γ_i^{thresh} is the QoS threshold. It is shown in [430] that (3.33) for each player *i* can be transformed into an equivalent non-cooperative game $G = [\mathcal{N}, \mathcal{S}, \{\log(P_i^{\text{max}} - p_i)\}_{i \in \mathcal{N}}]$, with the *coupled* strategy set

$$\mathcal{S} = \left\{ \mathbf{p} : f_i(\gamma_i(\mathbf{p})) \geq \gamma_i^{\text{thresh}}, p_i \in [0, P_i^{\text{max}}], \forall i \in \mathcal{N} \right\}.$$

Furthermore, it turns out that this power-control game G admits a potential function

$$\Phi(\mathbf{p}) = \sum_{i \in \mathcal{M}} \log(P_i^{\max} - p_i)$$

We can then maximize the potential function $\Phi(\mathbf{p})$ over the set S, and the corresponding maximizer(s) will be the pure-strategy Nash equilibria of the game G, and thus the optimal solution(s) of (3.33) for all $i \in \mathcal{N}$.

3.4.2 Stackelberg games

In many non-cooperative games, a hierarchy among the players might exist whereby one or more of the players declare and announce their strategies before the other players choose their strategies. In such a hierarchical decision-making scheme, the declaring players can be in a position to enforce their own strategies upon the other players. Thus, in these games, the player who holds the strong position and that can impose its own strategy upon the others is called the *leader* while the players who react to the leader's declared strategy are called *followers*. In some situations there could be multiple leaders as well as multiple followers.

Given a two-player non-cooperative game between a *leader* and a *follower*, with their strategy sets denoted by S_1 and S_2 , respectively, whenever the leader announces that it needs to play any strategy $s_1 \in S_1$, the follower must respond or *react* with a given strategy $s_2 \in S_2$. The follower might, indeed, have many possible reactions to a given strategy of the leader. Capturing this possibility, we first have:

DEFINITION 3.18 Given a two-person finite game, the set $\mathcal{R}_2(s_1)$, defined for each strategy $s_1 \in S_1$ by

$$\mathcal{R}_2(s_1) = \{ s_2 \in \mathcal{S}_2 : u_2(s_1, s_2) \ge u_2(s_1, t), \ \forall t \in \mathcal{S}_2 \}, \tag{3.34}$$

is the optimal response (or optimal reaction) set *of player* 2 *to the strategy* $s_1 \in S_1$ *of player* 1.

Given the concept of a reaction set, we can define the concept of a *Stackelberg equilibrium strategy* (or Stackelberg strategy for short) which will prove to be useful in defining an equilibrium point in games with hierarchical decision-making.

DEFINITION 3.19 In a two-person finite game with player 1 as the leader, a strategy $s_1^* \in S_1$ is called a Stackelberg equilibrium strategy for the leader, if

$$\min_{s_2 \in \mathcal{R}_2(s_1^*)} u_1(s_1^*, s_2) = \max_{s_1 \in \mathcal{S}_1} \min_{s_2 \mathcal{R}_2(s_1)} u_1(s_1, s_2) \triangleq u_1^*.$$
(3.35)

The quantity u_1^* is the Stackelberg utility for the leader. The same definition applies for the case where player 2 is the leader, with the subscripts 1 and 2 simply swapped.

For a finite game, given that S_1 and S_2 are finite sets, from (3.35) we have the following result [58]:

THEOREM 3.9 Every two-person finite game admits a Stackelberg strategy for the leader.

Although, in a given game with hierarchical decision-making, the Stackelberg utility for the leader has a unique value, the leader's Stackelberg strategy need not be unique. In (3.35), the Stackelberg strategy s_1^* of the leader ensures that the leader does not receive a utility that is lower than u_1^* . Hence, u_1^* constitutes a secured utility level for the leader. Nonetheless, whenever the follower's reaction set $\mathcal{R}_2(s_1)$ is a singleton set for each $s_1 \in S_1$, then the optimal response of the follower becomes unique for every strategy of the leader, and u_1^* becomes the actual utility level that the leader will attain. This now leads to the following:

DEFINITION 3.20 Let $s_1^* \in S_1$ be a Stackelberg strategy for the leader, i.e., player 1. Then, any strategy $s_2^* \in \mathcal{R}_2(s_1^*)$ that is in equilibrium with s_1^* is an optimal strategy for the follower. Thus, the pair (s_1^*, s_2^*) is a Stackelberg solution for the game with player 1 being

	U	D	М
U	(3,3)	(2,3)	(0,2)
D	(1,4)	(1,3)	(4,4)

Table 3.9 Stackelberg game example (player 1, player 2).

the leader, and the utility pair $(u_1(s_1^*, s_2^*), u_2(s_1^*, s_2^*))$ is the corresponding Stackelberg equilibrium outcome.

The Stackelberg solution of a non-cooperative game, i.e., the introduction of a hierarchy, does not necessarily yield payoffs for the leader and the follower that are better than the Nash equilibrium. However, it can be shown that, using the Stackelberg solution, the leader can improve its utility, as per the following proposition [58]:

PROPOSITION 3.4 For a given two-person finite game, let u_1^* and u_1^{NE} denote, respectively, the Stackelberg utility and the Nash equilibrium utility for player 1, i.e., the leader in the Stackelberg formulation. If the reaction set $\mathcal{R}_2(\mathfrak{s}_1)$ is a singleton set for all $\mathfrak{s}_1 \in \mathcal{S}_1$, then we have

$$u_1^* \ge u_1^{NE}.$$
 (3.36)

In other words, whenever the follower has a single optimal response for every strategy of the leader, then the leader can, at the Stackelberg solution, perform at least as well as at the Nash equilibrium. Note that this proposition holds if $\mathcal{R}_2(s_1)$ is a singleton set for all $s_1 \in S_1$ and not only at the Stackelberg strategy s_1^* of the leader.

To better illustrate the idea of a Stackelberg equilibrium, we consider the matrix game shown in Table 3.9. With no hierarchy, one can easily check that (D, M) and (U, U)are Nash equilibria, but (D, M) is the one with better payoffs for both players as it yields (4,4). Now consider the game in Table 3.9 with player 1 considered as a leader. If the leader chooses strategy U, then the reaction set of the follower is $\mathcal{R}_2(U) = \{U, D\}$. Hence, the Stackelberg utility for the leader, i.e., the secured utility, is $u_1^* = 2$. In contrast, if the leader chooses D, then $u_1^* = 1$. Consequently, $s_1^* = U$ would be the leader's Stackelberg strategy, and the Stackelberg utility would be $u_1^* = 2$. Note that, depending on whether the follower chooses U or D, at the Stackelberg equilibrium the leader might achieve either 2 or 3. In other words, the game admits two Stackelberg equilibria, (U, U)and (U, D), with payoffs (3, 3) and (2, 3), respectively. This demonstrates the difference between the secured Stackelberg utility $u_1^* = 2$ and the actual utility at the Stackelberg equilibrium $u_1(s_1^*, s_2^*)$, which could be either equal to u_1^* , if the follower plays U, or better than u_1^* , if the follower plays D. In this game, we also note that the Stackelberg equilibrium provides lower utilities for the leader and the follower than in the best Nash equilibrium case (note that, here, the result of Proposition 3.4 does not hold since the reaction sets are not singleton sets).

The Stackelberg solution easily extends to games with a single leader and multiple followers. Although the detailed treatment of this case will be done through an example
from communication networks in Chapter 14, the idea is relatively simple. In the multifollower case, given the leader's announced Stackelberg strategy, one can define a single reaction set for the group of followers. In this case, the optimal response (reaction set) of the followers is the set of joint strategies that maximize the utilities of the followers, where each follower's utility is a function of (i) the leader's announced strategy, and (ii) the strategies of the other followers. Consequently, the Stackelberg solution of a singleleader multi-follower non-cooperative game corresponds to the case where the leader maximizes its utility given the reaction set of the followers group while the followers react to the leader's announced strategy by playing according to a specific equilibrium concept (e.g., the Nash equilibrium). Similarly, one can also extend the Stackelberg solution to the multi-leader multi-follower case (we treat an example in the context of cellular networks in Chapter 9) as well as the case where more than two levels of hierarchy exist (e.g., a leader, a first follower, a second follower, etc.). However, a detailed treatment of this case is more complicated and is beyond the scope of this book (the interested reader is referred to [58] for further details).

3.4.3 Correlated equilibrium

The concept of a correlated equilibrium was introduced by Aumann [47, 48] as a concept suitable for scenarios that involve a decision process in between non-cooperation and cooperation. In this sense, the correlated equilibrium can be viewed as a generalization of the Nash equilibrium, whereby there exists an arbitrator or coordinator who can send (public or private) signals to the players, which help them to correlate their actions. In the context of the correlated equilibrium, an arbitrator is seen as an entity (not necessarily an intelligent one) that can generate signals that do not depend on the system (nor individual) states. Moreover, the arbitrator does not need any knowledge of the system, and can be a virtual entity. All the arbitrator has to do is to create some random signals (according to a randomized mechanism known by the players) which can help with the coordination of actions between the players. The players can also ignore the received signals and select their strategies independently of them.

The interest in correlated equilibrium stems from the fact that by allowing the players to utilize a joint action profile with a certain probability, a performance that is better than the Nash equilibrium can be achieved. In fact, it has been shown that the correlated equilibrium can be better than the convex hull of the Nash equilibria [47, 48]. For instance, given a non-cooperative strategic game $G = (\mathcal{N}, (S_i)_{i \in \mathcal{N}}, (u_i)_{i \in \mathcal{N}})$, we can define the *correlated strategy* $p(\mathbf{s})$ as a probability distribution over the strategy (or action) profile $\mathbf{s} \in S$. Given these basic notions, we can define the concept of a *correlated equilibrium* as follows:

DEFINITION 3.21 Given a strategic game G, a correlated strategy $p(\mathbf{s}) = p(s_i, \mathbf{s}_{-i})$ is said to be a correlated equilibrium if, for all $i \in \mathcal{N}$, $s_i, s'_i \in S_i$, and $\mathbf{s}_{-i} \in S_{-i}$, we have

$$\sum_{\mathbf{s}_{-i}\in\mathcal{S}_{-i}} p(s_i, \mathbf{s}_{-i}) [u_i(s'_i, \mathbf{s}_{-i}) - u_i(s_i, \mathbf{s}_{-i})] \le 0.$$
(3.37)

Table 3.10 Chicken game variant (driver 1, driver 2).

	Straight (ST)	Swerve (S)		
Straight (ST)	(0,0)	(5,1)		
Swerve (S)	(1,5)	(4,4)		

By dividing the inequality (3.37) by $p(s_i) = \sum_{\mathbf{s}_{-i} \in S_{-i}} p(s_i, \mathbf{s}_{-i})$ and using Bayes' rule, we get

$$\sum_{\mathbf{s}_{-i}\in\mathcal{S}_{-i}}p(\mathbf{s}_{-i}|s_i)[u_i(s'_i,\mathbf{s}_{-i})-u_i(s_i,\mathbf{s}_{-i})]\leq 0, \forall s'_i\in\mathcal{S}_i.$$
(3.38)

This implies that the expected payoff received by a player *i* choosing strategy s_i at the correlated equilibrium is greater than (or equal to) its expected payoff for choosing any other strategy s'_i . Furthermore, we note that the set of correlated equilibria is non-empty, closed, and convex in every finite game [47, 48], and may include distributions that are not in the convex hull of the Nash equilibrium solutions. Moreover, every Nash equilibrium is a point inside the set of correlated equilibria. In fact, a mixed-strategy Nash equilibrium is simply a correlated equilibrium where p(s) is a product distribution, i.e., a product of each individual player's probability for different strategies.

As an illustrative example of correlated equilibrium, we consider a variant of the game of Chicken, as shown in Table 3.10 (this is the same game as in Example 3.3 but with modified payoff values). The strategies ST and S refer, respectively, to staying straight and swerving. Using the payoffs in Table 3.10 and the inequalities in (3.37), we conclude that a probability distribution at the correlated equilibrium must satisfy the following:

$$egin{aligned} &(0-1)p_{11}+(5-4)p_{12}\geq 0,\ &(1-0)p_{21}+(4-5)p_{22}\geq 0,\ &(0-1)p_{11}+(5-4)p_{21}\geq 0,\ &(1-0)p_{12}+(4-5)p_{22}\geq 0, \end{aligned}$$

$$\sum_{\substack{i,j \in \{1,2\} \ 0 \le p_{ij} \le 1, \forall i, j \in \{1,2\},}} p_{ij} = 1,$$

where p_{ij} is the probability of player 1 choosing strategy *i* and player 2 choosing strategy *j*, where $i, j \in \{1, 2\}$ with strategy 1 being stay straight, i.e., strategy *ST*, and strategy 2 being swerve, i.e., strategy *S*. The first two inequalities represent the optimality of the distribution for player 1 by comparing the payoffs for the cases when player 1 chooses strategy *ST* (first inequality) and *S* (second inequality). Similarly, the third and fourth

	57	-	S			S	Т	S
ST	0		(0 or	1)	ST	1	L Ī	$\frac{1}{4}$
S	(1 or	0)	0		S	1	L Ī	$\frac{1}{4}$
	ST	S			ST	S		
ST	0	$\frac{1}{2}$]	ST	0	$\frac{1}{3}$		
S	$\frac{1}{2}$	0		S	$\frac{1}{3}$	$\frac{1}{3}$		

Table 3.11 Chicken game variant. (a) Nash equilibrium (upper left); (b) mixed Nash equilibrium (upper right); (c) correlated equilibrium (lower left); (d) correlated equilibrium (lower right).

inequalities are written for player 2's case. The last two equations simply state that p_{ij} are probability values. This system of inequalities admits an infinite number of solutions, i.e., correlated equilibria. Nonetheless, we can find several straightforward correlated equilibria, as shown in Table 3.11(a)–(d) (the probability of each joint distribution is shown in each matrix). In Table 3.11(a), we show the two pure-strategy Nash equilibria (which are in the correlated equilibria set). In Table 3.11(b), we show the correlated equilibrium which is the mixed-strategy Nash equilibrium of this game. This mixed-strategy Nash equilibrium dictates that each player use each strategy with a probability of $\frac{1}{2}$. The correlated equilibrium in Table 3.11(c) can be interpreted as a traffic light: a trusted arbitrator tosses a fair coin and, depending on the outcome, suggests a strategy to each player; neither player has an incentive to deviate from the suggested strategy. Note that the correlated equilibrium in Table 3.11(c) yields a better expected utility than the mixed-strategy Nash equilibrium. Finally, the correlated equilibrium in Table 3.11(d) is the one that maximizes the expected sum of utilities, obtained by a linear maximization over the set of correlated equilibria.

As seen in the previous example, the set of correlated equilibria can contain an infinite number of points. Therefore, it is useful in an application to define a metric through which one can choose a suitable correlated equilibrium. Two useful criteria for selecting a correlated equilibrium are the correlated optimal criterion and the max-min criterion. The correlated optimality criterion is a correlated equilibrium that maximizes the social welfare, i.e., the expected sum of utilities, as follows:

DEFINITION 3.22 A correlated strategy $p(\mathbf{s})$ is correlated optimal if it satisfies the following conditions:

$$p(\mathbf{s}) = \arg\max_{i \in S} \sum_{i \in \mathcal{N}} \mathbb{E}_p[u_i], \qquad (3.39)$$

s.t.
$$\sum_{\mathbf{s}_{-i}\in\mathcal{S}_{-i}}p(s_i,\mathbf{s}_{-i})\left[u_i(s'_i,\mathbf{s}_{-i})-u_i(s_i,\mathbf{s}_{-i})\right]\leq 0,$$

 $\forall s_i, s'_i \in S_i \text{ and } \forall i \in \mathcal{N}.$

In contrast, the max-min criterion attempts to find the correlated equilibrium that guarantees a minimum expected utility:

DEFINITION 3.23 A correlated strategy $p(\mathbf{s})$ satisfying the max-min criterion is given by

$$p(\mathbf{s}) = \arg\max_{i \in S} \min \mathbb{E}_{p}[u_{i}], \qquad (3.40)$$

s.t.
$$\sum_{\mathbf{s}_{-i} \in S_{-i}} p(s_{i}, \mathbf{s}_{-i}) [u_{i}(s_{i}', \mathbf{s}_{-i}) - u_{i}(s_{i}, \mathbf{s}_{-i})] \leq 0,$$

 $\forall s_i, s'_i \in S_i \text{ and } \forall i \in \mathcal{N}.$

Finally, the use of the correlated equilibrium in wireless and communication networks has recently attracted increased attention; see [283, 304] and the references therein. The main motivation for using the correlated equilibrium in wireless and communication networks stems from its ability to provide a balance between the fully cooperative solution (which requires a lot of overhead but can be highly efficient) and the fully non-cooperative solution (which can be easy to implement but yields poor performance). In this context, correlated equilibrium provides a scheme that can yield a better performance than the Nash equilibrium while requiring a reasonable overhead for implementation. In particular, correlated equilibrium has been used in solving problems related to multiple-access and wireless transmission (see [283] and references therein) as well as peer-to-peer networks [304].

3.4.4 Supermodular games

Supermodular games are a class of static non-cooperative games characterized by strategic complementarities. In a supermodular game, when a player takes a higher action according to a defined order, the other players are better off if they also take a higher action. In other words, supermodular games are characterized by increasing best responses. Supermodular games are analytically appealing since they have interesting properties regarding the existence of pure-strategy Nash equilibria and algorithms that can find these Nash equilibria.

Before formally defining a supermodular game, we need to introduce the concept of *increasing differences*, using the following definitions:

DEFINITION 3.24 A partial order \geq over a Euclidean space \mathbb{R}^{K} is defined as

$$\mathbf{x} \geq \mathbf{y} \Leftrightarrow x_k \geq y_k, \ \forall k = 1, \dots, K.$$

DEFINITION 3.25 Let \mathcal{T} be a partially ordered set with respect to order \geq and $\mathcal{X} \subseteq \mathbb{R}$. A function $f : \mathcal{X} \times \mathcal{T} \to \mathbb{R}$ has increasing differences in (x, t) if

$$\forall x' \ge x, \ \forall t' \ge t,$$

$$f(x',t')-f(x,t') \ge f(x',t)-f(x,t),$$

i.e., f(x', t) - f(x, t) is increasing in t and, symmetrically, f(x, t') - f(x, t) is increasing in x.

LEMMA 3.2 For a function f that is twice continuously differentiable, f has increasing differences if and only if

$$egin{aligned} t' &\geq t \Rightarrow rac{\partial f}{\partial x}(x,t') \geq rac{\partial f}{\partial x}(x,t), \ \Rightarrow rac{\partial^2 f}{\partial x \partial t}(x,t) \geq 0, \ orall x \in \mathcal{X}, \ orall t \in \mathcal{T} \end{aligned}$$

Given the definition of a function with increasing differences, we can formally define a supermodular game as follows:

DEFINITION 3.26 A strategic game $(\mathcal{N}, (\mathcal{S}_i)_{i \in \mathcal{N}}, (u_i)_{i \in \mathcal{N}})$ is a supermodular game *if*, for all *i*,

- S_i is a compact subset of \mathbb{R} .
- u_i is upper semi-continuous in s_i , and continuous in s_{-i} .
- *u_i* has increasing differences in (*s_i*, *s*_{-*i*}).

Interest in supermodular games stems from the following properties [160]:

THEOREM 3.10 Let $(\mathcal{N}, (\mathcal{S}_i)_{i \in \mathcal{N}}, (u_i)_{i \in \mathcal{N}})$ be a supermodular game. Then the set of strategies surviving iterated strict dominance has greatest and least elements $\bar{\mathbf{s}}, \bar{\mathbf{s}}$ which are bothNash equilibria. In other words,

- A pure-strategy Nash equilibrium exists.
- *The largest and smallest strategies compatible with iterated deletion, the correlated equilibrium, and the Nash equilibrium are the same.*
- If a supermodular game has a unique Nash equilibrium, then it is dominance solvable, and the best-response dynamics converges to it.

PROPOSITION 3.5 Suppose a supermodular game is indexed by t. Then the largest and smallest Nash equilibria are increasing in t.

PROPOSITION 3.6 Suppose a supermodular game has positive spillovers (for all *i*, $u_i(s_i, \mathbf{s}_{-i})$ is increasing in \mathbf{s}_{-i}). Then the largest Nash equilibrium is Pareto-preferred.

The application pool for supermodular games is quite large. In wireless networks, one popular application is in power control, such as in the uplink power-control game provided in [425] for CDMA networks. In this game, [425] considers the single cell of a wireless CDMA network where all users access the channel using orthogonal codes simultaneously, utilizing the entire available frequency spectrum. In such a system, the main objective is to allow the users, in a distributed manner, to control their *uplink* transmit power so as to optimize their quality of service, given the mutual interference that occurs among the users. Hence, a key question that game theory needs to answer is how to allocate the resources, i.e., the power.

For this purpose, one can formulate a non-cooperative strategic game where the players are the network users, the strategies are the transmit power values, and the utilities are a function of the power consumed by the users and the SINR they attain. Each user is assumed to act selfishly but rationally to optimize the following objective function:

$$u_i(p_i, \mathbf{p}_{-i}) = f(\gamma_i) - cp_i, \qquad (3.41)$$

where p_i is the transmit power of user *i*, \mathbf{p}_{-i} is the vector of transmit powers of all the users except *i*, *c* is a price factor, and $f(\cdot)$ is an increasing function (not necessarily concave). Furthermore, γ_i is the SINR, which is defined as

$$\gamma_i = \frac{h_i p_i}{\sum_{j \neq i} h_j p_j + \sigma^2},\tag{3.42}$$

where σ^2 is the Gaussian noise variance and h_i is the channel gain. For this powercontrol game, using the partial derivatives of u_i and by Lemma 3.2, one can show that the function u_i is supermodular if the function f is such that $-\gamma_i f''(\gamma_i) + f'(\gamma_i) > 0$. In [425], the authors provide an appropriate function f that makes the game supermodular, and discuss the various properties of this game.

Beyond wireless and communication networks, applications of supermodular games are abundant, and we provide a few examples below.

1. Investment game. Suppose *N* firms simultaneously make investments $s_i \in \{0, 1\}$ and the payoffs are

$$u_i(s_i, \mathbf{s}_{-i}) = \begin{cases} \pi(\sum s_i) - k & \text{if } s_i = 1, \\ 0 & \text{if } s_i = 0, \end{cases}$$
(3.43)

where π is increasing in aggregate investment and k is a positive constant. This is a supermodular game.

2. Diamond search model. Consider *N* agents who exert effort looking for trading partners. Let s_i denote the effort of agent *i*, and $c(s_i)$ the cost of this effort, where *c* is increasing and continuous. The probability of finding a partner is $s_i \sum_{i \neq i} s_j$. Then

$$u_i(s_i, \mathbf{s}_{-i}) = s_i \sum_{j \neq i} s_j - c(s_i),$$
 (3.44)

has increasing differences in (s_i, \mathbf{s}_{-i}) . Hence, it is supermodular.

3. Well-known games such as the Cournot duopoly and the Bertrand competition game can be supermodular under certain conditions on the payoffs and the strategies (see [160]).

3.4.5 Wardrop equilibrium

Introduced in the 1950s by Wardrop [501], the concept of the Wardrop equilibrium was first presented in the context of road traffic. The motivation was to introduce a concept that can capture key features of resource sharing among many selfish individuals. Since

then, the Wardrop equilibrium has been applied to many problems in transportation and communication networks [200, 458, 35, 396, 121, 28]. In essence, Wardrop considered the scenario of a network of roads and a *large number* (infinite in many cases) of vehicles traveling through the network from an origin to a destination. The vehicles are interested in minimizing their travel time, which is dependent on each road segment's characteristics and its congestion, i.e., the number of vehicles using it. Wardrop modeled this situation using a non-cooperative game, with the players being the vehicles attempting to find a shortest-path route while minimizing their travel time from origin to destination. In this context, the Wardrop equilibrium was introduced as a stable network flow which incurs equal and minimal latency on all used paths between a given origin-destination pair. In addition, the journey times (latency) at the Wardrop equilibrium are less than those experienced by a single vehicle on any unused route. Note that this equilibrium assumes that there exists a *large number* of vehicles, so that the contribution of a single vehicle to the delay is negligible, i.e., close to zero. Under this assumption of a large number of vehicles, and given that all vehicles select their strategies, i.e., routes, independently and rationally, the Wardrop equilibrium state can be seen as one where no arbitrary small fraction of the traffic assigned to some path can benefit from unilaterally deviating to another path.

Clearly, this situation from road traffic correlates with next-generation wireless and communication networks, where the vehicles can be seen as packets and the road network can be seen as a large-scale communication network such as the Internet or a large-scale wireless ad hoc network. The interdependency between the packets stems from the congestion that is caused mutually by the traffic traveling the same road.

Formally, we are given a directed graph $G(\mathcal{V}, \mathcal{E})$ the set of vertices \mathcal{V} and set of edges \mathcal{E} , and latency functions $\mathcal{L} = (l_e)_{e \in \mathcal{E}}$ attached to the edges, where

$$I_e: \mathbb{R}_0^+ \to \mathbb{R}_0^+. \tag{3.45}$$

The latency functions are considered to be non-decreasing, differentiable, and semiconvex, i.e., $x \cdot l_e(x)$ is convex. Moreover, we consider a set of commodities $[k] = \{1, ..., k\}$ specified, for every $i \in [k]$, by source–sink pairs $(s_i, t_i) \in \mathcal{V} \times \mathcal{V}$, a directed acyclic subgraph G_i of G connecting s_i and t_i , and flow demands d_i . The total demand is $d = \sum_{i \in [k]} d_i$. Thus, a Wardrop game is defined by a tuple (G, d).

Let \mathcal{P}_i denote all acyclic paths connecting s_i and t_i in G_i , also known as *admissible* paths of commodity i, and let $\mathcal{P} = \bigcup_{i \in [k]} \mathcal{P}_i$. For every path $p \in \mathcal{P}$, we let f_p denote the volume or traffic of agents on path p. Furthermore, a path-flow vector $(f_p)_{p \in \mathcal{P}}$ induces an edge-flow vector $(f_{e,i})_{e \in \mathcal{E}, i \in [k]}$ such that $f_{e,i} = \sum_{p \in \mathcal{P}_i: e \in p} f_p$, i.e., $f_{e,i}$ is the sum of flows on the paths in \mathcal{P}_i containing edge e. The total flow of an edge $e \in \mathcal{E}$ is, thus,

$$f_e = \sum_{i \in [k]} f_{e,i}.$$
(3.46)

The latency of an edge $e \in \mathcal{E}$ is given by $l_e(f_e)$. A non-negative path-flow vector $(f_p)_{p\in\mathcal{P}}$ is said to be *feasible* if it satisfies the flow demands, i.e., $\sum_{p\in\mathcal{P}_i} f_p = d_i$ for all $i \in [k]$. Note that an edge flow represented by an edge-flow vector $(f_{e,i})_{e\in\mathcal{E},i\in[k]}$ or by

the total edge-flow vector $(f_e)_{e \in \mathcal{E}}$ is considered *feasible* if it is induced by a feasible path-flow vector. By a slight abuse of notation, we denote $(f_p)_{p \in \mathcal{P}}, (f_{e,i})_{e \in \mathcal{E}, i \in [k]}$, and $(f_e)_{e \in \mathcal{E}}$ by f (these flows can in any case be expressed as functions of each other). We denote the set of all feasible flow vectors by \mathcal{F} . The latency of a given path $p \in \mathcal{P}$ is expressed by the sum of the edge latencies, i.e.,

$$I_{p}(f) = \sum_{e \in \mathcal{P}} I_{e}(f_{e}).$$
(3.47)

Note that this path latency is not a function of the corresponding path flow because it depends on the total flow on each of its edges. Consequently, the total latency C(f) of a flow f is defined as follows:

$$C(f) = \sum_{p \in \mathcal{P}} l_p(f_e) f_p \tag{3.48}$$

Given (3.46) and (3.47), it can be easily shown that C(f) depends solely on the edge flow, and is given by

$$C(f) = \sum_{e \in \mathcal{E}} l_e(f_e) f_e.$$
(3.49)

Hereafter, for notational brevity, we will drop the argument f whenever it is clear from the context.

Given these definitions, we can formulate a non-cooperative game where each flow is composed of an infinite number of players, each of which carries an infinitesimal amount of flow. Hence, the flow of a single player has almost no effect on the latency of a path. Each player selects a pure strategy by choosing one path from its origin to its destination. The cost for a player is the chosen path's latency. Here, each player attempts to minimize its latency.

Consequently, we can define the Wardrop equilibrium as follows [501]:

DEFINITION 3.27 A feasible flow vector f is at Wardrop equilibrium if for every commodity $i \in [k]$ and paths $p_1, p_2 \in \mathcal{P}_i$ with $f_{p_1} > 0$, it holds that

$$I_{\rho_1}(f) \le I_{\rho_2}(f). \tag{3.50}$$

In the Wardrop equilibrium, a flow vector is considered stable when no fraction of the flow can improve its sustained latency by moving unilaterally to another path. In this context, a network at the Wardrop equilibrium satisfies Wardrop's two principles [501]:

- Wardrop's first principle. All used paths from a source to a destination have equal mean latencies.
- Wardrop's second principle. Any unused path from a source to a destination has greater potential mean latency than that along the used paths.

Moreover, the latency of a flow f^* at the Wardrop equilibrium can be expressed as

$$C(f^*) = \sum_{i \in [k]} L_i(f^*) \cdot d_i, \qquad (3.51)$$

where $L_i(f^*)$ denotes the unique path latency of an equilibrium flow in commodity *i*. Note that, although the Nash equilibrium and the Wardrop equilibrium are related, in the sense that they both describe a stable network flow, they can still be different in many cases. For instance, while the Wardrop equilibrium satisfies the Wardrop principles, a Nash equilibrium may not. Notably, when the number of players is finite, a Nash equilibrium can be achieved without the latencies of all used paths being equal. This is mainly because the Wardrop equilibrium and its principles are motivated by the assumption that the contribution of an individual player is negligible, i.e., there exists an infinite number of players for each flow. Nonetheless, it has been shown that, under some conditions, the Wardrop equilibrium represents a limiting case of the Nash equilibrium when the number of players goes to infinity [200].

Furthermore, it can be shown [65] that the Wardrop equilibria are exactly the allocations that minimize the following potential function:

$$\Phi(f) = \sum_{e \in \mathcal{E}} \int_0^{f_e} l_e(u) du.$$
(3.52)

The existence of a potential function, as discussed in the case of potential games in Section 3.4.1, is sufficient to guarantee the existence of at least one Wardrop equilibrium. Moreover, every Wardrop equilibrium induces the same edge latencies and can be computed in polynomial time (owing to the presence of a potential function). Note that this existence result requires the latency functions to be continuous and monotone, but not necessarily semi-convex.

We make two observations on the potential function in (3.52). First, this function precisely absorbs progress: If an infinitesimal amount of flow du is shifted from path p_1 to path p_2 , thus improving its latency by $l_{p_1} - l_{p_2}$, the potential decreases by $(l_{p_1} - l_{p_2})du$. Furthermore, let f^* be a flow minimizing the potential function Φ . If an infinitesimal amount of flow du is shifted from path p_1 to path p_2 , transforming the flow f^* to g, it follows that $l_{p_1} - l_{p_2} = \Phi(g) - \Phi(f^*) \ge 0$. Hence, the fraction of deviating players are not able to benefit from the move.

The existence of a Wardrop equilibrium makes it a useful concept in many routing and flow-control problems. In wireless and communication networks, the Wardrop equilibrium has mainly been used in flow control and routing in communication networks [121, 28], in dense ad hoc networks [458, 35], as well as in routing in wireless networks [396]. In Chapter 14 we discuss the use of the Wardrop equilibrium within the context of flow control in communication networks, as per the work in [28].

Finally, we note that the Wardrop equilibrium also has some drawbacks. Two main ones are (i) the need for a large number of players, and (ii) the need for accurate knowledge of the network and its latency functions. In this context, several alternatives can be used,

such as the robust Wardrop equilibrium [151], the approximate Wardrop equilibrium for a finite number of players [375], or even the correlated equilibrium.

3.5 Summary

Non-cooperative games constitute an important branch of game theory. Many of the games that we will see in the following chapters, such as Bayesian games and differential games, are also generally classified under the umbrella of non-cooperative games. Static games, dynamic games, Stackelberg games, potential games, and others provide a wide variety of analytical tools for studying many important problems in various disciplines.

In wireless and communication networks, the use of non-cooperative games has been abundant, ever since the earliest work on power-control games [425]. For instance, there exist very strong connections and analogies between non-cooperative game theory and classical problems in wireless and communication networks such as routing, interference management, flow control, packet forwarding, and multiple access. In this context, readers interested in modeling communication problems using the non-cooperative-game framework are referred to Part III of this book, where we will develop several models that use many of the concepts introduced in this chapter, such as Nash equilibria and Stackelberg equilibria. The game models discussed in the preceding chapters were all built on the governing assumption that the players all have *complete information* on the elements of the game, particularly on the action spaces of all players and the players' payoff (or cost) functions, and that this is all common information to all players. However, in many situations, especially in a competitive environment, the *a priori* information available to a player may not be publicly available to other players. In particular, a player may not have complete information on his opponents' possible actions, strategies, and payoffs. For example, in the latter case, a player may not know the resulting payoff value for another player when all players have picked specific actions (or strategies). One way of addressing situations that arise as a result of such incompleteness of information is to formulate them as *Bayesian games*—the topic of this chapter. We first introduce this class of games in general terms, and then discuss applications in wireless communications and networking.

4.1 Overview of Bayesian games

4.1.1 Simple example

Let us consider the example of a game between two car companies responding to the possibility of a Clean Air Act, such as the one in 1990. If the Act is passed, then both car companies will be faced with the task of redesigning their cars, which will be costly. In order to prevent this from happening, they decide to start a lobbying campaign against the Act. But lobbying will cost money, although less than what would it cost to redesign their cars. Let us assume that this lobbying cost is known by both companies. With this information, the underlying game model can be described as follows: the players are the car companies (e.g., companies A and B); the strategies are whether to contribute or not to the lobbying effort; the payoff for a player depends on the actions taken by both players. In the first case, if neither company contributes, the Act will then pass and the payoffs for both companies will be low as a result of smaller profits (or higher costs). The Act will not pass if at least one company contributes to the lobbying effort. If only one company lobbies (the second case), then the payoff for that company will be higher than that in the first case. However, the other company without contribution will receive a free ride and the payoff for this company will be even higher than that of the company that contributes. In the third case, both companies contribute to the lobbying effort, and both end up with higher payoffs than those in the first case. These payoffs are also higher than that of the company that lobbies in the second case, but is smaller than the payoff for the company which does not contribute (again in the second case). A typical payoff matrix consistent with this description is as follows:

	B contributes	B does not contribute	
A contributes	(3,3)	(<u>2, 5</u>)	. (4.1)
A does not contribute	(<u>5, 2</u>)	(1,1)	-

The first and second figures in each pair are the payoffs for companies A and B, respectively. Clearly, if both companies knew the payoffs for each other perfectly, then Nash equilibrium would be the appropriate solution concept, which dictates that only one company should contribute to the lobbying effort.

What would happen if one company did *not* know the preference (e.g., payoff function) of the other company? For example, company B may not be sure whether company A has any new technology which will make it easier and less costly to comply with the Clean Air Act. If company A has the new technology, its payoff will be different. In this case, given that company B does not know whether company A has a new technology, we can introduce multiple types of players, such as company A with or without the new technology. In general, any player may have multiple types, with probabilities attached to each.

With the new technology (in our example), the payoff matrix in (4.1) would be modified as follows:

	B contributes	B does not contribute	
A contributes	(2,3)	(1, <u>5</u>)	. (4.2)
A does not contribute	(<u>4, 2</u>)	(<u>2</u> , 1)	-

This captures the scenario that if neither company contributes, then (as compared with the first game) the payoff for company A would be higher than that without the new technology. Otherwise, the payoff for company A is lower because of the cost of acquiring the new technology. This game has a unique Nash equilibrium, which dictates that B should contribute but A should not.

From this example, it can be seen that the sets of Nash equilibria are different in two games which differ only in the types of players. In general, the strategy of a player will depend not only on its own type but also on the probability distribution (i.e., belief) a player carries about the types of other players. To model and analyze such situations, we next introduce the static Bayesian game framework, where Bayesian Nash equilibrium is considered to be the appropriate solution concept, with uncertainty about the player types.

4.1.2 Static Bayesian game

In general, a game with incomplete information can be considered a Bayesian game [160], which is formally defined as follows. We have a set of players $\mathcal{N} = \{1, ..., N\}$, where N

is the total number of players. player *i*'s action space is denoted by \mathcal{A}_i and his type space by \mathcal{T}_i , for $i \in \mathcal{N}$. The payoff for player *i* is defined as a function of the types and actions of all players, i.e., $\mathcal{U}_i(\mathbf{a}; \mathbf{t})$, where **a** is a vector of actions taken by all players and **t** is a vector of types of all players. Belief for player *i* is defined as a conditional probability mass function of the types of all other players, given his own type, i.e., $\Pr_i(\mathbf{t}_{-i}|t_i)$, where $\mathbf{t}_{-i} \in \mathcal{T}_{-i}$ is a vector of types of all players except player *i* and $t_i \in \mathcal{T}_i$ is the type of player *i*. This belief formulation captures the uncertainty, which is at the heart of the formulation of a game with incomplete information.

In a Bayesian game, the different events occur in the following order:

- 1. Nature chooses the types of all players.
- 2. Players observe their own types, as drawn by nature. A full description of a particular player's type is known only by that player.
- Players simultaneously choose their actions. In particular, player *i* chooses an action a_i ∈ A_i based on his belief about the types of other players given his own type.
- 4. Players receive the payoff values.

As an example of a Bayesian game formulation, we consider the power-control game with incomplete information [202]. The players are the transmitters, and the action for each player is the transmit power. The type of player can be defined as the channel gain to the target receiver. This channel gain is known by the corresponding transmitter, but not by other transmitter–receiver pairs. Other transmitters know only the probability distribution for the channel gain of other players. In this case, the payoff for one player is a function of the transmit power of all players and of the channel gains.

In a Bayesian game, a strategy (i.e., the complete plan of action to be taken) is defined as a function of the self type, i.e., $s_i(t_i)$. In this case, strategy $s_i(t_i)$ determines the action from set A_i given the type t_i .

The formal definition of a Bayesian game, as a game of incomplete information, is given as follows:

DEFINITION 4.1 A game of incomplete information *is defined through the following elements:*

- *1. A set of players:* $i \in N = \{1, 2, ..., N\}$ *.*
- 2. A set of actions available to player i: A_i for $i \in \{1, 2, ..., N\}$, with $a_i \in A_i$ denoting a typical action for player i.
- 3. Sets of possible types for player i: T_i for $i \in \{1, 2, ..., N\}$, with $t_i \in T_i$ denoting a typical type of player i.
- 4. Let $\mathbf{a} = (a_1, \dots, a_N)$, $\mathbf{t} = (t_1, \dots, t_N)$, $\mathbf{a}_{-i} = (a_1, \dots, a_{i-1}, a_{i+1}, \dots, a_N)$, $\mathbf{t}_{-i} = (t_1, \dots, t_{i-1}, t_{i+1}, \dots, t_N)$.
- 5. Nature's move: **t** is selected according to a joint probability mass function $Pr(\mathbf{t})$ on $\mathcal{T} = \mathcal{T}_1 \times \cdots \times \mathcal{T}_N$, which induces natural conditional probabilities.
- 6. Strategies: $s_i : T \to A_i$, for $i \in \{1, 2, ..., N\}$, where $s_i(t_i) \in A_i$ is the action of player i of type i.
- 7. payoffs $\mathscr{U}_i(\mathbf{a}; \mathbf{t}), i \in \mathcal{N}$.

An appropriate solution concept for this game is the Bayesian Nash equilibrium, which is defined as the strategy N-tuple $\mathbf{s}^* = (\mathbf{s}_1^*, \dots, \mathbf{s}_N^*)$ satisfying for all $i \in \mathcal{N}$,

$$s_{i}^{*}(t_{i}) = \max_{a_{i} \in \mathcal{A}_{i}} \sum_{t_{-i} \in \mathcal{T}_{-i}} \mathscr{U}_{i}(s_{1}^{*}(t_{1}), \dots, s_{i-1}^{*}, a_{i}, s_{i+1}^{*}(t_{i+1}), \dots, s_{N}^{*}(t_{N}); \mathbf{t}) \mathsf{Pr}(\mathbf{t}_{-i}|t_{i}).$$
(4.3)

Note that what is being maximized above is the conditional average payoff for player *i* given his own type t_i . This Bayesian Nash equilibrium ensures that if the strategies of all players except the *i*th are fixed at s_{-i}^* , then player *i* cannot improve his payoff by moving away from s_i^* , this being so for all players.

What is introduced above can be qualified as a pure-strategy Bayesian Nash equilibrium (BNE). Just as in the case of finite (non-Bayesian) games we have considered earlier, a BNE may not exist in pure strategies. In this case we have to extend the definition of BNE to encompass mixed strategies, defined as a probability distribution for each player on his action set for each of his types (that is, a different probability distribution for each type). With such an extension of strategy spaces from pure to mixed strategies, it is known that every finite incomplete information game formulated as above has a BNE in mixed strategies. The reader can find more detail on non-cooperative games with incomplete information and BNE in [195].

4.1.3 Bayesian dynamic games in extensive form

Bayesian Nash equilibrium results in some complicated equilibria in a dynamic game, where players make multiple moves, gaining information along the way. The intricacies that arise here are similar to those in complete information games, which admit a plethora of equilibria under dynamic information. Such a multiplicity can be tamed in complete information games through various refinement schemes, such as subgame perfectness. This, however, is not always possible in incomplete information games there is only one subgame – the entire game – and so every Nash equilibrium is trivially subgame-perfect.

To refine the equilibria generated by the Bayesian Nash solution concept or subgame perfection, one can apply the perfect Bayesian equilibrium (PBE) solution concept. PBE is in the spirit of subgame perfection in that it demands that subsequent play be optimal. However, it places player beliefs on decision nodes that enable moves in non-singleton information sets to be dealt with more satisfactorily.

DEFINITION 4.2 A perfect Bayesian equilibrium *is a strategy profile and a set of beliefs for each player such that:*

- 1. At every information set, player i's strategy maximizes its payoff, given the strategies of all other players, and player i's beliefs.
- 2. At information sets reached with positive probability when PBE strategy is played, beliefs are formed according to strategy and Bayes' rule when necessary.

3. At information sets that are reached with probability zero when PBE strategy is played, beliefs may be arbitrary but must be formed according to Bayes' rule when possible.

So far in discussing Bayesian games, it has been assumed that information is perfect (or, if imperfect, play is simultaneous). In examining a dynamic game, however, it might be necessary to have the means to model imperfect information as well. PBE affords this means: players place beliefs on nodes occurring in their information sets, which means that the information set can be generated by nature (in the case of incomplete information) or by other players (in the case of imperfect information).

The beliefs held by players in a Bayesian game can be approached more rigorously in PBE. A belief system is an assignment of probabilities to every node in the game such that the sum of probabilities in any information set is 1. The beliefs of a player are exactly those probabilities of the nodes in all the information sets at which that player has the move (a player's belief might be specified as a function of the union of his information sets to [0,1]). A belief system is consistent for a given strategy profile if and only if the probability assigned by the system to every node is computed as the probability of that node being reached given the strategy profile, i.e., by Bayes' rule.

The notion of *sequential rationality* is what determines the optimality of subsequent play in PBE. A strategy profile is sequentially rational at a particular information set for a particular belief system if and only if the expected payoff for the player whose information set it is (i.e., who has the move at that information set) is maximal given the strategies played by all the other players. A strategy profile is sequentially rational for a particular belief system if it satisfies the above for every information set.

4.1.4 Cournot duopoly model with incomplete information

To demonstrate the formulation of a Bayesian game, the Cournot duopoly model with incomplete information is considered (Fig. 4.1). In this game, there is one product and two suppliers, namely suppliers 1 and 2 as the players, supplying the product in quantities s_1 and s_2 (which are actions) to the market. The price of the product is a function of the total supply, that is p(s) = A - s, where A is a constant and s is the total supply from two suppliers, i.e., $s = s_1 + s_2$. Note that the price decreases as the aggregated supply increases. Payoffs for the two suppliers are taken as the profit functions, defined as





follows:

$$\mathscr{U}_{i}(s_{1}, s_{2}) = P(s_{1} + s_{2})s_{i} - \mathscr{C}_{i}(s_{i}), \qquad (4.4)$$

where $\mathscr{C}_i(s_i)$ is the cost to supplier *i* of supplying the amount s_i to the market. For supplier 1, this cost is defined as $\mathscr{C}_1(s_1) = C_1 s_1$, where C_1 is the cost per unit of product. However, for supplier 2, the cost is defined as $\mathscr{C}_2(s_2) = C_{2H}s_2$ with probability α , where C_{2H} is the high-cost constant (e.g., the new production or logistic technology is not successfully deployed). Alternatively, the cost to supplier 2 is defined as $\mathscr{C}_2(s_2) = C_{2L}s_2$ with probability $1 - \alpha$, where $C_{2L} < C_{2H}$ is the low-cost constant (e.g., the new technology is successfully deployed). These high- and low-cost constants are defined as the type of supplier 2.

At the time of decision on the amount of supply to the market, suppliers 1 and 2 know their own cost functions (i.e., $\mathscr{C}_1(s_1)$ and $\mathscr{C}_2(s_2)$, respectively). Supplier 2 knows the cost function for supplier 1. However, supplier 1 does not know the exact cost function for supplier 2 (i.e., whether it is low- or high-cost). Supplier 1 knows, however, the two cost functions and the probability α for supplier 2. Since supplier 2 knows exactly its own cost, the best response (i.e., the optimal supply such that the profit is maximized) can be obtained given the cost (i.e., type), as follows:

$$s_{2}^{*}(C_{2H}) = \arg\max_{s_{2}} ((A - s_{1}^{*} - s_{2}) - C_{2H})s_{2}$$
(4.5)

for the high-cost constant, and

$$s_{2}^{*}(C_{2L}) = \arg\max_{s_{2}} ((A - s_{1}^{*} - s_{2}) - C_{2L})s_{2}$$
(4.6)

for the low-cost constant. $s_2^*(c)$, where $c \in \{C_{2H}, C_{2L}\}$, is obtained from the necessary condition of optimality, as follows (in this case this is also sufficient, because of strict concavity):

$$0 = \frac{\partial ((A - s_1^* - s_2) - c)s_2}{\partial s_2} = -2s_2 + A - s_1^* - c, \qquad (4.7)$$

$$s_2^*(c) = \frac{A - s_1^* - c}{2}.$$
(4.8)

However, supplier 1 knows that supplier 2 will choose the amount of supply $s_2^*(c)$ based on its own cost *c*. Given that the probability α is known by supplier 1, the best response can be defined as follows:

$$s_1^* = \arg\max_{s_1} \alpha((A - s_1 - s_2^*(C_{2H})) - C_1)s_1 + (1 - \alpha)((A - s_1 - s_2^*(C_{2L})) - C_1)s_1.$$
(4.9)

Again, s_1^* can be obtained from the necessary condition of optimality (which is, again, also sufficient), as follows:

$$0 = \frac{\partial \alpha ((A - s_1 - s_2^*(C_{2H})) - C_1)s_1 + (1 - \alpha)((A - s_1 - s_2^*(C_{2L})) - C_1)s_1}{\partial s_1},$$

$$s_1^* = \frac{\alpha (A - s_2^*(C_{2H}) - C_1) + (1 - \alpha)(A - s_2^*(C_{2L}) - C_1)}{2}.$$
(4.10)

The solutions for $s_2^*(C_{2H})$, $s_2^*(C_{2L})$, and s_1^* are

$$s_{2}^{*}(C_{2H}) = \frac{A - 2C_{2H} + C_{1}}{3} + \frac{(1 - \alpha)(C_{2H} - C_{2L})}{6}, \qquad (4.11)$$

$$s_{2}^{*}(C_{2L}) = \frac{A - 2C_{2L} + C_{1}}{3} - \frac{\alpha(C_{2H} - C_{2L})}{6}, \qquad (4.12)$$

$$s_1^* = \frac{A - 2C_1 + \alpha C_{2H} + (1 - \alpha)C_{2L}}{3}.$$
(4.13)

The solutions (4.11)–(4.13) constitute the Bayesian Nash equilibrium. Supplier 2 chooses a strategy according to the type (i.e., C_{2H} or C_{2L}), while supplier 1 chooses a strategy regardless of the type of supplier 2.

Note that the Nash equilibrium in the Cournot duopoly model with complete information for supplier *i* is $s_i^{n*} = \frac{(A-2c_i+c_j)}{3}$, where the other supplier is *j*, and c_i and c_j are the cost constants of supplier *i* and *j*, respectively. It is observed that in the Cournot duopoly model, the expected solution for supplier 1 in the incomplete information game equals the Nash equilibrium in the complete information game (owing to the quadratic nature of the payoff functions). However, in general this is not always true, and the Bayesian Nash equilibrium has to be computed by considering the probability distribution for the types of opponents rather than the expected value of type.

4.1.5 Auction with incomplete information

Competition among players in an auction can be formulated as a Bayesian game [166]. Specifically, we consider here the first-price, sealed-bid auction. In such an auction, two bidders (i.e., i = 1, 2) compete to buy a good by submitting non-negative bids (i.e., b_i and b_j) simultaneously. The bids are sealed and no bidder knows the bids of the others. The auctioneer determines the winner, the one with the highest bid. The winning bidder is committed to paying the price in the submitted bid. If both bidders submit the same price, then the winner is randomly selected. Bidder *i* values the good at v_i , and naturally his bid cannot exceed this valuation. This means that if the bidder *i* wins the auction, the price paid to the auctioneer to obtain the good is $p = b_i$ and the utility for this bidder *i* is $v_i - p$. Otherwise, his utility is zero. In this auction, each bidder knows its own value for the good, but not the valuation by the other bidder. Each bidder assumes the value of the other bidder to be random in $[0, v_{max}]$. The distribution of the value is assumed to be uniform and independent.

Since the bidders are rational and seek to maximize their payoffs, and the valuations of the bidder are not publicly available, the framework of the Bayesian game is appropriate here. The players in this game are the bidders. The action for each player is the bid submitted to the auctioneer. The type is defined to be the valuation for each player, and the payoff is the utility. The action space of bidder *i* is defined as $A_i = [0, \infty)$, and the

type space is $T_i = [0, v_{max}]$. The payoff function for bidder *i* can be expressed as

$$\mathscr{U}_{i}(\mathbf{b},\mathbf{v}) = \begin{cases} v_{i} - b_{i}, & \text{if } b_{i} > b_{j}, \\ \frac{v_{i} - b_{i}}{2}, & \text{if } b_{i} = b_{j}, \\ 0, & \text{if } b_{i} < b_{j}, \end{cases}$$
(4.14)

where $\mathbf{b} = \begin{bmatrix} b_1 & b_2 \end{bmatrix}^T$ and $\mathbf{v} = \begin{bmatrix} v_1 & v_2 \end{bmatrix}^T$ are the vectors of bids and valuations, respectively. In this case, the action of bidder *i* can be defined as a function of its own type, i.e., $b_i(v_i)$. The best response of bidder *i* can be obtained by maximizing his payoff, i.e.,

$$b_i^*(v_i) = \mathscr{B}_i(v_i, b_j(\cdot)) = \arg\max_{b_i} \int_0^{v_{\max}} \mathscr{U}_i(\mathbf{b}, \mathbf{v}) f_v(v_j) dv_j$$
(4.15)

$$= \arg \max_{b_i} (v_i - b_i) \Pr(b_i > b_j(v_j)) + \frac{1}{2} (v_i - b_i) \Pr(b_i = b_j(v_j)), \quad (4.16)$$

where $f_v(v_j)$ is the probability density function of v_j . The Bayesian Nash equilibrium is defined as $b_i^*(v_i) = \mathscr{B}_i(v_i, b_j^*(\cdot))$ and $b_j^*(v_j) = \mathscr{B}_j(v_j, b_i^*(\cdot))$, which can be obtained by solving (4.15). To obtain explicit results, let us restrict the players' strategies to affine functions, of the form $b_i(v_i) = \alpha_i + \beta_i v_i$.

If bidder *j* adopts the affine strategy $b_j = \alpha_j + \beta_j v_j$ where it is natural to assume that $\beta_j > 0$, the best response of bidder *i* can be expressed as follows:

$$b_i^*(v_i) = \mathscr{B}_i(v_i, b_j(\cdot)) = \arg\max_{b_i}(v_i - b_i) \Pr(b_i > \alpha_j + \beta_j v_j), \quad (4.17)$$

where the case of $b_i = b_j(v_j)$ is ignored because $Pr(b_i = b_j(v_j)) = 0$. We now obtain

$$\Pr(b_i > \alpha_j + \beta_j v_j) = \Pr\left(v_j < \frac{b_i - \alpha_j}{\beta_j}\right) = \frac{b_i - \alpha_j}{\beta_j v_{\max}}.$$
(4.18)

The best response of bidder *i* can be obtained from the necessary condition of optimality, as follows:

$$0 = \frac{\partial (v_i - b_i) \left(\frac{b_i - \alpha_j}{\beta_j}\right)}{\partial b_i} = v_i - 2b_i + \alpha_j.$$
(4.19)

That is,

$$b_i^*(v_i) = \mathscr{B}_i(v_i, b_j(v_j)) = \begin{cases} \frac{v_i + \alpha_j}{2}, & \text{if } v_i \ge \alpha_j, \\ \alpha_j, & \text{if } v_i < \alpha_j. \end{cases}$$
(4.20)



Fig. 4.2 Best response of bidder *i* given $\alpha_j = 1$.

Figure 4.2 shows the best response of bidder *i* given $\alpha_j = 1$. The best response of bidder *i* is constant and equal to α_j if v_i is less than α_j . However, if v_i is equal to or greater than α_j , the best response of bidder *i* will increase linearly with v_i . For the linear strategy, only the cases $\alpha_j \ge v_{\text{max}}$ and $\alpha_j \le 0$ will be considered. The case of $\alpha_j \ge v_{\text{max}}$ cannot be the equilibrium since the best response for a higher type to bid must be at least as much as a lower type's best response (i.e., $\beta_j \ge 0$). In this case, $\alpha_j \ge v_{\text{max}}$ will result in $b_j^*(v_j) \ge v_j$ which cannot be the best response. As a result, if $b_i^*(v_i)$ is linear, only the case $\alpha_j \le 0$ is valid, where $b_i^*(v_i) = \frac{v_i + \alpha_j}{2}$, $\alpha_i = \alpha_j/2$, and $\beta_i = 1/2$. When the same procedure is applied to bidder *j*, we have to consider only the case $\alpha_j = \alpha_i/2$ and $\beta_j = 1/2$. We conclude that at the Bayesian Nash equilibrium, $\alpha_i = \alpha_j = 0$ and $\beta_i = \beta_i = 1/2$, and $b_i^*(v_i) = v_i/2$.

4.2 Applications in wireless communications and networking

In this section, some applications of the Bayesian game framework in wireless communications and networking are discussed. The Bayesian game framework has been adopted to solve various problems that arise in this context, specifically when the players lack complete information on the types of their opponents.

4.2.1 Packet-forwarding game

In a multi-hop network, one major issue is packet forwarding. A node in a multi-hop network could be cooperative in forwarding a packet for the other nodes. In such a scenario, the specific type of the node will be important for the payoffs and actions (e.g., monitoring against attack) of the nodes in the multi-hop network. The problem can then be formulated as a Bayesian game where the type (i.e., regular or malicious) of a node is private information and cannot be known by the other nodes [498].





The system model considered for the Bayesian game formulated in [498] is shown in Fig. 4.3. There are two nodes, sender and receiver, denoted by *i* and *j*, respectively. These nodes are players of the game. The sender receives the packet from an upstream node. The game starts when the sender decides what to do with the received packet. The type of sender can be regular or malicious, i.e., $t_i \in T_i = \{\text{regular}, \text{malicious}\}$. A regular sender will forward the received packet to the receiver, while a malicious node (i.e., malicious sender) will attack the receiver (e.g., falsify the received packet and send it to the receiver). A receiver is only of one type or the other in this game. The strategy space (i.e., possible actions to be taken) of the sender depends on the type. If the type of node is regular, the action it will take is $s_i(t_i = \text{regular}) = \{\text{forward}\}$. That is, a regular sender will always forward the packet to the receiver. However, if the node is malicious, the possible actions it will take are $s_i(t_i = \text{malicious}) = \{\text{forward}, \text{attack}\}$. That is, a malicious sender can forward the packet or attack the receiver. The receiver can monitor to prevent the attack, or stay idle. Therefore, the strategy space of receiver *j* is $S_j = \{\text{monitor}, \text{idle}\}$.

We next define the payoffs for the sender and the receiver. The cost to a malicious sender *i* to attack is denoted by C_A . The probability of success of an attack is given by ψ . If the attack on the receiver is successful, the malicious sender receives a payoff of G_A . The cost for a regular sender to forward the packet is C_F . Monitoring the attack incurs a cost of C_M for the receiver, and staying idle has zero cost. The cost for the receiver of being attacked is $-G_A$. The payoff matrix for the case of a malicious sender can be expressed as follows:

MonitorStay idleAttack
$$(-G_A - C_A, G_A - C_M)$$
 $(G_A - C_A, -G_A)$ Forward $(-C_F, -C_M)$ $(-C_F, 0)$

and for the case of a regular sender it is as follows:

$$\frac{\text{Monitor Stay idle}}{\text{Forward } (-C_F, -C_M) (-C_F, 0)}$$
(4.22)

where the sender and receiver take actions in the rows and columns, respectively. The belief of the receiver is defined as the probability α for the sender to be malicious and $1 - \alpha$ for the sender to be regular. Note that the receiver can monitor and prevent attack successfully with a probability $1 - \beta$. The probability β could be the channel loss rate.

The Bayesian Nash equilibria in pure and mixed strategies are analyzed for this game below.

Pure strategy

For the pure strategy, the game can possess a Bayesian Nash equilibrium if the belief of the receiver is higher than a threshold denoted by α_0 . The analysis is as follows. The malicious sender always attacks and the regular sender always forwards. For the receiver *j*, the expected payoff for the monitoring strategy is obtained from

$$\mathscr{U}_{j}(\text{monitor}) = (G_{A} - C_{M})\alpha(1 - \beta) - \alpha\beta(\psi(G_{A} + C_{M}) + (1 - \psi)C_{M}) - (1 - \alpha)C_{M}.$$
(4.23)

 $(G_A - C_M)\alpha(1 - \beta)$ is the payoff for successfully monitoring and preventing the attack. $\alpha\beta(\psi(G_A + C_M) + (1 - \psi)C_M)$ is the payoff for failing to monitor and prevent the attack. $(1 - \alpha)C_M$ is the payoff if the sender is regular. The payoff for the receiver of staying idle is obtained from

$$\mathscr{U}_{i}(idle) = -\alpha \psi \mathcal{G}_{\mathcal{A}}.$$
(4.24)

If $\mathscr{U}_j(\text{idle}) > \mathscr{U}_j(\text{monitor})$, then the receiver will stay idle. Therefore, the best response for sender *i* would be to attack if the sender is malicious and to forward if the sender is regular. This best response of the sender and the staying-idle strategy of receiver is the Bayesian Nash equilibrium under the condition $\mathscr{U}_j(\text{idle}) > \mathscr{U}_j(\text{monitor})$. In this case, it can be shown that this Bayesian Nash equilibrium can be achieved if $\alpha < \frac{C_M}{(1-\beta)(1+\psi)G_A}$ [498].

For $\mathscr{U}_j(\text{idle}) < \mathscr{U}_j(\text{monitor})$, the best response for the receiver will be to monitor. Then, the best response of the sender to the monitoring strategy of the receiver will be the forwarding strategy, regardless of the type. Therefore, there is no Bayesian Nash equilibrium under this condition. In this case, the threshold is determined from $\alpha_0 = \frac{C_M}{(1-\beta)(1+\psi)G_A}$ to achieve the Bayesian Nash equilibrium.

Mixed strategy

The pure-strategy Bayesian Nash equilibrium may not be desirable since the receiver always stays idle. This solution may degrade the performance of the multi-hop network significantly since an attack can easily occur. Therefore, a mixed-strategy Bayesian Nash equilibrium is obtained as an alternative. Let ϕ and ϕ' denote, respectively, the probability that the malicious sender *i* chooses an attacking strategy and the probability that the receiver chooses a monitoring strategy. The mixed-strategy Bayesian Nash equilibrium is obtained as the probabilities ϕ and ϕ' such that both sender and receiver cannot improve their payoffs. For the receiver, the payoff for the monitoring strategy can be obtained similarly to (4.23), as follows:

$$\mathcal{U}_{j}(s_{j} = \text{monitor}) = \alpha \phi(\psi(G_{A} - C_{M})(1 - \beta) + (1 - \psi)(1 - \beta)(G_{A} - C_{M}))$$
$$-(1 - \psi)\beta C_{M} - \psi\beta(G_{A} + C_{M})) - (1 - \phi)\alpha C_{M} - (1 - \alpha)C_{M}$$
$$= \alpha \phi(G_{A} - G_{A}\beta(1 + \psi)) - C_{M}.$$
(4.25)

Similarly to (4.24), the payoff of the staying-idle strategy for receiver j can be obtained from

$$\mathscr{U}_{i}(s_{i} = idle) = -\phi\psi\alpha G_{A}.$$
(4.26)

For the best response, the payoff of the monitoring strategy must be equal to the payoff of the staying-idle strategy. Therefore, the probability ϕ can be obtained from

$$\phi = \frac{C_M}{\alpha G_A (1+\psi)(1-\beta)}.$$
(4.27)

The probability ϕ' can be obtained in a similar way. First, the payoff for sender *i* of choosing the attacking strategy is obtained from

$$\mathcal{U}_{i}(s_{i} = \text{attack}) = -(G_{A} + C_{A})(1 - \beta)\phi' + (G_{A} - C_{A})\psi(1 - \phi') + (G_{A} - C_{A})\psi\phi'\beta - C_{A}(1 - \psi)\beta\phi' - C_{A}(1 - \phi')(1 - \psi)$$
(4.28)

$$= \phi' G_{A}(\beta - 1)(1 + \psi) + G_{A}\psi - C_{A}.$$
(4.29)

The payoff for the sender of choosing the forwarding strategy is obtained from

$$\mathscr{U}_i(s_i = \text{forward}) = -C_F. \tag{4.30}$$

Again, for the best response, the condition $\mathcal{U}_j(s_j = idle) = \mathcal{U}_i(s_i = attack)$ must be true. Therefore, probability ϕ' is

$$\phi' = \frac{G_A \psi - C_A + C_F}{G_A (1 - \beta)(1 + \psi)}.$$
(4.31)

The mixed-strategy Bayesian Nash equilibrium is obtained as follows. If the sender is malicious, the sender chooses an attacking strategy, with probability $\phi = \frac{C_M}{\alpha G_A(1+\psi)(1-\beta)}$. If the sender is regular, the sender always chooses a forwarding strategy. The receiver chooses a monitoring strategy with probability $\phi' = \frac{G_A\psi - C_A + C_F}{G_A(1-\phi)(1+\psi)}$. It is also shown that this mixed-strategy Bayesian Nash equilibrium can be reached only when the condition of belief, i.e., $\alpha > \alpha_0$, is true. Figure 4.4 shows the probability of a malicious sender attacking the receiver under varied belief. Clearly, as the belief by the receiver that the sender is malicious becomes larger, the probability of attack will decrease. Figure 4.5 shows the probability of the receiver monitoring and preventing the attack by the sender. As the probability of successful attack increases, naturally the receiver will increase the probability of monitoring.

Note that a similar game model for cooperative diversity can be found in [126].

4.2.2 *K*-player Bayesian water-filling game

For multiple-access control, power control is one of the important issues in CDMA and OFDM systems. This problem becomes more challenging when the users are rational,



Fig. 4.4 Probability of malicious sender attacking the receiver.



Fig. 4.5 Probability of receiver monitoring and preventing the attack by the sender.

seeking to maximize their transmission rates. Game models have been formulated to study this problem, which is known as the "water-filling game" [276]. In this game, the objective of user i is to maximize the transmission rate, defined as follows:

$$R_{i} = \log\left(1 + \frac{P_{i}h_{i,i}}{\sigma^{2} + \sum_{j \neq i}P_{j}h_{j,i}}\right),\tag{4.32}$$

where σ^2 is the noise power, P_i is the transmission power of user *i*, and $h_{j,i}$ is the channel gain from the transmitter of user *j* to the receiver of user *i*. With complete channel-state information (CSI), it is shown that there is a unique Nash equilibrium for the two-user case (i.e., N = 2). However, the assumption of complete CSI may not be practical in an environment in which measurement of channel gain can be noisy.

In [202] the water-filling game is extended to consider the incompleteness of information on channel gain. In particular, the user knows the exact value of its own channel gain from transmitter to receiver, but only the probability distribution of channel gains of the other users. A Bayesian game model is developed to obtain the Bayesian equilibrium of this water-filling game with incomplete information. As in the original water-filling game, the *N* players of this Bayesian game are the users accessing the channel. The action variable is the transmit power, which is bounded by the maximum threshold P_i^{max} for user (player) *i*. The type of player is the channel gain, which can take continuous values from h_i^{hi} to h_i^{low} . The probabilities of the channel gain to be h_i^{hi} and h_i^{low} are denoted by α_i^{hi} and α_i^{low} , respectively. The payoff for a player is the achievable rate, which can be defined as follows:

$$U_{i} = R_{i} = \log_{2} \left(1 + \frac{P_{i}(h_{i,i})h_{i,i}}{\sigma^{2} + \sum_{j \neq i} P_{j}(h_{j,i})h_{j,i}} \right),$$
(4.33)

where the strategy (i.e., transmit power) is defined as a function of type (i.e., channel gain). However, since the channel gain is random, the objective of each player can be defined as follows:

$$\max \quad \overline{U}_{i} = E_{\mathbf{h}} \left(\log_{2} \left(1 + \frac{P_{i}(h_{i,i})h_{i,i}}{\sigma^{2} + \sum_{j \neq i} P_{j}(h_{j,i})h_{j,i}} \right) \right), \tag{4.34}$$

s.t.
$$E_{h_{i,i}}(P_i(h_{i,i})) \le P_i^{\max}$$
, (4.35)

$$P_i(h_{i,i}) \ge 0, \tag{4.36}$$

where $E(\cdot)$ is the expectation, $\mathbf{h} = \{h_{j,i} | j = 1, ..., N\}$ is a set of channel gains, and P_i^{\max} is the average maximum power for player *i*. The optimization model defined in (4.34)–(4.36) is a convex optimization problem. Therefore, the solution can be obtained using Lagrangian duality [202]:

$$E_{\mathbf{h}_{-i}}\left(1 + \frac{h_{i,i}}{\sigma^2 + P_i(h_{i,i})h_{i,i} + \sum_{j \neq i} P_j(h_{j,i})h_{j,i}}\right) = \lambda_i,$$
(4.37)

where \mathbf{h}_{-i} is the set of channel gains of all players except player *i*, and λ_i is the dual variable.

With the Bayesian game model for the water-filling game with incomplete information, the Bayesian equilibrium is obtained as the strategy profile in which no player can gain a higher transmission rate while the other players keep their transmit powers unchanged. The existence of this Bayesian equilibrium is proved by considering the strategy space P_i to be convex, compact, and non-empty for each player. Also, the payoff function \overline{U}_i is continuous in the strategies of all players. This payoff function \overline{U}_i is concave in P_i for any P_{-i} .

The uniqueness of the Bayesian equilibrium is proved by considering the sufficiency condition. That is, the game has a unique equilibrium if the non-negative weighted sum

of the payoffs is diagonally strictly concave [202]. The sum of the average payoffs is

$$u(\mathbf{p}, \mathbf{w}) = \sum_{i=1}^{N} w_i \overline{U}_i(\mathbf{p}), \qquad (4.38)$$

where $\mathbf{p} = [P_1, ..., P_N]^T$ is the transmit power vector, and $\mathbf{w} = [w_1, ..., w_N]^T$ is the nonnegative weight vector. This $u(\mathbf{p}, \mathbf{w})$ is diagonally strictly concave for any vector \mathbf{p} and fixed vector \mathbf{w} if, for any two different vectors \mathbf{p}^0 and \mathbf{p}^1 ,

$$(\mathbf{p}^{1} - \mathbf{p}^{0})\delta(\mathbf{p}^{0}, \mathbf{w}) + (\mathbf{p}^{0} - \mathbf{p}^{1})\delta(\mathbf{p}^{1}, \mathbf{w}) > 0, \qquad (4.39)$$

where $\delta(\mathbf{p}, \mathbf{w})$ is the pseudo-gradient of $u(\mathbf{p}, \mathbf{w})$, defined as follows:

$$\delta(\mathbf{p}, \mathbf{w}) = \begin{bmatrix} w_1 \frac{\partial \overline{U}_1}{\partial P_1} \\ \vdots \\ w_N \frac{\partial \overline{U}_N}{\partial P_N} \end{bmatrix}.$$
(4.40)

The detailed steps of the proof can be found in [202].

Note that the channel gain can be derived from the path-loss factor as in [465]. With a CDMA network (Fig. 4.6), let R be the random variable representing the distance from the transmitter to the base station. The probability density function of this distance is

$$f_R(r) = \frac{d}{dr} \left(\frac{\pi r^2 - \pi R_{\min}^2}{\pi R_{\max}^2 - \pi R_{\min}^2} \right) = \frac{2r}{R_{\max}^2 - R_{\min}^2},$$
(4.41)

where R_{min} and R_{max} are the minimum distance (i.e., antenna far-field reference distance) and the maximum distance (i.e., cell radius), respectively.



Fig. 4.6 CDMA cell with unknown channel gain.

As in standard channel modeling, the path loss is [202]

$$L(R) = \overline{L}(R_{\min}) + 10n\log R - 10n\log R_{\min} + X_{\sigma}, \qquad (4.42)$$

where $\overline{L}(R_{\min})$ is the constant path loss at the reference distance R_{\min} , *n* is the propagation exponent, X_{σ} is the shadowing, and σ is the standard deviation. The path-loss probability density function $\alpha_L(I)$, which is the belief of the other user, can be expressed as [202]:

$$\alpha_{L}(l) = \begin{cases} \int_{a-C_{0}}^{l-C_{0}+3\sigma} g(x)dx, & \text{if } a-3\sigma < l \le a+3\sigma, \\ \int_{l-C_{0}-3\sigma}^{l-C_{0}+3\sigma} g(x)dx, & \text{if } a+3\sigma < l \le b-3\sigma, \\ \int_{l-C_{0}-3\sigma}^{b-C_{0}} g(x)dx, & \text{if } b-3\sigma < l \le b+3\sigma, \\ 0, & \text{otherwise,} \end{cases}$$
(4.43)

where

$$g(x) = \frac{10^{x/5n}}{R_{\max}^2 - R_{\min}^2} \frac{\ln 10}{10n \text{erf}(3)} \frac{\exp(-(I - C_0 - x)^2/2\sigma^2)}{\sqrt{2\pi\sigma^2}}.$$
 (4.44)

erf(·) is the standard error function, $C_0 = \overline{L}(R_{\min}) - 10n \log R_{\min}$, $a = \overline{L}(R_{\min})$, and $b = 10n \log R_{\max} + C_0$, where $[a - 3\sigma, b + 3\sigma]$ is the range of possible path loss.

The best response of user *i* can be defined as

$$P_{i}^{*}(I_{i}) = \arg \max_{P_{i}} \int_{\mathcal{L}_{-i}} \alpha_{L}(\mathbf{I}_{-i}) U_{i}(P_{i}, \mathbf{p}_{-i}(\mathbf{I}_{-i}), (I_{i}, \mathbf{I}_{-i})) d\mathbf{I}_{-i}, \qquad (4.45)$$

where I_{-i} is a vector of path loss and \mathcal{L}_{-i} is the space of path loss of all users except user *i*. Note that the best response defined in (4.45) is similar to the solution of (4.34)–(4.36). The Bayesian Nash equilibrium can be obtained from the best response of all users, e.g., by numerical methods.

4.2.3 Channel-access game

In the wireless LAN (WLAN) environment, channel access can be competitive among rational users. In particular, if multiple users transmit at the same time, collision will occur. To maximize throughput, the user has to choose a transmit or backoff action strategically. This competitive situation is formulated within a game model in [393, 223]. In [284], a Bayesian game model is proposed in view of the incomplete information about the channel gain. Also, the interference among transmissions by different users to different access points is also taken into consideration. Figure 4.7 shows an example of this network model with four users and two access points.

In the network model under consideration in [284], the channel gain between the transmitter of user *i* and the target access point is denoted by h_i . The interference created by this transmitter to the other user is denoted by βh_i , where β is the crosstalk interference



Fig. 4.7

Network model for channel access by multiple users to multiple access points with interference.

ratio, which is assumed to be identical for all users. Transmit power is also assumed to be identical for all users. In this case, the transmission of any user *i* can be successful if the SINR is higher than the threshold γ_{thr} , i.e.,

$$\gamma_i = \frac{Ph_i}{\beta \sum_{j \neq i} Ph_j + \sigma^2} \ge \gamma_{\text{thr}},\tag{4.46}$$

where P is the transmitted power. To maintain the SINR higher than the threshold, only a certain number of users can transmit simultaneously. Therefore, strategies of the users to transmit or back off have to be optimized. In the Bayesian game model formulated in [284], the players are the users. The type of user is the channel gain. A set of actions for all types is defined as {transmit, back off}. The payoff function is defined as follows:

$$U_i(\mathbf{s}_i, \mathbf{s}_{-i}) = \begin{cases} 0, & \text{if } \mathbf{s}_i = \text{back off,} \\ R_i(\mathbf{s}_{-i}) - C, & \text{if } \mathbf{s}_i = \text{transmit,} \end{cases}$$
(4.47)

where *C* is the cost of transmission (e.g., energy consumption) and $R_i(\cdot)$ is the throughput calculated from

$$R_{i}(\mathbf{s}_{-i}) = \begin{cases} \log(1+\gamma_{i}), & \text{if } \gamma_{i} \ge \gamma_{\text{thr}}, \\ 0, & \text{otherwise}, \end{cases}$$
(4.48)

where γ_i is obtained as in (4.46) for all *j* with s_j = transmit. It is assumed that the channel follows independent Rayleigh fading, in which the probability distribution function of channel gain h_i is $\Pr(h_i) = \rho \exp(-\rho h_i)$, where $1/\rho$ is the average received power. Therefore, the belief of the user in this Bayesian game is given by $\Pr(\mathbf{h}_{-i}|h_i)$. Again, the Bayesian Nash equilibrium is defined as

$$s_i^*(h_i) = \arg\max_{s_i \in \mathcal{S}_i} E\left(U(s_i, \mathbf{s}_{-i}(\mathbf{h}_{-i}), h_i)\right), \tag{4.49}$$

for all h_i and for all users *i*. Note that expectation in (4.49) is conditioned on h_i . In this case, the desirable strategy is to transmit (i.e., to achieve non-zero payoff), which can be the Bayesian Nash equilibrium if

$$E\left(U_i(s_i = \text{transmit}, \mathbf{s}_{-i}(\mathbf{h}_{-i}), h_i)\right) \ge E\left(U_i(s_i = \text{back off}, \mathbf{s}_{-i}(\mathbf{h}_{-i}), h_i)\right).$$
(4.50)

An interesting result from [284] is that the Bayesian Nash equilibrium is a threshold strategy in the following form:

$$s_i(h_i) = \begin{cases} \text{transmit, if } h_i \ge h_{\text{thr},i}, \\ \text{back off, otherwise.} \end{cases}$$
(4.51)

In particular, user *i* will choose to transmit if his channel gain is larger than or equal to the threshold $h_{\text{thr},i}$. This finding is important since the user will know exactly whether to transmit by evaluating only his channel gain. The proof of this finding is based on the fact that the payoff is an increasing function of h_i for the transmitting strategy. The transmitting strategy will be the best response and, hence, Bayesian Nash equilibrium, if the payoff for this strategy is higher than that of the backoff strategy. In this case, the threshold will be the value such that the payoff for the transmitting strategy is equal to zero, i.e., $E(U_i(s_i = \text{transmit}, \mathbf{s}_{-i}(\mathbf{h}_{-i}, h_{\text{thr},i}))) = E(U_i(s_i = \text{back off}, \mathbf{s}_{-i}(\mathbf{h}_{-i}, h_{\text{thr},i}))) = 0$ as defined in (4.47). The user can choose the value of the threshold so as to achieve Bayesian Nash equilibrium.

For a simpler setting, the symmetric Bayesian Nash equilibrium is first considered with two identical users (i.e., $h_{\text{thr},i} = h_{\text{thr}}$). Let $\Lambda(x)$ denote the cumulative distribution function of the exponential random variable (due to Rayleigh fading channel), and $\Lambda'(x) = 1 - \Lambda(x)$. The payoff for user *i* for the transmitting strategy is obtained from

$$E(U_{i}(s_{i} = \text{transmit}, s_{j}(h_{j}), h_{i}))$$

$$= \Pr(h_{j} < h_{\text{thr}}) \left(\log \left(1 + \frac{h_{i}}{\sigma^{2}} \right) - C \right)$$

$$+ \Pr(h_{j} \ge h_{\text{thr}}) \left(\Pr\left(\frac{h_{i}}{\beta h_{j} + \sigma^{2}} \ge \gamma_{\text{thr}} \middle| h_{j} \ge h_{\text{thr}} \right)$$

$$\times E \left(\log \left(1 + \frac{h_{i}}{\beta h_{j} + \sigma^{2}} \right) - C \middle| h_{\text{thr}} \le h_{j} \le \frac{1}{\beta} \left(\frac{h_{i}}{\gamma_{\text{thr}}} - \sigma^{2} \right) \right)$$

$$+ \Pr\left(\frac{h_{i}}{\beta h_{j} + \sigma^{2}} < \gamma_{\text{thr}} \middle| h_{j} \ge h_{\text{thr}} \right) (0 - C) \right).$$

$$(4.52)$$

Then, because of the exponential distribution, the term $\Pr\left(\frac{h_i}{\beta h_j + \sigma^2} \ge \gamma_{\text{thr}} \middle| h_j \ge h_{\text{thr}}\right)$ becomes $\Lambda\left(\frac{1}{\beta}\left(\frac{h_i}{\gamma_{\text{thr}}} - \sigma^2\right) - h_{\text{thr}}\right)$, and (4.53) can be simplified to

$$E(U_{i}(s_{i} = \text{transmit}, s_{j}(h_{j}), h_{i}))$$

$$= \Lambda(h_{\text{thr}}) \left(\log \left(1 + \frac{h_{i}}{\sigma^{2}} \right) - C \right) + \Lambda'(h_{\text{thr}}) \left(\Lambda \left(\frac{1}{\beta} \left(\frac{h_{i}}{\gamma_{\text{thr}}} - \sigma^{2} \right) - h_{\text{thr}} \right) \right)$$

$$\times E \left(\log \left(1 + \frac{h_{i}}{\beta h_{j} + \sigma^{2}} \right) \right| h_{\text{thr}} \le h_{j} \le \frac{1}{\beta} \left(\frac{h_{i}}{\gamma_{\text{thr}}} - \sigma^{2} \right) - \beta \right).$$

$$(4.55)$$

Since $E(U_i(s_i = \text{transmit}, s_j(h_j), h_i))$ is an increasing function of h_i , there exists a symmetric h_{thr} for Bayesian Nash equilibrium such that $E(U_i(s_i = \text{transmit}, s_j(h_j), h_i)) \ge E(U_i(s_i = \text{back off}, s_j(h_j), h_i))$. Extensions to multiple users can be found in [284].

4.2.4 Bandwidth-auction game

When multiple users want to download/upload data from an access point, a bandwidthallocation problem arises. However, since the bandwidth of the access point is limited, the allocation can be based on an auction mechanism. In this case, the users submit bids (e.g., price per unit of bandwidth to be paid to the access point). Given the rule set by the access point, the bandwidth is allocated according to the bidding outcome. Based on the bandwidth demands, the users can strategically adjust their bidding strategies so that their payoff is maximized. Since the demand of all users may be larger than the total available bandwidth, the users will compete through the bidding strategy. This bandwidth auction can be modeled using game theory [231, 252, 218].

Without knowing the bandwidth demand of the other users, the bandwidth auction can be formulated as a Bayesian game [18]. In [18], a bandwidth auction among vehicular nodes in a vehicular network to download data from a roadside access point is considered. The access point is connected to the Internet. When a vehicular node moves into the coverage area of the access point, the vehicular node can connect to the access point and download data from the Internet. The total available bandwidth to connect to an access point is denoted by B, and is shared among N vehicular nodes. Vehicular node isubmits a bid (i.e., strategy) s_i to the access point, where s_i is the bidding price per unit of allocated bandwidth. Given the bids from all vehicular nodes, the amount of bandwidth g_i allocated to vehicular node i is

$$g_i = \frac{s_i t_i}{\sum_{j=1}^N s_j t_j} B,$$
(4.56)

where t_i is the duration of the connection of vehicular node *i* to the access point. The players of this game are the vehicular nodes connecting to the access point. The strategy is the bidding price as a function of the private information available to each player. The type is duration, which is a function of bandwidth demand. The payoff is the difference between utility and cost for the vehicular node, defined as follows:

$$\mathscr{U}_{i}(g_{i}, s_{i}, \mathbf{s}_{-i}, t_{i}, \mathbf{t}_{-i}) = t_{i}\mathscr{R}_{i}(g_{i}) - \delta_{i}g_{i}s_{i}t_{i}, \qquad (4.57)$$





where \mathbf{s}_{-i} is the vector of bidding prices and \mathbf{t}_{-i} is the vector of connection duration for all nodes except node *i*. δ_i is the cost weight, and $\mathcal{R}_i(g_i)$ is the utility, defined as follows:

$$\mathscr{R}_{i}(g_{i}) = \begin{cases} w_{1} \log(1 + w_{2}g_{i}), & g_{i} < \text{if } g_{\text{req},i}, \\ w_{1} \log(1 + w_{2}g_{\text{req},i}), & \text{if } g_{i} \ge g_{\text{req},i}, \end{cases}$$
(4.58)

where w_1 and w_2 are constants. $g_{req,i}$ is the bandwidth requirement of the vehicular node *i*.

In this Bayesian game model, the type of player is the connection duration t_i , whose value is exactly known by the vehicular node *i*. However, the vehicular node knows only the probability density function for the connection duration of other nodes (i.e., belief). This probability density function is assumed to be of the form

$$\alpha_{T}(t) = \frac{1}{\beta} \exp(-t/\beta) > 0, \qquad (4.59)$$

where β is the average connection duration. Then the best response (i.e., best bidding strategy) given the bidding and type of other players is

$$s_{i}^{*}(t_{i}) = \arg \max_{s_{i}} E\left(\mathscr{U}_{i}(g_{i}, s_{i}, \mathbf{s}_{-i}, t_{i}, \mathbf{t}_{-i})\right)$$

$$= \arg \max_{s_{i}} \int_{0}^{\infty} \cdots \int_{0}^{\infty} \alpha_{T}(t) \mathscr{U}_{i}(g_{i}, s_{i}, \mathbf{s}_{-i}, t_{i}, (t_{1}, \dots, t_{i-1}, t_{i+1}, \dots, t_{N}))$$

$$dt_{1} \cdots dt_{i-1} dt_{i+1} \cdots dt_{N}.$$

$$(4.61)$$

To obtain the Bayesian Nash equilibrium, an iterative algorithm is used:

- 1: Initialize iteration counter k = 1.
- 2: Access point receives $s_i[k]$, $t_i[k]$, and $g_{\text{thr},i}$ from all vehicular nodes.
- 3: Access point computes $g_i[k-1]$ from (4.56) and allocates bandwidth to the vehicular nodes.
- 4: repeat
- 5: $k \leftarrow k+1$.
- 6: $s_i^*[k] \leftarrow \arg\max_{s_i} E\left(\mathscr{U}_i(g_i[k-1], s_i[k-1], \mathbf{s}_{-i}[k-1], t_i[k-1], \mathbf{t}_{-i})\right)$
- 7: Vehicular node *i* sends $s_i^*[k]$ to access point.
- 8: Access point computes $g_i[k]$ from (4.56) and allocates bandwidth to the vehicular nodes.
- 9: **until** $\max_{i} |s_i^*[k] s_i^*[k-1]| \le \epsilon$.

 ϵ is a small number (e.g., $\epsilon = 10^{-6}$) used as the termination condition.

From the performance evaluation, it is shown that with Bayesian Nash equilibrium as the solution, the vehicular nodes can achieve higher utility than they can from the Nash equilibrium when the average duration is used to compute the payoff. This result confirms the usefulness of the Bayesian game formulation for the incomplete-information environment.

4.2.5 Bandwidth-allocation game

A bandwidth-allocation game is introduced to model users' competition to buy bandwidth from an Internet service provider (ISP) [450, 451] (Fig. 4.9). The ISP can adjust the price to maximize its profit. This results in a leader–follower (i.e., Stackelberg) game model where the ISP is the leader and the users are the followers. The total number of users is N. User *i* chooses the amount of bandwidth, denoted by s_i (i.e., strategy of follower), to buy from the ISP. The ISP determines the price to be charged to user *i*, denoted by p_i (i.e., strategy of leader). The analysis can be classified based on the availability of information, i.e., complete information, partially incomplete information, or totally incomplete information.

Complete information

The payoff is expressed as follows:

$$U_i = \mathscr{G}_i(s_i) + \mathscr{H}_i(\mathbf{s}) - p_i s_i, \qquad (4.62)$$

where **s** is a vector of bandwidth of all users. $\mathcal{G}_i(s_i)$ is the utility function due to the bandwidth. This function is assumed to be strictly increasing, strictly concave, and non-negative. $\mathcal{H}_i(\mathbf{s})$ is the service satisfaction function for the received QoS (e.g., delay and loss due to congestion). This service satisfaction depends on the bandwidth used by all other users (e.g., the outgoing link of the ISP is shared by all users). This function $\mathcal{H}_i(\mathbf{s})$ is assumed to be non-increasing and concave. For the ISP, the payoff is the profit function, which is defined as

$$F = \sum_{i=1}^{N} p_i s_i.$$
 (4.63)

The objective of user *i* is to obtain

$$s_i^* = \arg\max_{s_i} \left(\mathscr{G}_i(s_i) + \mathscr{H}_i(\mathbf{s}) - p_i s_i \right).$$
(4.64)

For all users, s^* can be obtained as the Nash equilibrium given p, a vector of prices charged to all users, i.e., $s^*(p)$. It can be shown that $s^*(p)$ is unique because of the



Fig. 4.9 System model of network game.

strict concavity of $\mathscr{G}_i(s_i)$ and of $\mathscr{H}_i(\mathbf{s})$ [451]. For the ISP as the leader, the price can be optimized as follows:

$$\mathbf{p}^* = \arg \max_{\mathbf{p}} \sum_{i=1}^{N} p_i \mathbf{s}^*(\mathbf{p}).$$
(4.65)

p^{*} is referred to as the Stackelberg equilibrium.

Partially incomplete information

In this case, the utility function $\mathscr{G}_i(\cdot)$ depends on the type t_i . Types of users are assumed to be independent. This type can be characterized by the user's application. Specifically, the utility function is defined as $\mathscr{G}_i(s_i, t_i)$. Given the price and known type (i.e., private information), the objective of the user becomes

$$s_i^* = \arg\max_{\mathbf{s}} \left(\mathscr{G}_i(s_i, t_i) + \mathscr{H}_i(\mathbf{s}) - p_i s_i \right). \tag{4.66}$$

In this case, the Nash equilibrium can be defined as $s^*(p_i, t_i)$. Then the ISP optimizes the profit as follows:

$$\mathbf{p}^{*} = \arg \max_{\mathbf{p}} \sum_{i=1}^{N} p_{i} E_{t_{i}} \left(\mathbf{s}^{*}(p_{i}, t_{i}) \right).$$
(4.67)

The difference between this and the complete-information case is that the Stackelberg equilibrium is the solution which maximizes the expected profit, since the ISP does not know the exact types of the users. However, the Nash equilibrium remains the same as that in the complete-information case, since the service satisfaction function depends only on the strategies but not the types of other users.

Totally incomplete information

In this case, both utility function $\mathscr{G}_i(\cdot)$ and service satisfaction function $\mathscr{H}_i(\cdot)$ are functions of type, whose exact value is not known. Therefore, the Nash equilibrium among the users will be affected and the objective of user *i* becomes

$$s_i^* = \arg\max_{s_i} \left(\mathscr{G}_i(s_i, t_i) + E_{\mathbf{t}_{-i}}(\mathscr{H}_i(\mathbf{s})) - p_i s_i \right).$$
(4.68)

The solution for all users (i.e., $s_i^*(\mathbf{p}, t_i)$) is the pure-strategy Bayesian Nash equilibrium. Then the Stackelberg equilibrium for ISP is

$$\mathbf{p}^* = \arg \max_{\mathbf{p}} \sum_{i=1}^{N} p_i E_{t_i} \left(\mathbf{s}^* (\mathbf{p}, t_i) \right).$$
(4.69)

4.3 Summary

Games with incomplete information (i.e., Bayesian games) can be used to analyze situations where a player does not know the preference (i.e., payoff) of his opponents. This is a common situation in wireless communications and networking where there is no centralized controller to maintain information on all users. Also, the users may not reveal private information to others. In this chapter, the Bayesian game framework has been studied in detail. Examples of this game in Cournot duopoly competition and firstprice sealed-bid auction have been presented. The framework has been applied to several problems in wireless communications and networking, including packet-forwarding, water-filling, power-control, channel-access, and bandwidth-allocation games. The differential games framework extends static non-cooperative continuous-kernel game theory into dynamic environments by adopting the tools, methods, and models of optimal-control theory. Optimal-control theory [58] has been developed to obtain the optimal solutions to planning problems that involve dynamic systems, where the state evolves over time under the influence of a control input (which is the instrument variable that is designed). Differential games can be viewed as extensions of optimal-control problems in two directions: (i) the evolution of the state is controlled not by one input but by multiple inputs, with each under the control of a different player, and (ii) the objective function is no longer a single one, with each player now having a possibly different objective function (payoff or cost), defined over time intervals of interest and relevance to the problem. The relative position of differential games in this landscape is captured in Table 5.1 [58]. Two main approaches that yield solutions to optimal-control problems are dynamic programming (introduced by Bellman) and the maximum principle (introduced by Pontryagin) [58]. The former leads to an optimal control that is a function of the state and time (closed-loop feedback control), whereas the latter leads to one that is a function only of the time and the initial state (open-loop control). These two approaches have also been adopted in differential games, where the common solution concepts of a differential game are again the Nash equilibrium and the Stackelberg equilibrium, for non-hierarchical and hierarchical structures, respectively. Using the techniques of optimal-control theory, not only can these solutions be obtained, but their stability can be analyzed.

The players in a differential game can interact with a dynamic system whose state affects the payoff, which is also affected directly by the controls of the players. For example, Fig. 5.1(a) shows a differential game formulation of bandwidth allocation among service providers (the players) to a group of users in a service area [264]. The underlying dynamic system has a proportion of users selecting different providers as the state, and this state affects the payoffs (i.e., revenue) for all providers. The providers, through their controls, can determine the amount of allocated bandwidth, which affects the evolution of the state.

The players themselves could also be dynamic systems, as depicted in Fig. 5.1(b), where the users (i.e., the players) transmit data to a base station. Buffer occupancy is the dynamic system, in which the state is the buffer length. This state evolves over time because of the arrival and transmission of packets. The user action is to control transmission rate or transmit power, and the payoff can be defined as the queueing delay of packets waiting in the buffer.

	Single player	Multiple players
Static	Mathematical programming	(static) Non-cooperative game
Dynamic	Optimal control theory	Dynamic and/or differential game theory

Table 5.1 Relationship between optimization, game theory, and differential games.



Fig. 5.1 Examples of a dynamic system in a differential game of (a) bandwidth allocation among service providers (b) buffer occupancy.

5.1 Optimal-control theory

Since optimal-control theory can be viewed as a special case of a differential game, the mathematical tools of optimal control will be useful for differential games in which the players can adopt an optimal-control Nash equilibrium or Stackelberg equilibrium concept. In an optimal-control game, there is one player, who is faced with an optimization problem with a single objective (e.g., maximization of payoff) over a period of time. In a differential game, on the other hand, even though there are multiple players, if the strategies of all but one player are fixed, essentially what we have is an optimal-control problem for the player whose action or strategy is to be determined. To solve this optimization problem, two approaches will be discussed here, both of them for the continuous-time case, dynamic programming, and the maximum principle. Both would also admit natural discrete-time counterparts.

5.1.1 Dynamic programming

Dynamic programming is based on the principle of optimality [58]. With this principle, an optimal course of action¹ has the property that whatever the initial state and time are, all remaining decisions must also constitute an optimal course of action [58]. In line with this principle, the solution can be obtained in retrograde time. In a discrete-time formulation, we start at all possible final states with the corresponding final times

¹ The terms "optimal control," "optimal strategy," and "optimal course of action" will be used interchangeably.

(e.g., stages). The optimal action at this final time is selected for each state, and we proceed back one step in time and determine the optimal action, again for each state. This process is repeated until the initial time or stage is reached. The core of dynamic programming, as applied to continuous-time optimal control, lies in a fundamental partial differential equation (PDE) called the Hamilton–Jacobi–Bellman (HJB) equation [58]. Before proceeding further, let us formulate a generic optimal-control problem in precise terms, as follows:

$$\frac{dx(t)}{dt} = \dot{x}(t) = F(x(t), a(t)), \quad t \ge 0,$$
(5.1)

$$a(t) = \gamma(x(t), t) \in \mathcal{A}, \tag{5.2}$$

$$J(a) = \int_0^1 \mathscr{U}(x(t), a(t)) dt + q(x(T)),$$
 (5.3)

where x(t) is the state variable at time t, with $x(0) = x_0$ as an initial condition (state); a(t) is an action at time t that is generated by a control law $\gamma(x(t), t)$; \mathcal{A} is the action space; $\mathcal{U}(x(t), a(t))$ is the instantaneous payoff function; J(a) is the objective function; T is the fixed terminal time; and q(x(T)) is the payoff at time T. From any state x and any initial time t, the maximum payoff-to-go (or cost-to-go in the minimization case) is determined by the value function, which is defined as follows:

$$V(x,t) = \max_{a(t'),t \leq t' \leq T} \left(\int_t^T \mathscr{U}(x(t'), a(t')) \mathrm{d}t' + q(x(T)) \right), \tag{5.4}$$

given x(t) = x. If $V_t(x)$ is jointly continuously differentiable in its two arguments, it satisfies the HJB equation, written as:

$$\frac{\partial V(x,t)}{\partial t} = \max_{a} \left(\frac{\partial V(x,t)}{\partial x} F(x,a) + \mathcal{U}(x,a) \right), \tag{5.5}$$

which has to be solved subject to the boundary condition $V_T(x) \equiv q_T(x)$. The HJB equation as a sufficient condition can be used to obtain the optimal control law γ^* (which is what maximizes the right-hand side of (5.5)), which in turn leads to the optimal action $a^*(t)$ through $a^*(t) = \gamma^*(x^*(t), t)$, where $x^*(\cdot)$ is the corresponding optimum trajectory [58].

5.1.2 The maximum principle

While the HJB equation is a sufficient condition for the optimal-control law (and hence optimal action), Pontryagin's maximum principle provides a necessary optimality condition for the optimal-control problem, and delivers the optimal solution in open-loop form. A restricted version of Pontryagin's maximum principle can be derived using the HJB equation with twice continuously differentiable V_t as the starting point. Let us first
introduce the function

$$\mathscr{G}(x,a) = \frac{\partial V(x,t)}{\partial x} F(x,a) + \mathscr{U}(x,a), \qquad (5.6)$$

and note that the HJB equation can be written as

$$-\frac{\partial V(x,t)}{\partial t} = \max_{a} \mathscr{G}(x,a).$$
(5.7)

For the optimal-control law γ^* , the HJB equation can be rearranged as follows:

$$\mathscr{G}(x,a^*) + \frac{\partial V(x,t)}{\partial t} = 0, \quad a^*(t) = \gamma^*(x,t).$$
(5.8)

Because V_t is twice continuously differentiable, by applying second partial derivatives with respect to x we obtain

$$\frac{\partial \mathscr{U}}{\partial x} + \frac{\mathsf{d}}{\mathsf{d}t} \left(\frac{\partial V(x,t)}{\partial x} \right) + \frac{\partial V(x,t)}{\partial x} \frac{\partial F}{\partial x} + \frac{\partial \mathscr{G}}{\partial a} \frac{\partial \gamma^*(x,t)}{\partial x} = 0.$$
(5.9)

If there is no constraint on the control *a*, then $\frac{\partial \mathscr{G}}{\partial a} = 0$ for $a = a^*$. On the other hand, if there are constraints on *a* and if a^* is not an interior point, then it can be shown that $\frac{\partial \mathscr{G}}{\partial a} \frac{\partial \gamma^*(x,t)}{\partial x} = 0$ as a result of the optimality condition, and $\frac{\partial \mathscr{G}}{\partial a}$ and $\frac{\partial \gamma^*(x,t)}{\partial x}$ are orthogonal. Therefore (5.9) becomes

$$\frac{\partial \mathscr{U}}{\partial x} + \frac{\mathsf{d}}{\mathsf{d}t} \left(\frac{\partial V(x,t)}{\partial x} \right) + \frac{\partial V(x,t)}{\partial x} \frac{\partial F}{\partial x} = 0.$$
(5.10)

Then an auxiliary variable, called the co-state $\lambda(t)$, is introduced to simplify the derivation for the given state trajectory x^* and the optimal action $a^*(t) = \gamma^*(x^*(t), t)$. Equation (5.10) can be expressed in the following form:

$$\frac{\mathrm{d}\lambda}{\mathrm{d}t} = -\frac{\partial}{\partial x} \left(\mathscr{U}(x^*, a^*) + \lambda(t)F(x^*, a^*) \right) = -\frac{\partial}{\partial x} \mathscr{H}(\lambda, x^*, a^*), \tag{5.11}$$

where

$$\mathscr{H}(\lambda, x, a) = \mathscr{U}(x, a) + \lambda F(x, a).$$
(5.12)

The boundary condition for the terminal time of the co-state is

$$\lambda(T) = \frac{\partial V(x^*, T)}{\partial x} = \frac{\partial q(x^*)}{\partial x}.$$
(5.13)

This leads to the maximum principle of Pontryagin, which can in fact be derived more directly, without the need for differentiability of the value function, or the HJB equation. Furthermore, here the necessary condition of the maximum principle yields the control as an open-loop one, i.e., as a function of t and x_0 , and not of the state x.

The maximum principle may be stated as follows. For the optimal-control problem, if $a^*(\cdot)$ is an optimal open-loop control, and $x^*(\cdot)$ is the corresponding state trajectory, there exists a co-state function $\lambda(\cdot)$ such that

$$\dot{x}^{*}(t) = F(x^{*}, a^{*}), \quad x(t_{0}) = x_{0},$$
(5.14)

$$\dot{\lambda}(t) = -\frac{\partial \mathscr{H}(\lambda, x^*, a^*)}{\partial x}, \qquad (5.15)$$

$$\lambda(T) = \frac{\partial q_T(x^*)}{\partial x},\tag{5.16}$$

$$\mathscr{H}(\lambda, x, a) = \mathscr{U}(x, a) + \lambda F(x, a), \qquad (5.17)$$

$$a^{*}(t) = \arg \max_{a \in \mathcal{A}} \mathcal{H}(\lambda, x^{*}, a).$$
(5.18)

Here, \mathcal{H} is known as the Hamiltonian associated with the optimal-control problem.

5.2 Differential games

5.2.1 Main ingredients and general results

The main ingredients of differential games are the state variable, the control variable, the action (control) set of each player, the objective functions of the players, the information structure, and the relevant solution concept. As in the case of an optimal-control problem, in a differential game the state variable evolves over time, driven by the players' actions. The actions are generated by the strategies of the players, which are defined for each player as mappings from the information available to that player to his action set. A differential game is played over time $t \in [0, T]$, where the time horizon of the game can be finite (i.e., $T < \infty$) or infinite (i.e., $T = \infty$). Let \mathcal{N} denote the set of players, defined as $\mathcal{N} = \{1, ..., N\}$. The state vector for the game is described by $x(\cdot)$, evolving as in (5.1) according to

$$\dot{x}(t) = F(x(t), \mathbf{a}(t)), \qquad (5.19)$$

where $\mathbf{a}(t) = [a_1(t), ..., a_N(t)]^T$ is the collection of actions at time t (which can also be viewed as a vector), with a_i standing for player *i*'s action, and $i \in \{1, 2, ..., N\}$.

The objective function of a player (i.e., the payoff) is the benefit to be maximized (or minimized in case of a negative payoff, equivalently cost). The payoff function in a differential game can be defined in general as the discounted value of the flow of instantaneous payoff over time. Let $\mathscr{U}_i(\cdot)$ denote the instantaneous payoff function at time *t* (e.g., the utility) for player *i*. This instantaneous payoff for player *i* is a function of the actions and state variables of all players. The cumulative payoff is defined as the integral of instantaneous payoff over time, properly discounted, that is,

$$J_{i} = \int_{0}^{T} \mathscr{U}_{i}(x(t), a_{i}(t), \mathbf{a}_{-i}(t)) e^{-\rho t} \mathrm{d}t, \qquad (5.20)$$

where $\mathbf{a}_{-i}(t)$ is the vector of actions of all players except player *i*, and $\rho > 0$ is the discount factor. Note that to keep the presentation simple, we have not included a cost on the terminal state here, such as $q_i(x(T))$. For each player, the objective is to optimize this cumulative payoff by choosing an action $a_i(\cdot)$, i.e., $\max_{a_i(\cdot)} J_i$, more generally by choosing a strategy γ_i . For this, we need to introduce possible information structures for the players in the game.

Even though a much higher number of information structures is possible, we will consider here the three most commonly used ones:

- *Open-loop (OL) information.* The players have common knowledge of the value of the state vector at initial time *t* = 0, and acquire no further information.
- *Feedback (FB) information.* At time *t*, each player has access to the value of the state vector at time *t*, that is *x*(*t*), and no further information.
- Closed-loop (CL) information. At time t, players have access to the values of the state variables from time 0 to t, namely $\{x(s), 0 \le s \le t\}$, that is, to *perfect information* on the past and present as far as the state goes.

It is possible to have a mixture of these three information structures for the players, some having OL, some FB, and others CL information, but we will not consider such structures here. We will consider only the open-loop case (for all players) and the feedback case (for all players), in the context of Nash equilibrium.

For the OL structure, the derivation of Nash equilibrium involves the solution of N optimal-control problems, where, in the generic *i*th one, the actions of all players except the *i*th are held fixed as OL policies (that is as functions of time, and not of state), and maximization of the payoff, $J_i(a_i, a_{-i})$ is carried out with respect to $a_i(\cdot)$, the action variable of player *i* [58]:

$$\max_{a_{i}(\cdot)} J_{i}(a_{i}, \mathbf{a}_{-i}) = \int_{0}^{T} \mathscr{U}_{i}(x(t), a_{i}(t), \mathbf{a}_{-i}(t)) e^{-\rho t} dt, \qquad (5.21)$$

s.t.
$$\frac{dx(t)}{dt} = \dot{x}(t) = F(x(t), \mathbf{a}(t)), \quad x(0) = x_0.$$
 (5.22)

This problem can be solved for each a_{-i} using the maximum principle of Pontryagin, discussed earlier. The Hamiltonian function is defined as

$$\mathscr{H}_{i}(x(t), a_{i}(t), \mathbf{a}_{-i}(t); \lambda_{i}(t)) = e^{-\rho t} \mathscr{U}_{i}(x(t), a_{i}(t), \mathbf{a}_{-i}(t)) + \mu(t) F(x(t), \mathbf{a}(t)),$$
(5.23)

where $\lambda_i(t) = \mu_i(t)e^{-\rho t}$ is the co-state.

A set of necessary conditions for the open-loop solution of (5.21)–(5.22) now arise from the maximum principle:

$$\frac{\partial \mathscr{H}_i(\mathbf{x}(t), \mathbf{a}_i(t), \mathbf{a}_{-i}(t); \lambda_i(t))}{\partial \mathbf{a}_i(t)} = 0,$$
(5.24)

$$-\frac{\partial \mathscr{H}_{i}(x(t), a_{i}(t), \mathbf{a}_{-i}(t); \lambda_{i}(t))}{\partial x(t)} = \frac{\mathrm{d}\lambda_{i}(t)}{\mathrm{d}t},$$
(5.25)

given the two-point boundary conditions $x_i(0) = x_0$, $\lambda_i(T) = 0$. For the infinite-horizon problem, we require in addition that the system be stable under optimal control, that is, $\lim_{t\to\infty} x_i(t) = 0$. Introducing $\tilde{\mathscr{H}}_i := e^{\rho t} \mathscr{H}_i$, we obtain a relationship equivalent to (5.25) but without the exponential term, and in terms of μ_i , and subject again to the boundary condition $\mu_i(T) = 0$, for all *i*:

$$-\frac{\partial \tilde{\mathscr{H}}_{i}(x(t), a_{i}(t), \mathbf{a}_{-i}(t))}{\partial x(t)} + \rho \mu_{i}(t) = \frac{d \mu_{i}(t)}{dt}.$$
(5.26)

For the feedback Nash equilibrium, say $\{\gamma_1^*(x(t), t), \dots, \gamma_N^*(x(t), t)\}\$, the underlying optimization problem for each player, say player *i*, as the counterpart of (5.21)–(5.22), would be

$$\max_{\gamma_i} J_i(\gamma_i, \gamma_{-i}^*) = \int_o^T \mathcal{U}_i(x(t), \gamma_i(x(t), t), \gamma_{-i}^*(x(t), t)) e^{-\rho t} \mathrm{d}t, \qquad (5.27)$$

subject to dynamics (5.22) with **a** replaced by γ . The tool to be used in this case is dynamic programming, and particularly the HJB equation. If $V_i(x, t)$ denotes the value function (cost-to-go function) associated with player *i*, assuming that it is jointly continuously differentiable in *x* and *t*, we have as a sufficient condition for a feedback Nash equilibrium solution the following set of coupled PDEs:

$$-\frac{\partial V_i(x,t)}{\partial t} = \max_{a_i} \left[\frac{\partial V_i(x,t)}{\partial x} F(x,a_i,\gamma_{-i}^*(x,t)) + e^{-\rho t} \mathscr{U}_i(x(t),a_i,\gamma_{-i}^*(x(t),t)) \right],$$
(5.28)

with boundary conditions $V_i(x, T) \equiv 0$, for i = 1, ..., N. The a_i that maximize the righthand side of the HJB PDEs above are clearly functions of both x and t (in general), and they constitute the feedback Nash equilibrium solution $\{\gamma_1^*(x(t), t), ..., \gamma_N^*(x(t), t)\}$ of the differential game.

If we introduce, as we did in the case of the open-loop solution, the transformation $\tilde{V}_i(x, t) = e^{\rho t} V_i(x, t), i = 1, ..., N$, then, given that

$$\frac{\partial \tilde{V}_i(x,t)}{\partial t} = \rho \tilde{V}_i(x,t) + e^{\rho t} \frac{\partial V_i(x,t)}{\partial t}, \qquad (5.29)$$

we arrive at the equivalent set of PDEs without the exponential term:

$$-\frac{\partial \tilde{V}_{i}(x,t)}{\partial t} + \rho \tilde{V}_{i}(x,t) = \max_{a_{i}} \left[\frac{\partial \tilde{V}_{i}(x,t)}{\partial x} F(x,a_{i},\gamma_{-i}^{*}(x,t)) + \mathscr{U}_{i}(x(t),a_{i},\gamma_{-i}^{*}(x(t),t)) \right],$$
(5.30)

where the boundary conditions are again $\tilde{V}_i(x, T) \equiv 0, i = 1, ..., N$.

5.2.2 Stackelberg differential game

While under the Nash equilibrium in a differential game all players decide on their actions or strategies simultaneously, under the Stackelberg solution there is a hierarchy in decision-making, either in the announcement and execution of actions or in the

announcement of strategies. We will call such games Stackelberg differential games. They arise in many applications, with one area being management science, particularly supply-chain management [204]. Here we will consider the two-player case, where one player is the leader and the other one a follower, both making decisions over a time horizon T. In an open-loop Stackelberg differential game, the leader chooses the action path $a_{\rm le}(t)$ first. Then the follower determines its optimal action path $a_{\rm fo}(t)$ in response to the action path of the leader. A Stackelberg solution is obtained when the leader reaches the maximum payoff, anticipating (and taking into account) the optimal action path of the follower. If the information structure is closed-loop instead of open-loop, then the leader will have to announce a strategy using the dynamic information it has (instead of a course of action), and obtaining the Stackelberg solution in that case is very complicated since its derivation involves the solution of functional optimization problems; for details see [58]. In its place, a solution concept that is introduced is that of a "feedback Stackelberg solution," which provides the leader with an advantage in decision-making only stagewise (in a discrete time framework) or at each point in time, and not during the entire course of the game. Such a definition allows for the solution to be computed backwards in time, by solving static Stackelberg games at each point in time, assuming that both the leader and the follower have access to the current value of the state. Below we discuss the derivation of the Stackelberg solution, first for the open-loop case and then under the feedback Stackelberg concept with closed-loop feedback information available to both players.

Open-loop solution of a Stackelberg differential game

An open-loop solution can be obtained as follows. First, the best response action path of the follower is anticipated by the leader. Specifically, the optimal-control problem of the follower is formulated and solved given the action path of the leader. Then, the optimal response action path of the follower (i.e., best response) is used in the optimalcontrol problem formulated for the leader, the solution of which leads to the Stackelberg solution.

The given action path of the leader is denoted by $a_{le}(\cdot)$. The optimal-control problem of the follower is defined as follows:

$$\max_{a_{fo}} J_{fo}(x_0, a_{le}(t), a_{fo}(t)) = \int_0^T e^{-\rho_{fo}t} \mathscr{U}_{fo}(x(t), a_{le}(t), a_{fo}(t)) dt + e^{-\rho_{fo}T} q_{fo}(x(T)),$$
(5.31)

given the constraint on the state variable,

$$\frac{\mathrm{d}\lambda}{\mathrm{d}t} = \dot{x}(t) = F(x(t), a_{\mathrm{le}}(t), a_{\mathrm{fo}}(t)), \qquad (5.32)$$

for $x(0) = x_0$, where $\rho_{fo} > 0$ is the discount rate of the follower, $\mathcal{U}_{fo}(x(t), a_{le}(t), a_{fo}(t))$ is the instantaneous payoff function, $q_{fo}(x(T))$ is the terminating condition, and x_0 is the initial state. The Hamiltonian of the follower is

$$\mathscr{H}_{fo}(x,\lambda,a_{le},a_{fo}) = \mathscr{U}_{fo}(x,a_{le},a_{fo}) + \lambda F(x,a_{le},a_{fo}), \qquad (5.33)$$

where λ is the co-state. The adjoint equation is

$$\dot{\lambda} = \rho \lambda - \frac{\partial}{\partial x} \mathscr{H}_{fo}(x, \lambda, a_{le}, a_{fo}), \qquad (5.34)$$

$$\lambda(T) = \frac{\partial}{\partial x} q_{\rm fo}(x(T)). \tag{5.35}$$

The optimal action path of the follower is

$$a_{f_{o}}^{*}(t) = \arg \max_{a_{f_{o}} \in \mathcal{A}_{f_{o}}} \mathscr{H}_{f_{o}}(x(t), \lambda, a_{le}(t), a_{f_{o}}),$$
(5.36)

where \mathcal{A}_{fo} is the action set of the follower. Assuming that this solution is unique for each action path of the leader, let us write it as follows to show the dependence on the action path of the leader: $a_{f_0}^*(t) = \mathcal{B}_{f_0}(x(t), \lambda(t), a_{le}(t))$. Note that, assuming differentiability and that the action set of the follower is open, $\mathcal{B}_{f_0}(x(t), \lambda(t), a_{le}(t))$ must satisfy the first-order condition

$$\frac{\partial}{\partial a_{fo}} \mathscr{H}_{fo}(x, \lambda, a_{le}, \mathscr{B}_{fo}) = 0.$$
(5.37)

Then, the optimal-control problem faced by the leader is defined as follows:

$$\max_{a_{le}} J_{le}(x_{0}, a_{le}), \qquad (5.38)$$

$$J_{le}(x_{0}, a_{le}) = \int_{0}^{T} e^{-\rho_{le}t} \mathscr{U}_{le}(x, a_{le}, \mathscr{B}_{fo}(x, \lambda, a_{le})) dt + e^{-\rho_{le}T} q_{le}(x(T)), \qquad \dot{x} = F(x, a_{le}, \mathscr{B}_{fo}(x, \lambda, a_{le})), \qquad \dot{\lambda} = \rho_{le}\lambda - \frac{\partial}{\partial x} \mathscr{H}_{fo}(x, \lambda, a_{le}, \mathscr{B}_{fo}(x, \lambda, a_{le})), \qquad \lambda(T) = \frac{\partial}{\partial x} q_{fo}(x(T)),$$

for $x(0) = x_0$, where ρ_{le} is the discount rate of the leader. The Hamiltonian of the leader is then

$$\mathcal{H}_{\mathsf{le}}(x,\lambda,\phi,\theta,\mathsf{a}_{\mathsf{le}}) = \mathcal{U}_{\mathsf{le}}(x,\mathsf{a}_{\mathsf{le}},\mathcal{B}_{\mathsf{fo}}(x,\lambda,\mathsf{a}_{\mathsf{le}})) + \phi F(x,\mathsf{a}_{\mathsf{le}},\mathcal{B}_{\mathsf{fo}}(x,\lambda,\mathsf{a}_{\mathsf{le}})) + \theta \left(\rho_{\mathsf{fo}}\lambda - \left(\frac{\partial}{\partial x}\mathcal{H}_{\mathsf{fo}}(x,\lambda,\mathsf{a}_{\mathsf{le}},\mathcal{B}_{\mathsf{fo}}(x,\lambda,\mathsf{a}_{\mathsf{le}}))\right)\right), \quad (5.39)$$

where ϕ and θ are the co-states of the state variable x and of the co-state λ of the follower, respectively. The adjoint equations are

$$\dot{\phi} = \rho_{\mathsf{le}}\phi - \frac{\partial}{\partial x}\mathscr{H}_{\mathsf{le}}(x,\lambda,\phi,\theta,\mathsf{a}_{\mathsf{le}},\mathscr{B}_{\mathsf{fo}}(x,\lambda,\mathsf{a}_{\mathsf{le}}))$$
(5.40)

$$= \rho_{\mathsf{le}}\phi - \frac{\partial}{\partial x}\mathscr{U}_{\mathsf{le}}(x, a_{\mathsf{le}}, \mathscr{B}_{\mathsf{fo}}(x, \lambda, a_{\mathsf{le}})) - \phi \frac{\partial}{\partial x}F(x, a_{\mathsf{le}}, \mathscr{B}_{\mathsf{fo}}(x, \lambda, a_{\mathsf{le}})) + \theta \frac{\partial^2}{\partial x^2}\mathscr{H}_{\mathsf{fo}}(x, \lambda, a_{\mathsf{le}}, \mathscr{B}_{\mathsf{fo}}(x, \lambda, a_{\mathsf{le}})),$$
(5.41)

$$\phi(T) = \frac{\partial}{\partial x} q_{\mathsf{le}}(x(T)) - \theta(T) \frac{\partial^2}{\partial x^2} q_{\mathsf{fo}}(x(T)), \tag{5.42}$$

$$\dot{\theta} = \rho_{\mathsf{le}}\theta - \frac{\partial}{\partial\lambda}\mathscr{H}_{\mathsf{le}}(x,\lambda,\phi,\theta,\mathsf{a}_{\mathsf{le}},\mathscr{B}_{\mathsf{fo}}(x,\lambda,\mathsf{a}_{\mathsf{le}}))$$
(5.43)

$$= \rho_{\mathsf{le}}\theta - \frac{\partial}{\partial\lambda}\mathscr{U}_{\mathsf{le}}(x, a_{\mathsf{le}}, \mathscr{B}_{\mathsf{fo}}(x, \lambda, a_{\mathsf{le}})) - \theta \frac{\partial}{\partial\lambda}F(x, a_{\mathsf{le}}, \mathscr{B}_{\mathsf{fo}}(x, \lambda, a_{\mathsf{le}})) + \theta \frac{\partial}{\partial\lambda}\frac{\partial}{\partial x}\mathscr{H}_{\mathsf{fo}}(x, \lambda, a_{\mathsf{le}}, \mathscr{B}_{\mathsf{fo}}(x, \lambda, a_{\mathsf{le}})),$$
(5.44)

$$\theta(0) = 0. \tag{5.45}$$

The second-order terms in these adjoint equations are due to the hierarchical game structure, in which the optimal-control problem of the leader has as a constraint the solution of the optimal-control problem of the follower. The leader's optimal action path is defined as

$$a_{\mathsf{le}}^{*}(t) = \arg\max_{a_{\mathsf{le}} \in \mathcal{A}_{\mathsf{le}}} \mathcal{H}_{\mathsf{e}}(x(t), \lambda(t), \phi(t), \theta(t), a_{\mathsf{le}}),$$
(5.46)

where A_{le} is the action set of the leader, which we assume to be open. Then, a necessary optimality condition for the leader's optimal action path is computed from

$$\frac{\partial}{\partial a_{\mathsf{le}}} \mathscr{H}_{\mathsf{le}}(x(t), \lambda(t), \phi(t), \theta(t), a_{\mathsf{le}}^*) = 0, \qquad (5.47)$$

and the optimal action path of the follower a_{fo}^* at the Stackelberg equilibrium can be obtained by substituting a_{le}^* into $\mathscr{B}_{fo}(x(t), \lambda(t), a_{le}^*)$.

However, the best response for the follower, i.e., $\mathscr{B}_{fo}(\cdot)$, cannot always be expressed in closed form. In this case, the equality constraint from the follower's optimal-control problem is introduced into the optimal-control problem formulation of the leader. Let a_{le}^* and a_{fo}^* denote the open-loop Stackelberg equilibria for the leader and follower, respectively, and let x^* be the corresponding state variable trajectory. Assume that $F(\cdot)$, $\mathscr{U}_{le}(\cdot)$, $q_{le}(\cdot)$, $\mathscr{U}_{fo}(\cdot)$, and $q_{fo}(\cdot)$ are twice continuously differentiable on x; $\mathscr{H}_{fo}(\cdot)$ is continuously differentiable and strictly convex on a_{fo} ; and $F(\cdot)$, $\mathscr{U}_{le}(\cdot)$, and $\mathscr{U}_{fo}(\cdot)$ are continuously differentiable on a_{le} . Then there exist continuously differentiable functions $\lambda(\cdot)$, $\phi(\cdot)$, $\theta(\cdot)$, and $\mu(\cdot)$ which satisfy the following relationships:

$$\dot{x}^* = F(x, a_{\mathsf{le}}^*, a_{\mathsf{fo}}^*), \quad x^*(0) = x_0,$$
(5.48)

$$\dot{\lambda} = \rho_{\rm fo} \lambda - \frac{\partial}{\partial x} \mathscr{H}_{\rm fo}(x^*, \lambda, a_{\rm le}^*, a_{\rm fo}^*), \qquad (5.49)$$

$$\lambda(T) = \frac{\partial}{\partial x} q_{\rm fo}(x^*(T)), \qquad (5.50)$$

$$\dot{\phi} = \rho_{\mathsf{le}}\phi - \frac{\partial}{\partial x}\mathscr{H}_{\mathsf{le}}(x^*, \lambda, \phi, \theta, \mu, \mathbf{a}^*_{\mathsf{le}}, \mathbf{a}^*_{\mathsf{fo}}),$$
(5.51)

$$\phi(T) = \frac{\partial}{\partial x} q_{\mathsf{le}}(x^*(T)) - \theta(T) \frac{\partial^2}{\partial x^2} q_{\mathsf{fo}}(x^*(T)), \qquad (5.52)$$

$$\dot{\theta} = \rho_{\mathsf{le}}\theta - \frac{\partial}{\partial\lambda}\mathscr{H}_{\mathsf{le}}(\mathbf{x}^*, \lambda, \phi, \theta, \mu, \mathbf{a}_{\mathsf{le}}^*, \mathbf{a}_{\mathsf{fo}}^*),$$
(5.53)

$$\theta(\mathbf{0}) = \mathbf{0},\tag{5.54}$$

$$0 = \frac{\partial}{\partial a_{\mathsf{le}}} \mathscr{H}_{\mathsf{le}}(x^*, \lambda, \phi, \theta, \mu, a_{\mathsf{le}}^*, a_{\mathsf{fo}}^*), \qquad (5.55)$$

$$0 = \frac{\partial}{\partial a_{fo}} \mathscr{H}_{le}(x^*, \lambda, \phi, \theta, \mu, a_{le}^*, a_{fo}^*), \qquad (5.56)$$

$$0 = \frac{\partial}{\partial a_{fo}} \mathscr{H}_{fo}(x^*, \lambda, a_{le}^*, a_{fo}^*), \qquad (5.57)$$

where \mathscr{H}_{fo} is defined as in (5.33), and \mathscr{H}_{le} is defined as follows:

$$\mathcal{H}_{\mathsf{le}}(x,\lambda,\phi,\theta,\mu,\mathsf{a}_{\mathsf{le}},\mathsf{a}_{\mathsf{fo}}) = \mathcal{U}_{\mathsf{le}}(x,\mathsf{a}_{\mathsf{le}},\mathsf{a}_{\mathsf{fo}}(x,\lambda,\mathsf{a}_{\mathsf{le}})) + \phi F(x,\mathsf{a}_{\mathsf{le}},\mathsf{a}_{\mathsf{fo}}(x,\lambda,\mathsf{a}_{\mathsf{le}})) + \theta \left(\rho\lambda - \frac{\partial}{\partial x}\mathcal{H}_{\mathsf{fo}}(x,\lambda,\mathsf{a}_{\mathsf{le}},\mathsf{a}_{\mathsf{fo}})\right) + \mu \frac{\partial}{\partial \mathsf{a}_{\mathsf{fo}}}\mathcal{H}_{\mathsf{fo}}(x,\lambda,\mathsf{a}_{\mathsf{le}},\mathsf{a}_{\mathsf{fo}}).$$
(5.58)

Note that the open-loop Stackelberg equilibrium is static, in the sense that the players have to commit to action paths at the initial time, regardless of how the state evolves. In this sense, the open-loop Stackelberg equilibrium is not time-consistent.

Feedback Stackelberg solution

Unlike the open-loop Stackelberg equilibrium, the feedback Stackelberg equilibrium differential game at any time t is a function of the value of the state variable at that time. Therefore, the feedback Stackelberg equilibrium can be considered to be the perfect state-space equilibrium, since the necessary optimality condition must be satisfied for all values of the state variable and at each point in time. In other words, the feedback Stackelberg equilibrium is subgame-perfect, since the solution does not depend on the initial condition, and the solution remains optimal at any time instance after the game starts.

The feedback Stackelberg equilibrium can be physically interpreted as the sequence of solutions applied to a discrete-time dynamic game. In particular, the original differential game is divided into multiple sampled-stage games. Each solution is applied to each discrete time in which the player can observe the value of the state variable. Therefore, the feedback Stackelberg equilibrium of each sampled-state game can be obtained by solving a sequence of approximately defined open-loop Stackelberg games. Let $V_{le}(x)$ and $V_{fo}(x)$ denote the feedback Stackelberg value-to-go of the leader and the follower, respectively. This value-to-go is defined in current-value form at any time given the value

of the state variable x. For a given policy $a_{le}(t, x)$, the HJB equation of the follower is [532, 204]

$$\rho_{fo} V_{fo} - \frac{\partial V_{fo}(x)}{\partial t} = \max_{a_{fo} \in \mathcal{A}_{fo}} \left(\frac{\partial V_{fo}(x)}{\partial x} F(x, a_{le}(t, x), a_{fo}) + \mathcal{U}_{fo}(x, a_{le}(t, x), a_{fo}) \right),$$
(5.59)

for $V_{fo}(x(T)) = q_{fo}(x(T))$. The best response of the follower at each point in time is expressed as

$$\mathscr{B}_{fo}(x, a_{le}, \frac{\partial V_{fo}}{\partial x}) = \arg \max_{a_{fo} \in \mathcal{A}_{fo}} \left(\frac{\partial V_{fo}(x)}{\partial x} F(x, a_{le}, a_{fo}) + \mathscr{U}_{fo}(x, a_{le}, a_{fo}) \right).$$
(5.60)

Then the HJB equation of the leader can be written as

$$\rho_{\rm le} V_{\rm le} - \frac{\partial V_{\rm le}(x)}{\partial t} = \max_{a_{\rm le} \in \mathcal{A}_{\rm le}} \left(\frac{\partial V_{\rm le}(x)}{\partial x} F\left(x, a_{\rm le}(t, x), \mathscr{B}_{\rm fo}\left(x, a_{\rm le}, \frac{\partial V_{\rm fo}}{\partial x}\right)\right) + \mathscr{U}_{\rm le}\left(x, a_{\rm le}, \mathscr{B}_{\rm fo}\left(x, a_{\rm le}, \frac{\partial V_{\rm fo}}{\partial x}\right)\right)\right),$$
(5.61)

for $V_{le}(x(T)) = q_{le}(x(T))$. The maximization of the HJB equation of the leader then yields the optimal feedback action,

$$a_{le}^{\dagger}(t,x) = \arg \max_{a_{le} \in \mathcal{A}_{le}} \left(\frac{\partial V_{le}(x)}{\partial x} F\left(x, a_{le}(t,x), \mathscr{B}_{fo}\left(x, a_{le}, \frac{\partial V_{fo}}{\partial x}\right) \right) + \mathscr{U}_{le}\left(x, a_{le}, \mathscr{B}_{fo}\left(x, a_{le}, \frac{\partial V_{fo}}{\partial x}\right) \right) \right).$$
(5.62)

The optimal action path of the follower is

$$a_{\rm fo}^{\dagger}(t,x) = \mathscr{B}_{\rm fo}\left(x, a_{\rm le}^{\dagger}, \frac{\partial V_{\rm fo}}{\partial x}\right). \tag{5.63}$$

With some manipulation, the HJB equation for the entire game can be expressed as

$$\rho_{\mathsf{le}} V_{\mathsf{le}} - \frac{\partial V_{\mathsf{le}}(x)}{\partial t} = \frac{\partial V_{\mathsf{le}}(x)}{\partial x} F(x, a_{\mathsf{le}}^{\dagger}(t, x), a_{\mathsf{fo}}^{\dagger}(t, x)) + \mathscr{U}_{\mathsf{le}}(x, a_{\mathsf{le}}^{\dagger}(t, x), a_{\mathsf{fo}}^{\dagger}(t, x)),$$
(5.64)

$$\rho_{\rm fo}V_{\rm fo} - \frac{\partial V_{\rm fo}(x)}{\partial t} = \frac{\partial V_{\rm fo}(x)}{\partial x}F(x, a_{\rm le}^{\dagger}(t, x), a_{\rm fo}^{\dagger}(t, x)) + \mathscr{U}_{\rm fo}(x, a_{\rm le}^{\dagger}(t, x), a_{\rm fo}^{\dagger}(t, x)),$$
(5.65)

for $V_{le}(x(T)) = q_{le}(x(T))$ and $V_{fo}(x(T)) = q_{fo}(x(T))$. These equations for $V_{le}(x)$ and $V_{fo}(x)$ can be solved to obtain the feedback Stackelberg equilibrium of $a_{le}^{\dagger}(t,x)$ and $a_{fo}^{\dagger}(t,x)$ for the leader and follower, respectively.

5.3 Applications of differential games in wireless communications and networking

For routing in a mobile ad hoc network (MANET), the forwarding nodes, as players, have an incentive from the destination, in terms of price, to allocate transmission rate to forward packets from the source. A differential game for duopoly competition is applied to model this competitive situation.

In [299], the traffic routing in an ad hoc network is formulated as a differential game. The network under consideration is shown in Fig. 5.2, in which the source transmits data to the destination. There are two forwarding nodes, considered to be the players in this game. As the destination pays these forwarding nodes according to the amount of forwarded data, the two nodes compete with each other by adjusting the forwarding rate (i.e., action denoted by $a_i(t)$ for player *i* at time *t*) to maximize their utility over the time duration $[0, \infty]$. Let the payment from the destination at time *t* be denoted by P(t). The payoff function for player *i* can be expressed as

$$J_{i} = \int_{0}^{+\infty} e^{-\rho t} \left(P(t)a_{i}(t) - ca_{i}(t) - \frac{1}{2}a_{i}(t)^{2} - g(\mathbf{a}) \right) dt, \qquad (5.66)$$

where a quadratic cost function (i.e., $ca_i(t) + \frac{1}{2}a_i(t)^2 + g(\mathbf{a})$) is considered, *c* is a cost parameter, and $g(\mathbf{a})$ is a cost function given vector **a** of the actions of the players. Note that the cost is quadratic, for example because of the precipitous discharge of battery life. $\rho > 0$ is the discount rate. For the payment, the following evolution (i.e., a differential equation of Tsutsui and Mino [480]) is considered:

$$\frac{\mathrm{d}P(t)}{\mathrm{d}t} = \dot{P}(t) = K(E - a_1(t) - a_2(t) - P(t)), \qquad (5.67)$$

where K and E are constants. If $g(\mathbf{a}) = 0$, this game is reduced to an infinite-horizon duopolistic competition. The feedback Nash equilibrium strategies of this game can be



expressed as [299]

$$a_i^* = P^*(t) - c - K(AP^*(t) - B), \qquad (5.68)$$

for i = 1, 2, where

$$P^{*}(t) = \left(P_{0} - \frac{E + 2(c - KB)}{3 - 2KA}\right)e^{-K(3 - 2KA)t} + \frac{E + 2(c - KB)}{3 - 2KA},$$
 (5.69)

$$A = \frac{\rho + 6K - \sqrt{(\rho + 8K)^2 - 12K^2}}{6K^2},$$
(5.70)

$$B = \frac{-EKA + c - 2KcA}{\rho - 3K^2A + 3K},\tag{5.71}$$

where P_0 is the initial payment at time t = 0. However, for general $g(\mathbf{a})$ the payoff optimization becomes a coupling constraint. In [299], an iterative approach based on greedy adjustment is proposed to obtain the solution. In particular, the algorithm gradually increases the forwarding rate of the player as long as the payoff is non-decreasing. If the payoff for one player decreases, the algorithm will allow the other players to adjust the forwarding rate until none of players can gain a higher payoff.

5.4 Summary

In this chapter, the topic of differential games has been introduced. In a differential game, the players adapt their actions to gain the highest payoffs over the time period. Therefore, optimal-control theory is applied to obtain the equilibrium solution. The basics of optimal control and its solution methods based on dynamic programming and Pontryagin's maximum principle have been presented. The general form of a non-cooperative differential game has been introduced, with examples from oligopoly competition to obtain open-loop, closed-loop, and feedback equilibria. The topic of hierarchical (Stackelberg) differential game has also been covered, and solution methods for both open-loop and feedback Stackelberg equilibria have been discussed. To this end, to applications of a differential game wireless communications and networking have been presented.

Evolutionary-game theory has been developed as a mathematical framework to study the interaction among rational biological agents in a population [152]. In evolutionarygame theory, the agent adapts (i.e., evolves) the chosen strategy based on its fitness (i.e., payoff). In this way, both static and dynamic behavior (e.g., equilibrium) of the game can be analyzed.

Evolutionary-game theory has the following advantages over the traditional noncooperative game theory we have studied in the previous chapters:

- As we have seen, the Nash equilibrium is the most common solution concept for non-cooperative games. An *N*-tuple of strategies in an *N*-player game is said to be in Nash equilibrium if an agent (player) cannot improve his payoff by moving to another strategy, given that the other players stay with their strategies at Nash equilibrium. Specifically, the strategy of a player at Nash equilibrium is the best response to the strategies of the other players, again at Nash equilibrium. However, the Nash equilibrium is not necessarily efficient, as it would be possible for all players to benefit from a collective behavior. Also, there could be multiple Nash equilibria in a game, and if the agent is restricted to adopting only pure strategies, the Nash equilibrium may not exist. In this case, the solution of the evolutionary game (i.e., *evolutionarily stable strategies* (ESS) or *evolutionary equilibrium*) can serve as a refinement to the Nash equilibrium, especially when multiple Nash equilibria exist.
- In a traditional non-cooperative game, the agents are assumed to be rational. That is, an agent will always be able to maximize his payoff, which is consistent with his preferences among different alternative outcomes. This rationality of the agent requires complete information and a well-defined and consistent set of choices (e.g., actions). However, in reality, this assumption rarely holds. A number of results from experimental economics and the social sciences have shown that strong rationality (i.e., so-called hyperrational behavior) rarely exists and cannot describe the behavior of real human beings. For example, people tend to have limited information about available choices and consequences. Also, people occasionally make decisions irrationally. On the other hand, evolutionary-game theory has been developed to model the behavior of biological agents (e.g., insects and animals). Hence, a strong rationality assumption is not required. Therefore, an evolutionary-game formulation will be suitable for scenarios that involve human beings as agents who may not display hyperrational behavior.

• Traditional non-cooperative game theory has mostly been developed in a static setting. It cannot capture the adaptation of agents to change their strategies and reach equilibrium over time. Although in a non-cooperative game the dynamics of the decision-making process can be modeled in extensive form, such an extensive-form description becomes intractable for most game settings with reasonable complexity. In addition, it relies on the selection of a strategy at the outset of the game. This cannot capture the fact that an agent can observe his opponent's behavior, learn from this observation, and optimize the strategy selection according to the knowledge gained. In contrast, an evolutionary game is based on an evolutionary process, which is dynamic in nature. An evolutionary game establishes the dynamics of interactions among agents in the population (i.e., strategy adaptation over time).

6.1 The evolutionary process

In an evolutionary game, the game is played repeatedly by agents who are selected from a large population. Two major mechanisms of the evolutionary process and the evolutionary game are *mutation* and *selection*. Mutation is the mechanism of modifying the characteristics of an agent (e.g., genes of the individual or strategy of the player). As a result, agents with new characteristics are introduced into the population. The selection mechanism is then applied to retain the agents with high fitness while eliminating agents with low fitness [170]. In particular, while the mutation mechanism is used to maintain the diversity of a population, the selection mechanism is used to promote agents with higher fitness over other agents. In the evolutionary game, the mutation mechanism is described by *evolutionarily* stable strategies (ESS), and the selection mechanism is described by *replicator dynamics*. In other words, ESS is used to study the static evolutionary game while replicator dynamics is used for the dynamic evolutionary game.

6.1.1 Evolutionarily stable strategies

ESS is the key concept in the evolutionary process in which a group of agents choosing one strategy will not be replaced by other agents choosing a different strategy when the mutation mechanism is applied. In the game context, a pure or mixed strategy *s* chosen by the initial group of agents in a population is referred to as the *incumbent strategy*. A small group of agents whose population share is $\epsilon \in (0, 1)$ may choose a different pure or mixed strategy *s'*, referred to as the *mutant strategy*. In this case, an agent selected from the population will use strategies *s* and *s'* with probabilities $1 - \epsilon$ and ϵ , respectively. The payoff for the selected agent in this game is identical to that in the traditional noncooperative game when the player chooses a mixed strategy *s* given that the opponent chooses strategy *s'*. Strategy *s* is called *evolutionarily stable* if, for each strategy *s' \ne s*, there is $\hat{\epsilon} \in (0, 1)$ such that the following condition is satisfied:

$$u(s,\epsilon s'+(1-\epsilon)s) > u(s',\epsilon s'+(1-\epsilon)s),$$
(6.1)

for all $\epsilon \in (0, \hat{\epsilon})$. With ESS, the mutant population share will tend to decrease, since it gains lower payoff and, hence, has lower growth rate. In this case, strategy *s* is then immune to mutation.

We now establish the relationship between ESS and the Nash equilibrium. If we consider the linear payoff function from (6.1), which would definitely be the case in a matrix game with mixed strategies, we have the following condition:

$$(1-\epsilon)u(s,s) + \epsilon u(s,s') > (1-\epsilon)u(s',s) + \epsilon u(s',s').$$
(6.2)

As the value of ϵ approaches zero, by continuity we have $u(s, s) \ge u(s', s)$, which shows (by symmetry) that ESS is a mixed-strategy Nash equilibrium (MSNE). This shows that for *s* to be ESS, it is necessary that (s, s) be a MSNE. However, this may not be sufficient, unless u(s, s) > u(s', s). If, on the other hand, u(s, s) = u(s', s), then we need as a second-order condition u(s, s') > u(s', s'), which again provides a sufficient condition for ESS.

Hawk–dove game

The classical example of an evolutionary game is the Hawk–Dove game [463]. In this game, there are two types of agents competing for a resource (i.e., food) of a fixed value V. Each agent chooses his strategy from a set of two possibilities (i.e., hawk and dove). If an agent chooses to be hawk, it will display aggressive behavior and will not stop fighting until it is injured or until the opponent retreats. On the other hand, if the agent chooses to be a dove, it will display mild behavior and always retreat instantly if the opponent initiates aggressive behavior. Therefore, we have the following scenarios in this game:

- If both agents adopt hawk behavior (i.e., competing aggressively), the competition will result in both being equally injured. The cost of competition reduces the value of the resource by a constant *C*.
- If one agent adopts hawk behavior and the other agent adopts dove behavior, the dove immediately retreats and earns zero payoff, while the hawk captures the resource *V*.
- When both adopt dove behavior, they will share the resource equally.

According to these scenarios, the payoff matrix for the Hawk–Dove game can be expressed as follows:

Hawk
 Dove

 Hawk

$$1/2(V-C), 1/2(V-C)$$
 $V, 0$

 Dove
 $0, V$
 $V/2, V/2$

where the first and second elements correspond to the payoffs of agents choosing strategy row-wise or column-wise, respectively. The strategy will be evolutionarily stable if, when almost all agents in the population adopt this strategy, no mutant (i.e., a small number of agents adopting a different strategy) can invade. In other words, the small number of agents cannot make the large group of agents move from the ESS. This property can be demonstrated as follows. Let $\phi(s_1, s_2)$ denote the change in fitness for an agent adopting strategy s_1 against an opponent adopting strategy s_2 , and let f(s) denote the total fitness of an agent adopting strategy s. Let f_0 denote the initial fitness, s denote the ESS, and s' denote the mutant strategy. The fitness of the agents adopting the different strategies can be expressed as follows:

$$f(s) = f_0 + (1 - \epsilon)\phi(s, s) + \epsilon\phi(s, s'), \qquad (6.4)$$

$$f(s') = f_0 + (1 - \epsilon)\phi(s', s) + \epsilon\phi(s', s'), \qquad (6.5)$$

where ϵ is the proportion of the population adopting the mutant strategy s'. For ESS, the fitness of an agent adopting strategy s must be larger than that of those members of the population choosing strategy s', i.e., f(s) > f(s'). If ϵ approaches zero, it is required that either of these conditions holds, i.e.,

$$\phi(\mathbf{s},\mathbf{s}) > \phi(\mathbf{s}',\mathbf{s}),\tag{6.6}$$

$$\phi(\mathbf{s}, \mathbf{s}) = \phi(\mathbf{s}', \mathbf{s}) \text{ and } \phi(\mathbf{s}, \mathbf{s}') > \phi(\mathbf{s}', \mathbf{s}'). \tag{6.7}$$

With the payoff matrix of the Hawk–Dove game, the dove is not ESS since a pure population of doves can be invaded by a hawk mutant. In this case, if the value V of the resource is larger than the cost of both agents behaving aggressively (i.e., V > C), then the hawk is ESS as there is value in both agents competing for a resource even though they would be hurt. Otherwise, there is no ESS in this game if agents adopt a pure strategy. However, if mixed strategies are considered, there could be ESS in this Hawk–Dove game.

6.1.2 Replicator dynamics

The population can be divided into multiple groups, and each group adopts a different pure strategy. Replicator dynamics can model the evolution of the group size over time. Unlike ESS, in replicator dynamics agents will play only pure strategies. A large but finite group of agents adopts strategy $s \in S$, where S is a set of strategies. Let $n_s(t)$ denote the number of agents using strategy s at time t. The total number of agents in a population is denoted by $N(t) = \sum_{s \in S} n_s(t)$. In this way, the proportion or fraction of agents using a pure strategy s (i.e., population share) is denoted by $x_s(t) = n_s(t)/N(t)$. The population state can then be defined as the vector $\mathbf{x}(t) = \begin{bmatrix} \cdots & x_s(t) & \cdots \end{bmatrix}^T$ of dimension |S|. Let the payoff to an agent using strategy s given the population state **x** be denoted by $u(s, \mathbf{x})$. The average payoff for the population, which is the payoff to an agent selected randomly from a population, is given by $\overline{u}(\mathbf{x}) = \sum_{s \in S} x_s u(s, \mathbf{x})$. Naturally, the reproduction rate of each agent (i.e., the rate at which the agent switches from one strategy to another) depends on the payoff. In other words, agents will switch to the strategy that leads to a higher payoff. The larger the payoff, the faster the strategy switching. As a result, the group size of agents ensuring higher payoff will grow over time because the agents having low payoff will switch their strategies. Based on this fact, the dynamics of the population share can be expressed as follows:

$$\dot{x}_{s} = x_{s} \left(u(s, \mathbf{x}) - \overline{u}(\mathbf{x}) \right), \tag{6.8}$$

where \dot{x}_s is the time derivative of the population state x_s . With these dynamics, the groups of agents with payoffs higher than the average will grow in size over time.

The evolutionary equilibrium can be determined at $\dot{x}_s = 0$. That is, the fractions of the population choosing different strategies cease to change. It is important to analyze the stability of the replicator dynamics to determine the evolutionary equilibrium. Evolutionary equilibrium can be stable (i.e., equilibrium is robust to the local perturbation) in the following two cases:

- Given an initial point of replicator dynamics sufficiently close to the evolutionary equilibrium, the solution path of replicator dynamics will remain arbitrarily close to the equilibrium. This is referred to as *Lyapunov stability*.
- Given an initial point of replicator dynamics close to the evolutionary equilibrium, the solution path of replicator dynamics converges to the equilibrium. This is referred to as *asymptotic stability*.

Two main approaches to prove the stability of evolutionary equilibrium are based on the Lyapunov function and eigenvalues of a corresponding matrix of the linearized system, as in the study of the stability of non-linear systems. The details for such a stability analysis can be found in [462].

The Prisoner's Dilemma

Replicator dynamics can be established for the classical game of The Prisoner's Dilemma. In this game, two agents choose a strategy of either *cooperate* or *defect*. The payoff matrix of this game can be written as follows:

$$\begin{array}{c|cccc}
Cooperate & Defect \\
\hline
Cooperate & R, R & S, T \\
\hline
Defect & T, S & P, P \\
\end{array}$$
(6.9)

where T > R > P > S. For simplicity, it is assumed that the payoffs of The Prisoner's Dilemma are identical for all agents in a large population. Let x_C and x_D denote the proportions of the population adopting cooperate and defect strategies, respectively. The average fitnesses of agents adopting these two strategies are denoted by u_C and u_D , respectively. The average fitness of the entire population is \overline{u} . These fitness values can be obtained from

$$u_{\mathsf{C}} = u_0 + x_{\mathsf{C}}\phi(\mathsf{C},\mathsf{C}) + x_{\mathsf{D}}\phi(\mathsf{C},\mathsf{D}), \qquad (6.10)$$

$$u_{\rm D} = u_0 + x_{\rm C}\phi({\rm D},{\rm C}) + x_{\rm D}\phi({\rm D},{\rm D}),$$
 (6.11)

$$\overline{u} = x_{\rm C} u_{\rm C} + x_{\rm D} u_{\rm D}, \qquad (6.12)$$

where u_0 is an initial payoff and $\phi(s_1, s_2)$ is the change in fitness for an agent adopting strategy s_1 against an opponent adopting strategy s_2 . The future proportion of the

population adopting the strategies (i.e., \hat{x}_C and \hat{x}_D) depends on the current proportion (i.e., x_C and x_D). This relationship can be expressed as follows:

$$\hat{x}_{\mathsf{C}} = \frac{x_{\mathsf{C}} u_{\mathsf{C}}}{\overline{u}},\tag{6.13}$$

$$\hat{x}_{\mathsf{C}} - x_{\mathsf{C}} = \frac{x_{\mathsf{C}} \left(u_{\mathsf{C}} - \overline{u} \right)}{\overline{u}},\tag{6.14}$$

$$\hat{x}_{\rm D} = \frac{x_{\rm D} \, u_{\rm D}}{\overline{u}},\tag{6.15}$$

$$\hat{x}_{\rm D} - x_{\rm D} = \frac{x_{\rm D} \left(u_{\rm D} - \overline{u} \right)}{\overline{u}}.$$
(6.16)

If we consider a small time interval, the difference equations can be approximated by the differential equations

$$\frac{\mathrm{d}x_{\mathrm{C}}}{\mathrm{d}t} = \dot{x}_{\mathrm{C}} \approx \frac{x_{\mathrm{C}} \left(u_{\mathrm{C}} - \overline{u} \right)}{\overline{u}},\tag{6.17}$$

$$\frac{\mathrm{d}x_{\mathrm{D}}}{\mathrm{d}t} = \dot{x}_{\mathrm{D}} \approx \frac{x_{\mathrm{D}} \left(u_{\mathrm{D}} - \overline{u}\right)}{\overline{u}}.$$
(6.18)

These differential equations are the replicator dynamics. For The Prisoner's Dilemma, we have $u_{\rm C} = u_0 + x_{\rm C}R + x_{\rm D}S$ and $u_{\rm D} = u_0 + x_{\rm C}T + x_{\rm D}P$. Since T > R and P > S, it is clear that $u_{\rm D} > u_{\rm C}$, and $\frac{u_{\rm D} - \overline{u}}{\overline{u}} > 0$ and $\frac{u_{\rm C} - \overline{u}}{\overline{u}} < 0$. Therefore, as time increases, the proportion of the population adopting the cooperate strategy will approach zero (i.e., become extinct). From replicator dynamics, the defect strategy constitutes the evolutionary equilibrium. Also, it can be proven that the defect strategy is the ESS of The Prisoner's Dilemma game [504].

6.1.3 The evolutionary game and reinforcement learning

Reinforcement learning (i.e., Q-learning) of agents can be modeled as an evolutionary game [481]. For a game with two players, let U_1 and U_2 denote payoff matrices for players 1 and 2, respectively. With two strategies (i.e., s_1 and s_2), these matrices are defined as follows:

$$\mathbf{U}_{1} = \begin{bmatrix} u_{1}(s_{1}, s_{1}) & u_{1}(s_{1}, s_{2}) \\ u_{1}(s_{2}, s_{1}) & u_{1}(s_{2}, s_{2}) \end{bmatrix}, \quad \mathbf{U}_{2} = \begin{bmatrix} u_{2}(s_{1}, s_{1}) & u_{2}(s_{1}, s_{2}) \\ u_{2}(s_{2}, s_{1}) & u_{2}(s_{2}, s_{2}) \end{bmatrix}, \quad (6.19)$$

where $u_i(s, s')$ is the payoff for player *i* when this player chooses strategy *s* while the other player chooses *s'*. The dynamics of player 1 can be expressed as follows:

$$\frac{\mathrm{d}x_{s,1}}{\mathrm{d}t} = \dot{x}_{s,1} = x_{s,1}\alpha((\mathbf{U}_1\mathbf{x}_2)_s - \mathbf{x}_1\mathbf{U}_1\mathbf{x}_2) + x_{s,1}\alpha\sum_{s'}x_{s',1}\ln\left(\frac{x_{s',1}}{x_{s,1}}\right), \quad (6.20)$$

and the dynamics of player 2 can be expressed as follows:

$$\frac{dx_{s,2}}{dt} = \dot{x}_{s,2} = x_{s,2}\alpha((\mathbf{U}_2\mathbf{x}_1)_s - \mathbf{x}_2\mathbf{U}_2\mathbf{x}_1) + x_{s,2}\alpha\sum_{s'} x_{s',2}\ln\left(\frac{x_{s',2}}{x_{s,2}}\right).$$
(6.21)

These dynamics represent the evolution of both players using the Q-learning algorithm in terms of probability of selecting a strategy (i.e., $x_{s,i}$ is the probability of selecting strategy *s* of player *i*). It can be observed that the first terms of (6.20) and (6.21), which account for the strategy-selection process of the players, are the same as those for replicator dynamics. The second terms account for the mutation process. Specifically, the mutation and selection processes can be considered as the exploration and exploitation steps in reinforcement learning. Details of the relationship between evolutionary games and reinforcement learning can be found in [481].

6.2 Applications of evolutionary games in wireless communications and networking

In this section, selected applications of evolutionary-game theory in wireless communications and networking are discussed. These applications are from various aspects and protocols of such networks. Analyses based on both ESS and replicator dynamics are presented.

6.2.1 Congestion control

The competition between two types of behaviors (i.e., aggressive and peaceful) in wireless nodes to access the channel using a certain protocol can be modeled as an evolutionary game [29]. Congestion control is required in the transport protocol to avoid performance degradation by the ongoing users. Congestion control limits the transmission rate according to the available network resources. In particular, the protocol observes the success of data transmission. The transmission rate can be increased if the transmitted packet is successful, and decreased if the transmitted packet fails. The transmission rate can be adjusted by changing the congestion window size (i.e., the maximum number of packets to be transmitted). The increase or decrease in the rate of transmission defines the aggressiveness of the protocol. The transmission control protocol (TCP) with an additive-increase multiplicative-decrease (AIMD) mechanism can control this aggressiveness through the parameters determining the increase and decrease [29]. In particular, if the transmitted packet is successful, the window size will linearly increase by α packets for every round trip time. Otherwise, the window size will decrease by β proportional to the current size. For the TCP New-Reno protocol, the values of these parameters are $\alpha = 1$ and $\beta = 1/2$.

With congestion control, when multiple flows share the same link, a competitive situation arises. In a wired network environment, it is found that the aggressive strategy of all flows (i.e., large values of α and β) becomes the Nash equilibrium, and the performance will degrade significantly because of the congestion in the network. However, in a wireless environment, the congestion due to playing Nash equilibrium not only degrades the performance but also shortens the battery life of a mobile node. The analysis of the TCP protocol in a wireless environment is performed in [29], where an evolutionary-game model similar to the Hawk–Dove game [302, 162] is introduced. The

TCP New-Reno protocol is considered because it is shown to be well behaved in wireless networks [461]. However, the analysis can be applied to other TCP variants as well.

Static game

The evolutionary game is formulated as follows. There are two populations (i.e., groups) of flows with AIMD TCP. The flow from population $i \in \{1,2\}$ is characterized by parameters α_i and β_i , the increase and decrease rates, respectively. Strategy *s* of flow is to be aggressive (i.e., hawk or H) or to be peaceful (i.e., dove or D), i.e., $s \in \{H, D\}$. The parameters associated with these strategies are given as $(\alpha_i, \beta_i) \in \{(\alpha_H, \beta_H), (\alpha_D, \beta_D)\}$ for $\alpha_H \ge \alpha_D$, and $\beta_H \ge \beta_D$. The transmission rate at time *t* is denoted by $r_i(t)$. It is assumed that there is light traffic in the network in which the flow can utilize almost all resources. Packet loss occurs when the total transmission rate of all flows reaches the capacity *C*, i.e., $x_1r_1 + x_2r_2 = C$. The payoff u_i of flow in population *i* is defined as

$$u_i = \tau_i - \omega L, \tag{6.22}$$

where τ_i is the average throughput, *L* is the loss rate, and ω is the weight for the loss. The throughput of flow from population *i* is given by

$$\tau_i = \frac{1 + \beta_i}{2} \frac{\alpha_i \overline{\beta}_j}{\alpha_i \overline{\beta}_i + \alpha_j \overline{\beta}_i} C, \qquad (6.23)$$

where $\overline{\beta}_i = 1 - \beta_i$. The average loss rate is the same for all flows, i.e.,

$$L = \frac{1}{C} \left(\frac{\alpha_i}{\overline{\beta}_i} + \frac{\alpha_j}{\overline{\beta}_j} \right).$$
(6.24)

The average throughput and loss rate can be defined as functions of the strategies of two populations, i.e., $\tau_i(s_i, s_j)$ and $L(s_i, s_j)$, respectively. In this case, it is shown that $\tau_i(H, H) = \tau_i(D, D)$. When the loss rate is considered, it increases as the flow becomes more aggressive, i.e., larger values of α_i and β_i . Therefore, it can be shown that $u_i(H, H) < u_i(D, D)$ and $u_i(D, H) < u_i(D, D)$. In this form, the game becomes a Hawk–Dove game whose solution is the ESS. The mixed ESS (i.e., the probability of using strategy H) is given by

$$x^*(\omega) = \frac{\eta_1 - \eta_2 \omega}{\eta_3},\tag{6.25}$$

where

$$\eta_1 = \left(\overline{\mu}\frac{1+\beta_i}{2} - \frac{1+\beta_j}{4}\right)C, \quad \eta_2 = \frac{1}{C}\left(\frac{\alpha_i}{\overline{\beta}_i} - \frac{\alpha_j}{\overline{\beta}_j}\right), \tag{6.26}$$

$$\eta_3 = C(1/2 - \mu) \frac{\beta_i - \beta_j}{2}, \quad \mu = \frac{\alpha_j \beta_i}{\alpha_j \overline{\beta}_i + \alpha_i \overline{\beta}_j}, \quad (6.27)$$

where $\overline{\mu} = 1 - \mu$. In this case, η_2 and η_3 are positive numbers. As a result, the probability x^* decreases linearly with the weight of packet loss ω . It can be concluded that the application that is loss-sensitive will tend to use a less aggressive strategy at ESS [29].

In addition, it is observed that when there is a change of wireless link (e.g., due to fading), the aggressive flow will take a longer time to adjust the transmission rate to the available capacity. Therefore, the aggressive flow will experience more loss from this longer transient period.

Dynamic game

The dynamics of strategy selection by the flows in two populations can also be analyzed using replicator dynamics, expressed in this case as follows:

$$\dot{x}_{s}(t) = \frac{\mathsf{d}x_{s}(t)}{\mathsf{d}t} = x_{s}(t) \mathcal{K}\left(u(s, \mathbf{x}(t)) - \sum_{s'} x_{s'}(t) u(s', \mathbf{x}(t))\right), \tag{6.28}$$

where x_s is the proportion of the population choosing strategy s and $\mathbf{x}(t)$ is a vector of x_s at time t; $u(s, \mathbf{x}(t))$ is the payoff for using strategy s, and K is a positive constant standing for the speed of strategy change.

However, the payoff may not immediately affect the strategy selection of the population. Let \hat{t} denote the time duration (i.e., delay) for the population to change strategy after the payoff has changed. The replicator dynamics with delay can be expressed as follows:

$$\dot{x}_{s}(t) = x_{s}(t) \mathcal{K}\left(\sum_{s'} x_{s'}(t-\hat{t})u(s,s') - \sum_{s',s''} x_{s'}(t)u(s',s'')x_{s''}(t-\hat{t})\right).$$
(6.29)

Given the speed K and delay \hat{t} , the strategy adaptation of two populations will be stable if the stability condition $\hat{t}K < \theta$ is satisfied, where

$$\theta = \frac{(u(s',s) - u(s,s)) + (u(s,s') - u(s',s'))}{(u(s',s) - u(s,s))(u(s,s') - u(s',s'))}.$$
(6.30)

If the delay is large, the strategy change can fluctuate. It can be shown that the evolutionary equilibrium can be reached if the delay is small.

6.2.2 Evolutionary game for the Aloha protocol

In a network with the Aloha protocol, the nodes can access a channel independently to maximize their utility. An evolutionary game is formulated for the node population using the Aloha protocol in [473]. For the network model, the number of nodes, which are the players, in the population is random, and the nodes are randomly located in the service area. Each node has a transmission range of r, and the node can either transmit or back off (i.e., use strategy). The cost of transmission (e.g., energy consumption) is denoted by δ , and transmission loss is due to collision only. Collision occurs when there is more than one node transmitting at the same time. The cost to the node due to collision is

denoted by Δ , but if there is no collision the node gains benefit, denoted by V, where $V > \delta$. If none of the nodes transmits (i.e., they all choose to back off) the regret cost is denoted by κ .

With the evolutionary-game setting, the node chooses pure strategy in which the state of the evolutionary game indicates the fraction of nodes in the population choosing these strategies. Let x denote the proportion of the population choosing the transmit strategy, so (1 - x) denotes the proportion choosing the backoff strategy. The payoff of the node with k other interfering nodes when strategy s is used can be expressed as

$$u(s,k,x) = \begin{cases} -(\Delta - \delta)(1 - \eta_k) + (V - \delta)\eta_k, & s = \text{transmit,} \\ -\kappa(1 - x)^k, & s = \text{back off,} \end{cases}$$
(6.31)

where $\eta_k = (1-x)^k$.

Three cases are studied in [473]. In the first case, the node does not know the number of interfering nodes. In the second case, the node knows that there is an interfering node in the range of its receiver. Therefore, this node will transmit with a certain probability if no potential interfering nodes are present. The third case corresponds to the dense network in which there is always an interfering node. Properties of this evolutionary game (i.e., the existence and uniqueness of ESS) are analyzed for these three cases using the ratio between the collision cost and the difference between the global cost for the node and the benefit. This ratio is defined as

$$\psi = \frac{\Delta + \delta}{V + \Delta + \kappa}.\tag{6.32}$$

In the first case, the game has a unique ESS if $\Pr(k = 0) < \psi$, where $\Pr(k)$ is the probability of the number of interfering nodes being *k*. That is, the probability of the number of interfering nodes being zero is less than the ratio ψ . The ESS is at $x^* = \phi^{-1}(\psi)$, where $\phi(\cdot) = \sum_{k>0} \Pr(k)(1-x)^k$. In the second case, the game has a unique ESS if $\Pr(k=0) < \frac{\Delta+\delta}{V+\Delta}$. The ESS is at $x^* = \phi^{-1}\left(\frac{\Delta+\delta+\kappa\Pr(k=0)}{V+\Delta+\kappa}\right)$. In the third case, the game always has a unique ESS, which is the solution of $\sum_{k>1} \Pr(k)(1-x)^k = \psi$.

The game can be further analyzed by considering the spatial node distribution. Two scenarios are considered, namely when a fixed number of nodes are in a local interaction, and when the number of nodes is random.

Fixed number of nodes

With *n* nodes, for $n \ge 2$, the average payoff can be expressed as

$$\overline{u} = x \left((-\Delta - \delta)(1 - (1 - x)^{n-1}) + (V - \delta)(1 - x)^{n-1} \right) - \kappa (1 - x)^n.$$
(6.33)

It is shown in [473] that the ESS exists and is unique. This ESS is at $x^* = 1 - \psi^{1/(n-1)}$.

Random number of nodes

In this scenario, the nodes are distributed in the area following a Poisson distribution with density λ . The ESSs for different cases can be obtained. For example, in the first

case (i.e., when the node does not know the number of interfering nodes), the probability of having k interfering nodes is

$$\Pr(k) = \frac{\left(\lambda \pi r^2\right)^k}{k!} \exp\left(-\lambda \pi r^2\right), \qquad (6.34)$$

for $k \ge 1$, where *r* is the transmission range of the node; hence πr^2 is the transmission area, which is a circle. The corresponding ESS, x^* , is the solution of the following equation:

$$\exp\left(-\lambda\pi r^2 x\right) = \psi. \tag{6.35}$$

Then, replicator dynamics can be used to evaluate the stability of ESS. The corresponding replicator dynamics can be expressed as

$$\dot{x}(t) = (V + \Delta + \kappa)x(t)(1 - x(t))(\phi(x(t)) - \psi).$$
(6.36)

Since $\phi(\cdot)$ is decreasing in (0,1), the derivative of $x(1-x)(\phi(x) - \psi)$ at the ESS is negative. Therefore, ESS is stable.

6.2.3 Evolutionary game for WCDMA access

The evolutionary game is now formulated for the WCDMA system. The number of interfering nodes is random, and depends on the geographical location of the mobile nodes. The decentralized power-control problem of WCDMA access is considered [473]. The mobile nodes have two strategies, to use high or low power levels, which correspond to transmit power $P_{\rm H}$ and $P_{\rm L}$, respectively. Given the random location of the mobile nodes, the signal-to-interference-plus-noise ratio (SINR) with distance *r* between transmitter and receiver of node *i* is given by

$$\gamma_{i}(P_{i}, x, r) = \begin{cases} \frac{gP_{i}/r_{0}^{\alpha}}{\sigma + \beta I(x)}, & \text{if } r \leq r_{0}, \\ \frac{gP_{i}/r^{\alpha}}{\sigma + \beta I(x)}, & \text{if } r > r_{0}, \end{cases}$$
(6.37)

where P_i is the strategy of node *i* (i.e., $P_i \in \{P_H, P_L\}$), *x* is the proportion of the population choosing P_H , *g* is the channel gain, r_0 is the radius-of-reception circle of the receiver, α is the attenuation order, with a value between 3 and 6, σ is the noise power, and β is the inverse of the processing gain. I(x), the total interference from all nodes to the receiver of node *i*, is given by

$$I(x) = g\lambda\pi(xP_{\mathsf{H}} + (1-x)P_{\mathsf{L}})\left(\frac{\alpha}{\alpha-2}r_{0}^{-(\alpha-2)}\right), \qquad (6.38)$$

where λ is the density of the nodes. The payoff for node *i* is then defined as

$$u_i(P_i, x) = \int_0^R \log(1 + \gamma_i(P_i, x, r))\zeta(r) dr - w_p P_i, \qquad (6.39)$$

where *R* is the transmission range and w_p is the cost weight due to adopting transmit power P_i (e.g., energy consumption). $\zeta(r)$ is the probability density function given the density ν of receivers (e.g., $\zeta(r) = \nu e^{-\nu r}$).

Based on this evolutionary-game formulation, the sufficient condition [29] for existence and uniqueness of the ESS in WCDMA access is established as follows. For all density functions $\zeta(r)$, if $\mathscr{H}(1) < w_p(P_H - P_L) < \mathscr{H}(0)$, then there exists a unique ESS x^* given by $x^* = \mathscr{H}^{-1}(w_p(P_H - P_L))$. In this case, function $\mathscr{H}(x)$ is defined as follows:

$$\mathscr{H}(x) = \log\left(\frac{1+\gamma_i(P_{\mathsf{H}}, x, r)}{1+\gamma_i(P_{\mathsf{L}}, x, r)}\right)\zeta(r).$$
(6.40)

It is shown that this function $\mathscr{H}(x)$ is continuous and strictly monotonic, which is required for the proof of stability based on sufficient condition. With ESS, it is observed that the cost weight w_p can be adjusted so that the node will select the optimal transmit power. In particular, x^* decreases as cost weight w_p increases. Therefore, the node becomes less aggressive in its use of high transmit power, and the optimal transmission rate can be achieved because of reduction of interference in the network.

The dynamics of the evolutionary-game formulation of WCDMA access can be established based on replicator dynamics, as follows:

$$\dot{x}(t) = x(t)(1 - x(t))(\mathscr{H}(x(t)) - w_{\rm p}(P_{\rm H} - P_{\rm L})).$$
(6.41)

Again, it can be seen that the function $\mathscr{H}(\cdot)$ is decreasing according to x(t). As a result, the derivative of $x(1-x)(\mathscr{H}(x) - w_p(P_H - P_L))$ at $x^* = \mathscr{H}^{-1}(w_p(P_H - P_L))$ can be evaluated, and it can be shown that the system is asymptotically stable at x^* [473].

6.2.4 Routing-potential game

In [30], non-cooperative traffic routing among non-cooperative users is formulated as a potential game and subsequently analyzed by evolutionary dynamics (i.e., potentialgame replicator dynamics). Consider a general network topology modeled as a directed graph, where \mathcal{L} is a set of links and \mathcal{N} is a set of users. In this case, the total numbers of links and users are denoted by $L = |\mathcal{L}|$ and $N = |\mathcal{N}|$, respectively. The cost of utilizing the link *l* is a function of the aggregated traffic rate or total load, i.e., $\lambda_l(A_l\lambda_l + B_l)$, where λ_l is the total load on link $l \in \mathcal{L}$ from all users and A_l and B_l are constants. There are *N* users using this network (i.e., *N* players), and user *i* transmits data from source S(i) to destination D(i). If we let $\lambda_{l,i}$ denote the transmission rate of user *i* on link *l*, the cost for user *i* on link *l* can be expressed as

$$\mathscr{C}_{l,i}(\lambda_l) = \lambda_{l,i}(A_l\lambda_l + B_l). \tag{6.42}$$

Therefore, the total cost for user *i* is defined as

$$\mathscr{C}_{i}(\boldsymbol{\lambda}) = \sum_{l \in \mathcal{L}} \lambda_{l,i} (A_{l} \lambda_{l} + B_{l}) = \sum_{l \in \mathcal{L}} \mathscr{C}_{l,i}(\boldsymbol{\lambda}), \qquad (6.43)$$

where λ is a vector of the transmission rates on all links for all users. The objective of each user is to choose a transmission rate on the link so that the demand is met and the cost is minimized. The demand (i.e., total transmission rate from source S(i) to destination D(i)) of user *i* is denoted by Λ_i . The strategy is $\lambda_{I,i}$ and the negative payoff is $\mathscr{C}_i(\lambda)$. In addition to the above objective, user *i* has to meet feasibility conditions: positivity (i.e., $\lambda_{I,i} \ge 0$) and flow conservation,

$$r_i(v) + \sum_{l \in \mathcal{I}} \lambda_{l,i} = \sum_{l \in \mathcal{O}} \lambda_{l,i}, \qquad (6.44)$$

where

$$r_i(v) = \begin{cases} \Lambda_i & \text{if } v = S(i), \\ -\Lambda_i & \text{if } v = D(i), \\ 0 & \text{otherwise.} \end{cases}$$
(6.45)

 \mathcal{I} and \mathcal{O} are the sets of links that are the inputs and outputs of node v, respectively. It is proved in [30] that the above game formulation is the potential game whose potential function is

$$\mathscr{P}(\boldsymbol{\lambda}) = \sum_{l \in \mathcal{L}} \frac{A_l}{2} \left(\sum_{i=1}^N (\lambda_{l,i})^2 + \left(\sum_{i=1}^N \lambda_{l,i} \right)^2 \right) + \sum_{l \in \mathcal{L}} B_l \lambda_l.$$
(6.46)

The Nash equilibrium can be obtained as the solution of the constrained optimization $\min_{\lambda} \mathscr{P}(\lambda)$. The Lagrangian function of this optimization problem is

$$\mathscr{L}(\boldsymbol{\lambda},\boldsymbol{\mu}) = \mathscr{P}(\boldsymbol{\lambda}) - \sum_{i=1}^{N} \left(\sum_{\boldsymbol{\nu}\in\mathcal{V}} \mu_{i,\boldsymbol{\nu}} \left(r_i(\boldsymbol{\nu}) + \sum_{l\in\mathcal{I}} \lambda_{l,i} - \sum_{l\in\mathcal{O}} \lambda_{l,i} \right) \right), \quad (6.47)$$

where \mathcal{V} is a set of nodes in the network, and $\boldsymbol{\mu}$ is a vector of Lagrange multipliers (i.e., $\mu_{i,v}$ for user *i* at node *v*).

Then, the evolutionary-game model is formulated as follows. It is assumed that the demand Λ_i of user *i* is constant. User *i* is considered to be the population with mass Λ_i of infinitesimal users. As a result, the proportion of population *i* (of user *i*) choosing strategy *l* (i.e., link *l*) is $x_{l,i} = \lambda_{l,i}/\Lambda_i$. With this population *i*, the cost can be rewritten as

$$\mathscr{C}_{l,i}(x_l) = (x_{l,i}\Lambda_i) \left(A_l \sum_{i=1}^N x_{l,i}\Lambda_i + B_l \right), \qquad (6.48)$$

where $\mathbf{x}_{l} = \begin{bmatrix} x_{l,1} & \cdots & x_{l,i} & \cdots & x_{l,N} \end{bmatrix}^{T}$. The replicator dynamics can be expressed as

$$\dot{x}_{l,i} = x_{l,i} \left(\mathscr{F}_i(\mathbf{x}) - f_{l,i}(x_l) \right),$$
(6.49)

where $f_{l,i}(x_l)$ is the marginal cost and $\mathscr{F}_i(\mathbf{x})$ is the average cost for the population; \mathbf{x} is a vector of the proportions of the population of users. These costs can be obtained from

$$f_{l,i}(x_l) = \frac{\partial \mathscr{C}_i(\boldsymbol{\lambda})}{\partial \lambda_{l,i}} = A_l \sum_{i'=1}^N x_{l,i'} \Lambda_{i'} + A_l x_{l,i'} \Lambda_{i'} + B_l, \qquad (6.50)$$

$$\mathscr{F}_{i}(\mathbf{x}) = \sum_{l \in \mathcal{L}} x_{l,i} f_{l,i}(x_{l}).$$
(6.51)

In the special case when $B_l = B$, $\forall l$, that is, a constant, it can be proved that the Nash equilibrium \mathbf{x}^* is the stable point where $\dot{x}_{l,i} = 0$, $\forall l$ and $\forall i$. In this case, the Nash equilibrium for each user on each link is

$$x_{l,i}^* = \frac{1/A_l}{\sum_{l=1}^{L} 1/A_l}.$$
(6.52)

However, in some cases the Nash equilibrium may not be the stationary point of the dynamics (i.e., when B_l is not identical for all links and at the equilibrium the traffic rate on each link is positive).

The stability (i.e., Lyapunov stability) of this Nash equilibrium is investigated by constructing a Lyapunov function. This function can be derived from the potential function defined in (6.46), i.e.,

$$\mathscr{Y}(\mathbf{x}) = \mathscr{P}(\mathbf{x}) - \mathscr{P}(\mathbf{x}^*), \tag{6.53}$$

where $\mathscr{P}(\mathbf{x})$ is the potential function, a function of the proportion \mathbf{x} of the population. The proof in [30] is based on the fact that for all $\mathbf{x} \neq \mathbf{x}^*$, $\mathscr{Y}(\mathbf{x}) > 0$ and also $\frac{d\mathscr{Y}(\mathbf{x})}{dt} \leq 0$. Therefore, the Nash equilibrium is Lyapunov-stable.

6.2.5 Cooperative sensing in cognitive radio

In a cognitive-radio network, spectrum sensing plays an important role for unlicensed users (i.e., secondary users) to opportunistically access the spectrum allocated to the licensed users (i.e., primary users). Spectrum sensing must be performed by secondary users to ensure that the spectrum is not occupied by primary users. This can be achieved by sampling the signal with hypotheses that a primary user is present or absent, denoted by H_1 and H_0 , respectively. The received signal is expressed as

$$r(t) = \begin{cases} gd(t) + w(t), & \text{if } H_1, \\ w(t), & \text{if } H_0, \end{cases}$$
(6.54)

where g is the channel gain between a primary user and a sensing secondary user, d(t) is the signal from the primary user, and w(t) is additive white Gaussian noise (AWGN). Given the slot duration, denoted by T, and the sensing time, denoted by T_{sense} for

 $T_{sense} < T$, the average throughput for a secondary user when the spectrum is idle (i.e., H_0 event) is

$$\tau_{H_0} = \frac{T - T_{\text{sense}}}{T} (1 - P_{\text{fal}}) R_i, \qquad (6.55)$$

where P_{fal} is the false-alarm probability (i.e., the probability that the spectrum is idle, but the secondary user falsely detects activity of a primary user), and R_i is the transmission rate of the secondary user *i*. Similarly, the average throughput when the spectrum is occupied by a primary user (i.e., H_1 event) is

$$\tau_{H_1} = \frac{T - T_{\text{sense}}}{T} (1 - P_{\text{det}}) R_i, \qquad (6.56)$$

where P_{det} is the detection probability (i.e., the probability that the secondary user correctly detects activity of a primary user).

Multiple secondary users can cooperate and share sensing results, to reduce sensing time while maintaining the detection and false-alarm probabilities at the target levels. The shorter sensing time results in lower energy consumption, but longer transmission times. The sensing result obtained by each node can be shared by broadcasting it over the dedicated channel. However, there will be secondary users who contribute or choose not to contribute to cooperative spectrum sensing because they are rational. Secondary users refusing to participate in cooperative spectrum sensing will have more time for data transmission. However, if none of the secondary users performs cooperative sensing, throughput will be low because the detection probability is low and the false-alarm probability is high. This conflict situation can be analyzed using the evolutionary-game framework to find the equilibrium strategy of secondary users for participation in cooperative spectrum sensing [496].

The evolutionary game is defined as follows. The players are the secondary users (N players in total), and the strategies are to contribute or deny (refuse), denoted by C and D, respectively. The payoff for the secondary user is the throughput. For the contributing secondary user *i*, the payoff function is

$$u_{\mathsf{C},i} = P_{H_0} \left(1 - \frac{T_{\mathsf{sense}}}{|\mathcal{C}|T} \right) \left(1 - P_{\mathsf{fal}}(\mathcal{C}) \right) R_i, \tag{6.57}$$

where P_{H_0} is the probability that the spectrum is idle (i.e., a primary user is absent), C is a set of contributing secondary users (i.e., $i \in C$), |C| is the cardinality of this set, $P_{fal}(C)$ is the false-alarm probability given a set of contributing secondary users C, and R_i is the transmission rate for user *i*. For the denying secondary user *j*, the payoff function is

$$u_{\mathrm{D},j} = P_{H_0} \left(1 - P_{\mathsf{fal}}(\mathcal{C}) \right) R_j. \tag{6.58}$$

Since denying secondary users do not need to spend time on sensing, their throughput is larger. However, if there is no contributing secondary user (i.e., $C = \emptyset$), the payoff for the denying secondary users will be $u_{D,j} = 0$. With cooperative spectrum sensing, the logical-OR rule applies. That is, if any contributing secondary user detects the signal of a primary

user, the final decision will be H_1 . In this case, the detection and false-alarm probabilities, given a set of contributing users C, are $P_{det}(C) = 1 - \prod_{i \in C} (1 - P_{det,i})$ and $P_{fal}(C) = 1 - \prod_{i \in C} (1 - P_{fal,i})$, respectively. $P_{det,i}$ and $P_{fal,i}$ are, respectively, the detection and false-alarm probabilities for the individual secondary user *i*. Let x_i denote the probability of secondary user *i* selecting a contributing strategy. The replicator dynamics can be expressed as follows:

$$\dot{x}_i = \left(u_{\mathsf{C},i}(\mathbf{x}_{-i}) - \overline{u}_i(\mathbf{x})\right) x_i, \tag{6.59}$$

where **x** is the vector of x_i for all secondary users, \mathbf{x}_{-i} is the vector of the probabilities of all secondary users except user *i* contributing, and $\overline{u}_i(\cdot)$ is the average payoff for user *i*.

To simplify the equilibrium analysis of the above game, the homogeneous case is considered. That is, all secondary users are taken to be identical (i.e., with the same detection and false alarm probabilities and the same transmission rates). In this case, the respective payoff functions for contributing and denying users become

$$u_{\mathsf{C}}(J) = P_{H_0}(1 - P_{\mathsf{fal}}) \left(1 - \frac{T_{\mathsf{sense}}}{JT}\right), \tag{6.60}$$

$$u_{\rm D}(J) = P_{H_0}(1 - P_{\rm fal}),$$
 (6.61)

where J = |C|. Let x denote the probability of each secondary user contributing to cooperative sensing. The average payoff for a pure strategy of contributing is

$$\overline{u}_{\mathsf{C}} = \sum_{j=0}^{F-1} {\binom{F-1}{j} x^j (1-x)^{F-1-j} u_{\mathsf{C}}(j+1)}, \tag{6.62}$$

where *F* is the number of channels. In this case, $\binom{F-1}{j} x^j (1-x)^{F-1-j}$ is the probability that *j* users contribute to cooperative spectrum sensing. Similarly, the average payoff for the deny pure strategy is

$$\overline{u}_{\mathsf{D}} = \sum_{j=0}^{F-1} {\binom{F-1}{j} x^j (1-x)^{F-1-j} u_{\mathsf{D}}(j)}.$$
(6.63)

The replicator dynamics can be modified to

$$\dot{x} = x(1-x)(\overline{u}_{\mathsf{C}} - \overline{u}_{\mathsf{D}}). \tag{6.64}$$

At the equilibrium, we have $\dot{x} = 0$. Therefore, the ESS can be obtained by solving the following equation for x^* :

$$\frac{T_{\text{sense}}}{T}(1-x^*)^F + Fx^*(1-x^*)^{F-1} - \frac{T_{\text{sense}}}{T} = 0.$$
(6.65)

In addition to the ESS, the distributed learning algorithm to achieve Nash equilibrium in this cooperative spectrum-sensing game is presented in [496]. From the numerical

study, it is shown that as the number of secondary users and the cost of spectrum sensing increase, there is less incentive for secondary users to contribute to cooperative sensing. However, with more than two secondary users, they still gain higher average throughput than without cooperative sensing.

6.2.6 TCP throughput adaptation

When TCP is used on a wireless link, the packet loss that affects the throughput of the protocol can be from network congestion or from wireless channel error. To optimize the performance of TCP in the WiMAX network, a TCP throughput adaptation scheme based on the evolutionary game in [38] (i.e., selection of adaptive modulation and coding mode) is proposed. The network model considers one WiMAX user transmitting a group of packets (λ packets in total). This user selects adaptive modulation and coding for the packets so that the end-to-end throughput is maximized. From an evolutionary-game perspective, the population is the packets, and the strategy is the mode of modulation and coding. Let S denote a set of available modes (i.e., of strategies). The number of packets transmitted using mode s is denoted by λ_s , so $\lambda = \sum_{s \in S} \lambda_s$. The normalized number of packets using mode s is denoted by x_s , which is the population share. The corresponding vector of population share is $\mathbf{x} = [x_1 \cdots x_s \cdots x_s]^T$, where S = |S| is the total number of modulation and coding modes.

TCP throughput

The throughput of TCP in packets per second can be approximated by [62]

$$\tau \approx \min\left(\frac{1}{\mathsf{RTT}}\sqrt{\frac{3}{\mathsf{2PER}}}, \tau_{\mathsf{max}}\right),\tag{6.66}$$

where RTT is the average round-trip time, PER is the total packet error rate, and τ_{max} is the maximum TCP throughput. In this case, the packet loss can occur because of network congestion or wireless channel error. Let L_{cong} and L_{err} denote the packet loss due to network congestion and channel error, respectively. The PER is given by

$$\mathsf{PER} = 1 - (1 - L_{\mathsf{cong}})(1 - L_{\mathsf{err}}). \tag{6.67}$$

Throughput dynamics

The modulation and coding mode selection works as follows. First, a user selects random mode for the packets. Then, the throughput is measured for the sampled packets with different modes. The user uses this measured throughput to adjust the mode selection in the future. In this case, the changing of modulation and coding mode is modeled as the replicator dynamics, which depends on the throughput and the current population state **x**. Let the sampling rate for the throughput from mode *s* be denoted by r_s . The probability of switching from mode *s* to mode *s'* is denoted by $p_{s \to s'}(\mathbf{x})$, and the number of packets changing from mode *s* to mode *s'* is denoted by $x_s r_s(\mathbf{x}) p_{s \to s'}(\mathbf{x})$. With the balance of

mode switching, the differential equation for population share is

$$\dot{x}_{s} = \sum_{s' \in \mathcal{S}, s' \neq s} x_{s'} r_{s'}(\mathbf{x}) p_{s' \to s}(\mathbf{x}) - x_{s} r_{s}(\mathbf{x}) (1 - p_{s \to s}(\mathbf{x}))$$
(6.68)

$$=\sum_{s'\in\mathcal{S}}x_{s'}r_{s'}(\mathbf{x})p_{s'\to s}(\mathbf{x})-x_sr_s(\mathbf{x}),\tag{6.69}$$

which can be considered to be the replicator dynamics.

Let the throughput of packets using modulation and coding mode *s* be denoted by τ_s . The conditional probability that the packet assigned with mode *s* switches to mode *s'* is

$$\phi(\tau_{s'} - \tau_s) = \Pr(\tau_{s'} - \tau_s > 0). \tag{6.70}$$

The conditional switching probability is

$$p_{s \to s'}(\mathbf{x}) = \begin{cases} x_{s'}\phi(\tau_{s'} - \tau_s), & \text{if } s \neq s', \\ 1 - \sum_{\hat{s} \neq s, \hat{s} \in S} x_{\hat{s}}\phi(\tau_{\hat{s}} - \tau_s), & \text{if } s = s'. \end{cases}$$
(6.71)

For simplicity, $r_s(\mathbf{x}) = 1$, and the dynamics in (6.69) can be expressed as follows:

$$\dot{x}_{s} = x_{s} \sum_{s' \in \mathcal{S}, s \neq s'} x_{s'} \left(\phi(\tau_{s} - \tau_{s'}) - \phi(\tau_{s'} - \tau_{s}) \right).$$
(6.72)

It is shown in [62] that under long-term rain-fading conditions, there is a unique modulation and coding mode that maximizes TCP throughput. In addition, under fading conditions, the ESS is the strategy for transmitting a packet with a single modulation and coding mode.

Incentive protocol for peer-to-peer networks

An evolutionary game is used to model the learning process of a rational agent in peerto-peer (P2P) networks in [532]. In P2P networks, multiple users (i.e., peers) share a file by relying on other peers to perform an uploading service. Since there are many peers in the network, server overload can be avoided. However, the major issue in P2P networks is the incentive for a peer to share or upload a file, as uploading a file consumes resources and may overload the uploading peer. In a common approach [146], a peer uploads file to other peers, but when many free riders (i.e., peers performing downloading without uploading contributions) are detected, this peer may switch to a more selfish strategy. To study the adaptation of peers' strategies in a P2P network, an evolutionary-game model is introduced [532] to study the *current-best learning model* (CBLM).

In the CBLM, all peers find and switch to the current best strategy. The available strategies are: cooperate, reciprocate, and defect. In the cooperate strategy, a peer uploads data to other peers unconditionally. In the reciprocate strategy, a peer uploads data depending on the requesting peers. In the defect strategy, a peer refuses to upload any data. The set of these strategies is denoted by $S = \{C, R, D\}$. To decide on the strategy to switch to, a peer will observe the other peers. Let $s'(t) \in S$ denote the strategy that has the highest expected payoff among all strategies at the end of time slot *t*. In this case, the peer with strategy $s \in S$ will switch to s' at time t + 1 with probability $\gamma_{nor}(u_{s'}(t) - u_s(t))$, where γ_{nor} is a normalizing factor. If the peer switches to the new strategy with probability γ_{swt} , the learning rate of the peer can be defined as $\gamma = \gamma_{nor} \gamma_{swt}$. This CBLM can be implemented by using a centralized controller, selected from the nodes in the P2P network. This controller collects the payoffs for all peers, and computes the average payoff for all strategies. The average payoff is used to determine the best strategy, and informs all peers in the P2P network.

Let $x_{s'}(t)$ denote the fraction of peers using strategy s' at time t. The update of this fraction can be expressed as the following difference equations:

$$x_{s'}(t+1) = x_{s'}(t) - \gamma x_{s'}(t)(u_s(t) - u_{s'}(t)), \qquad (6.73)$$

$$x_{\mathfrak{s}}(t+1) = x_{\mathfrak{s}}(t) + \gamma \sum_{\hat{\mathfrak{s}} \in \mathcal{S}, \hat{\mathfrak{s}} \neq \mathfrak{s}} x_{\hat{\mathfrak{s}}}(u_{\mathfrak{s}}(t) - u_{\hat{\mathfrak{s}}}(t)).$$
(6.74)

These difference equations can be transformed into a differential equation in a continuous-time model (i.e., when the time step in the difference equations is small) as follows:

$$\dot{x}_{s} = \gamma \sum_{\hat{s} \neq s} x_{\hat{s}}(t) (u_{s}(t) - u_{\hat{s}}(t))$$
(6.75)

$$= \gamma \left(u_{\mathfrak{s}}(t) - \sum_{\hat{\mathfrak{s}} \in \mathcal{S}} x_{\hat{\mathfrak{s}}}(t) u_{\hat{\mathfrak{s}}}(t) \right), \qquad (6.76)$$

which is the replicator dynamics.

The payoff function can be derived based on the generosity parameter. The generosity matrix ${f G}$ is defined as

$$\mathbf{G} = \begin{bmatrix} G_{\mathbf{C} \to \mathbf{C}} & G_{\mathbf{C} \to \mathbf{R}} & G_{\mathbf{C} \to \mathbf{D}} \\ G_{\mathbf{R} \to \mathbf{C}} & G_{\mathbf{R} \to \mathbf{R}} & G_{\mathbf{R} \to \mathbf{D}} \\ G_{\mathbf{D} \to \mathbf{C}} & G_{\mathbf{D} \to \mathbf{R}} & G_{\mathbf{D} \to \mathbf{D}} \end{bmatrix},$$
(6.77)

where element $G_{s \to s'}$ is the probability that peer type *s* uploads data to peer type *s'*. In this case, a peer with type *s* gains a benefit α from uploading data to another peer with cost $\beta = 1$ (e.g., resource consumption due to data upload). $D_s(t)$, the expected amount of downloaded data in one time slot, can be obtained from

$$D_{s}(t) = \sum_{\hat{s} \in S} x_{\hat{s}}(t) G_{\hat{s} \to s}.$$
(6.78)

Then the expected amount of uploaded data can be approximated from

$$U_{s}(t) \approx \sum_{\hat{s} \in S} x_{\hat{s}} G_{s \to \hat{s}}.$$
(6.79)

The payoff for peer type *s* can be calculated from

$$u_s(t) = \alpha D_s(t) - U_s(t). \tag{6.80}$$

The generosity parameter $G_{s\to\hat{s}}$ depends on the incentive policy. Three incentive policies are considered: mirror, proportional, and linear.

Mirror incentive policy

When a reciprocative peer (i.e., s = R) receives a request for a data download, this peer evaluates the reputation of the requester. This peer will upload data with the same probability as that of the requester uploading data to other peers. As the name of the policy implies, if the requester is a cooperator, this peer will behave exactly the same as the requester. Therefore, the generosity matrix is

$$\mathbf{G}_{\text{mirror}} = \begin{bmatrix} 1 & 1 & 1 \\ 1 & G_{\text{R} \to \text{R}} & 0 \\ 0 & 0 & 0 \end{bmatrix},$$
(6.81)

where

$$G_{R\to R} = x_{C}(t) + x_{R}(t)G_{R\to R} = \frac{x_{C}(t)}{1 - x_{R}(t)}.$$
 (6.82)

Proportional incentive policy

A reciprocative peer uploads data to another peer s' with a probability equal to the requester's uploading and downloading ratio $U_{s'}(t)/D_{s'}(t)$, a ratio bounded by 1. The generosity matrix of this policy is similar to that of the mirror incentive policy except that $G_{R\to R} = U_R(t)/D_R(t)$. The expected amounts of uploaded and downloaded data are

$$U_{\mathsf{R}}(t) = x_{\mathsf{C}}(t)G_{\mathsf{R}\to\mathsf{C}} + x_{\mathsf{R}}(t)G_{\mathsf{R}\to\mathsf{R}} + x_{\mathsf{D}}(t)G_{\mathsf{R}\to\mathsf{D}}$$
(6.83)

$$= x_{\mathsf{C}}(t) + x_{\mathsf{R}}(t)G_{\mathsf{R}\to\mathsf{R}}, \qquad (6.84)$$

$$D_{\mathsf{R}}(t) = x_{\mathsf{C}}(t)G_{\mathsf{C}\to\mathsf{R}} + x_{\mathsf{R}}(t)G_{\mathsf{R}\to\mathsf{R}} + x_{\mathsf{D}}(t)G_{\mathsf{D}\to\mathsf{R}}$$
(6.85)

$$= x_{\mathsf{C}}(t) + x_{\mathsf{R}}(t)G_{\mathsf{R}\to\mathsf{R}}.$$
(6.86)

It can be concluded that $G_{R \to R} = 1$.

Linear incentive policy class

The generosity matrix of this policy class can be expressed as follows:

$$\mathbf{G}_{\text{linear}} = \begin{bmatrix} 1 & 1 & 1 \\ p_{\text{C}} & p_{\text{R}} & p_{\text{D}} \\ 0 & 0 & 0 \end{bmatrix}.$$
 (6.87)

In particular, a reciprocative peer uploads data to other peers according to the fixed parameters $p_{\rm C}$, $p_{\rm R}$, and $p_{\rm D}$, which are independent of $x_s(t)$. The proportional incentive

policy is a special case of the linear incentive policy class, while the mirror incentive policy is not.

Using the notion of replicator dynamics of CBLM, these three incentive policies can be analyzed. For example, with the mirror incentive policy, there are two equilibria, E_1 and E_2 . At equilibrium E_1 , the fraction of reciprocative peers $x_R(t)$ will be $1/\alpha$, while at equilibrium E_2 the defector will dominate (i.e., $x_D(t) = 1$). It can be shown that equilibrium E_1 is not stable. The network can change from E_1 to E_2 (e.g., when a new peer enters the network or an existing peer leaves the network). Therefore, the mirror incentive policy with CBLM is not robust, and eventually all peers will be defectors.

6.2.7 User churning behavior

The churning of mobile users from one service provider to another is expected to become a common feature when mobile users have the freedom to choose the best wireless service. This churning behavior impacts both the system and the economic aspects of wireless network design. In [366], the churning behavior of wireless service users is analyzed using the theory of evolutionary games. The wireless service under consideration is based on WLAN hotspots, where a wireless user can choose among different IEEE 802.11-based WLAN access points (APs) based on performance and/or price. In this WLAN hotspot scenario, multiple APs operate in a service area using non-overlapping channels (Fig. 6.1). These APs use the distributed coordination function (DCF)-based MAC protocol in the IEEE 802.11 standard, and the APs are operated by different service providers. In the service area, the total number of service providers is S. There are multiple wireless users in the service area, and the number of ongoing connections (i.e., active users) is random because of connection arrival and departure processes (i.e., users initiate and terminate connections randomly). We assume that the ongoing users always have data to transmit (i.e., an infinite backlog situation). Service provider s charges each user a price p_s per connection.



Fig. 6.1 Service area with churning, arriving, and departing users.

It is assumed that the users in a service area show the following behavior:

- A user tends to choose and churn to the wireless service provider that returns a higher payoff.
- Owing to the lack of information about performance by different service providers and/or inadequate information about the decisions of other users, a user has to gradually learn and change his/her decisions about choosing a particular wireless service.
- A user can make a wrong decision, choosing a wireless service provider that provides a lower payoff. This event, which is referred to as irrational churning, occurs randomly with a small probability.
- An individual user has no intention of influencing the decisions of other users in the service area.

With these behaviors, the stochastic dynamic evolutionary game with a single population can be formulated as follows [489]. The players in this game are the users in a service area, and the strategies of these users correspond to the selection of a wireless service provider. The payoff for a user choosing wireless service provider s (i.e., strategy s) is

$$u(s) = \mathscr{U}(\tau_s(n_s)) - p_s, \tag{6.88}$$

where $\mathscr{U}(\tau_s)$ is a concave utility function representing user satisfaction as a function of throughput τ_s , and p_s is the price charged by service provider s to a user. A logarithmic utility function $\mathscr{U}(\tau_s) = \log(1 + \tau_s)$ is adopted. τ_s , the MAC-layer throughput for a wireless user attached to service provider s, is given by [72]:

$$\tau_{s}(n_{s}) = \frac{P_{succ} Pack}{\overline{\Phi} + P_{succ} T_{succ} + (1 - P_{succ}) T_{coll}},$$
(6.89)

where P_{succ} is the successful transmission probability, Pack is the average packet length, $\overline{\Phi}$ is the average number of consecutive idle slots, T_{succ} is the average time the channel is sensed busy as a result of successful transmission, and T_{coll} is the average time the channel is sensed busy because of collision. This throughput is a function of the total number of users n_s attached to service provider *s*.

Stochastic dynamic evolutionary-game formulation

It is assumed that connections in the service area are initiated at an average rate of λ and that connection initiation follows a Poisson process. The connection holding time is exponentially distributed with mean $1/\mu$. Since the price impacts the decision of a user to initiate a connection, the arrival rate is a function of the average charging price $\overline{p} = \sum_{s=1}^{S} p_s/S$ and the normal price p_0 . That is, when the average charged price is higher than the normal price, the demand (i.e., connection arrival rate) decreases. Here, the normal price defines the highest price that is acceptable to the users. With this demand

function, the effective connection arrival rate is computed from [216]:

$$\tilde{\lambda} = \lambda \exp\left(-\left(\frac{\sum_{s=1}^{S} p_s / S}{p_0} - 1\right)^2\right).$$
(6.90)

A user randomly initiates a wireless connection with one of the S service providers. Therefore, the connection arrival rate for each service provider is $\lambda_s = \tilde{\lambda}/S$.

Based on these arrival and departure processes for connections, and the churning behavior of the users, the stochastic dynamic evolutionary game can be modeled as a continuous-time Markov chain. The state space of this Markov chain is $\Delta = \{(N_1, ..., N_s, ..., N_S) | 0 \le N_s \le N\}$, where N is the maximum number of ongoing connections and N_s is a random variable representing the number of ongoing connections with service provider s. Based on this state space, the transition matrix is **Q**, whose element $q_{(n_1,...,n_s,...,n_s),(n'_1,...,n'_s,...,n'_s)}$ is the transition rate from state $(n_1,...,n_s,...,n_s)$ to state $(n'_1,...,n'_s,...,n'_s)$. These elements can be obtained for the different cases as follows:

- Connection arrival: q_{(...,n_s,...),(...,n_s+1,...)} = λ_s, where the number of connections of service provider s increases by one (i.e., one connection arrives at service provider s).
- Connection departure: $q_{(...,n_s,...),(...,n_s-1,...)} = n_s \mu$, where the number of connections of service provider *s* decreases by one (i.e., one connection to service provider *s* departs).
- *Rational churning:* q_{(...,n_s,...,n_{s'},...),(...,n_s-1,...,n_{s'}+1,...)} = n_s (u(s') u(s)), for u(s') > u(s), where the number of connections to service provider s decreases by one and the number to service provider s' increases by one. In this case, the rational churning rate is proportional to the difference in payoffs from service providers s and s'.
- *Irrational churning:* q_{(...,n_s,...,n_{s'},...),(...,n_s-1,...,n_{s'}+1,...)} = n_sρ, for u(s') ≤ u(s). Here, ρ is a parameter for irrational churning, or the perturbation rate of the users who choose the wireless service provider with lower payoff.

For the case of two wireless service providers (i.e., S = 2), the transition matrix is

$$\mathbf{Q} = \begin{bmatrix} \mathbf{q}_{0,0} & \mathbf{q}_{0,1} & & \\ \mathbf{q}_{1,0} & \mathbf{q}_{1,1} & \mathbf{q}_{1,2} & & \\ \ddots & \ddots & \ddots & \\ & \mathbf{q}_{n,n-1} & \mathbf{q}_{n,n} & \mathbf{q}_{n,n+1} \\ & \ddots & \ddots & \ddots \\ & & \mathbf{q}_{N,N-1} & \mathbf{q}_{N,N} \end{bmatrix},$$
(6.91)

where each row represents the number of connections in the service area, denoted by *n*. The maximum number of ongoing connections is *N*. Each row of element $\mathbf{q}_{n,n'}$ is defined

as follows:

$$\mathbf{q}_{n,n+1} = \begin{bmatrix} q_{(n_1,0),(n_1+1,0)} & q_{(n_1,0),(n_1,1)} \\ & q_{(n_1-1,1),(n_1,0)} & q_{(n_1-1,1),(n_1-1,2)} \\ & \ddots & \ddots \end{bmatrix},$$
$$\mathbf{q}_{n,n-1} = \begin{bmatrix} q_{(n_1,0),(n_1-1,0)} & q_{(n_1-1,1),(n_1-2,1)} \\ & q_{(n_1-1,1),(n_1-1,0)} & q_{(n_1-1,1),(n_1-2,1)} & q_{(n_1-2,2),(n_1-3,2)} \\ & \ddots & \ddots & \ddots \end{bmatrix},$$
$$\mathbf{q}_{n,n} = \begin{bmatrix} q_{(n_1,0),(n_1,0)} & q_{(n_1,0),(n_1-1,1)} \\ & q_{(n_1-1,1),(n_1,0)} & q_{(n_1-1,1),(n_1-1,1)} & q_{(n_1-1,1),(n_1-2,2)} \\ & \ddots & \ddots & \ddots \end{bmatrix},$$

where n_1 and n_2 are the numbers of users choosing service providers 1 and 2, respectively. The diagonal elements of matrix **Q** are

$$q_{(n_1,n_2),(n_1,n_2)} = -\sum_{n'_1,n'_2} q_{(n_1,n_2),(n'_1,n'_2)},$$
(6.92)

for $n_1 \neq n'_1$, $n_2 \neq n'_2$, and the off-diagonal elements are zero.

The steady-state probability (i.e., vector π) of this Markov chain,

$$\boldsymbol{\pi} = \begin{bmatrix} \cdots \pi_{(\dots, n_s, \dots, n_{s'}, \dots)} \cdots \end{bmatrix}^T, \qquad (6.93)$$

can be obtained by solving $\pi^T \mathbf{Q} = \mathbf{0}$ and $\pi^T \mathbf{1} = 1$, where **0** and **1** are the vectors of zeros and ones, respectively. The evolutionary equilibrium of the stochastic dynamic evolutionary game is defined as the state at which the steady-state probability is non-zero. The average number of users in a service area choosing service provider *i* is

$$\overline{n}_{s} = \sum_{n_{s}=1}^{N} n_{s} \sum_{n_{s'} \neq n_{s}} \pi_{(\dots, n_{s}, \dots, n_{s'}, \dots)}.$$
(6.94)

Competitive and cooperative pricing

Given the churning behavior of the user, service providers can apply non-cooperative and cooperative pricing schemes to maximize their revenues. An interaction between the evolutionary game of user churning and the non-cooperative game of competitive pricing is shown in Fig. 6.2.

Non-cooperative pricing

Service providers compete with each other in terms of price to gain the highest revenue. To model this competitive situation, a non-cooperative-game framework is applied. The players of this game are the wireless service providers, who are rational to maximize



Fig. 6.2 Interaction between pricing and users' churning.

their revenue. The total number of players is the number of wireless service providers S. The strategy of each service provider is the offered price, and the payoff for each is the revenue. The revenue earned by service provider s is

$$\mathscr{R}_{s}(p_{s},\mathbf{p}_{-s}) = \overline{n}_{s}(p_{s},\mathbf{p}_{-s})p_{s}, \qquad (6.95)$$

where p_s is the price offered by service provider s, \mathbf{p}_{-s} is the vector of prices of all service providers except s, and $\overline{n}_s(p_s, \mathbf{p}_{-s})$ is the average number of users choosing service provider s, which can be obtained from the evolutionary-game model.

The solution of this price competition among the service providers is the Nash equilibrium, for which the condition is

$$\mathscr{R}_{s}(\boldsymbol{p}_{s}^{*}, \mathbf{p}_{-s}^{*}) \geq \mathscr{R}_{s}(\boldsymbol{p}_{s}, \mathbf{p}_{-s}^{*}), \qquad (6.96)$$

 $\forall s$ where p_s^* is the Nash equilibrium price of service provider *s*, and \mathbf{p}_{-s}^* is the vector of Nash equilibrium prices of all service providers except *s*. In this case, the Nash equilibrium is obtained by using the best-response function $\mathscr{B}_s(\mathbf{p}_{-s})$. The best-response function for service provider *s* is defined as

$$\mathscr{B}_{s}(\mathbf{p}_{-s}) = \arg\max_{p_{s}} \mathscr{R}_{s}(p_{s}, \mathbf{p}_{-s}).$$
(6.97)

The Nash equilibrium price for this non-cooperative game among wireless service providers is $p_s^* = \mathscr{B}_s(\mathbf{p}_{-s}^*)$. Note that the best-response functions and the Nash equilibrium can be obtained using numerical optimization methods.

Cooperative pricing

With cooperative pricing, all wireless service providers agree (i.e., collude) to choose the price so that their revenue is maximized. In this case, the total revenue is simply the sum of individual revenues for each service provider. In cooperative pricing, the optimal price is defined as

$$\mathbf{p}^* = \arg \max_{\mathbf{p}} \sum_{s=1}^{S} \mathscr{R}_s(\mathbf{p}), \qquad (6.98)$$
where \mathbf{p} is the vector of prices offered by all wireless service providers. Again, numerical methods can be used to obtain the solution.

6.2.8 Dynamic bandwidth allocation with evolutionary network selection

In [264], the bandwidth-allocation problem for different service classes in heterogeneous wireless networks is formulated as a differential game with the dynamics of network selection. In particular, a two-level game framework is developed (Fig. 6.3). The underlying dynamic service selection is modeled as an evolutionary game based on replicator dynamics. An upper bandwidth-allocation differential game is formulated to model the competition among different service providers. In this case, the service selection distribution of the underlying evolutionary game describes the state of the upper differential game.

Service area *A* in the coverage of a heterogeneous wireless environment consists of *M* access networks and N(t) active users at time *t*, as shown in Fig. 6.4. Each access network is owned by the corresponding service provider. Service provider $i \in \{1, 2, ..., M\}$ can provide K_i service classes to users to satisfy different quality-of-service (QoS) requirements. $K = \sum_{i=1}^{M} K_i$ is the total number of service classes. For differentiating









services, the Paris Metro Pricing (PMP) model [444] is used, on the basis of which the access network is partitioned into several logically separated channels, each channel for one class. $B_i(t)$ denotes the total available bandwidth from the service provider or access network *i* at time *t*. It is assumed that all users subscribed to the same service class will share the available bandwidth equally (e.g., a WiMAX base station allocates equal-size time slots to all users). The bandwidth for user *k* of service class *j* from service provider *i* at time *t* is denoted by $\tau_k^{ij}(t) = B_{ij}(t)/N_{ij}(t)$, where $B_{ij}(t)$ is the allocated bandwidth of service class *j* from service provider *i*, $N_{ij}(t)$ is the total number of users choosing service class *j* from service provider *i* at time *t*, and $\sum_{i=1}^{M} \sum_{j=1}^{K_i} N_{ij}(t) = N(t)$. Users with multi-mode terminals can choose different service classes from different service providers freely and independently, according to the perceived instantaneous utility. Users in the service area compete to select the available access networks from candidate service providers. The objective of this selection is to maximize the satisfaction (i.e., utility) from QoS performance.

Network selection as an evolutionary game

As in [367], an underlying evolutionary game is formulated to model the dynamic competition in service selection among users. This is the lower-level game in the proposed two-level game framework (Fig. 6.3). In this lower-level evolutionary-game model, the players are the N(t) active users at time t. In the context of an evolutionary game, a group of users constitutes the population. The strategies of the players are the choices of a particular service class from a certain service provider (i.e., available access network). The payoff for a player is the utility representing the QoS satisfaction level. Let $x_{ij}(t) \in [0, 1]$ denote the proportion of users in the service area choosing service class jfrom service provider i at time t. The bandwidth allocated to each proportion of users at time t is $\tau_k^{ij}(t) = \frac{B_{ij}(t)}{N(t)x_{ij}(t)}$ and the payoff for user k is

$$\mathscr{P}(\tau_k^{ij}(t)) = \alpha \tau_k^{ij}(t) = \alpha \frac{B_{ij}(t)}{N(t) x_{ij}(t)}, \qquad (6.99)$$

where α is a constant indicating the increasing rate of utility. The average payoff (utility) for the population is then

$$\overline{\mathscr{P}}(t) = \sum_{i=1}^{M} \sum_{j=1}^{K_i} x_{ij}(t) \mathscr{P}(\tau_k^{ij}(t)).$$
(6.100)

The replicator dynamics used to model the evolution process of service-selection strategy for all $i \in \{1, 2, ..., M\}, j \in \{1, 2, ..., K_i\}$ can be described using the following differential equations:

$$\frac{\partial x_{ij}(t)}{\partial t} = \dot{x}_{ij}(t) = \delta x_{ij}(t) \left(\mathscr{P}(\tau_k^{ij}(t)) - \overline{\mathscr{P}}(t) \right),$$

$$\sum_{i=1}^{M} \sum_{j=1}^{K_i} x_{ij}(t) = 1,$$
(6.101)

with initial condition $\mathbf{x}(0) = \mathbf{x}_0 \in \mathcal{X}$, where $\mathbf{x}(t) = [x_{11}(t) \cdots x_{ij}(t) \cdots x_{MK_M}(t)]^T$ is a vector describing the population state, δ is the learning rate of the population, and \mathcal{X} is the set of all possible initial states.

Bandwidth allocation as a differential game

With the dynamic service-selection behavior of users, the service providers can optimally allocate bandwidth to achieve maximum profits. Increasing the allocated bandwidth of a certain service is a natural way to improve performance and also to attract more users for this service class. However, with the limited capacity of the access network, increasing the bandwidth allocated to one service class will decrease the bandwidth allocated to other service classes, which may result in a reduced total profit for the service provider. Therefore, the differential game model for bandwidth allocation by service providers is formulated to obtain the equilibrium solution of this conflicting situation. This is the upper-level game in the proposed two-level game framework (Fig. 6.3). This differential game model takes the dynamic service selection of users into account.

Each of the *M* non-cooperative service providers competes to maximize the present value of its objective function, derived over an infinite time horizon, by controlling the bandwidth-allocation action. To achieve this, a simultaneous-play differential game is formulated as follows. The set of players is composed of all service providers of the available access networks. For a service provider as a player, the strategy is the dynamic control of the proportion of bandwidth allocated to different service classes. Specifically, the proportion of bandwidth of service provider *i* allocated to service classes *j* at time *t* is denoted by $a_{ij}(t)$. The control action of service provider *i* is denoted by the vector $\mathbf{a}_i(t) = [a_{i1}(t) \cdots a_{ij}(t) \cdots a_{iK_i}(t)]^T$ in which $a_{ij}(t) \in [0,1]$, $\sum_{j=1}^{K_i} a_{ij}(t) = 1$, and $B_{ij}(t) = B_i(t)a_{ij}(t)$ for all $t \in [0, +\infty)$. Let $\mathbf{a}_i(t)$ denote the action profile of player *i* and $\mathbf{a}_{-i}(t)$ the action profile (i.e., vector) of all players except player *i*.

The open-loop control action of the service provider is considered because of its simplicity of implementation (i.e., a centralized controller is not required), suitable for a loosely coupled, heterogeneous wireless network. In the bandwidth-allocation differential game, all service providers (i.e., players) choose their bandwidth-allocation control actions simultaneously, thereby influencing the evolution of the state of the differential game, as well as their own and their opponents' objective functions. The state of the differential game is represented by the population state $\mathbf{x}(t)$ of the underlying serviceselection game. The replicator-dynamics differential equations (6.101) describe how the current state $\mathbf{x}(t)$ and the service providers' control $\mathbf{a}_i(t)$ at time t influence the rate of change of the state at time t. For a service provider, the problem becomes optimal control subject to the constraints (e.g., state-evolution differential equations) given the control actions of other service providers. The instantaneous payoff for service provider i choosing control action $\mathbf{a}_i(t)$ is

$$\mathscr{U}_{i}(\mathbf{a}_{i}(t),\mathbf{a}_{-i}(t)) = \sum_{j=1}^{K_{i}} (p_{ij}N(t)x_{ij}(t) - c_{j}(a_{ij}(t)B_{i}(t))^{2}),$$

where c_j is a cost parameter, and p_{ij} is price charged by service provider *i* for service class *j* per user per unit of time.

In non-cooperative bandwidth allocation, for each rational service provider $i \in \{1, 2, ..., M\}$, the optimal control can be expressed as follows:

$$\max J_i(\mathbf{a}_i(t), \mathbf{a}_{-i}(t)) = \int_0^\infty e^{-\rho t} \sum_{j=1}^{K_i} (p_{ij} N(t) x_{ij}(t) - c_j (a_{ij}(t) B_i(t))^2) dt, \quad (6.102)$$

subject to

$$\dot{x}_{ij}(t) = \delta x_{ij}(t) \left(\mathscr{U} \left(\frac{B_i(t)a_{ij}(t)}{N(t)x_{ij}(t)} \right) - \overline{\mathscr{U}}(t) \right), \ \mathbf{x}(0) = \mathbf{x}_0, \tag{6.103}$$

for $i \in \{1, ..., M\}$ and $j \in \{1, ..., K_i\}$, where

$$\sum_{i=1}^{M} \sum_{j=1}^{K_{i}} x_{ij}(t) = 1, \quad x_{ij}(t) \in [0, 1],$$
$$\sum_{j=1}^{K_{i}} a_{ij}(t) = 1, \quad a_{ij}(t) \in [0, 1], t \in [0, +\infty), \quad (6.104)$$

where ρ is the discount rate of payoff for the service provider.

The Nash equilibrium concept is chosen as the solution of the above bandwidthallocation differential game. An optimal bandwidth-allocation action is defined as follows:

DEFINITION 6.1 A bandwidth-allocation control path $\mathbf{a}_i^*(t)$ is optimal for service provider *i* if the inequality condition $J_i(\mathbf{a}_i^*(t), \mathbf{a}_{-i}(t)) \ge J_i(\mathbf{a}_i(t), \mathbf{a}_{-i}(t))$ holds for all feasible control paths $\mathbf{a}_i(t)$, $\forall i$ in the non-cooperative bandwidth-allocation differential game.

According to Definition 6.1, an open-loop Nash equilibrium for the bandwidthallocation differential game is defined as follows:

DEFINITION 6.2 Denote by $\mathbf{a}_i(t)$ the open-loop bandwidth-allocation action of service provider *i*. The action profile $\mathbf{a}^* = {\mathbf{a}_i^*(t), \mathbf{a}_{-i}^*(t)}$ is an open-loop Nash equilibrium *if*, for each service provider $i \in {1, 2, ..., M}$, $\mathbf{a}_i^*(t)$ is an optimal-control path given other service providers' control actions $\mathbf{a}_{-i}^*(t)$.

To obtain the open-loop Nash equilibrium, each service provider must solve an optimal-control problem. In this case, Pontryagin's maximum principle can be used [132]. First, the definitions of the Hamiltonian function \mathcal{H} , the maximized Hamiltonian function \mathcal{H}^* , and the adjoint equation $\dot{\lambda}(t)$ for a bandwidth-allocation differential game are given. The Hamiltonian function \mathcal{H}_i of service provider *i* is

$$\mathcal{H}_{i}(\mathbf{x}(t), \boldsymbol{\gamma}_{i}(t), \mathbf{a}_{-i}(t), \lambda_{ij}(t), t)$$

$$= \sum_{j=1}^{K_{i}} \left(p_{ij} N(t) x_{ij}(t) - c_{j} (\boldsymbol{\gamma}_{ij}(t) B_{i}(t))^{2} \right)$$

$$+ \sum_{i=1}^{M} \sum_{j=1}^{K_{i}} \lambda_{ij}(t) \delta x_{ij}(t) \left(\mathscr{U} \left(\frac{B_{i}(t) a_{ij}(t)}{N(t) x_{ij}(t)} \right) - \overline{\mathscr{U}} \right), \quad (6.105)$$

where $\lambda_{ij}(t)$ is the co-state associated with $\mathbf{x}(t)$. The corresponding maximized Hamiltonian function \mathscr{H}^* is defined as

$$\mathscr{H}_{i}^{*}(\mathbf{x}(t),\lambda_{ij}(t),t) = \max\{\mathscr{H}_{i}(\mathbf{x}(t),\mathbf{a}_{i}(t),\mathbf{a}_{-i}(t),\lambda_{ij}(t),t)|\mathbf{a}_{i}(t)\in[0,1]^{K_{i}}\}.$$
(6.106)

The adjoint equation is

$$\dot{\lambda}_{ij}(t) = \rho \lambda_{ij}(t) - \frac{\partial \mathscr{H}_i^*(\mathbf{x}(t), \lambda_{ij}(t), t)}{\partial x_{ij}(t)}.$$
(6.107)

Based on the above Hamiltonian functions and the linear utility function, the following derivation can be carried out:

$$\frac{\partial \mathscr{H}_{i}(\mathbf{x}(t), \mathbf{a}_{i}(t), \mathbf{a}_{-i}(t), \lambda_{ij}(t), t)}{\partial x_{ij}(t)} = p_{ij}N(t) - \frac{\alpha\delta B(t)\lambda_{ij}(t)}{N(t)}, \quad (6.108)$$

where $B(t) = \sum_{i=1}^{M} B_i(t)$. Therefore,

$$\frac{\partial^2 \mathscr{H}_i(\mathbf{x}(t), \mathbf{a}_i(t), \mathbf{a}_{-i}(t), \lambda_{ij}(t), t)}{\partial x_{ij}(t)^2} = 0, \qquad (6.109)$$

and similarly

$$\frac{\partial^2 \mathscr{H}_i(\mathbf{x}(t), \mathbf{a}_i(t), \mathbf{a}_{-i}(t), \lambda_{ij}(t), t)}{\partial \lambda_{ij}(t) \partial x_{ij}(t)} = 0.$$
(6.110)

According to (6.109) and (6.110), the bandwidth-allocation differential game defined in (6.102)–(6.104) is a linear-state differential game. Therefore, this bandwidthallocation differential game has the property that the open-loop Nash equilibria are Markovian-perfect.

To solve for the optimal-control action, the first-order condition is defined as follows:

$$\frac{\partial \mathscr{H}_i}{\partial a_{ij}(t)} = -2c_j B_i(t)^2 a_{ij}(t) + \lambda_{ij}(t) \delta \alpha \frac{B_i(t)}{N(t)} = 0.$$
(6.111)

The solution can be expressed as

$$a_{ij}^{*}(t) = \frac{\lambda_{ij}(t)\delta\alpha}{2c_j B_i(t)N(t)}.$$
(6.112)

It can be observed that the optimal-control path is independent of the system state x(t) and only relates to the co-state $\lambda_{ij}(t)$. This co-state can be obtained by solving the adjoint equations

$$\dot{\lambda}_{ij}(t) = \rho \lambda_{ij}(t) - \frac{\partial \mathscr{H}_i^*(\mathbf{x}(t), \lambda_{ij}(t), t)}{\partial x_{ij}(t)}, \qquad (6.113)$$

where the maximized Hamiltonian function $\mathscr{H}_i^*(\mathbf{x}(t), \lambda_{ij}(t), t)$ can be obtained by substituting (6.112) into the Hamiltonian function in (6.105). Denote the solution of (6.113) by $\overline{\lambda}_{ij}(t)$. Substituting this into (6.112), the optimal bandwidth-allocation control path $a_{ij}^*(t)$ for service class *j* of service provider *i* can be obtained. Similarly, the optimalcontrol path for all service classes of all service providers can be derived. Then, the action profile $\mathbf{a}^* = \{a_{ij}^*(t) | i \in \{1, ..., M\}, j \in \{1, ..., K_i\}\}$ is obtained. Since the state space \mathcal{X} is a convex set, the solution to the state-evolution differential equation (6.103) exists and is unique [132]. Also, for all $t \in [0, \infty)$, the maximized Hamiltonian function \mathscr{H}^* is concave and continuously differentiable with respect to **x**. Therefore, the obtained action profile \mathbf{a}^* is a Nash equilibrium for the non-cooperative bandwidth-allocation differential game.

The service providers can cooperate to allocate bandwidth to service classes. In particular, they can adjust their bandwidth-control paths in a cooperative manner to maximize the social welfare in terms of aggregated profits. As in the non-cooperative case, the optimal-control problem for the cooperative bandwidth allocation can be expressed as

$$\max sJ(\mathbf{a}_{i}(t), \mathbf{a}_{-i}(t)) = \int_{0}^{\infty} e^{-\rho t} \sum_{i=1}^{M} \sum_{j=1}^{K_{i}} (p_{ij}N(t)x_{ij}(t) - c_{j}(a_{ij}(t)B_{i}(t))^{2}) dt,$$
(6.114)

with the same constraints as defined in (6.103) and (6.104).

To obtain the optimal solution of cooperative bandwidth allocation, Pontryagin's maximum principle is used. In this case, the Hamiltonian function \mathscr{H}_i^c , the maximized Hamiltonian function \mathscr{H}_i^{\star} , and the adjoint equation $\dot{\lambda}_{ij}^c(t)$ for service provider *i* for the cooperative bandwidth allocation are

$$\mathcal{H}_{i}^{c}(\mathbf{x}(t), \mathbf{a}_{i}(t), \mathbf{a}_{-i}(t), \lambda_{ij}^{c}(t), t)$$

$$= \sum_{i=1}^{M} \sum_{j=1}^{K_{i}} (p_{ij}N(t)x_{ij}(t) - c_{j}(a_{ij}(t)B_{i}(t))^{2})$$

$$+ \sum_{i=1}^{M} \sum_{j=1}^{K_{i}} \lambda_{ij}^{c}(t)\delta x_{ij}(t) \left(\mathcal{U}\left(\frac{B_{i}(t)a_{ij}(t)}{N(t)x_{ij}(t)}\right) - \overline{\mathcal{U}} \right),$$
(6.115)

$$\mathscr{H}_{i}^{\star}(\mathbf{x}(t),\lambda_{ij}^{c}(t),t) = \max\{\mathscr{H}_{i}^{c}(\mathbf{x}(t),\mathbf{a}_{i}(t),\mathbf{a}_{-i}(t),\lambda_{ij}^{c}(t),t)|\mathbf{a}_{ij}(t) \in [0,1]\}, \quad (6.116)$$

$$\dot{\lambda}_{ij}^{c}(t) = \rho \lambda_{ij}^{c}(t) - \frac{\partial \mathscr{H}_{i}^{\star}(\mathbf{x}(t), \lambda_{ij}^{c}(t), t)}{\partial x_{ii}(t)}.$$
(6.117)

According to (6.115), we can verify that the cooperative bandwidth allocation is a linear-state optimal control. With methods similar to those used in the non-cooperative case, we can obtain the cooperative optimal control $a_{ij}^{\star}(t)$ and accordingly the cooperative action profile \mathbf{a}^{\star} . Then the efficiency of non-cooperative behavior of the service providers can be evaluated. According to the first-order condition, let $\frac{\partial \mathcal{H}_i^c}{\partial a_{ij}(t)} = 0$. The optimal action can be expressed as

$$a_{ij}^{\star}(t) = \frac{\lambda_{ij}(t)\delta\alpha}{2c_j B_i(t)N(t)}.$$
(6.118)

The adjoint equation for the cooperative case is

$$\begin{aligned} \dot{\lambda}_{ij}^{c}(t) &= \rho \lambda_{ij}^{c}(t) - \frac{\partial \mathscr{H}_{i}^{\star}(\mathbf{x}(t), \lambda_{ij}^{c}(t), t)}{\partial x_{ij}(t)} \\ &= \rho \lambda_{ij}^{c}(t) + \frac{B(t)\delta\alpha\lambda_{ij}^{c}(t)}{N(t)} - p_{ij}N(t), \end{aligned}$$
(6.119)

which is the same as the adjoint equation for the non-cooperative case. Accordingly, the co-state is given by $\lambda_{ij}^c(t) = \lambda_{ij}(t)$, for all $i \in \{1, 2, ..., M\}, j \in \{1, 2, ..., K_i\}$. Thus, we obtain $a_{ij}^*(t) = a_{ij}^*(t)$, which shows that the selfish behavior of service providers can also maximize the social welfare.

The dynamic behavior of service selection by users under the bandwidth-allocation control is shown in Fig. 6.5 for two SPs and two service classes. The distribution of users



Fig. 6.5 Dynamics of service selection. "SP" denotes a service provider.

and

selecting different service classes from different providers converges to a distribution in which every user in the service area receives the same utility as the average utility for the population.

6.3 Summary

In this chapter, the basics of evolutionary games have been presented. The evolutionarygame framework has some advantages over the classical non-cooperative game, among which are equilibrium selection, bounded rationality, and dynamic behavior of players. There are two approaches in the evolutionary-game framework, namely the static and the dynamic models. For the static case, the solution concept in terms of evolutionarily stable strategies (ESS) has been considered. For the dynamic model, the replicator dynamics have been used to model the adaptation of the strategies of the players. Evolutionary equilibrium has been considered as the solution of this dynamic model of the evolutionary game. Since an evolutionary game can model the decisions of a number of players in a population, it can be applied to address various issues in different protocol layers of wireless communications and networking. Some of these applications have been discussed in this chapter, including congestion control, contention-based (i.e., Aloha) protocol adaptation, power control in CDMA, routing, cooperative sensing in cognitive radio, TCP throughput adaptation, and service-provider selection. While non-cooperative game theory studies the strategic choices resulting from the interactions among competing players, cooperative game theory provides analytical tools to study the behavior of rational players when they cooperate. In this context, in a cooperative-game scenario, the players are allowed to form agreements among themselves that can impact the strategic choices of these players as well as their utilities. Cooperative games encompass two main branches: bargaining theory and coalitional games. The former describes the bargaining process between a set of players that need to agree on the terms of cooperation, while the latter describes the formation of cooperating groups of players, referred to as coalitions, that can strengthen the players' positions in a game. In this chapter, we examine the key characteristics, properties, and solution concepts of both branches of cooperative games as well as sample applications within wireless and communication networks.

7.1 Bargaining theory

7.1.1 Introduction

In economics, many problems involve a number of entities that are interested in reaching an agreement over a trade or the sharing of a resource but have a conflicting interest on how to reach this agreement and on the terms of the agreement. In this context, a *bargaining situation* is defined as a situation in which two (or more) players can mutually benefit from reaching a certain agreement but have conflicting interests on the terms of the agreement. Certainly, in a bargaining situation, no agreement can be imposed on any player without the player's approval. *Bargaining theory* is an established field that deals with studying and analyzing bargaining situations in a variety of problems. The foundations of bargaining theory were first laid by the work of Nash in the 1950s [350, 352]; however, the field only began to blossom following the seminal work of Ariel Rubinstein in [405]. In fact, the contributions of Rubinstein constitute the origin of most of the work that followed, so it lies at the heart of the development of the *theory* of bargaining.

The motivation and ever-growing interest in developing models and theories for analyzing bargaining situations stems from the prevalence of such situations in many disciplines. In economics, the simplest example of a bargaining situation is a trade between a buyer and a seller. An example would be a seller who owns a painting that he values at \$1,000 and who wishes to sell this painting to a buyer who values it at \$1,500. Clearly, in this situation the seller and the buyer have an incentive to agree on completing the transaction, but at the same time they have conflicting interests regarding the price at which to trade: the seller would like to trade at a high price, while the buyer would like to trade at a low price. In wireless and communication networks, an example of a bargaining situation is the sharing of a resource (e.g., time, bandwidth, rate) between two or more nodes. While the nodes have an interest in agreeing on a division of the resource (e.g., in order to communicate), they have conflicting interests on the share: each node is interested in maximizing the amount of resource that it receives.

While bargaining theory involves a broad range of challenging issues, one can identify the following four key issues that are of central concern for studying any bargaining problem:

- Efficiency. A central issue in bargaining theory is to identify a unique, mutually beneficial agreement. From the set of all possible outcomes of a bargaining situation, the bargainers need to identify the set of *jointly efficient outcomes*.
- **Distribution.** Once the efficient bargaining outcomes are identified, the key issue is to select one of the outcomes as a solution to the considered problem.
- **Strategy coordination.** Once the chosen outcome is identified, the bargainers must identify the joint strategy that they can use to reach this outcome.
- Agreement enforcement. Once the joint strategy is found, the terms of the agreement must be identified so as to enforce the implementation of the strategies by the bargainers.

In order to develop a theoretical analysis of these four key issues of bargaining, one has to use solid mathematical and analytical frameworks. In this context, bargaining theory is strongly linked to game theory. For instance, a significant portion of cooperative games, which are the main focus of this chapter, deals mainly with efficient distribution of resources or utilities, and is thus a suitable framework for tackling the efficiency and distribution issues of bargaining. In addition, non-cooperative game theory, notably dynamic games and evolutionary games (which are treated in Chapters 3 and 6), deals with the coordination of strategies and the enforcement of agreements.

In this section, we mainly focus on bargaining theory as it relates to cooperative games through the concept of Nash bargaining. We start by introducing the central concept of Nash bargaining, and then discuss a few representative applications of bargaining theory in wireless and communication networks.

7.1.2 The Nash bargaining solution

In bargaining theory, we distinguish between two key concepts: the bargaining process and the bargaining outcome. The bargaining process is the procedure that the involved bargainers must follow in order to reach an agreement over a certain bargaining outcome. The bargaining outcome is, thus, the result of the bargaining process. In his seminal paper, Nash [350] adopted an axiomatic approach that abstracts the bargaining process and considers only the *bargaining outcome*. Hence, instead of studying *how* the bargainers can reach an agreement, Nash looked at the possible outcomes or agreements that satisfy "reasonable" properties identified by a set of axioms that he defined.

In this context, consider two players, labeled i = 1, 2, that are trying to come to an agreement over an outcome in a space \mathcal{X} . Each player *i* has a utility function u_i defined over the space $\mathcal{X} \cup \{D\}$, where *D* is the outcome if the two players fail to reach an agreement, i.e., the *disagreement outcome*. Define the space \mathcal{S} as the set of all possible utilities that the two players can achieve, i.e.,

$$S = \{ (u_1(x_1), u_2(x_2)) | x = (x_1, x_2) \in \mathcal{X} \}.$$
(7.1)

Furthermore, we define the pair $d = (d_1, d_2)$ with $d_1 = u_1(D)$ and $d_2 = u_2(D)$ as the *disagreement point* or the *threat point*.

Subsequently, a *bargaining problem* is defined as the pair (S, d) where $S \subset \mathbb{R}^2$ and $d \in S$ such that

- S is a convex and compact set.
- There exists some s ∈ S such that s > d, i.e., s₁ > d₁ and s₂ > d₂ (note that, in some references, this property is considered as a *feasibility axiom* rather than part of the definition of a bargaining problem).

We are interested in a *bargaining solution* that is a function f that specifies a unique outcome $f(S, d) \in S$ for every bargaining problem (S, d) ($f_i(S, d)$ would represent the component of player i in the bargaining outcome). As previously mentioned, instead of looking at the bargaining process, Nash stated the following four axioms, specifying properties that the bargaining outcome (or solution) must satisfy:

- Pareto efficiency: A bargaining solution f(S, d) is Pareto-efficient if there does not exist a point (s₁, s₂) ∈ S such that s ≥ f(S, d) and s_i > f_i(S, d) for some i. Any reasonable bargaining scheme must choose a Pareto-efficient outcome since, otherwise, there would exist another outcome which is better for both players.
- 2. Symmetry: Let (S, d) be symmetric around $s_1 = s_2$, i.e., $(s_1, s_2) \in S$ if and only if $(s_2, s_1) \in S$ and $d_1 = d_2$; then $f_1(S, d) = f_2(S, d)$. In other words, the bargaining solution would not discriminate among the players if these players were indistinguishable.
- 3. *Invariance to equivalent utility representation:* If we transform a bargaining problem (S, d) into another different bargaining problem (S', d') by taking $s'_i = \alpha_i s_i + \beta_i$ and $d'_i = \alpha_i d_i + \beta_i$ where $\alpha_i > 0$, then $f_i(S', d') = \alpha_i f_i(S, d) + \beta_i$.
- 4. *Independence of irrelevant alternatives:* Given two bargaining problems (S, d) and (S', d) such that $S' \subseteq S$, if $f(S, d) \in S'$ then f(S', d) = f(S, d). This axiom states that, if bargaining in the utility region S results in a solution f(S, d) that lies in a subset S' of S, then a hypothetical bargaining in the smaller region S' would result in the same outcome.

Nash's result shows that there exists a unique bargaining solution that satisfies the four axioms, which we state through the following theorem:

THEOREM 7.1 There exists a unique solution satisfying the four axioms, and this solution is the pair of utilities (s_1^*, s_2^*) that solves the following optimization problem:

$$\max_{(s_1,s_2)} (s_1 - d_1)(s_2 - d_2), \ s.t. \ (s_1, s_2) \in \mathcal{S}, \ \ (s_1, s_2) \ge (d_1, d_2), \tag{7.2}$$

where $(s_1 - d_1)(s_2 - d_2)$ is known as the Nash product and the solution of the optimization problem (7.2) is the Nash bargaining solution.

Proof For the purpose of the proof, we first note that the optimization problem (7.2) admits a unique optimal solution since: (i) S is a compact set, and (ii) the objective function is continuous and strictly quasi-concave. In order to prove the existence and uniqueness of a solution that solves (7.2) and satisfies the four axioms (i.e., the Nash bargaining solution), we proceed in two steps. In the first step, we prove that the Nash bargaining solution (i.e., the solution to (7.2)) satisfies the four axioms. Then we show that if a bargaining solution satisfies the four axioms it is a Nash bargaining solution.

Step 1:

- 1. Pareto efficiency: Since the objective function of (7.2) is increasing in s_1 and s_2 , the solution of (7.2) is, indeed, Pareto-optimal.
- Symmetry: Assume that d₁ = d₂ and let s* = (s₁*, s₂*) be the Nash bargaining solution of (7.2). Then we easily see that (s₂*, s₁*) would also be an optimal solution of (7.2). Because of the uniqueness of the optimal solution of (7.2), we must have s₂* = s₁*; thus, the Nash bargaining solution satisfies the symmetry axiom.
- 3. Invariance to equivalent utility representation: Consider the Nash bargaining outcome $f^*(S', d')$ of the transformed bargaining problem (S', d'); by definition, this outcome is the optimal solution of the problem

$$\max_{(s_1, s_2)} (s_1 - \alpha_1 d_1 - \beta_1) (s_2 - \alpha_2 d_2 - \beta_2), \ s.t. \ (s_1, s_2) \in \mathcal{S}'.$$
(7.3)

Performing the change of variables $s'_1 = \alpha_1 s_1 + \beta_1$, $s'_2 = \alpha_2 s_2 + \beta_2$, it follows immediately that the Nash bargaining solutions of (S, d) and its transformed problem (S', d') satisfy $f_i^*(S', d') = \alpha_i f_i^*(S, d) + \beta_i$ for i = 1, 2.

4. Independence of irrelevant alternatives: Let $S' \subseteq S$. From (7.2), it is clear that the objective function value at the Nash bargaining solution $f^*(S, d)$ is greater than or equal to the one at $f^*(S', d)$. If $f^*(S, d) \in S'$, then the objective function values must be equal; thus, $f^*(S, d)$ is optimal for S'. By the uniqueness of the solution, we will have $f^*(S, d) = f^*(S', d)$.

Step 2:

Consider a bargaining solution f(S, d) satisfying the four axioms, and let us prove that this solution is, indeed, the Nash bargaining solution $f^*(S, d)$ of (7.2). Let $t = f^*(S, d)$

denote the Nash bargaining solution of (7.2), and define the set

$$\mathcal{S}' = \{ \alpha' \mathbf{s} + \beta | \mathbf{s} \in \mathcal{S}; \alpha' \mathbf{t} + \beta = \left(\frac{1}{2}, \frac{1}{2}\right)'; \alpha' \mathbf{d} + \beta = (\mathbf{0}, \mathbf{0})' \}.$$
(7.4)

In other words, we map the point t to $(\frac{1}{2}, \frac{1}{2})$ and the disagreement point d to (0,0). Since f(S,d) and $f^*(S,d)$ both satisfy the axiom of invariance to equivalent utility representations (Axiom 3), then we have $f(S,d) = f^*(S,d)$ if and only if f(S',0) = $f^*(S',0) = (\frac{1}{2},\frac{1}{2})$. Thus, to conclude our proof, it is sufficient to prove that f(S',0) = $(\frac{1}{2},\frac{1}{2})$.

For this purpose, we first show that there exists no $s \in S'$ such that $s_1 + s_2 > 1$. Assume that such a point s exists. Then let $z = (1 - \lambda)(\frac{1}{2}, \frac{1}{2}) + \lambda(s_1, s_2)$ for some $\lambda \in (0, 1)$. Since S' is convex, then $z \in S'$. Although we can choose λ sufficiently small such that $z_1z_2 > \frac{1}{4} = f^*(S', 0)$, this contradicts the optimality of $f^*(S', 0)$, which shows that for all $s \in S'$ we have $s_1 + s_2 \leq 1$. Then, since S' is bounded, we can find a rectangle \mathcal{R} that is symmetric around the line $s_1 = s_2$ and such that $S' \subseteq \mathcal{R}$ and $(\frac{1}{2}, \frac{1}{2})$ lies on the boundary of \mathcal{R} .

Consequently, by Axioms 1 and 2, we have $f(\mathcal{R}, 0) = (\frac{1}{2}, \frac{1}{2})$, and, by Axiom 4, since $S' \subseteq \mathcal{R}$, we have $f(S', 0) = (\frac{1}{2}, \frac{1}{2})$, which completes the proof.

In consequence, the Nash bargaining solution provides a unique outcome of a bargaining process that satisfies a set of desired properties as conveyed by the axioms of Nash. In order to give a better idea of this solution, consider the following two classic examples (detailed analyses of these examples can be found in [345] and references therein).

Example 7.1 Two players are negotiating over sharing a pie of size 1. Let x be the share that player 1 will receive, so the share obtained by player 2 will be 1 - x. Denote by $u_1(x)$ and $u_2(x)$ the utilities (assumed to be concave) of players 1 and 2, respectively. Clearly, u_1 is increasing in x while u_2 is decreasing in x, since x is the share of player 1. For this example, the Nash bargaining solution is the share x^* that maximizes the Nash product $(u_1(x) - d_1)(u_2(x) - d_2)$, where $x \in [0, 1]$ and (d_1, d_2) are the utilities at the disagreement point. If we choose $u_1(x) = x$ and $u_2(x) = 1 - x$ with $d_1 = d_2$, then the Nash bargaining solution dictates that the two players would split the pie equally, i.e., $x^* = \frac{1}{2}$.

Example 7.2 Consider two men, one rich and one poor, who run into a genie on the street. The genie offers to let them share \$100 on the condition that they agree on how to split the money.¹ This problem highlights a common bargaining situation in which two individuals have an incentive to cooperate but are negotiating on how to cooperate.

¹ A variant of this problem considers the case in which the rich man and the poor man need to share the payment of a debt to a bank.

One solution to this problem is through the Nash bargaining concept. We assume that the rich man initially has a wealth of $w_1 = \$10^{10}$ while the poor man has initially a wealth of $w_2 =$ \$10. In order to formulate the problem as a bargaining problem, we first need to define the space S and the disagreement point d. Consider logarithmic utilities, and choose the disagreement point as the log of the initial wealth of the two men, i.e., $d_1 = \log w_1$ and $d_2 = \log w_2$. This choice is natural since, if the two men do not agree on the division of the 100, then, the genie will not give it to them. If we let x denote the share of the money that the rich man gets, then the utility for the rich man would be $u_1 = \log(10^{10} + x)$ and that of the poor man $u_2 = \log(10 + (100 - x))$. By solving the maximization of the Nash product $(u_1 - d_1)(u_2 - d_2)$ over $x \in [0, 100]$, the Nash bargaining solution yields $x^* \approx$ \$66. This solution can be found graphically by intersecting the Pareto boundary of the utility region (i.e., the boundary of the utility region which is Pareto-efficient) with the hyperbola parameterized by a constant m such that $(u_1 - d_1)(u_2 - d_2) = m$. To find the Nash bargaining solution, the parameter m is chosen so as to have a single intersection point with the Pareto boundary. The Nash bargaining solution for this problem favors the rich man, i.e., the player who had a larger initial wealth, and thus greater bargaining power. The reason is that, from the rich man's perspective, the loss incurred from staying at the disagreement is not as big as the loss of the poor man (owing to the relative importance of 100 with respect to each man's initial wealth). As a matter of fact, in the case where w_1 and w_2 are comparable, the Nash bargaining solution goes toward equal division.

One important aspect of Nash bargaining is the selection of the disagreement point. In general, this selection is dependent on the application, as some scenarios have a natural value for this point. For example, in Example 7.2, it is quite intuitive to select the initial wealth of the two men as their disagreement point. In many other cases, the disagreement point can be simply chosen as the origin. Nonetheless, one rule of thumb in the selection of the disagreement point is to select a point that the players can improve upon by bargaining. For instance, in many applications, it is of interest to select the Nash equilibrium as the disagreement point and, subsequently, use the Nash bargaining solution to improve the utilities of the players with respect to their non-cooperative payoffs.

While the Nash bargaining solution was initially derived for the two-player case, it can be extended to the general *N*-player case, N > 2, by expanding *S* to become an *N*-dimensional utility space. Hence, the Nash bargaining solution for an *N*-player bargaining problem (S, d), with $d = (d_1, ..., d_N)$ being the disagreement point, becomes the unique solution of the following optimization problem:

$$\max_{(s_1,\ldots,s_N)} \prod_{i=1}^N (s_i - d_i), \text{ s.t. } (s_1,\ldots,s_N) \in \mathcal{S}, \quad (s_1,\ldots,s_N) \ge (d_1,\ldots,d_N).$$
(7.5)

Certainly, this solution satisfies the four axioms of Nash, in the *N*-dimensional space.

Although the extension of the Nash bargaining solution concept to the *N*-player case is straightforward, the complexity of finding the solution, i.e., solving the optimization problem in (7.5), becomes higher. While in the two-player case a graphical method can be used, e.g., using the intersection of the Pareto boundary of the utility region S with the hyperbola given by the Nash product parameterized by a constant *m*, in the *N*-player case advanced optimization techniques may be required. In general, when approaching an *N*-player Nash bargaining solution, depending on the structure of the problem and the properties of the considered utility functions, one can utilize the well-known optimization methods in [81], such as the bisection method, to find the solution.

In a given *N*-player bargaining problem (S, d) with $N \ge 2$, it is sometimes useful to give some additional weight, i.e., power, to some of the players. For example, some large companies might need to be assured of a large share of the pie (in a bargaining situation) before they indulge in the bargaining process. In order to do so, one can assign, for every player *i*, a value $\alpha_i \in [0, 1]$ which represents the *bargaining power* of *i*. The bargaining-power values are chosen such that $\sum_{i=1}^{N} \alpha_i = 1$. The idea of bargaining power allows us to give some weight to the negotiation capabilities of every player. A player having a higher bargaining power would thus be a candidate to obtain an advantage in the final outcome of the bargaining process. In such a bargaining problem, if we drop the symmetry axiom of Nash (Axiom 2), we can define the *generalized Nash bargaining solution*, which is the solution of the following optimization problem:

$$\max_{(s_1,\ldots,s_N)} \prod_{i=1}^N (s_i - d_i)^{\alpha_i},$$
(7.6)

with the same constraints as (7.5). Clearly, as seen in (7.6), the use of bargaining power allows the maximization to become more biased towards the player having a higher bargaining power α_i . Furthermore, one can see that, whenever the weights α_i are equal for all i = 1, ..., N, then (7.6) reduces back to the standard Nash bargaining solution of (7.5). In the pie-splitting Example 7.1, if we assign a bargaining power α to the first player, then we can use the first-order conditions to solve (7.6) and obtain the generalized Nash bargaining solution. In this case, the solution dictates that $x^* = d_1 + \alpha(1 - d_1 - d_2)$ and $1 - x^* = d_1 + (1 - \alpha)(1 - d_1 - d_2)$. Hence, in this example, each player will obtain his disagreement point *plus* a certain share of the extra utility $1 - d_1 - d_2$, i.e., the total utility available in surplus of the utility at the disagreement point.

Although the Nash bargaining solution is a widely popular concept for solving bargaining problems, it is worth noting that it is by no means the *only* approach for doing so. As a matter of fact, the Nash bargaining solution suffers from several drawbacks that have led to the emergence of other concepts or solutions for bargaining situations. One drawback of the Nash bargaining solution is that it requires convexity of the utility space, which can be quite restrictive in many applications. For overcoming this restriction, some extensions of the Nash bargaining solution to other regions, such as log convex regions [76], have been proposed. Moreover, ever since the introduction of the Nash bargaining solution, many objections have been raised with regard to the "independence of irrelevant alternatives" axiom (Axiom 4) [305, 237]. An important criticism of Axiom 4 pertains to its claim that expanding the utility region in a direction favorable to one of the players does not yield any extra benefit for this player in the bargaining process. In many scenarios, this restriction has been deemed unfair since it does not provide any advantage to the player when its bargaining power in the utility region (maximum utility) increases. To overcome this problem, a number of alternative bargaining solutions have been proposed [305, 237, 345]. One popular alternative is the Kalai–Smorodinsky solution, which was first introduced in [237]. The Kalai–Smorodinsky approach suggests replacing Axiom 4 with a montonicity axiom. This new monotonicity axiom states that if, for every utility level that player 1 may demand, the maximum feasible utility level that player 2 can simultaneously reach is increased, then, in the bargaining outcome, the share of the utility for player 2 would also increase. In other words, the Kalai–Smorodinsky solution of a bargaining problem (S, d) requires that whenever the maximum feasible utility achievable by one of the players is increased, i.e., S is expanded in a direction favorable to one of the players, this player obtains a larger utility at the bargaining outcome.

Beyond the shortcomings of the axioms and the mathematical properties, one drawback of the Nash bargaining solution is, as previously mentioned, the fact that it is concerned *solely* with the outcome of the bargaining situation, while ignoring completely the details of the bargaining process that can achieve this outcome. This implies that many questions such as "how do the players negotiate so as to reach the Nash bargaining solution?" are left unanswered by Nash's approach. In this regard, important aspects of bargaining such as the delay in reaching the agreement and the cost of haggling and negotiation are abstracted and somewhat ignored by the Nash solution. In this context, dynamic bargaining is a major branch of bargaining theory that deals with the bargaining process and its connection to the outcome of a given bargaining situation. One of the first approaches to modeling the bargaining process was presented by Rubinstein in [405] for the case of two players negotiating over shares of a pie of size 1 (a scenario somewhat similar to Example 7.1). The Rubinstein process is based on an offer/counteroffer procedure, and, through concepts from non-cooperative games, Rubinstein shows that there exists a unique outcome to the bargaining process and, whenever no delay for negotiations is taken into account, this outcome is the Nash bargaining solution.

An in-depth analysis of the alternatives to the Nash bargaining solution, as well as dynamic bargaining approaches, are beyond the scope of this chapter, since we are mainly interested in the aspects of bargaining theory that fall under the umbrella of cooperative-game theory, i.e., the Nash bargaining solution. However, interested readers are referred to [345] for an in-depth treatment of the field of bargaining theory which, like game theory, admits numerous applications in the context of wireless and communication networks.

7.1.3 Sample applications in wireless and communication networks

In the context of wireless and communication networks, the Nash bargaining solution turns out to be very useful for characterizing the outcome of many bargaining situations, notably in the area of resource allocation and management. As a demonstration of the strong connection between Nash bargaining and communication networks, it has been shown that, whenever the disagreement point *d* is chosen as the origin, i.e., d = (0, 0), the Nash bargaining solution, when applied to resource allocation in a wireless or communication network, coincides with the famous *proportional fair* distribution of resources (see [76] and references therein), which was introduced by Kelly [242]. Hence, in this subsection, we discuss several applications of the Nash bargaining solution in wireless and communication networks.

Nash bargaining for downlink beamforming in an interference channel

Enabling wireless systems to efficiently operate in the same spectral band is a key challenge for next-generation networks. An illustrative example of this problem is the scenario whereby two base stations (BS) equipped with multiple antennas are trying to transmit their information, in the downlink, to two mobile stations (MS). In this scenario, considering that both BSs utilize the same spectrum, whenever the communication occurs simultaneously, the received signal at each MS will suffer from the interference caused by the other transmission. The scenario is depicted in Fig. 7.1 and studied in detail in [281].

Formally, consider two BSs, each equipped with *k* antennas and belonging to *different operators*, with BS 1 transmitting data to MS 1 and BS 2 transmitting data to MS 2. By considering that each BS performs single-stream transmission, i.e., scalar coding followed by beamforming (which is optimal under certain conditions on the channel knowledge and interference modeling [180]) and given that all channels are frequency-flat, we have the following complex baseband data symbols y_1 and y_2 received at MS 1





and MS 2, respectively:

$$y_1 = \mathbf{h}_{11}^T \mathbf{w}_1 s_1 + \mathbf{h}_{21}^T \mathbf{w}_2 s_2 + n_1, \tag{7.7}$$

$$y_2 = \mathbf{h}_{22}^T \mathbf{w}_2 s_2 + \mathbf{h}_{12}^T \mathbf{w}_1 s_1 + n_2, \qquad (7.8)$$

where s_1 and s_2 represent the transmitted symbols, \mathbf{h}_{ij} represents the $k \times 1$ channel-vector between BS *i* and MS *j*, \mathbf{w}_i is the $k \times 1$ beamforming vector used by BS *i*, and n_1 , n_2 are the noise terms modeled as i.i.d. (independently and identically distributed) Gaussian with zero mean and variance σ^2 . The maximum transmit power per BS is normalized to 1, which yields the following power constraint on each BS *i*: $||\mathbf{w}_i||^2 \leq 1$.

In this scenario, we highlight an important conflict of interest that arises between the two BSs. While every BS *i* has an incentive to optimize its weight vector \mathbf{w}_i so as to maximize the quality of service received by its served MS, this optimization yields increased interference in the transmission of BS *j*, thus requiring BS *j* to change its weight vector, which in turn would affect the optimization of BS *i*. This cycle continues, and two key questions need to be answered:

- If the BSs act in a non-cooperative manner, with each BS being self-interested, what are the resulting beamforming vectors **w**₁ and **w**₂?
- Can some sort of cooperation improve the performance of the BSs?

In order to answer the second question, one must first assess the outcome of the noncooperative situation, i.e., the first question. To tackle the first question, we revert to a concept from non-cooperative non-zero-sum games. In this scenario, we formulate a non-cooperative game with the following components:

- The players are the two BSs.
- The strategy of each BS *i* is the choice of a beamforming vector \mathbf{w}_i such that $||\mathbf{w}_i||^2 \le 1$.
- The utility for each BS *i* is simply the rate that it achieves at the MS.

Considering no interference-cancellation techniques at the receivers, for a given pair of beamforming vectors ($\mathbf{w}_1, \mathbf{w}_2$), expressions for the rates R_1 and R_2 achievable, respectively, by BS 1 and BS 2 (using codebooks approaching Gaussian) are given by

$$R_1 = \log_2\left(1 + \frac{|\mathbf{w}_1^T \mathbf{h}_{11}|^2}{\sigma^2 + |\mathbf{w}_2^T \mathbf{h}_{21}|^2}\right),\tag{7.9}$$

$$R_2 = \log_2 \left(1 + \frac{|\mathbf{w}_2^T \mathbf{h}_{22}|^2}{\sigma^2 + |\mathbf{w}_1^T \mathbf{h}_{12}|^2} \right).$$
(7.10)

Consequently, the utilities of BSs 1 and 2 can be defined simply as $u_1(\mathbf{w}_1, \mathbf{w}_2) = R_1$ and $u_1(\mathbf{w}_1, \mathbf{w}_2) = R_2$. As the utilities depend on the strategies of the competing players, we have a non-cooperative game among the BSs.

In order to specify the outcome of this game, we can use the notion of the Nash equilibrium as defined in Chapter 3. Nevertheless, prior to determining the Nash equilibrium of this game, we first define an achievable-rate region for the considered multiple-input single-output (MISO) interference channel. Although, in general, the capacity region of the MISO interference channel is not known, for the purpose of studying this problem, we can use the following achievable-rate region [281]:

$$\mathcal{R} = \bigcup_{\mathbf{w}_1, \mathbf{w}_2, ||\mathbf{w}_1||^2 \le 1} (R_1, R_2).$$
(7.11)

In the absence of coordination among the BSs, the outcome of the game will generally be the Nash equilibrium. In this regard, a pair of strategies $(\mathbf{w}_1^{\text{NE}}, \mathbf{w}_2^{\text{NE}})$ at the Nash equilibrium must satisfy

$$u_1(\mathbf{w}_1^{\text{NE}}, \mathbf{w}_2^{\text{NE}}) \ge u_1(\mathbf{w}_1, \mathbf{w}_2^{\text{NE}})$$

$$(7.12)$$

for BS 1 and for all \mathbf{w}_1 , $||\mathbf{w}_1||^2 \le 1$, as well as

$$u_2(\mathbf{w}_1^{\text{NE}}, \mathbf{w}_2^{\text{NE}}) \ge u_2(\mathbf{w}_1^{\text{NE}}, \mathbf{w}_2)$$

$$(7.13)$$

for BS 2 and for all \mathbf{w}_2 , $||\mathbf{w}_2||^2 \leq 1$. By substituting (7.9) and (7.10) into (7.12) and (7.13), it is shown that, for this game, there exists a *unique* Nash equilibrium in pure strategies determined by the maximum-ratio transmission beamforming vectors [281]:

$$\mathbf{w}_{1}^{\text{NE}} = \frac{\mathbf{h}_{11}^{\dagger}}{||\mathbf{h}_{11}||}, \mathbf{w}_{2}^{\text{NE}} = \frac{\mathbf{h}_{22}^{\dagger}}{||\mathbf{h}_{22}||},$$
(7.14)

where $\mathbf{h}_{ij}^{\dagger}$ is the complex conjugate. This result follows immediately from the fact that when BS *i* uses the beamforming vector \mathbf{w}_i^{NE} at the Nash equilibrium, there exists no other vector that can yield a larger rate while satisfying the power constraint.

Although the Nash equilibrium is a natural outcome of the considered scenario, as mentioned in previous chapters the Nash equilibrium point in many cases is not necessarily Pareto-efficient. As a matter of fact, in the studied scenario, it can be shown that the Nash equilibrium does not lie at the Pareto boundary of \mathcal{R} . Moreover, at high SNR the Nash equilibrium yields a poor performance for both BSs [281, Proposition 4].

Because of the inefficiency of the Nash equilibrium, a better solution must be sought. The inefficiency of the Nash equilibrium solution is partially due to the uncoordinated actions of the BSs. For instance, one can remark that, by a small exchange of information (without any need for a centralized controller), the BSs might be able to coordinate their beamforming vectors so as to improve upon their non-cooperative performance. However, each BS is self-interested, and would seek to maximize its own individual rate. Hence, we have a bargaining situation between the two BSs whereby they would certainly benefit from cooperation but they need to agree on a way to cooperate, i.e., on a point in the achievable-rate region. Thus, we formulate a Nash bargaining problem between the two BSs whereby

- The achievable-rate region \mathcal{R} is compact but can be non-convex; thus, in order to define a bargaining problem, we consider the convex hull \mathcal{S} of the achievable-rate region \mathcal{R} as the bargaining utility region.²
- The disagreement point is chosen as the Nash equilibrium point $d = (u_1(\mathbf{w}_1^{\text{NE}}, \mathbf{w}_2^{\text{NE}}), u_2(\mathbf{w}_1^{\text{NE}}, \mathbf{w}_2^{\text{NE}})) \in S.$

Hence, the outcome of the Nash bargaining problem described above would be a pair of utilities (i.e., rates) in the rate region which improve upon the disagreement point (the Nash equilibrium) and which can be mapped onto a pair of beamforming vectors constituting the Nash bargaining solution. Given the beamforming Nash bargaining problem (S, d) previously defined, the Nash bargaining solution is a point on the Pareto boundary that corresponds to the solution of the following optimization problem (recall that the utility is the rate):

$$\max_{(R_1,R_2)\in\mathcal{S}^+,} (R_1 - R_1^{\rm NE})(R_2 - R_2^{\rm NE}), \text{ s.t. } (R_1,R_2) \ge (R_1^{\rm NE}, R_2^{\rm NE}).$$
(7.15)

In order to find the Nash bargaining solution, (7.15) can be solved graphically by intersecting the Pareto boundary of S with a hyperbola $(R_1 - R_1^{\text{NE}})(R_2 - R_2^{\text{NE}}) = m$, where *m* is a parameter chosen in such a way as to have a single intersection with the Pareto boundary. By finding the Nash bargaining solution for the problem (S, d), one would find a point in the rate region that outperforms the Nash equilibrium and is Pareto-optimal. For example, given channels that are randomly chosen so as to have a convex rate region with a signal-to-noise ratio of 0 dB, the Nash equilibrium solution yields, approximately, the rates (1.05, 0.19) (in bits per channel use), while the Nash bargaining solution outperforms the Nash equilibrium in any given scenario, and, in some cases of this model, this solution is as good as the sum-rate maximizing centralized solution, as shown through extensive numerical simulations in [281].

While in this example we have focused on the case of the MISO interference channel, the use of the Nash bargaining solution for picking Pareto-optimal rate vectors in wireless channels extends to numerous scenarios and problems. For instance, in [288], the use of the Nash bargaining solution for finding efficient rate vectors in the Gaussian interference channel is studied. The authors show that, under certain conditions on the signal-to-noise ratio, the Nash bargaining solution using frequency-division multiplexing significantly outperforms the Nash equilibrium solution in a frequency-flat channel. The results are further extended to a frequency-selective channel whereby it is shown that, using convex optimization techniques, the Nash bargaining solution for an *N*-player game can be found with reasonable complexity, and its performance is significantly better than that of the Nash equilibrium. In OFDMA networks, the authors in [224] discuss the use of bargaining for rate allocation and its relation to well-known concepts such as proportional fairness

² For an interpretation of the convex hull in the context of the considered scenario, the reader is referred to [281].

and max-min fairness. The use of the Nash bargaining solution in MIMO interference channels is discussed in [104]. The link between rate allocation in wireless networks and the Nash bargaining solution is further surveyed in [282] and the references therein.

In summary, beyond the presented example, whenever one is faced with the problem of resource allocation, notably rate allocation in a wireless channel, it is of interest to study the Nash bargaining solution because it can provide an efficient outcome as well as important insights on the operating point of the studied problem.

Nash bargaining for multimedia resource management

In recent years, the need for deploying resource-demanding applications such as multimedia streaming, video surveillance, and video gaming over bandwidth-constrained network infrastructures has increased. In this context, ensuring the required quality-ofservice parameters for these applications becomes very challenging under constrained resources. Hence, an efficient resource management scheme is required to enable multimedia communication over resource-constrained communication networks. This problem is tackled using bargaining theory in [384].

Consider N video users that are seeking to share the bandwidth of a wired or wireless network. The users have an incentive to jointly agree on a division of the network resources so as to optimize their performance. However, each user is self-interested and would aim at obtaining the largest amount of resources for its own use. Clearly, this gives rise to a bargaining problem with the following characteristics:

- The players are the *N* video users.
- Each player *i* seeks to obtain a share x_i of the bandwidth. Hence, we define $u_i(x_i)$ as the utility for player *i*, which is a function of its share of the bandwidth x_i . We let S denote the *N*-dimensional bargaining region.
- A disagreement point *d* ∈ S represents the minimum utility that each user demands before agreeing on a bargaining outcome.

Given this (S, d) bargaining problem, we seek to characterize the Nash bargaining solution and its implications in this multimedia resource-management context. Using an adequate distortion-rate model, the utility function of any video user *i* can be defined as [384]

$$u_i(x_i) = \frac{255 \cdot (x_i - x_{0_i})}{D_{0_i}(x_i - x_{0_i}) + \mu_i},$$
(7.16)

where x_{0_i} is the bandwidth share at the disagreement point, D_{0_i} and μ_i are rate-distortion parameters (D_{0_i} is non-negative, and μ_i is positive) that are dependent on the video sequence characteristics, resolutions, and delay. This utility function is chosen in such a way that PSNR_i = $10 \log_{10} u_i(x_i)$ represents the PSNR achieved by video user *i*. Consequently, there is an optimization of the utility which maps into an optimization of the PSNR.

From (7.16), we first remark that, at the disagreement point, i.e., at $x_i = x_{0_i}$, the utility is 0, and thus the disagreement point *d* can be defined as the origin: d = (0, 0). Furthermore, with this utility definition, it is shown in [384] that the utility region *S* is compact and convex.

Consequently, we assign to every video user a bargaining power $\alpha_i \in [0, 1]$, such that $\sum_{i=1}^{N} \alpha_i = 1$, and we formally define a generalized Nash bargaining problem (S, d) among the video users, with its solution given by the solution to

$$\max_{(u_1,\ldots,u_N)\in\mathcal{S}}\prod_{i=1}^N u_i(x_i)^{\alpha_i}.$$
(7.17)

The outcome of this problem would determine the resource allocation that would provide Pareto-optimal utilities, i.e., PSNR, for the video users. To obtain the generalized Nash bargaining outcome, optimization techniques such as the bisection method can be used (the details of these techniques are beyond the scope of this example but can be found in [384, Algorithm 1]).

Using (7.16), the logarithm of the Nash product in (7.17) becomes

$$10\log_{10}\prod_{i=1}^{N}u_{i}(x_{i})^{\alpha_{i}}=\sum_{i=1}^{N}\alpha_{i}\text{PSNR}_{i}^{*},$$
(7.18)

where $PSNR_i^*$ is the PSNR achieved by user *i* at the generalized Nash bargaining outcome. In other words, the Nash bargaining solution maximizes a weighted sum of PSNRs (weighted in terms of bargaining power, i.e., importance of the user in the game) given the total available bandwidth. In some sense, the Nash bargaining solution in the considered resource-management problem optimizes the total system utility in terms of the weighted sum of PSNRs.

Furthermore, for multimedia resource management, the choice of the bargaining powers allows more importance to be given to some users relative to others, depending on the characteristics of the video (e.g., content characteristics, spatio-temporal resolution, delay requirements) or on the communication channel characteristics (e.g., fading when dealing with wireless channels). For example, to facilitate the transmission of highly delay-sensitive content, one would assign higher bargaining power to users with stringent delay requirements, thus giving them a higher share of the resource so that they maintain an acceptable quality of service.

Moreover, it is shown in [384] that using the generalized Nash bargaining solution in this problem yields a significant performance improvement for the system, notably when compared with equal allocation. Using various bargaining powers and video scenarios, the results in [384] show that, as the resources become scarcer, the use of the generalized Nash bargaining solution becomes more crucial in that it allows important improvements in terms of resource allocation and achieved PSNR performance. Thus, in summary, this example demonstrates how the Nash bargaining solution, in the context of multimedia resource management, provides a fair and Pareto-optimal outcome.

Finally, beyond the presented examples, bargaining theory and the Nash bargaining solution admit numerous applications in wireless communications and networking such as in cognitive-radio networks [91, 516, 46], vehicular communications [455], common radio resource management [247].

7.2 Coalitional game theory: basics

In this section, we introduce the main fundamental concepts of coalitional-game theory, which is a key branch of cooperative games.

7.2.1 Introduction

Coalitional-game theory provides suitable analytical tools that have been widely explored in different disciplines such as economics and political science. With the recent emergence of cooperation as a new networking paradigm, coalitional-game theory started to become a central framework for modeling cooperation in wireless and communication networks. For instance, coalitional games prove to be a very powerful tool for designing fair, robust, practical, and efficient cooperation strategies in communication networks. The goal of the remainder of this chapter is to introduce the main concepts of coalitional-game theory as well as to address the major opportunities and challenges in applying coalitional games to the understanding and designing of modern communication systems, with emphasis on both new analytical techniques and potential application scenarios.

This section starts by laying out the main components of coalitional games before zooming in on an in-depth study of these games, their solution concepts, and their properties. Since the literature on coalitional games is sparse, we will follow the engineering-oriented classification of coalitional games that is given in [412]. Hence, based on various properties of the considered game, we group coalitional games into three distinct classes [412]:

- Class I. Canonical (coalitional) games³
- Class II. Coalition-formation games
- Class III. Coalitional graph games.

The key features of these classes are summarized in Fig. 7.2, and an in-depth study of each class is provided in the rest of this chapter.

7.2.2 Coalitional-game theory: preliminaries

In essence, coalitional games involve a set of players, denoted by \mathcal{N} , who seek to form cooperative groups, i.e., coalitions, in order to strengthen their positions in a given situation. Any coalition $S \subseteq \mathcal{N}$ represents an agreement between the players in S to act as a single entity. The formation of coalitions or alliances is ubiquitous in many applications. For example, in political games, parties or groups of individuals can form coalitions for improving their voting power. In addition to the player set \mathcal{N} , the second fundamental concept of a coalitional game is the coalition *value*. Mainly, the coalition value, denoted by v, quantifies the worth of a coalition in a game. The definition of the

³ We will use the terms "canonical coalitional games" and "canonical games" interchangeably throughout this book.





coalition value determines the *form* and *type* of the game. Nonetheless, independent of the definition of the value, a coalitional game is uniquely defined as follows:

DEFINITION 7.1 A coalitional game (or game in coalitional form) is defined by the pair (\mathcal{N}, v) , where \mathcal{N} is the set of the players, and v is a mapping that determines the payoffs that these players receive in the game.

It must be noted that the value v is, in many instances, referred to as *the game*, since for every v a different coalitional game may be defined.

The most common form of a coalitional game is the *characteristic form*, whereby the value of a coalition *S* depends *solely* on the members of that coalition, with no dependence on how the players in $\mathcal{N} \setminus S$ are structured. The characteristic form was introduced, along with a category of coalitional games known as games with *transferable utility* (TU), by von Neuman and Morgenstern [492]. The value of a game in characteristic form with TU is defined as follows:

DEFINITION 7.2 The characteristic function of a coalitional game with transferable utility is a function v over the real line defined as follows: $v : 2^{\mathcal{N}} \to \mathbb{R}$ with $v(\emptyset) = 0$.

This characteristic function associates with every coalition $S \subseteq \mathcal{N}$ a real number quantifying the gains of *S*. The TU property implies that the total utility represented by this real number can be divided in any manner between the coalition members. The values in TU games are thought of as monetary values that the members in a coalition can distribute among themselves using an appropriate *fairness* rule (one such rule being an equal distribution of the utility). The amount of utility that a player $i \in S$ receives from the division of v(S) constitutes the player's *payoff* and is denoted by x_i hereinafter. The vector $\mathbf{x} \in \mathbb{R}^S$, with each element x_i being the payoff for player $i \in S$, constitutes a *payoff allocation* for the players in *S*. Although the TU characteristic function can model a broad range of games, many scenarios exist in which the coalition value cannot be assigned a single real number, or rigid restrictions exist on the distribution of the utility. These games are known as *coalitional games with non-transferable utility (NTU)* and were first introduced by Aumann and Peleg using non-cooperative strategic games as a basis [50, 347]. In an NTU game, the payoff that each player in a coalition *S* receives is dependent on the joint actions that the players of coalition *S* select.⁴ The value of a coalition *S* in an NTU game, $v(S) \subseteq \mathbb{R}^{S}$, where each element x_i of a vector $\mathbf{x} \in v(S)$ represents a payoff that player $i \in S$ can obtain within coalition *S* given a certain strategy selected by player *i* while being a member of *S*. Given this definition, a TU game can be seen as a particular case of the NTU framework [347]. Coalitional games in characteristic form with TU or NTU constitute one of the most important types of cooperative games, and their solutions are explored in detail in the rest of this chapter.

Furthermore, although the characteristic form covers a broad range of applications, there has been a recent interest in games in which the value of a coalition depends on the partition of \mathcal{N} that is in place at any time during the game. In such games, unlike the characteristic form, the value of a coalition S will have a strong dependence on how the players in $\mathcal{N} \setminus S$ are structured. For this purpose, Thrall and Lucas [475] introduced the concept of games in *partition form*. In these games, given a *coalitional structure* \mathcal{B} , defined as a *partition* of \mathcal{N} , i.e., a collection of coalitions $\mathcal{B} = \{B_1, \ldots, B_l\}$ such that $\forall i \neq j, B_i \cap B_j = \emptyset$, and $\bigcup_{i=1}^l B_i = \mathcal{N}$, the value of a coalition $S \in \mathcal{B}$ is defined as $v(S, \mathcal{B})$. This definition imposes a dependence on the coalitional structure when evaluating the value of S. Coalitional games in partition form are inherently complex to solve; however, the potential of these games is interesting and, although in this chapter the main focus will be on the characteristic form, we will also provide some insights on games in partition form in the following sections.

As an example of the difference between characteristic and partition forms, consider a five-player game with $\mathcal{N} = \{1, 2, 3, 4, 5\}$, and let $S_1 = \{1, 2, 3\}$, $S_2 = \{4\}$, $S_3 = \{5\}$, and $S_4 = \{4, 5\}$. Given two partitions $\mathcal{B}_1 = \{S_1, S_2, S_3\}$ and $\mathcal{B}_2 = \{S_1, S_4\}$ of \mathcal{N} , evaluating the value of coalition S_1 depends on the form of the game. If the game is in *characteristic form*, then $v(S_1, \mathcal{B}_1) = v(S_1, \mathcal{B}_2) = v(S_1)$, while in *partition form* $v(S_1, \mathcal{B}_1) \neq v(S_1, \mathcal{B}_2)$ (the value here can be either TU or NTU). The basic difference is that, unlike in the characteristic form, the value of S_1 in partition form depends on whether players 4 and 5 cooperate or not. This is illustrated in Fig. 7.3(a).

In many coalitional games, the players are interconnected and communicate through pairwise links in a graph. In such scenarios, both the characteristic form and the partition form may be unsuitable since, in both forms, the value of a coalition *S* is independent of how the members of *S* are connected. For modeling the interconnection graphs, coalitional games in *graph form* were introduced by Myerson in [346], in which *connected* graphs were mapped into coalitions. This work was generalized in [229] by making the value of each coalition $S \subseteq \mathcal{N}$ a function of the graph structure connecting the members

⁴ The action space depends on the underlying non-cooperative game (see [50] for examples).



Fig. 7.3 (a) Coalitional games in characteristic form vs. partition form; (b) example of a coalitional game in graph form.

of *S*. Hence, given a coalitional game (\mathcal{N}, v) and a graph G_S (directed or undirected) connecting the members of a coalition $S \subseteq \mathcal{N}$ (the coalition members are, thus, the vertices of the graph G_S), the value of *S* in *graph form* is given by $v(G_S)$. For games in graph form, the value can also depend on the graph $G_{\mathcal{N}\setminus S}$ interconnecting the players in $\mathcal{N} \setminus S$. An example of a coalitional game in graph form is shown in Fig. 7.3(b). In this figure, given two graphs $G_S^1 = \{(1,2), (2,3)\}$ and $G_S^2 = \{(1,2), (1,3)\}$ (a pair (i,j) is a link between two players *i* and *j*) defined over coalition $S = \{1,2,3\}$, a coalitional game in graph form could assign a different value for coalition *S* depending on the graph.⁵ Hence, in graph form, it is possible that $v(G_S^1) \neq v(G_S^2)$, while in characteristic or partition form, the presence of the graph does not affect the value.

⁵ In this example we have considered an undirected graph and a single link between each pair of nodes. However, multiple links between pairs of nodes as well as directed graphs can also be considered within the graph form of coalitional games.

7.3 Class I: canonical coalitional games

7.3.1 Main properties of canonical coalitional games

Canonical coalitional games are a class of coalitional games in which the value is considered in characteristic form (TU or NTU) and cooperation, i.e., the formation of large coalitions, is always beneficial to the players. In such games, it is assumed that when forming a larger coalition the players cannot do worse than by acting alone (non-cooperatively). This characteristic maps to the mathematical property of superadditivity, defined as follows for NTU games:

DEFINITION 7.3 An NTU coalitional game (\mathcal{N}, v) is said to be superadditive if and only if

$$v(S_1 \cup S_2) \supset \{ \mathbf{x} \in \mathbb{R}^{S_1 \cup S_2} | (x_i)_{i \in S_1} \in v(S_1), (x_j)_{j \in S_2} \in v(S_2) \}$$

$$\forall S_1 \subset \mathcal{N}, S_2 \subset \mathcal{N}, S_1 \cap S_2 = \emptyset, \qquad (7.19)$$

where **x** is a payoff allocation for coalition $S_1 \cup S_2$.

In other words, superadditivity implies that, in an NTU game, given any two disjoint coalitions S_1 and S_2 , if coalition $S_1 \cup S_2$ forms, this coalition can always give its members the payoffs that they would receive if they acted separately in the disjoint coalitions S_1 and S_2 .

For a TU game, the superadditivity property given in (7.19) reduces to

$$v(S_1 \cup S_2) \ge v(S_1) + v(S_2) \ \forall S_1 \subset \mathcal{N}, S_2 \subset \mathcal{N}, \text{ s.t. } S_1 \cap S_2 = \emptyset.$$

$$(7.20)$$

From (7.20), the concept of a superadditive game is better understood. For instance, a game is superadditive if cooperation, i.e., the formation of a large coalition out of disjoint coalitions, guarantees at least the value that is obtained by the disjoint coalitions separately. The rationale behind superadditivity is that, within a coalition, the players can always revert back to their non-cooperative behavior to obtain their non-cooperative payoffs. Thus, in a superadditive game, cooperation is never detrimental to any of the players. Note that, in some references, superadditivity is considered as part of the definition of a coalitional game, because these references deal solely with canonical games. However, in this book, we will deal with superadditivity as a separate property that may or may not be satisfied by a coalitional game.⁶

Since canonical games are superadditive by definition, it is to the joint benefit of the players to always form the *grand coalition* \mathcal{N} (i.e, the coalition of *all* the players) since the payoff received from $v(\mathcal{N})$ is at least as large as the amount received by the players in any disjoint set of coalitions they could form (for both the TU and the NTU cases). The formation of the grand coalition in canonical games implies that the main emphasis

⁶ As will be seen in later chapters, many applications map to coalitional games that are inherently non-superadditive.

is on studying the properties of this grand coalition. Two key aspects are of importance in canonical games: (i) finding a payoff allocation that guarantees that no group of players has an incentive to leave the grand coalition (having a *stable* grand coalition), and (ii) assessing the gains that the grand coalition can achieve as well as the fairness criteria that must be used for distributing these gains (having a *fair* grand coalition). For solving canonical coalitional games, the literature presents a number of concepts [347, 377] that we will explore in detail in the following sections.

7.3.2 The core as a solution for canonical coalitional games

The most renowned solution concept for canonical coalitional games is the *core*. The core of a canonical game is directly related to the grand coalition's stability. In a canonical coalitional game (\mathcal{N}, v) , as a result of superadditivity, the players have an incentive to form the grand coalition \mathcal{N} . In this regard, the core of a canonical game is the set of payoff allocations that guarantees that no group of players has an incentive to leave \mathcal{N} in order to form another coalition $S \subset \mathcal{N}$. For a TU game, given the grand coalition \mathcal{N} , we make the following definitions.

DEFINITION 7.4 A payoff vector $\mathbf{x} \in \mathbb{R}^{\mathcal{N}}$ for dividing $v(\mathcal{N})$ is group-rational if $\sum_{i \in \mathcal{N}} x_i = v(\mathcal{N})$.

DEFINITION 7.5 A payoff vector $\mathbf{x} \in \mathbb{R}^{N}$ is individually rational if every player can obtain a benefit no less than acting alone, i.e., $x_i \ge v(\{i\}), \forall i \in N$.

DEFINITION 7.6 An imputation is a payoff vector that is both individually rational and group-rational.

Following the definition of an imputation, we can define the core of a TU canonical coalitional game as follows:

DEFINITION 7.7 Given a TU canonical coalitional game (\mathcal{N}, v) , the core is defined as the set of imputations in which no coalition $S \subset \mathcal{N}$ has an incentive to reject the proposed payoff allocation, deviate from the grand coalition, and form coalition S instead. Mathematically, the core C_{TU} of a canonical TU game is given by

$$\mathcal{C}_{\mathrm{TU}} = \left\{ \mathbf{x} : \sum_{i \in \mathcal{N}} x_i = v(\mathcal{N}) \text{ and } \sum_{i \in S} x_i \ge v(S) \forall S \subseteq \mathcal{N} \right\}.$$
(7.21)

In other words, the core guarantees that the players have no incentive to deviate from the grand coalition, because any payoff allocation **x** that is in the core guarantees at least an amount of utility equal to v(S) for every $S \subset \mathcal{N}$. Clearly, whenever one is able to find a payoff allocation that lies in the core, then the grand coalition is a stable and optimal solution for the coalitional game.

For solving NTU canonical games using the core, the value v of the NTU game must satisfy the following conditions, for any coalition *S* [347]:

- 1. The value v(S) of any coalition S must be a closed and convex subset of \mathbb{R}^{S} .
- 2. The value v(S) must be *comprehensive*, i.e., if $\mathbf{x} \in v(S)$ and $\mathbf{y} \in \mathbb{R}^{S}$ are such that $\mathbf{y} \leq \mathbf{x}$, then $\mathbf{y} \in v(S)$.
- 3. The set $\{\mathbf{x} | \mathbf{x} \in v(S) \text{ and } x_i \ge z_i, \forall i \in S\}$ with $z_i = \max\{y_i | \mathbf{y} \in v(\{i\})\} < \infty \forall i \in \mathcal{N}$ must be a bounded subset of \mathbb{R}^S .

The comprehensive property implies that if a certain payoff allocation x is achievable by the members of a coalition S, then, by changing their strategies, the members of S can achieve any allocation y where $y \le x$. The last property implies that, for a coalition S, the set of vectors in v(S) in which each player in S receives no less than the maximum that it can obtain non-cooperatively, i.e., z_i , is a bounded set. These properties are often assumed to be part of the definition of an NTU game because they are needed for solving the game using the core, notably when the game is canonical. However, since the NTU framework is more general, we consider these characteristics as properties that may or may not be satisfied by an NTU game.

For a canonical NTU game (\mathcal{N}, v) with v satisfying the above properties, the core is defined as

$$\mathcal{C}_{\text{NTU}} = \{ \mathbf{x} \in v(\mathcal{N}) | \forall S, \nexists \mathbf{y} \in v(S), \text{ such that } y_i > x_i, \forall i \in S \}.$$
(7.22)

This definition for NTU also guarantees a stable grand coalition. The basic idea is that any payoff allocation in the core of an NTU game guarantees that no coalition S can leave the grand coalition and provide a better allocation *for all* of its members. The difference from the TU case is that, in the NTU core the grand coalition's stability is acquired over the elements of the payoff vectors, while in the TU game it is acquired by the sum of the payoff vectors' elements.

Properties and existence

The cores of TU or NTU canonical games are not always guaranteed to exist. In fact, in many games, the core is empty and, hence, the grand coalition cannot be stabilized. In these cases, alternative solution concepts may be used, as we will see in the following sections. However, coalitional-game theory provides several categories of games that fit under our canonical game class, where the core is guaranteed to be non-empty. Before exposing the existence results for the core, we provide a simple example of the core in a TU canonical game:

Example 7.3 Consider a majority-voting TU game (\mathcal{N}, v) , where $\mathcal{N} = \{1, 2, 3\}$. The players, on their own, have no voting power, hence $v(\{1\}) = v(\{2\}) = v(\{3\}) = 0$. Any two-player coalition wins two-thirds of the voting power; hence, $v(\{1,2\}) = v(\{1,3\}) = v(\{2,3\}) = \frac{2}{3}$. The grand coalition wins the whole voting power, and thus $v(\{1,2,3\}) = 1$. Clearly, this game is superadditive and is in characteristic form, and is thus classified as canonical. By (7.21), solving the following inequalities yields the core and shows what

allocations stabilize the grand coalition.

$$\begin{aligned} x_1 + x_2 + x_3 &= v(\{1, 2, 3\}) = 1, \ x_1 \ge 0, \ x_2 \ge 0, \ x_3 \ge 0, \\ x_1 + x_2 \ge v(\{1, 2\}) &= \frac{2}{3}, \ x_1 + x_3 \ge v(\{1, 3\}) = \frac{2}{3}, \ x_2 + x_3 \ge v(\{2, 3\}) = \frac{2}{3} \end{aligned}$$

By manipulating these inequalities, the core of this game is found to be the unique vector $\mathbf{x} = \begin{bmatrix} \frac{1}{3} & \frac{1}{3} & \frac{1}{3} \end{bmatrix}$, which corresponds to an equal division of the total utility for the grand coalition among all three players.

In general, given a TU coalitional game (\mathcal{N}, v) and an imputation $\mathbf{x} \in \mathbb{R}^{\mathcal{N}}$, the core is found by the following linear program (LP):

$$\min_{\mathbf{x}} \sum_{i \in \mathcal{N}} x_i, \text{ s.t. } \sum_{i \in S} x_i \ge v(S), \forall S \subseteq \mathcal{N}.$$
(7.23)

The existence of the TU core is related to the feasibility of the LP in (7.23). In general, determining whether the core is non-empty through this LP, is NP (non-deterministic polynomial time) -complete [113] owing to the number of constraints growing exponentially with the number of players N (this is also true for NTU games). However, for determining the non-emptiness of the core, as well as finding the allocations that lie in the core, several techniques exist; these are summarized in Table 7.1.

The *first* technique in Table 7.1 deals with TU games with up to three players. In such games, the core can be found using an easy graphical approach. The main idea is to plot the constraints of (7.23) in the plane $\sum_{i=1}^{3} x_i = v(\{1, 2, 3\})$. By doing so, the region containing the core allocation can be easily identified. Several examples of the graphical techniques are found in [377], and the technique for solving them is straightforward. Although the graphical method can provide a lot of intuition into the core of a canonical game, its use is limited to TU games with up to three players.

The *second* technique in Table 7.1 utilizes the dual of the LP in (7.23) to show that the core is non-empty. The main result is given through the Bondareva–Shapley Theorem [347, 377], which relies on the *balanced* property. We define a balanced TU game as follows:

DEFINITION 7.8 A canonical TU game $(\mathcal{N}, \mathbf{v})$ is known as a balanced game if and only if the inequality

$$\sum_{S \subseteq \mathcal{N}} \mu(S) v(S) \le v(\mathcal{N})$$
(7.24)

is satisfied for all non-negative weight collections $\mu = (\mu(S))_{S \subseteq \mathcal{N}}$ (μ is a collection of weights, i.e., numbers in [0, 1], associated with each coalition $S \subseteq \mathcal{N}$) that satisfy $\sum_{S \ni i} \mu(S) = 1, \forall i \in \mathcal{N}$; this set of non-negative weights is known as a balanced set.

This notion of a balanced game is interpreted as follows. Each player $i \in N$ possesses a single unit of time, which can be distributed between all the coalitions of which *i* can

Table 7.1 Approaches for finding the core of a canonical coalitional game.

Game-theoretic and mathematical approaches

- (T1) A *graphical* approach can be used for finding the core of TU games with up to three players.
- (T2) Using duality theory, a necessary and sufficient condition for the non-emptiness of the core exists through the *Bondareva–Shapley Theorem* (Theorem 7.9) for TU and NTU [347, 377].
- (T3) A class of canonical games, known as *convex coalitional games* always has a non-empty core.
- (T4) A necessary and sufficient condition for a non-empty core exists for a class of canonical games known as *simple games*, i.e., games in which $v(S) \in \{0, 1\}$, $\forall S \subseteq \mathcal{N}$, and $v(\mathcal{N}) = 1$.

Application-specific approaches

- (T5) In several applications, it suffices to determine whether payoff distributions that are of interest in a given game, e.g., fair distributions, lie in the core.
- (T6) In many games, exploiting game-specific features such as the value's mathematical definition or the underlying nature and properties of the game model, helps in finding the imputations that lie in the core.

be a member. Every coalition $S \subseteq \mathcal{N}$ is active during a fraction of time $\mu(S)$ if all of its members are active during that time, and this coalition achieves a payoff of $\mu(S)v(S)$. In this context, the condition $\sum_{S \ni i} \mu(S) = 1$, $\forall i \in \mathcal{N}$ is simply a feasibility constraint on the players' time allocation, and the game is balanced if there is no feasible allocation of time that can yield a total payoff for the players that exceeds the value of the grand coalition $v(\mathcal{N})$. Subsequently, given a TU balanced canonical game, the following result holds [347, 377]:

THEOREM 7.2 (Bondareva–Shapley Theorem) The core of a game is non-empty if and only if the game is balanced.

For NTU canonical games, two different definitions for balancedness exist (one of which is analogous to the TU case) and can be found in [347, 377]. The definitions for NTU accommodate the fact that the value v in an NTU game is a set and not a function. In the NTU case, the Bondareva–Shapley Theorem is also true; however, in that case the balanced condition presented is sufficient but not necessary as a second definition for a balanced game to also exist (see [347, 377]). Therefore, in a given canonical game, one can always show that the core is non-empty by proving that the game is balanced through (7.24) for TU games or its counterparts for NTU [347, Section 9.7].

The *third* technique in Table 7.1 pertains to *convex* games, which are defined as follows:

DEFINITION 7.9 A TU canonical game (\mathcal{N}, v) is convex if

$$v(S_1) + v(S_2) \le v(S_1 \cup S_2) + v(S_1 \cap S_2) \ \forall \ S_1, S_2 \subseteq \mathcal{N}.$$
(7.25)

Alternatively, a convex coalitional game is defined as any coalitional game that satisfies $v(S_1 \cup \{i\}) - v(S_1) \le v(S_2 \cup \{i\}) - v(S_2)$, whenever $S_1 \subseteq S_2 \subseteq \mathcal{N} \setminus \{i\}$. This alternative definition implies that a game is convex if and only if for each player $i \in \mathcal{N}$ the marginal contribution of this player, i.e., the difference between the value of a coalition with and without this player, is non-decreasing with respect to set inclusion. The convexity property can also be extended to NTU in several ways, and the reader is referred to [377, Chapter 9.9] for more details. For both TU and NTU canonical games, a convex game is balanced and *has a non-empty core*, but the converse is not always true [377]. Thus, convex games constitute an important class of games in which the core is non-empty.

The *fourth* technique in Table 7.1 pertains to *simple games*, which are an interesting class of canonical games in which the core can be shown to be non-empty. A simple game is a coalitional game in which the values are either 0 or 1, i.e., $v(S) \in \{0, 1\}$, $\forall S \subseteq \mathcal{N}$ and the grand coalition has $v(\mathcal{N}) = 1$. These games model numerous scenarios, notably voting games. It is known that a simple game that contains at least one *veto* player $i \in \mathcal{N}$, i.e., a player i such that $v(\mathcal{N} \setminus \{i\}) = 0$ has a *non-empty core* [377]. Moreover, in such simple games, the core is fully characterized, and it consists of all non-negative payoff profiles $\mathbf{x} \in \mathbb{R}^{\mathcal{N}}$ such that $x_i = 0$ for each player i that is a *non-veto* player, and $\sum_{i \in \mathcal{N}} x_i = v(\mathcal{N}) = 1$.

The first four techniques in Table 7.1 rely mainly on well-known game-theoretic properties. In many practical scenarios, notably in wireless and communication networking applications, alternative techniques may be needed to find the allocations in the core. These alternatives are inherently application-specific, and depend on the nature of the defined game and the properties of the defined value function. One of these alternatives, the *fifth* technique in Table 7.1, is to investigate whether well-known allocation rules yield vectors that lie in the core. In many communication applications (and even gametheoretic settings), the objective is to assess whether certain well-defined types of fair allocation, such as equal fairness or proportional fairness, are in the core or not, without finding all the allocations that are in the core. In such games, showing the non-emptiness of the core is done by testing whether such well-known allocations lie in the core or not, using the intrinsic properties of the considered game and using (7.21) for TU games or (7.22) for NTU games. A simple example of such a technique is Example 7.3, where one can easily check the non-emptiness of the core by showing that the equal allocation lies in the core. In many canonical games, the nature of the defined value for the game can be explored for showing the non-emptiness of the core; this is done in many applications, such as [323], where information-theoretical properties are used; [191], where network properties are used; as well as [42, 447], where the value is given as a convex optimization and, through duality, a set of allocations that lie in the core can be found. Hence, whenever techniques (T1)-(T4) are too complex or difficult to apply for solving a canonical game, as in the *sixth* technique in Table 7.1, one can explore the properties of the considered game model, as in [323, 191, 310, 42, 447].

In summary, the core is one of the most important solution concepts in coalitional games, notably in our canonical games class. It must be stressed that the existence of the core shows that the grand coalition \mathcal{N} of a given (\mathcal{N}, v) canonical coalitional game is stable, optimal (from a payoff perspective), and desirable.

7.3.3 The Shapley value

As a solution concept, the core suffers from three main drawbacks:

- The core can be empty.
- The core can be quite large, so selecting a suitable core allocation can be difficult.
- In many scenarios, the allocations that lie in the core can be unfair to one or more players.

These drawbacks have motivated the search for a solution concept that can associate with each coalitional game (\mathcal{N}, v) a *unique* payoff vector known as the *value* of the game (which is quite different from the value of a coalition). Shapley approached this problem axiomatically by defining a set of desirable properties, and he characterized a unique mapping ϕ that satisfies these axioms, later known as the *Shapley value* [347]. The Shapley value was essentially defined for TU games; however, extensions to NTU games exist. In this chapter, we restrict our attention to the Shapley value for TU canonical games, and we refer the reader to [347, Chapter 9.9] for insights on how the Shapley value is extended to NTU games. Shapley provided four axioms⁷ as follows (ϕ_i is the payoff given to player *i* by the Shapley value ϕ):

- 1. *Efficiency axiom*: $\sum_{i \in \mathcal{N}} \phi_i(v) = v(\mathcal{N}).$
- 2. Symmetry axiom: If player *i* and player *j* are such that $v(S \cup \{i\}) = v(S \cup \{j\})$, for every coalition *S* not containing player *i* and player *j*, then $\phi_i(v) = \phi_i(v)$.
- Dummy axiom: If player i is such that v(S) = v(S ∪ {i}), for every coalition S not containing i, then φ_i(v) = 0.
- 4. Additivity axiom: If u and v are characteristic functions, then $\phi(u+v) = \phi(v+u) = \phi(u) + \phi(v)$.

Shapley showed that there exists a unique mapping, the Shapley value $\phi(v)$, from the space of all coalitional games to \mathbb{R}^N , that satisfies these axioms. Hence, for every game (\mathcal{N}, v) , the Shapley value ϕ assigns a unique payoff allocation in \mathbb{R}^N that satisfies the four axioms. The efficiency axiom is in fact group rationality. The symmetry axiom implies that, when two players have the same contribution in a coalition, their assigned payoffs must be equal. The dummy axiom assigns no payoff to players that do not improve the value of any coalition. Finally, the additivity axiom links the value of different games u and v and asserts that ϕ is a unique mapping over the space of all coalitional games.

The Shapley value also has an alternative interpretation that takes into account the order in which the players join the grand coalition \mathcal{N} . In the event in which the players join the grand coalition in a *random* order, the payoff allotted by the Shapley value to a player $i \in \mathcal{N}$ is the expected marginal contribution of player i when it joins the grand coalition. The basis of this interpretation is that, given any canonical TU game (\mathcal{N}, v) ,

⁷ In some references, the Shapley axioms are compressed into three by combining the dummy and efficiency axioms.

for every player $i \in \mathcal{N}$ the Shapley value $\phi(v)$ assigns the payoff $\phi_i(v)$ given by

$$\phi_i(\mathbf{v}) = \sum_{S \subseteq \mathcal{N} \setminus \{i\}} \frac{|S|! (\mathcal{N} - |S| - 1)!}{\mathcal{N}!} [\mathbf{v}(S \cup \{i\}) - \mathbf{v}(S)].$$
(7.26)

In (7.26), it is clearly seen that the marginal contribution of every player *i* in a coalition *S* is $v(S \cup \{i\}) - v(S)$. The weight that is used in front of $v(S \cup \{i\}) - v(S)$ is the probability that player *i* faces the coalition *S* when entering in a random order, i.e., the players in front of *i* are the ones already in *S*. In this context, there are |S|! ways of positioning the players of *S* at the start of an ordering, and (N - |S| - 1)! ways of positioning the remaining players except *i* at the end of an ordering. The probability that such an ordering occurs (when all orderings are equally probable) is therefore $\frac{|S|!(N-|S|-1)!}{N!}$; consequently, the resulting payoff $\phi_i(v)$ is the expected marginal contribution under random-order joining of the players for forming the grand coalition.

In general, the Shapley value is unrelated to the core. However, in some applications, one can show that the Shapley value lies in the core. Such a result is of interest, since if such an allocation is found, it combines both the stability of the core as well as the axioms and fairness of the Shapley value. In this regard, an interesting result from game theory is that for convex games the Shapley value lies in the core [347, 377]. The Shapley value presents an interesting solution concept for canonical games, and it has numerous applications in both game theory and communication networks. For instance, in coalitional voting simple games, the Shapley value of a player *i* represents its power in the game. In such games, the Shapley value is used as a power index (known as the Shapley-Shubik index), and it has a large number of applications in many game-theoretic and political settings [377]. In communication networks, the Shapley value presents a suitable fairness criteria for allocating resources or data rates, as in [275, 191, 89]. The computation of the Shapley value is generally done using (7.26); however, in games with a large number of players the computational complexity of the Shapley value grows significantly. For computing the Shapley value in reasonable time, several analytical techniques have been proposed, such as multi-linear extensions [377] and sampling methods for simple games [92].

7.3.4 The nucleolus

Another prominent and interesting solution concept for canonical games is the *nucleolus*,⁸ which was introduced mainly for TU games [377]. Extensions of the nucleolus for NTU games are not yet formalized in game theory, and hence this chapter will only focus on the nucleolus for TU canonical games. The basic motivation behind the nucleolus is that, instead of applying a general fairness axiomatization for finding a unique payoff allocation (i.e., a value for the game), one can provide an allocation that minimizes the dissatisfaction of the players with the allocation they can receive in a given (\mathcal{N} , v) game. For this purpose, we introduce the concepts of *excess* and *kernel*, as follows.

⁸ In the game-theory literature, sometimes the term *prenucleolus* is used to indicate a concept similar to the nucleolus but without individual rationality [377].

DEFINITION 7.10 The measure of dissatisfaction with an allocation $\mathbf{x} \in \mathbb{R}^{N}$ for a coalition *S* is defined as the excess $e(\mathbf{x}, S) = v(S) - \sum_{j \in S} x_j$. A kernel of *v* is the set of all allocations $\mathbf{x} \in \mathbb{R}^{N}$ such that

$$\max_{S \subseteq \mathcal{N} \setminus \{j\}, i \in S} e(\mathbf{x}, S) = \max_{G \subseteq \mathcal{N} \setminus \{i\}, j \in G} e(\mathbf{x}, G).$$
(7.27)

The kernel states that if players i and j are in the same coalition, then the highest excess that i can make in a coalition without j is equal to the highest excess that j can make in a coalition without i.

Clearly, an allocation x that can ensure that all excesses (or dissatisfactions) are minimized is of particular interest as a solution⁹ and, hence, constitutes the main motivation behind the concept of the *nucleolus*, defined as follows.

DEFINITION 7.11 Let $\mathbf{O}(\mathbf{x})$ be the vector of all excesses in a canonical game $(\mathcal{N}, \mathbf{v})$ arranged in non-increasing order (except the excess of the grand coalition \mathcal{N}). A vector $\mathbf{y} = (y_1, ..., y_k)$ is said to be lexographically less than a vector $\mathbf{z} = (z_1, ..., z_k)$ (denoted by $\mathbf{y} \prec_{lex} \mathbf{z}$) if $\exists l \in \{1, ..., k\}$, where $y_1 = z_1, y_2 = z_2, ..., y_{l-1} = z_{l-1}, y_l < z_l$. An imputation \mathbf{x} is a nucleolus if, for every other imputation δ , $\mathbf{O}(\mathbf{x}) \prec_{lex} \mathbf{O}(\delta)$. Hence, the nucleolus is the imputation \mathbf{x} that minimizes the excesses in a non-increasing order starting with the maximum excess.

The nucleolus of a canonical coalitional game exists and is unique. The nucleolus is group- and individually rational (since it is an imputation), lies in the kernel of the game, and satisfies the symmetry and dummy axioms of Shapley. If the core is not empty, the nucleolus is in the core. Thus, the nucleolus is the best allocation under a min-max criterion. The process for computing the nucleolus is more complex than for the Shapley value, and is described as follows. First, we start by finding the imputations that distribute the worth of the grand coalition in such a way that the maximum excess (dissatisfaction) is minimized. In the event where this minimization has a unique solution, this solution is the nucleolus. Otherwise, we search for the imputations that minimize the second-largest excess. The procedure is repeated for all subsequent excesses, until a unique solution is found, which would be the nucleolus. These sequential minimizations are solved using linear programming techniques such as the simplex method [128]. The applications of the nucleolus are numerous in game theory. One of the most prominent examples is the marriage contract problem that first appeared in the Babylonian Talmud (0–500 AD):

Example 7.4 A man has three wives, and he is committed to a marriage contract that specifies that they should receive 100, 200, and 300 units respectively, after his death. This implies that, given a total of α units left after the man's death, the three wives can only claim 100, 200, and 300, respectively, out of the α units. If after the man dies, the

⁹ In particular, an imputation x lies in the core of (N, v) if and only if all its excesses are negative or zero.

amount of money left is not enough for this distribution, the Talmud recommends the following:

- If $\alpha = 100$ is available after the man dies, then each wife gets $\frac{100}{3}$.
- If $\alpha = 200$ is available after the man dies, wife 1 gets 50, and the other two get 75 each.
- If $\alpha = 300$ is available after the man dies, wife 1 gets 50, wife 2 gets 100 and wife 3 gets 150.

Note that the Talmud does not specify the allocation for other values of α but, certainly, if $\alpha \ge 600$ each wife simply claims her full right. A key question that puzzled mathematicians and researchers in game theory was how this allocation was made, and it turns out that the nucleolus is the answer. Let us model the game as a coalitional game $(\mathcal{N}, \mathbf{v})$, where \mathcal{N} is the set of all three wives, who constitute the players, and \mathbf{v} is the value defined for any coalition $S \subseteq \mathcal{N}$ as $\mathbf{v}(S) = \max(0, \alpha - \sum_{i \in \mathcal{N} \setminus S} c_i)$, and where $\alpha \in \{100, 200, 300\}$ is the total units left after the death of the man and c_i is the claim that wife *i* must obtain $(c_1 = 100, c_2 = 200, c_3 = 300)$. It then turns out that, with this formulation, the payoffs that were recommended by the Talmud coincide with the nucleolus of the game! This result highlights the importance of the nucleolus in allocating fair payoffs in a game.

In summary, the nucleolus is quite an interesting concept, since it combines a number of fairness criteria with stability. However, the communications applications that utilize the nucleolus remain scarce, one example being [191], where it was used for allocating the utilities in the modeled game. The main drawback of the nucleolus is its computational complexity in some games. Nonetheless, with appropriate models, the nucleolus can be an optimal and fair solution to many applications in wireless and communication networks.

7.3.5 Sample applications in wireless and communication networks

Canonical coalitional games are a very powerful analytical tool for modeling several problems in wireless and communication networks. The application of canonical games range from physical-layer applications, such as rate allocation and stable cooperation, to network-layer applications, such as cooperation in packet-forwarding networks. In this section, we discuss a few state-of-the-art applications of canonical games.

Rate allocation in a multiple-access channel

An elegant and interesting use of canonical games within communication networks is presented in [275] for the study of rate allocation in multiple-access channels. The model in [275] tackles the problem of how to fairly allocate the transmission rates between a number of users accessing a wireless Gaussian multiple-access channel. In this model, the users are trying to obtain a fair allocation of the total transmission rate available. Every user, or group of users (coalition), that does not obtain a fair allocation of the rate can threaten to act on its own, which can reduce the rate available for the remaining
users. Consequently, the game is modeled as a canonical coalitional game defined by $(\mathcal{N}, \mathbf{v})$, where $\mathcal{N} = \{1, \dots, N\}$ is the set of players, i.e., the wireless network users that need to access the channel, and v is the maximum sum-rate that a coalition S can achieve. In order to have a characteristic function, [275] assumes that, when evaluating the value of a coalition $S \subset \mathcal{N}$, the users in $S^c = \mathcal{N} \setminus S$, known as jammers, cooperate in order to *jam* the transmission of the users in S. The jamming assumption is a neat way of maintaining the characteristic form of the game, and it was previously used in game theory for deriving a characteristic function from a strategic-form non-cooperative game [50, 347]. Subsequently, when evaluating the sum-rate utility v(S) of any coalition $S \subseteq \mathcal{N}$, the users in S^c form a single coalition to jam the transmission of S, and, hence, the coalitional structure of S^{c} is always predetermined, yielding a characteristic form. For a coalition S, the characteristic function in [275], v(S), represents the capacity, i.e., the maximum sum-rate, that S achieves under the jamming assumption. Hence, v(S)represents a rate that can be apportioned in an arbitrary manner between the players in S, and thus the game is a TU game. It is proven in [275] that the game is superadditive since the sum of sum-rates achieved by two disjoint coalitions is no less than the sumrate achieved by the union of these two coalitions, since the jammer in both cases is the same (from the assumption of a single coalition of jammers). Consequently, the problem lies in allocating the payoffs (i.e., the transmission rates) between the users in the grand coalition \mathcal{N} that forms in the network. The grand coalition \mathcal{N} has a capacity region $C = \{ \mathbf{R} \in \mathbb{R}^N | \sum_{i=1}^N R_i \leq C(\Gamma_S, \sigma^2), \forall S \subseteq \mathcal{N} \}$, where Γ_S captures the power constraints on the users in S, σ^2 is the Gaussian noise variance, and, hence, $C(\Gamma_S, \sigma^2)$ is the maximum sum-rate (capacity) that coalition S can achieve. Based on these properties, the rate allocation game in [275] is clearly a *canonical coalitional game*, and the key question that [275] seeks to answer is how to allocate the capacity of the grand coalition $v(\mathcal{N})$ among the users in a fair way that stabilizes \mathcal{N} . In answering this question, two main concepts from canonical games are used: the core and the Shapley value.

In this rate-allocation game, it is shown that the core, which represents the set of rate allocations that stabilize the grand coalition, is *non-empty*, using technique (T5) from Table 7.1. By considering the *imputations* that lie in the capacity region C, i.e., the rate vectors $\mathbf{R} \in \mathcal{C}$ such that $\sum_{i=1}^{N} R_i = C(\Gamma_{\mathcal{N}}, \sigma^2)$, it is shown that any such vector lies in the core. Therefore, the grand coalition $\mathcal N$ of the Gaussian MAC canonical game can be stabilized. However, the core of this game is *big* and contains a large number of rate vectors. Thus, the authors in [275] sought to answer the next question, how to select a single fair allocation which lies in the core. For this purpose, the authors investigated the use of the Shapley value as a fair solution for rate allocation that accounts for the random order of joining of the players in the grand coalition. In this setting, the Shapley value simply implies that no rate is left unallocated (efficiency axiom), dummy players receive no rate (dummy axiom), and the labeling of the players does not affect the rate that they receive (symmetry axiom). However, the authors show that (i) the fourth Shapley axiom (additivity) is not suitable for the rate-allocation game, and (ii) the Shapley value does not lie in the core and, hence, cannot stabilize the grand coalition. Based on these results for the Shapley value, the authors propose a new fairness criterion, named "envyfree" fairness. The envy-free fairness criterion relies on the first three axioms of Shapley Table 7.2 The main steps in solving the Gaussian MAC rate allocation canonical game as per [275].

- 1. The player set is the set N of users in a Gaussian MAC channel.
- For a coalition S ⊆ N, a superadditive value function in characteristic form with TU is defined as the maximum sum-rate (capacity) that S achieves under the assumption that the users in coalition S^c = N \ S attempt to jam the communication of S.
- 3. Through technique (T5) of Table 7.1 the core is shown to be non-empty and containing all imputations in the capacity region of the grand coalition.
- 4. The Shapley value is discussed as a fairness rule for rate allocation, but is shown to be outside the core, thus rendering the grand coalition unstable.
- 5. A new application-specific fairness rule, known as "envy-free" fairness, is shown to lie in the core and is presented as a solution to the rate-allocation game in Gaussian MAC.

(without the additivity axiom), and complements them with a fourth axiom, the *envy-free allocation axiom* [275, Eq. (6)]. This axiom states that, given two players *i* and *j* with power constraints $\Gamma_i > \Gamma_j$, an envy-free allocation ψ gives a payoff $\psi_i(v)$, for user *j* in the game (\mathcal{N}, v), equal to the payoff $\psi_i(v^{i,j})$ for user *i* in the game ($\mathcal{N}, v^{i,j}$), where $v^{i,j}$ is the value of the game in which user *i* utilizes a power $\Gamma_i = \Gamma_j$. Mathematically, this axiom implies that $\psi_j(v) = \psi_i(v^{i,j})$. With these axioms, it is shown that a unique allocation exists and that this allocation lies in the core. Thus, the envy-free allocation is presented as a fair and suitable solution for the rate-allocation game in [275]. Finally, the approach used for solving the rate-allocation canonical coalitional game in [275] is summarized in Table 7.2.

Canonical games for receiver and transmitter cooperation

In [323], canonical games were used for studying the cooperation possibilities between single-antenna receivers and transmitters in an interference channel. The model considered in [323] consisted of a set of transmitter-receiver pairs, in a Gaussian interference channel. The authors studied the cooperation between the receivers under two coalitional-game models: a TU model in which the receivers communicate through noise-free channels and jointly decode the received signals, and an NTU model in which the receivers cooperate by forming a linear multi-user detector (in this case the interference channel is reduced to a MAC channel). Furthermore, the authors studied the transmitters' cooperation problem under perfect cooperation and partial decode-and-forward cooperation, while considering that the receivers have formed the grand coalition. Since all the considered games are canonical (as we will see later), the main interest is in studying the properties of the grand coalitions for the receivers and for the transmitters.

For receiver cooperation using joint decoding, the coalitional-game model is as follows. The player set \mathcal{N} is the set of links (the players are the receivers of these links) and, assuming that the transmitters do *not* cooperate, the value v(S) of a coalition $S \subseteq \mathcal{N}$ is the maximum sum-rate achieved by the links whose receivers belong to S. Under this model, one can easily see that the utility is transferable since it represents a sum-rate; hence the game is TU. The game is also in characteristic form, since, as the transmitters are considered non-cooperative, the sum-rate achieved when the receivers in S cooperate depends solely on the receivers in S, while the signal from the links in $\mathcal{N} \setminus S$ is treated as interference. In this game, the cooperation channels between the receivers are considered noiseless; hence, cooperation is always beneficial, and the game is shown to be superadditive. This game is clearly a canonical game, and the interest is in studying the properties of the grand coalition of receivers. Under this cooperation scheme, the network can be seen as a single-input multiple-output (SIMO) MAC channel, and the coalitional game is shown to have a non-empty core that contains all the imputations that lie in the SIMO MAC capacity region. The technique used for this proof is similar to that for the game in [275], which selects a particular set of rate vectors, those that are in the SIMO MAC region, and shows that they lie in the core as per (T5) from Table 7.1. The core of this game is very large, and for selecting fair allocations, it is proven in [323] that the Nash bargaining solution, and in particular a proportional fair rate allocation, lie in the core, and hence constitute suitable fair and stable allocations. For the second receiver-cooperation game, the model is similar to the joint decoding game, with one major difference: instead of jointly decoding the received signals, the receivers form linear multi-user detectors (MUD). The MUD coalitional game is inherently different from the joint decoding game since, in a MUD, the SINR ratio achieved by a user i in coalition S cannot be shared with the other users, and hence the game becomes an NTU game with the SINR representing the payoff for each player. In this NTU setting, the value v(S) of a coalition S becomes the set of SINR vectors that a coalition S can achieve. For this NTU game, the grand coalition is proven to be stable and sum-rate maximizing in a high SINR regime using limiting conditions on the SINR expression, hence technique (T6) in Table 7.1.

For modeling the transmitter-cooperation problem as a coalitional game, two assumptions are made: (i) the receivers jointly decode the signal and hence form a grand coalition, and (ii) a jamming assumption similar to [275] is considered for the purpose of maintaining the characteristic form. In the transmitters' game, from the set of links \mathcal{N} , the transmitters are the players. When considering the transmitters' cooperation along with the receivers' cooperation, the interference channel is mapped onto a MIMO MAC channel. For maintaining a characteristic form, it is assumed, in a manner analogous to [275], that whenever a coalition of transmitters S forms, the users in $S^c = \mathcal{N} \setminus S$ form one coalition and aim to jam the transmission of coalition S. Without this assumption, the maximum sum-rate that a coalition can obtain depends highly on how the users in S^{c} structure themselves; hence, it requires a partition form that may be difficult to solve. With these assumptions, the value of a coalition S is defined as the maximum sum-rate achieved by S when the coalition S^c seeks to jam the transmission of S. Using this transmitters-with-jamming coalitional game, the authors show that, in general, the game has an empty core. This game is not totally canonical since it does not satisfy the superadditivity property. However, by proving through [323, Theorem 19] that the grand coalition is the optimal partition, from a total-utility point of view the grand coalition becomes the main candidate partition for the core. The authors conjecture that, in some cases, the core can also be non-empty, depending on the power and channel gains. However, no existence results for the core can be provided in this game. Finally, the authors in [323] provide a discussion of the grand coalition and its feasibility

Table 7.3 Main results for receiver and transmitter-cooperation coalitional games as per [323].

- 1. The coalitional game between the receivers, where cooperation entails joint decoding of the received signal, is a canonical TU game that has a non-empty core. Hence, the grand coalition is the stable and sum-rate-maximizing coalition.
- 2. The receivers' coalitional game, in which cooperation entails forming linear multi-user detectors, is a canonical NTU game that has a non-empty core. Hence, the grand coalition is the stable and sum-rate maximizing coalition.
- 3. For transmitter cooperation under the jamming assumption, the coalitional game is not *superadditive*, and hence it is non-canonical. However, the grand coalition is shown to be the rate-maximizing partition.
- 4. For transmitter cooperation under the jamming assumption, no results for the existence of the core can be found, owing to mathematical intractability.

when the transmitters employ a partial decode-and-forward cooperation. The main results of this receiver- and transmitter-cooperation canonical game are summarized in Table 7.3.

Other sample applications of canonical coalitional games

Canonical coalitional games cover a broad range of communications and networking applications and, indeed, most research activities in these areas utilize the tools that fall within the canonical coalitional-games class. In addition to the previous examples, numerous applications can use models that involve canonical games.

For instance, in [191] canonical coalitional games are used to solve an inherent problem in packet-forwarding ad hoc networks. In such networks, the users located in the center of the network, known as backbone nodes, have a mutual benefit in forwarding each others' packets. In contrast, users located at the boundary of the network, known as boundary nodes, are not helped by the backbone nodes because the backbone nodes do not need the help of the boundary nodes at any time. Hence, in such a setting the boundary nodes end up having no way of sending their packets to other nodes, a problem known as the *curse of the boundary nodes*. In [191], a canonical coalitional-game model is proposed between a player set \mathcal{N} that includes all boundary nodes and a *single* backbone node. The details of this model are discussed and developed further in Chapter 12 of this book. Beyond packet forwarding, many other applications, such as in [42, 310, 89], utilize several of the techniques in Table 7.1 for studying the grand coalition in a variety of communications applications.

In summary, canonical games are an important tool for studying cooperation and fairness in communication networks, notably when cooperation is always beneficial. Future applications are numerous, e.g., to study cooperative-transmission capacity gains, distributed cooperative source coding, and cooperative relaying in cognitive radio. In brief, whenever a cooperative scheme that yields significant gains at any layer is devised, one can utilize canonical coalitional games for assessing the stability of the grand coalition and identifying fairness criteria in allocating the gains that result from cooperation. Finally, it has to be noted that canonical games are not restricted to link-level analysis, but also extend to network-level studies.

7.4 Class II: coalition-formation games

7.4.1 Main properties of coalition-formation games

Coalition-formation games encompass coalitional games where, unlike the canonical class, *network structure* and *cost* for cooperation play major roles. While in canonical games the focus was on the stability of the grand coalition, the main challenge in coalition-formation games is to study the *network coalitional structure*, i.e., to answer questions such as: Which coalitions will form? What is the optimal coalition size? How does the network structure evolve over time? How can we assess the structure's characteristics? and so on. In contrast to canonical games, a coalition-formation game is generally not superadditive and can support both the characteristic form and partition-form models (in TU or NTU). Another important characteristic that classifies a game as a coalition-formation game is the presence of a cost for forming coalitions. In canonical games, as well as in most of the literature, there is an implicit assumption that forming a coalition requires a negotiation process or an information-exchange process that can incur a cost, thus reducing the gains from forming the coalition.

In general, coalition-formation games can be divided into two types:

- Static coalition-formation games
- Dynamic coalition-formation games.

In static coalition-formation games, an external factor imposes a certain coalitional structure, and the objective is to study the properties of this structure, such as its stability. In contrast, dynamic coalition-formation games constitute a richer framework. In these games, the main objectives are to analyze the formation of a coalitional structure through players' interaction, and to study the properties of this structure and its adaptability to environmental variations or externalities. In contrast to canonical games, with their formal rules and analytical solution concepts, such as the core, solving a coalition-formation game, notably dynamic coalition formation, is more difficult, and application-specific.

The rest of Section 7.4 is devoted to dissecting the key properties of coalitionformation games.

7.4.2 Impact of a coalitional structure on solution concepts for canonical coalitional games

In canonical games, the solution concepts defined, such as the core, the Shapley value, and the nucleolus, assumed that the grand coalition would form because of the superadditivity property. In a coalition-formation game, the presence of a coalitional structure (a partition of \mathcal{N}) affects the definition and use of these concepts, as pointed out in [49] for a given static coalition-formation game. For instance, consider a TU coalitional game, in the presence of a static coalitional structure $\mathcal{B} = \{B_1, \dots, B_l\}$ (each B_i is a coalition). In this setting, the coalitional game can be defined as the triplet $(\mathcal{N}, v, \mathcal{B})$ in which v is a characteristic function.

In such coalition-formation games, the concept of group rationality can be replaced by the following concept:

DEFINITION 7.12 For a coalition-formation game $(\mathcal{N}, \mathbf{v}, \mathcal{B})$, an allocation vector $\mathbf{x} \in \mathbb{R}^{\mathcal{N}}$ is said to satisfy the relative efficiency property if and only if, for each coalition $B_k \in \mathcal{B}$, we have $\sum_{i \in B_k} x_i = \mathbf{v}(B_k)$.

Hence, relative efficiency implies that for every present coalition B_k in \mathcal{B} the total value available for coalition B_k is divided among its members, unlike in canonical games in which the value of the grand coalition $v(\mathcal{N})$ is distributed among all players (group rationality).

Given this definition, we investigate the impact of the presence of the coalitional structure \mathcal{B} on canonical solutions. First, we turn our attention to the Shapley value. For a coalition-formation game $(\mathcal{N}, v, \mathcal{B})$, the Shapley axioms defined for a canonical game remain valid, except for the efficiency axiom, which is replaced by a *relative efficiency* axiom. With this modified axiom, the Shapley value of $(\mathcal{N}, v, \mathcal{B})$, referred to as the \mathcal{B} -value, has the *restriction property* [49]. The restriction property implies that, for finding the \mathcal{B} -value, one can consider the *restricted* coalitional games $(\mathcal{B}_k, v|\mathcal{B}_k), \forall \mathcal{B}_k \in \mathcal{B}$, where $v|\mathcal{B}_k$ is the value v of the original game $(\mathcal{N}, v, \mathcal{B})$ defined over player set (coalition) \mathcal{B}_k . As a result, for finding the \mathcal{B} -value, we proceed in two steps, using the restriction property:

- 1. Consider the games $(B_k, v|B_k), k = 1, ..., l$, separately and, for each such game $(B_k, v|B_k)$, find the Shapley value using the canonical definition (7.26).
- 2. The *B*-value of the game is the $1 \times N$ vector ϕ of payoffs constructed by combining the resulting allocations of each restricted game $(B_k, v|B_k)$.

In the presence of a coalitional structure \mathcal{B} , the canonical definitions of the core and the nucleolus are also mainly modified by replacing group rationality with relative efficiency. However, unlike the Shapley value, the restriction property does not apply to the core, nor to the nucleolus. This can be easily deduced from the fact that both the core and the nucleolus depend on *all* coalitions of \mathcal{N} , i.e., all coalitions in the power set $2^{\mathcal{N}}$. Hence, in the presence of \mathcal{B} , the core and the nucleolus depend on the values of coalitions $B_j \in \mathcal{B}$ as well as the values of coalitions that are not in \mathcal{B} , i.e., coalitions $S \subset \mathcal{N}, \nexists B_k \in \mathcal{B}$ such that $B_k = S$. Hence, the problem of finding the core and the nucleolus of $(\mathcal{N}, v, \mathcal{B})$ is more complex than for the Shapley value. An approach for finding these solutions for 0-normalized games, i.e., games in which $v(\{i\}) = 0, \forall i \in \mathcal{N}$ is possible. The approach is based on finding a game equivalent to v by redefining the value, from which the core and nucleolus can be found for this equivalent game. For a detailed analysis, we refer the reader to [49, theorems 4 and 5].

Even though the analysis presented in this section is restricted to static coalitionformation games with TU and in characteristic form, it shows that finding solutions for coalition-formation games is not straightforward. The difficulty of such solutions increases whenever an NTU game, a partition-form game, or a dynamic coalitionformation game are considered, notably when the objective is to compute the solution in a distributed manner. For example, when considering a dynamic coalition-formation game, one would need to evaluate the payoff allocations *jointly* with the formation of the coalitional structure, and hence solution concepts become even more complex to find (although the restriction property of the Shapley value makes things easier). For this purpose, the solution concepts are either redefined or alternative concepts, specific to the game being studied, are introduced. Hence, unlike canonical games for which formal solutions exist, the solution of a coalition-formation game depends on the model and the objectives that are being considered.

7.4.3 Dynamic coalition-formation algorithms

In a coalition-formation game, the most important aspect is the formation of the coalitions in the game. While in static games these coalitions are already formed by an external factor, in dynamic coalition-formation games a challenging question is how to form a coalitional structure that is suitable to the studied game. In addition, the evolution of this structure is important, notably when changes to the game's nature can occur because of external or internal factors (e.g., if one or more players leave the game). In many applications, coalition formation entails either finding a coalitional structure that maximizes the total utility (social welfare) if the game is TU, or finding a structure with Pareto-optimal payoff distribution for the players if the game is NTU. For achieving such a goal, a *centralized* approach can be used; however, such an approach is generally NP-complete [465, 398, 41, 44]. The main reason is that finding an optimal partition requires iterating over all the partitions of the player set \mathcal{N} . The number of partitions of a set \mathcal{N} grows exponentially with the number of players in \mathcal{N} and is given by a value known as the Bell number [465]. For example, for a game where N has only 10 elements, the number of partitions that a centralized approach must go through is 115 975 (computed using the Bell number). Hence, finding an optimal partition from a centralized approach is, in general, computationally complex and impractical. In some cases, it may be possible to explore the properties of the game, notably of the value v, for reducing the centralized complexity. For example, if the value v of a TU game is concave, a procedure with computational time consumption polynomially bounded in the number of players in \mathcal{N} exists for finding the social-welfare-maximizing partition in a centralized manner.

However, in many coalition-formation games, v may not be concave, especially when there is a cost for coalition formation. Moreover, for utilizing coalition-formation games in practical applications, it is desirable that the coalition-formation process take place in a distributed manner, whereby the players have autonomy on the decision of whether or not they join a coalition. In fact, the complexity of the centralized approach as well as the need for distributed solutions has sparked a huge growth in the coalition-formation literature that aims to find low complexity and distributed algorithms for forming coalitions [398, 41, 44, 465]. The approaches used for distributed coalition formation are quite varied and range from heuristic approaches [465] and Markov-chain-based methods [398] to set-theory-based methods [41], as well as approaches that use bargaining theory or other negotiation techniques from economics [44]. Although there are no general rules for distributed coalition formation, some work, such as [41], provides generic rules that can be used to derive application-specific coalition-formation algorithms. Next, we will delve into the details of how to build a coalition-formation algorithm using these rules.

The merge-and-split algorithm

The main ingredients needed in order to construct a coalition-formation algorithm based on the generic rules given in [41] are three:

- Well-defined orders suitable for comparing *collections* of coalitions
- Two simple rules for forming or breaking coalitions
- Adequate notions for assessing the stability of a partition.

First, we present the following definitions:

DEFINITION 7.13 A collection of coalitions in the grand coalition \mathcal{N} , denoted \mathcal{S} , is defined as the set $\mathcal{S} = \{S_1, ..., S_l\}$ of mutually disjoint coalitions $S_i \subseteq \mathcal{N}$. In other words, a collection is any arbitrary group of disjoint coalitions S_i of \mathcal{N} not necessarily spanning all players of \mathcal{N} . If the collection spans all the players of \mathcal{N} , i.e., $\bigcup_{j=1}^{l} S_j = \mathcal{N}$, the collection is simply a partition of \mathcal{N} .

DEFINITION 7.14 *A* preference relation *or* comparison relation \triangleright *is an order defined* for comparing two collections $\mathcal{R} = \{R_1, ..., R_l\}$ and $\mathcal{S} = \{S_1, ..., S_p\}$ that are partitions of the same subset $\mathcal{A} \subseteq \mathcal{N}$ (i.e., same players in \mathcal{R} and \mathcal{S}). Thus, $\mathcal{R} \triangleright \mathcal{S}$ implies that the way \mathcal{R} partitions \mathcal{A} is preferred to the way \mathcal{S} partitions \mathcal{A} .

For comparing collections of coalitions, a number of preference relations can be defined. These can be divided into two categories: coalition-value orders and individual-value orders. Coalition-value orders compare two collections (or partitions) using the value of the coalitions inside these collections, such as in the *utilitarian order*. The utilitarian order, which is mainly suitable for TU games, states that a group of players prefers to organize themselves into a collection $\mathcal{R} = \{R_1, ..., R_k\}$ instead of $\mathcal{S} = \{S_1, ..., S_l\}$ if the total social welfare achieved in \mathcal{R} is strictly greater than in \mathcal{S} , i.e., $\sum_{i=1}^{k} v(R_i) > \sum_{i=1}^{l} v(S_i)$. Individual-value orders perform the comparison using the individual payoffs received by the players. One example of an individual-value order is the *Pareto order*, defined as follows:

DEFINITION 7.15 Given two payoff allocations \mathbf{x} and \mathbf{y} that are allotted by two collections \mathcal{R} and \mathcal{S} , respectively, to the same players, \mathcal{R} is preferred over \mathcal{S} by Pareto order if and only if $\mathbf{x} \ge \mathbf{y}$, with at least one element x_i of \mathbf{x} such that $x_i > y_i$.

Pareto order is an individual-value order in which a collection \mathcal{R} is preferred over another collection \mathcal{S} if at least one player improves in \mathcal{R} without hurting the other players. It must be noted that the Pareto order is suitable for both TU and NTU games.

Using these preference relations, two main rules for forming or breaking coalitions, referred to as *merge* and *split*, can be defined as follows [41]:

DEFINITION 7.16 (*Merge Rule*) Any set of coalitions $\{S_1, ..., S_l\}$ may be merged whenever the merged form is preferred by the players; i.e., where $\{\bigcup_{j=1}^l S_j\} \triangleright \{S_1, ..., S_l\}$, therefore, $\{S_1, ..., S_l\} \rightarrow \{\bigcup_{j=1}^l S_j\}$. DEFINITION 7.17 (**Split Rule**) Any coalition $\bigcup_{j=1}^{l} S_j$ may be split whenever a split form is preferred by the players; i.e., where $\{S_1, \ldots, S_l\} \triangleright \{\bigcup_{j=1}^{l} S_j\}$, thus, $\{\bigcup_{j=1}^{l} S_j\} \rightarrow \{S_1, \ldots, S_l\}$.

The basic idea behind the rules is that, given a set of players \mathcal{N} , any collection of disjoint coalitions $\{S_1, ..., S_l\}, S_i \subset \mathcal{N}$ can agree to merge into a single coalition $G = \bigcup_{i=1}^{l} S_i$, if this new coalition G is preferred by the players over the previous state, depending on the selected comparison order. Similarly, a coalition S splits into smaller coalitions if the resulting collection $\{S_1, \dots, S_l\}$ is preferred by the players over S. For example, when the Pareto order is selected, coalitions will merge only if at least one player can enhance its individual payoff through this merge without decreasing the other players' payoffs. Similarly, a coalition will split only if at least one player in that coalition is able to strictly improve its individual payoff through the split without hurting other players. A decision to merge or split is thus tied to the fact that all players must benefit; thus, any merged (or split) form is reached only if it allows all involved players to maintain their payoffs with that of at least one user improving. Hence, an algorithm based on the merge-and-split rules with Pareto order is a dynamic coalition-formation algorithm with partially reversible agreements, where the players enter into a binding agreement to form a coalition through the merge operation (if all players are able to improve their individual payoffs from the previous state) and can only split this coalition if splitting does not decrease the payoff to any coalition member (partial reversibility). Independent of the selected order or the starting partition of the network, any arbitrary sequence of these two rules is shown to converge to a final partition of \mathcal{N} . Consequently, one can build a dynamic coalition-formation algorithm using the merge-and-split rules, as the players in the game (or the users in a network) can interact to form the coalition and converge to a suitable network partition. Notice that each decision to merge or split can be taken in a distributed manner by each individual player or by each already-formed coalition without any reliance on a centralized entity.

Once a merge-and-split coalition-formation algorithm is constructed, the next step is to study the properties of the resulting partition. For instance, in order to assess the stability of any partition of \mathcal{N} , the concept of a *defection function* can be used. A defection function is defined as follows:

DEFINITION 7.18 A defection function \mathbb{D} is a function that associates with each partition S of \mathcal{N} a group of collections in \mathcal{N} . A partition $S = \{S_1, ..., S_l\}$ of \mathcal{N} is \mathbb{D} -stable if no group of players benefits from leaving \mathcal{N} when the players who leave can only form the collections allowed by \mathbb{D} .

By defining various types of defection functions, one can assess whether, in a given partition \mathcal{T} of \mathcal{N} , there is an incentive for the players to deviate and form other partitions or collections. One example of such functions is a weak equilibrium-like stability known as \mathbb{D}_{hp} stability, suitable for algorithms that rely on the merge-and-split rule. A \mathbb{D}_{hp} -stable partition simply implies that, in this partition, no group of players has an interest in performing a merge or a split operation. This type of stability can be thought of as

merge-and-split proof of a partition, or a kind of equilibrium with respect to merge-andsplit.

Another type of stability of importance to merge-and-split algorithms is \mathbb{D}_c -stability. A \mathbb{D}_c -stable partition presents numerous attractive properties:

- A D_c-stable partition is a *unique* outcome of any arbitrary merge-and-split iteration. Hence, starting from any given partition, one would always reach the D_c-stable partition by merge-and-split.
- Depending on the selected order, the players prefer the \mathbb{D}_c -stable partition over *all* other partitions. On the one hand, if the selected order is the utilitarian order, this implies that the \mathbb{D}_c -stable partition maximizes the social welfare (total utility); on the other hand, if the selected order is the Pareto order, the \mathbb{D}_c -stable partition has a Pareto-optimal payoff distribution for the players.
- No group of players in a D_c-stable partition has an incentive to leave this partition to form any other collection in N.

Nonetheless, the existence of a \mathbb{D}_c -stable partition is not always guaranteed. The \mathbb{D}_c -stable partition $S = \{S_1, ..., S_l\}$ of the whole space N exists if a partition of \mathcal{N} verifies two necessary and sufficient conditions:

- 1. For each $i \in \{1, ..., I\}$ and each pair of disjoint *coalitions* A and B, such that $\{A \cup B\} \subseteq S_i$, we have $\{A \cup B\} \triangleright \{A, B\}$.
- For the partition S = {S₁,..., S_l}, a coalition G ⊂ N formed of players belonging to different S_i ∈ S is S-incompatible if for no i ∈ {1,..., l} we have G ⊂ T_i. D_c stability requires that, for all S-incompatible coalitions, {G}[S] ▷ {G}, where {G}[S] = {G ∩ S_i ∀ i ∈ {1,..., l}} is the projection of coalition, G in partition S.

If no partition of \mathcal{N} can satisfy these conditions, then no \mathbb{D}_c -stable partition of \mathcal{N} exists and, in this case, using merge-and-split algorithms yields suboptimal \mathbb{D}_{hp} -stable partitions.

Although this section has presented the concept of a defection function for merge-andsplit algorithms, it should be noted that, depending on the application being investigated, one can possibly define different suitable defection functions, as this concept is not limited to a particular problem or algorithm. Moreover, other stability notions can also be defined for dynamic coalition-formation games, some of which are direct extensions of notions such as the core. The selection of a particular stability notion is applicationdependent.

Finally, by periodically taking distributed merge-and-split decisions, the players can adapt their coalitional structure to environmental changes such as the arrival of new players, the departure of other players, the change of position of players, or an increase or decrease in the utility.

Hedonic coalition-formation games

Coalition formation-games are diverse, and are by no means limited to the concepts in [41]. For example, a type of coalition-formation game, known as a *hedonic coalition-formation* game, has been widely studied in game theory. Hedonic games are interesting

because they allow the formation of coalitions (whether dynamic or static) based on the individual preferences of the players. In addition, these games admit different stability concepts that are extensions to well-known concepts such as the core or the Nash equilibrium used in a coalition-formation setting [77]. In this regard, hedonic games constitute a very useful analytical framework, which has a very strong potential to be adopted in modeling problems in wireless and communication networks (only a few contributions, including [406, 409] have used this framework in wireless communication problems).

Beyond merge-and-split and hedonic games, dynamic coalition-formation games encompass a multitude of algorithms and concepts such as those in [465, 398, 41, 44]. This chapter cannot provide an exhaustive survey of all such algorithms. Nonetheless, as will be seen later, many coalition-formation algorithms and concepts can be tailored and adapted for communication applications.

7.4.4 Sample applications in wireless and communication networks

In this section, we present a couple of sample applications of coalition-formation games in wireless and communication networks, and discuss some potential applications of this class of coalitional games.

Transmitter cooperation with cost in a TDMA system

The formation of virtual MIMO systems through distributed cooperation has received increasing attention recently (see [323, 411] and references therein). The problem involves a number of single-antenna users that cooperate and share their antennas in order to benefit from spatial diversity or multiplexing, and hence form a virtual MIMO system. Beyond the exhaustive research devoted to analyzing the link-level information-theoretical gains from distributed cooperation, or to assessing the stability of the grand coalition for cooperation with no cost, there is an interest in studying how a network of users can interact to form virtual MIMO systems, notably when there is a cost for cooperation. For this purpose, coalition-formation games are quite an appealing tool for performing this network-level analysis of virtual MIMO formation in wireless networks, as demonstrated in [411].

For instance, consider the model in [411] in which a network of single-antenna transmitters that send data in the uplink of a TDMA system to a receiver with multiple antennas is studied. In such a non-cooperative approach, each single-antenna transmitter sends its data in an allotted slot. To improve their capacity, the transmitters can interact form coalitions, whereby each coalition S is seen as a single-user MIMO that transmits in the slots that were previously held by the users of S. After cooperation, the TDMA system schedules one coalition per time slot. An illustration of the model is shown in Fig. 7.4. The use of coalitional games is highly suited to the study of network problems such as this one.

In this model, in order to cooperate, the transmitters must exchange data. This exchange of information incurs a cost in terms of power, which increases with distance inside the coalition as well as coalition size. This cost, as shown in [411], renders the game non-superadditive. For example, when two users are far away from one another, information



Fig. 7.4 System model for virtual MIMO-formation game.

exchange can consume the total available power, so, the utility received by the users (e.g., in terms of rate) when cooperating is smaller than in the non-cooperative case. Similarly, adding more users to a coalition does not always yield an increase in the utility; a coalition consisting of a large number of users increases the cost of information exchange, for example, so superadditivity cannot be guaranteed. As a consequence, for the game in [411] the grand coalition seldom forms¹⁰ and the game is modeled as a dynamic coalition-formation game between the transmitters (identified by the set \mathcal{N}) that seek to form cooperating coalitions. The dynamic aspect stems from the fact that many environmental changes, such as the mobility of the transmitters or the deployment of new users, may affect the coalitional structure that will form, and any algorithm must be able to cope with these changes accurately.

For this virtual MIMO coalition-formation game, the value function represents the sum-rate, or capacity, that the coalition can achieve, while taking into account the power cost. Owing to the TDMA nature of the problem, a power constraint \tilde{P} per time slot, and hence per coalition, is considered. Whenever a coalition forms, a fraction of \tilde{P} is used for information exchange, which constitutes a cost for cooperation, while the remaining fraction is used by the coalition to transmit its data, as a single-user MIMO, to the receiver. For a coalition *S*, the fraction used for information exchange is the sum of the powers that each user $i \in S$ needs to transmit its data to the user $j \in S$ that is farthest from *i*; because of the broadcast nature of the wireless channel, all other users in *S* can receive this data as well. This power cost scales with the number of users in the coalition, as well as with the distance between these users. Hence, the sum-rate that a coalition can achieve is limited by the fraction of power spent on information exchange. For instance,

¹⁰ In this game, the grand coalition only forms in extremely favorable cases, such as when the network contains only two users and these users are located quite close to one another.

if the power for information exchange for a coalition S is larger than \tilde{P} , then v(S) = 0. Otherwise, v(S) represents the sum-rate achieved by the coalition using the remaining fraction of power. Clearly the sum-rate is a transferable utility, so we are dealing with a TU game.

In this framework, a dynamic coalition-formation algorithm based on the merge-andsplit rules previously described can be built. Using [411], we start with a non-cooperative network, in which each user discovers its neighbors, starting with the closest, and attempts to merge based on the utilitarian order, i.e., if cooperating with a neighbor improves the total sum-rate that the involved users can achieve, then merging occurs (merging is accomplished through pairwise interactions between a user or coalition and a user or coalition in the vicinity). Furthermore, if a formed coalition finds that splitting into smaller coalitions improves the total utility achieved by its users, then a split occurs. Starting from the initial non-cooperative network, the coalition-formation algorithm involves sequential merge-and-split rules. Any coalition in the network can autonomously decide whether to perform a merge or split, based on the utility evaluation. Convergence is guaranteed by the definition of merge-and-split. Furthermore, if an optimal \mathbb{D}_c -stable partition exists, the studied algorithm converges to it. The existence in this model of a \mathbb{D}_c -stable partition cannot always be guaranteed, as it depends on random locations of the users; however, convergence, when it does exist, is guaranteed.

The coalition-formation algorithm proposed in [411] can handle any network size, as implementation is inherently distributed so that each coalition (or user) can detect the strength of the other users' uplink signals (using techniques as in ad hoc routing), and discover nearby candidate partners. Consequently the distributed users can exchange the required information and assess the kinds of merge or split decisions they can make. The transmitters engage in merge-and-split periodically, adapting the topology to environmental changes such as the mobility or joining/leaving of transmitters. The topology is always dynamically changing, through individual and distributed decisions by the network's coalitions. As the model is TU, one of several rules for dividing the coalition's value can be used, from well-known fairness criteria such as the proportional fair division to coalitional-game-specific rules such as the Shapley value or the nucleolus. Owing to the distributed nature of the problem, the nucleolus or the Shapley value are applied at the level of the coalitions that are forming or splitting. Although for the Shapley value this allocation coincides with the Shapley value of the whole game, as previously discussed, for the nucleolus the resulting allocations lie in the nucleolus of the restricted games only. In this game, for any coalition $S \subseteq N$ that forms through merge-and-split, the Shapley value presents a division of the payoff that takes into account the random order in which transmitters can join the coalition S (this division is also efficient at the coalition level and treats the players symmetrically within S). In contrast, the division by the nucleolus at the level of every coalition $S \subseteq \mathcal{N}$ that forms through merge-and-split ensures that the dissatisfaction of any transmitter within S is minimized by minimizing the excesses inside S. Finally, in this virtual MIMO game, one can use either the utilitarian order or the Pareto order for merge-and-split. Using the Pareto order, the model allows every user of the coalition to assess the improvement to its own payoff during merge or split, instead of relying on the entire coalitional value as in the utilitarian order [411]. Using the Pareto order, the fairness criteria chosen for payoff division would impact the network structure, so that, for each fairness type, one obtains a different topology.

Coalition-formation games for roadside-unit cooperation in vehicular networks

Recent advances in the integration of communication and sensor technologies have led to the emergence of intelligent transportation systems (ITS), which enable numerous attractive applications for road transportation systems (e.g., providing road traffic conditions, remote vehicle monitoring, accident prevention, payment services, and security applications) [374]. In order to support different ITS applications, both vehicle-to-roadside (V2R) and vehicle-to-vehicle (V2V) communications must be supported in vehicular networks. On the one hand, V2R communications allow vehicles to connect, through their on-board units (OBUs), to roadside units (RSUs) belonging to one or several service providers, in order to download (or upload) data related to a variety of applications. On the other hand, V2V communications enable a group of vehicles to exchange information for different purposes. The emergence of ITS applications raises numerous challenges for vehicular networks, such as advanced communication technologies for V2R and V2V communication, modeling of content-sharing through V2V cooperation, and analysis of non-cooperative data delivery in V2R communications (see [374] and references therein).

In this example, we tackle one challenge for vehicular networks: the design of cooperative strategies among RSUs. For instance, RSUs can cooperate in order to improve the diversity of the data circulating in the network, or to exploit the data-exchange capabilities of the underlying V2V networks. Instead of non-cooperatively sending information on the traffic in the same geographical location to their served vehicles, two RSUs can cooperate to send information on the traffic conditions at different locations, relying on V2V data exchange to disseminate this data to all the vehicles traveling between them. Thus, with an efficient V2V data-exchange protocol, all the vehicles moving between the two RSUs can acquire traffic information on various geographical areas without passing multiple RSUs. In doing so, the RSUs can obtain more revenue since they are providing more diverse information through cooperation. Furthermore, from the vehicles' perspective, because of the short time a vehicle spends at each RSU [374], a vehicle commonly has time to download only a limited number of data chunks or packets, e.g., related to a single class of data. By enabling cooperation among RSUs, vehicles could obtain more diverse data. The need for modeling cooperation in this scenario is a motivation for the use of coalitional-game theory as demonstrated in [415].

In [415], a network of RSUs is considered, with each having a number of data classes from which it can pick which data to send to its served vehicles. For each class of data there is a corresponding priority level that determines the importance of this data to the vehicles (and the RSUs). Because of the short period that a vehicle may spend at an RSU, each RSU selects only *one* class of data to transmit at a time to the vehicles in a given direction. From the chunks of data downloaded, the RSUs generate revenue by charging the vehicles a price proportional to the priority of the data.

Moreover, in the considered network, between each pair of RSUs there exists a V2V content-sharing scheme that allows vehicles to exchange data when possible. However,

when RSUs act in a non-cooperative way, as is often the case, they are not aware of the underlying V2V content-sharing network. Without coordination, the RSUs cannot estimate the fraction of vehicles that can potentially share content, nor the number moving in their direction, so they are unaware of the vehicle-to-vehicle content-sharing that can potentially occur. As a result, it is beneficial for a non-cooperative RSU to transmit data in the class with the highest priority for the vehicles passing by it, in order to maximize its revenue. This scheme is also the most beneficial for the vehicles, considering the time of entry of these vehicles into the network. Once a vehicle enters the network and meets its first RSU, the most beneficial action for this vehicle is to acquire the most important class of data from this RSU. In non-cooperative model, the utility for each RSU corresponds to the *total revenue* generated from the data transmitted to the vehicles. This utility is a function of the total data downloaded, the priorities of each data class, and the pricing scheme.

To increase their revenues, the RSUs can cooperate and exploit the underlying V2V content-sharing network. We consider that a pairwise content-sharing scheme exists between each pair of RSUs, i.e., between each pair of RSUs, a certain average number of *pairs of vehicles* can meet and engage in V2V content sharing. The average number of vehicles that can meet between each pair of RSUs depends on many factors, including the distance between the RSUs and the mobility of the vehicles. In order to exploit the V2V content-sharing scheme, RSUs can form *coalitions*, as explained in [415], using the following two-step cooperative protocol:

- 1. The RSUs exchange information on their estimates of the average number of vehicles they are serving, their distances, and so on.
- 2. The members of each cooperative group (i.e., coalition) coordinate the classes of data that each RSU needs to transmit to its served vehicles. The classes are agreed upon by the coalition members so as to maximize amount and diversity of data received by the vehicles through joint V2R and V2V communication. This coordination maps into agreeing on a selection of the classes of data that will be transmitted to the vehicles circulating between any two RSUs inside the same coalition.

Using this cooperative protocol, the RSUs in a vehicular network can increase the revenue from V2R communication with their served vehicles. A sample network structure is shown in Fig. 7.5 for five RSUs and three data classes (class 1 is the highest priority, class 2 is the second highest, and so on). In this figure, a number of RSUs with relatively high traffic among them form a cooperative coalition, while coordinating the classes of data transmitted so as to exploit V2V content-sharing.

In order to characterize the RSU network structure that will form, one can formulate a coalitional game whereby:

- The players are the RSUs.
- The value or utility function of a coalition captures the total revenue received by the RSUs from the downloaded data, as well as the cost due to synchronization and information exchange. Hence, the utility represents a total amount of money received by the RSUs, which is a transferable entity by definition.



Fig. 7.5 Example of coalition formation for cooperation among RSUs for a vehicular network with five RSUs and three data classes.

When a coalition of RSUs forms, the potential mutual benefit from cooperation depends highly on the traffic between the RSUs as well as on the average number of vehicles that can utilize V2V content sharing. Hence, cooperation might not be always beneficial. In addition, because of the cost of information exchange, which is function of the coalition size, the formation of a grand coalition is not guaranteed. Clearly the RSUs' coalitional game is a *coalition-formation game* with transferable utility.

In [415] a distributed algorithm for coalition formation based on a simple *switch operation* is considered. A switch operation allows each RSU to decide on leaving its current coalition and joining a new coalition, depending on the potential payoff as well as on the approval of the new coalition. That is, an RSU can break from a coalition and join another one if it can improve its payoff without decreasing the payoff to any member of the new coalition. It is shown that, using well-defined preferences, an algorithm based on switch operations converges to a Nash-stable network partition, i.e., a partition whereby no RSU prefers to leave its current coalition and join another. In this partition, each coalition of RSUs can utilize the two-step cooperative protocol to improve its performance.

Simulations offer insights into the application of the coalition-formation game to RSU cooperation in [415]. Numerical examples demonstrate that, if the number of vehicles that can potentially meet between two RSUs depends mainly on the distance, then closely located RSUs are more apt to form coalitions. Furthermore, it is shown that cooperation among RSUs can present gains, in terms of average revenue per RSU, of up to 33.2

percent relative to the non-cooperative case. An additional interesting result is that, in general, after coalition formation the resulting network of RSUs is mainly composed of a small number of relatively large coalitions rather than a large number of small coalitions.

In a nutshell, this example demonstrates that coalition-formation games constitute an important analytical framework for the study of vehicular networks. Although this example focuses on RSU cooperation while abstracting the details of the V2V protocol, extensions can investigate the use of this coalition-formation model in conjunction with realistic wireless channels and practical V2V protocols. Further extensions can look at joint coalition formation among RSUs and vehicles (for an insight on vehicles-coalition formation, readers are referred to [371]).

Other sample applications of coalition-formation games

Potential applications of coalition-formation games in wireless and communication networks are numerous and diverse. Coalition-formation games have been applied to cognitive-radio networks for collaborative sensing, as will be discussed in Chapter 13. They have also been applied (see [407]) to improving the physical-layer security of wireless nodes through cooperation among transmitters. In [406] coalition formation among autonomous agents, such as unmanned aerial vehicles, is studied in the context of data collection and transmission in wireless networks. Recently, there has been a significant increase in interest in autonomic communication systems, networks that are self-configuring, self-organizing, self-optimizing, and self-protecting. In such networks, the users should be able to learn and adapt to their environment (changes in topology, technologies, service demands, application context, etc.), thus providing much-needed flexibility and functional scalability. Coalition-formation games present an adequate framework for the modeling and analysis of these self-organizing next-generation communication networks. Potential applications of coalition-formation games encompass cooperative networks, peer-to-peer networks, delay-tolerant networks, wireless sensor networks, next-generation IP networks, ad hoc self-configuring networks, and many others. In general, whenever there is a need for distributed algorithms for autonomic networks, coalition formation is a strong tool for modeling such problems. Any study of cooperative wireless nodes' behavior when a cost is present is a candidate for modeling using coalition-formation games.

Finally, although the main applications of coalition-formation games explored in this book require a characteristic form, coalition-formation games in partition form are of major interest and are ripe for application in wireless and communication networks (e.g., see [408] for insights into the application of coalition-formation games in partition form to cooperation in peer-to-peer networks).

7.5 Class III: coalitional graph games

7.5.1 Main properties of coalitional graph games

In canonical and coalition-formation games, the utility or value of a coalition does *not* depend on how the players are interconnected within the coalition. However, in

many scenarios, the *underlying communication structure* between the players in a coalitional game can have a major impact on the utility and other characteristics of the game [346, 205]. By the underlying communication structure, we imply a graph structure representing the connectivity of the players among one another, i.e., which player communicates with which one inside every coalition. We provided examples of such interconnections in Section 7.2.2 and Fig. 7.3(b). In general, a coalitional graph game is in graph form, and can be TU or NTU with the possibility that the value of a coalition depends on the external network structure, as explained in Section 7.2.2.

In coalitional graph games, the main theme is the presence of a graph of the communication between the players. Typically, there are two objectives in coalitional graph games. The first and most important objective is to derive low-complexity distributed algorithms for players who wish to build a network *graph* (directed or undirected) and not just coalitional groups as in coalition-formation games. A second objective is to study the properties (stability, efficiency, etc) of the formed network graph, which are inherently different from those of a coalitional structure because of the presence of the graph. In some scenarios, the network graph is given, and hence analyzing its stability and efficiency is the only goal of the game. The following sections provide an in-depth study of coalitional graph games.

7.5.2 Coalitional graph games and network-formation games

The idea of a value which is dependent on a graph of communication between players was introduced by Myerson [346], through the graph-function form for TU games. To define the graph-function form of a game, a TU canonical coalitional game (\mathcal{N}, v) is considered along with an undirected graph G that interconnects the players in the game. In this setting, a fair solution can be found by defining a new value function u that depends on the graph. To evaluate the value u of a coalition S, the coalition is divided into smaller coalitions depending on the players connected through S. For example, given a three-player coalition $S = \{1, 2, 3\}$ and a graph $G = \{(2, 3)\}$ (only players 2) and 3 are connected by a link in G), the value u(S,G) is $v(\{2,3\}) + v(\{1\})$, where v is the original value of the canonical game. Using the new value u, an axiomatic approach, similar to the Shapley value, is provided by Myerson for solving the game in graph-function form. In this context, the so-called Myerson value, defined as the Shapley value of the game (\mathcal{N}, u) , where u is the newly defined value, presents a fair solution of the canonical game (\mathcal{N}, v) in the presence of a graph structure G. The main drawback to using the Myerson value as a solution concept for coalitional graph games is that the value u of a coalition depends solely on the connected players in the coalition, with no dependence on the structure. For example, for both graphs G_5^1 and G_5^2 in Fig. 7.3(b), the values u are equal, although the individual payoffs received by the players in G_5^1 and G_5^2 through the Myerson value allocation would be different because of the different graphs.

Nevertheless, the Myerson value is a cornerstone for coalitional graph solutions, and it has resulted in several extensions. Notably, the value function as presented by Myerson can be extended so as to depend on the graph structure, and not only on the connected

components. In this regard, a general definition for a coalitional graph game, with a general value, is as follows:

DEFINITION 7.19 Consider a coalitional graph game (\mathcal{N}, v) and denote by $\mathcal{G}(S)$ the set of all possible graphs with vertices being the players in any coalition $S \subseteq \mathcal{N}$. In this setting, for any coalition $S \subseteq \mathcal{N}$, connected by any graph $G_S \in \mathcal{G}(S)$, the value v is defined as the set $v(S, G_S) \subset \mathbb{R}^S$. If the game is TU, then $v(S, G_S) \in \mathbb{R}$ is a real number representing the worth of coalition S when connected by graph G_S .

Using this definition for the value, coalitional graph games offer a framework that is richer than the Myerson value but is more complex to solve. Furthermore, while the Myerson value was proposed in [346] as a solution to a game in which the graph is *given*, coalitional graph games also support games in which the *formation of the graph* is a key issue. One prominent tool in this area is non-cooperative game theory, which was extensively used for forming the network graph. For instance, in [347, Section 9.5], using the Myerson framework of [346], an extensive-form game is proposed for forming the network graph. However, the extensive-form approach is impractical in many situations, as it requires listing all possible links in the graph, which is a complex combinatorial problem.

Nonetheless, following this work, a new breed of games started to appear known as *network-formation games*. The main objective in these games is to study the interactions within a group of players that wish to form a graph. Although in some references these games are decoupled from coalitional-game theory, we discuss them here for two reasons:

- The basis of all network-formation games is the work on coalitional graph games, starting in [346].
- Network-formation games share many aspects with coalitional graph games, such as the ideas of a value and an allocation rule, and the need for stability. The solutions of network-formation games are quite correlated with those of coalition-formation games (in terms of forming the graph) and canonical games (in terms of having stable allocations).

Network-formation games can be thought of as hybrids of coalitional graph games and non-cooperative games because, in forming the network, non-cooperative game theory plays a prominent role. In network-formation games one needs to form a network graph as well as to ensure the stability of this graph, using concepts analogous to those in canonical coalitional games. For forming the graph, a broad range of approaches exists, grouped into two types: *myopic* and *far-sighted*.¹¹ In myopic approaches, players choose their strategies given the current state of the network, while in far-sighted algorithms, players adapt their strategy by learning and by predicting the future strategies of other players. In both approaches, well-known concepts from non-cooperative game theory are used. The most popular approach is to consider the network formation as a non-zero-sum,

¹¹ These approaches are sometimes referred to as dynamics of network formation, for example in the terminology of [228].

non-cooperative game in which a player's strategy is to select one or more links to form or break. One approach to solving the game is to play *myopic best-response dynamics*, whereby each player selects the strategy, i.e., the link(s) to form or break, that maximizes its utility. Under certain conditions on the utilities, the best-response dynamics converge to a Nash equilibrium, which constitutes a Nash network. These approaches are widespread in network-formation games [127, 61, 129], and several refinements to the Nash equilibrium suitable for network formation are used [127, 61, 129]. The main drawback of aiming for a Nash network is that, in many network-formation games, the Nash networks are trivial graphs, such as the empty graph, or can be inefficient. For these reasons, a new type of network-formation game has been developed, which utilizes new concepts for stability such as *pairwise stability* and *coalitional stability* [228]. The basic idea is to present stability notions that depend on deviations by a group of players instead of the unilateral deviations allowed by the Nash equilibrium. In the following, given an undirected graph G and a link between two players $i, j \in \mathcal{N}$, the notation G + ij indicates that link ij is added to the graph G, while G - ij indicates that link ij is removed from the graph G. The concept of *pairwise stability* is defined as follows:

DEFINITION 7.20 Given a coalitional graph game $(\mathcal{N}, \mathbf{v})$, an undirected graph G over \mathcal{N} , and a payoff allocation $\mathbf{x}(G) \in \mathbb{R}^{\mathcal{N}}$ (the payoffs in a coalitional graph game depend on both the coalitional value \mathbf{v} and the graph G in place), the undirected graph G is said to be pairwise stable with respect to the payoff allocation $\mathbf{x}(G)$ if (1) for all links $ij \in G$, $x_i(G) \ge x_i(G - ij)$ and $x_j(G) \ge x_j(G - ij)$, and (2) for all $ij \notin G$, if $x_i(G + ij) > x_i(G)$ then $x_j(G + ij) < x_j(G - ij)$.

The concept of pairwise stability considers an undirected graph, but it can be easily extended to directed graphs. It can be also applied to subgraphs (e.g., graphs on coalitions $S \subseteq N$). Although pairwise stability is an appealing concept because of its simplicity, in many games no network that is pairwise-stable exists. Moreover, as its name implies, pairwise stability only considers deviations by a pair of players in the network-formation game. In contrast, coalitional (strong) stability allows deviations by groups of players. The concept of strong stability is defined as follows:

DEFINITION 7.21 Given a coalitional graph game (\mathcal{N}, v) , and any graph G over \mathcal{N} , a graph G' is said to be obtainable from graph G via deviations by a coalition $S \subseteq \mathcal{N}$ if (1) ij $\in G'$ and ij $\notin G$ implies ij $\subset S$, and (2) ij $\in G$ and ij $\notin G'$ implies ij $\cap S = \emptyset$.

DEFINITION 7.22 Consider a coalitional graph game (\mathcal{N}, v) and define a graph G over \mathcal{N} as well as a payoff allocation $\mathbf{x}(G) \in \mathbb{R}^{\mathcal{N}}$. Then, the graph G is said to be coalitional stable (or strongly stable) with respect to $\mathbf{x}(G)$ if, for any coalition $S \subset \mathcal{N}$, graph G' obtainable from graph G via deviations by coalition S, and $i \in S$ such that $x_i(G') > x_i(G)$, there exists $j \in S$ such that $x_j(G') < x_j(G)$.

The concept of coalitional stability is a strong refinement of the pairwise stability concept, but it is a very demanding concept.

Independent of the stability concept, a key design issue in network-formation games is the tradeoff between stability and efficiency. It is desirable to devise algorithms for forming stable networks that can also be efficient in terms of payoff distribution or total social welfare. Several approaches for devising such algorithms exist, notably using stochastic processes, graph-theoretical techniques, or non-cooperative games. For a comprehensive survey of such algorithms, we refer the reader to [228].

Finally, the Myerson value and network-formation games are not the only approaches for solving coalitional graph games. Other approaches, closely tied to canonical games, have been proposed. Herings, van der Laan, and Talman [205] formulate a canonical game-like model for an NTU game, whereby the graph structure is taken into account. The authors propose an extension to the core called the *balanced core*, which takes into account the graph structure. Furthermore, under certain conditions, analogous to the balanced conditions of canonical games, they show that this balanced core is non-empty. Hence, coalitional graph games constitute quite a rich analytical framework, with a broad range of applications.

7.5.3 Sample applications in wireless and communication networks

The need to analyze network architectures, routing, and graph structures in the context of wireless communications and networking has led to the emergence of interesting applications of coalitional graph games. In this section, we survey a few sample applications.

In next-generation wireless networks such as LTE-Advanced and the most recent WiMAX standard, the IEEE 802.16j, a new node, the relay station, has been introduced in order to improve network capacity and coverage [7, 11]. This impacts the network architecture of next-generation networks, which will be governed by a tree architecture connecting a base station to subordinate relay stations. The deployment of relay stations leads to new and important design challenges, such as devising distributed algorithms for building the network's architecture. In this context, network-formation games, as will be seen in Chapter 9, can be an important tool in studying the architecture and structure of next-generation LTE-Advanced or IEEE 802.16j networks in the presence of three types of network nodes: mobile stations, base stations, and relay stations.

In fact, the presence of network graphs is ubiquitous in many wireless and communication applications. In designing, understanding, and analyzing such graphs, coalitional graph games are an accurate tool. Using concepts such as network formation, stability, fairness, and others, one can model a wide range of problems. For instance, networkformation games have been widely used in routing problems. For example, in [43], a stochastic approach to network formation is proposed, in which a network of nodes wishing to form a graph for routing traffic among themselves is considered. Each node aims to minimize its costs, (e.g., for routing, link maintenance, disconnection, etc.). For network formation, a myopic dynamic best-response algorithm is proposed. Each round of this algorithm begins by randomly selecting a pair of nodes *i* and *j* in the network. The algorithm proceeds in two steps. In the first step, if the link (i, j) is already formed in the network, node *i* is allowed to break this link, while in the second step node *i* is allowed to form a new link with a certain node k, if k accepts the formation of the link (i, k). The benefit from the link (i, j) can be seen as a kind of cost sharing between nodes i and j. Using a stochastic-process approach, the authors show that the myopic algorithm always converges to a pairwise stable and efficient tree network. Under certain conditions on the cost function, this network is a simple star network. Efficiency is measured in terms of Pareto optimality of the utilities, as the game is inherently NTU. Although the network-formation algorithm in [43] converges to a stable and efficient network, it suffers from a major drawback, which is the slow convergence time, notably for large networks. The algorithm is mainly intended for undirected graphs, although the authors provide insights on how this work can be extended to directed graphs.

The use of network-formation games in routing applications is not restricted to network formation, but can be used for studying existing networks. In [234], the authors study the stability and flow of traffic in a given directed graph. Several concepts from network-formation games, including pairwise stability, are used. In addition the authors generalize the concept of pairwise stability, making it suitable for directed graphs. Non-cooperative game theory is used to determine the network flows at different nodes while taking into account the stability of the network graph.

Applications of coalitional graph games are not limited to routing problems. In fact, the main potential of this class of games lies in solving problems beyond network routing. For instance, coalitional graph games are suitable tools for analyzing information trust management in wireless networks, multi-hop cognitive radio, relay selection in cooperative communication, intrusion detection, peer-to-peer data transfer, multi-hop relaying, packet forwarding in sensor networks, and many other applications.

7.6 Summary

Cooperative game theory provides a variety of tools useful in many applications. From bargaining theory to coalitional games, one can identify numerous analytical tools suited to modeling different aspects of wireless and communication networks. While bargaining theory is useful in the allocation and distribution of resources among nodes in a wireless or wired network, coalitional-game theory allows us to model a broad range of problems, including cooperative behavior, fairness in cooperation, network formation, cooperative strategies, and incentives for cooperation.

In this chapter, we have covered the key elements of cooperative game theory from an engineering and communications perspective. In order to illustrate the application of cooperative games in a wireless and communication context, we have studied a number of representative applications drawn from state-of-the-art research, including rate and resource allocation, virtual MIMO systems, cooperation in vehicular networks, distributed cooperation, and network formation in next-generation networks. Auction theory is an applied branch of game theory that deals with how people act in auction markets, and it studies the game-theoretic properties of auction markets. There are many possible designs (or sets of rules) for an auction, and typical issues studied by auction theorists include the efficiency of a given auction design, optimal and equilibrium bidding strategies, and revenue comparison. Auction theory is also used as a tool to inform the design of real-world auctions, most notably auctions for the privatization of public-sector companies or the sale of licenses for use of the electromagnetic spectrum.

Mechanism design is a subfield of game theory studying solution concepts for a class of private-information games. The distinguishing features of these games are as follows. First, a game "designer" chooses the game structure rather than inheriting one. Thus, the mechanism design is often called "reverse game theory." Second, the designer is interested in the game's outcome. Such a game is called a "game of mechanism design" and is usually solved by motivating players to disclose their private information. The 2007 Nobel Memorial Prize in Economic Sciences was awarded to Leonid Hurwicz, Eric Maskin, and Roger Myerson "for having laid the foundations of mechanism design theory."

Auction theory and mechanism design are important for communications and networking for practical, empirical, and theoretical reasons. First, a large number of wireless networking and resource-allocation problems can be formulated within the framework of auction theory – for example, the routing problem for self-interested users [39]. Second, auction theory has a simple game setup, and many theoretical results are available for analysis. Third, the Federal Communications Commission (FCC) spectrum auction has garnered a significant amount of money. There are 51 major trading areas (MTAs), with a 30 MHz spectrum per MTA, and 492 basic trading areas (BTAs), each with one 30 MHz and four 10 MHz blocks. So there are $51 \times 1 + 492 \times (1 + 4) = 2511$ items. From 1994 to 2001, the total revenue was more than \$40 billion.

In this chapter, we first discuss the basics of auction theory. Second, we introduce mechanism design to show how to control the game outcome by cleverly designing the game rules. Then, we discuss some typical types of auctions. Finally, some communications and networking applications of auction theory and mechanism design are described.

8.1 Introduction and auction basics

In this section, we first introduce the concept of an auction. Then, we present the auction within a game-theoretic model. Some basic properties are also discussed. Finally, we list several basic types of auctions, followed by a simple example.

Auctions take many forms but always satisfy two conditions. First, they can be used to sell any item, so they are universal; second, the outcome of the auction does not depend on the identity of the bidders, i.e., auctions are anonymous. Most auctions have the feature that participants submit bids, or the amounts of money they are willing to pay. Following is the definition of an auction.

DEFINITION 8.1 An auction is a market mechanism in which an object, service, or set of objects, is exchanged on the basis of bids submitted by participants. It provides a specific set of rules that will govern the sale or purchase (procurement auction) of an object to the submitter of the most favorable bid. Specific mechanisms include first-price, second-price, English, and Dutch auctions.

A game-theoretic auction model is a mathematical game represented by a set of players, a set of actions (strategies) available to each player, and a payoff vector corresponding to each combination of strategies. Generally, the players are the buyer(s) and the seller(s). The action set of each player is a set of bid functions or reservation prices. Each bid function maps the player's value (in the case of a buyer) or cost (in the case of a seller) to a bid price. The payoff for each player under a combination of strategies is the expected utility (or expected profit) for that player under that combination of strategies.

Game-theoretic models of auctions and strategic bidding generally fall into one of two categories. In a private-value model, each participant (bidder) assumes that each of the competing bidders obtains a random private value from a probability distribution. In a common-value model, each participant assumes that any other participant obtains a random signal from a probability distribution common to all bidders. Usually, but not always, a private-value model assumes that the values are independent across bidders, whereas a common-value model usually assumes that the values are independent up to the common parameters of the probability distribution. When it is necessary to make explicit assumptions about bidders' value distributions, most of the published research assumes symmetric bidders. This means that the probability distribution from which the bidders obtain their values (or signals) are identical across bidders. In a privatevalue model that assumes independence, symmetry implies that the bidders' values are independently and identically distributed (i.i.d.).

There are various properties for an auction. First, allocative efficiency means that in all such auctions the highest bidder always wins (i.e., there are no reserve prices). Second, it is desirable for an auction to be computationally efficient. Finally, to study the revenue (expected selling price) of different auctions, we have the following theorem:

THEOREM 8.1 (*Revenue Equivalence Theorem*) Any two auctions with the following properties

- the bidder with the highest value wins
- the bidder with the lowest value expects zero profit

- bidders are risk-neutral ¹
- value distributions are strictly increasing and atomless

have the same revenue and also the same expected profit for each bidder.

It is worth mentioning the phenomenon of the winner's curse, which can occur in common-value settings when the actual values to the different bidders are unknown but correlated, and the bidders make bidding decisions based on estimated values. In such cases, the winner will tend to be the bidder with the highest estimate, and that winner will frequently have bid too much for the auctioned item.

There are many ways to categorize auctions. For example, standard auctions require that the winner of the auction be the participant with the highest bid. A non-standard auction (e.g., a lottery) does not require this. There are traditionally four types of auctions that are used for the allocation of a single item, as follows.

A *first-price auction* is an auction in which the bidder who submitted the highest bid is awarded the object being sold and pays a price equal to the amount of the bid. Alternatively, in a procurement auction the winner is the bidder who submits the lowest bid and is paid an amount equal to that bid. In practice, first-price auctions are either sealed-bid, in which bidders submit bids simultaneously, or Dutch. In first-price auctions, bidders shade their bids below their true value.

A *second-price auction* is an auction in which the bidder who submitted the highest bid is awarded the object being sold and pays a price equal to the second-highest amount bid. Alternatively, in a procurement auction the winner is the bidder who submits the lowest bid, and is paid an amount equal to the next-lowest submitted bid. In practice, second-price auctions are either sealed-bid, in which bidders submit bids simultaneously, or English, in which bidders continue to raise each other's bids until only one bidder remains. The theoretical nicety of second-price auctions, first pointed out by William Vickrey, is that bidding one's true value is a dominant strategy. Alternatively, firstprice auctions award the object to the highest bidder, but the payment is equal to the amount bid.

An *English (or open ascending-bid) auction* is a type of sequential second-price auction in which the auctioneer directs participants to beat the current standing bid. New bids must increase the current bid by a predefined increment. The auction ends when no participant is willing to outbid the current standing bid. Then, the participant who placed the current bid is the winner and pays the amount bid. An English auction, in which the highest bidder pays the amount bid, is termed a second-price auction since the winning bidder need only outbid the next-highest bidder by the minimum increment. Thus, the winner effectively pays an amount equal to (or slightly higher than) the second-highest bid.

A Dutch (or open descending-bid) auction is a type of first-price auction in which a "clock" initially indicates a price for the object for sale which is substantially higher

¹ In economics, risk-neutral behavior is in between risk aversion and risk seeking. If offered either \$50 or a 50 percent chance of \$100, a risk-averse person will take the \$50, a risk-seeking person will take the 50 percent chance of \$100, and a risk-neutral person will have no preference between the two options.

than any bidder is likely to pay. Then, the clock gradually decreases the price until a bidder "buzzes in" or indicates a willingness to pay. The auction is then concluded, and the winning bidder pays the amount reflected on the clock at the time the process was stopped. These auctions are named after a common market mechanism for selling flowers in Holland, but they also mirrored in stores successively reducing prices on sale items.

Most auction theory revolves around the above four "standard" auction types. However, other auction types have also received some academic study.

A *Japanese auction* is a type of sequential second-price auction, similar to an English auction, in which the auctioneer regularly raises the current price. Participants must signal at every price level their willingness to stay in the auction and pay the current price. Thus, unlike an English auction, each participant must bid at each level to stay in the auction. The auction concludes when only one bidder indicates his willingness to stay in. This auction format is also known as a button auction.

In an *all-pay auction*, bidders place their bids in sealed envelopes and simultaneously hand them to the auctioneer. The envelopes are opened, and the individual with the highest bid wins, paying a price equal to the exact amount bid. All losing bidders are also required to make a payment to the auctioneer equal to their own bid. This auction format is nonstandard, but it can be used to understand things such as election campaigns (in which bids can be interpreted as campaign spending) and queuing for a scarce commodity (in which your bid is the amount of time that you are prepared to remain in the queue). The most straightforward form of an all-pay auction is a Tullock auction, sometimes called a Tullock lottery, in which everyone submits a bid but both the losers and the winners pay their submitted bids. This is instrumental in describing certain ideas in public-choice economics. A dollar auction is a two-player Tullock auction, or a multi-player game, in which only the two highest bidders pay their bids. Other forms of all-pay auctions exist, such as the war of attrition, in which the highest bidder wins, but all (or both, more typically) bidders pay only the lower bid. The war of attrition is used by biologists to model conventional contests, or agonistic interactions resolved without recourse to physical aggression.

A *unique-bid auction* is a type of strategy game related to traditional auctions, in which the winner is usually the individual with the lowest unique bid, although less commonly the auction rules may specify that the highest unique bid is the winner. Unique-bid auctions are often used as a form of competition or lottery.

A generalized second-prize (GSP) auction is a non-truthful auction mechanism for multiple items. First thought of as a natural extension of the Vickrey auction [257], it actually does not conserve some good properties of the Vickrey auction (such as truthfulness, for example). It is used mainly in the context of keyword auctions, in which sponsored search slots are sold on an auction basis.

Finally, we study a simple first-price auction model with two buyers who are bidding for an object. Each buyer might assume that the rival buyer's private value is drawn from the uniform distribution over the interval [0, 1], with the cumulative distribution function² F(v) = v. We assume that (i) the value of the object for the seller is 0,

² Since F is symmetric between the two buyers, this is an auction model with symmetric bidders.

and (ii) the seller's reservation price is also 0. Each buyer's expected utility U can be written as

$$U(p) = (v - p)Pr[p > B(v_o)], \qquad (8.1)$$

where *p* is the bid price, (v - p) is the consumer surplus that the buyer will receive, conditional upon winning, and $Pr[p > B(v_o)]$ is the likelihood that he or she will be the buyer with the highest bid price. That likelihood is given by the probability that this buyer's bid price *p* exceeds the other buyer's bid price *B* (expressed as a function of the other buyer's value v_o).

Assume that each buyer's equilibrium bid price is monotonically increasing in that buyer's value; this implies that the bid function *B* has an inverse function. Let *Y* be the inverse of *B*: $Y = B^{-1}$. Then $U(p) = (v - p)Pr[Y(p) > v_o]$. Since v_o is distributed according to $F(v_o)$, we have

$$Pr[Y(p) > v_o] = F(Y(p)) = Y(p),$$
 (8.2)

which implies that U(p) = (v - p)Y(p). A bid price p maximizes U if U'(p) = 0. Differentiating U with respect to p and setting to zero, we have

$$U'(p) = -Y(p) + (v - p)Y'(p) = 0.$$
(8.3)

Since the buyers are symmetric, in equilibrium it must be the case that p = B(v) or (equivalently) Y(p) = v. Therefore, we have

$$-Y(p) + (Y(p) - p)Y'(p) = 0.$$
(8.4)

A solution \hat{Y} of this differential equation is an inverse Nash equilibrium strategy of this game.

At this point, we may conjecture that the (unique) solution is the linear function $\hat{Y}(p) = \alpha p$ and $\hat{Y}'(p) = \alpha$ for some real number α . Substituting into U'(p) = 0, we have

$$-\alpha p + (\alpha p - p)\alpha = 0. \tag{8.5}$$

Solving for α yields $\hat{\alpha} = 2$. Therefore, $\hat{Y}(p) = 2p$ satisfies U'(p) = 0. $\hat{Y}(p) = \hat{\alpha}p$ implies $\hat{Y}(p)/\hat{\alpha} = p$, or $v/\hat{\alpha} = \hat{B}(v)$. Thus, the (unique) Nash equilibrium strategy bidding function of this game is established as $\hat{B}(v) = v/2$, at least within the set of invertible bidding functions.

8.2 Mechanism design

A game designer tries to consider all possible games and choose the one that best influences other players' selections. Mechanism design is used to define the game rules so as to achieve the desired outcome. This is different from game analysis, in which the game rules are predefined and then the outcome is investigated. In addition, the game designer has to consider the situation in which players may lie. Fortunately, by the revelation principle, it is only necessary to consider games in which players truthfully report their private information. In this section, we discuss mechanism design in some detail.

First, let us introduce the following ingredients:

- Outcome set: Ω
- Players *i* ∈ *I*, where *I* is the set of players. The set has size |*I*| = *N*, with preference types θ_i ∈ Θ_i
- Utility $u_i(o, \theta_i)$, over outcome $o \in \Omega$
- Mechanism M = (S, g), which defines
 - a strategy space $S^N = S_1 \times \cdots \times S_N$, s.t. player *i* chooses a strategy $s_i(\theta_i) \in S_i$ with $s_i : \Theta_i \to S_i$
 - an outcome function $g: S^N \to \Omega$, s.t. outcome $g(s_i(\theta_1), \dots, s_N(\theta_N))$ is implemented given strategy profile $s = (s_1(\cdot), \dots, s_N(\cdot))$
- Game: The utility to player *i* from strategy profile *s* is $u_i(g(s(\theta)), \theta_i)$, denoted $u_i(s, \theta_i)$.

The objective of a mechanism M = (S, g) is to achieve the desired game outcome $f(\theta)$ such that

$$g(s_1^*(\theta_1), \dots, s_N^*(\theta_N)) = f(\theta), \ \forall \theta \in \Theta^N$$
(8.6)

for an equilibrium strategy $(s_1^*, ..., s_N^*)$. The desired properties of a mechanism are:

- Efficiency. Select the outcome that maximizes total utility.
- Fairness. Select the outcome that achieves a particular fairness criterion for utility.
- **Revenue maximization.** Select the outcome that maximizes revenue to a seller (or more generally, utility to one of the players).
- Budget balance. Implement outcomes that have balanced transfers across players.
- **Pareto optimality.** Only implement outcomes o^* for which, for all $o' \neq o^*$, either $u_i(o'; \theta_i) = u_i(o^*; \theta_i)$ for all *i* or $\exists i \in \mathcal{I}$ with $u_i(o', \theta_i) < u_i(o^*, \theta_i)$.

In the following, we first discuss several design concepts, then explain the revelation principle, and finally explain the concepts of impossibility and possibility. The Groves mechanism is studied as an example.

8.2.1 Equilibrium concepts

We define three types of mechanism: Nash implementation, Bayes–Nash implementation, and dominant implementation, in order of increasing constraints and difficulty. DEFINITION 8.2 Nash implementation mechanism M = (S,g) implements $f(\theta)$ in a Nash equilibrium if, for all $\theta \in \Theta$, $g(s^*(\theta)) = f(\theta)$, where $s^*(\theta)$ is a Nash equilibrium; i.e.,

$$u_i(s_i^*(\theta_i), s_{-i}^*(\theta_{-i}), \theta_i) \ge u_i(s_i'(\theta_i), s_{-i}'(\theta_{-i}), \theta_i), \ \forall i, \forall \theta_i, \forall s_i' \neq s_i^*.$$
(8.7)

DEFINITION 8.3 Bayes–Nash implementation with common prior $F(\theta)$, mechanism M = (S, g) implements $f(\theta)$ in a Bayes–Nash equilibrium if, for all $\theta \in \Theta$, $g(s^*(\theta)) = f(\theta)$, where $s^*(\theta)$ is a Bayes–Nash equilibrium; i.e.,

$$E_{\theta_{-i}}[u_i(s_i^*(\theta_i), s_{-i}^*(\theta_{-i}), \theta_i)] \ge E_{\theta_{-i}}[u_i(s_i'(\theta_i), s_{-i}'(\theta_{-i}), \theta_i)], \ \forall i, \forall \theta_i, \forall s_i' \neq s_i^*.$$
(8.8)

DEFINITION 8.4 Dominant implementation mechanism M = (S, g) implements $f(\theta)$ in a dominant-strategy equilibrium if, for all $\theta \in \Theta$, $g(s^*(\theta)) = f(\theta)$, where $s^*(\theta)$ is a dominant-strategy equilibrium; i.e.,

$$u_i(s_i^*(\theta_i), s_{-i}^*(\hat{\theta}_{-i}), \theta_i) \ge u_i(s_i'(\theta_i), s_{-i}'(\hat{\theta}_{-i}), \theta_i), \ \forall i, \forall \theta_i, \forall \hat{\theta}_{-i}, \forall s_i' \neq s_i^*.$$
(8.9)

8.2.2 Participation and incentive compatibility

Next, we define three rationality concepts: *ex ante* individual rationality, interim individual rationality, and *ex post* individual rationality, in order of increasing constraints and difficulty.

Let $\bar{u}_i(\theta_i)$ denote the (expected) utility to player *i* with type θ_i as its outside option, and recall that $u_i(f(\theta); \theta_i)$ is the equilibrium utility for player *i* from the mechanism. We state the definitions of three types of rationality:

• *Ex ante* individual rationality: players choose to participate before they know their own types:

$$E_{\theta \in \Theta}[u_i(f(\theta); \theta_i)] \ge E_{\theta_i \in \Theta_i}[\bar{u}_i(\theta_i)].$$
(8.10)

• Interim individual rationality: players can withdraw once they know their own type:

$$E_{\theta_{-i}\in\Theta_{-i}}[u_i(f(\theta,\theta_{-i});\theta_i)] \ge \bar{u}_i(\theta_i). \tag{8.11}$$

• *Ex post* individual rationality: players can withdraw from the mechanism at the end:

$$u_i(f(\theta);\theta_i) \ge \bar{u}_i(\theta_i). \tag{8.12}$$

A special kind of mechanism, called the *direct-revelation mechanism* (DRM), has a strategy space $S = \Theta$, and players simply report their type to the mechanism with outcome rule $g: \Theta \rightarrow \Omega$. For the DRM, we have the following definitions:

DEFINITION 8.5 A DRM is Bayesian Nash incentive-compatible if truth revelation is a Bayesian Nash equilibrium, i.e., $s_i^*(\theta_i) = \theta_i$, for all $\theta \in \Theta$.

DEFINITION 8.6 *A DRM is* strategy-proof *if truth revelation is a dominant-strategy* equilibrium for all $\theta \in \Theta$.

8.2.3 Revelation principle

The revelation principle states that for any Bayesian Nash equilibrium there corresponds a Bayesian game with the same equilibrium outcome but in which players truthfully report their types. The principle allows one to solve for a Bayesian equilibrium by assuming all players truthfully report their types (subject to an incentive-compatibility constraint), which eliminates the need to consider either strategic behavior or lying. So, no matter what the mechanism, a designer can confine attention to equilibria in which players truthfully report type.

THEOREM 8.2 For any mechanism M there is a direct, incentive-compatible mechanism with the same outcome.

Proof Consider a mechanism M = (S, g) that implements $f(\theta)$, in a dominant-strategy equilibrium. In other words, $g(s^*(\theta)) = f(\theta)$ for all $\theta \in \Theta$, where s^* is a dominant-strategy equilibrium. We construct the direct mechanism M = (S, g). In contradiction, we suppose

$$\exists \theta_i' \neq \theta_i, \ \text{s.t.} \ u_i(f(\theta_i', \theta_{-i}), \theta_i) > u_i(f(\theta_i, \theta_{-i}), \theta_i)$$
(8.13)

for some $\theta'_i \neq \theta_i$. But because $f(\theta) = g(s^*(\theta))$, this implies that

$$u_i(g(s_i^*(\theta_i'), s_{-i}^*(\theta_{-i})), \theta_i) > u_i(g(s_i^*(\theta_i), s_{-i}^*(\theta_{-i})), \theta_i),$$
(8.14)

which contradicts the strategy-proofness of s^* in mechanism M.

The practical implications are obvious for the above theorem. First, incentivecompatibility is free, i.e., any outcome implemented by mechanism M can be implemented by an incentive-compatible mechanism M'. Second, fancy mechanisms are unnecessary, i.e., any outcome implemented by a mechanism with complex strategy space S can be implemented by a DRM.

8.2.4 Budget balance and efficiency

Before we define budget balance, we first introduce transfers or side payments. Define the outcome space $\mathcal{O} = \mathcal{K} \times \mathbb{R}^N$ such that an outcome rule $o = (k, t_1, ..., t_N)$ defines a choice $k(s) \in \mathcal{K}$ and a transfer $t_i(s) \in \mathbb{R}$ from player *i* to the mechanism, given strategy profile $s \in S$. For example, the utility can be written as

$$u_i(o,\theta_i) = v_i(k,\theta_i) - t_i, \qquad (8.15)$$

where $v_i(k, \theta_i)$ is the value of player *i* and t_i is the payment (transfer) to the auctioneer.

DEFINITION 8.7 Budget balance introduces constraints over the total transfers made from the players to the mechanism. Let $s^*(\theta)$ denote the equilibrium strategy of a mechanism. We have:

- Weak budget balance if:
 - (a) expost: $\sum_{t} t_i(s^*(\theta)) \ge 0, \forall \theta$
 - (b) exante: $E_{\theta \in \Theta}[\sum_t t_i(s^*(\theta))] \ge 0.$
- Strong budget balance if:
 - (a) expost: $\sum_{t} t_i(s^*(\theta)) = 0, \forall \theta$
 - (b) exante: $E_{\theta \in \Theta}[\sum_t t_i(s^*(\theta))] = 0.$

Obviously, strong budget balance is harder than weak budget balance, and *ex post* is harder than *ex ante*.

Next, we define efficiency and discuss the trade-off between efficiency and budget balance.

DEFINITION 8.8 A choice rule $k^* : \Theta \to \mathcal{K}$ is expost efficient if, for all $\theta \in \Theta$, $k^*(\theta)$ maximizes the sum of individual value function, $\sum_{k \in \mathcal{K}} v_i(k, \theta_i)$.

Unfortunately, according to the Green–Laffont Impossibility Theorem [174], if Θ allows all valuation functions from \mathcal{K} to \mathbb{R} , then no mechanism can be efficient and *ex post* budget balance in dominant strategy. So we can either (i) restrict the space of preferences, (ii) drop budget balance, (iii) drop efficiency, or (iv) drop dominant strategy.

8.2.5 Groves mechanism

A special mechanism is the Groves mechanism.

DEFINITION 8.9 A Groves mechanism $M = (\Theta, k, t_1, ..., t_N)$ is defined by means of a choice rule,

$$k^*(\hat{\theta}) = \arg\max_{k \in \mathcal{K}} \sum_i v_i(k, \hat{\theta}_i), \qquad (8.16)$$

and transfer rules,

$$t_i(\hat{\theta}) = h_i(\hat{\theta}_{-i}) - \sum_{j \neq i} v_j(k^*(\hat{\theta}), \hat{\theta}_j), \qquad (8.17)$$

where $h_i(\cdot)$ is an (arbitrary) function that does not depend on the reported type $\hat{\theta}_i$ of player *i*.

It has been proven that the Groves mechanism is strategy-proof and efficient [175]. The Groves mechanisms is, furthermore, unique, in the sense that any mechanism that implements efficient choice $k^*(\theta)$ in truthful dominant strategy must implement Groves transfer rules.

8.2.6 Impossibility and possibility

Of the properties discussed so far, some combinations are possible, while others are impossible. We list some well-known cases in the form of the following theorems:

THEOREM 8.3 (Gibbard–Satterthwaite Impossibility Theorem [165, 427]) If agents have general preferences, and there are at least two agents and at least three different optimal outcomes over the set of all agent preferences, then a social-choice function is dominant-strategy implementable if and only if it is dictatorial (i.e., one or more agents always receives one of its preferred alternatives).

THEOREM 8.4 (*Hurwicz Impossibility Theorem [222]*) It is impossible to implement an efficient, budget-balanced, and strategy-proof mechanism in a simple exchange economy³ with quasi-linear preferences.

THEOREM 8.5 (*Myerson–Satterthwaite Theorem [348]*) It is impossible to achieve allocative efficiency, budget balance, and (interim) individual rationality in a Bayesian Nash incentive-compatible mechanism, even with quasi-linear utility functions.

An interesting extension of the Groves mechanism, the dAGVA (or "expected Groves" [45, 123]) mechanism, demonstrates that it is possible to achieve efficiency and budget balance in a Bayesian Nash equilibrium, even though this is impossible in a dominant-strategy equilibrium (Hurwicz Impossibility Theorem). However, the dAGVA mechanism is not individually-rational, which we should expect from the Myerson–Satterthwaite Impossibility Theorem.

8.3 Special auctions

In this section we discuss some popular types of auctions and their properties. (In the next section we will provide the examples of implementation of these auctions in wireless communication.)

8.3.1 VCG auction

A Vickrey auction [256] is a sealed-bid auction in which bidders (players) submit written bids without knowing the bids of the other bidders. The highest bidder wins, but the price paid is the second-highest bid. This auction, created by William Vickrey, is strategically similar to an English auction, and gives bidders an incentive to bid their true value.

If only a single, indivisible good is being sold, the terms "Vickrey auction" and "second-price sealed-bid auction" are equivalent, and they are used interchangeably. When multiple identical units (or a divisible good) are being sold in a single auction, the most obvious generalization is to have all winning bidders pay the amount of the highest non-winning bid. This is known as a uniform-price auction. The uniform-price auction does not, however, result in bidders bidding their true valuations, as they do in a second-price auction, unless each bidder has demand for only a single unit. A generalization of the Vickrey auction that maintains the incentive to bid truthfully is known as the Vickrey–Clarke–Groves (VCG) mechanism [256]. In the VCG, each player in the auction pays the opportunity cost that their presence introduces for all the other players.

³ A simple exchange environment is one in which there are buyers and sellers, selling single units of the same good.

DEFINITION 8.10 The VCG mechanism implements an efficient outcome $k^* = \max_k \sum_i v_j(k, \hat{\theta}_j)$ and computes transfers

$$t_i(\hat{\theta}) = \sum_{j \neq i} v_j(k^{-i}, \hat{\theta}_j) - \sum_{j \neq i} v_j(k^*, \hat{\theta}_j), \qquad (8.18)$$

where $k^{-i} = \max_k \sum_{j \neq i} v_j(k, \hat{\theta}_j)$.

In other words, the payment equals the performance loss for all other users because of the inclusion of user *i*.

For example, suppose two apples are being auctioned among three bidders. Bidder A wants one apple and bids \$5 for it. Bidder B wants one apple and is willing to pay \$2 for it. Bidder C wants two apples and is willing to pay \$6 to have both of them, but is uninterested in buying only one without the other. First, the outcome of the auction is determined by maximizing bids: the apples go to bidder A and bidder B. Next, to decide on payments, the opportunity cost each bidder has imposed on the rest of the bidders is considered. Currently B has a utility of \$2. If bidder A had not been present, C would have won and had a utility of \$6, so A pays 6-\$2 = \$4. Currently A has a utility of \$5 and C has a utility of 0. If bidder B had been absent, C would have won and had a utility of \$6 = \$1. The outcome is identical whether or not bidder C participates, so C does not need to pay anything.

In a Vickrey auction with independent private values (IPV), each bidder maximizes his or her expected utility by bidding (revealing) his or her true valuation. A Vickrey auction is *ex post* efficient (the winner is the bidder with the highest valuation) under the most general circumstances; it thus provides a baseline model against which the efficiency properties of other types of auctions can be compared. The auction is also strategy-proof. Because of the above strengths, VCG auctions are widely used in wireless networking, especially in situations where it is important to prevent players from lying.

Despite the Vickrey auction's strengths, it also has some shortcomings:

- It does not allow for price discovery that is, discovery of the market price if the buyers are unsure of their own valuations without sequential auctions.
- Sellers may use shill bids to increase profit.⁴
- In iterated Vickrey auctions, the strategy of revealing true valuations is no longer dominant.

The VCG mechanism has the following additional shortcomings:

- It is vulnerable to collusion by the losing bidders.
- It is vulnerable to shill bidding with respect to the buyers.
- It does not necessarily maximize seller revenues; these may even be zero in a VCG auction. If the purpose of holding the auction is to maximize profit for the seller rather than just to allocate resources among buyers, then VCG may be a poor choice.
- The seller's revenues are non-monotonic with regard to the sets of bidders and offers.

⁴ A *shill* is a person who helps another person or organization to sell goods or services without disclosing that he or she has a close relationship with the seller.

8.3.2 Share auction

A share auction [505, 51, 324] is concerned with allocating a perfectly divisible good among a set of bidders. The most common example in the literature comes from the financial markets (e.g., the auction of treasury notes) [53, 497, 212]. Other examples include the allocation of emission permits [470] and the sale of electricity [144]. There are two basic pricing structures in a share auction. In a uniform-price auction, all the winners (typically more than one) get some portion of the good and pay the same unit price. In a discriminatory price auction (sometimes called a pay-you-bid auction [51]), winning bids are filled at the bid price. The references above largely focus on how different pricing and information structures affect auction results such as the final price, the seller's revenue, and the allocation of the divisible good.

Unlike the well-studied single-unit-good auction, in which bidders typically submit one-dimensional bids, some share auctions allow bidders to submit multiple combinations of price and quantity as bids (e.g., [53, 497, 470]). This significantly complicates the auction design since the bidders have large strategy spaces. When using a share auction to allocate resources such as bandwidth in communication networks, researchers typically adopt simple one-dimensional bidding rules, as in [235, 517, 422, 314, 313, 312]. The allocation is proportional to the value of the bids. Some researchers have focused on developing bounds of efficiency loss in such simple bidding games: Johari and Tsitsiklis [235] show that with a uniform-pricing scheme the Nash equilibrium (NE) of a share auction achieves at least 3/4 of the total utility in a socially optimal solution. Sanghavi and Hajek [422] show that the efficiency loss could be reduced to roughly 1/8 if discriminatory pricing (pay-you-bid) is used. Yang and Hajek [517] and Maheswaran and Basar [314, 316, 317, 318] advance the results further by showing that more complex pricing functions can reduce the efficiency loss to zero under certain conditions on the bidders' utility functions. Maheswaran and Başar have considered several network resource-allocation games using share auctions, focusing on the effects of coalition [313] or the design of decentralized negotiation methods [311, 312, 315].

Let us consider how a share auction can be used in spectrum sharing. We consider the case in which there is a measurement point in the network. The aggregated interference generated by all users at the measurement point should be no larger than a threshold P, i.e., $\sum_{i=1}^{N} p_i \leq P$. Here, p_i is the allocated power for the *i*th user.

In a share auction, users submit one-dimensional bids b_i representing their willingness to pay, and the manager simply allocates the available resource P in proportion to the bids. The users then pay an amount proportional to their performance gain γ_i . The manager announces a non-negative reserve bid β . In contrast to the situation in which the manager submits a reserve bid to extract more revenue from the other bidders [339], here the main purpose of the reserve bid is to guarantee a unique desirable outcome of the auction. A share-auction mechanism can be expressed as follows:

1. The manager announces a reserve bid $\beta \ge 0$ and a price $\pi > 0$.

2. After observing β and π , user *i* submits a bid $b_i \ge 0$.

3. The resources are allocated to each user *i*, whose share *p_i* is proportional to its bid, i.e.,

$$p_i = \frac{\beta}{\sum_i b_i + \beta} P. \tag{8.19}$$

For example, if P is the overall transmitted power, for the interference case we have the resulting SINR for user i as

$$\gamma_i(\mathbf{p}) = \frac{p_i h_{ii}}{n_0 + \sum_{i \neq i} p_j h_{ji}},\tag{8.20}$$

where h_{ij} is the channel gain for user *i* to receiver *j* and n_0 is the noise level. 4. In a share auction, user *i* pays $C_i = \pi \gamma_i$.

A bidding profile is the vector $\mathbf{b} = (b_1, ..., b_N)$ containing the users' bids. The bidding profile of user *i*'s opponents is defined as $b_{-i} = (b_1, ..., b_{i-1}, b_{i+1}, ..., b_N)$, so that $\mathbf{b} = (b_i; b_{-i})$. Typically, each user *i* submits a bid b_i to maximize its surplus function,

$$S_i(b_i; b_{-i}) = U_i(\gamma_i(b_i; b_{-i})) - C_i$$

Here, we have suppressed the dependence on β and π . An NE of the auction is a fixed point of all users' best responses.

These auction mechanisms differ from some previously proposed auction-based network resource-allocation schemes (e.g., [235, 314]) in that the bids here are not the same as the payments. Instead, the bids are signals of willingness to pay. The manager can, therefore, influence the NE by choosing β and π . This alleviates the typical inefficiency of the NE, and in some cases allows us to achieve socially optimal solutions.

With a properly chosen price π , the share auction can achieve either fair or efficient allocation. In a fair allocation, users achieve the performance by some predefined fairness criteria. In an efficient allocation, the total utility for the network is maximized.

8.3.3 Double auction

In a double auction [156], there are *l* buyers and *N* sellers. Each buyer *i* wants to purchase x_i items, and each seller *n* wants to sell y_n items. The information about x_i and y_n are available publicly. A buyer *i* reports price $p_i^{(b)}$ (i.e., bidding price), while a seller *n* reports price $p_n^{(s)}$ (i.e., asking price). These prices are per unit of item. Without loss of generality, we may assume $p_1^{(b)} \ge p_2^{(b)} \ge \cdots \ge p_i^{(b)}$ and $p_1^{(s)} \le p_2^{(s)} \le \cdots \le p_N^{(s)}$. Note that if two prices are equal, their indexes are interchangeable. Also, each seller and each buyer can set different prices for different items, with each item being bought and sold separately.

To determine the trading price in a double auction, the demand quantities from all buyers are arranged in descending price order, and the supply quantities from all sellers are arranged in ascending price order (Fig. 8.1). At the trading point T^* , the aggregate



Fig. 8.1 Example of ordered demand and supply in a double auction.

demand and supply intersect, so n' - 1 sellers will trade with i' - 1 buyers. There are two cases in determining the trading price and trading quantity:

- 1. The bidding and asking prices satisfy the condition $p_{i'}^{(b)} \ge p_{n'}^{(s)} \ge p_{i'+1}^{(b)}$ and aggregate demand and supply satisfy $\sum_{n=1}^{n'-1} y_n \le \sum_{i=1}^{i'} x_i \le \sum_{n=1}^{n'} y_n$. In this case, the sellers $n \in \{1, ..., n'\}$ sell all their items y_n at price $p_{n'}^{(s)}$, and the buyers $i \in \{1, ..., i'\}$ buy at price $p_{i'}^{(b)}$. Each buyer buys a quantity $\left[x_i \frac{\sum_{j=1}^{i'-1} x_j \sum_{j=1}^{n'-1} y_j}{i'-1}\right]$, where $\lfloor x \rfloor$ denotes a floor function.
- 2. The bidding and asking prices satisfy the condition $p_{n'+1}^{(s)} \ge p_{i'}^{(b)} \ge p^{(s)}$ and aggregate demand and supply satisfy $\sum_{i=1}^{i'-1} x_i \le \sum_{n=1}^{n'} y_n \le \sum_{i=1}^{i'} x_i$. In this case, the buyers $i \in \{1, ..., i'\}$ buy at price $p_{i'}^{(b)}$, and the sellers $n \in \{1, ..., n'\}$ sell at price $p_{n'}^{(s)}$. Each seller sells a quantity $\left\lfloor y_n \frac{\sum_{j=1}^{i'-1} y_j \sum_{j=1}^{j'-1} x_j}{n'-1} \right\rfloor$.

However, when a central controller is available in this double auction, an optimization problem can be formulated to obtain the quantity of items to be traded. Let the reserve price of each buyer and seller be fixed and denoted by $\hat{p}_i^{(b)}$ and $\hat{p}_n^{(s)}$, respectively. Let $\hat{x}_{i,n}$ and $\hat{p}_{i,n}$ be the solutions: the quantities buyer *i* buys from seller *n* and the trading price, respectively. The utility for buyer *i* in a double auction is

$$U_i^{(b)} = \sum_{n=1}^{N} (\hat{p}_i^{(b)} - \hat{p}_{i,n}) \hat{x}_{i,n}, \qquad (8.21)$$

and that of seller n is

$$U_n^{(s)} = \sum_{i=1}^{l} (\hat{p}_{i,n} - \hat{p}_n^{(s)}) \hat{x}_{i,n}.$$
(8.22)
To maximize the utility for both the seller and the buyer, an optimization can be formulated as a linear programming problem, as follows:

$$\max \sum_{i=1}^{l} \sum_{n=1}^{N} \hat{x}_{i,n} \left(\hat{p}_{i}^{(b)} - \hat{p}_{n}^{(s)} \right), \qquad (8.23)$$

s.t.
$$\sum_{i=1}^{l} \hat{x}_{i,n} \le y_n, \forall n, \sum_{n=1}^{N} \hat{x}_{i,n} \le x_i, \forall i, \hat{x}_{i,n} \ge 0, \forall i, n.$$
 (8.24)

The constraints limit the quantities to be traded not to exceed the supply and demand quantities for seller and buyer, respectively.

8.4 Examples of communication applications

As explained in the previous sections, an auction is a process of selling and buying a commodity (or service) whose price is undetermined. In an auction process, the bidders submit their bids (e.g., in terms of bidding price and quantity) to the auctioneer, who then determines the winning bidder. Then, the commodity is sold at the trading price. There are three types of auctions: supply auction, demand auction, and double auction (Fig. 8.2). In a supply auction, multiple sellers offer a commodity to a buyer. In a demand auction, multiple buyers bid for a commodity being sold by a seller. In a double auction, multiple buyers bid to buy commodities from multiple sellers.



The ingredients of in an auction market are as follows:

- *Seller* is an entity in the market who wants to sell the commodity. A seller offers the price (i.e., *asking* price) and the amount of commodity to be traded by auction.
- *Buyer* is an entity who wants to buy the commodity. A buyer submits a bid in terms of price and the amount of the commodity to be bought through the auction.
- *Trading/clearing price* is the price of each commodity to be traded in an auction market. This trading price has to satisfy the asking price from the seller and the bidding price from the buyer (e.g., higher than or equal to asking price, but lower than or equal to the bidding price).

In the remainder of this section, we provide examples of the application of auction theory to cognitive-radio networks (Section 8.4.1) and to communication with physicallayer security (Section 8.4.2).

8.4.1 Cognitive radio

In a cognitive-radio network, a spectrum auction may be jointly designed within a resource-allocation framework (e.g., scheduling). An example of such a spectrum auction is shown in Fig. 8.3 [251]. In this case, the downlink and uplink schedulers will use information about the auction mechanism from both the network service provider and the bidding strategy of the user. The user can bid for the spectrum based on quality-of-service (QoS) requirements, while the network service provider can charge a price according to the bids from all users.

General framework of a spectrum auction

To support spectrum allocation and pricing, a real-time spectrum-auction framework was proposed in [161]. In the system model, the seller sells the available spectrum to the buyers (Fig. 8.4). The seller could be a spectrum owner or a primary user with spectrum opportunity to be sold. The buyers could be service providers, cognitive-radio users, or secondary users.

This framework was designed to support a large number of cognitive-radio users under interference constraints. A multi-unit auction scheme was used to support diverse





Fig. 8.4 System model of a spectrum auction.

spectrum demand, resulting from both short-term and long-term traffic load. The objective is to maximize the auction revenue of the spectrum owner and also the spectrum utilization. To obtain a solution, a fast auction-clearing algorithm was proposed.

In order to communicate bidding information between the spectrum seller and the buyer, a bidding language, piecewise linear price-quality (PLPQ), was proposed in [161]. Buyers express their demands in terms of spectrum bandwidth and per-unit price. Bids are submitted to the seller. In PLPQ, spectrum demand is expressed as a concave piecewise linear demand function such as

$$p_i(f_i) = -a_i f_i + b_i,$$
 (8.25)

where p_i is the spectrum price, f_i is the spectrum bandwidth, and a_i and b_i are parameters of the linear function. In this case, the size of the demand spectrum can be expressed as a function of price by $f_i(p_i) = \frac{b_i - p_i}{a_i}$, and the revenue can be expressed as $R_i(p_i) = p_i f_i(p_i) = \frac{b_i p_i - p_i^2}{a_i}$.

The spectrum seller collects the spectrum bids (i.e., spectrum demand) from all buyers, and an auction-clearing algorithm is used to obtain the optimal clearing price. Two pricing models were used, namely *uniform* and *discriminatory* pricing. In the case of uniform pricing, the clearing price p is fixed for all users and the spectrum will be sold to the buyer if $b_i \ge p$, where b_i is the bidding parameter in (8.25). An optimization problem was formulated to obtain price p such that the revenue is maximized, as follows:

$$\max \sum_{\{i|b_i > p\}} \frac{b_i p - p^2}{a_i}, \text{ s.t. } f_i = \frac{b_i - p}{a_i}, f_i + \sum_{j \in \mathbb{N}_L(i)} f_j \le 1.$$
(8.26)

The constraint in (8.26) is the interference, i.e., the sum of the spectrum shares f_i and f_j must be less than or equal to one. $\mathbb{N}_L(i)$ is the set of nodes lying to the left of node *i* in a two-dimensional network, and it is used (instead of the set of all neighboring nodes) to reduce the complexity of optimization.

In discriminatory pricing, the clearing price could be different for each buyer. In this case, an optimization problem can be formulated as follows:

$$\max_{f_i} \sum_{i} (-a_i f_i^2 + b_i f_i), \text{ s.t. } -a_i f_i + b_i \ge 0, f_i \ge 0, f_i + \sum_{j \in \mathbb{N}_L(i)} f_j \le 1, \quad (8.27)$$

where f_i is a function of p_i , which is the optimal discriminatory price.

To obtain the optimal solution, approximation algorithms of linear programming [230] were proposed. For uniform pricing, the feasible region for clearing price p is first obtained. The algorithm searches for an optimal clearing price which maximizes the revenue. With discriminatory pricing, the algorithm uses separable programming [210] to solve a special class of non-linear programs using linear programming. The details of the derivation can be found in [161].

As for the performance evaluation, it is shown that the discriminatory-pricing model can achieve higher revenue and higher utilization than a uniform-pricing model. Under the discriminatory-pricing model, the spectrum broker charges cognitive-radio users based on their condition. Consequently, the spectrum broker can optimize to achieve higher revenue and utilization. These two pricing models were also evaluated under different user behaviors: aggressive, normal, and conservative, with unit price functions defined as follows: p(f) = -f + 1, p(f) = 1/2(-f + 1), and p(f) = 2(-f + 1), respectively. In the uniform-pricing model, the aggressive user obtains all the spectrum. Since the clearing price is high, only an aggressive user can afford it. In contrast, with a discriminatory-pricing model, although an aggressive user obtains the largest portion of the spectrum, normal and conservative users also obtain part of the spectrum. Using an approximation algorithm, it was observed that performance is only lower than the optimal solution by only about 10 percent. However, the optimal solution takes much longer to compute.

Dynamic spectrum allocation by multi-bid auction

In [255], a one-shot multi-bid auction framework for dynamic spectrum allocation was proposed. This framework explicitly considers the various wireless services, in the same or different regions, that could interfere with each other. As in [161], the framework determines the solution in terms of spectrum bandwidth and the price for service providers, which bid for the spectrum from the spectrum owner (Fig. 8.4). However, in this framework a service provider submits a bid as a set of two-dimensional bids, rather than the piecewise linear function of [161]. Consequently the solution space is smaller, which results in lower computational complexity.

Let the set of bids from service provider *i* be represented as

$$\mathbb{B}_{i} = \{ b_{i,1}, \dots, b_{i,D_{i}} \}, \tag{8.28}$$

for $i \in \{1, ..., l\}$, where D_i is the total number of bids from service provider *i*, and $b_{i,d} = (q_{i,d}, \Theta_i(q_{i,d}))$, where $q_{i,d}$ denotes the requested spectrum bandwidth, and $\Theta_i(\cdot)$ denotes the total price the provider is willing to pay for the requested spectrum bandwidth.

The profile of the bids from all service providers is denoted by $\mathbb{B} = \{\mathbb{B}_1, ..., \mathbb{B}_l\}$. The optimum feasible allocation of spectrum is defined as $\mathbb{A}^*(\mathbb{B}) = \{a_1, ..., a_l\}$, where $a_i = q_{i,d_i^*}$,

$$(d_1^*, \dots, d_i^*, \dots, d_i^*) = \arg \max_{(d_1, \dots, d_i)} \sum_{i=1}^{l} \Theta_i(q_{i, d_i}),$$
(8.29)

and q_{i,d_i} is an element from the set of feasible allocations \mathbb{B} . This solution is obtained using a simulated annealing algorithm.

The price charged to the service provider is based on an exclusion-compensation principle (i.e., a second-price auction) [161]. That is, service provider *i* considers price c_i , which covers the social opportunity cost. This cost quantifies the loss of utility due to the presence of service provider *i*. In this case, the set of all bids except that of service provider *i* is defined as $\mathbb{B}_{-i} = \{\dots, \mathbb{B}_{i-1}, \mathbb{B}_{i+1}, \dots\}$. The optimum feasible spectrum allocation is defined as

$$\mathbb{A}_{-i}^* = \{a_1^{-i*}, \dots, a_{i-1}^{-i*}, a_{i+1}^{-i*}, \dots, a_l^{-i*}\}.$$

The price for service provider *i* is then

$$c_i(\mathbb{B}) = \sum_{j \neq i} \Theta_j(a_j^{-i*}) - \Theta_j(a_j).$$
(8.30)

The proposed scheme was evaluated by considering two regions, A and B. In these regions, there are two service providers and one provider of terrestrial digital video broadcasting (DVB-T). The ultra wideband (UWB) provider is only in region B. It was observed that the DVB-T provider pays the highest price since it covers both regions and its interference limit is low. In addition, the cost for the UWB provider is zero because this network does not interfere with the other networks, owing to its small coverage area.

Dynamic spectrum allocation by double auction

When multiple sellers (i.e., primary users) and multiple buyers (i.e., secondary users) are involved in a spectrum auction, a double-auction model can be applied to obtain a competitive equilibrium solution [232]. The model assumes that all sellers and buyers are rational, seeking to maximize their payoffs. However, sellers and buyers may not provide private information, or they may even cheat each other if this can improve the payoff. Therefore, each seller and each buyer has to develop its own belief regarding the information from others in order to reach a solution.

First, a static-pricing game model was formulated for spectrum trading. In this case, the payoff for the primary user *i* can be expressed as:

$$U_{i} = \sum_{j=1}^{n_{i}} \left(\phi_{a_{i}^{j}} - c_{i}^{j} \right) x_{i}^{j}, \qquad (8.31)$$

where n_i is the total number of channels of primary user i, $\phi_{a_i^j}$ is the payment received by the primary user from selling channel a_i^j to the secondary user, c_i^j is the cost for primary

user *i* of selling channel *j*, and x_i^j is a binary variable indicating whether the channel *j* is sold $(x_i^j = 1)$ or not $(x_i^j = 0)$. Similarly, the payoff for the secondary user *k* can be expressed as:

$$U_{k} = \sum_{k=1}^{N} \left(v_{k}^{j} - \phi_{j} \right) y_{k}^{j}, \qquad (8.32)$$

where *N* is the total number of channels, v_i^j is the benefit gained from obtaining channel *j* by secondary user *k*, and y_k^j is again a binary variable indicating whether or not channel *j* is obtained by secondary user *k*. Since the primary and secondary users are rational, optimization problems can be formulated to obtain the payments $\phi_{a_i^j}$ and ϕ_j such that the payoffs for the primary and secondary users are maximized. To avoid the complexity resulting from the multi-objective nature of this optimization problem, competitive equilibrium from double-auction theory can be used to obtain the solution. This competitive equilibrium determines the price at which the number of channels to be bought is equal to the number of channels to be sold. In this case, the supply function is defined as the relationship between the number of channels to be sold and the cost for the primary users. The demand function is defined as the relationship between the number of channels to be bought and the benefit to the secondary users.

Since the payoff function for a primary/secondary user depends on the decision of the secondary/primary user, some primary and secondary users can cheat by posting incorrect information. In this case, all primary and secondary users must establish their own beliefs regarding the available information.

The static-pricing game model was also extended to the dynamic case for which there are multiple periods, and the parameters in the payoff functions for both primary and secondary users can be varied in each period. The objective functions for both types of users now include a factor γ , which discounts the payoff in the future. These objective functions are in the form of the Bellman equation, which can be solved by the standard dynamic programming method [232]. The budget of secondary users for buying a channel can be integrated into the discounted objective function.

The algorithm for obtaining the solution in the two-user case is presented in [232] as follows. First, the subcarrier assignment is initialized so that the minimum throughput requirements of all users are satisfied. Second, the subcarriers are sorted according to their channel quality. Third, a group of subcarriers is divided into two subgroups, and these subgroups are assigned to different users. Fourth, the subcarrier assignment is updated, and the performance is measured. The algorithm stops if the utility cannot be improved. The convergence of this algorithm was proved.

The performance evaluation of the proposed scheme was compared with the theoretical competitive equilibrium (i.e., perfect knowledge for all users). When the number of secondary users increases, the total payoff increases. It is observed that the total payoff obtained from the proposed scheme is slightly smaller than that from the theoretical model. The overhead of the proposed scheme is compared with the traditional continuous double auction. It is clear that the overhead from the proposed scheme is much smaller than that of a traditional double auction, since the belief-update algorithm can reduce the amount of information exchange.

Dynamic spectrum allocation by sequential and concurrent auction

The problem of dynamic spectrum access has been formulated as multi-unit, sealedbid, sequential, and concurrent auctions [437]. In this case, service providers bid for the spectrum from a spectrum broker. In a sequential auction, the spectrum bands are auctioned one by one, while in a concurrent auction all spectrum bands are auctioned at once. In this spectrum-auction environment there are N bands and I service providers. These service providers are rational, seeking to maximize their profit, calculated as the revenue gained from serving the corresponding users minus the price paid to the spectrum broker. In each auction round, service provider i submits bid b_i to the spectrum broker. Given the bids from all service providers, the spectrum broker determines the winning service provider (with the highest bid), and allocates the spectrum band to that service provider. At the end of an auction round, the spectrum broker broadcasts the maximum bid for that round. In the next auction round, the winning service provider will decrease the bid to increase its profit. These steps are repeated until a steady state is reached.

The value of the band for service provider *i* is denoted by v_i . In the case of a substitutable band, this value is fixed for all bands. The profit for the service provider is

$$\pi_i = \begin{cases} v_i - b_i, & \text{if service provider wins,} \\ 0, & \text{if service provider loses.} \end{cases}$$
(8.33)

In a sequential auction, the probability-density function of bid b is assumed to be uniformly distributed, and can be expressed as

$$f(b) = \frac{1}{v_{\max} - b_{\min}},\tag{8.34}$$

where v_{max} is the maximum value of the spectrum band, and b_{\min} is the lowest bid of all submitted bids. If k bands have already been auctioned, the probability that any bid b_j is less than b_i , where $j \in (l - k - 1)$, $j \neq i$, and $i \in (l - k)$, is $\int_{b_{\min}}^{b_i} f(b) db$, or

$$P(b_j < b_i) = \frac{b_i - b_{\min}}{v_{\max} - b_{\min}}.$$
 (8.35)

To win the auction, the condition $\forall b_j < b_i$ has to be satisfied, and the probability of service provider *i* winning the sequential auction is

$$P_{\rm win}^{\rm seq}(i) = \left(\frac{b_i - b_{\rm min}}{v_{\rm max} - b_{\rm min}}\right)^{I-k-1}.$$
(8.36)

The expected profit of service provider *i* is

$$\overline{\pi}_i = \left(v_i - b_i\right) \left(\frac{b_i - b_{\min}}{v_{\max} - b_{\min}}\right)^{l-k-1}.$$
(8.37)

The optimal bid to maximize this profit can be obtained by differentiating the expected profit $\overline{\pi}_i$ with respect to bid b_i , giving

$$b_{i_{seq}}^* = \frac{(I-k-1)v_i + b_{min}}{I-k}.$$
 (8.38)

In a concurrent auction, the probability of winning is the probability of a bid b_i such that $b_j < b_i$ for $j \neq i$. That is, bid b_i will win the auction if all l - N other bids are smaller. The probability of winning the concurrent auction is

$$P_{\rm win}^{\rm con}(i) = \left(\frac{b_i - b_{\rm min}}{v_{\rm max} - b_{\rm min}}\right)^{I-N}.$$
(8.39)

Similarly, the optimal bid for service provider *i* in a concurrent auction is

$$b_{i_{\rm con}}^* = \frac{(I-N)v_i + b_{\rm min}}{I-N+1}.$$
(8.40)

Sequential and concurrent auctions were compared in [437]. The comparison was divided into three cases: transient-state case 1, transient-state case 2, and the steady state (defined when all service providers reach fixed bids and cannot unilaterally change the bids to obtain higher profit). In contrast, in a transient state the service providers adjust their bids before reaching the steady state. Transient states were divided into two cases, when no band has been auctioned (transient-state case 1) and when *k* bands have already been auctioned (transient-state case 2). In all three cases it was shown that the optimal bids in a sequential auction are higher than those in a concurrent auction. Therefore, the spectrum broker will prefer a sequential auction since higher revenue from service providers can be achieved.

Dynamic spectrum allocation by knapsack auction

A model for a spectrum auction was proposed in [436], formulated as a knapsack problem. In the system model, *I* service providers bid for *N* frequency bands from the spectrum broker. Service provider *i* submits bid b_i , which is defined as $b_i = \{x_i, v_i\}$, where x_i and v_i are the number of requested bands and the bidding price, respectively. Given the bids from all service providers, the spectrum broker solves the knapsack problem, defined as follows:

$$\max \sum_{i \in \mathbb{W}} v_i, \text{ s.t. } \sum_{i \in \mathbb{W}} x_i \le N,$$
(8.41)

where \mathbb{W} is the set of service providers winning the auction.

For an asynchronous auction, a service provider submits a bid to the spectrum broker whenever a spectrum band is required (e.g., because of instantaneous load). If the spectrum broker has enough spectrum bands, they are immediately allocated to the service providers. However, if the number of requested bands is larger than the number of available bands, the knapsack auction in (8.41) is solved to obtain the optimal solution for the spectrum broker. For the winning service providers, the spectrum bands will be allocated for the duration requested in each bid. A similar auction process is used in synchronous auctions. However, the spectrum bands in that case will be allocated for equal durations for all service providers.

It was proved in [436] that the revenue generated by service providers in an asynchronous auction cannot be better than that in a synchronous auction for a given set of bids. This is because, in a synchronous auction, the spectrum band is allocated for a fixed and equal duration for all service providers. Therefore, revenue can be optimized in each period. Let us consider an example in which the number of bands is N = 2. At time slot t, only service provider 1 submits a bid $b_1 = \{1, 1\}$ (i.e., 1 band for price 1) for two time slots. At time slot t + 1, service provider 2 submits a bid $b_2 = \{2, 5\}$ (i.e., 2 bands for price 5) for one time slot. In an asynchronous auction, 1 band is allocated to service provider 1 in each of time slots t and t + 1, while another band is allocated to service provider 2 in time slot t + 1 only. In this case, both spectrum bands cannot be allocated in time slot t + 1 to service provider 2, whose bidding price is higher than that of service provider 1. In contrast, in a synchronous auction, 1 band is allocated to service provider 1, while 2 bands are allocated to service provider 2 in time slot t + 1. Therefore, the total revenue earned by the spectrum broker in a synchronous auction is $(1 \times 1) + (2 \times 5) = 11$, which is higher than $(1 \times 1) + (1 \times 1 + 1 \times 5) = 7$ in a asynchronous auction. Since the solution of the auction is obtained at each time slot, the spectrum-band allocation can be adjusted according to the bids, and the revenue of the spectrum broker is higher for a synchronous auction.

Dynamic spectrum allocation by weighted proportional fairness

A spectrum-allocation algorithm based on weighted proportional fairness for an OFDMA-based cognitive-radio network was proposed in [149]. This algorithm is based on an asymmetric Nash bargaining solution (ANBS) utility function which captures the spectrum-sensing contribution of the secondary users. These secondary users are grouped to cooperate in spectrum sensing and sharing. The user who contributes more in spectrum sensing will receive a better allocation, while fairness and efficiency can be achieved for all users. The coordination among secondary users in the same group is performed by a central base station.

In the system model, for subcarrier *n* the throughput of the secondary user *i* is denoted by $\tau_{i,n}$ and its allocation is denoted by $x_{i,n} = 1$ ($x_{i,n} = 0$ if subcarrier *n* is not allocated to user *i*). This throughput is computed from the channel quality of the secondary user on that subcarrier, where adaptive modulation is used for transmission. If ω_i denotes the spectrum-sensing contribution of secondary user *i*, an optimization problem can be defined as follows:

$$\max_{x_{i,n}} \quad U = \prod_{i=1}^{l} \left(\sum_{n} x_{i,n} \tau_{i,n} - T_{\min}^{(i)} \right)^{\omega_i}, \tag{8.42}$$

s.t.
$$\sum_{n} x_{i,n} \tau_{i,n} \ge T_{\min}^{(i)}, \sum_{i=1}^{l} x_{i,n} = 1, \sum_{n} p_{i,n} \le p_{\max}, \sum_{i=1}^{l} \omega_i = 1,$$
 (8.43)

where *U* is the utility, *I* is the total number of secondary users, $T_{\min}^{(i)}$ is the minimum throughput requirement of user *i*, $p_{i,n}$ is the transmit power of user *i* on subcarrier *n*, and p_{\max} is the power budget. An algorithm to obtain the solution was presented, and the convergence of the algorithm was proved, in [149].

In the performance evaluation, the OFDMA system with adaptive modulation was considered. As the spectrum-sensing contribution of one secondary user increases, from the optimization formulation those of other users decrease. There is an optimal point where the aggregated throughput is maximized. This scheme was compared with maximum-total-throughput and max-min-fairness formulations. Intuitively, the maximum-total-throughput formulation provides the highest throughput, but it is not fair for the user with low channel quality. On the other hand, the max-min-fairness formulation achieves a fair solution (i.e., the individual throughput of the user is almost the same), but the total system throughput is not maximized. In the scheme proposed in [149], which is based on an asymmetric Nash bargain solution, is a compromise between these formulations. That is, the user with bad channel quality will receive higher throughput, while the throughput of the user is not significantly different. Also, the user who contributes more in spectrum sensing is rewarded with higher throughput.

Dynamic spectrum allocation by spectrum policy server

A spectrum policy server (SPS) can be used as a broker to allocate spectrum to service providers. These service providers then use the allocated spectrum to provide wireless access service to cognitive-radio users. A non-cooperative game has been formulated for spectrum allocation in this SPS [226]. In this system, the user can accept the service from the provider based on price and satisfaction with performance. In particular, the service-acceptance probability by users from provider *i* is

$$A(U(r), p) = 1 - e^{-Cu^{\mu}p^{-\epsilon}}$$
, where $U(r) = \frac{(r/k)^{\beta}}{1 + (r/k)^{\beta}}$, (8.44)

where *u* is the utility (i.e., satisfaction) defined as a function of the transmission rate *r*, *p* is the price of the service, and *C*, μ , ϵ , *k*, and β are constants of this acceptanceprobability function. Basically, when the transmission rate becomes higher, the utility becomes higher. Therefore, the user will accept the service with higher probability. However, when the price is high, this acceptance probability becomes low because of the larger cost for the user. If there is more than one service available, the user will choose the service with higher acceptance probability.

For service provider *i*, the profit P_i is defined as

$$P_i(r_i, p_i) = p_i - F_i - V_i b_i,$$
 (8.45)

where F_i is the fixed cost for service provider and V_i is the price per unit of bandwidth b_i charged by the spectrum server. This bandwidth is a function of the transmission rate r_i and the spectral efficiency c_i , i.e., $b_i = r_i/c_i$. Owing to the acceptance probability of

the user, the expected profit for the service provider becomes $\overline{P}_i(r_i, p_i) = A(U(r_i), p_i) \times P_i(r_i, p_i)$.

Each service provider adjusts the service price and the transmission rate to achieve the highest profit. In this case, a non-cooperative game can be formulated as follows. The players of this game are the service providers. The strategy of each player is the price and the offered transmission rate, and the payoff for each player is the resulting expected profit, defined as follows:

$$\phi_i((r_i, p_i), (r_j, p_j)) = \begin{cases} 0, & \text{if } A(U(r_i), p_i) < A(U(r_j), p_j), \\ 1/2\overline{P}_i(r_i, p_i), & \text{if } A(U(r_i), p_i) = A(U(r_j), p_j), \\ \overline{P}_i(r_i, p_i), & \text{if } A(U(r_i), p_i) > A(U(r_j), p_j). \end{cases}$$
(8.46)

The Nash equilibrium, which is the solution of this non-cooperative game, is defined as follows:

$$(r_i^*, p_i^*) = \arg\max_{(r_i, p_i)} \phi_i((r_i, p_i), (r_j, p_j)).$$
(8.47)

Since there could be multiple service providers in the system, the service selection can be performed for the user by the spectrum server to reduce the complexity and overhead of the system. This service selection is referred to as a bidding process, and it works as follows:

- 1. A new user connects to the spectrum server and also submits an acceptance-probability function.
- 2. Service providers compete with each other to provide service to the user. This competition is iterated until the Nash equilibrium solution is achieved for the service price and transmission rate.
- 3. The winning service provider reports the winning bid, in terms of service price and transmission rate, to the user. The user then makes a decision on whether to accept the service based on the acceptance-probability function.

These steps are shown in Fig. 8.5.



Fig. 8.5 Iterative bidding through a spectrum server.

The multiple-user case was also considered. In this case, service providers compete for each user individually, using the same steps for iterative bidding as in Fig. 8.5. The spectrum server can maximize its revenue by adjusting the bandwidth allocation to each user, and an optimization problem can be formulated as follows:

$$\max R_{server}(\mathbf{b}) = \sum_{m=1}^{M} V_m A_m(U(r), p) B_m(\mathbf{b}) \text{ s.t. } \sum_{m=1}^{M} b_m \ge b_{\max}, \qquad (8.48)$$

where **b** is a vector of bandwidth allocated to each user (i.e., $\mathbf{b} = [\cdots b_m \cdots]$), *M* is the total number of users, b_{max} is the maximum size of the spectrum that can be allocated to all service providers, and B_m is the transmission rate offered by the winning service provider (i.e., $B_n = r_n^*/c_n$).

With two service providers, the acceptance probabilities of both providers with competition are higher than without competition. The competition benefits the user since the price is lower. However, the expected profits of the service providers are lower when there is competition. In the case of multiple users, the studied scheme was compared with an equal-bandwidth partition scheme (i.e., the same amount of bandwidth is allocated to each user). When the number of users was varied, the proposed scheme based on the spectrum server achieved higher expected bandwidth utilization and higher average acceptance probability because of the game and optimization formulation. From the performance evaluation, it was also observed that when the cost of unit bandwidth increases, the service provider with higher fixed costs becomes more competitive in attracting users, in order to improve its revenue.

Bilateral bargaining in spectrum access

Within a microeconomic framework, dynamic spectrum allocation can be optimized using the concepts of auction and bargaining [173]. In particular, two algorithms based on an Anglo-Dutch split-award auction and bilateral bargaining models have been applied in [173] to short-term and long-term spectrum trading, respectively, among multiple radio access networks (RANs). In the first algorithm, based on an auction model, spectrum trading is divided into four stages. In the first stage, service providers owning RANs submit bids for the auction. These service providers are allocated the minimum pre-specified amount of available spectrum. In addition, the two service providers with the highest bid prices are selected to proceed to the second stage. In the spectrum remaining from the first stage. In the third stage, the spectrum owner allocates the spectrum to maximize the total bids. Finally, in the fourth stage, two service providers compete with each other to sell their services to the market using the obtained spectrum.

In the second algorithm, based on a bilateral bargaining model, there are leasing and renting RANs negotiating for the spectrum to maximize their utilities [173]. In this case, the leasing RAN submits the asking price while the renting RAN submits the bidding price. Let *c* and *v* denote the cost and the value of the spectrum for leasing and renting RANs, respectively. Based on asking price p_{ask} and bidding price p_{bid} , the trading price

is determined from $p_{tr} = \frac{p_{bid} + p_{ask}}{2}$, and the profits of these RANs can be expressed as follows:

$$P_{\text{leasing}} = \frac{p_{\text{ask}} + p_{\text{bid}}}{2} - c, \qquad (8.49)$$

$$P_{\rm renting} = v - \frac{p_{\rm ask} + p_{\rm bid}}{2}, \qquad (8.50)$$

for $p_{ask} \le p_{bid}$. Both leasing and renting RANs search for the optimal asking and bidding prices to maximize their profits. However, they do not have information about each other: the leasing RAN does not have perfect information on value *v* for the renting RAN, and the renting RAN does not have perfect information on cost *c* for the leasing RAN. Therefore, an optimization problem is formulated for the leasing RAN, as follows:

$$\max_{\rho_{ask}} P_{\text{leasing}} = \frac{1}{2} \left(p_{ask} + E \left[p_{\text{bid}}^{\text{lease}}(v) | p_{\text{bid}}^{\text{lease}}(v) \ge p_{ask} \right] - c \right)$$
$$\times \Pr \left(p_{\text{bid}}^{\text{lease}}(v) \ge p_{ask} \right),$$
(8.51)

where $\tilde{p}_{\text{bid}}^{\text{lease}}(v)$ is an estimated value for the bidding price from the renting RAN, which is used by the leasing RAN; $E\left[\tilde{p}_{\text{bid}}^{\text{lease}}(v)|\tilde{p}_{\text{bid}}^{\text{lease}}(v) \ge p_{\text{ask}}\right]$ is the expected value of the estimated bidding price given that this estimated bidding price is larger than or equal to the asking price; and $\Pr\left(\tilde{p}_{\text{bid}}^{\text{lease}}(v) \ge p_{\text{ask}}\right)$ is the probability that the estimated bidding price is larger than or equal to the asking price. Similarly, an optimization problem for the renting RAN can be formulated as follows:

$$\max_{p_{\text{bid}}} P_{\text{renting}} = v - \frac{1}{2} \left(p_{\text{bid}} + E \left[\tilde{p}_{\text{ask}}^{rent}(c) | p_{\text{bid}} \ge \tilde{p}_{\text{ask}}^{rent}(c) \right] \right) \\ \times \Pr \left(p_{\text{bid}} \ge \tilde{p}_{\text{ask}}^{rent} \right), \qquad (8.52)$$

where $\tilde{p}_{ask}^{rent}(c)$ is the estimated value for the asking price from the leasing RAN, which is used by the renting RAN. An optimal Bayesian equilibrium, which is considered to be the solution of this bilateral bargaining model, is defined as follows:

$$\frac{\mathrm{d}P_{\text{leasing}}}{\mathrm{d}p_{\text{ask}}} = 0, \quad \frac{\mathrm{d}^2 P_{\text{leasing}}}{\mathrm{d}p_{\text{ask}}^2} < 0, \tag{8.53}$$

$$\frac{\mathrm{d}P_{\text{renting}}}{\mathrm{d}p_{\text{bid}}} = 0, \quad \frac{\mathrm{d}^2 P_{\text{renting}}}{\mathrm{d}p_{\text{bid}}^2} < 0.$$
(8.54)

Note that the second order is considered for optimality since P_{leasing} and P_{renting} are functions of the single variables p_{ask} and dp_{bid} , respectively.

After the RAN obtains the spectrum, it sells this radio resource to users through auction (i.e., auction between UMTS RAN and UMTS users). Since the UMTS system was considered, the bidding radio resource is defined as the code in CDMA and the time frame, which is divided into 15 slots, each of which can be allocated to a user. Each user has to bid for the radio resource in both uplink and downlink. The bid consists of the



Fig. 8.6 Bidding message.

amount of radio resource and the bidding price. The bid from each user is submitted to the service provider by using the bidding message (Fig. 8.6). The service provider allocates radio resources according to the users' bids to maximize its revenue. This allocation is based on a discriminatory auction.

The simulation result shows that the solution of the auction among service providers for spectrum incurs smaller overhead than that from the scheme without estimation. This result is similar to that in [232]. For the auction between the UMTS service provider and the user, the discriminatory-pricing solution of the auction is compared with the uniform-pricing model. As in [161], the discriminatory-pricing model can achieve higher revenue than the uniform-pricing model.

8.4.2 Physical-layer security

The design of future wireless networks will have to put a huge effort into security. The main reason is that future networks will be decentralized and ad hoc in nature, and, hence, will allow various types of network mobile terminals to join and leave. This makes the entire network vulnerable and very sensitive to attack. Because of the broadcast nature of wireless transmission, anyone within communication range can intercept data not intended for them. In such a complex environment, current cryptographic methods with high-level security may not work. This may happen because of the difficulty of exchanging public keys in such an ad hoc network. Therefore it is of great importance to investigate the design of decentralized networks with perfect security on a physical layer. For this reason, physical-layer security is gaining new attention. The goal is a decentralized system that will protect broadcast data and make it impossible for an eavesdropper to receive packets even if it knows the encoding and decoding schemes used by the transmitter and receiver. In systems in which physical-layer security is studied, the main objective is to maximize the rate of reliable information from the source to the intended destination, while all malicious nodes are kept as ignorant of that information as possible. This maximum reliable rate is known as secrecy capacity.

Secrecy capacity work was pioneered by Aaron Wyner, who defined the wiretap channel and established fundamental results that allowed the creation of almost perfectly secure communication without the exchange of private (secret) keys [512]. Wyner showed that when an eavesdropper channel is a degraded (weaker) version of the main channel, the source and destination can exchange perfectly secure messages at a positive rate. With his scheme, a maximal equivocation (i.e., uncertainty) is induced at the eavesdropper, i.e., a maximal level of secrecy is obtained. By ensuring that the equivocation rate is arbitrarily close to the message rate, one can achieve perfect secrecy in the sense that the eavesdropper can now learn *almost nothing* about source-destination messages. Follow-up work by Leung-Yan-Cheong and Hellman characterized the secrecy capacity of the additive white Gaussian noise (AWGN) wiretap channel [289]. In their seminal paper, Csiszár and Körner generalized Wyner's approach by considering the transmission of confidential messages over broadcast channels [118]. Recently, research in the area of physical-layer security has exploded. There have been considerable efforts to generalize these studies to wireless-channel and multi-user scenarios (see [206, 289, 293, 357, 383, 440, 294, 172, 133] and references therein). Jamming [240, 441, 83, 54], long studied to analyze the hostile behavior, of malicious nodes has been applied to physical-layer security to reduce an eavesdropper's ability to decode the source's information [278]: friendly jamming in this context.

In this section, we use auction theory to investigate the interaction between source– destination pairs and a friendly jammer. Although the friendly jammer helps by reducing the rate of data "leaking" to a malicious node, it also reduces the useful data rate from source to destination. Using well-chosen amounts of power from the friendly jammer, the secrecy rate⁵ can be maximized. In the auction defined here, the source–destination pairs provide bids for the jammer to interfere with the malicious eavesdropper, and thus to increase the secrecy rate. Our analysis uses the VCG auction model [256].

System model

We consider a network with multiple sources s_i , destinations d_i , a malicious eavesdropper node m, and a friendly jammer node J, as shown in Fig. 8.7. The malicious node tries to eavesdrop on transmitted data from the source nodes. When the eavesdropper channel from the source to the malicious node is a degraded version of the main source– destination channel, the source and destination can exchange perfectly secure messages at a non-zero rate. By transmitting a message at a rate higher than the rate to the malicious node, the malicious node can learn almost nothing about the message.

Suppose the source s_i transmits with power P_i . The channel gains from the source to the destination and from the source to the malicious node are $G_{s_id_i}$ and G_{s_im} , respectively. The friendly jammer *J* transmits with power P_i^J , and the channel gains from *J* to the destination and the malicious node are G_{Jd_i} and G_{Jm} , respectively. The large-scale path-loss model is used with a particular path-loss coefficient. The thermal noise for each channel is σ^2 , and the bandwidth is *W*. The channel capacity from source *i* to destination *i* is

$$C_{1}^{i} = W \log_{2} \left(1 + \frac{P_{i} G_{s_{i} d_{i}}}{\sigma^{2} + P_{i}^{j} G_{J d_{i}}} \right).$$
(8.55)

⁵ The secrecy rate is an achievable rate that is smaller than the secrecy capacity.



Fig. 8.7 System model. S_i are multiple sources, d_i are multiple destinations, m is a malicious eavesdropper, and J is a friendly jammer.

The channel capacity from source to the malicious node is

$$C_2^i = W \log_2 \left(1 + \frac{P_i G_{s_i m}}{\sigma^2 + P_i^J G_{Jm}} \right).$$
(8.56)

We assume that there is no interference from other sources, since only one source at a time transmits its own data.

The secrecy rate is defined as

$$C_{s_i} = \max(C_1^i - C_2^i, 0). \tag{8.57}$$

We observe that with an increase in jamming power P_i^J , both C_1 and C_2 are reduced. The questions are whether C_{s_i} can be increased, and how jamming power may be controlled in a distributed manner. We approach the problem using auction theory.

VCG auction

In the VCG auction model, the jammer asks all sources for their evaluations of its power, and calculates the optimal power allocation accordingly. Each source pays a "performance loss" to other sources arising from its own participation in the auction. In the context of wireless secrecy rate, the performance upper bound can be described as follows:

• *Information*: Publicly available information includes the noise density σ^2 and the bandwidth *W*. Source s_i knows the channel gains $G_{s_id_i}$ and G_{s_im} . The jammer knows the channel gains G_{Jd_i} for all *i*, and can estimate the channel gain G_{Jm} when it receives bids from the sources.

- *Bids*: Source s_i submits $\Delta C_{s_i} (P_i^J(b_i; b_{-i}))$ to the jammer, i.e., the increase in the secrecy rate as a function of jammer power P_i^J .
- Allocation: The jammer determines the power allocation $\mathbf{P} = [P_1^J \dots P_N^J]$ by solving the following problem:

$$\mathbf{P}^{*} = \arg\max_{\mathbf{P}} \sum_{j \in \mathcal{I}} C_{\mathbf{s}_{j}} \left(P_{j}^{J} \right).$$
(8.58)

• *Payments*: For each source *s_i*, the jammer solves the following problem:

$$\mathbf{P}^{*/i} = \arg \max_{\mathbf{P}, P_i = 0} \sum_{j} C_{\mathbf{s}_j} \left(P_j^J \right), \qquad (8.59)$$

i.e, the total distortion decreases without allocating resources to source *i*. The payment for source *i* is then

$$c_{i} = \sum_{j \neq i, j \in \mathcal{I}} C_{s_{j}}\left(P_{j}^{*/i}\right) - \sum_{j \neq i, j \in \mathcal{I}} C_{s_{j}}\left(P_{j}^{*}\right), \qquad (8.60)$$

i.e., the performance loss to all other sources arising from the inclusion of source *i* in the allocation.

The resource allocation calculated in (8.58) achieves an *efficient* allocation, as shown in [256]. The VCG auction can achieve the efficient allocation in one shot, by allowing the jammer to gather a considerable amount of information and perform heavy but local computation.

Although a VCG auction has the desirable social optimality, it is usually computationally expensive to solve l + 1 non-convex optimization problems, where l is the size of \mathcal{I} . To solve a non-convex optimization by the interior point method, a complexity of $O(l^2)$ is required. As the result, the overall complexity for the performance upper bound is $O(l^3)$.

8.5 Summary

From the standpoint of auction theory as applied to wireless networking and resource allocation, the individual user can elect to pay for services such as channels, routes, or power, with payment, made via a central "bank." However, this requires more control than other game-theory approaches. Moreover, in order to achieve particular goals such as the network's total benefit, the auction must be designed according to the available information. Mechanism design is a tool for game and auction design.

Mechanism design is a subfield of microeconomics and game theory that considers system-wide solutions to problems that involve multiple self-interested players, each with private information about their preferences. The goal is a social choice function for distributed systems with private information and rational players. The design criteria are: efficiency, fairness, revenue maximization, budget balance, and Pareto optimality. As discussed in this chapter, there are some basic concepts and theorems for mechanism design. The revelation principle states that under quite weak conditions any mechanism can be transformed into an equivalent incentive-compatible, direct-revelation mechanism [256] that implements the same social-choice function. One of the most important families of mechanisms is the VCG mechanism, the only type that is allocatively efficient and strategy-proof [256]. A detailed discussion of mechanism design can be found in [3].

Beyond the VCG auction, there are many other types of auctions. We have studied the share auction and the double auction in detail. In a share auction, resources are allocated to users in proportion to their bids. A double auction is a process of buying and selling goods in which potential buyers submit their bids and potential sellers simultaneously submit their asking prices to the auctioneer, who chooses a price p that clears the market: all the sellers who asked less than p sell and all buyers who bid more than p buy at this price p. Beyond those auctions, the sequential auction [437] and the multiple-item auction [434, 438, 435] have also attracted much attention recently.

Using auction theory, we have studied several examples of applications in wireless communication networks. Dynamic spectrum access in cognitive-radio networks can be formulated as an auction among different secondary users. In physical-layer security, the friendly jammer's power can be allocated by auction theory. Many other applications can be formulated and solved in a similar way.

Part II

Applications of game theory in communications and networking

9 Cellular and broadband wireless access networks

In the past two decades, cellular communication has witnessed a significant growth. Today, millions of mobile users utilize cellular phones worldwide. In essence, a cellular network is designed to provide a large number of users with access to wireless services over a large area. The basic architecture of a cellular network relies on dividing a large area (e.g., a city) into smaller areas, commonly referred to as cells. Each cell typically represents the coverage area of a single base station that is often located at the center of the cell. By dividing the network into cells, one can ensure a reliable coverage and access to wireless services. Despite the emergence of ad hoc networks with no infrastructure, the cellular architecture remains prevalent in the majority of existing or soon-to-be deployed wireless networks, because of its proven success. In fact, cellular communication has been the pillar architecture in key wireless systems, from traditional 2G systems such as GSM to 3G systems such as UMTS, and the emerging 4G and 5G systems. Beyond traditional macrocell-based networks (e.g., 3G and 4G networks), the use of small cells, covered by low-cost, low-power stations known as femtocell access points that can be overlaid with existing architectures, has recently become of central importance in the design of next-generation wireless networks. Thus, cellular technology is expected to remain as one of the most important paradigms in future wireless communication systems. In Chapter 2, we provided a comprehensive introduction to cellular communication, its key challenges, as well as its past and projected future evolution.

Broadband wireless access refers to a range of wireless radio systems used primarily to convey broadband services between users' premises and core networks. A typical broadband wireless access network supports a connection to many user premises within a radio coverage area. In a way, a broadband wireless network provides a pool of bandwidth that is shared automatically amongst the network's users. Broadband wireless networks are seen as an alternative to many existing wired backhaul and last-mile coverage deployments such as cable modems, digital subscriber lines (DSL), T- and E-carrier systems, and optical-carrier technologies. The importance of this technology is obvious, notably in areas that are sparsely settled across difficult terrain, as well as in areas where running cables is infeasible. The definition of the term "broadband" has been debated for many years, but often the term simply implies the capability to deliver a significant bandwidth to network users, enabling the delivery of a plethora of services and applications. Fueled by the phenomenal growth of broadband Internet access and Web browsing demands,

users' quest for broadband links on the move has been escalating steadily toward pervasive broadband connectivity irrespective of locality, mobility, or type of connectivity. Although broadband wireless access does not fulfill all the requirements for ubiquitous communication, it nevertheless tackles the growing metropolitan and rural connectivity gap. In this regard, IEEE introduced a standard, IEEE 802.16, for the development of broadband wireless access technology. Furthermore, the WiMAX forum has been set up and given the task of ensuring interoperability and conformance among the systems and solutions developed by various vendors, based on the IEEE 802.16 standard for broadband wireless access. In Chapter 2, we presented a detailed and comprehensive introduction to broadband wireless access, IEEE 802.16, and, notably, WiMAX networks.

The range of applications covered by cellular and broadband wireless access networks is very wide and evolving quickly. It includes voice, data, gaming, video, and a variety of entertainment applications. Next-generation cellular and broadband networks are expected to enable their users to explore a varied mix of wireless services, while efficiently coping with rapid changes in users' environment and conditions as these users roam in the network and terminate/establish network connections. In particular, these networks must efficiently serve a large number of wireless users while providing seamless connectivity and access to a broad range of services with stringent performance requirements. Consequently, numerous technical challenges arise such as architecture management, power control, resource allocation, admission control, handover, interference management, small-cell overlay, and new-node deployment. Many of these challenges involve the design of models that can accurately capture the numerous interactions between wireless users. Moreover, it is desirable to design low-cost, low-complexity, distributed algorithms that can enable devices using the cellular and broadband networks to operate efficiently, optimize their performance metrics, and meet QoS demands. This need for distributed optimization, efficient operation, and fair resource allocation in cellular and broadband networks, coupled with their large-scale nature, is a motivation for the use of game-theoretic techniques for addressing some central technical issues. In this chapter, using a variety of tools from game theory, we tackle the following key technical challenges in cellular and broadband networks:

- Uplink power control in CDMA networks. In the uplink of CDMA networks, as the users operate using the same frequency, it is imperative that they control the transmit power in such a way as to ensure good performance (e.g., in terms of throughput) while minimizing interference. Hence, to manage interference and maintain a desired QoS, it is essential to design low-complexity, distributed algorithms for power control. In Section 9.1, we address the problem of power control in the uplink of CDMA cellular networks from different perspectives. First, we study the problem in a single-cell system, and, then, we analyze a multi-cell system and the additional challenges and constraints that it incurs with regard to the design of efficient power-control schemes.
- **Resource allocation in OFDMA networks.** OFDMA is a promising technology that is expected to become the standard for multiple access in numerous cellular and

broadband wireless access networks. A key challenge in an OFDMA system is to allocate the resources [187] (often considered as the OFDMA subcarriers) among users in an efficient way. This resource-allocation problem gives rise to interesting competitive (and cooperative) situations among the users of a wireless OFDMA system. In Section 9.2, we analyze these situations and develop game-theoretic models to address the challenges of resource allocation in single-cell multi-user OFDMA networks.

- **Deployment of femtocell access points.** The need for pervasive wireless coverage, as well as the high cost incurred in installing wireless equipment such as base stations, has led to the use of low-cost transceivers known as femtocell access points. These can be deployed, at a low cost, in conjunction with existing infrastructure such as cellular or broadband networks. However, their deployment has been accompanied by numerous technical challenges, such as maintaining an efficient co-existence with existing infrastructure and designing suitable resource-allocation schemes for the femtocell network. In Section 9.3, we analyze, using non-cooperative Stackelberg games, the problem of power control in a network in which macrocell base stations need to co-exist with femtocell access points.
- **IEEE 802.16 broadband wireless access.** The deployment of broadband wireless access networks such as IEEE 802.16 entails numerous challenges, including resource allocation, power control, admission control, and architecture design. In Section 9.4, we develop game-theoretic tools for addressing a number of these issues. In particular, we develop non-cooperative game models for bandwidth allocation and admission control in 802.16 networks. In addition, we show how network-formation games can be used for efficient relay station deployment in next-generation broadband networks such as the upcoming IEEE 802.16 j standard.
- Vertical handover in heterogeneous wireless networks: With the emergence of the IEEE 802.21 standard, it is envisioned that next-generation wireless devices will be able to utilize, simultaneously, a variety of technologies such as cellular, broad-band, and WiFi. Thus, heterogeneity, in terms of wireless technologies, will be a key characteristic enabling seamless mobility in next-generation wireless networks such as 4G and 5G systems. In such heterogeneous networks, the wireless devices need to decide autonomously which technology to use, given their location and performance requirements. This challenging network-selection problem is often referred to as *vertical handover*. In Section 9.5, we tackle the vertical-handover problem in heterogeneous networks, using non-cooperative and Bayesian games. We show how game theory can be used to enable users to make decisions on which network to use, in a distributed manner, taking into account their individual optimization objectives.

9.1 Uplink power control in CDMA networks

In wireless communication systems, mobile users need to adjust their transmit power in such a way as to respond to the time-varying nature of the channel. In code-division multiple-access (CDMA) systems, power control is essential for the efficient operation of the network. In such systems, the users perform power control for two main reasons:

- The limited battery energy available to mobile users
- The increase in capacity that is possible through interference minimization and management.

By using power control, CDMA network users aim to achieve a certain SINR level, independent of channel conditions, while minimizing interference. Much research has been dedicated to studying uplink power control in CDMA networks. In [519], a power-control scheme was studied in which each user updates its transmit power based on the total received power at the base station, and it was shown that, using non-cooperative game theory for uplink power control, one can implement a distributed algorithm that converges to a Nash equilibrium under a wide variety of interference models. This line of work, using non-cooperative game theory, was further pursued in [171], where a utility suited for data services was proposed. In this context, [171] studies how pricing can improve the efficiency of the resulting equilibria, in the Pareto sense. Following this seminal work on using non-cooperative game theory for power control in cellular and wireless networks, the popularity of the approach increased significantly, and it was applied to numerous scenarios and networks (e.g., see [187] for a comprehensive survey on power control in CDMA networks).

In this section, we study the uplink power-control problem in wireless CDMA systems. First, we study how non-cooperative game theory can be used to model the uplink powercontrol problem in single-cell CDMA networks. We discuss the existence and properties of the resulting equilibria. Then, we analyze multi-cell networks using a non-cooperative game. We highlight the properties of the multi-cell power-control game, and discuss how pricing can be implemented in a multi-cell CDMA network to achieve equilibrium efficiency for the power-control game.

9.1.1 Single-cell CDMA networks

In this subsection, we study the problem of uplink power control in single-cell CDMA cellular networks.

Single-cell uplink power-control game: formulation and equilibrium

Consider a single-cell CDMA system with up to N users. In this network, the number of users is limited under an admission-control scheme that ensures the minimum necessary SINR for each user in the cell. The users seek to regulate their transmit power so as to improve their performance in the uplink. For this purpose, we consider a non-cooperative strategic game among the network users in which the strategy of any user *i* is to select a value for the uplink transmit power p_i so as to optimize the following cost function:

$$c_i(p_i, \mathbf{p}_{-i}) = \lambda_i p_i - \alpha_i \log(1 + \gamma_i), \ p_i \ge 0, \forall i,$$
(9.1)

where \mathbf{p}_{-i} is the power profile of all users except *i*, and γ_i is the SINR function for user *i*, which is given by

$$\gamma_i = \mathcal{L} \frac{h_i p_i}{\sum_{j \neq i} h_j p_j + \sigma^2},\tag{9.2}$$

with $L = \frac{W}{R}$ being the spreading gain of the CDMA system, where *W* is the chip rate and *R* is the total rate. Hereinafter, it is assumed that L > 1. The parameter h_j , $0 < h_j < 1$, is the channel gain from user *j* to the base station in the cell, and $\sigma^2 > 0$ is the Gaussian noise variance. The cost function in (9.1) represents the difference between a pricing function that assigns a linear price λ_i per power unit and the benefit that a user draws from a better SINR, which is assumed to be logarithmic, with $\alpha_i > 0$ a parameter capturing the user's level of "desire" for SINR. The parameter α_i will be referred to as the *benefit parameter*. Note that the linear pricing in (9.1) implies that the benefit (in terms of increased SINR) that a user receives by increasing its power level is reduced by a linear function of this power level. Hence, each user pays a price in terms of the power consumed. Such a pricing scheme ensures that users aim not only to improve their SINR but also to reduce their energy consumption if possible (which eventually helps in minimizing the interference in the system).

For notational convenience, we let $y_i = h_i p_i$ denote user *i*'s received power level at the base station, and $\bar{y}_{-i} = \sum_{j \neq i} y_j$. Furthermore, we define a parameter a_i as follows: $a_i \triangleq \frac{\alpha_i h_i}{\lambda_i} - \frac{\sigma^2}{L}$.

Consequently, by defining the players, the strategies, and the utilities, we have formulated a strategic non-cooperative power-control game among the CDMA users. This game is a continuous-kernel game since the strategy spaces (power levels) and the utilities are continuous (see Chapter 3). In this power-control game, each user *i* aims to minimize the cost function in (9.1) given the sum of powers of the other users as received at the base station, i.e., \bar{y}_{-i} as well as the noise. Given this strategic game, the next step is to study and analyze the existence and uniqueness of the Nash equilibrium.

To perform this study, we first derive each user's best-response function. Using the first-order condition and invoking the positivity constraint ($p_i \ge 0$), the best-response function can be found as [23]:

$$p_i = b_i(\bar{y}_{-i}, a_i) = \begin{cases} \frac{1}{h_i}(a_i - \frac{1}{L}\bar{y}_{-i}), & \text{if } \bar{y}_{-i} \leq La_i, \\ 0, & \text{otherwise.} \end{cases}$$
(9.3)

We can see that the best-response function of each user depends not only on user-specific parameters such as λ_i , α_i , and h_i but also on the spreading factor L (which is a network parameter) and the total power level at the base station, $\sum_{j=1}^{N} y_j$. This dependence can be seen by adding $-\frac{p_i}{L}$ to both sides of (9.3) and dividing both sides by $(1 - \frac{1}{L})$. The base station can, in the downlink, provide the user with the total received power level, so the user can update its best-response function. Using the best-response functions of the users and the set of fixed-point equations that they yield, the following result holds [23]:

THEOREM 9.1 In the studied uplink power-control game with N users, let the indexing be done such that $a_i < a_j \Rightarrow i > j$, with the order being chosen arbitrarily if $a_i = a_j$. Let

 $N^* \leq N$ be the largest integer \tilde{N} with the following condition satisfied:

$$a_{\tilde{N}} > \frac{1}{(L+\tilde{N}-1)} \sum_{i=1}^{\tilde{N}} a_i.$$
 (9.4)

Then, the power-control game admits a unique Nash equilibrium with the property that users $N^* + 1, ..., N$ have zero power levels, $p_j^* = 0, j \ge N^* + 1$. The equilibrium power levels of the first N^* users are

$$p_{i}^{*} = \frac{1}{h_{i}} \left\{ \frac{L}{L-1} \left[a_{i} - \frac{1}{L+N^{*}-1} \sum_{j \in \mathcal{N}^{*}} a_{j} \right] \right\}, \ i \in \mathcal{N}^{*} = \{1, \dots, N^{*}\}.$$
(9.5)

If no \tilde{N} can be found for which (9.4) is satisfied, then there still exists a unique Nash equilibrium, but this equilibrium assigns zero power to all N users in the network.

Thus, as proven in [23], the modeled game admits a unique Nash equilibrium (in transmit powers), which dictates that the users utilize the powers in (9.5) at the equilibrium. As long as an integer exists with (9.4) satisfied, this equilibrium would have at least one user with non-zero power; otherwise, an equilibrium exists in which no user is admitted. It is interesting to note that condition (9.5) is a function of channel gains, noise, price, desired utility level, and spreading factor.

Convergence to the Nash equilibrium: update schemes and stability

While the Nash equilibrium of the studied non-cooperative uplink power-control game has been well characterized in the previous subsection, the next step is to construct distributed algorithms for reaching the equilibrium. Iterative-update algorithms, in which each user updates its power value until it reaches the equilibrium, are popular in game theory and communications. In this context, we can devise two interesting schemes, as described in [23]: a parallel-update scheme and a random-update scheme.

In the parallel-update scheme, at each iteration (assuming discrete time intervals), all users update their power levels using the best-response function in (9.3). Thus, this algorithm can be described as

$$y_i^{(k+1)} = \max\left(0, a_i - \frac{1}{L} \sum_{j \neq i} y_j^{(k)}\right).$$
(9.6)

It is shown in [23] that, if the following condition is satisfied:

$$\frac{N-1}{L} < 1,$$
 (9.7)

then the parallel-update algorithm would be globally stable, i.e., starting from any initial point, it converges to the unique Nash equilibrium, which is given by

$$p_i^* = \max\left(0, \frac{1}{h_i}(a_i - \frac{1}{L}\sum_{j \neq i} h_j p_j^*)\right), \ p_i \ge 0, \ \forall i.$$
(9.8)

The parallel-update algorithm considers that the users update their power levels in discrete time, in a deterministic manner. This update scheme can be enhanced by devising a random-update scheme whereby the users optimize their power levels with a predefined probability $0 < \prod_i < 1$. In the random-update scheme, at each iteration a set of randomly picked users (among the *N* users) update their power levels. This algorithm can be described as follows:

$$y_{i}^{(k+1)} = \begin{cases} h_{i} \max\left(0, \frac{1}{h_{i}}\left(a_{i} - \frac{1}{L}\sum_{j \neq i} y_{j}^{(k)}\right)\right), & \text{with probability } \Pi_{i}, \\ y_{i}^{(k)}, & \text{with probability } 1 - \Pi_{i}. \end{cases}$$
(9.9)

It is demonstrated in [23], using the Borel–Cantelli lemma, that the random-update scheme converges asymptotically under the condition that

$$\frac{N-1}{L}\bar{\Pi} + (1-\underline{\Pi}) < 1, \tag{9.10}$$

where $\overline{\Pi}$ and $\underline{\Pi}$ are, respectively, the upper and lower limits for the update probability of any user *i*, i.e., $\underline{\Pi} < \Pi_i < \overline{\Pi}$. Note that when all of the update probabilities are equal to 1, (9.10) reduces to (9.7).

In both algorithms, the users are myopic, in the sense that they update their power levels based on instantaneous parameters, while ignoring the future implications of their actions. The difference is that, in the random-update scheme, not all users act in every iteration. Whether a user acts is determined probabilistically. When the probabilities are equal to 1 for all users, then random update is equivalent to parallel update.

Using simulations, it is shown in [23] that, in a system with no delay and in which all users have the same initial power, the random-update scheme outperforms the parallelupdate scheme. This is a consequence of the myopic behavior of users, as well as the inherent randomization in the case of random update. In contrast, the opposite is true for a system with delay because variations in delay provide sufficient randomization, and the parallel-update scheme becomes more advantageous as a result of frequent updates.

Pricing strategies at the base station

In order to align the users' goals with those of the network, it is usually useful to design adequate pricing strategies. In the considered CDMA system, the price λ_i per unit power for any user *i* is determined by the base station. The Nash solution previously computed, does not by itself guarantee that users who are transmitting, i.e., with non-zero power levels, will be able to meet their minimum SINR requirements for establishing a connection with the base station. Thus, given that, in different wireless systems, the users might have to meet different SINR requirements, we discuss, as in [23], the use of two pricing mechanisms:

• *Centralized pricing.* The users are divided into multiple classes, depending on the value of their benefit parameters α_i . All users within a class have the same SINR requirement. The objective of the base station is to set prices for these different classes such that, at the resulting Nash equilibrium, the SINR targets are met.

• **Decentralized market-based pricing.** In this scheme, the base station sets a single price for all users, while the users choose their benefit parameter α_i to satisfy their quality-of-service requirements. This scheme, in contrast to the centralized scheme, allows users to compete for the system resources by adjusting their individual α_i .

In both approaches, we consider that the base station sets the price proportionally to the channel gain of the *i*th user, i.e., $\lambda_i = k_i h_i$ with k_i a constant. For the centralized pricing scheme, it is shown in [23] that, when every mobile user has the same SINR requirement, the following result holds:

THEOREM 9.2 Consider a network with symmetric users where $\alpha_i = 1$, $\forall i$, the minimum SINR requirement is γ^* for all users, and the users are charged in proportion to their channel gain, $\lambda_i = kh_i$, with the parameter k being user-independent. Then, in this network the maximum number of users N^* that the system can accommodate is bounded by

$$N^* < \frac{L}{\gamma^*} + 1. \tag{9.11}$$

Moreover, the user-independent pricing parameter k under which $N \le N^*$ users achieve the SINR level γ^* is

$$k = \frac{\lambda_i}{h_i} = \frac{L}{\sigma^2} \frac{L - \gamma^* (N - 1)}{L(\gamma^* + 1)}.$$
(9.12)

The proof is presented in [23]. The pricing approach in Theorem 9.2 is equivalent to a centralized power control in which the base station determines the price in such a way that the users utilize power levels determined by the unique Nash equilibrium. The base station can also set the prices so that the SINR requirements of the users are satisfied. Note that if $N > N^*$ then all users would fall below their SINR level because of symmetry. The system can, in this case, decrease the number of users admitted (i.e., N) below the threshold N^* , so as to obtain a viable solution.

Although Theorem 9.2 deals with symmetric users having the same SINR requirements, this result can easily be extended to the case of multiple service levels and multiple schemes. To do so, the users can be split into groups or cells, where each group has the same desired SINR level, i.e., each group has symmetric users. In this case, the base station can apply the result of Theorem 9.2 to each group, and thus obtain multiple pricing schemes for the different groups.

The centralized approach discussed so far does not take into account the fact that each user *i* can, dynamically, adjust its benefit parameter α_i so that it can always maintain a minimum SINR level γ_i^* , given the interference at the base station. In this case, the base station can limit aggressive requests for SINR by setting an upper bound y_{max} on the received power of any user *i*, i.e., $y_i \leq y_{\text{max}}$. This strategy would be implemented by the base station in order to preserve the network resources. It is demonstrated in [23] that, for this market-based pricing strategy, given that α_i is lower- and upper-bounded, then by limiting the number of mobiles to N_{max} the base station can provide the following

minimum SINR level γ_{\min}^* :

$$\gamma_{\min}^{*} = \frac{Ly_{\max}}{\sigma^{2} + (N_{\max} - 1)y_{\max}}.$$
(9.13)

This highlights a tradeoff in the choice of the design parameters N_{max} and γ_{min}^* . If the network wants to provide guarantees of a high SINR level, then it has to sacrifice by limiting the number of users. Furthermore, the users may implement a distributed admission-control scheme according to their budget constraints and desired SINR levels. In such a scheme, if, for a user *i*, the price necessary to achieve a SINR level exceeds a certain budget η_i , i.e.,

$$\frac{k\gamma_i^*}{L}(\bar{y}_{-i}+\sigma^2) \ge \eta_i, \tag{9.14}$$

then the user might choose not to transmit at all.

In summary, in this subsection we have formulated the uplink power-control problem in a CDMA system using non-cooperative game theory, discussed its solution, and studied two possible algorithms for finding the equilibrium as well as the impact of pricing on the game's solution.

9.1.2 Multi-cell wireless CDMA networks

While we dealt in the previous subsection with the problem of uplink power control in a single-cell CDMA network, it is also of interest to study the problem within a multi-cell network. In a multi-cell environment, each mobile terminal experiences interference, not only from the users within the cell but also from users in other cells. Another important problem to consider in a multi-cell environment is the *base-station assignment* problem: each mobile needs not only to adjust its transmit power but also to select the base station to which it needs to connect. The problem of power control and base-station assignment in a multi-cell network has been widely studied in the literature using a variety of approaches, ranging from game theory to control theory and optimization [482, 424, 20, 24, 254, 193]. A comprehensive survey of uplink power control in cellular networks can be found in [106].

In this subsection, we formulate a non-cooperative game for the multi-cell powercontrol problem for data services, based on the seminal work of [424]. Furthermore, we study the properties of the game and discuss possible pricing strategies that can improve the system performance.

Network model for uplink power control in a multi-cell network

Consider the uplink of a CDMA network with K cells serving N users. We let \mathcal{N} the set of all users and \mathcal{K} the set of all base stations. We let h_{ia_i} and d_{ia_i} denote, respectively, the path gain and the distance between user $i \in \mathcal{N}$ and base station $a_i \in \mathcal{K}$. The users are assumed to be stationary, so the path gains do not change. Each user transmits data at a rate of R bits s⁻¹ over a spectrum of W Hz. In this setting, the SINR γ_{ia_i} of any user i

at a base station $a_i \in \mathcal{K}$ is given by

$$\gamma_{ia_i} = \frac{W}{R} \frac{h_{ia_i} p_i}{\sum_{k \in \mathcal{N} \setminus \{i\}} h_{ka_i} p_k + \sigma^2},$$
(9.15)

where p_i is the transmit power of user *i* and σ^2 is the variance of the Gaussian noise.

We assume no base-station diversity, i.e., each user is assigned to a single base station at a given time. In this setting, each user $i \in \mathcal{N}$ attempts to select a base station $a_i \in \mathcal{K}$ and a transmit power p_i , so as to maximize the following utility function:

$$u_{i,a_i}(\mathbf{p}) = \frac{RL}{M} \frac{f(\gamma_{i,a_i})}{p_i},$$
(9.16)

where $\mathbf{p} = [p_1, ..., p_N]$ is the vector of transmit powers, *L* is the number of information bits in a packet of size *M* bits, and $f(\cdot)$ is an efficiency function that approximates the probability of successful reception. For this section, we consider that, for a given SINR γ_{ia_i} , the efficiency function is given by

$$f(\gamma_{ia_i}) = (1 - 2P_e(\gamma_{ia_i}))^M, \qquad (9.17)$$

where $P_e(\gamma_{ia_i})$ is the bit error rate (BER). The function in (9.17) is an approximation of the frame success rate, which yields zero utility at zero power. This function depends on factors such as channel quality, coding, modulation, and packet size. Further discussion on the selection of the function $f(\cdot)$ can be found in [424]. The utility function in (9.16) is a ratio between the expected number of bits received correctly and the energy consumed in transmission. Thus, this utility allows us to measure the performance of a wireless user in bits per joule.

We can formulate a non-cooperative continuous-kernel game among the users, whereby the strategy of each user is two-dimensional, as each user aims to select a base station $a_i \in \mathcal{K}$ and a power value p_i so as to maximize its utility function in (9.16). In other words, we have a non-cooperative game among the N users in which each user $i \in \mathcal{N}$ attempts to solve the following optimization problem:

$$\max_{p_i,a_i} u_{i,a_i}(p_i, \mathbf{p}_{-i}), \tag{9.18}$$

where the transmit powers of user *i* are selected from a convex and compact set with minimum and maximum power constraints, i.e., $\underline{p}_i \leq p_i \leq \overline{p}_i$. Throughout this section, we assume that $\overline{p}_i = \overline{p}$, $\forall i \in \mathcal{N}$.

The optimization problems in (9.18) define the studied non-cooperative game. We note that the user has to select a two-dimensional strategy to optimize his utility. Furthermore, the vector $\mathbf{a} = [a_1, ..., a_N]$ of base-station assignments can be arbitrary. There are a total of \mathcal{K}^N different possible base-station assignments. In the remainder of this section, we discuss two possible assignments: a maximum-received-signal-strength assignment and a maximum-SINR assignment.

Power-control game with maximum-received-signal-strength base-station assignment

To solve the formulated non-cooperative game, we will first assume that, for any user *i*, the assigned base station a_i is determined by the received signal strength of the base-station pilot signal, i.e., a_i is the base station with the highest channel gain h_{ia_i} . Thus, each user $i \in \mathcal{N}$ is assigned to the base station $a_i \in \mathcal{K}$ such that

$$a_i = \underset{j \in \mathcal{K}}{\arg \max} h_{ij} \equiv \underset{j \in \mathcal{K}}{\arg \min} d_{ij}.$$
(9.19)

The assignment in (9.19) is equivalent to assigning the user to the closest base station since we assume a channel gain based only on the distance to the base station. This base-station assignment based on the received signal strength is fixed and independent of the utility function. As a result, under a *fixed* base-station assignment, as in (9.19), solving the non-cooperative game in (9.18) reduces to choosing the best transmit power, instead of the original two-dimensional problem. Because of this, in the remainder of this subsection, for any user $i \in \mathcal{N}$ we drop the subscript pertaining to the base station, i.e., $u_{i,a_i}(\mathbf{p}) = u_i(\mathbf{p})$. Unless stated otherwise, we will assume that $p_i = 0$, $\forall i \in \mathcal{N}$.

In this fixed maximum-received-signal-strength base-station assignment, the next objective is to solve the non-cooperative game and find the Nash equilibrium. First, it has been shown in [424] that the utility function in (9.16) is quasi-concave. This proof is done by showing that the local maximum of the function is at the same time a global maximum. By establishing the quasi-concavity of the utility function in (9.16), the existence of the Nash equilibrium is guaranteed by Theorem 3.2 in Chapter 3, which states that, for a continuous-kernel game, a Nash equilibrium exists if the strategy sets (here the power values) are compact and convex while the utility is continuous in the profile **p** of strategies and quasi-concave in the strategy p_i .

With the existence of a Nash equilibrium established, the next step is to determine whether the equilibrium is unique. First, for any given interference vector \mathbf{p}_{-i} and assuming user *i* is assigned to base station a_i as per (9.19), it is found that the power that maximizes $u_i(p_i, \mathbf{p}_{-i})$ satisfies

$$f'(\gamma_{ia_i})\gamma_{ia_i} - f(\gamma_{ia_i}) = 0, \qquad (9.20)$$

by the first-order optimality condition $(f'(\cdot))$ is the first-order derivative of $f(\cdot)$). Let $\tilde{\gamma} = \gamma_{ia_i}$ be the SINR value satisfying (9.20), which is the same for all $i \in \mathcal{N}$. The value of $\tilde{\gamma}$ depends on parameters such as packet length and modulation. The transmit power p_i that maximizes the utility is, thus, the power that achieves $\tilde{\gamma}$. However, because of power constraints, some users might not be able to achieve $\tilde{\gamma}$ at the equilibrium, so the best these users can do is to transmit at maximum power \bar{p} . Consequently, the best-response function for any user *i* is the value of the power p_i that maximizes the utility for a given interference vector \mathbf{p}_{-i} , namely

$$b_i(\mathbf{p}_{-i}) = \min\left(\bar{p}, \frac{\tilde{\gamma}\left(\sum_{k \neq i} h_{ka_i} p_k + \sigma^2\right)}{\frac{W}{R} h_{ia_i}}\right).$$
(9.21)

By showing that the best-response functions for all users $i \in N$, as given in (9.21), are *standard functions* and using Theorem 3.1 of Chapter 3, it is established, as in [424], that the Nash equilibrium is *unique*.

Thus, for the considered power-control game with maximum-received-signal strength base-station assignment, there exists a unique Nash equilibrium. However, as is often the case, it is shown in [424] that this equilibrium is inefficient. The main idea is that there exists a power vector smaller than the equilibrium vector (component-wise) where users obtain utilities that are higher than those at the equilibrium. As in the single-cell case in Section 9.1.1, to improve the efficiency at the equilibrium we can introduce a pricing scheme.

We consider that the base station takes care of pricing. Thus, we define a vector $\mathbf{c} = [c_1, ..., c_K]$ of pricing factors where each element c_j denotes the price at a given base station $j \in \mathcal{K}$. We consider that each base station charges the users a usage-based price, i.e., a price proportional to the consumed power. In this case, given the base-station assignment in (9.19), the non-cooperative game with pricing can be mapped to the case in which each user *i* (assigned to base station a_i) solves the following optimization problem:

$$\max_{p_i} u_i(p_i, \mathbf{p}_{-i}) - c_{a_i} p_i, \ \forall i \in \mathcal{N}.$$
(9.22)

Notice that, under the base-station assignment scheme in (9.19), the base-station assignment of a user is fixed with or without pricing.

To study the Nash equilibria of the game with pricing, we note that the utility being optimized in (9.22) is no longer quasi-concave with the introduction of pricing. Therefore, we can no longer utilize the existing results for the case of no pricing. Instead, we can show that the game with pricing is, in fact, a supermodular game. Recall from Chapter 3 that a game is supermodular if the utility function has increasing differences, i.e., is supermodular. Also note that in Chapter 3 we provided an example of how a power-control game with pricing can be supermodular in a single-cell network. The following analysis is similar but applied to multi-cell networks.

In the multi-cell power-control game with pricing, the supermodular property implies that we need to have increasing differences in (p_i, \mathbf{p}_{-i}) , which is true if and only if $\frac{\partial^2 u_i(\mathbf{p})}{\partial p_i \partial p_k} \ge 0$ for all $i \ne k$. It is shown in [424], in a straightforward manner, that this second-order condition is satisfied for all $i, j \in \mathcal{N}$ if $\gamma_{ia_i} \ge \hat{\gamma}$ for all *i*, where $\hat{\gamma} = 2 \log M$. This constraint on the SINR of a user *i* can be translated into a lower bound on the strategy of that user, as

$$\underline{p}_{i} = \frac{\hat{\gamma}\left(\sum_{k \neq i} h_{ka_{i}} p_{k} + \sigma^{2}\right)}{\frac{W}{R} h_{ia_{i}}}.$$
(9.23)

Let $\underline{\mathbf{p}}$ denote the power vector corresponding to $\gamma_{ia_i} = \hat{\gamma}$ for all *i* (assuming that it exists), and let $\overline{\mathbf{p}}$ denote the power vector corresponding to the maximum power constraints. Then, the game having the strategy space that is a compact and convex set with

minimum power level \underline{p}_i and maximum power level \overline{p} (assumed non-empty) is clearly supermodular. As discussed in Chapter 3, a supermodular game always possesses Nash equilibrium. In fact, the set of Nash equilibria for a supermodular game has the smallest and the largest (component-wise) power vector (in the context of the multi-cell powercontrol game). We let \mathcal{E} denote the set of Nash equilibria. For the multi-cell power-control supermodular game, all the equilibria $\mathbf{p} \in \mathcal{E}$ are located such that $\mathbf{p}_5(\mathbf{c}) \leq \mathbf{p} \leq \mathbf{p}_L(\mathbf{c})$, where $\mathbf{p}_5(\mathbf{c})$ and $\mathbf{p}_L(\mathbf{c})$ are, respectively, the smallest and largest power vectors. These vectors are shown in [424] to be non-increasing in the price vectors \mathbf{c} . Given this set of equilibria, an asynchronous and distributed algorithm can be implemented to reach an equilibrium. The algorithm, as discussed in [424], allows each user to update its power by maximizing the net utility (with price) at that instance.

We remark that the pricing factors can be set by the base stations in two ways. A global pricing scheme can be used in which all base stations utilize the same cost factor, i.e., $c_j = c$, $\forall j \in \mathcal{K}$. The factor can be chosen so as to optimize the overall system utilities, for all users and all cells. Alternatively, a local pricing scheme can be used in which each base station $j \in \mathcal{K}$ chooses a pricing factor proportional to the traffic in its cell. In this local pricing scheme, the pricing factors at the base stations are different, so going through all vectors of the form $\mathbf{c} = [c_1, ..., c_K]$ to find the best pricing factor is impractical. Instead, we can utilize, as in [424], a pricing strategy in which each base station $i \in \mathcal{K}$ calculates its pricing factor in proportion to the number of users N_i in its cell, i.e., $c_i = \alpha N_i$, where α is a scalar. This choice allows the pricing vector to be varied by changing the scalar α .

So far we have fixed the base-station assignment as per (9.19), and studied the resulting multi-cell power-control game. In what follows, we extend the analysis by re-including the base-station assignment as part of the strategy space in the non-cooperative game, as per (9.18).

Power-control game with maximum-SINR base-station assignment

By re-inspecting the optimization problem in (9.18), we can see that the main difficulty is that the optimization space is two-dimensional: transmit power and base-station selection. In [424], this difficulty is reduced by showing that, given an interference vector \mathbf{p}_{-i} , for any user $i \in \mathcal{N}$ the following result holds:

$$\max_{p_i} \max_{a_i \in \mathcal{K}} u_{ia_i}(p_i, \mathbf{p}_{-i}).$$
(9.24)

The result in (9.24) allows us to simplify the joint power and base-station assignment problem by identifying the base-station assignment as a maximum-SINR assignment. Then, we select the transmit power that maximizes the utility for a user *i*, which is defined as

$$u_i(\mathbf{p}) = \max_{a_i \in \mathcal{K}} u_{ia_i}(p_i, \mathbf{p}_{-i})$$
(9.25)

Thus the optimization problem equivalent to (9.18) can be written as

$$\max_{p_i} u_i(p_i, \mathbf{p}_{-i}), \ \forall i \in \mathcal{N}.$$
(9.26)

At the equilibrium of the game as formulated in (9.26), the base-station assignment can be given as a maximum-SINR assignment, as follows:

$$a_i = \underset{j \in \mathcal{K}}{\arg\max} u_{ij}(\mathbf{p}) \equiv \underset{j \in \mathcal{K}}{\arg\min} \gamma_{ij}.$$
(9.27)

The problem in (9.26), referred to as a power-control continuous-kernel game under the maximum-SINR base-station assignment rule, is shown in [424] to have a utility function quasi-concave in the transmit power, which shows, using Theorem 3.2 in Chapter 3, that a Nash equilibrium exists for this game. Furthermore, by showing that the best-response functions for this game are standard (as in the game with fixed assignment), the uniqueness of the equilibrium is established. The powers at the equilibrium can be found as follows. Recall that $\gamma_{ia_i} = \tilde{\gamma}$ is the utility maximizer, and suppose that for all base-station assignments $\gamma_{ia_i} = \tilde{\gamma}$ is feasible for all $i \in \mathcal{N}$. Since at the equilibrium the users will attain the same SINR, the base-station assignment that maximizes the utility is the one that results in minimum transmit powers (such a power vector is shown to exist in [520]). In case $\gamma_{ia_i} = \tilde{\gamma}$ is not feasible for all *i*, then some terminals will not be able to reach the utility-maximizing SINR, so they transmit with maximum power, it follows that the equilibrium. If at the equilibrium the user transmits at maximum power, it follows that the equilibrium base-station assignment is the assignment with the highest SINR.

As with the base-station assignment case based on the maximum received signal strength, the Nash equilibrium can be inefficient, and pricing strategies can be implemented to improve this efficiency. For each user i, we can express the power-control game, under maximum-SINR base-station assignment, as

$$\max_{p_i} \left(u_i(p_i, \mathbf{p}_{-i}) - c_{a_i} p_i \right), \ \forall i \in \mathcal{N},$$
(9.28)

where $u_i(p_i, \mathbf{p}_{-i})$ is given by (9.25) and a_i is given by the assignment in (9.27). For this game with pricing, finding an analytical proof of the existence of the equilibrium is complex, as indicated in [424]. However, in [424], simulations were run to assess this existence, and the results suggest that such an equilibrium does exist. Furthermore, one can introduce global or local pricing. Under global pricing, as in the game with no pricing, the maximum-SINR and maximum-utility base-station assignment are equivalent. However, the actual base-station assignment at the equilibrium of the game with global pricing is possibly different from that for the game with no pricing. With no pricing, the utility is maximized for all users in the system. With pricing, each user targets a different SINR to maximize his net utility, depending on the value of the pricing factor and the location of the terminal within the system. As a result, the equilibrium base-station assignment could be different from the case without pricing. Certainly, with local pricing, the performance can be further improved, as discussed in [424].

From the simulations in [424] it was observed that, in general, the equilibrium utilities for the games with pricing increase as compared with the equilibrium utilities without pricing, regardless of the base-station assignment rule. Similarly, using pricing, the equilibrium transmit powers are smaller than those in the no-pricing case. For the maximum-SINR base-station assignment scheme, local pricing is shown to yield a minimum of around 25 percent of improved utility and 25 percent of power savings, relative to the no-pricing case.

9.2 Resource allocation in single-cell OFDMA networks

OFDMA is a promising multiple-access technique enabling high-data-rate transmission over wireless radio channels. Because of its potential, OFDMA has been chosen as the key multiple-access technique for next-generation wireless systems such as LTE [13] and WiMAX [487, 40]. In OFDMA networks, efficient resource allocation, involving bit loading, transmission power allocation, and subcarrier assignment, can greatly improve system performance, so it has drawn a great attention in recent research.

Numerous resource-allocation methods have been developed for OFDMA networks (see [187] and references therein). In its most basic form, the optimal (total-rate-maximizing) resource allocation under total-power constraints for a single user across parallel orthogonal channels can be found using the well-known water-filling method [116]. In single-cell, multi-user systems with a given set of subcarriers allocated to each user, the water-filling solution can also be applied since resource allocation can be considered independently for each user [187].

The problem becomes more complicated when dealing with a multi-user environment in which each user has a different channel and communication link quality. The difficulty stems from the discrete nature of the subcarrier-assignment problem. Nonetheless, by adaptively assigning subcarriers of various frequencies, one can take advantage of channel diversity (including independent path loss and fading) among users in different locations, i.e., the well-known concept of *multi-user diversity*. Exploring multi-user diversity has been shown to improve the efficiency of OFDMA systems under different objectives and constraints [187, 507, 477, 522, 83, 401, 508, 260]. Most of these approaches focus on efficiently maximizing the total transmission rate or minimizing the total transmitted power under certain constraints. The formulated problems and their solutions are mainly concerned with efficiency. In resource allocation, another important issue that arises when assigning subcarriers is *fairness*. In many system-optimizing approaches, users who have high power or who are closer to the base station are usually given higher benefit, which can be unfair to other users. In this regard, subcarrierassignment schemes that can not only improve the system performance but also provide a fair distribution of resources to the users are of interest. One example in which fairness is considered can be found in [401], where a max-min criterion has been defined for OFDMA channel allocation. Other fair approaches for OFDMA resource allocation are discussed in [187]. In a nutshell, when dealing with OFDMA channel allocation, it is imperative to develop a model that jointly considers fairness of resource allocation, system efficiency, and complexity.

In this section we develop, based on [183], a fair scheme to allocate subcarriers, rate, and power for a single-cell, multi-user OFDMA system. After presenting the studied system model, we develop, using bargaining theory, an algorithm for the two-user case that allows these users to bargain over the subcarrier usage. Then, we study and analyze a multi-user bargaining algorithm based on optimal groupings of pairs among the users.

9.2.1 OFDMA resource-allocation model

Consider the uplink of a single-cell multi-user OFDMA network with *N* users randomly located within the cell. Let \mathcal{N} denote the set of all users and \mathcal{K} the set of all subcarriers. The users want to share their transmissions among *K* different subcarriers, with each subcarrier having a bandwidth of \mathcal{W} . Each user *i*'s transmission rate R_i is allocated to different subcarriers, as $R_i = \sum_{j=1}^{K} r_{ij}$, where r_{ij} is the *i*th user's transmission rate in the *j*th subcarrier. Define the rate allocation matrix **r** with $[\mathbf{r}]_{ij} = r_{ij}$ and the subcarrier assignment matrix **A** with $[\mathbf{A}]_{ij} = a_{ij}$, where

$$a_{ij} = \begin{cases} 1, & \text{if } r_{ij} > 0, \\ 0, & \text{otherwise.} \end{cases}$$
(9.29)

For single-cell, multi-user OFDMA networks, no subcarrier can support the transmissions of more than one user, i.e., $\sum_{i=1}^{N} a_{ij} = 1, \forall j$.

By using adaptive modulation, each user *i* can match each subcarrier's transmission rate r_{ij} , according to its channel condition. For this purpose, we adopt for M-quadrature amplitude modulation (MQAM), high spectrum efficiency (without loss of generality). For MQAM, the BER can be approximated by a function of rate and SNR, as follows:

$$BER_{ij} \approx c_1 e^{-c_2 \frac{\gamma_{ij}}{2^{r_{ij}-1}}},$$
(9.30)

where $c_1 \approx 0.2$, $c_2 \approx 1.5$, and γ_{ij} is the *i*th user's SNR at the *j*th subcarrier, given by

$$\gamma_{ij} = \frac{p_{ij}h_{ij}}{\sigma^2},\tag{9.31}$$

where h_{ij} is the subcarrier channel gain and p_{ij} is the transmit power for the *i*th user on the *j*th subcarrier. The thermal noise power for each subcarrier is assumed to be the same and equal to σ^2 . We define a power-allocation matrix **p** with $[\mathbf{p}]_{ij} = p_{ij}$. From (9.30), without loss of generality, we assume the same BER for all users in all subcarriers. Then we have

$$r_{ij} = W \log_2\left(1 + \frac{p_{ij}h_{ij}c_3}{\sigma^2}\right), \qquad (9.32)$$

where $c_3 = c_2 / \ln(c_1 / \text{BER})$ with $\text{BER} = \text{BER}_{ij}, \forall i, j$.


Fig. 9.1 System model for a single-cell, multi-user OFDMA network.

We make a few assumptions regarding the system studied:

- The channel's fading is slow and stable within each OFDM frame.
- The channel conditions for different subcarriers for each user are assumed to be perfectly estimated.
- There exist reliable feedback channels from the base station to the users without any delay.
- The base station and the mobile users are synchronized, a common assumption (e.g., see [183]).

In Fig. 9.1, a three-user example is shown for the considered system with eight subcarriers (each subcarrier is occupied by one user). According to the channel conditions, a user selects an adaptive modulation level and adjusts its rate for a given subcarrier. The key challenge in this system is to design a subcarrier-allocation scheme while taking into account the conflicting situation that can arise when a certain subcarrier is good for more than one user. In this case, the scheme must be able to decide which user should take the subcarrier that is subject to a conflict of interest. Thus, the objective of this section is to devise a game-theoretic approach for allowing users to negotiate a subcarrier division so that each user can obtain its minimal rate while the system' overall performance is optimized.

We can see that the users have an incentive to agree on a division of the subcarriers (since otherwise they cannot communicate), but they are in conflict on how to do so. This situation is reminiscent of the bargaining situation in Chapter 7. For this multi-user OFDMA situation, we can formulate a bargaining problem among the users in \mathcal{N} that are bargaining over their rates. In fact, each user *i* aims to optimize its rate R_i , which is bounded from above and has a non-empty, closed, and convex support. In this bargaining problem, we let S denote the set of all feasible rates that satisfy $R_i \ge R_{\min}^i, \forall i$, where R_{\min}^i

is a minimum rate for user *i* (this can also be seen as a feasible range for the rate-allocation matrices **r**). To formally define a bargaining problem, we let the vector of minimal rates \mathbf{R}_{\min} represent the disagreement point. Thus, we have an (S, \mathbf{R}_{\min}) bargaining problem among the network users, and the objective is to choose an outcome in S that the users can agree on.

9.2.2 Nash bargaining solution for subcarrier allocation

As previously mentioned, the key conflict in the bargaining problem under consideration stems from the fact that a competition among the users arises whenever the channel conditions for a specific subcarrier are good for more than one user. The users thus compete over subcarriers having good channel gains. We consider that, in addition to the minimal rate R_{\min}^{i} , each user *i* has a maximum transmitted power p_{\max} . To solve the bargaining problem in this model, we need to determine different users' transmission functions and power values (for the different subcarriers), i.e., **A** and **P**. For this purpose, one suitable solution would be the Nash bargaining solution (NBS), which, as defined in Chapter 7, reduces to solving the following optimization problem:

$$\max_{\mathbf{A},\mathbf{P}} \prod_{i=1}^{N} \left(R_{i} - R_{\min}^{i} \right), \quad \text{s.t.} \begin{cases} \sum_{i=1}^{N} a_{ij} = 1, \forall j \in \mathcal{K}, \\ R_{i} \ge R_{\min}^{i}, \forall i, \\ \sum_{j=1}^{K} p_{ij} \le p_{\max}, \forall i \in \mathcal{N}. \end{cases}$$
(9.33)

Recall that the product $\prod_{i=1}^{N} (R_i - R_{\min}^i)$ is the Nash product. Note that although one can consider other criteria such as sum-rate maximization or max-min fairness, the choice of the Nash bargaining solution is made for two reasons. First, the Nash bargaining solution, as will be shown later in this section, is a generalized proportional fair solution. Second, as discussed in Chapter 7, a solution to the Nash bargaining problem, satisfying Nash's axioms, is shown to exist and to be Pareto-efficient.

To illustrate the use of the Nash bargaining solution, in Fig. 9.2 we show a two-user example where \mathbf{R}_{min} is assumed to be zero. The shaded area represents the feasible range S for R_1 and R_2 . For the Nash bargaining solution, the optimal point is B at $(\tilde{R}_1, \tilde{R}_2)$ with $R_1R_2 = \tilde{C}$, where \tilde{C} is the largest constant for the feasible set S. In a sense, the Nash bargaining solution implies that, after the users are assigned their minimum rates, the remainder of the resources would be divided between the users in proportion to the rates at which the utility can be transferred (see more insights on such an interpretation in [377]). A geometrical interpretation is that an isosceles triangle ABC can be drawn with $(\tilde{R}_1, \tilde{R}_2)$ as the apex, such that one of its sides is tangent to the set S and the other side passes by (R_{\min}^1, R_{\min}^2) , i.e., the origin in this example. Since line BC is also tangent to the curve $R_1R_2 = \tilde{C}$, the ratio with which two rates can be exchanged within the set S is equal to the ratio of the two rates. In the figure, we also show other possible solutions. For example, the solution that maximizes the social welfare, i.e., the total rate, occurs at $R_1^* + R_2^* = C^*$, that is, by definition, the point within the feasible set S where the sum C^* of R_1 and R_2 is maximized. Certainly the Nash bargaining solution has an overall rate that is slightly smaller than the maximal rate point, but this solution trades off this





performance loss in order to provide better fairness. The max-min fairness criterion that maximizes the worst-case rate is also shown in Fig. 9.2 at the point with $R'_1 = R'_2 = C'$, where C' is the largest constant for the feasible set S. Clearly, the overall rate for the Nash bargaining solution is higher than that in the max-min case.

As briefly mentioned in Chapter 7, when the disagreement point is zero, the Nash bargaining solution coincides with the famous proportional fair distribution of resources, which is defined as follows:

DEFINITION 9.1 A rate distribution is said to be proportionally fair whenever any change in the distribution of rates results in the sum of the proportional changes in the utilities becoming non-positive, i.e.,

$$\sum_{i} \frac{R_{i} - \tilde{R}_{i}}{\tilde{R}_{i}} \leq 0, \ \forall R_{i} \in \mathcal{S},$$
(9.34)

where \tilde{R}_i and R_i are, respectively, the proportionally fair rate distribution and any other feasible rate distribution for the *i*th user.

The relation between the Nash bargaining solution and proportional fairness can be formally drawn via the following theorem [183]:

THEOREM 9.3 When $R_{\min}^i = 0, \forall i$, the Nash bargaining solution coincides with the proportional fair rate distribution.

Proof The proof is taken from [183]. Taking the logarithm of the Nash product (recall that the logarithm function is concave and monotone) when $R_{\min}^i = 0, \forall i$, the Nash

bargaining solution is equivalent to

$$\max_{\mathbf{r}\in\mathcal{S}}\sum_{i=1}^{K}\log(R_i).$$
(9.35)

Define $\hat{U}_i = \log(R_i)$. The gradient of \hat{U}_i at the Nash bargaining solution \tilde{R}_i is $\frac{\partial \hat{U}_i}{\partial R_i}|_{\tilde{R}_i}$. Since the Nash bargaining solution optimizes (9.35), for any point deviating from the Nash bargaining point, the following optimality condition holds:

$$\sum_{i} \frac{\partial \hat{U}_{i}}{\partial R_{i}} |_{\tilde{R}_{i}} (R_{i} - \tilde{R}_{i}) = \sum_{i} \frac{R_{i} - \tilde{R}_{i}}{\tilde{R}_{i}} \leq 0.$$
(9.36)

The above equality implies that, for all feasible $R_i \in S$, $\forall i$, different from Nash bargaining point \tilde{R}_i , the overall change of benefits is negative, according to the gradients. In fact, the above equation is the same as the definition of proportional fairness in (9.34).

Thus, proportional fairness is a special case of the Nash bargaining solution when $R_{\min}^{i} = 0, \forall i$. In practice, a minimum rate is desired by the users, so we focus on the Nash bargaining solution, which, as already shown, encompasses the proportional fair case.

In [183] it is shown that there exists a unique and optimal solution to (9.33), when the feasible set satisfying the constraints is not empty. This is shown in two steps. Uniqueness and optimality are proven when the channel assignment matrix **A** is fixed. But it is shown in [183] that the probability of having more than one optimal point is zero for different channel assignment matrices **A**.

Using the concept of the Nash bargaining solution, we have studied the outcome of the modeled bargaining problem. The next step is to devise algorithms for finding this solution.

9.2.3 Algorithms for reaching the Nash bargaining solution

Two-user case

For the case of two users, i.e., N = 2, we can develop an algorithm to reach the Nash bargaining solution. As with bargaining in a real market, the idea is to allow two users to negotiate and exchange their subcarriers for mutual benefits. The main challenge is to determine how to optimally exchange subcarriers, which can be considered a complex integer programming problem. To solve this problem, one can use Algorithm 9.1, which is developed in [183]. First, all subcarriers are initially assigned and, for each user *i*, a positive weight factor ρ_i is computed as follows (ϵ is a small positive number):

$$\varrho_i = \begin{cases} \frac{1}{R_i - R_{\min}^i}, & \text{if } R_i \ge R_{\min}^i + \epsilon, \\ \frac{1}{\epsilon}, & \text{otherwise.} \end{cases}$$
(9.37)

Then, the two users' subcarriers are sorted and a two-band partition algorithm is applied for them to negotiate the exchange of subcarriers. To achieve the Nash bargaining solution, an intermediate parameter needs to be updated for every iteration. From Algorithm 9.1 Two-user algorithm for finding the Nash bargaining solution.

1. Initialization:

Initialize the subcarrier assignment with the minimum rate requirements. For the Nash bargaining solution, calculate ρ_1 and ρ_2 .

2. Sort the subcarriers:

Arrange the indices from largest to smallest $\frac{g_{1j}^{e_1}}{g_*^{e_2}}$.

3. For j = 1, ..., K - 1

User 1 occupies and water-fills subcarriers 1 to *j*;

User 2 occupies and water-fills subcarriers j + 1 to K.

Calculate the Nash product in (9.33).

End

4. Choose the two-band partition (the corresponding *j*) that generates the largest Nash product satisfying the constraints.

Calculate \mathbf{A} , \mathbf{p} , R_1 , and R_2 .

5. Update channel assignment.

If the Nash product cannot be increased by updating ρ_1 and ρ_2 , the iteration ends; otherwise, update $\rho_1 = 1/(R_1 - R_{\min}^1)$, $\rho_2 = 1/(R_2 - R_{\min}^2)$; go back to step 2.

simulations, as shown in [183], the iterations between steps 2 and 5 converge within two to three rounds. The algorithm has a complexity of $O(N^2)$ for each iteration and can be further improved by using a binary search algorithm with a complexity of only $O(N \log N)$ for each iteration. Notice that all the iterations in Algorithm 9.1 are performed within the base station, so there is no need for signalling between users and base stations.

For Algorithm 9.1, we highlight the following result, which is shown in [183]:

PROPOSITION 9.1 Algorithm 9.1 is nearly optimal for finding the Nash bargaining solution in (9.33) with N = 2 users when the SNR of each subcarrier for all users in (9.31) is much greater than 1 and there exists a feasible solution.

Multiple-user case

For the case in which N > 2, the computational complexity for allocating the subcarriers is quite high, and for this reason most researchers have approached the problem from a centralized perspective (see [187, 507, 477] and references therein). To approach the problem in a distributed manner, we can use an iterative scheme, as in [183], based on the following steps:

- 1. The users are grouped into pairs.
- 2. For each pair, Algorithm 9.1 is applied for the two users to negotiate and improve their performances by exchanging subcarriers.

After the two steps are completed, the users are regrouped and they renegotiate again and again until convergence. The remaining key question is how to group users into pairs. One straightforward approach is to form pairs randomly and let the users bargain in an arbitrary manner. This is referred to as the *random method*, and is described in

Algorithm 9.2 Multi-user subcarrier allocation algorithm.

1. Initialize the channel assignment:

Assign all subcarriers to users.

2. Group the pairs:

- If the number of users is even, the users are grouped into coalitions; otherwise, a dummy user is created to make the total number of users even. No user can exchange its resource with this dummy user.
- Random method: randomly form groups of two users.
- Hungarian method: form user pairs using the Hungarian algorithm.

3. Bargain within each pair:

Users in each pair negotiate to exchange subcarriers using two-user Algorithm 9.1.

4. Repeat:

Repeat steps 2 and 3 until no further improvement is achieved.

Algorithm 9.2. In the initialization stage, the objective is to assign all subcarriers to users while trying to satisfy the minimum-rate and maximum-power constraints. If the user having the best channel conditions has a rate that is higher than or equal to R_{\min}^{i} , then it is removed from the assignment list. After each user is given a sufficient rate, the remaining subcarriers are greedily assigned to the users according to their channel gains. Note that there is no need for the initial assignment to satisfy all the constraints; these can be satisfied during subsequent iterations (i.e., negotiations).

Despite its straightforward implementation, the convergence speed of the random method, evaluated by the number of rounds of negotiations, decreases as the number of users increases. As discussed in [183], this is because most negotiations within arbitrarily grouped pairs end up with little or no improvement compared with the means the channel allocation prior to the start of negotiations. Therefore, one must consider an alternative for grouping the users into pairs. It is shown in [183] that this problem can be mapped into a well-known assignment problem, as follows [259].

The benefit for the *i*th user in negotiating with the *j*th user is defined as b_{ij} . Obviously $b_{ii} = 0, \forall i \in \mathcal{N}$ because no user would negotiate with itself. From (9.33), the remaining elements of the benefit matrix **b** can be expressed as

$$b_{ij} = \max(U(\tilde{R}_i, \tilde{R}_j) - U(\hat{R}_i, \hat{R}_j), 0), \qquad (9.38)$$

where \tilde{R}_i and \tilde{R}_j are the respective rates if the negotiation happens, and \hat{R}_i and \hat{R}_j are the original rates. $U(R_i, R_j) \triangleq (R_i - R_{\min}^i) (R_j - R_{\min}^j)$ is the Nash product. We note that **b** is symmetric.

Now, define a $N \times N$ assignment matrix **X**. Each component of this matrix represents whether or not there is a coalition between two users, as follows:

$$X_{ij} = \begin{cases} 1, & \text{if user } i \text{ negotiates with user } j, \\ 0, & \text{otherwise.} \end{cases}$$
(9.39)

This matrix is clearly symmetric: $\sum_{i=1}^{N} X_{ij} = 1, \forall j \in \mathcal{N}$, and $\sum_{j=1}^{N} X_{ij} = 1, \forall i \in \mathcal{N}$.

Algorithm 9.3 The Hungarian algorithm.

- 1. Subtract the entries of each row of **B** by the row minimum, so that each row has at least one zero and all entries are positive or zero.
- 2. Subtract the entries of each column by the column minimum, so that each row and each column has at least one zero.
- 3. Select rows and columns across which lines are drawn, in such a way that all the zeros are covered and that no more lines have been drawn than necessary.
- 4. Test for optimality:
 - (i) If the number of the lines is *N*, choose a combination **A** from the modified cost matrix in such a way that the sum is zero.
 - (ii) If the number of the lines is less than *N*, go to step 5.
- 5. Find the smallest element that is not covered by any of the lines. Then subtract it from each entry that is not covered by the lines and add it to each entry that is covered by a vertical and a horizontal line. Go back to step 3.

The assignment problem involves selecting pairs of users that must negotiate while maximizing the overall benefit, i.e.,

$$\max_{\mathbf{X}} \sum_{i=1}^{N} \sum_{j=1}^{N} X_{ij} b_{ij}, \text{ s.t.} \begin{cases} \sum_{i=1}^{N} X_{ij} = 1, \quad j = 1, \dots, N, \forall i \in \mathcal{N}; \\ \sum_{j=1}^{N} X_{ij} = 1, \quad i = 1, \dots, N, \forall j \in \mathcal{N}; \\ X_{ij} \in \{0, 1\}, \quad \forall i, j \in \mathcal{N}. \end{cases}$$
(9.40)

In order to solve (9.40) and obtain the optimal user pairs, one can use the *Hungarian method* as presented in [183]. As the Hungarian method typically deals with minimization problems, for convenience we map the maximization problem in (9.40) into a minimization problem. To do so, we define $B_{ij} = -b_{ij} + \max(b_{ij})$. The Hungarian algorithm (Algorithm 9.3) can then be used to find the optimal pairs of users.

Thus, in each round of negotiations, the optimal pairs **A** are determined by the Hungarian method and then the users bargain using the two-user Algorithm 9.1. The algorithm ends when no bargaining can further improve the performance, i.e., **b** is equal to a zero matrix. Since, in each iteration, the optimization function U (i.e., the Nash product) is nondecreasing in steps 2 and 3, the optimal solution is upper bounded. Consequently, the multi-user algorithm will always converge, as discussed in [183]. Nonetheless, because of the non-linearity and non-convexity of the problem in (9.33) and despite the optimality of the Hungarian method, the algorithm can end up with a local optimum. However, it is shown in [183] that these local optima can still have a reasonable performance.

The complexity of the Hungarian method is $O(N^4)$. As a result, the overall complexity for each iteration of the studied scheme is $O(N^2 \ K \log_2 K + N^4)$. Since the number of users is, in general, much smaller than the number of subcarriers, the complexity of the algorithm is reasonable compared with other methods [183]. Moreover, the practical implementation of the algorithm requires little or no signalling between users.

In [183], extensive simulations were run to assess the performance of the algorithms for finding Nash bargaining solutions in the two- and multiple-user cases. From these



Fig. 9.3 Total rate (Mbps) achieved by the system as the number of users *N* varies.

simulations, given an OFDMA network with 128 subcarriers over the 3.2 MHz band and where each user has a minimum rate requirement of $R_{\min}^i = 25$ kbs for each *i*, we show in Fig. 9.3 the sum of all users' rates as the the number of users in the system varies. The figure compares the performances of the Nash bargaining solution, the maximum-rate solution, and the max-min fair solution. As the number of users increases, the performance of all three schemes obviously increases. This gain is a result of the multi-user diversity provided by the independent variation of channels across different users. The Nash bargaining solution shows a performance comparable to that of the scheme that maximizes the total rate outperforms the max-min scheme. The performance gap between the maximum-rate scheme and the Nash bargaining solution becomes smaller as the number of users increases, since more choices for bargaining pairs become available.

In Fig. 9.4, we show a histogram of the number of rounds necessary for convergence of the random method and the Hungarian method for a network with eight users. The Hungarian method converges in about one to six rounds, while the random method may converge very slowly. The average number of convergence rounds for the random method is 4.25 times larger than that for the Hungarian method. This is because the Hungarian method can find, quickly, the best pairs to group during negotiations. Further simulations in [183] demonstrated the advantages of this algorithm.

In summary, this subsection has showed how one can formulate a bargaining problem within the context of subcarrier allocation in a single-cell OFDMA network. While the focus has mainly been on single-cell OFDMA networks, game-theoretic approaches can also be used within the context of multi-cell OFDMA networks. The reader is referred to [184] for further information on how non-cooperative-game theory and the Nash



Fig. 9.4 Histogram of convergence for the studied algorithm.

equilibrium solution can be used to solve problems of resource allocation in multi-cell OFDMA networks.

9.3 Power allocation in femtocell networks

Imperfect network coverage, especially in indoor locations such as buildings and houses, is a key problem in existing wireless systems such as cellular networks. Traditionally, in order to solve this problem, the network operator would deploy additional base stations to increase coverage. Because of the high cost of such a deployment, operators are reluctant to install new base stations, notably in areas that are not too dense. To overcome this problem, the concept of *femtocell access points* (FAPs) has recently emerged as a means to overlay, on existing cellular network technologies (e.g., 2G, 3G, WiMAX), *low-power* and *low-cost* base stations. Femtocell access points are connected by an IP backhaul network through a local broadband connection such as DSL, cable, or fiber. Femtocell access points could, for example, be low-cost, plug-and-play devices similar to WiFi access points.

Various benefits of using FAPs have been identified [99, 521, 250]. First, the deployment of FAPs improves indoor coverage where the signal of a macro-cell base station can be weak. Second, FAPs provide high data rates and improved quality of service to subscribers, while at the same time ensuring longer battery life for mobile devices. The extended battery life arises from the fact that, in a femtocell environment, mobile devices do not need to communicate with a distant base station. Finally, from the operator's perspective, deploying femtocell access points saves the backhaul cost since FAP traffic is carried over wired residential broadband connections. In fact, for the operators, FAPs can be seen as a new way towards the convergence of landline and mobile services. For a general survey of this technology, the reader is referred to [99, 521, 250].

Recently, research has turned to deploying femtocell access points in wireless networks. In [98], uplink capacity and interference avoidance for two-tier femtocell networks are considered. Downlink power control in a femtocell network is considered in [292], where the objective is to minimize the transmit power under signal-to-noise constraint. Similarly, [100] deals with additional power-allocation approaches in femtocell networks. The work in [233] deals with bandwidth partitioning in a femtocell-like network.

When FAPs are deployed on top of an existing cellular system, one can envision the emergence of a hierarchical overlay network. In such a hierarchical network, it is of interest to study the problem of transmit-power control in the downlink. Since FAPs are expected to operate on the same frequency bands as macrocell base stations, cochannel interference can impede the overall performance of the network. As the number of FAPs increases, the accumulated interference becomes a critical issue. To highlight the importance of power control in such a network deployment, one can consider the interesting situation in which a mobile user is connected to an FAP while another cell-edge user is connected to a base station. In this case, there is strong interference from the base station to the mobile within the coverage of the FAP. As the cell-edge user is outside the coverage of the FAP, it can only be served by the base station, requiring a large transmit power. As a result, suitable power-control schemes need to account for the fact that some users are served by FAPs while others are served by macrocell base stations.

In this section, we adopt the approach of [178] for studying this power-control problem from a game-theoretic perspective. First, we model the problem as a Stackelberg game. Then, we discuss the properties of the considered game and its solution. Finally, we develop a low-complexity algorithm to reach the desired outcome.

9.3.1 Femtocell power control as a Stackelberg game

Consider a system of base-station transceivers and femtocell access points. Let the set of base-station transceivers be given by $\mathcal{M} = \{1, ..., M\}$ and the set of femtocell access points by $\mathcal{N} = \{1, ..., N\}$, where \mathcal{M} and \mathcal{N} are two disjoint sets. We further assume that there is a wire-line backhaul connecting the femtocell access points to the base-station transceivers, enabling them to exchange relevant information. Fig. 9.5 illustrates the deployment of such FAPs within a wireless cellular network.

For multiple access, we consider that both the base stations and the FAPs utilize OFDMA, with the spectrum divided in such a way that each transceiver $i \in \mathcal{M} \cup \mathcal{N}$ is assigned a set of orthogonal channels $\mathcal{L}_i = \{1, ..., L_i\}$. Each channel has the same bandwidth W_i . For any $i, j \in \mathcal{M} \cup \mathcal{N}, k \in \mathcal{L}_i, l \in \mathcal{L}_j$, let h_{ij}^{kl} be the link gain for channel *l* of transmitter *j* to channel *k* of receiver *i*. We assume that the channels are slow, flatfading channels. The transmitters estimate the channel gain based on feedback from the receivers, and the channel state does not change within a given slot. The orthogonality of the channels of user *i* ensures that the link gain $h_{ii}^{kl} = 0$ for $k \neq l, k, l \in \mathcal{L}_i$. Note that the orthogonality of the channels does not need to be retained for different users.



Fig. 9.5 Example of the deployment of femtocell access points within a wireless cellular network.

The noise is assumed to be additive white Gaussian, with power n_i^k in channel k of receiver *i*. Let the transmit power of transmitter *j* in channel *l* be denoted by p_j^l . The interference and noise power as observed by receiver *i* in channel k is given by

$$\nu_{i}^{k} = \sum_{j \neq i} \sum_{l \in \mathcal{L}_{j}} h_{ij}^{kl} p_{j}^{l} + n_{i}^{k}.$$
(9.41)

Normalizing the link gains and noise powers such that $h_{ii}^{kk} = 1$ for all $i \in \mathcal{M} \cup \mathcal{N}$ and $k \in \mathcal{L}_i$, the SINR of receiver *i* in channel *k*, which equals $\frac{h_{ii}^{kk} \rho_i^k}{\nu_i^k}$, can be simplified as

$$\frac{p_i^k}{\nu_i^k} = \frac{p_i^k}{\sum_{j \neq i} \sum_{l \in \mathcal{L}_j} h_{ij}^{kl} p_j^l + n_i^k}.$$
(9.42)

If all users divide the spectrum in the same manner, that is, the number of channels are identical, and if the overlapping channels share the same spectrum,¹ then we say that the channel is a *parallel Gaussian interference channel*. For such a channel we have $L_i = L_j$ for every $i, j \in \mathcal{M} \cup \mathcal{N}$, and the link gain $h_{ij}^{kl} = 0$ whenever $k \neq l$. Hence, the SINR of receiver *i* at channel *k* simplifies to

$$\frac{p_i^k}{\nu_i^k} = \frac{p_i^k}{\sum_{j \neq i} h_{ij}^{kk} p_j^k + n_i^k}.$$
(9.43)

Let $\mathbf{p}_i = [p_i^1, \dots, p_i^{L_i}]^T$ be the transmit *power vector* of transmitter *i* and $\boldsymbol{\nu}_i = [\nu_i^1, \dots, \nu_i^{L_i}]^T$ be the *interference (plus noise) vector* of receiver *i*. The power of each

¹ Here we assume that frequency re-use is equal to 1 among users in different cells, along with uniform and random frequency-hopping patterns, where full overlapping is allowed.

user *i* is subject to the total power constraint $\sum_{k \in \mathcal{L}_i} p_i^k \leq \bar{p}_i$, and individual power constraint (also known as *spectral mask*) $p_i^k \leq \bar{m}_i^k$ for all $i \in \mathcal{N} \cup \mathcal{M}$ and $k \in \mathcal{L}_i$. We assume that $\bar{p}_i \leq \sum_{k \in \mathcal{L}_i} \bar{m}_i^k$ for all *i*, so as to avoid trivial cases. If this inequality does not hold, then $p_i^k = \bar{m}_i^k$ for all *i* and *k*. Let \mathcal{P}_i denote the set of all feasible power vectors of transmitter *i*:

$$\mathcal{P}_{i} = \left\{ \mathbf{p}_{i} \in \prod_{k \in \mathcal{L}_{i}} [0, \bar{m}_{i}^{k}] : \sum_{k \in \mathcal{L}_{i}} p_{i}^{k} \leq \bar{p}_{i} \right\}.$$
(9.44)

We assume that the base stations and FAPs seek to allocate their transmit power so as to maximize the total throughput. Given the power-allocation vector, from Shannon's capacity formula for additive white Gaussian channels, the maximal data rate that user *i* can achieve is

$$C_{i} = C_{i}(\mathbf{p}_{1}, \dots, \mathbf{p}_{i}, \dots, \mathbf{p}_{M+N}) = C_{i}(\mathbf{p}_{i}, \boldsymbol{\nu}_{i})$$
$$= W_{i} \sum_{k \in \mathcal{L}_{i}} \log\left(1 + \frac{p_{i}^{k}}{\nu_{i}^{k}}\right).$$
(9.45)

In order to tackle the power-control problem using game theory, we utilize the framework of a Stackelberg game as defined in Chapter 3. Recall that in a Stackelberg game a hierarchy exists between the players in which a subset of the players can be designated as leaders of the game, i.e., players that announce their strategy before the other players, known as followers, choose their strategy.

In the studied femtocell deployment model, we consider that the macrocell base stations are the leaders and the FAPs are the followers in a Stackelberg game. While in Chapter 3 we dealt mainly with games having a single leader and a single follower, treatment of a multi-leader multi-follower Stackelberg game follows the same line of analysis. In essence, in a multi-leader multi-follower Stackelberg game, there exists a competitive game between the leaders and the followers, a competitive game between the leaders themselves, and a competitive game between the followers themselves. Nonetheless, the game maintains a distinct hierarchy between leaders and followers such that the leaders can anticipate, and take into consideration, the behavior of the followers before making their own moves. The followers do not have this power to anticipate the leaders' moves.

To model the studied femtocell problem, we consider a Stackelberg game with complete and perfect information. As already mentioned, the leaders are the set of macrocell base-station transceivers \mathcal{M} , the followers the set of femtocell access points \mathcal{N} . Therefore, the total set of players in the Stackelberg game is $\mathcal{M} \cup \mathcal{N}$.

The strategy space of the leaders is given by $\mathcal{P}^{up} = \prod_{i \in \mathcal{M}} \mathcal{P}_i$, and any point in \mathcal{P}^{up} is called a leader strategy. The leaders compete with each other in a non-cooperative manner in order to maximize their individual throughput, while at all times anticipating the strategic responses of the followers. This game among the leaders will be referred to as the *upper subgame*, and its equilibrium will be referred to as the *upper subgame*, and its equilibrium will be referred to as the *upper subgame*, and its equilibrium will be referred to as the *upper subgame*, and its equilibrium will be referred to as the *upper subgame* equilibrium. After the leaders apply their strategies, the followers make their moves in response to the leaders' strategies. The strategy space of the followers is $\mathcal{P}^{low} = \prod_{i \in \mathcal{N}} \mathcal{P}_i$, and any point in \mathcal{P}^{low} is called a follower strategy. The followers also compete with

each other in a non-cooperative manner to maximize their own throughput² and this competition among the followers is referred to as the *lower subgame*, with its equilibrium designated as the *lower subgame equilibrium*. Finally, the strategy space of the entire game is given by the Cartesian product $\mathcal{P} = \mathcal{P}^{up} \times \mathcal{P}^{low}$.

For any user $i \in \mathcal{M} \cup \mathcal{N}$, we define the best-response function:

$$\mathbf{p}_{i} = \underset{\mathbf{p}_{i}}{\operatorname{argmax}} C_{i}(\mathbf{p}_{i}, \mathbf{p}_{-i})$$
$$= b_{i}(\mathbf{p}_{-i}; \bar{p}_{i}, \bar{\mathbf{m}}_{i}), \qquad (9.46)$$

where $\bar{\mathbf{m}}_i = (\bar{m}_i^k)_{k \in \mathcal{L}_i}$ is chosen so as to maximize user *i*'s capacity function subject to the power constraints. The notation -i refers to all of the users in the set $\mathcal{M} \cup \mathcal{N}$ except user *i*.

We define the lower subgame equilibrium as any fixed point $\mathbf{p}^{\text{low}^*} = (\mathbf{p}_1^*, \dots, \mathbf{p}_N^*) \in \mathcal{P}^{\text{low}}$ such that

$$\mathbf{p}_i^* = b_i \big(\mathbf{p}_{-i}^*, \mathbf{p}^{\mathrm{up}}; \bar{p}_i, \bar{\mathbf{m}}_i \big), \tag{9.47}$$

where $\mathbf{p}^{up} \in \mathcal{P}^{up}$ is a *fixed* but *arbitrary* leader strategy, for all $i \in \mathcal{N}$. Note that this definition is the same as a Nash equilibrium (of the lower subgame).

For any user $i \in \mathcal{M} \cup \mathcal{N}$, assuming that the set \mathcal{L}_i is partitioned into three disjoint subsets,

- A subset A_i that contains all channels that are active, i.e., having an optimal (capacity-maximizing) non-zero power (after water-filling) that is strictly lower than the upper bound m
 _i^k
- A subset S_i that contains all active channels that are *saturated*, i.e., where the power is equal to the upper bound \bar{m}_i^k
- The subset $\mathcal{L}_i \setminus \mathcal{A}_i \cup \mathcal{S}_i$ of all remaining channels (neither active nor saturated).

Since every user in the lower subgame will myopically maximize their individual throughput, the best response $b_i(\cdot)$ of each user in the subgame will be given by the following water-filling function:

$$\mathbf{p}_{i} = F(\mathbf{p}_{-i}; \bar{p}_{i}, \bar{\mathbf{m}}_{i})$$

$$= \mathbf{W}_{i}(\mathcal{A}_{i})\boldsymbol{\nu}_{i} + \mathbf{r}_{i}(\mathcal{A}_{i}, \mathcal{S}_{i}),$$
(9.48)

where $\mathbf{W}_i(\mathcal{A}_i)$ is an $L_i \times L_i$ symmetric matrix whose (k, l)-th member is given by

$$[\mathbf{W}_i(\mathcal{A}_i)]_{kl} = \begin{cases} 0 & \text{if } k \text{ or } l \notin \mathcal{A}_i, \\ \frac{1}{|\mathcal{A}_i|} & \text{if } k, l \in \mathcal{A}_i \text{ and } k \neq l, \\ -1 + \frac{1}{|\mathcal{A}_i|} & \text{if } k, l \in \mathcal{A}_i \text{ and } k = l, \end{cases}$$

² This can be referred to as the "noisy-neighbors" problem.

and $\mathbf{r}_i(\mathcal{A}_i, \mathcal{S}_i)$ is an L_i -dimensional column vector given by

$$r_i^k(\mathcal{A}_i, \mathcal{S}_i) = \begin{cases} 0 & \text{if } k \notin \mathcal{A}_i \cup \mathcal{S}_i, \\ \bar{m}_i^k & \text{if } k \in \mathcal{S}_i, \\ \frac{1}{|\mathcal{A}_i|}(\bar{p}_i - \sum_{l \in \mathcal{S}_i} \bar{m}_i^k) & \text{if } k \in \mathcal{A}_i, \end{cases}$$

which is analytically derived in [178]. We note that several algorithms can be used for finding the elements of sets A_i and S_i , as discussed in [178].

By letting $b^{\text{low}} \equiv (b_i(\cdot))_{i=1}^N$, we can express the lower subgame equilibrium as any fixed point of the system-power space $\mathbf{p}^* \in \mathcal{P}$ such that

$$\mathbf{p}^* = b^{\text{low}}(\mathbf{p}^*). \tag{9.49}$$

Note that the function $b^{\text{low}}(\cdot)$ does not impact the upper subgame strategy.

We now define the upper subgame equilibrium as any fixed point $\mathbf{p}^{up*} = (\mathbf{p}_1^*, \dots, \mathbf{p}_M^*) \in \mathcal{P}^{up}$ such that

$$\mathbf{p}_i^* = b_i (\mathbf{p}_{-i}^*, \mathbf{p}^{\text{low}^*}; \bar{p}_i, \bar{\mathbf{m}}_i), \qquad (9.50)$$

where $\mathbf{p}^{\text{low}*} \in \mathcal{P}^{\text{low}}$ is an *equilibrium* follower strategy conditioned on the upper subgame strategy, for all $i \in \mathcal{M}$. Equivalently, let $b^{\text{up}} \equiv (b_i(\cdot))_{i=1}^M$; then we can define the upper subgame equilibrium as the fixed point $\mathbf{p}^{\text{up}*} \in \mathcal{P}^{\text{up}}$ such that

$$\mathbf{p}^{\text{up}*} = b^{\text{up}}(\mathbf{p}^{\text{up}*}; b^{\text{low}}(\mathbf{p}^{\text{low}*}; \mathbf{p}^{\text{up}*})).$$
(9.51)

For convenience, the notation can be further simplified by writing the upper subgame equilibrium in terms of a system-power vector, i.e., as any fixed point $\mathbf{p}^* \in \mathcal{P}$ such that

$$\mathbf{p}^* = b^{\mathrm{up}}(b^{\mathrm{low}}(\mathbf{p}^*)). \tag{9.52}$$

Note that although the function $b^{up}(\cdot)$ acts only on the upper subgame strategy, the lower subgame equilibrium strategy (the reaction of the followers, in Stackelberg game terminology) associated with each upper subgame strategy needs to be computed as well, since the leaders compute their strategies given their knowledge of what the followers might play.

9.3.2 Multi-leader multi-follower Stackelberg equilibrium

A suitable solution for the formulated hierarchical non-cooperative game between the base stations and the FAPs is the Stackelberg equilibrium. In such a multi-leader multi-follower game, the Stackelberg equilibrium is defined as any fixed point $(\mathbf{p}^{up^*}, \mathbf{p}^{low^*}) = \mathbf{p}^* \in \mathcal{P}$ that satisfies (9.49) and (9.52). In other words, let $b : \mathcal{P} \to \mathcal{P}$ be a composition of the two vector functions:

$$b \equiv b^{\rm up} \circ b^{\rm low}$$

Then we have the Stackelberg equilibrium as any fixed point of the function b,

$$\mathbf{p}^* = b(\mathbf{p}^*),\tag{9.53}$$

such that $\mathbf{p}^* = (\mathbf{p}^{up^*}, \mathbf{p}^{low^*})$. This definition of a Stackelberg equilibrium in a multi-leader multi-follower setting is analogous to the notion of a subgame-perfect equilibrium, which is a refinement of the Nash equilibrium for dynamic games discussed in Chapter 3.

It is important to note that the best-response function for the lower sub-game, $b^{\text{low}}(\cdot) \equiv (F_i(\cdot))_{i=1}^N$, where $F_i(\cdot)$ is given by the water-filling function from (9.49), is a piecewise affine continuous function of **p** [178]. By assuming the continuity of the best-response function of the upper subgame b^{up} , we can prove the existence of a Stackelberg equilibrium in the studied game, using the Schauder Fixed-Point Theorem stated below [169]:

THEOREM 9.4 (Schauder Fixed-Point Theorem) Every continuous function from a convex compact subset \mathcal{K} of a Banach space to \mathcal{K} itself has a fixed point.

Using the Schauder Fixed-Point Theorem, [178] showed that the following result holds:

THEOREM 9.5 In the studied multi-leader multi-follower power-allocation game, given that the best-response function of the upper subgame is continuous, at least one Stackelberg equilibrium exists.

Proof The proof is from [178]. Since the best-response functions $b^{up} : \mathcal{P} \to \mathcal{P}$ and $b^{low} : \mathcal{P} \to \mathcal{P}$ are continuous functions, the composition of these two functions, $b = b^{up} \circ b^{low}$, is also continuous. It is also easy to see that \mathcal{P} (which is a Cartesian product) is convex, closed, and bounded, and, being finite-dimensional, it is convex compact. Since a Stackelberg equilibrium is defined as any fixed point of $b(\cdot)$, using the Schauder Fixed-Point Theorem we have that the studied game will admit at least one Stackelberg equilibrium.

From this theorem, we can immediately remark that the following holds [178]:

COROLLARY 9.1 At least one upper subgame and one lower subgame equilibrium exist.

For instance, it can be shown that the lower subgame equilibrium is, in fact, unique. It is straightforward to check that, for any fixed leader strategy in \mathcal{P}^{up} , the sum of capacities $\sum_{i \in \mathcal{N}} C_i$ for the lower game is diagonally strictly concave. Therefore, from Theorem 3.3 in Chapter 3, for any given leader strategy there exists at most one lower subgame equilibrium. However, the uniqueness of the upper subgame equilibrium cannot be guaranteed.

If, however, the leaders also adopt their strategies based on the water-filling function, assuming the interference vector to be constant, then the Stackelberg equilibrium will actually coincide with the Nash equilibrium of the whole game in static form (in this case, there would no longer be any hierarchy). In such a case, the upper subgame would then be a concave game as well, thus guaranteeing the uniqueness of the Nash equilibrium.

9.3.3 Algorithm for reaching the Stackelberg equilibrium

Finding, iteratively, the fixed point of the lower subgame using the water-filling algorithm usually yields an unstable system for a random channel gain matrix. In fact, the system is stable only under specific conditions on the water-filling function.³ Therefore, based on [178], in order to ensure that a stable system is reached, one can use the following iterative technique, known as the *Mann iterative methods*, which allows a weaker stability criterion:

$$\mathbf{p}_i(t+1) = (1-\lambda(t))\mathbf{p}_i(t) + \lambda(t), F_i(\mathbf{p}_{-i}(t))$$
(9.54)

such that the scalar sequence $\{\lambda(t)\}$ satisfies [69]

- $\lambda(t=0)=1$
- $\lambda(t) \in (0, 1)$ for t > 0
- $\sum_{t=0}^{\infty} \lambda(t) = \infty.$

In order to reach the lower subgame equilibrium iteratively, one can assume the set \mathcal{P}^{low} to be a uniformly convex Banach space and $F(\cdot)$ to be a quasi-non-expansive operator on \mathcal{P}^{low} (provided that $\{\lambda(t)\}$ is *bounded away* from 0 and 1; see [69, Theorem 4.5]). The $\{\lambda(t)\}$ sequence is generated as

$$\lambda(t) = \frac{t}{2t+1}, \quad t > 0.$$
 (9.55)

It is straightforward to verify that the sequence satisfies the constraints on $\lambda(t)$ as $\lim_{t\to\infty} \lambda(t) = 1/2$.

In contrast to the lower subgame equilibrium, no simple method is known for computing the upper subgame equilibrium. This is, as discussed in [178], mainly because, every time the leaders' strategy is updated, the lower subgame equilibrium needs to be computed. Such a problem is inherently difficult to solve, although a number of algorithms have been proposed. In [467, Algorithm 1], the authors suggest a Lagrangian dual approach for computing the best response of a leader. The idea is to approximate the Lagrangian dual function by locally optimizing the Lagrangian with respect to the individual frequency bin while keeping the power in other bins constant, for a fixed, dual variable. This algorithm then updates the dual variable by using the bisection search and repeats the procedure until convergence is achieved. However, the algorithm is very sensitive to the initial starting point and the ordering of iterations. Also, since it does not perform an exhaustive search, it is, at best, suboptimal since it does not decouple the power allocated in each frequency bin from the interference generated in the other frequency bins to compute the Lagrangian dual function.

For the case in which both the leaders and the followers use the water-filling function as their best response, i.e., the case where the Nash equilibrium and the Stackelberg equilibrium coincide, the equilibrium can be found iteratively as follows. First, the

³ In fact, for the system to be stable, the water-filling function should be a contraction (see [456]).

upper subgame equilibrium is computed, assuming an initial lower subgame equilibrium. The lower subgame equilibrium is then computed while keeping the upper subgame equilibrium fixed. Subsequently, the upper subgame and the lower subgame equilibrium are iteratively computed until the system-power vector of the whole game converges. As mentioned earlier, the Mann iterative method is used to obtain the subgame equilibria. From the simulations in [178] it is seen that, in general, the game without hierarchy yields a better performance for both the leaders and the followers than the hierarchical case using [467, Algorithm 1]. This advantage stems from the fact that it is quite difficult, as mentioned earlier, to find an algorithm that can compute an optimal solution for the multi-leader multi-follower game (the algorithm based on [467, Algorithm 1] is suboptimal).

9.4 IEEE 802.16 broadband wireless access networks

Broadband wireless access based on the IEEE 802.16 technology [487] is a promising technique for last-mile access. As discussed in Chapter 2, the IEEE 802.16 standard has been proposed to provide high-speed broadband wireless connectivity through a predefined QoS framework for multimedia traffic. Even though the physical-layer specifications and the medium access control protocol signaling are well defined in the standard, the resource-allocation and admission-control policies for the IEEE 802.16 air interface remain open issues and are the subject of much research work [186, 298, 334, 404].

In the context of IEEE 802.16 broadband networks, because of the existence of different classes of services [487], any bandwidth-allocation and admission-control scheme must explicitly consider each user's satisfaction level in terms of delay and throughput. Furthermore, in IEEE 802.16, physical-layer aspects such as adaptive modulation and coding need to be taken into consideration in optimizing system performance, while satisfying the users' QoS requirements. In this respect, a number of approaches for providing bandwidth allocation and admission control within 802.16 have been studied in [361, 179, 362, 364].

In this section, we study in detail how non-cooperative game theory can be used to model the problem of bandwidth allocation and admission control in an 802.16 network, given the different classes of service. After introducing key performance measures in IEEE 802.16 networks, we formulate a non-cooperative game for performing QoS-aware bandwidth allocation and admission control. Beyond resource allocation, this section also presents a network-formation games-based approach for modeling the network structure resulting from the deployment of relay stations in next-generation broadband and cellular networks, notably in the emerging IEEE 802.16 j standard.

9.4.1 Resource allocation and admission control

In this subsection, based on [364], we study a QoS-aware bandwidth-allocation and admission-control method based on queueing analysis and game-theoretic formulation. First, we present a general overview of the performance measures for resource allocation in 802.16, and describe the considered model. Then we proceed to study and formulate the bandwidth-allocation and admission-control game.

Resource allocation and admission control in IEEE 802.16: performance measures In the IEEE 802.16 architecture, as mentioned in Chapter 2, there are two types of stationary stations: subscriber stations (SS) and base stations. The subscriber stations cannot communicate with one another directly, and the base station governs all communications in the network. Here, we consider a single base station serving multiple connections, presumably from different subscriber stations through a TDMA/TDD access mode using single-carrier modulation (e.g., as in WirelessMAN-SC, which operates in the 10-66 GHz band and where the signal propagation between a base station and an SS should be line-of-sight). For each connection, a separate queue with protocol data units (PDUs) of size X is maintained for buffering the PDUs from the corresponding application. We consider SSs of the GPC type; that is, a certain amount of bandwidth is reserved for each connection when bandwidth allocation and connection admission control are performed. Adaptive modulation and coding is used to adjust the transmission rate adaptively in each frame according to the channel quality, as per the IEEE 802.16 standard [487]. Hereinafter, we consider the performance measures for uplink transmission. However, the same model can be used for analyzing the performance of downlink transmission as well.

First, in order to capture the peak arrival rate of traffic, we use a Markov modulated Poisson process (MMPP) to model each source of traffic. MMPP is, in fact, a general traffic-source model that is able to model multimedia traffic as well as Internet traffic (further discussion of the MMPP is found in [364]). With MMPP, the PDU arrival rate λ_s is determined by the phase *s* of a Markov chain having a total number of phases *S* (i.e., s = 1, 2, ..., S). An MMPP can be represented by matrices **M** and **A**, where the former is the transition probability matrix of the modulating Markov chain, and the latter is the matrix corresponding to the Poisson arrival rates. These matrices are defined as follows:

$$\mathbf{M} = \begin{bmatrix} m_{1,1} & \cdots & m_{1,S} \\ \vdots & \vdots & \vdots \\ m_{5,1} & \cdots & m_{5,S} \end{bmatrix}, \quad \mathbf{\Lambda} = \begin{bmatrix} \lambda_1 & & \\ & \ddots & \\ & & \lambda_S \end{bmatrix}.$$
(9.56)

The queueing analysis is performed in discrete time; therefore, we consider discretetime MMPP (dMMPP), which is equivalent to MMPP in continuous time. In this case, the rate matrix Λ is represented by the diagonal probability matrix Λ_a when the number of PDUs arriving in one frame is *a*. Note that $a \in \{0, 1, \dots, A\}$, where *A* is the maximum batch size for PDU arrival. In this case, we can establish the matrices for an MMPP traffic source as follows:

$$\mathbf{\Lambda}_{a} = \mathbf{M} \begin{bmatrix} f_{a}(\lambda_{1}) & & \\ & \ddots & \\ & & f_{a}(\lambda_{S}) \end{bmatrix}, \quad \mathbf{\Lambda}_{A} = \mathbf{M} \begin{bmatrix} F_{A}(\lambda_{1}) & & \\ & \ddots & \\ & & F_{A}(\lambda_{S}) \end{bmatrix}, \quad (9.57)$$

where

$$f_{a}(\lambda_{s}) = \frac{e^{-\lambda_{s}T}(\lambda_{s}T)^{a}}{a!}, \quad F_{a}(\lambda_{s}) = \sum_{j=a}^{\infty} f_{j}(\lambda_{s})$$
(9.58)

denote, respectively, the probability mass function and the complementary cumulative probability mass function corresponding to the occurrence of *a* Poisson events during time interval T (i.e., one frame duration) with mean rate λ_s .

For the channel, we consider a Nakagami-*m* channel model in which the channel quality is determined by the instantaneous SNR γ at the receiver. With adaptive modulation, the SNR at the receiver is divided into N + 1 non-overlapping intervals (N = 7 in the IEEE 802.16 specifications) by thresholds Γ_n $(n \in \{0, 1, ..., N\})$, where $\Gamma_0 < \Gamma_1 < ... < \Gamma_{N+1} = \infty$. The channel is said to be in state *n* (i.e., rate ID *n* will be used) if $\Gamma_n \leq \gamma < \Gamma_{n+1}$. To avoid possible transmission error, no PDU is transmitted when $\gamma < \Gamma_0$. Note that these thresholds correspond to the required SNR specified in the IEEE 802.16 standard [487]. With Nakagami-*m* fading, the probability of using rate ID *n* (i.e., Pr(*n*)) is given by

$$\Pr(n) = \frac{\Gamma(m, m\Gamma_n/\bar{\gamma}) - \Gamma(m, m\Gamma_{n+1}/\bar{\gamma})}{\Gamma(m)},$$
(9.59)

where $\bar{\gamma}$ is the average SNR, *m* is the Nakagami fading parameter ($m \ge 0.5$), $\Gamma(m)$ is the gamma function, and $\Gamma(m, \gamma)$ is the complementary incomplete gamma function.

Here, bandwidth *b* is defined as the number of PDUs that can be transmitted in one frame using *rate* ID = 0. For a given amount of bandwidth and the transmission *rate* ID, the number of transmitted PDUs can be calculated from the number of information bits per symbol. For example, with b = 1, if *rate* ID = 1, two PDUs can be transmitted in one frame. Similarly, with *rate* ID = 6, nine PDUs (i.e., 2×4.5) can be transmitted in one frame. We assume that the channel for one connection remains stationary over a frame interval (≤ 2 ms) and all PDUs corresponding to a connection transmitted during one frame period use the same *rate* ID.

We can define row vector \mathbf{D}_b , whose element D_{k+1} corresponds to the probability of transmitting *k* PDUs in one frame using *b* units of bandwidth, as

$$\mathbf{D}_b = \begin{bmatrix} D_0 \cdots D_k \cdots D_{9 \times b} \end{bmatrix},\tag{9.60}$$

where

$$D_{(I_n \times 2 \times b)} = \Pr(n), \tag{9.61}$$

where I_n is the number of information bits per symbol corresponding to the transmission rate ID n, and $D_0 = 1 - \sum_{k=1}^{9 \times b} D_k$. With b units of bandwidth, the transmission rate for a connection is

$$\tau = \sum_{k=1}^{9 \times b} k \times D_k. \tag{9.62}$$

For each connection, a separate queue with size X PDUs is used for buffering data from the higher-layer application. The state of a queue (i.e., the number of PDUs in the queue and the phase of arrival) is observed at the beginning of each frame. A PDU arriving in

frame f will not be transmitted until the next frame f + 1 at the earliest. The state space of a queue can be defined as

$$\Phi = \{ (\mathcal{X}, \mathcal{M}), | 0 \le \mathcal{X} \le X, \mathcal{M} \in \{1, \dots, S\} \},$$
(9.63)

where \mathscr{X} and \mathscr{M} represent, respectively, the number of PDUs in the queue and the phase of MMPP arrival. The transition matrix **P** for a queue can be expressed as

$$\mathbf{P} = \begin{bmatrix} \mathbf{p}_{0,0} & \cdots & \mathbf{p}_{0,A} & & & \\ \vdots & \ddots & \ddots & \ddots & \\ \mathbf{p}_{U,0} & \cdots & \mathbf{p}_{U,U} & \cdots & \mathbf{p}_{U,U+A} & \\ & \ddots & \ddots & \ddots & \ddots & \ddots \\ & & \mathbf{p}_{x,x-U} & \cdots & \mathbf{p}_{x,x} & \cdots & \mathbf{p}_{x,x+A} \\ \hline & & \ddots & \ddots & \ddots & \ddots \\ & & & \mathbf{p}_{X,X-U} & \cdots & \mathbf{p}_{X,X} \end{bmatrix} .$$
(9.64)

The element $\mathbf{p}_{x,x'}$ denotes the probability matrix for the case in which the number of PDUs in the queue changes from x in the current frame to x' in the next frame. Here, U denotes the maximum number of PDUs that can depart within a frame time, and is given by $I_n \times 2 \times b$.

In the matrix **P**, the first set of rows represents the cases in which the number of possible PDU departures is larger than the queue size and there is *no* PDU dropping effect. The second set of rows represents the cases in which the number of possible PDU departures is larger than the queue size and there exists a PDU dropping effect. Since the size of a queue is finite, some of the arriving PDUs will be dropped, owing to the lack of buffer space. The bottom part, from row X - A + 1 to row X, represents cases in which some of the incoming PDUs are dropped.

Let $\mathbf{D}^{(x)}$ denote the transmission probability when there are x PDUs in the queue:

$$\mathbf{D}^{(x)} = \begin{bmatrix} D_0 \cdots D_{U'} \end{bmatrix},\tag{9.65}$$

where $U' = \min(x, U)$ and

$$D_{U'} = \begin{cases} D_{U,} & \text{if } U' = U, \\ \sum_{k=x}^{U} D_k, & \text{if } U' = x. \end{cases}$$
(9.66)

Note that the maximum total PDU transmission rate can be greater than the number of PDUs in the queue, while the maximum number of transmitted PDUs cannot be larger than the number of PDUs in the queue.

The elements in the first and the second parts of matrix **P** can be obtained as follows:

$$\mathbf{p}_{x,x-u} = \sum_{k-j=u} \mathbf{\Lambda}_j \times \left(\left[\mathbf{D}^{(x)} \right]_{k+1} \mathbf{I}_S \right), \tag{9.67}$$

$$\mathbf{p}_{x,x+\nu} = \sum_{j-k=\nu} \mathbf{\Lambda}_j \times \left(\left[\mathbf{D}^{(x)} \right]_{k+1} \mathbf{I}_S \right), \tag{9.68}$$

$$\mathbf{p}_{x,x} = \sum_{k=j} \mathbf{\Lambda}_j \times \left(\left[\mathbf{D}^{(x)} \right]_{k+1} \mathbf{I}_S \right), \tag{9.69}$$

for u = 1, ..., U' and v = 1, ..., A, where $k \in \{0, 1, 2, ..., U'\}$ and $j \in \{0, 1, 2, ..., A\}$ represent the number of departed PDUs and the number of PDU arrivals, respectively, and I_S is an identity matrix of size $S \times S$. Note that $[\mathbf{D}^{(x)}]_{k+1}$ indicates the element at column k + 1 of row vector $\mathbf{D}^{(x)}$.

Considering both the PDU arrival and PDU departure events, (9.67), (9.68), and (9.69) represent the transition probability matrices for the cases in which the number of PDUs in the queue decreases by u, increases by v, and does not change, respectively.

The bottom part of matrix **P** ({x = X - A + 1, X - A + 2, ..., X}) must capture the PDU dropping effect. Therefore, for $x + v \ge X$, (9.68) becomes

$$\mathbf{p}_{x,x+\nu} = \sum_{i=\nu}^{A} \hat{\mathbf{p}}_{x,x+i}.$$
(9.70)

For x = X, (9.69) becomes

$$\mathbf{p}_{x,x} = \hat{\mathbf{p}}_{x,x} + \sum_{i=1}^{A} \hat{\mathbf{p}}_{x,x+i}, \qquad (9.71)$$

where $\hat{\mathbf{p}}_{x,x'}$ is obtained from (9.67), (9.68), and (9.69) when there is no PDU dropping effect. Equations (9.70) and (9.71) indicate the case in which the queue will be full if the number of incoming PDUs is greater than the available space in the queue.

In consequence, to capture the required QoS measures, the steady-state probabilities for the queue need to be found. In this context, the steady-state probability matrix π is obtained by solving the equations $\pi \mathbf{P} = \pi$ and $\pi \mathbf{1} = 1$, where $\mathbf{1}$ is a column matrix of ones. The matrix π will thus contain the steady-state probabilities for the number of PDUs in the queue and the phases of the MMPP traffic source. This matrix π can be decomposed into $\pi(x, s)$, i.e., the steady-state probability that there are x PDUs in the queue and the phase of the MMPP arrival is s, as follows:

$$\pi(\mathbf{x}, \mathbf{s}) = [\boldsymbol{\pi}]_{S\mathbf{x}+\mathbf{s}}.$$
(9.72)

It follows that the average number of PDUs in a tagged queue is

$$\bar{x} = \sum_{x=1}^{X} x \left(\sum_{s=1}^{S} \pi(x, s) \right).$$
 (9.73)

Moreover, we need to define the PDU dropping probability, the probability that an incoming PDU will be dropped because of the unavailability of buffer space. This can be derived from the average number of dropped PDUs per frame. Given that there are x PDUs in the queue and the number of PDUs in the queue increases by v, the number of dropped PDUs is v - (X - x) for v > X - x, and zero otherwise. The average number of dropped PDUs per frame is

$$\bar{x}_{drop} = \sum_{s=1}^{S} \sum_{x=0}^{X} \sum_{\nu=X-x+1}^{A} \pi(x,s) \left(\sum_{j=1}^{S} [\mathbf{p}_{x,x+\nu}]_{s,j} \right) (\nu - (X-x)), \quad (9.74)$$

where the term $\left(\sum_{j=1}^{S} [\mathbf{p}_{x,x+v}]_{s,j}\right)$ indicates the total probability that the number of PDUs in the queue increases by *v* at every arrival phase. After calculating the average number of dropped PDUs per frame, we can obtain the probability that an incoming PDU is dropped:

$$P_{\rm drop} = \frac{\bar{x}_{\rm drop}}{\bar{\lambda}},\tag{9.75}$$

where $\bar{\lambda}$ is the average number of PDU arrivals per frame,

$$\bar{\lambda} = \sum_{j=1}^{A} \sigma \Lambda_j \mathbf{1}, \qquad (9.76)$$

where σ denotes the steady-state probability of an MMPP source, which can be obtained by solving $\sigma \mathbf{M} = \sigma$ and $\sigma \mathbf{1} = 1$.

Two final important measures that we take into account are the queue throughput and the average delay. The queue throughput, measuring the number of PDUs transmitted during one frame, is

$$\eta = \lambda (1 - P_{\rm drop}). \tag{9.77}$$

Furthermore, the average delay is defined as the number of frames that a PDU waits in the queue from its arrival until it is transmitted to the base station. This average delay is obtained from Little's law [71] as follows:

$$\bar{d} = \frac{\bar{x}}{\eta},\tag{9.78}$$

where η is the effective arrival rate at the queue and \bar{x} is the average number of PDUs in the queue.

Having formulated the key performance measures, notably the delay and queueing aspects of the problem, the next step is to provide a game-theoretic analysis.

A non-cooperative game for bandwidth allocation and admission control

In this subsection, we formulate a non-cooperative two-person non-zero-sum game for bandwidth allocation and admission control in IEEE 802.16. The key elements of this game are the following.

- The two players are the base station (or service provider) and any new connection that receives a service from the base station.
- The strategy of the base station corresponds to allocating a certain bandwidth to the new connection. Note that a strictly positive allocated bandwidth implies that the base station accepts the new connection, while a zero bandwidth allocation implies that the base station rejects the new connection.
- There are two possible strategies for a new connection: to accept or to reject the service offered by the base station.
- The utility functions are dependent on the service type. The IEEE 802.16 standard defines the following four service types, each of which has different QoS requirements:
 - Unsolicited grant service (UGS) supports constant-bit-rate (CBR) traffic, for which a static resource allocation is generally used.
 - Real-time polling service (rtPS) supports real-time traffic in which the delay is an important QoS requirement. The amount of bandwidth required for this type of service is determined based on the required QoS performance (e.g., delay), the channel quality, and the traffic arrival rates of the sources.
 - Non-real-time polling service (nrtPS) requires a QoS guarantee that is not as stringent as in the rtPS case. This type is suitable for applications such as file transfer with guaranteed throughput. The bandwidth allocation is also adaptive, as in the case of rtPS.
 - *Best-effort service (BE)* is used for best-effort traffic with no QoS guarantee. However, user satisfaction depends on perceived performance measures (e.g., transmission rate).

Note that UGS service does not need any adaptive bandwidth allocation.

We note that the considered game is non-cooperative because the decisions are made by the two players (base station and new connection) independently. The base station can decide to admit a new connection if the performance of the ongoing connections does not degrade below a desired level. Similarly, the new connection can accept or deny the services offered by the base station according to whether its delay and/or throughput requirements are met.

For the base station, the procedure for establishing its strategies, i.e., deciding on the amount of bandwidth to allocate to a new connection (including rejecting this connection), is as follows:

1. In order to allocate bandwidth to a new connection, some portion of bandwidth needs to be taken from the set of ongoing rtPS (C_{rt}), nrtPS (C_{nrt}), and BE (C_{be}) connections. The maximum amounts of bandwidth from rtPS (b'_{rt}), nrtPS (b'_{nrt}), and BE (b'_{be})

connections are, respectively,

$$b_{rt}' = \begin{cases} \beta_{1}^{(rt)}, & \sum_{i \in C_{rt}} b(i) < B_{TH}^{(rt)}, \\ \beta_{2}^{(rt)}, & \sum_{i \in C_{rt}} b(i) \ge B_{TH}^{(rt)}, \end{cases}$$

$$b_{nrt}' = \begin{cases} \beta_{1}^{(nrt)}, & \sum_{j \in C_{nrt}} b(j) < B_{TH}^{(nrt)}, \\ \beta_{2}^{(nrt)}, & \sum_{j \in C_{nrt}} b(j) \ge B_{TH}^{(nrt)}, \end{cases}$$

$$b_{be}' = \begin{cases} \beta_{1}^{(be)}, & \sum_{j \in C_{be}} b(j) < B_{TH}^{(be)}, \\ \beta_{2}^{(be)}, & \sum_{j \in C_{be}} b(j) \ge B_{TH}^{(be)}. \end{cases}$$

(9.79)

Here, $\beta_1^{(rt)}$, $\beta_2^{(rt)}$, $\beta_1^{(nrt)}$, $\beta_2^{(nrt)}$, $\beta_1^{(be)}$, $\beta_2^{(be)}$, $B_{TH}^{(rt)}$, $B_{TH}^{(nrt)}$, and $B_{TH}^{(be)}$ are system parameters. In particular, if the amount of bandwidth allocated to rtPS connections is higher than the threshold $B_{TH}^{(rt)}$, the maximum amount of bandwidth taken from C_{rt} is $\beta_2^{(rt)}$, and $\beta_1^{(rt)}$ otherwise. A similar method is used for nrtPS and BE services. Using these thresholds, we can prioritize bandwidth allocation among rtPS, nrtPS, and BE services. For example, if the amount of bandwidth used by BE service is larger than that of rtPS and nrtPS, the portion of the bandwidth for the new connection coming from C_{be} should be larger than the ones coming from C_{rt} and C_{nrt} . Note that $\beta_1^{(rt)}$, $\beta_2^{(nrt)}$, $\beta_2^{(nrt)}$, $\beta_1^{(be)}$, and $\beta_2^{(be)}$ are chosen according to the type of new connection. For example, if the new connection is of type rtPS, a larger amount of bandwidth should come from C_{rt} .

- 2. Subsequently, the set of base station strategies can be defined as $\mathcal{BS} = \{bs_{0,0,0}, bs_{0,0,1}, \dots, bs_{r,n,e}, \dots, bs_{b'_{rt},b'_{nt},b'_{be}}\}$, where *r*, *n*, and *e* units of bandwidth (i.e., $r \in \{0, 1, \dots, b'_{rt}\}$, $n \in \{0, 1, \dots, b'_{nrt}\}$, and $e \in \{0, 1, \dots, b'_{be}\}$) are taken from rtPS, nrtPS, and BE connections, respectively, and assigned to a new connection. The total number of possible strategies is $(b'_{rt} + 1) \times (b'_{nrt} + 1) \times (b'_{be} + 1)$.
- 3. After obtaining the amount of bandwidth, the base station needs to decide from which connections in C_{rt} , C_{nrt} , and C_{be} the bandwidth will be taken. In this case, the base station can iteratively search for the ongoing connections with the highest utility. We denote by \mathbf{B}_{r+n+e} the row vector pertaining to the remaining amount of bandwidth allocated to ongoing connections if the base station allocates r + n + e units of bandwidth to the new connection. Note that \mathbf{B}_0 denotes the initial amount of bandwidth of the ongoing connections before the algorithm starts and $\sum_{\forall i} [\mathbf{B}_k]_i = B$, $(k = \{0, 1, ..., b'_{rt} + b'_{nrt} + b'_{be}\})$, where *B* is the total amount of bandwidth for rtPS, nrtPS, and BE connections.

This matrix \mathbf{B}_{r+n+e} is obtained as follows:

$$i_h = \max(u_i([\mathbf{B}_{r+n+e-1}]_i)) \text{ for } r+n+e>0$$
, (9.80)

$$[\mathbf{B}_{r+n+e}]_{i_h} = [\mathbf{B}_{r+n+e-1}]_{i_h} - 1,$$
(9.81)

where $u_i(\cdot)$ is the utility function of connection *i*, and i_h is the index of the connection with the highest utility in \mathbf{B}_{r+n-1} .

For each new connection, the set of strategies can be defined as $\mathcal{NC} = \{nc_1, nc_2\}$, where nc_1 and nc_2 denote the strategies in which the new connection accepts or denies the service, respectively. Having defined the strategy spaces for the two players, i.e., the base station and the new connection, we can now zoom in on the utility functions.

For an rtPS connection *i*, the utility depends on average delay d(b(i)). As discussed in [364], a suitable utility for this type of service would be based on the modified sigmoid function, which, given a delay requirement d_{tar} , can be expressed as

$$u_i(b(i)) = 1 - \frac{1}{1 + \exp\left(-g_{rt} \times (\bar{d}(b(i)) - d_{tar} - h_{rt})\right)},$$
(9.82)

where g_{rt} and h_{rt} are two parameters of the modified sigmoid function. While g_{rt} indicates the steepness (i.e., sensitivity to delay), h_{rt} represents the center of the curve (i.e., satisfaction with the performance as perceived by the users). This utility function is well suited for rtPS services since its value ranges between zero and one, and it can represent the case in which the connection is either fully satisfied with the QoS performance or totally unsatisfied. We also note that the degree of satisfaction can be adjusted through the function's slope.

For an nrtPS connection *i*, the utility depends on the transmission rate $\tau(b(i))$, which is a function of allocated bandwidth. Given a throughput requirement τ_{tar} , the utility for an nrtPS connection can be expressed as

$$u_i(b(i)) = \frac{1}{1 + \exp(-g_{\rm nrt} \times (\tau(b(i)) - \tau_{\rm tar} - h_{\rm nrt}))},$$
(9.83)

where g_{nrt} and h_{nrt} are the parameters of the sigmoid function. We note that the sigmoid used for the nrtPS implies that whenever the perceived transmission rate is lower than a particular level, the nrtPS user is unsatisfied with the service.

For BE connections, no strict QoS guarantees are required. Therefore, as in [364], we consider the following logarithmic utility function in which the utility is an increasing function of the transmission rate $\tau(b(i))$:

$$u_i(b(i)) = g_{be} \log(1 + h_{be} \tau(b(i))), \qquad (9.84)$$

where g_{be} and h_{be} are the parameters of the utility function. Note that with this logarithmic utility function, the rate of increase in utility decreases as the transmission rate increases.

The utilities defined so far pertain to each individual connection. From the base station's perspective, the utility can represent the total revenue from the ongoing connections, where the revenue for each connection depends on the QoS performance.

Since we have a two-player non-cooperative game, as studied in Chapter 3, it is useful to develop the matrix representations of the payoffs. For convenience, and because of the large number of strategies for the base station, instead of representing the payoffs to both players for each combination of strategies in a single matrix, we use two matrices for each player: a matrix that shows the possible payoffs for each strategy of the base station when the new connection uses strategy nc_1 (i.e., accepts a connection), and a

matrix that shows the possible payoffs for each strategy of the base station when the new connection uses strategy nc_2 (i.e., rejects a connection). This is somewhat similar to the popular *bimatrix* representation [377] of a two-player game, with the difference being that in the bimatrix case a single matrix per player is defined. For the base station, the payoff matrices Ψ_{bs} are defined as follows:

$\rho_{bs}(bs_{0,0,0}, nc_1)$	$\rho_{bs}(bs_{0,0,1}, nc_1)$		$\rho_{bs}(bs_{0,b'_{ntt},b'_{be}},nc_1)$	
$\rho_{bs}(bs_{1,0,0}, nc_1)$	$\rho_{bs}(bs_{1,0,1}, nc_1)$		$\rho_{bs}(bs_{1,b'_{\mathrm{nrt}},b'_{\mathrm{be}}},nc_1)$	
:	:		:	,
$\boxed{\rho_{bs}(bs_{b'_{\rm rt},0,0},nc_1)}$	$\rho_{bs}(bs_{b'_{\mathrm{rt}},0,1},nc_1)$		$\rho_{bs}(bs_{b'_{\mathrm{rt}},b'_{\mathrm{nrt}},b'_{\mathrm{be}}},nc_{1})$	
F		ı		
$\rho_{bs}(bs_{0,0,0}, nc_2)$	$ ho_{bs}(bs_{0,0,1}, nc_2)$	•••	$\rho_{bs}(bs_{0,b'_{nrt},b'_{be}},nc_2)$	
$\rho_{bs}(bs_{1,0,0}, nc_2)$	$\rho_{bs}(bs_{1,0,1}, nc_2)$		$ ho_{bs}(bs_{1,b_{\mathrm{nrt}}^{\prime},b_{\mathrm{be}}^{\prime}},nc_{2})$	(0.05)
:	:		÷	. (9.85)
$\rho_{bs}(bs_{b'_{\mathrm{rt}},0,0},nc_2)$	$\rho_{bs}(bs_{b'_{t},0,1},nc_2)$		$\rho_{bs}(bs_{b'_{\mathrm{rt}},b'_{\mathrm{nrt}},b'_{\mathrm{be}}},nc_2)$	

Column *j* of these matrices corresponds to strategy $bs_{r,n,e}$, where $n = \lfloor \frac{j-1}{b'_{nrt}+1} \rfloor$ and $e = j - \lfloor \frac{j-1}{b'_{nrt}+1} \rfloor (b'_{nrt}+1) - 1$. The elements of this payoff matrix are

$$\rho_{bs}(bs_{r,n,e}, nc_1) = \sum_{\forall i} u_i([\mathbf{B}_{r+n+e}]_i) + u_{c_{(n)}}(r+n+e), \qquad (9.86)$$

$$\rho_{bs}(bs_{r,n,e}, nc_2) = \rho_{bs}(bs_{0,0,0}, nc_2), \qquad (9.87)$$

where subscript $c_{(n)}$ denotes a new connection. Equation (9.86) is obtained based on the fact that if the new connection is accepted and is allocated r + n + e units of bandwidth, the payoff for the base station becomes the total utility for the ongoing connections after some portion of bandwidth has been taken away (which results in a decrease in the total utility for the ongoing connections) plus the utility for the new connection. However, if the new connection chooses to deny the offered service, the payoff for the base station is the same as that when the base station rejects the new connection.

The payoff matrices Ψ_{nc} for a new connection are defined as follows:

$\rho_{nc}(bs_{0,0,0}, nc_1)$	$\rho_{nc}(bs_{0,0,1}, nc_1)$		$\rho_{nc}(bs_{0,b'_{nt},b'_{be}},nc_1)$	
$\rho_{nc}(bs_{1,0,0}, nc_1)$	$\rho_{nc}(bs_{1,0,1}, nc_1)$		$\rho_{nc}(bs_{1,b'_{nrt},b'_{be}},nc_1)$	
÷	÷		:	,
$\overline{\rho_{nc}(bs_{b_{rt}^{\prime},0,0},nc_{1})}$	$\rho_{nc}(bs_{b'_{\mathrm{rt}},0,1},nc_1)$		$\rho_{nc}(bs_{b'_{rt},b'_{nrt},b'_{be}},nc_1)$	
$\rho_{nc}(bs_{0,0,0}, nc_2)$	$\rho_{nc}(bs_{0,0,1}, nc_2)$		$\rho_{nc}(bs_{0,b'_{nrt},b'_{be}},nc_2)$	
$\rho_{nc}(bs_{1,0,0}, nc_2)$	$\rho_{nc}(bs_{1,0,1}, nc_2)$		$\rho_{nc}(bs_{1,b'_{nrt},b'_{be}},nc_2)$	(0.00)
:	:		:	(9.88)
$\boxed{\rho_{nc}(bs_{b'_{rt},0,0},nc_2)}$	$\rho_{nc}(bs_{b'_{rt},0,1},nc_2)$		$\rho_{nc}(bs_{b'_{\mathrm{rt}},b'_{\mathrm{nrt}},b'_{\mathrm{be}}},nc_2)$	

where

$$\rho_{nc}(bs_{r,n,e}, nc_1) = u_{c_{(n)}}(r+n+e), \qquad (9.89)$$

$$\rho_{nc}(bs_{r,n,e}, nc_2) = 1 - u_{c_{(n)}}(r+n+e).$$
(9.90)

Equation (9.89) assigns the payoff, as per the previously defined utility functions, for a new connection when it accepts the service offered by the base station with r + n + e units of bandwidth. In case the connection rejects this service because of poor QoS performance, the payoff is presented in (9.90). This is selected in such a way that, if r + n + e is large enough to satisfy the target QoS performance, the payoff for denying the service is low (i.e., it is better for the new connection to accept the offered service).

Given this formulation of a non-cooperative game, in the next subsection we investigate the possible solution and discuss the algorithm details as well as some results on performance.

Nash equilibrium of the bandwidth-allocation and admission-control game

For the studied non-cooperative game, the Nash equilibrium represents a suitable solution concept. As discussed in [364] for this game, it is quite complex to analytically study the existence and uniqueness of the Nash equilibrium. However, to determine the Nash equilibrium in simulations similar to those in [364], one can utilize the best-response functions of the players. For the base station, the best-response function $BR_{bs}(nc'_i)$, given that the new connection chooses strategy nc'_i (which is either to accept or to reject the connection), is

$$BR_{bs}(nc'_{i}) = \max_{bs_{r,n,e}} (\rho_{bs}(bs_{r,n,e}, nc'_{i})).$$
(9.91)

For a new connection, we can define the best-response function $BR_{nc}(bs'_{r,n,e})$, given that the base station chooses strategy $bs'_{r,n,e}$, as follows:

$$BR_{nc}(bs'_{r,n,e}) = \max_{nc_i} \left(\rho_{nc}(bs'_{r,n,e}, nc_i) \right).$$
(9.92)

Recall from Chapter 3 that the pair of strategies $(bs_{r,n,e}^*, nc_i^*)$ is a Nash equilibrium if and only if $bs_{r,n,e}^* = BR_{bs}(nc_i^*)$ and $nc_i^* = BR_{nc}(bs_{r,n}^*)$. Certainly, for this strategy pair the following holds:

$$\rho_{bs}(bs^*_{r,n,e}, nc^*_i) \ge \rho_{bs}(bs'_{r,n,e}, nc^*_i), \tag{9.93}$$

$$\rho_{nc}(bs_{r,n,e}^{*}, nc_{i}^{*}) \ge \rho_{bs}(bs_{r,n,e}^{*}, nc_{i}^{\prime}), \qquad (9.94)$$

for any other strategies $bs'_{r,n,e}$ and nc'_i .

For admission control, a new connection is accepted by the base station if there exists a Nash equilibrium that maximizes the utility for the base station, and is rejected otherwise. To reach a Nash equilibrium, when it exists, one can use the best-response strategies of the players, as discussed in [364] using simulations. For bandwidth allocation, if the new connection is accepted, the base station takes r, n, and e units of bandwidth from the

groups of rtPS, nrtPS, and BE connections, respectively, and assigns r + n + e units of bandwidth to the new connection.

Thus, the operation of the network will be as follows. When a new connection is initiated, the base station invokes the bandwidth-allocation and admission-control algorithm. In this case, the new connection informs the base station of the connection's type (i.e., rtPS, nrtPS, or BE), traffic source parameters, and its QoS (i.e., delay or throughput) requirement. Then, the base station establishes a set of strategies and computes the expected payoff corresponding to each strategy. Next, the game is solved to obtain the Nash equilibrium, which is then used to make a decision on whether the connection will be rejected or accepted as well as on the amount of bandwidth assigned to a new connection (if accepted). This bandwidth-allocation and admission-control algorithm is shown in Fig. 9.6, as per [364].

In [364], extensive simulations were run to assess the performance of this algorithm. Figures 9.7(a) and 9.7(b) show the payoffs to both the base station and a new connection



Fig. 9.6 Bandwidth-allocation and admission-control procedure.



Fig. 9.7 Payoff for (a) the base station, and (b) a new connection, for the case when a new connection is accepted.

for the case in which the base station accepts the new connection (with $b'_{rt} = b'_{nrt} = b'_{be} =$ 10). The results in these figures are obtained using the best-response functions and the matrices Ψ_{bs} in (9.85) and Ψ_{nc} in (9.88), respectively. This case corresponds to a scenario in which, from the perspective of the base station, accepting a new connection yields a higher payoff than rejecting it. Similar results can be seen in [364] for the case where there is no Nash equilibrium, in which the payoff in accepting the new connection is higher than that resulting from rejecting the new connection (i.e., it is better for the base station to reject the connection). Finally, the reader is referred to [364] for additional simulation results and insights on the properties and performance advantages of the game-theoretic approach.

9.4.2 Relay-station deployment in IEEE 802.16j

In the most recent WiMAX standard, the IEEE 802.16j, a new node, the relay station (RS), has been introduced to improve the network's capacity and coverage [11]. One important motivation for deploying RSs is that it enables the use of advanced communication techniques such as cooperative communication (i.e., cooperative relaying), which can significantly improve the performance for the wireless users. This has also encouraged the incorporation of relay-station nodes in other important next-generation wireless networks (beyond 802.16j), such as LTE-Advanced [7].

For an efficient deployment of RSs in next-generation networks, several key technical challenges need to be addressed at both the uplink and downlink levels. For the downlink of 802.16j networks, in [296] the authors study the optimal placement of one RS that maximizes the total rate of transmission. This optimal RS placement is further studied in [297] for multiple RSs, aiming to maximize throughput using the concept of dual relaying. Moreover, the work in [524] provides an algorithm for finding the optimal locations of the BS and the RSs, thus minimizing the cost for deployment of a full-scale IEEE 802.16j network. In [429], the authors study the capacity gains and resource utilization in a multi-hop LTE network in the presence of RSs. Furthermore, the performance of different relaying strategies in an LTE-Advanced network is studied in [389]. Resource-allocation and network-planning techniques for 802.16j networks in the presence of RSs are proposed in [369]. Other aspects of RS deployment in next-generation networks are considered in [490, 506, 474, 285, 410, 414].

Most of these contributions focus on the performance-assessment and operational aspects of RS deployment in next-generation multi-hop networks such as 802.16j and LTE-Advanced. Beyond these aspects, one challenging area that is of central interest in the design of next-generation broadband wireless networks is the distributed formation of the network architecture that will connect the base stations, relay stations, and mobiles in the network. In particular, we note that the introduction of the RS strongly impacts the network architecture of next-generation networks such as IEEE 802.16j and LTE-Advanced, which will be governed by a *tree architecture* connecting the base station to its subordinate RSs. Efficient design of the tree topology in 802.16j and similar networks is a challenging problem, notably because the RSs can be nomadic or mobile. Existing WiMAX and LTE standards do not provide any algorithm for the tree formation.

although they state that both distributed and centralized approaches may be used. One contribution to tackling this problem in 802.16j networks has been made in [285] using a centralized approach. However, this work neither provides a clear algorithm for the tree formation, nor considers cooperative transmission or multi-hop delay. In addition, a centralized approach can yield some significant overhead and complexity, namely in networks with a rapidly changing environment caused by RS mobility or incoming traffic load. Distributed approaches for the tree formation, using game theory, are derived in [410, 414].

In this section, based on the work in [414], we tackle the problem of forming the tree architecture in 802.16j-like multi-hop networks, in a distributed manner. First, we describe the studied model and formulate it as a network-formation game. Then, we develop an algorithm for network formation, and study its performance in a simulated environment.

Network-formation game for uplink tree formation

Consider the uplink of an IEEE 802.16j (or LTE-Advanced) network with *M* RSs (fixed, mobile, or nomadic) and one base station. The RSs transmit their data in the uplink to a central base station through multi-hop links, so a tree architecture needs to form, in the uplink, between the RSs and their serving base station. In an 802.16j network, this tree architecture is imposed in the standard [11]. Once the uplink network structure is formed, mobile stations (MSs) can connect to the network by selecting a serving RS or by directly connecting to the base station. In this context, we consider that the MSs deposit their data packets to the serving RSs using direct transmission. Subsequently, the RSs in the network that received the data from the external MSs can act as source nodes, transmitting the received MS packets to the base station through one or more hops in the formed tree, using *cooperative transmission*. The considered direct transmission between an MS and its serving RS enables us to consider a tree-formation algorithm that can be easily incorporated into new or existing wireless networks without the need for coordination with external entities such as the MSs.

To perform cooperative transmission between the RSs and the base station, we consider a decoded relaying multi-hop diversity channel, i.e., decode-and-forward relaying as required by the IEEE 802.16j standard (see [414] for further information on the multi-hop diversity channel that is used in the model). In this relaying scheme, each intermediate node on the path between a transmitting RS and the BS combines, encodes, and reencodes the received signal from all preceding terminals before relaying (decode-andforward). Formally, every MS *k* in the network constitutes a source of data traffic that follows a Poisson distribution with an average arrival rate λ_k . With such Poisson streams at the entry points of the network (the MSs), for every RS, the incoming packets are stored and transmitted in a first-in first-out (FIFO) fashion, and we consider that we have the Kleinrock independence approximation [71, Chapter 3], with each RS being an M/D/1 queueing system.⁴

⁴ Any other queueing model (e.g., M/M/1) can also be accommodated.



Fig. 9.8 Example of a tree topology formed using a distributed network-formation game.

With this approximation, the total traffic that an RS *i* receives from the MS that it is serving is a Poisson process with an average arrival rate of $T_i = \sum_{l=1}^{L_i} \lambda_l$, where L_i is the number of MSs served by RS *i*. Moreover, RS *i* also receives packets from RSs that are connected to it, with a total average rate R_i . For these R_i packets, the sole role of RS *i* is to relay them to the next hop. In addition, any RS *i* that has no assigned MSs ($L_i = 0$ and $T_i = 0$), transmits HELLO packets, generated with a Poisson arrival rate of η_0 , in order to maintain its link to the BS as active during periods of no actual MS traffic. An illustrative example of this model is shown in Fig. 9.8.

For modeling the interactions among the RSs seeking to form the uplink tree structure, as in Fig. 9.8, network-formation games, which are a branch of coalitional graph games, provide a suitable framework, as discussed in Chapter 7. For the considered tree-formation model, we formulate a network-formation game, with the RSs being the players. The result of the interactions among the RSs is a *directed* graph $G(\mathcal{V}, \mathcal{E})$ with $\mathcal{V} = \{1, ..., M + 1\}$ denoting the set of all vertices (*M* RSs and the base station) and \mathcal{E} denoting the set of all edges (links) between pairs of RSs. Each link between RSs *i* and *j*, denoted $(i, j) \in \mathcal{E}$, corresponds to an uplink traffic flow from RS *i* to RS *j*.

For the considered model, it is useful to define the notion of a path as follows:

DEFINITION 9.2 A path between two nodes *i* and *j* in the graph *G* is defined as a sequence of nodes $i_1, ..., i_K$ such that $i_1 = i$, $i_K = j$, and each directed link $(i_k, i_{k+1}) \in G$

for each $k \in \{1, ..., K - 1\}$. We denote the set Q_i as the set of all paths from node *i* to the BS, and thus $|Q_i|$ represents the number of paths from node *i* to the BS.

As the 802.16j standard imposes a *tree structure* between the RSs and the base station, we adopt the following convention throughout the remainder of this book:

CONVENTION 9.1 Each RS *i* is connected to the BS through at most one path, and thus $|Q_i| \in \{0, 1\}$, $\forall i \in V$. Hence, we denote by $q_i \in Q_i$ the path between any RS *i* and the BS.

This convention is also reasonable in the context of other next-generation networks such as LTE-Advanced, because it will be common that, in the uplink, each RS communicates with its serving base station over a single path.

For the RSs' network-formation games, we delineate the possible actions or strategies that each RS can take in the network-formation game. The strategy space of each RS *i* consists of the RSs (or the base station) that *i* wants to connect to. Consequently, the strategy of an RS *i* is to select the link that it wants to form from the available strategy space. We note that an RS *i* cannot connect to an RS *j* that is already connected to *i*, in the sense that if $(j, i) \in G$, then $(i, j) \notin G$.

Formally, for a current network graph *G*, let $A_i = \{j \in V \setminus \{i\} \mid (j, i) \in G\}$ be the set of RSs from which RS *i* has already accepted a link (j, i), and $S_i = \{(i, j) \mid j \in V \setminus \{i\} \cup A_i\}$ be the set of links corresponding to the nodes (RSs or the base station) with which RS *i* wants to connect (note that RS *i* cannot connect to RSs that are already connected to it, i.e., RSs in A_i). In consequence, the strategy of an RS *i* is to select the link $s_i \in S_i$ that it wants to form, i.e., choose the RS that it will connect to. Based on Convention 9.1, an RS can be connected to at most *one* other node in our game, so selecting to form a link s_i implies that RS *i* will *replace* its previously connected link (if any) with the new link s_i .

Having clearly defined the players and the strategies of the RS network-formation game, the next step is to introduce a suitable utility that can capture the objectives of the RSs. In the considered model, each RS aims to optimize the tradeoff between the achieved packet success rate from transmitting data to the base station and the delay incurred by multi-hop transmission. Although longer hops can, in most cases, yield an improved packet success rate because of bit error rates from cooperative transmission, this gain comes at the cost of an increased delay from using more hops (using more hops can also increase the likelihood of passing through a congested RS). Hereinafter, we consider that the RSs are utilizing voice over IP (VoIP) services, in which optimizing the tradeoff between packet success rate and delay is significantly important.⁵ For VoIP services, given the delay and the packet success rate (which is a function of the bit error rate), an appropriate utility function can be defined using the concept of the R-factor, which links the delay and the packet loss to voice quality, as follows [483]:

$$u_{i}(G) = \Omega_{a} - \epsilon_{1}\tau_{i,q_{i}}(G) - \epsilon_{2}(\tau_{i,q_{i}}(G) - \alpha_{3})H - \upsilon_{1}$$

- $\upsilon_{2}\ln(1 + 100\upsilon_{3}(1 - \rho_{i,q_{i}}(G))),$ (9.95)

⁵ The analysis done in this subsection can readily be extended to other types of services by using a different utility (e.g., see [410]).

where τ_{i,q_i} is the multi-hop delay (given by the Kleinrock approximation; see [410]) expressed in milliseconds; $100(1 - \rho_{i,q_i})$ represents the packet loss percentage, with ρ_{i,q_i} being the packet success rate achieved by RS *i* over path q_i , which is a function of the packet size and the bit error rate achieved when using the previously described decode-and-forward cooperative-transmission scheme (we consider only the packet loss due to errors, ignoring packet loss due to overloaded links). The remaining parameters are constants: $\Omega_a = 94.2$, $\epsilon_1 = 0.024$, $\epsilon_2 = 0.11$, $\epsilon_3 = 177.3$, H = 0 if $\tau_{i,q_i} < \epsilon_3$, and H = 1 otherwise. The parameters v_1 , v_2 , and v_3 depend on the voice speech codec. The relationship between the R-factor and VoIP service quality is such that as the R-factor increases, voice quality improves. For different voice codecs, different R-factor ranges provide an indication of voice quality varying from poor, low, medium, high, to best as the R-factor increases.

Thus, in the formulated RS network-formation game, each RS attempts to find the strategy (next hop) that maximizes its R-factor. Each strategy profile adopted by the relay stations maps into a formed network graph G, which yields a different utility for each RS, as per (9.95).

Now that we have formulated the network-formation game by clearly defining the players, their strategies, and their utilities, the next step is to develop a distributed algorithm for forming the network.

Network-formation algorithm

Before delving into the details of the algorithm for the considered game, we highlight the following property, based on [414]:

PROPOSITION 9.2 Any network graph *G* resulting from a dynamics applied to the studied network-formation game is a connected, directed tree structure rooted at the BS.

This property can be seen by inspecting the utility in (9.95), where we notice that if an RS *i* is not connected to the base station through direct transmission or other multihop paths ($Q_i = \emptyset$), we have $\rho_{i,q_i} = 0$, as all packets are lost (no packet reaches the base station). As a result, the last term in (9.95) is maximized, and the whole utility is minimized (the delay is also infinite if $Q_i = \emptyset$). Hence, a disconnection by any RS drastically decreases its utility and there is no incentive for any RS to disconnect from the base station. Consequently, any graph *G* formed in the considered RS network-formation game is a connected graph, i.e., a tree rooted at the base station.

Because of the high disconnection cost, if an RS is unable to find any partner suitable for forming a link, it will connect to the base station by direct transmission. Thus, our network initially starts with all the RSs connected to the base station (star topology), before engaging in the network-formation game. Denote by $G_{s_i,s_{-i}}$ the graph formed when RS *i* plays a strategy $s_i \in S_i$, while all other RSs maintain their vector of strategies $\mathbf{s}_{-i} = [s_1, \dots, s_{i-1}, s_{i+1}, \dots, s_M]$. We define the *best response* for an RS as follows:

DEFINITION 9.3 A strategy $s_i^* \in S_i$ is a best response for an RS $i \in \mathcal{V}$ if

$$u_i(G_{\mathbf{s}_i^*,\mathbf{s}_{-i}}) \ge u_i(G_{\mathbf{s}_i,\mathbf{s}_{-i}}), \ \forall \mathbf{s}_i \in S_i.$$

$$(9.96)$$

Thus, the best response for RS *i* is to select the link that maximizes its utility, given that the other RSs maintain their vector of strategies. Notice that this is similar to the standard definition of a best response in non-cooperative games (see Chapter 3), but in network-formation games a key feature is that the strategy profile maps to a network graph.

By using the different properties of the RS network-formation game, we can construct, as in [414], a distributed network-formation algorithm using the best responses of the RSs. In this algorithm, the RSs are assumed to be myopic, in the sense that they aim at improving their payoffs considering only the current state (graph), without taking into account the future evolution of the network. The network-formation algorithm under consideration is composed of several rounds with each round consisting mainly of two phases: a fair-prioritization phase and a tree-formation phase. In the fair-prioritization phase, a priority function is used to assign a priority to each RS. In the tree-formation phase, an iterative procedure is developed in which the RSs act by increasing priorities.

Therefore, each round of the considered algorithm begins with the fair-prioritization phase, in which each RS is assigned a priority depending on its actual perceived bit error rate: RSs with higher bit error rates are assigned higher priorities. The motivation behind this procedure is to fairly give RSs that are experiencing a bad channel an advantage in selecting their partners, for the purpose of improving their performance. Thus, the RSs experiencing bad channel conditions (high bit error rates) can select their partners out of a larger space of strategies during the tree-formation phase (recall that an RS cannot choose to connect to an RS that is already connected to it). Other priority functions can also be used, and in general, a random priority function can be defined.

Following prioritization, the RSs start selecting their strategies sequentially in order of priority. During its turn, each RS *i* chooses to play its best response $s_i^* \in S_i$ in order to maximize its utility at each round given the current network graph resulting from the strategies of the other RSs. The best response of each RS can be seen as a *replace* operation, whereby the RS will replace its current link to the BS with another link that maximizes its utility (if such a link is available).

Multiple rounds consisting of the above two phases will be run until convergence to the final tree structure G^{\dagger} , where the RSs can no longer improve their utilities using best responses. The studied algorithm is summarized as Algorithm 9.4.

The stability of the final graph G^{\dagger} is given using the concept of a Nash network, defined as follows:

DEFINITION 9.4 A Nash network is a network graph G in which no node i can improve its utility by a unilateral change in its strategy $s_i \in S_i$.

A Nash network, in the context of RS network formation, is a network in which no RS has an incentive to unilaterally decide to replace its current link with another link, given that the other RSs do not change their connections. In a Nash network, each link selected by an RS $i \in \mathcal{V}$ is a best response to the graph formed by the other RSs, i.e., the graph with vertices $\mathcal{V} \setminus \{i\}$. Algorithm 9.4 is based on best-response interactions, so when it converges it will reach a Nash network as per [414, Lemma 1].

Although, in this setting, an analytical proof for the convergence of the algorithm is complicated because of the characteristics of the game (R-factor-based utility function

Algorithm 9.4	Network-form	ation algorithm.
---------------	--------------	------------------

Initial State

All the RSs are connected in a start network rooted at the base station.

Two Phases in each round of network formation:

Phase 1, fair prioritization:

Prioritize the RSs from the highest to the lowest current bit error rate.

Phase 2, myopic tree formation:

The RSs take action sequentially by priority:

- a) Each RS *i* plays its best response s_i^* , maximizing its utility (R-factor).
- b) The best response s_i^* of each RS is a *replace link* operation through which an RS *i* splits from its current parent RS and replaces it with a new RS that maximizes its utility.

Multiple rounds are run until convergence to the final Nash tree G^{\dagger} , where no RS can improve its utility by a unilateral change of strategy.

and discrete strategies), using simulations it is shown in [414] that the algorithm does indeed converge to a Nash network for the considered simulation setting. Note that, in cases where the algorithm does not converge (e.g., when no Nash network exists), one can impose a restriction on the RSs, such as forbidding them to visit certain graphs or excluding some deviations, in order to guarantee convergence. Alternatively, in the non-convergent case, the RSs can be allowed to utilize multiple graphs, i.e., to cycle periodically between a fixed number of tree structures.

Figure 9.9 shows the tree that forms among M = 10 RSs deployed within the coverage area of a single base station. The network-formation game starts with the star topology in which all RSs are connected directly to the base station. Prior to the presence of the MSs in the network (only HELLO packets present), the RSs interact and converge to a final Nash tree structure, shown by the solid lines in the figure. This figure clearly shows how the RSs connect to their nearby partners, forming the tree structure. Upon deployment of 50 MSs in the network, the RSs adapt to this incoming traffic by forming a new tree structure shown by the dashed lines in Fig. 9.9. To adapt to the incoming traffic, each RS can, in a distributed manner, take a decision to change its current connection in the network. As an example, when the MSs are deployed, RS 3 can no longer accommodate the traffic generated by RS 2 as this drastically decreases its utility. As a result, RS 2 takes the decision to disconnect from RS 3 and improve its R-factor by connecting directly to the base station. Similarly, RS 7 disconnects from RS 10 and connects directly to the less-loaded RS 6.

In Fig. 9.10, we show the average achieved utility per MS as *M*, the number of RSs in the network, increases. The results are shown for the studied network-formation algorithm, for a star topology in which each RS is directly connected to the base station and for the scenario in which no RSs exist in the network. In this figure, we clearly see that as the number of RSs in the network increases, the performance of the network-formation case as well as for the star topology improves. However, for the star topology the rate of improvement is much slower than for network formation. In addition, network



Fig. 9.9 Snapshot of a tree topology formed using network-formation algorithm with 10 RSs before (solid line) and after (dashed line) random deployment of 50 MSs (MS positions not shown for clarity).



Fig. 9.10 Performance assessment of distributed tree-formation algorithm for a network with 40 MSs, shown through average achieved MS utility vs. number of RSs *M* in network (average over random positions of MSs and RSs).
formation presents a clear performance advantage, increasing with the number of the RSs, and reaching up to 40.3 percent and 42.8 percent (at M = 25 RSs) relative to the star topology and the no-RSs case, respectively. Also, this figure highlights the fact that, owing to the delay cost for transmission over multi-hop networks, a significant number of RSs must be deployed within the coverage area of a single base station in order to benefit from performance gains.

In summary, in this subsection we have showed how, based on [414], networkformation games can be used to construct the tree topology that will govern the architecture of next-generation networks such as IEEE 802.16j and LTE-Advanced, in the presence of RSs. Future extensions of this model can tackle various aspects of network topology formation in next-generation networks such as devising a probabilistic approach to network formation, expanding the analytical results on convergence by choosing alternate stability or algorithmic concepts (e.g., pairwise stability or interactions that are not based on best responses), or imposing a structure on the current utility function. Further extensions can consider coalitional graph game concepts such as the balanced core introduced in [205] to characterize the architecture of next-generation wireless networks.

9.5 Network selection in multi-technology wireless networks

In recent years, the wireless market has been served by a variety of technologies (GSM, UMTS, HSDPA, WiMAX, LTE, WiFi). Because each technology possesses advantages and drawbacks, the deployment of multiple technologies in adjacent locations is increasing. Moreover, with the emergence of the norm IEEE 802.21 [9, 472], radio access equipment is becoming multi-standard, offering the possibility of connecting, concurrently, using a number of technologies. Hence, heterogeneity in terms of network types and capabilities is a key characteristic of emerging wireless networks. An example of a heterogeneous wireless network with three different access technologies is shown in Fig. 9.11. It is expected that future wireless devices will be able to switch between different networks with different technologies, this capability is often referred to as *vertical handover*.

Vertical handover reflects an important paradigm shift towards next-generation wireless (e.g., 4G) networks where seamless mobility across heterogenous technologies and services becomes a possibility [140, 468, 155, 246]. Because of this heterogeneity, dynamic network selection (i.e., vertical handover) is required not only to achieve seamless mobility, but also to support quality-of-service improvement and load balancing. For example, one key characteristic of 4G networks is the possibility of vertical handover between wireless LAN and cellular communication (e.g., UMTS or CDMA2000). In such a case, a suitably equipped wireless device would be able to exploit the benefits both of wireless LANs (high throughput) and of cellular technology (ubiquitous coverage) for Internet access, for example. IEEE 802.21 addresses the technical aspects of handover between cellular, GSM, GPRS, 802.16, 802.11, Bluetooth, and WiFi [9, 472].

Vertical handover in heterogeneous wireless networks can be grouped into two categories: network-driven selection and user-driven selection. In a network-driven



Fig. 9.11 Example of a heterogeneous wireless network with three different technologies.

approach, the selection decision is made on the network side, e.g., by the service provider or network operator. Such a scheme is suitable for tightly integrated environments in which a central controller distributes traffic flows among different networks. In a userdriven approach, the users can, individually and in a distributed fashion, take choose the network that is best suited to their objectives. As a result, a user-driven verticalhandover scheme does not require any centralized coordination or the integration of additional inter-networking protocols among the different wireless technologies.

The interest in vertical handover spawned a variety of research activities tackling its technical challenges, notably for the user-driven scheme [529, 526, 109, 120, 531, 68, 63, 365, 114]. In [529], vertical handover between wireless WAN and wireless LAN is implemented and tested using a connection manager that can detect condition changes. Guidelines for efficient optimization of the performance of vertical handover between cellular networks and wireless LANs are presented in [526]. An opportunistic vertical-handover scheme that reduces packet in loss time VoIP services is studied in [109] using a two-state Markov model that exploits the on-off characteristics of voice traffic. Providing vertical handover between IEEE 802.11 and IEEE 802.16 is studied in [120, 531] on the basis of a variety of quality-of-service metrics. Additional approaches for network association and vertical handover are found in [68, 63, 365, 114].

Vertical handover is inherently a decentralized operation, as each user must take a decision on which network type to use, depending on its QoS needs and service requirements. This has encouraged the development of approaches based on game theory to tackle the network-selection problem in heterogeneous, next-generation wireless networks. We study this problem using the approach in [131].

We start by formulating a non-cooperative game for network association in wireless networks. Then, we study the solution of the game and develop an algorithm for finding this solution.

9.5.1 Network selection as a non-cooperative game

In this subsection, we tackle the problem of network selection (vertical handover) in heterogeneous wireless networks using a non-cooperative-game model with complete information. First, we formulate the problem of vertical handover in a multi-technology network as a non-cooperative game. We then discuss the possible solutions of this game. Finally, we study a non-cooperative-game model where network selection and resource allocation are jointly considered.

Network selection as a non-cooperative game with complete information

Consider a wireless network in which two access points, each belonging to a different standard (e.g., HSDPA, WiMAX, WiFi, cellular, etc.), coexist. In this network, we consider N users that are able to monitor both standards and to select the best standard to connect to, depending on their quality-of-service requirements. We also assume that there are MAC schedulers at every access point (or base station), which, over an intermediate time scale, allow each user to use a channel and transmit according to the number of users connected to a given base station, and to its own channel conditions. For any user *i* connected to an access point or base station *r*, the throughput is given by

$$T_{ir} = \frac{c_{ir}}{N_r},\tag{9.97}$$

where N_r is the number of users connected to access point *r* and $c_{ir} = \log(1 + \gamma_{ir})$ is referred to as the *coupling coefficient*, the throughput that user *i* could achieve if it were connected on its own to the base station $(\gamma_{ir} = \frac{h_{ir}P_i}{\sigma^2})$ is the SNR of user *i*, with P_i being the transmit power of user *i*, h_{ir} the channel gain from user *i* to base station *r*, and σ^2 the variance of the Gaussian noise). This formula is strictly valid in the round-robin scheduling case, although the functional form of the user-perceived throughput on the instantaneous SNR and system load *N* are similar to those in other scheduling systems (e.g., proportional fair). Note that in some cases (e.g., when user throughput is averaged over several time windows) c_{ir} should represent the ergodic capacity instead of the instantaneous Shannon capacity; however, this case is beyond the scope of the analysis in this section.

In this model, the key problem is to determine which user will connect to which network standard (out of the two considered network types), i.e., we have a verticalhandover problem. Clearly this suggests the use of a non-cooperative-game model for the users.

Formally, we define a non-cooperative game (in strategic form) with the following components:

- The *players* are the users in the set \mathcal{N} .
- The *strategy* of each user $i \in \mathcal{N}$ can be represented by variable σ_i , where $\sigma_i = +1$ if user *i* selects standard 1, and $\sigma_i = -1$ if user *i* selects standard 2.

• The *utility* for each user *i* is given by the throughput (9.97). In light of the definition of a user's strategy, (9.97) can be rewritten as:

$$u_i(\boldsymbol{\sigma}) = \frac{1}{N} \left(\frac{c_{i1}(1+\sigma_i)}{1+m} + \frac{c_{i2}(1-\sigma_i)}{1-m} \right), \tag{9.98}$$

where c_{i1} and c_{i2} represent, respectively, the coupling coefficients when user *i* selects the first or second standard, and $m = \frac{1}{N} \sum_{i=1}^{N} \sigma_i$ is a value referred to as the *aggregate strategy*.

In this game, each user aims to select the wireless standard that can maximize its utility as given in (9.98). We investigate the possibility of finding *mixed-strategy* Nash equilibria. A mixed strategy in this game denotes a probability distribution over the two possible pure strategies $\sigma_i = +1$ and $\sigma_i = -1$. This would mean that user *i* would select the first technology with a certain probability p_i^1 and the second technology with a probability $p_i^2 = 1 - p_i^1$. Each user would then aim to find the probability distribution that maximizes its *expected* payoff.

As discussed in Chapter 3, a mixed-strategy Nash equilibrium always exists. Hence, the key question is how to devise a scheme to reach a mixed-strategy Nash equilibrium for the modeled vertical-handover game.

Dynamics of the vertical-handover non-cooperative Game

In this subsection, we develop a scheme to describe how the users reach a mixed-strategy Nash equilibrium. We can utilize the approach in [131], where each user keeps track of the performance it could have achieved at each time t when selecting any standard r. The game is iterated allowing the users to keep track of their strategies' performance and to choose the one that performs best for them. Each player keeps a recursive score,

$$U_i^r(t+1) = U_i^r(t) + u_i^r(t), \ r = 1, 2, \tag{9.99}$$

where $u_i^r(t)$ is the payoff, as per (9.98), that user *i* would have received at time *t* if it had used standard *r*, r = 1, 2. In this model, each user can, at time *t*, use standard *r* with probability $p_i^r(t)$, as given by the exponential *logit* model:

$$p_i^1(t) = \frac{1}{1 + e^{+\eta \Delta U_i(t)}}, \quad p_i^2(t) = 1 - p_i^1(t) = \frac{1}{1 + e^{-\eta \Delta U_i(t)}}, \quad (9.100)$$

where $\Delta U_i = U_i^1 - U_i^2$ and η is parameter that controls the users' learning rate. Considering a continuous time limit, we can derive the dynamics of the system:

$$\frac{dp_i^r}{d_t} = \eta p_i^r (u_i^r(t) - \langle u_i(t) \rangle), \ r = 1, 2,$$
(9.101)

where $\langle \cdot \rangle$ denotes an averaging over the strategies p_i^r of user *i*. Equation (9.101) is reminiscent of the standard replicator dynamics equation used in evolutionary games, as discussed in Chapter 6. It is well known that the stable points of this equation are Nash equilibria. Thus, by adhering to the selfish scheme of selecting the wireless standard that

yields the greatest individual payoff with a probability that depends on the disparity of the payoffs, the users eventually converge to a steady state that is a Nash equilibrium.

To analyze the performance of vertical handover based on the mixed-strategy Nash equilibrium as given by (9.100), a variety of simulations were run in [131]. In essence, the results show that the network reaches a mixed-strategy Nash equilibrium in a relatively small number of iterations, i.e., the vertical-handover game converges quickly. Moreover, compared to the case with no handover as well as to another dynamic vertical-handover algorithm, it is seen that using the studied game-theoretic framework yields the best performance, at the cost of an increased vertical-handover rate. Another interesting result is that the Nash equilibrium is reasonably efficient in that it achieves about 87.3 percent of the performance of the global optimal case, i.e., the social welfare (aggregate throughput) maximizing case. In essence, by formulating the vertical handover as a non-cooperative game, we see that, even though users do not communicate with one another and act on the basis of a completely selfish agenda, they quickly learn to perform with an unexpected efficiency. In fact, their performance closely rivals the (exponentially hard to calculate) optimal distribution, which maximizes the aggregate throughput and would be difficult to implement even within a centrally controlled network. Future studies could extend the work to multiple technologies, as discussed in [131], as well as taking into account the cost of vertical handover (in terms of additional delay or reduced throughput efficiency).

We highlight that in this section, it is assumed that all users have complete information on the game in terms of payoffs, load, downlink SNR, etc. In the next subsection, we investigate a model for vertical handover with incomplete information.

9.5.2 Network selection with incomplete information

Most of the existing work on vertical handover assumes that users have complete information on one another. Within a game-theoretic framework, this implies that each user is perfectly aware of the behavior and decisions of the other users. In practice, although the users can make a best response to the current state of the network, they lack the ability to predict the behaviors of others based on past actions. In this case, it is appropriate to utilize a non-cooperative game with incomplete information, i.e., a Bayesian game, as explained in Chapter 4. In particular, we adopt the approach of [263], where a Bayesian game-based approach is formulated by considering a vertical-handover game in which the users have different bandwidth requirements. Since the preference (i.e., utility) for a mobile user is private information, each user has to make a network selection given only the distribution of the preferences of other users.

In the rest of this section, we describe the studied network-selection model, formulate it as a static Bayesian game, and study its possible dynamics. We conclude by investigating the impact of different system parameters on the performance of the studied game in a practical setting.

Network selection as a Bayesian game

Consider a service area in the coverage of a heterogeneous wireless environment consisting of multiple access networks types (e.g., WLAN, WiMAX, 3G). Without loss of generality, we consider a service area *a* with three access networks, similar to the one shown in Fig. 9.11. *N* users are deployed, with each user periodically receiving beacon signals from the base stations (or access points) of the available access networks. We let C_i denote the set of candidate access networks for user *i*. Each user *i* can choose to connect to any network in its candidate set C_i . Hereinafter, we assume that all users in area *a* have the same candidate access set $C_i = C$, which consists of K = |C| access networks, where $|\cdot|$ is the cardinality of a set operator.

Each access network *k* employs a fixed-price scheme setting a price p_k per connection per unit of time for using its network. Furthermore, all users accessing the same network are considered to be sharing, equally, the available bandwidth. Thus, the bandwidth of user *i* received from network *k* is $\tau_i^k = \frac{B_k}{N_k}$, where B_k is the available bandwidth of network *k* and N_k is the total number of users choosing network *k* in service area *a*, with $N = \sum_{k \in C} N_k$.

Each user needs to select the most appropriate network, given its own objectives. For this purpose, we can formulate a non-cooperative game with incomplete information, i.e., a Bayesian network-selection game. In this game, we will use the minimum bandwidth requirement, which is private information to represent the type of a user. Furthermore, the uncertainty of the minimum bandwidth requirement will be taken into account. Formally, the studied Bayesian network-selection game has the following components:

- The *players* are the *N* users in service area *a*.
- The *action* of any user is to select an access network k from the candidate access set C. Let $\Delta = \{\mathbf{y} = \begin{bmatrix} y_1 & \cdots & y_k & \cdots & y_K \end{bmatrix}^T \in \mathbb{R}_+^K | \sum_{k \in C} y_k = 1 \}$ denote the set of probability distributions over the actions, where y_k represents the probability of choosing network k.
- The *type* of any user *i* is the minimum bandwidth requirement b_i ∈ Γ, where Γ is the type space. We assume that all users of a particular type have the same probability distribution with a probability density function denoted by *f*(b_i).
- The *strategy* of any user $i, s_i : \Gamma \to \Delta$, is a mapping from the type space to the action distribution set. $s_i(b_i) = [s_i^1(b_i) \cdots s_i^k(b_i) \cdots s_i^k(b_i)]^T$ represents the probability distribution over the actions given the Bayesian strategy s_i and the minimum bandwidth requirement b_i , where $s_i^k(b_i)$ is equal to y_k . For simplicity, hereinafter we use $s_i(b_i)$ and s_i interchangeably. The set of all strategies is denoted by Ω .
- For the underlying Bayesian network-selection game, we let $\overline{\pi}_i$ denote the expected *payoff* for user *i*, which is defined as the utility received from the bandwidth minus the connection fee. Later, when we deal with the evolutionary process, the handover cost needs to be considered, and the instantaneous payoff for user *i* at decision epoch *m* will be denoted by $\pi_i(m)$.

Given the above-formulated Bayesian game, let us look more closely at a possible utility function. The instantaneous utility for user i in selecting network k can be expressed as

$$\pi_{i}^{k} = \begin{cases} u(\tau_{i}^{k}) - p_{k}, & \tau_{i}^{k} \ge b_{i}, \\ -p_{k}, & \tau_{i}^{k} < b_{i}, \end{cases}$$
(9.102)

for $i \in \{1, 2, ..., N\}$ and $k \in C$, where $u(\tau_i^k) = \alpha \log (1 + \beta \tau_i^k)$. In particular, $u(\tau_i^k)$ is a concave function representing the utility that user *i* can extract, given an allocated bandwidth τ_i^k , from network *k*, and p_k is the price charged by network *k* (i.e., the connection fee). The utility function in (9.102) implies that, whenever the minimum bandwidth requirement of a user *i* cannot be met (i.e., the received bandwidth is less than the threshold b_i), the payoff for the user becomes equal to the negative value of the price paid for the connection. Otherwise, the utility in (9.102) monotonically increases as the allocated bandwidth increases. This utility function is used in many applications on the Internet (e.g., elastic services using transmission control protocol (TCP)) [263].

Let $\delta = \{s_1, s_2, ..., s_N\}$ denote the strategy profile in the considered Bayesian networkselection game, which is the set of strategies adopted by the *N* users. For ease of presentation, the strategy profile can be represented as $\delta = \{s_i, \mathbf{s}_{-i}\}$, where s_i is the strategy of user *i* and \mathbf{s}_{-i} represents the vector of strategies of all users except user *i*. Similarly, the set of types of all users can be denoted as $\{b_i, \mathbf{b}_{-i}\}$, where \mathbf{b}_{-i} is the vector of types of all users except user *i*.

The expected number of users choosing network k, given all other users' strategies \mathbf{s}_{-i} and types \mathbf{b}_{-i} , is

$$i_k(\mathbf{s}_{-i}, \mathbf{b}_{-i}) = \sum_{j=1, j \neq i}^N s_j^k(b_j), \qquad (9.103)$$

and the expected number of users choosing network k containing all possible type combinations is expressed as

$$i_{k}(\mathbf{s}_{-i}) = \int_{b_{1}} \dots \int_{b_{j}} \dots \int_{b_{N}} i_{k}(\mathbf{s}_{-i}, \mathbf{b}_{-i}) \prod_{j=1}^{N} f(b_{j}) db_{N} \dots db_{j} \dots db_{1}, \qquad (9.104)$$

for $j \neq i$. Therefore, if user *i* chooses to access network *k*, the total number of users expected to choose network *k* becomes

$$I(N_k) = 1 + i_k(\mathbf{s}_{-i}). \tag{9.105}$$

Given all other users' strategies, the bandwidth allocated to user i by network k is

$$\tau_i^k(\mathbf{s}_{-i}) = \frac{B_k}{l(N_k)}.\tag{9.106}$$

Let $\Phi_i^k(\mathbf{s}_{-i}, b_i)$ denote the probability that user *i* satisfies the minimum bandwidth requirement by choosing network *k*, given all other users' strategies \mathbf{s}_{-i} . $\Phi_i^k(\mathbf{s}_{-i}, b_i)$ can be defined as

$$\Phi_i^k(\mathbf{s}_{-i}, b_i) = \Pr[\tau_i^k(\mathbf{s}_{-i}) > b_i].$$
(9.107)

If user *i* chooses network *k*, the expected payoff to user *i* can be expressed as

$$\overline{\pi}_{i}^{k}(\mathbf{s}_{-i}, b_{i}) = \Phi_{i}^{k}(\mathbf{s}_{-i}, b_{i}) \left(u(\tau_{i}^{k}(\mathbf{s}_{-i})) - p_{k} \right) - [1 - \Phi_{i}^{k}(\mathbf{s}_{-i}, b_{i})]p_{k}.$$
(9.108)

In (9.108) we have the expected payoff to the users for the formulated Bayesian networkselection game. Nonetheless, the formulated game is static, in the sense that the users make their decisions only once. For vertical handover, it is of interest to allow the users to adapt their network-selection decisions over time. To accurately model the dynamics of network selection, we consider an evolutionary process in which the static Bayesian game evolves over time. For this dynamic process, which is performed iteratively, we need to consider the cost of handover (e.g., due to delay and loss). Thus, at any decision epoch m-1, if a user decides to perform a handover, i.e., to switch its selected network from k(m-1) to k(m) (at decision epoch m), with $k(m) \neq k(m-1)$, then a cost c_i is incurred by user *i*. As a result, in the dynamic network-selection process, the instantaneous payoff to user *i* at decision epoch m can be expressed as

$$\pi_i(m) = \begin{cases} \overline{\pi}_i^{k(m)}, & k(m) = k(m-1), \\ \overline{\pi}_i^{k(m)} - c_i, & k(m) \neq k(m-1). \end{cases}$$
(9.109)

Nash equilibrium of the static Bayesian game

We now investigate the Nash equilibria of the static Bayesian network-selection game. To find a Nash equilibrium, the expected payoff to user *i*, considering the action distribution **y** and the strategy s_i , are derived. From (9.108),

$$\overline{\pi}_i(\mathbf{y}, \mathbf{s}_{-i}, b_i) = \sum_{k \in \mathcal{C}} \overline{\pi}_i^k(\mathbf{s}_{-i}, b_i) y_k.$$
(9.110)

According to (9.110), $\overline{\pi}_i(s_i, \mathbf{s}_{-i}, b_i)$ can be written as

$$\overline{\pi}_i(s_i, \mathbf{s}_{-i}, b_i) = \sum_{k \in \mathcal{C}} s_i^k(b_i) \overline{\pi}_i^k(\mathbf{s}_{-i}, b_i).$$
(9.111)

Therefore, the expected payoff for user *i*, given strategy profile $\{s_i, \mathbf{s}_{-i}\}$, can be expressed as

$$\overline{\pi}_{i}(s_{i}, \mathbf{s}_{-i}) = \int_{\Gamma} \overline{\pi}_{i}(s_{i}, \mathbf{s}_{-i}, b_{i})f(b_{i})db_{i}$$

$$= \int_{\Gamma} \sum_{k \in \mathcal{C}} s_{i}^{k}(b_{i}) \{\Phi_{i}^{k}(\mathbf{s}_{-i}, b_{i}) \left(\mathscr{U}(\tau_{i}^{k}(\mathbf{s}_{-i})) - P_{k}\right)$$

$$- \left(1 - \Phi_{i}^{k}(\mathbf{s}_{-i}, b_{i})\right) P_{k} \}f(b_{i})db_{i}.$$
(9.112)

Let $r(\mathbf{s}_{-i})$ denote the best response of user *i* given other users' strategies \mathbf{s}_{-i} . For every type of user *i*, the best response is

$$r_i(\mathbf{s}_{-i}, b_i) = \arg\max_{\mathbf{y} \in \Delta} \overline{\pi}_i(\mathbf{y}, \mathbf{s}_{-i}, b_i).$$
(9.113)

Recall from Chapter 4 that a strategy profile δ^* is a Bayesian Nash equilibrium if and only if no user can benefit by unilaterally changing its strategy, or even just an action under

a certain type. Formally, a strategy profile $\delta^* = \{s_i^*, \mathbf{s}_{-i}^*\}$ is a Nash equilibrium if and only if $\forall s_i \in \Omega, \overline{\pi}_i(s_i^*, \mathbf{s}_{-i}^*) \geq \overline{\pi}_i(s_i, \mathbf{s}_{-i}^*)$ for all $i \in \{1, 2, ..., N\}$, and $s_i^*(b_i) = r_i(\mathbf{s}_{-i}^*, b_i)$ for every *i* and b_i . Next we discuss how this equilibrium can be obtained for the studied network-selection game.

Bayesian Dynamics of network selection

While in the previous subsections we formulated the static Bayesian game for vertical handover, in this subsection we develop strategies by which the game can evolve over time. The evolutionary strategy that will be studied is based on the process of Bayesian best-response dynamics common for modeling the behavior of Bayesian games. We also inspect how this process can be utilized to obtain the Bayesian Nash equilibrium. To derive a process of network selection based on Bayesian best-response dynamics, we first define two important operators:

- The operator e : Ω → Δ is a mapping from the set of all strategies to the set of probability distributions over the actions.
- The operator g : Ω → Δ is a mapping from the set of probability distributions over the actions to the set of all strategies.

We note that these two operations are not specific to any user *i*, since this network-selection game is symmetric (i.e., the action set and type distribution are identical for every user in the same area). Thus, for notational convenience and because of symmetry, the strategy of any user *i* will be denoted by *s* instead of s_i . Furthermore, for studying the dynamics of the considered network-selection game, the following definitions are needed:

DEFINITION 9.5 Let e(s) denote the aggregate network-selection distribution induced by a Bayesian strategy $s \in \Omega$. This aggregate network-selection distribution can be expressed as

$$e(s) = (e(s)_1, e(s)_2, \dots, e(s)_K), \qquad (9.114)$$

where $e(s)_k = \int_{\Gamma} s^k(b) f(b) db$, $k \in C$ denotes the proportion of users in a service area choosing network k under strategy s.

DEFINITION 9.6 The best response $g(\mathbf{x})$ corresponding to the social aggregate network-selection distribution $\mathbf{x} = [x_1 \cdots x_k \cdots x_K]^T$, where x_k represents the aggregate proportion of users choosing network k, can be expressed as

$$g(\mathbf{x}) = \arg \max_{\mathbf{y} \in \Delta} \pi(\mathbf{y}, \mathbf{x}, b), \qquad (9.115)$$

where $\pi(\mathbf{y}, \mathbf{x}, \mathbf{b})$ is the payoff obtained under the selection distribution \mathbf{y} , the social aggregate distribution \mathbf{x} , and the minimum bandwidth requirement \mathbf{b} .

According to Definition 9.5, the aggregate distribution **x** can be induced by certain Bayesian strategies. Therefore, $\overline{\pi}_i(\mathbf{y}, \mathbf{s}_{-i}, b_i)$ is equivalent (i.e., after operation *e*) to

 $\pi(\mathbf{y}, \mathbf{x}, b)$ in the underlying static Bayesian game. For the dynamics (i.e., considering the handover cost), $\pi(\mathbf{y}, \mathbf{x}, b)$ can be obtained jointly from (9.108), (9.109), and (9.110). In fact, it is easy to see the equivalence between the best response $r(\mathbf{s}_{-i})$ defined in (9.113) and the best-response correspondence $g(\mathbf{x})$ in the underlying static game.

In games with complete information, it is often the case that the best response can contain multiple strategies. In contrast, in a Bayesian setting such as the considered Bayesian best-response dynamics, if the type distribution is sufficiently diverse and smooth, then the best response $g(\mathbf{x})$ returns a single value (and, hence, is a function). Hereinafter, we assume that $g(\mathbf{x})$ yields a single value.

According to Definitions 9.5 and 9.6, each Bayesian strategy *s* induces the network-selection distribution e(s), and the best response to the distribution can be expressed as g(e(s)). The definition of Bayesian best-response dynamics for the considered vertical-handover game is, thus [263]:

DEFINITION 9.7 For the considered network-selection game, the Bayesian bestresponse dynamics is described by the law of motion in the space of Bayesian strategies, as follows:

$$\dot{s} = g(e(s)) - s.$$
 (9.116)

For continuity of (9.116), the L^1 norm is used to measure the distances of Bayesian strategies. Furthermore, the rest points of the Bayesian best-response dynamics form the set of Bayesian Nash equilibria. In [263], using the Lipschitz continuity property, it is shown that the existence and uniqueness of the solution to the Bayesian best-response dynamics of the network-selection game are guaranteed.

Aggregate best-response dynamics for network selection

Because of the complexity in analyzing the Bayesian best-response dynamics in the L^1 space, as an alternative one can apply the *aggregate best-response dynamics*, defined as follows for the considered vertical-handover game:

$$\dot{\mathbf{x}}_t = \gamma (\boldsymbol{e}(\boldsymbol{g}(\mathbf{x}_t)) - \mathbf{x}_t), \qquad (9.117)$$

where \mathbf{x}_t is the aggregate network-selection distribution at time *t*, and γ is the learning rate representing the proportion of users adjusting their strategies towards their best response to the current network-selection distribution at each selection epoch. Operators $e(\cdot)$ and $g(\cdot)$ have the same definitions as in the Bayesian best-response dynamics, i.e., (9.114) and (9.115), respectively.

We let $\mathbf{x}(m)$ denote the social network-selection distribution at selection epoch m. Notice that $\mathbf{x}(m)$ is different from \mathbf{x}_t . $\mathbf{x}(m)$ is a network-selection distribution point that is the weighted best response (i.e., considering the learning rate) to $\mathbf{x}(m-1)$, while \mathbf{x}_t describes the path from $\mathbf{x}(m-1)$ to $\mathbf{x}(m)$.

As with the Bayesian best-response dynamics, if the type distribution is sufficiently diverse and smooth, $g(\mathbf{x})$ is single-valued and Lipschitz-continuous [263]. Therefore, the solution to the aggregate best-response dynamics exists and is unique, as proven in [263]. The rest points (i.e., solution points of (9.117)) of the aggregate dynamics for

the Bayesian vertical-handover game constitute the set of equilibrium network-selection distributions. A one-to-one correspondence is established between the Bayesian Nash equilibrium and the equilibrium distribution of aggregate best-response dynamics as discussed in [263], which enables an analysis of the Bayesian dynamics through the aggregate dynamics.

For a given initial aggregate network-selection distribution $\mathbf{x}(0)$, the best-response Bayesian strategy $g(\mathbf{x}(0))$ can be obtained from (9.115), and $e(g(\mathbf{x}(0)))$, the expected network-selection distribution induced by $g(\mathbf{x}(0))$, can be calculated according to Definition 9.5. To reach the equilibrium distribution, many iterations of network selection need to be performed to construct a convergence trajectory. Within an epoch, \mathbf{x}_t describes the path from the initial distribution state to a best-response distribution. At the next selection epoch, this best-response distribution is considered as the initial state. Therefore,

$$\mathbf{x}(m) = \gamma e(g(\mathbf{x}(m-1))) + (1-\gamma)\mathbf{x}(m-1), \qquad (9.118)$$

where γ is the learning rate.

Having formulated the dynamics of the studied game, we next study various properties of the game, based on simulation results from [263].

Impact of system parameters on the equilibrium distributions

To get some insight into the performance of the formulated Bayesian game in a practical heterogeneous network, we use the simulation setup of [263]. In this setup, we consider the coverage area *a* of an IEEE 802.11b access point which is also in the coverage area of an IEEE 802.16 cell and of a CDMA-based cellular network cell, as in Fig. 9.11. Area *a* is totally overlapped as shown in Fig. 9.11. The parameters of the networks are found in [263]. In particular, in this subsection we are interested in the impact of system parameters (e.g., learning rate γ and handover cost c_i) on the equilibrium distributions.

First, in Fig. 9.12, we present the phase portrait and the convergence property of the aggregate best-response dynamics for the considered network-selection game. The network-selection distribution states are mapped from three-dimensional space to a triangle in two-dimensional space. The vertices A, B, and C represent the selection distributions $[x_1 = 1 \quad x_2 = 0 \quad x_3 = 0]$, $[x_1 = 0 \quad x_2 = 1 \quad x_3 = 0]$, and $[x_1 = 0 \quad x_2 = 0 \quad x_3 = 1]$, respectively. The phase portrait shows solution trajectories of the aggregate dynamics, with convergence to equilibrium distributions from different initial states. For example, starting from an initial state *D*, the dynamics follows a trajectory composed of linear orbits toward the best responses, finally reaching equilibrium distribution. The equilibrium distributions correspond to the cyclically stable set, which may have one or multiple points.

Figure 9.13 shows the adaptation of network selection for different handover costs, using the WiMAX network as an example. The handover cost is due to handover delay or packet loss. When this cost is small (e.g., $c_i = 0.05$), users are more willing to churn to another network if the payoff is higher (e.g., because of larger allocated bandwidth or cheaper price). For example, for $c_i = 0.05$, the proportion of users choosing WiMAX access network fluctuates. When the handover cost is large (e.g., $c_i = 0.25$), users have



Fig. 9.12 Phase portrait of aggregate best-response dynamics.



Fig. 9.13 Impact of handover cost on dynamics.

no incentive to switch even if the new allocated bandwidth is larger or the price is lower. Thus, the proportion of users choosing to churn to other networks is much lower for large costs.

We highlight how price affects the equilibrium in Fig. 9.14 by varying the price for the WLAN connection. As the price increases, the proportion of users choosing WiMAX and cellular increases. In this case, the proportion of users choosing the cellular access



Fig. 9.14 Impact of price of WLAN on equilibrium distribution.



Fig. 9.15 Impact of learning rate on dynamics.

network is smaller than that of WiMAX, owing to the smaller capacity of the cellular network. It is worth noting that even in the case where the price of WLAN is zero, some users would still prefer another network at a higher price because of their QoS needs. In this example, with a low price the WLAN quickly becomes congested, which drives users to less congested technology.

In Fig. 9.15, we show how the selection of a particular technology (e.g., WiMAX in this figure) is affected by variations in the learning rate. When the learning rate is low, the

impact of the adjusted strategies on the aggregate network-selection distribution is small, and variations in the solution trajectory are small. In contrast, at high learning rates, wide fluctuations are observed in the trajectory, before it converges to the equilibrium.

9.6 Summary

Cellular and broadband access networks are set to continue dominating the wireless industry in the next few years, whether it be through traditional systems such as 3G networks and WiMAX or through upcoming technologies such as 5G and femtocell networks. These networks confront a number of technical challenges arising from competitive and cooperative behavior from wireless devices, making them candidates for modeling using game-theoretic tools. Game theory provides analytical techniques suitable for tackling key problems such as resource allocation, network formation, admission control, network selection, and others. This chapter has showed that concepts such as the Nash equilibrium, the Stackelberg equilibrium, and network-formation games provide useful approaches to characterizing the outcome of a variety of problems such as allocation of subcarriers in OFDMA multi-cell networks, power control in CDMA networks, bandwidth allocation in IEEE 802.16 networks, and architecture formation in multi-hop next-generation wireless networks. With the emergence of novel technologies such as green communication, interference alignment, heterogeneous technologies, and femtocell deployment, it is expected that game-theory-based models will become more abundant, notably in the context of communication problems involving user interaction, resource allocation, fairness, and distributed optimization.

IEEE 802.11 wireless local area networks (WLANs) have been widely deployed in many places for both residential and commercial use. The IEEE 802.11 standard supports two major configurations - i.e., the point - coordination function (PCF) and the distributedcoordination function (DCF). With PCF, the transmission in the network is based on a central node (i.e., an access point). Client nodes listen to the channel and wait for the signal from the access point. Once permission is sent by the access point, the client node can start data transmission. On the other hand, with DCF, the nodes employ carrier-sense multiple-access with collision avoidance (CSMA/CA) for MAC protocol. Each node can transmit independently, based on the availability of the channel. In particular, with CSMA/CA, the nodes listen for the channel status. If the channel is busy, the node defers its transmission by waiting for a backoff period. If a node senses a channel is idle, it will wait for a certain period of time and start transmission. In this case, multiple nodes can start transmissions at the same time, which results in collision. The colliding nodes will wait for the backoff period, and then sense for transmission again. To avoid performance degradation arising from packet collision, the backoff period can be adjusted, according to a specific rule, on the basis of the congestion level in the network (e.g., the rate of packet collisions). The IEEE 802.11 standard specifies a rule for the CSMA/CA protocol so that the network can work efficiently. However, if this rule is modified by the user (e.g., setting a smaller contention window), although this user will receive higher throughput, it will degrade the performances of other users. This conflict situation can be analyzed using game theory. Also, many radio-resource-management issues in IEEE 802.11-based WLAN can be optimized using game theory – for example, power and rate control, access point selection, and service/access pricing. The equilibrium solution (e.g., Nash) can provide stable solution for a WLAN users.

In this chapter, we review the game models developed to analyze the performance of WLANs with rational users and service providers. These models have been developed to deal with the following issues:

• MAC protocol design. A WLAN user can modify the CSMA/CA protocol to achieve higher throughput. However, this will degrade performance for other users. This conflict situation is studied in [88, 103, 119], where static and dynamic game models are formulated. While users can strategically choose the size of the contention window or the channel-access probability, a solution in terms of equilibrium strategy can be obtained to ensure efficient and fair throughput for the users in a WLAN.

- Access point selection. In some areas, multiple WLAN access points may be available (e.g., commercially operated by service providers or network operators). A rational user will select the access point with the lowest cost and/or the highest throughput. In an uncoordinated environment, the transient and steady-state behavior of such users can be studied using game theory [442, 366] (e.g., an evolutionary game). With this game model, the access point's owner (i.e., the service provider) can adjust the parameters (e.g., the service price) so that revenue is maximized.
- Admission control. Because of the limited radio resource, an admission-control mechanism is required for WLAN to ensure that the quality-of-service (QoS) performance for the user will not be degraded below an acceptable level. A non-cooperative game model is formulated for making decisions on admission control [268]. The benefits for both service providers and new users are taken into account. In particular, the revenue of the service provider is maximized, and the new user is satisfied with the received QoS performance.
- Service pricing. With commercial WLAN access, a service provider needs to optimize the price charged to the user in order to maximize revenue. Users with different applications have different preferences for service and QoS performance. In the absence of knowledge about the preferences of the users, a game with incomplete information can be formulated to obtain perfect Bayesian equilibrium for the price charged by the service provider [344].

10.1 MAC protocol design

IEEE 802.11-based WLAN employs the carrier-sense multiple-access with collision avoidance (CSMA/CA) protocol [72]. This protocol relies on random packet transmission for the wireless channel shared among multiple active users. IEEE 802.11-based WLAN can operate efficiently and flexibly when users follow the CSMA/CA protocol strictly. For example, users must perform a binary exponential backoff process before starting transmission, to avoid congestion of the network. However, with the emergence of the programmable IEEE 802.11 adaptor [399], users can override the CSMA/CA protocol to maximize their performance in a WLAN.

In [88], the stability and efficiency of a WLAN with a greedy user (i.e., a cheater [272]) are studied. This greedy user aims to modify the programmable network adaptor [454] by changing the backoff mechanism to gain higher throughput (e.g., fixed and smaller contention window size). Different types of games are formulated to analyze the behavior of the greedy user and the performance of the network. First, the case of non-cooperative users (i.e., both normal users with standard CSMA/CA protocol and cheaters with modified protocol) is considered. The Nash equilibria of the cheaters in this single-stage (i.e., one-shot) game are investigated. Then the case of cooperative users is considered. The Pareto-optimal strategy of the cheaters is defined, and the equilibria of a dynamic game are analyzed. An algorithm to reach such an efficient strategy is proposed. It is shown that, with cooperative behavior by the users, system throughput is higher. However, the individual throughput of any user may not be maximized because the cooperation may

not be credible. Therefore, detection and punishment mechanism for non-cooperative users is introduced.

In the system model, there are N wireless nodes (i.e., active WLAN users). All nodes always have data to transmit (i.e., they are infinitely backlogged). All nodes operate on CSMA/CA with distributed coordination function (DCF) mode [14]. N_{cht} out of N nodes are the cheaters, where the binary exponential backoff mechanism is overridden, and a fixed contention window W_i is used for cheater *i*. In the following game formulation, the players are the users in the WLAN. The strategy is the size of the contention window, and the payoff is the throughput, denoted by r_i .

10.1.1 Static game

First, the static game with a cheater in an IEEE 802.11-based WLAN is considered. In this game, all players make their decisions (i.e., choice of the size of the contention window W_i) simultaneously. The game is assumed to be played only once. The payoff (i.e., throughput) to node *i* with the standard CSMA/CA protocol can be determined from [72]:

$$r_{i} = \frac{P_{i}^{\text{suc}}\overline{L}}{P^{\text{suc}}T_{\text{suc}} + P^{\text{col}}T_{\text{col}} + P^{\text{idl}}T_{\text{idl}}},$$
(10.1)

where P_i^{suc} is the probability that any node successfully transmits the packet and $P^{\text{suc}} = \sum_k P_k^{\text{suc}}$. \overline{L} is the average size of a packet. $P^{\text{idl}} = \prod_k (1 - \tau_k)$ is the probability of a channel being idle, where τ_k is the access probability of node k. $P^{\text{col}} = 1 - P^{\text{idl}} - \sum_k P_k^{\text{suc}}$ is the probability of collision. T_{suc} is the average time duration of successful packet transmission, and T_{idl} is the average duration for a channel to be idle. T_{col} is the average time of collision. However, the cheater will have a fixed contention window, so the model used to obtain the throughput of the CSMA/CA protocol is extended. In this case, the throughput for cheater *i* becomes

$$r_{i} = \frac{\tau_{i}^{\text{acc}} c_{i,1}}{\tau_{i}^{\text{acc}} c_{i,2} + c_{i,3}},$$
(10.2)

where $c_{i,1}$, $c_{i,2}$, and $c_{i,3}$ are positive constants and τ_i^{acc} is the access probability. To maximize throughput r_i , the contention window can be adjusted. As an approximation, W_i is assumed to be a continuous variable. Then the first-order derivative of the throughput expression can be expressed as

$$\frac{\partial r_i}{\partial W_i} = \frac{c_{i,1}c_{i,3}}{\left(\tau_i^{\text{acc}}c_{i,2} + c_{i,3}\right)^2} \frac{-2}{(W_i + 1)^2}.$$
(10.3)

It is observed that this first derivative is always negative. Thus, the throughput of the cheater monotonically increases as W_i decreases. In non-cooperative environment, the cheater will always choose the smallest contention window to achieve the highest throughput (i.e., $W_i = 1$). This result is verified by simulation.

Based on the throughput (i.e., payoff) expression, the Nash equilibria of the game among cheaters can be studied. Let $W_i \in \{1, ..., W_{max}\}$ the strategy of cheater *i* and $W_{-i} = (W_1, ..., W_{i-1}, W_{i+1}, ..., W_{N_{cht}})$ denote the strategies of all cheaters except cheater *i*. N_{cht} is the total number of cheaters, and W_{max} is the largest size of the contention window. Since the throughput r_i is a monotonically decreasing function of W_i , the best response will be $W_i = 1$. Therefore, the Nash equilibrium is defined as $W^* = (W_1^*, ..., W_{N_{cht}}^*)$ for $\exists i$ such that $W_i = 1$. In this case there are multiple Nash equilibria, characterized by the condition that any cheater selects the size of the contention window to be 1. There are two types of Nash equilibria:

- One cheater with contention window of 1. In this case, cheater *i* with $W_i = 1$ and $W_j > 1$ for $j \neq i$ will receive the highest throughput (i.e., $r_i > 0$) since it can transmit before the other cheaters with its larger contention window. However, other cheaters *j* with any $W_j > 1$ will receive zero throughput $r_j = 0$, since they will never succeed in the contention.
- More than one cheater with contention window of 1. In this case, the throughput of all cheaters will be zero, since collision always occurs for cheaters *i* and *j* for which $W_i = W_i = 1$.

Since there are multiple Nash equilibria, in order to maximize the throughput of the WLAN the Nash equilibrium for which there is a single cheater *i* with $W_i = 1$ is selected. It is clear that although this Nash equilibrium yields the highest network throughput, it is totally unfair since only one cheater can enjoy this throughput. Therefore, a fair and efficient solution is introduced.

To achieve a fair and efficient solution, Pareto optimality and a bargaining game [154] are adopted. In a bargaining game, the disagreement point is defined as $r^0 = (r_1^0, ..., r_{N_{cht}}^0)$. The threat point is a set of strategies by which the cheaters obstruct one another, so $r^0 = (0, ..., 0)$. The solution of this bargaining game is

$$r^{\star} = (r_{1}^{\star}, \dots, r_{N_{\text{cht}}}^{\star}) = \arg \max_{r = (r_{1}, \dots, r_{N_{\text{cht}}})} \prod_{i=1}^{N_{\text{cht}}} (r_{i} - r_{i}^{0}),$$
(10.4)

for $r \ge r_i^0$. Although, in this bargaining game, the set of feasible payoffs to all cheaters is not compact and convex, the solution r^* is unique, Pareto-optimal, and fair.

10.1.2 Dynamic game

Since a bargaining solution is desirable, a method to allow the cheaters to reach this strategy is studied in the context of a dynamic game [160]. Assuming that the game is played for infinitely long, and that the players (i.e., cheaters) make decisions based on past outcomes, the utility for cheater i is

$$J_i = r_i - z_i, \tag{10.5}$$

where z_i is a penalty function, defined as $z_i = k_i(\tau_i - \underline{\tau})$, where τ_i is the access probability and k_i and $\underline{\tau}$ are positive constants. With decision variable τ_i , an optimization problem can be defined as follows:

$$\max_{0 \le \tau_i \le 1} = \frac{\tau_i c_{i,1}}{\tau_i c_{i,2} + c_{i,3}} - k_i \times \begin{cases} (\tau_i - \underline{\tau}), & \text{if } \tau_i > \underline{\tau}, \\ 0, & \text{if } \tau_i \le \underline{\tau}. \end{cases}$$
(10.6)

 γ_i^* , the solution of $\frac{\partial J_i}{\partial \tau_i} = 0$, is

$$\gamma_i^* = \frac{1}{c_{i,2}} \left(\sqrt{\frac{c_{i,1}c_{i,3}}{k_i}} - c_{i,3} \right).$$
(10.7)

It is found that the strategy,

$$\tau_{i} = \begin{cases} 0, & \text{if } \gamma_{i}^{*} < 0, \\ \gamma_{i}^{*}, & \text{if } 0 \le \gamma_{i}^{*} \le 1, \\ 1, & \text{if } \gamma_{i}^{*} > 1 \end{cases}$$
(10.8)

is a unique Nash equilibrium of a dynamic game for $T_{suc} = T_{col}$. To reach the Nash equilibrium strategy – i.e., $\tau_i^* = 1 - \frac{1}{\alpha_i}$ – a the gradient-based algorithm [70] is introduced as follows:

$$\alpha_i[t+1]) = \alpha_i[t] - \phi\left(\tau_i^*[t] - \underline{\tau}\right),\tag{10.9}$$

where ϕ is the step size and $\alpha_i[t]$ is a value of α at iteration t. The convergence of this algorithm can be demonstrated by showing that the function $f(\alpha_i[t]) = \alpha_i[t] - \phi(\tau_i^*[t] - \underline{\tau})$ is a contraction mapping.

Moving this Nash equilibrium to the Pareto-optimal point is performed as follows. In the first step, an arbitrary access probability τ_i is selected, and the algorithm in (10.9) is used to obtain the Nash equilibrium. At this equilibrium, one cheater *i* will decrease its τ_i by a small value. This triggers the other cheaters to run the algorithm in (10.9) to reach a new Nash equilibrium. These cheaters compare their new and previous payoffs. If the difference is small, the cheaters stop moving the Nash equilibrium since the set of strategies is already close to Pareto-optimal.

10.1.3 Deviation detection and penalization

To avoid any cheater deviating from the Pareto-optimal strategy, each cheater applies a deviation and penalization mechanism to the deviating cheater. In general, a cheater can deviate from the Pareto-optimal strategy by reducing the size of the contention window to achieve higher throughput. To detect such a deviation, a cheater observes a difference in throughput compared with other cheaters. Because of the broadcasting nature of the CSMA/CA protocol, the cheater measures an average throughput of the other cheaters during a certain time period (i.e., observation-time window size T_{obs}). If the throughput of any cheater is different from other cheaters by more than a tolerance margin κ , then this cheater is identified as a deviating cheater.

Once a deviation is detected, the deviating cheater will be penalized by the other cheaters in the network, using selective jamming. In this case, the packet from the deviating cheater will be jammed by one the other cheaters. In particular, by observing the net allocation vector (NAV) information in the packet from the deviating cheater, the penalizing cheater transmits a signal to jam this packet at the data field for a certain number of bits so that the receiver of the deviating cheater cannot receive the packet correctly. In this way, the throughput of the deviating cheater is decreased. The duration of jamming is $T_{jam} = \left(\frac{r_j}{r_i} - 1\right) T_{obs}$, where r_j and r_i are the throughputs of the deviating and penalizing cheaters, respectively. Once the deviating cheater observes the penalization, it will increase the size of the contention window so that the strategy becomes Pareto-optimal again.

10.1.4 Related work

In [253], the problem of backoff attack in IEEE 802.11-based WLAN is studied. In this work, the user can be honest, greedy, or selfish. The minimum sizes of the contention windows of honest, greedy, and selfish users are those in the standard, 1, and 2, respectively. The repeated game is formulated to ensure that the users behave properly such that the received throughput is maximized and is fair for each user. The strategy – i.e., cooperation via randomized inclination to selfish/greedy play (CRISP) – is proposed to achieve this objective. In [102], a repeated non-cooperative game is presented, and the users apply a Tit-for-Tat (TFT) strategy to avoid deviation from the efficient solution. It is found that there are multiple Nash equilibria in this game in which an efficient equilibrium can be selected. Both single-hop and multi-hop networks are considered in this game [253].

10.2 Random-access control

The CSMA/CA protocol is based on random access, in which the user randomly accesses the channel. This randomness depends on the congestion in the network. In particular, users observe the level of congestion (e.g., collision probability or the average number of idle slots) and adjust the channel-access probability accordingly. The problem of random access by multiple users is formulated as a non-cooperative game [103, 119]. In this case, the dynamics of the access probability for user *i* can be expressed in the form of a feedback system with function $\mathscr{F}_i(\cdot)$, as follows:

$$\tau_i[t+1] = \mathscr{F}_i(\tau_i[t], \mathbf{q}_i[t]), \qquad (10.10)$$

where $\tau_i[t]$ is the access probability at time t, $\mathbf{q}_i[t]$ is a vector of the measured congestion level in the network, and $\boldsymbol{\tau}[t]$ is a vector of the access probabilities of all users. As shown in (10.10), access probability is adjusted as a function of congestion level. If all users are rational, access probability τ_i has to be determined – e.g., to be Nash equilibrium – so that the network is stable. The random-access control game can be formulated as follows. The players are the users in the network. The strategy is the access probability τ_i , and the payoff is the utility minus cost. The payoff can be defined as $u_i(\tau) = \mathscr{U}_i(\tau_i) - \tau_i \mathscr{C}_i(\mathbf{q}_i[t])$, where $\mathscr{U}_i(\tau_i)$ is the utility function (e.g., throughput) and $\mathscr{C}_i(\mathbf{q}_i[t])$ is the cost due to congestion. The general random-access game model is presented in [103], where $\mathscr{C}_i(\mathbf{q}_i[t]) = q_i$ (i.e., cost is the congestion measure). It is shown that there is a channel-access probability vector $\boldsymbol{\tau}^* = [\tau_1^* \cdots \tau_i^* \cdots \tau_N^*]$ which is the Nash equilibrium. *N* is the total number of users in the network. This Nash equilibrium $\boldsymbol{\tau}^*$ is a non-trivial equilibrium if τ_i^* satisfies the following condition:

$$\frac{\partial u_i(\tau_i^*, \boldsymbol{\tau}_{-i})}{\partial \tau_i} = \mathbf{0}, \quad \forall i,$$
(10.11)

where τ_{-i} is the vector of access probabilities of all users except user *i*. This nontrivial Nash equilibrium can achieve efficient and fair performances for all the users. In this case, the utility function $\mathscr{U}(\cdot)$ must be twice continuously differentiable, strictly concave, and increasing. Also, there must exist positive constants μ and ξ such that $\frac{1}{\mu} \geq \frac{1}{\mathscr{U}'_i(\tau_i)} \geq \frac{1}{\xi}$. Finally, the inverse function $(\mathscr{U}'_i)^{-1}(\mathscr{C}_i(\mathbf{q}_i))$ maps any \mathbf{q}_i into any point within the strategy space of τ_i for all users.

10.2.1 Choice of utility function

In [119], various utility functions are summarized for this access-control game. For the homogeneous-user case [208], the utility function is defined as

$$\mathscr{U}_i(\tau_i) = \tau_i + e^{-\zeta} \log(1 - \tau_i), \qquad (10.12)$$

where ζ satisfies the condition $1 - \zeta = \eta e^{-\zeta}$, and $\eta = 1 - T_{slot}/T_{col}$, where T_{slot} and T_{col} are the slot duration and average collision duration, respectively. In this case, the measured congestion level q_i is determined as a function of idle probability, which is obtained by solving the following equation:

$$(1-\tau_i)(1-q_i) = \frac{\phi}{\phi+1} = e^{-\zeta},$$
 (10.13)

where $\phi = \frac{e^{-\zeta}}{1-e^{-\zeta}}$. It is shown that there exist an infinite number of equilibria for this game. In this case, a refinement of the utility function is introduced so that this game has a unique, non-trivial Nash equilibrium.

For heterogeneous users [392], the weight ω_k is assigned to the class k whose \mathcal{U}_k denotes the set of users in this class. The fairness constraint can be defined for traffic classes k and j such that

$$\frac{\tau_k(1-\tau_j)}{\omega_k} = \frac{\tau_j(1-\tau_k)}{\omega_j}.$$
(10.14)

The utility function of user *i* in class *k* is defined as

$$\mathscr{U}_{k,i}(\tau_k) = 1 - \frac{\omega_k \zeta}{\alpha + 1} (1 - \tau_k)^{\alpha + 1}, \qquad (10.15)$$

where $\alpha > 0$ and ζ is defined in the same way as for the homogeneous case. The cost is defined as

$$\mathscr{C}_{k,i}(\mathbf{q}_i) = \frac{\sum_{j=1}^{K} q_{k,i}\omega_j/\tau_j}{q_{k,0}\tau_k},$$
(10.16)

where *K* is the total number of classes, and \mathbf{q}_i is a vector of $q_{k,i}$. The probability of an idle slot is $q_{k,0} = \prod_k (1 - \tau_k)^{|\mathcal{U}_k|}$, $q_{k,i} \approx \zeta_k e^{-\zeta}$, and $\zeta_k = \frac{|\mathcal{U}_k|\omega_k}{\sum_{j=1}^{K} |\mathcal{U}_j|\omega_j} \zeta$. In this case, there is a non-trivial equilibrium in this game for $\tau_i < 1/N$, where *N* is the total number of users.

Alternatively, the payoff function can be defined as

$$U_i(\boldsymbol{\tau}) = \mathscr{U}_i(\tau_i) - \tau_i \prod_{j \neq i} (1 - \tau_j).$$
(10.17)

In this case, the cost to the user is a function of received throughput. It is shown that this game is a supermodular game. That is, the payoff function $U_i(\tau_i, \tau_{-i})$ has increasing differences in (τ_i, τ_{-i}) . As a result, this game possesses at least one Nash equilibrium. In addition, for the utility function

$$\mathscr{U}_i(\tau_i) = \frac{1}{a_i} \left(\frac{(a_i - 1)b_i}{a_i} \ln(a_i \tau_i - b_i) - \tau_i \right), \qquad (10.18)$$

where a_i and b_i are constants, there could be a condition such that the game has a unique, non-trivial Nash equilibrium.

10.2.2 Dynamics of a random-access game

The above discussion is based on a static game model. In [119], the dynamics of a random-access game model is analyzed. In this case, three distributed strategy-update mechanisms are considered for the users.

Best response. The best response is the simplest strategy-update mechanism. All users choose the best response in each iteration *t*, i.e.,

$$\tau_i[t+1] = \max\left(\arg\max_{\tau} U_i(\tau, \boldsymbol{\tau}_{-i}[t])\right).$$
(10.19)

However, if there is more than one best response for the access probabilities, the largest probability will be chosen. In this case, at the steady state, if stationary point is reached, this point will be a Nash equilibrium. In addition, for a supermodular game it is guaranteed that this best-response strategy update converges towards the Nash equilibrium.

Gradient play. In gradient-play strategy update, all users adjust their access probabilities in a specific direction [153]. This direction, $\mathcal{M}_i(\cdot)$, depends on the measured congestion level of the network. The strategy update is

$$\tau_i[t+1] = \mathscr{M}_i(\tau_i[t] + \phi_i[t])(\mathscr{U}'_i(\tau_i[t]) - \mathscr{C}_i(\mathbf{q}_i(\boldsymbol{\tau}[t])))), \quad (10.20)$$

where $\phi_i[t] = \phi[t]$ is the step size. The convergence of this gradient-play strategy update (e.g., given step size) is proved (e.g., in [153]) by evaluating the smallest eigenvalue of the Jacobian of utility and cost functions. It can be shown in [119] that this game has a unique, non-trivial Nash equilibrium.

Jacobi play. In Jacobi-play strategy update, all users adjust access probabilities towards the best-response strategy [273], i.e.,

$$\tau_i[t+1] = \mathscr{M}\left(\tau_i[t] + \phi_i[t](\mathscr{B}_i(\boldsymbol{\tau}[t]) - \tau_i[t])\right), \qquad (10.21)$$

where $\mathscr{B}_i(\cdot)$ is the best response of user *i*. For $\phi_i[t] = 1$, this Jacobi-play update becomes the best-response strategy update. However, for $\phi_i[t] < 1$, it can still be shown (e.g., as in [273]) that the sequence of $\tau_i[t]$ is non-increasing and that the update converges to the Nash equilibrium.

10.2.3 Extension with propagation delay and estimation error

In addition to the above basic strategy-update mechanisms, their dynamics under propagation delay and estimation error can also be analyzed. With propagation delay, the information on network congestion used to update the strategy at time *t* is the result of measurement by user *i* at time \hat{t}_i for $0 \le \hat{t}_i \le t$. The best-response strategy update is

$$\tau_i[t+1] = \mathscr{B}_i(\boldsymbol{\tau}[\hat{t}_i]) = \max\left(\arg\max_{\tau} U_i(\tau, \boldsymbol{\tau}_{-i}[\hat{t}_i])\right).$$
(10.22)

Again, it can be shown that the best-response strategy update with a propagation delay of $t - \hat{t}_i$ will converge to the Nash equilibrium under certain conditions. Similar results are observed for gradient-play and Jacobi-play strategy-update mechanisms.

In general, the congestion level is estimated from conditional collision probability by observing the idle slots between transmissions. In particular, each user updates $\bar{n} = \beta \bar{n} + (1 - \beta) \frac{T_{\text{idl}}}{T_{\text{trans}}}$, where β is the estimation weight and T_{idl} is the number of idle slots during T_{trans} transmission. Then the conditional probability is $q_i = \frac{1 - (\bar{n} + 1)\tau_i}{(\bar{n} + 1)(1 - \tau_i)}$. Because of estimation error, the cost becomes

$$\hat{\mathscr{C}}_i(\mathbf{q}_i(\boldsymbol{\tau}[t])) = \mathscr{C}_i(\mathbf{q}_i(\boldsymbol{\tau}[t])) + w_i[t], \qquad (10.23)$$

where $w_i[t]$ is the estimation error for player *i*. Again, it can be proved, based on evaluation of the smallest eigenvalue of the Jacobian of the payoff function, that the gradient-play and Jacobi-play strategy-update mechanisms with estimation error converge to the Nash equilibrium.

10.2.4 Related work

A similar game model has been developed for the Aloha protocol [307, 227, 107, 108, 372], which is a fundamental of the CSMA/CA protocol. In [307], non-cooperative-game and Stackelberg-game models are proposed to obtain equilibrium channel-access probability for slotted Aloha. The selfishness of the user in accessing the channel is

also considered. In [227], a one-shot, random-access game model is introduced for heterogeneous and selfish users in a network with slotted Aloha protocol. Also, necessary and sufficient conditions for the Nash equilibria are derived. The game model with channel capture is considered in [108].

10.3 Rate selection for VoIP service on WLAN

To support voice service in WLAN using voice over IP (VoIP), the QoS guarantee for data transmission is crucial [286, 97, 449]. In [502], game theory is applied to competition between VoIP users for channel access in a WLAN environment. The user selects the transmission rate through the codec and forward error correction (FEC) mode [439]. In this case, if the high-quality codec is selected, although the quality of voice is better, the communication requires higher bandwidth; consequently, congestion can occur. This congestion can degrade the performance of the communication. Therefore, in a competitive environment, VoIP users must select equilibrium strategies to maximize their payoff, which is defined as the QoS performance of VoIP. This adaptation of the transmission rate is referred to as *end-user congestion control*. In [502], experiment, simulation, and an analytical model based on evolutionary-game theory are used to analyze the consequences of a situation in which all users are allowed to freely choose transmission rate. Controlled laboratory experiments have shown that the aggregated rate (i.e., total transmission rate for all users) is close to the capacity of the WLAN, even though the users are non-cooperative.

10.3.1 Game formulation

The evolutionary-game formulation of rate selection can be described as follows. The players are the VoIP users, and the strategies are the transmission rates in which the user can select the codec and FEC modes accordingly. The payoff is the voice quality. Let *N* denote the total number of VoIP users. *S* is a set of actions or strategies (i.e., a set of available codec and FEC modes) in which the total number of modes is K = |S|. The state of the game can be defined as $\mathbf{n} = (n_1, ..., n_l, ..., n_K)$, where n_l is the number of users choosing strategy *l*. The total number of states is $\frac{(N+K-1)!}{N!(K-1)!}$. Given a state \mathbf{n} , the utility function $U_i(l, \mathbf{n})$ of user *i* using strategy *l* can be determined. To obtain equilibrium, a continuous-time Markov chain whose state is \mathbf{n} can be established as in [327]. Let $\mathbf{n}(l, m) = (n_1, ..., n_l - 1, ..., n_m + 1, ..., n_K)$ denote the state when the user changes strategy from *l* to *m*. The transition rate of the Markov chain from state \mathbf{n} to $\mathbf{n}(l, m)$ is

$$\alpha(\mathbf{n} \to \mathbf{n}(l, m)) = \begin{cases} n_l \phi(U_i(m, \mathbf{n}(l, m)) - U_i(l, \mathbf{n})), & \text{if } (U_i(m, \mathbf{n}(l, m)) > U_i(l, \mathbf{n})) \\ & \text{and } (U_i(l, \mathbf{n}) > \gamma), \\ E_x, & \text{if } U_i(l, \mathbf{n}) \le \gamma, \\ \epsilon, & \text{otherwise.} \end{cases}$$
(10.24)

In this case, the user does not have complete information about the voice quality. Therefore, the user will gradually adapt the strategy to that with higher payoff. The rate of changing strategy is proportional to the difference between current strategy / and new strategy m. The user can make a mistake with small rate ϵ by changing strategy from l to m even though $U_i(l, \mathbf{n}) > U_i(m, \mathbf{n}(l, m))$. In addition, owing to the strict QoS requirement of VoIP, the user will not tolerate poor voice quality. In particular, if the payoff of the current strategy / drops below a threshold γ (i.e., the tolerance level), the user randomly switches to another strategy m with rate E_x . This is referred to as the *exploration rate*. Also, if the voice quality is too poor, the user can stop accessing the channel (i.e., quit the game). Strategy l = 0 corresponds to this case. Again, the rate at which the user stops using VoIP is proportional to the difference between the current payoff and the threshold γ . The user starts using VoIP again given strategy *l* with probability $P_{retry}(l)$. The transition rate from state **n** to $\mathbf{n}(0, m)$ is denoted by $n_0 RP_{retry}(m)$, where n_0 is the number of users who have temporarily stopped using VoIP service and R is the retrial rate. At the steady state (i.e., $t \to \infty$) and $\epsilon \to 0$, the Nash equilibria can be determined as the subset of the state space with non-zero stationary probability [327].

10.3.2 Payoff function

In the above evolutionary-game model, the Nash equilibria can be determined given the payoff function. In [502], a method for quantifing the payoff function based on experiment and simulation is proposed. A mean opinion score (MOS) is used to determine the voice quality. The MOS is obtained from laboratory measurement and network simulation. With this MOS from experiment and simulation, a random neural network (RNN) [341] is trained under various settings to estimate the MOS. The major steps are shown in Fig. 10.1.

First, the wireless channel performances (e.g., loss rate, delay, jitter) are estimated for all states of the Markov-chain model of this game. The traffic of voice flow is generated by a Tangram-II Traffic Generator [130]. Then, in an experiment, the lost packet, mean loss burst size, jitter, and packet delay are measured. In addition, given





the same settings, a simulation is performed to validate the experimental result and to expand the evaluation to a variety of scenarios. As a result, there will be more data to train the RNN.

Then, the QoS performance, in terms of MOS, for VoIP is estimated. This MOS is a subjective quality score that ranges between 1 (unacceptably poor) and 5 (excellent). In the experiment, the method of *pseudo-subjective quality assessment* (PSQA) [122] is used. In the first step of PSQA, the choice of system parameters that affect the transmission quality is made. These parameters are the performance measures for the wireless channel (i.e., loss rate, mean loss burst size, jitter, and delay) and the codec and FEC modes. In the second step, subjective assessment test is performed, and in the third step the results are used to train and validate the RNN. The output of the RNN is MOS. The training data are obtained from the experiment based on the VoIP tool – i.e., VivaVoz, which supports eight codec rates and two FEC modes. The subjects (e.g., students) in the experiment carry on a conversation and provide MOS data on voice quality. This data is processed to obtain the MOS.

The performance evaluation is carried out using experiment, simulation, and analytical modeling. In the first experiment, the subjects (i.e., users) speak freely. They can change the codec rate and FEC mode (i.e., strategy) to maximize perceived voice quality. The experiment ends when no users change strategy within a given period of time. The final strategy adopted by all users is referred to as the convergence point of the behavioral experiment. This experiment is based on the fact that the users change strategy asynchronously with incomplete information. In this case, users know only their own voice quality. However, with these constraints, it is found that the convergence point results in a total transmission rate close to that of the effective network capacity. This result shows that even in an environment without coordination, rational users will perform so that an efficient state (i.e., the network resource is almost fully utilized) is reached. Also, congestion is avoided by user adaptation without any intervention from a centralized controller.

10.4 Access-point selection

IEEE 802.11-based WLAN is deployed to provide commercial Internet access in many places. Also, in one place, there could be multiple available IEEE 802.11-based WLANs. Therefore, the issue of access-point selection arises [442]. In this case, the users can search for the access point that provides a WLAN connection with the best performance (i.e., the highest signal strength or the highest data rate) and the lowest cost. Without the standard protocol in IEEE 802.11, it has been shown [207] that all users with infinite-backlog traffic to transmit will have the same throughput. Therefore, multihoming of users is introduced to split the data transmission to multiple access points [17]. With this multihoming capability, users can take advantage of the diversity of channel quality, to maximize their throughput. The general system model of access-point selection with multihoming of the user is shown in Fig. 10.2. In this system model, there are multiple access points that can be accessed by multiple groups of users (i.e., populations). The



Fig. 10.2 Access-point selection in IEEE 802.11-based WLAN.

payoff for the user is defined as the difference between the utility from throughput and the charged price.

10.4.1 Formulation of a population game

In [442], it can be assumed that the user's station has a single interface that can connect to one access point at a time. Therefore, multihoming means connecting to each access point during different time intervals. Both transmission-control protocol (TCP) and userdatagram protocol (UDP) are considered for the transport-layer protocol. Therefore, the connection time interval is the session duration for TCP and UDP connections. Naturally, a user's connection time that yields a higher transmission rate or has a lower price will be longer. A multihoming protocol is proposed for the user to connect to each access point. In this case, the price charged to the user depends on channel occupancy; this is called *cost-price charging*. With a population game, it is shown that this multihoming of WLAN users under cost-price charging is efficient (i.e., the network throughput is maximized).

A population game [421] with a continuum player set (i.e., an infinite number of players) can be described as follows. There are K classes (i.e., groups) of users. Each class can correspond to a group of users in the same geographical location. Class k has a (population) mass \hat{d}_k . The set of strategies for a class k player is denoted by $S_k = \{1, ..., S_k\}$, where S_k is the number of available access points. The strategy distribution (i.e., the fraction of time) that the users in a group select access point *i* is denoted by n_k^i , and the vector \mathbf{n}_k is defined as $\mathbf{n}_k = [n_k^1 \cdots n_k^s \cdots n_k^{S_k}]$, where $\sum_{i=1}^{S_k} n_d^i = \hat{d}_k$. The vector of strategy distributions for all classes is $\mathbf{n} = [\mathbf{n}_1 \quad \mathbf{n}_2 \quad \cdots \quad \mathbf{n}_K]$. This vector \mathbf{n} is referred to as the state of the system. The payoff for the user in any class depends on this system state. Let the payoff function for users from class k selecting access point *i* be $u_k^i(\mathbf{n})$. The total payoff for the users in class k is then $u_k = \sum_{i=1}^{S_k} u_k^i(\mathbf{n}) n_k^i$.

Given the rationality of users, the multihoming protocol will make a decision on the strategy distribution for each available access point, and this decision will be based on the current estimate of the utility. The decision adaptation of the multihoming protocol can

be modeled by replicator dynamics [211]. With replicator dynamics, the rate of change for a strategy distribution – i.e., $\frac{\dot{n}_k^s}{n^s}$ for users in class k selecting access point s – is

$$\dot{n}_{k}^{s} = n_{k}^{s} \left(u_{k}^{s}(\mathbf{n}) - \frac{1}{\hat{d}_{k}} \sum_{i=1}^{S_{k}} n_{k}^{i} u_{k}^{i}(\mathbf{n}) \right),$$
 (10.25)

where \dot{n}_k^s is the derivative of n_k^s with respect to time *t*, i.e., $\dot{n}_k^s = \frac{dn_k^s}{dt}$. In particular, replicator dynamics is defined for evolutionary success. This success is a function of the difference between the fitness of strategy *s* and the average fitness of all strategies. The dynamics defined in (10.25) will maintain the fixed size of the population, i.e., $\sum_{i=1}^{S_k} n_k^i = \hat{d}_k$.

Alternatively, the dynamics of strategy distribution can be modeled using Brown–von Neumann–Nash (BNN) dynamics [84]. With BNN dynamics, the excess marginal payoff for strategy s relative to the average payoff to users in the same class is computed from

$$\gamma_k^s = \max\left(u_k^s(\mathbf{n}) - \frac{1}{\hat{d}_k} \sum_{i=1}^{S_k} n_k^i u_k^i(\mathbf{n}), 0\right).$$
(10.26)

BNN dynamics is defined as

$$\dot{n}_{k}^{s} = \hat{d}_{k}\gamma_{k}^{s} - n_{k}^{s}\sum_{j=1}^{S_{k}}\gamma_{j}^{s}.$$
(10.27)

The difference between BNN and replicator dynamics is that BNN allows extinct strategies (i.e., strategies *s* with $n_k^s = 0$) to be revived. This capability is matched by the actual system, since a user can try a different access point (even though there is nobody using that access point) to determine the possibility of higher payoff.

Then, given the dynamics of strategy distribution, the payoff for a user has to be defined. This payoff is a function of throughput and cost. However, throughput depends on the transport layer-protocol, which could be UDP or TCP. For both protocols, the throughput per unit mass of users (in bits per second) in class k selecting access point s has the same form:

$$r_k^s(\mathbf{n}^s) = \frac{\eta_k}{\sum_{j=1}^K \mu_j n_j^s},\tag{10.28}$$

where η_k and μ_k are constants. The exact throughput expressions for UDP and TCP protocols can be obtained from [261] and [85, 269], respectively. The payoff function is then

$$u_k^s(\mathbf{n}) = r_k^s(\mathbf{n}) - \mathscr{C}_k^s(\mathbf{n}), \qquad (10.29)$$

where $\mathscr{C}_k^s(\mathbf{n})$ is the cost for users in class k, i.e.,

$$\mathscr{C}_{k}^{s}(\mathbf{n}) = \psi_{k}^{s}(\mathbf{n}) \sum_{i=1}^{K} R_{i,k}^{s}(\mathbf{n}), \qquad (10.30)$$

and ψ_k^s is the occupancy factor. $R_{i,k}^s$ is the total throughput that users in class k receive when connecting to access point s and the system state is **n**, i.e., $R_{i,k}^s(\mathbf{n}) = n_i^s r_i^s(\mathbf{n})$. Given the payoff function $u_k^s(\mathbf{n})$ and either BNN or replicator dynamics, the equilibrium system state **n** can be obtained by the standard method of solving differential equations.

It is observed that with the cost-price mechanism $\mathscr{C}_k^s(\mathbf{n})$, the revenue obtained in a cell by access points is identical to the total throughput. Also, the system throughput $R(\mathbf{n}) = \sum_{s=1}^{S} \sum_{i=1}^{K} n_i^s r_i^s(\mathbf{n})$ is the potential function [71]. Therefore, the stationary points of the dynamics are asymptotically stable, and also maximize the throughput. In addition, it is proved that the stationary point selected by replicator dynamics will be a Wardrop equilibrium. For BNN dynamics, the stationary point is also a Wardrop equilibrium.

10.4.2 Price of anarchy

In general, a Wardrop equilibrium can be inefficient (e.g., payoff is not maximized). This is referred to as the *price of anarchy*. However, in the population game with a multihoming protocol, the stationary point denoted by \mathbf{n}^* is efficient. With either replicator or BNN dynamics, the stationary point satisfies the condition $\dot{n}_k^s = 0$, which indicates that either $u_k^s(\mathbf{n}^*) = \frac{1}{\partial_k} \sum_{i=1}^{S_k} n_k^{i*} u_k^i(\mathbf{n}^*)$ or $n_k^{s*} = 0$. These conditions indicate that the users in class *k* will receive identical payoffs from all access points if the equilibrium strategy is applied. This stationary point is also the solution of the constrained optimization problem defined as follows:

$$\max \sum_{s=1}^{S} \sum_{i=1}^{K} n_{i}^{s} r_{i}^{s}(\mathbf{n}), \qquad (10.31)$$

s.t.
$$\sum_{j=1}^{S_i} n_i^j = \hat{d}_i, \quad i \in \{1, \dots, K\}.$$
 (10.32)

In particular, the stationary point maximizes the total system throughput. Therefore, anarchy has no price in this multihoming protocol.

10.4.3 Access pricing

The pricing of wireless access with a multihoming protocol is considered here. All access points are assumed to be owned by the same ISP. The actual mass of users in the system is defined as a function of the price p_s for access point *s* (i.e., there are different prices for different access points). The prices will not only determine the total number of users, but will also partition the subgroup of users selecting different access points. Users in class *k* will select access point *s* if $p_s \leq \Lambda_k$, where Λ_k is a threshold value [443]. The payoff per unit mass of users becomes

$$u_k^s(\mathbf{n}) = r_k^s(\mathbf{n}) - \mathscr{C}_k^s(\mathbf{n}) - p_s.$$
(10.33)

The objective of the ISP is to maximize its profit, which is total revenue minus cost. The cost is assumed to be identical to the actual throughput. The profit function of the ISP

can be expressed as

$$F_{\rm mul}(\mathbf{p}) = \sum_{s=1}^{S} p_s \sum_{i=1}^{K} n_i^s, \qquad (10.34)$$

where $\mathbf{p} = [p_1 \cdots p_s \cdots p_{S_k}]$ is a vector of prices. This profit function is evaluated with and without a multihoming protocol. Without a multihoming protocol, a user in class *k* selects the access point with the lowest price $p_{\min(k)}$. This is referred to as unihoming. In this case, the total profit is

$$F_{\text{uni}}(\mathbf{p}) = \sum_{i=1}^{K} p_{\min(i)} \hat{d}_i, \qquad (10.35)$$

and $\hat{d}_i = 0$ if $p_{\min(i)} > \Lambda_i$. It can be shown that the profit of the ISP from the system with a multihoming protocol is higher than that with unihoming, as follows. The profit with a multihoming protocol is $F_{mul}(\mathbf{p}) = \sum_{s=1}^{S} p_s \sum_{i=1}^{K} n_i^s$. It can be shown that

$$\sum_{s=1}^{S} p_s \sum_{i=1}^{K} n_i^s \ge \sum_{s=1}^{S} \sum_{i=1}^{K} p_{\min(i)} n_i^s = \sum_{i=1}^{K} p_{\min(i)} \sum_{s=1}^{S} n_i^s$$
(10.36)

$$=\sum_{i=1}^{K} p_{\min(i)} \hat{d}_i = F_{\mathrm{uni}}(\mathbf{p}).$$
(10.37)

In addition, the throughput with multihoming is equal to or higher than that with unihoming. To achieve maximum profit, there are multiple solutions for price **p**. The pricing solution p_s^* that achieves the highest throughput can be selected.

10.4.4 Related work

User churning behavior in wireless networks can be modeled using an evolutionarygame framework [170]. The stochastic evolutionary-game model based on the Markov chain is studied in [366]. Mobility arising from users' arrival in and departure from the service area is also considered. Competitive and cooperative pricing schemes are proposed whose solutions are the Nash equilibrium and the optimal solution, respectively. While the Nash equilibrium maximizes individual profit given the price offered by the other service provider, the optimal solution maximizes the total profit of all service providers. Similarly, in [367], an evolutionary game is applied to the problem of network selection in a heterogeneous wireless network, where different options (i.e., WLAN, cellular network, and broadband wireless access) are available to users. A reinforcement learning algorithm for network selection is introduced to achieve evolutionary equilibrium. A user association game is proposed to analyze user behavior in choosing an access point [138]. The Nash equilibrium is identified in this association game, with an iterative algorithm being used WLAN users to converge to the Nash equilibrium. To improve the efficiency of the Nash equilibrium, different social cost schemes are introduced in [138], and the prices of anarchy are analyzed for these schemes. A usernetwork association game for IEEE 802.11-based WLAN and 3G UMTS hybrid cell can be found in [262].

10.5 Admission control

Admission control is important for wireless networks, since it can be used to avoid network congestion. In particular, admission control limits the number of ongoing users in the network so that performance can be maintained above the requirement or the users' satisfaction can be maximized. In [268], non-cooperative admission control for IEEE 802.11-based WLAN is proposed. The system model considers differentiated services [176] based on the enhanced distributed coordination function (EDCF). There are *K* classes in total, and the number of ongoing users in class *k* is denoted by n_k . The rate of data transmission in class *k* is denoted by r_k . The user connects to a particular access point owned by one service provider. If the network is not congested (i.e., it is in an under-loaded condition), admitting new users will not affect performance for ongoing users. If the network is congested and a new user is admitted, the performance for ongoing users will be degraded, and the access point may lose revenue since some ongoing users could leave the system. Because of this tradeoff situation, the service provider must have a proper admission-control policy to maximize its revenue. The new user must also make a decision, so that satisfaction is maximized.

10.5.1 Two-player game formulation

The non-zero-sum game with two players is formulated in [268] for admission control in WLAN. The players are the access point (i.e., the service provider) and the new user. The strategies of the access point are to accept or to reject the request from a new user to connect to the access point. The strategies of the new user are to stay or to leave the current access point. The payoff for the access point is the revenue from new and ongoing users. The payoff for the user is the satisfaction. The payoff matrix for the access point is

$$\begin{array}{c|cccc}
 Leave & Stay \\
\hline Admit & R + (R_k - O) - O_k & R + (R_k - O) \\
\hline Reject & R - O_k & R
\end{array}$$
(10.38)

where the access point and the new user take row-wise and column-wise strategies, respectively, and

- *R* is the total revenue of the access point from ongoing users.
- R_k is the revenue gained from admitting a new user in class k.
- O_k is the revenue loss if a user from class k leaves the access point.
- O is the potential revenue loss from all classes at the access point.

The payoff matrix for the new user is

LeaveStayAdmit
$$U_k - Y_k$$
 $U_k - D(B_k)$ Reject $V_k - Y_k$ $V_k - D(B_k)$

where

- U_k is the satisfaction if the new user is admitted.
- V_k is the dissatisfaction if the new user is rejected, for $U_k > V_k$.
- Y_k is the dissatisfaction if a user in class k leaves the access point.
- $D(B_k)$ is the QoS dissatisfaction ratio for a user in class k.
- B_k is the bandwidth violation ratio,

$$B_{k} = \begin{cases} \frac{r_{k} - \rho_{k}/n_{k}}{r_{k}}, & \text{if } \rho_{k}/n_{k} \le r_{k}, \\ 0, & \text{if } \rho_{k}/n_{k} > r_{k}, \end{cases}$$
(10.40)

where r_k is the required rate, ρ_k is the effective bandwidth for class k [267], and ρ_k/n_k is the effective bandwidth per user. The QoS dissatisfaction ratio $D(B_k)$ indicates the inclination for a user in class k to leave the current access point because of the bandwidth violation ratio B_k . This QoS dissatisfaction ratio is $D(B_k) = 1 - S(B_k)$, where $S(B_k)$ is the QoS satisfaction ratio for the received QoS performance, i.e., $0 \le S(B_k) \le 1$. This QoS satisfaction ratio is approximated by a sigmoid function, defined as

$$S(B_k) = \frac{1}{1 + e^{-\alpha_k(\beta_k - B_k)}},$$
(10.41)

where α_k and β_k are constants to control, respectively, the shape and location of the curve for users in class k. An example of this sigmoid function is shown in Fig. 10.3. In this case, the revenue loss due to the admission of user in class k is $O = \sum_{k=1}^{K} n_k D(B_k) O_k$.



Fig. 10.3 Example of a sigmoid function.

10.5.2 Interpretation of payoff

From the payoff matrix in (10.38), if the access point admits a new user but a new user decides to leave the system (i.e., row 1 and column 1), the payoff for the access point is the revenue gained from ongoing and new users minus the possible loss of revenue from ongoing and new users leaving the system. If the access point admits a new user and the new user decides to stay (i.e., row 1 and column 2), the payoff for the access point is the revenue gained from ongoing and new users minus only the loss from ongoing users leaving the system. If the access point rejects a new user and the new user decides to leave (i.e., row 2 and column 1), the payoff for the access point is the remaining revenue from ongoing users minus the loss of revenue from the new user. Finally, if the access point rejects a new user but the new user decides to stay (i.e., row 2 and column 2), the payoff is only the revenue from ongoing users. The payoff in (10.39) for the new user can be derived in a similar way. However, in this case, if the new user decides to stay, the dissatisfaction $D(B_k)$ has to be taken into account.

With the above non-cooperative game formulation, there is at least one Nash equilibrium for the admission-control game played between the access point and the new user. The proof is based on the domination of strategy s_i or s_j . A strategy s_i is said to dominate strategy s_j if s_i yields a higher payoff regardless of the nature of s_j . There are three cases in the proof of the existence of Nash equilibria: $R_k > O$, $R_k = O$, and $R_k < O$. For $R_k > O$, the admitting strategy of the access point dominates the rejecting strategy, so the access point will always admit a new user. In this case, a new user will decide to stay if $U_k - Y_k > U_k - D(B_k)$, i.e., the dissatisfaction due to leaving the system is lower. For $R_k = O$, all combinations of strategies of the access point and the new user are possible. The solution depends on the specific values of the parameters. For $R_k < O$, the rejecting strategy dominates the admitting strategy, so the access point will always reject a new user.

In [268] a simulation was performed to evaluate the effectiveness of the proposed noncooperative admission control. In this case, the network which was nearly overloaded was emphasized. It was shown that admission control can successfully avoid congestion of the network. In addition, the impact of the bandwidth violation ratio was extensively investigated.

10.6 WiFi access-point pricing

Since IEEE 802.11-based WLAN has become common for commercial Internet access (e.g., in airports and libraries), the pricing of WLAN connection has become an important issue. Two major pricing models adopted by WiFi-based WLAN service providers are the direct-payment and aggregator models. In the direct-payment model, the user pays a price directly to the service provider for the WLAN connection. In the aggregator model, a broker (i.e., the aggregator) rents WLAN access points from service providers. The user subscribes to the aggregator. The aggregator is responsible for providing the user with access to all rented access points, and for billing the users. Money paid by the user

to the aggregator is distributed to the WLAN service providers according to the actual usage by the user. Naturally, the aggregator model is not preferred by either the service provider or the user. Either the user has to pay a higher price or the service provider receives lower revenue, since some money has to be given to the aggregator. However, the aggregator model has remained popular because of the absence of an effective pricing scheme in the direct-payment model. In particular, in the direct-payment model the user may not trust the service provider, and vice versa. For example, if a prepaid scheme is used (i.e., the user pays the service provider before connecting and transferring data at the access point), the service provider may cheat by not delivering the contracted service to the user. On the other hand, if a post-pay scheme is used (i.e., the user pays after connecting and transferring data at the access point), the user may cheat by not paying the service provider.

10.6.1 Pricing scheme for direct payment

In [344], a dynamic-game model is used to find the equilibrium strategy for the user and the service provider such that direct payment is viable. In this game, the connection to the access point and payment by the user are divided into multiple periods (e.g., with a length of one minute). When the user connects to the access point, a corresponding payment will be made at the end of the time slot. This technique of *micropayment* [335] is adopted to minimize transaction overhead. In the system model under consideration, a single access point and one user are considered, and are the players (Fig. 10.4).

After the user connects to the access point, the service provider proposes an access price p_t for the period t. This price is the strategy of the service provider. The user's strategy is to either accept the price and connect to the access point or to reject it and stay disconnected. If the user decides to reject, the game ends. The user's utility is $\mathscr{U}(T,\psi)$, where T is the actual number of periods of connection and ψ is the intended connection duration (i.e., the maximum number of periods of connection to the access point). In this utility function, T is a decision variable, which can be determined from the decision of the user to remain connected or to disconnect in the current period. ψ is random, which determines the *type* of the user. Again, the user knows the exact value of ψ at the beginning of the game. However, only the probability distribution of ψ is known by the service provider. Given the utility function $\mathscr{U}(T,\psi)$, the payoff for the user is $U_{user} = \mathscr{U}(T,\psi) - \sum_{t=1}^{T} p_t$. The payoff for the service provider is the





total revenue obtained from the user in all periods, i.e., $\sum_{t=1}^{T} p_t$. In this incompleteinformation environment, a perfect Bayesian equilibrium (PBE) is the solution of this dynamic game with a finite time horizon. As in the general case, the players have perfect recall of all strategies used.

10.6.2 User with Web browsing

First, a dynamic game for a user with a Web browsing utility function is considered. With Web browsing, the user's utility is a function of the connection duration T. The utility function is defined as

$$\mathscr{U}(T,\psi) = u\min(T,\psi), \qquad (10.42)$$

where *u* is the utility per period, which is random in each game. Therefore, this utility *u* is also the *type* of the user. Again, the user knows the exact value of *u* at the beginning of the game, but the service provider knows only its distribution. Assuming that *u* and ψ are independent, the perfect Bayesian equilibrium (PBE) can be described as follows:

- The user connects to the access point in period *t* if and only if *t* ≤ ψ and *u* > *p_t*. This is referred to as the *myopic strategy*.
- The service provider charges a non-decreasing sequence of prices such that pt ∈ arg maxp Pr(u > p), where Pr(u > p) is the probability of u > p.

For the service provider, the strategy selection depends only on the probability distribution of *u*. It is found in [344] that the PBE for the service provider is to choose a single value of p^* for the entire sequence of prices, i.e., $p_t = p^*$ for all *t*. (This result is counter-intuitive, since if the service provider observes that the user accepts price p_{t-1} , the better strategy would presumably be to change $p_t > p_{t-1}$.)

This model of two players is also extended in [344] to three players with a reseller (Fig. 10.5). The reseller is introduced, for example, to extend the service area to the user. In this case, the user requests connection to the reseller. The reseller then contacts the root access point, and the access point replies with the price c_t to be charged to the



reseller. The reseller can accept or reject this price c_t . If the reseller accepts the price c_t , the charging price p_t will be calculated and then proposed to the user. The user decides to accept or reject the price. Again, the connection duration of the user is random, with only the user knowing its exact value, while the reseller and the access point know only the probability distribution. The user's payoff is again the utility minus the cost: $\mathscr{U}(T, \psi) - \sum_{t=1}^{T} p_t$, where *T* is the number of periods of connection by the user. The reseller's payoff is the difference between the revenue gained from the user and the cost paid to the access point, i.e., $\sum_{t=1}^{T} (p_t - c_t)$. The access point's payoff is simply the revenue gained from the reseller, i.e., $\sum_{t=1}^{T} c_t$. Let the reseller's history be defined as follows: $\mathbf{h}_t = \{(c_{\hat{t}}, p_{\tilde{t}}); \hat{t} = \{1, ..., t\}, \tilde{t} = \{1, ..., t - 1\}\}$. The strategy of the access point is c_t , the strategy of the reseller is $p_t(\mathbf{h}_t)$, and the strategy of the user is to accept or reject the offered price p_t . With this three-player game, the PBE can be described as follows:

- The user adopts a myopic strategy, i.e., connects to the reseller if and only if *u* > *p*_t for *t* ≤ ψ.
- The reseller chooses price $p^*(c)$ such that $p^*(c) \in \arg \max_p (p-c) \Pr(u > p)$ and $p^*(c') \ge p^*(c)$ for c' > c.
- The access point chooses the non-decreasing price sequence $c_t \in \arg \max_c (c \Pr(u > p^*(c)))$.

In this case, the reseller's strategy depends only on the current price c_t proposed by the access point.

Note that this three-player model is similar to that of a Stackelberg game. In particular, the follower (i.e., the reseller) changes strategy p_t according to the strategy c_t of the leader (i.e., the access point). Also, the user changes strategy, accepting or rejecting the proposed price according to p_t . With this knowledge, the leader can seek the strategy c_t that maximizes its payoff. In this case, at PBE the access point will receive a share of the revenue from the user which is not less than that for the reseller [344].

10.6.3 User with file transfer

Here, a game model for a user with file transfer is considered. Without a reseller, the user connects to the access point until the file is completely downloaded. The utility function for the user is

$$\mathscr{U}(T,\psi) = \begin{cases} 0, & \text{if } T < \psi, \\ u\psi, & \text{if } T = \psi, \end{cases}$$
(10.43)

for $T \le \psi$. Again, the type of user is determined by u and ψ , where u is the utility per unit of file length. $u \in [I, h]$ with $0 \le I \le h$, where I and h are, respectively, the lower and upper bounds of utility per unit of file length, and $\psi \in \{1, ..., \psi_{max}\}$, where ψ_{max} is the maximum file length. The PBE can be described as follows:

• The user accepts price $p_t = 0$ for $t < \psi$. When $t = \psi$, the user accepts the price if in all previous periods the user connects to the reseller and $u\psi > p_t$. This is referred to as the *pessimistic strategy* for the user.
• The service provider charges

$$\rho_t = \begin{cases} 0, & \text{if } t < t^*, \\ u^* t^*, & \text{otherwise,} \end{cases}$$
(10.44)

where $(u^*, t^*) \in \arg \max_{(u,t)} ut \Pr(u > \tilde{u}, \psi = t)$, and \tilde{u} is constant.

10.6.4 Model for uncertain application

Here, a dynamic game is developed for the case that the service provider does not know whether the user uses Web browsing or file transfer. The service provider knows only the probability x for the user to use Web browsing and 1 - x to use file transfer. This model is referred to as the Bayesian model, which considers two periods of the game (i.e., $\psi = 2$). Three observations can be made about this special game structure:

For x ∈ [0, 0.516], the strategies s^{*}_{Se} and s^{*}_{Us} of the service provider and the user are PBE. The strategy of the service provider, s^{*}_{Se} = (p^{*}₁, p^{*}₂), is the price charged in periods 1 and 2, which can be obtained as follows [344]:

$$p_1^*(x) = \frac{4-5x}{2(1-x)(4-x)}, \quad p_2^*(x) = \frac{4-3x}{2(1-x)(4-x)}.$$
 (10.45)

The strategy s_{Us}^* of the user depends on the usage. If the user uses Web browsing (i.e., s_{Us}^* (web)), then the user in period 1 if and only if $p_1 < u$, and in period 2 if and only if the user decides to connect in period 1 and $p_2 < u$ (i.e., myopic strategy). If the user uses file transfer (i.e., s_{Us}^* (file)), then the user will connect in period 1 if and only if $p_1 + \hat{p}_2 < 2u$ for $\hat{p}_2 = p_2^*$, and in period 2 if and only if the user connects in period 1 and $p_2 < 2u$.

- For x ∈ [0.382, 1], strategies s⁺_{Se} and s⁺_{Us} of the service provider and the user are PBE. For the service provider, s⁺_{Se} = (0, 1/(2 − x)). For the user with Web browsing, a myopic strategy is applied, i.e., s⁺_{Us}(web) = s^{*}_{Us}(web). For the user with file transfer, the pessimistic strategy is applied. In particular, the user connects in period 1 if and only if p₁ = 0, and in period 2 if and only if the user connects in period 1 and 2u > p₂.
- For x > 0, there is no PBE if the service provider charges a constant price $p_1 = p_2$.

Perfect Bayesian equilibria for the above observations are shown in Fig. 10.6. As observed, this game possesses multiple equilibria for $x \in [0.382, 0.516]$. Note that this range [0.382, 0.516] is obtained from the numerical analysis in [344].

In summary, when the game between the service provider (i.e., the access point's owner) and the user is played for a finite number of periods, a PBE can be obtained for the direct-payment model. The PBE depends on the utility function of the user. For Web browsing, it is found in [344] that the service provider can charge a constant price to the user. If the price p_t is smaller than the utility per period of user u, then the user will accept the price. On the other hand, for a file-transfer utility function, the service provider charges a zero price in the initial periods and a constant price in the later period. It is noted that the Web-browsing utility function is more practical for actual use by the user.



Fig. 10.6 Perfect Bayesian equilibria given probability x of user to use file transfer.

10.7 Summary

In IEEE 802.11-based WLAN, it is typical for users to be rational, seeking to achieve the highest benefit from data transmission. Various game-theoretic models have been developed the behavior of the users and service providers WLAN. First, as IEEE 802.11based WLAN uses the CSMA/CA protocol for medium access control, the user can modify the protocol (e.g., change the contention window) to gain higher throughput. However, this modification will impact the performance of other users. Therefore, a game model for rational users was developed to analyze network performance. Access-point or network selection by users was formulated as the game. Since the users are rational, they will connect to the access point that yields the highest benefit (e.g., throughput minus cost). Equilibrium of access-point selection determines the average number of users in each network and, hence, the performance can be obtained from the game model. WLAN with a service provider as the owner was then considered. While users pay for connection to the WLAN access point, the service provider can decide to admit the user for service or not. If there are many ongoing users in the network, performance will be degraded. A game model between the new user and the service provider can be established to obtain an equilibrium admission-control policy. To this end, a game model for the pricing scheme for WLAN service was presented. The equilibrium price for a service provider to charge users can be determined for different applications (e.g., Web browsing or file transfer).

Recent advances in wireless communication have made possible the large-scale deployment of wireless networks, which consist of small, low-cost nodes with simple processing and networking capabilities. In order to reach the desired destination such as the data sink, transmissions depending on multiple hops are necessary. As a result, the optimization of routing is a critical problem that involves many aspects such as link quality, energy efficiency, and security. Moreover, the nodes may not be willing to fully cooperate. For example, from the node's perspective, forwarding the arriving packets consumes its limited battery power, so it may not be in the node's interest to forward all arriving packets. But doing so will adversely affect network connectivity. Hence, it is crucial to design a distributed-control mechanism that encourages cooperation among participating multi-hop nodes.

This chapter studies game-theoretic approaches to routing in multi-hop networks. We first introduce important models and examples of routing games. We provide two detailed examples, a repeated-routing game and a hierarchical-routing game for enforcing cooperation. Finally, we list other approaches from the literature.

11.1 Routing-game basics

A network is given by a directed graph G = (V, E), with vertex set V and edge set *E* (either directed or undirected). A set $\{(s_1, d_1), ..., (s_K, d_K)\}$ consists of sourcedestination vertex pairs, which we also call commodities. Each player is identified with one commodity. Different players can originate from different source vertices and pass information to different destination vertices. We use P_i to denote the $s_i - d_i$ paths of a multi-hop network. We consider only networks in which $P_i \neq \emptyset, \forall i$, and we define $\mathcal{P} = \bigcup_{i=1}^{K} P_i$. We allow the graph G to contain parallel edges, and a vertex can participate in multiple source–destination pairs.

Each edge e of a network has a cost function c_e . The cost functions are often assumed to be non-negative, continuous, and monotonous. All of these assumptions are reasonable in wireless applications. Notice that the cost of each edge is determined by both nature (such as the link quality) and players' actions. For example, the cost represents a quantity that increases with the network congestion (when players utilize the edge too much). The utility for a player U_i is usually formulated as the sum of the cost over the selected paths, and depends on the flows. In the routing game of a multi-hop network, the strategy s_i for each player is to choose the optimal flow (for example with the minimal overall cost). The flow is a non-negative vector indexed by the set \mathcal{P} of source–destination paths. There are two constraints on the flows and paths. First, a player can select multiple source–destination paths for transmission. The summation of flows over different paths is equal to the player's source–destination rate. Second, for each vertex, the sum of the input flows and the flow generated by this vertex is equal to output flows. This is similar to the Kirchhoff's circuit laws [111].

In such a routing game, several independent decision-makers (players) interact in order to form a network graph. Depending on the goals of each player, the final network flows result from individual players' decisions. Denote by $G_{s_i,s_{-i}}$ the graph *G* formed when player *i* plays a strategy s_i while all other nodes maintain their strategies $\mathbf{s}_{-i} = [s_1, ..., s_{i-1}, s_{i+1}, ..., s_K]$. To analyze the outcome of such a game, we can first define the best response for a player and then the Nash equilibrium, as follows:

DEFINITION 11.1 A strategy s_i^* is the best response for a player *i* if $U_i(G_{s_i^*,s_{-i}}) \ge U_i(G_{s_i,s_{-i}}), \forall s_i$. Therefore, the best response for player *i* is to make the selection of the flow that optimizes its utility, given that the other players maintain their strategies. The Nash equilibrium is the stable point at which no player can unilaterally improve its performance by changing its own strategy alone.

Next, we give an example of Braess' paradox [136] and its applications in wireless networks, to show how to formulate the game in a multi-hop network. Braess' paradox states that adding extra capacity to a network, when the moving entities selfishly choose their route, can in some cases reduce the overall performance. This is because the Nash equilibrium of such a system is not necessarily globally optimal. Formally, the paradox is stated as follows [136]:

For each point of a road network, let there be given the number of cars starting from it, and the destination of the cars. Under these conditions, one wishes to estimate the distribution of traffic flow. Whether one street is preferable to another depends not only on the quality of the road, but also on the density of the flow. If every driver takes the path that looks most favorable to him, the resultant running times need not be minimal. Furthermore, it is indicated by an example that an extension of the road network may cause a redistribution of the traffic that results in longer individual running times.

The reason for this is that, in a Nash equilibrium, drivers have no incentive to change their routes. If the system is not in a Nash equilibrium, selfish drivers must be able to improve their respective travel times by changing the routes they take. In the case of Braess' paradox, drivers will continue to switch until they reach the Nash equilibrium, despite the reduction in overall performance.

Now consider the network shown in Fig. 11.1 as an example, in which 4000 drivers wish to travel from Start to End. The travel time in minutes on the Start–A road is the number of travelers T divided by 100, and on Start–B it is a constant 45 minutes (similar to the other roads). If the dashed road does not exist (so the traffic network has four roads in total), the time needed to drive the Start–A–End route with A drivers would be



Fig. 11.1 Example of Braess' paradox.

 $\frac{A}{100}$ + 45, and the time needed to drive the Start–B–End route with B drivers would be $\frac{B}{100}$ + 45. If one route is shorter than the other, this would not be a Nash equilibrium because any rational driver would switch from the longer route to the shorter route. As there are 4000 drivers, the fact that A + B = 4000 can be used to solve that A = B = 2000 when the system is at equilibrium, and therefore each route takes $\frac{2000}{100} + 45 = 65$ minutes.

Next, we suppose that the dashed line is a road with a very small travel time, of approximately 0 minutes. In this situation, all drivers will choose the Start–A–B path, because Start–A–B will take 40 minutes at worst, while Start–B is always 45 minutes. Upon reaching A, every rational driver will elect to take the "free" road to B and continue to End, as A–End is always 45 minutes, while A–B–End is at worst 40 minutes. Each driver's travel time is $\frac{4000}{100} + \frac{4000}{100} = 80$ minutes, an increase from the 65 minutes required when the fast A-B road did not exist. No driver has an incentive to switch, as the two original routes (Start–A–End and Start–B–End) are both now 85 minutes. If every driver were to agree not to use the A–B path, all drivers would benefit, reducing their travel times by 15 minutes. However, because any single driver will always benefit by taking the A–B path, the socially optimal distribution is not stable, and so Braess' paradox occurs.

We now consider a wireless network scenario in which the Braess' paradox occurs. We consider a single-cell network with two access points (APs) to which a number of mobile users (terminals) are connected, as shown in Fig. 11.2. Assume that the first access point, denoted by AP_1 , offers a fixed rate r_F and that the rate offered by the second access point, denoted by AP_2 , offers a rate $r_V(n)$ that depends on the number of users connected to it. An example of such a system in practice would be a network that has two types of access points. The first type is an access point that uses an orthogonal, multiple-access scheme (e.g., TDMA) in which each user gets a *fixed rate*. The second type is an access point that offers a *variable rate* that depends on the number of users connected to it. An example of such an access point could be based on a CDMA system, in which the more users that are connected to the system, the greater is the interference. This leads to a lower signal-to-interference-and-noise ratio (SINR) and therefore to a lower rate. We can show that if we try to improve the system by allowing an *intersystem connection*, i.e., a connection between the two types of APs, it may eventually lead to a performance that is worse than in the original system. This happens because of the selfishness of the users, and it is similar to the original Braess' paradox that was studied in transport



Fig. 11.2 Single cell with two different access points, one offering a fixed rate r_F and the other offering a rate $r_V(n)$ that depends on the number *n* of users connected to it.

networks, in which adding a new fast road may not necessarily improve the throughput of the vehicles.

Finally, there are different ways to enforce cooperation in the wireless multi-hop network. In the rest of this section, we study some literature. Since the distributed nodes do not have exact information about the others, they act selfishly to optimize their own performances. This leads us to apply a game-theory approach [377] to the routing problem [403, 52]. In [274], repeated-game theory is applied to routing problems. In [182], the authors propose a repeated-game framework for multiple access using cartel maintenance. A Tit-for-Tat solution is proposed in [338] for multi-hop wireless networks. In [309], multiple-access resource allocation is studied using a game-theory approach. In [32], the authors consider a less aggressive punishment policy, in which the node uses the minimum forwarding probability among its neighborhoods as its forwarding probability after detecting the misbehavior. Felegyhazi et al. [147] consider a model to show cooperation among participating nodes and provide sufficient conditions on the network topology under which each node employing the punishment strategy results in a Nash equilibrium. Srinivasan et al. [464] provide a mathematical framework for cooperation in ad hoc networks, which focuses on the energy-efficient aspects of cooperation. In [337], the authors focus on the properties of the cooperation-enforcement mechanisms used to detect and prevent selfish behavior of nodes in an ad hoc network. They show that the formation of large coalitions of cooperating nodes is possible when a mechanism similar to CORE [336] is used. In [336] and [86], the authors define protocols that are based on a reputation system. In [30, 527], evolutionary dynamics and potential games are studied for non-cooperative routing. In [358, 359], a game-theoretic model is proposed for collaborative protocols in selfish, tariff-free, multi-hop wireless networks. A dynamic Bayesian game approach is studied in [373] for routing in wireless ad hoc networks.

11.2 Cooperation enforcement and learning using a repeated game

For multi-hop networks with packet forwarding, distributed control to enforce cooperation for nodes' packet-forwarding probabilities is essential for maintaining connectivity. In this section, we study a self-learning repeated-game framework [188] to optimize packet-forwarding probabilities of distributed users. The framework has two major steps. First, an adaptive repeated-game scheme ensures cooperation among users for the current cooperative packet-forwarding probabilities. Second, a self-learning scheme tries to find better cooperation probabilities. Some special cases are analyzed to evaluate the framework. From the simulation results, the framework demonstrates the near-optimal solutions in both symmetrical and asymmetrical networks.

11.2.1 System model and problem formulation

In multi-hop networks, packet forwarding is essential for distributed users to get connected to destinations. Suppose there are K users. The kth user has a total of N_k routes for its packet transmission. We assume that the routes are determined and known. Define I_k^i as the set of the nodes on the *i*th route for the kth user. Suppose each user is willing to forward other users' packets with a probability of α_i . For each user, the successful transmission or reception of one packet will have the benefit G, and forwarding others' packets will cost F per packet. Suppose the kth user transmits its packet with a probability of P_k^i to the *i*th route. Obviously, we have $\sum_{i=1}^{N_k} P_k^i = 1$. So the utility function U_k for the kth user is

$$U_k = \sum_{i=1}^{N_k} P_k^i G\Pi(\alpha_j, j \in I_k^i) - F\alpha_k B_k, \qquad (11.1)$$

where Π is the successful transmission probability, which is a function of the packetforwarding probabilities α_j along the routes. B_k is the forward-request probability from other users. The first term on the right-hand side of (11.1) is the average benefit for the *k*th user, which depends on other users' willingness to forward packets. The second term is the cost of forwarding other users' packets, which depends on the user's own willingness to do so.

We can formulate this problem as a non-cooperative game (Chapter 3) in which each user adjusts its forward probability to maximize its own utility function:

$$\max_{0 \le \alpha_k \le 1} U_k(\alpha_k, \alpha_{-k}), \tag{11.2}$$

where $\alpha_{-k} = [\alpha_1, ..., \alpha_{k-1}, \alpha_{k+1}, ..., \alpha_K]^T$ denotes the other users' packet-forwarding behaviors. At the Nash equilibrium every user will select a utility-maximizing strategy given the strategy of every other user.

Unfortunately, the Nash equilibrium for the packet-forwarding game is usually $\hat{\alpha}_k = 0$, $\forall k$, because each user's benefit depends on the other users' willingness to forward, and does not depend on its own behavior, while the user's cost depends solely on

its willingness to forward. So each user will greedily drop its packet-forwarding probability to reduce the cost and increase the utility. However, if all users do not forward, drop to successful packet-transmission probabilities might zero is, the benefit for users zero, and the system shuts down. That is, if the users play non-cooperatively and reach Nash equilibrium, all users' utilities might be zero, while if they cooperate and maintain a positive packet-forwarding probability, they can benefit.

The problem is to design a method of enforcing cooperation among users. We want to find the best packet-forwarding vector such that the utilities of all users are strictly better than those at the Nash equilibrium, design a mechanism to enforce such cooperation among users, and use the repeated game (Chapter 3) to solve the problem.

For a T-period *repeated game*, at each period *t* the moves of all players during periods 1, ..., t - 1 are known to every player. β is the discount factor. The total discounted payoff for each player is $\sum_{t=1}^{T} \beta^{t-1} U_k(t)$, where $U_k(t)$ denotes the payoff to player *k* in period *t*. If $T = \infty$, the game is referred to as an infinitely repeated game. The average payoff to player *k* is then

$$\bar{U}_{k} = (1 - \beta) \sum_{t=1}^{\infty} \beta^{t-1} U_{k}(t).$$
(11.3)

11.2.2 Self-learning cooperation-enforcing framework

The basic idea for the studied algorithm is to let distributed users learn the optimal packet transmission probability step by step, while within each step the strategy of a repeated game is applied to ensure cooperation among the users. For simplicity, we omit the user index. A block diagram of the algorithm is shown in Fig. 11.3.

During **initialization**, all users play a non-cooperative game, and all users are balanced in an inefficient Nash equilibrium $\hat{\alpha}$. We set the time counter at n = 0, the punishment time at T = 0, and the trigger threshold at $V = \hat{\alpha}$.

In the next step, we play a **repeated-game** strategy. If all users play cooperatively, every user will get a positive benefit. However, from (11.1), if any user deviates from cooperation by playing non-cooperatively and other users still play cooperatively, the non-cooperative user will have higher benefit, while the others will suffer with lower benefit because of this user's greediness. In order to prevent users from deviating, the repeated-game strategy provides a punishment mechanism. The basic idea is that each user checks whether the utility function is lower than the threshold *V*. If so, this means some user may have deviated; therefore, this user also chooses to play non-cooperatively, for a period *T*. In this case, the greedy user's short-term benefit is eliminated by the long-term punishment. If all users consider the long-term payoff to be (11.3), which is true under the assumption of rational users, then none of them will have an incentive to deviate from cooperation.

The repeated-game scheme with parameters (V, T) for all users is explained in detail as follows. Each user's utility U is compared with the threshold V. If U < V, i.e., someone has deviated, the time counter n is set to zero, the punishment time is increased by 1, and the user plays non-cooperatively for a period T. Since we assume that all



Fig. 11.3 Self-learning repeated-game framework.

users are rational, with increasing T the benefit of a one-time deviation will eventually be eliminated. Finally, no user wants to deviate, and $U \ge V$. At this point, the counter *n* starts increasing. If the system is stable in cooperation for a period N, a predefined constant, the algorithm assumes that cooperation has been enforced, and moves to the next step.

In the next step, the algorithm tries to **self-learn** the optimal forward probabilities by modifying α , with the goal of optimizing the performances. The simplest way is to randomly generate $\alpha \in [0, 1]$, where different users may have different α . In the next time slot, all users observe whether their performances become better. If they do not, then α is changed to the previous value. Otherwise, each user selects its packet-forwarding probability as α , updates its threshold to the current benefit V = U, calculates the change in utility

$$\Delta U = U(\text{new }\alpha) - U(\hat{\alpha}), \qquad (11.4)$$

and calculates the change ΔD in benefit. If the network is symmetric, the optimal punishment time is

$$T = \frac{\Delta D}{\Delta U},\tag{11.5}$$

where T is the estimated punishment time that prevents deviation. Then the algorithm goes back to the repeated-game case to update the punishment time T such that all users are willing to cooperate.

Notice that during the first time slot after α is modified, all users will act cooperatively, since deviation eliminates the chance of future utility improvement. In the repeated-game step, the benefit of instantaneous deviation is eliminated sooner or later, as long as the discount factor β is close enough to 1. Thus T will converge to some value. In the self-learning step, if the new α are not good for all users, the original value of α is restored. If the new α are good, cooperation can be enforced in a future repeated-game step. So the framework will converge.

In summary, the framework uses the threat of punishment to maintain cooperation for the current α , and tries to learn whether there is a better value of α for cooperation.

11.2.3 Asynchronous network

In the previous analysis, we assumed the networks were synchronous, i.e., each user's utility could be observed instantaneously whenever other users deviated. This might not be true in a real network. Here we discuss the problem introduced by asynchronous networks and some possible solutions.

When a network is asynchronous, deviations by users will be detected by other users with some time delay. The problem is that when the punishment period is over, the users may return to the cooperation phase at different times. This may trigger some users to continue punishment because they cannot distinguish between users who are deviating and those who are still in the punishment phase. This will make the network fluctuate, and the punishment time T cannot converge. In order to solve such a problem, we study a modification to the repeated-game step of the framework.

This is shown in Fig. 11.4, where an extra step is added at the end of the punishment period. After switching back to cooperation, the user will wait for time T' and then observe whether others deviate. This time T' is reserved for the other users to return to cooperation, and its value is determined by the scale and topology of the network. If the value of T' is too small, the network will not be stable, and the punishment period will always prevail because some users' delayed return to cooperation will trigger others' new punishment periods. If the value of T' is too large, it gives opportunities for greedy users to deviate to gain benefits without being detected by other users. Hence, there are tradeoffs in the choice of T'.

The other concern occurs during the step when α is modified after the system is stable, n > N. The message for all users to modify α can be implemented by protocols such as



flooding. This message will take time to arrive at each node. In order to check whether the system becomes better, a user needs to wait for a period of time that may be similar to T'.

11.2.4 Case analysis and performance evaluations

We analyze two cases: symmetrical networks and asymmetrical networks. Some simple examples are given, and analytical optimal results are deduced. Simulations are conducted to evaluate performance.

First, we analyze the characteristics of a symmetric network. The topology of such a network is symmetric, so the resulting Nash equilibrium and the optimum of the packet-forwarding probabilities should be the same for all users, i.e., $\hat{\alpha}_k = \hat{\alpha}_j$, $\alpha_k = \alpha_j$, $\forall k, j$. In general, networks are asymmetric. However, at the edges of networks, where some nodes may equally access the networks, a symmetrical topology may exist, and symmetric analysis can be applied.

As an example of a synchronous symmetrical network, consider the network in Fig. 11.5. In this network, there are six fixed routes: $1 \rightleftharpoons 4$, $2 \rightleftharpoons 5$, and $3 \rightleftharpoons 6$. All destinations are three hops away from the source, and we consider the node's utility function as the reward obtained from successfully transmitting or receiving a packet. We assume that forwarding of others' packets consumes resources such as energy, and therefore contributes a cost (negative reward) to the utility function. The utility functions for each of the nodes in Fig. 11.5 are represented as follows:

$$\begin{split} &U_1 = 2G[1 - (1 - \alpha_2\alpha_3)(1 - \alpha_5\alpha_6)] - F[\alpha_1 + \alpha_1\alpha_2], \\ &U_2 = 2G[1 - (1 - \alpha_1\alpha_6)(1 - \alpha_3\alpha_4)] - F[\alpha_2 + \alpha_2\alpha_3], \\ &U_3 = 2G[1 - (1 - \alpha_1\alpha_2)(1 - \alpha_4\alpha_5)] - F[\alpha_3 + \alpha_3\alpha_4], \\ &U_4 = 2G[1 - (1 - \alpha_2\alpha_3)(1 - \alpha_5\alpha_6)] - F[\alpha_4 + \alpha_4\alpha_5], \\ &U_5 = 2G[1 - (1 - \alpha_1\alpha_6)(1 - \alpha_3\alpha_4)] - F[\alpha_5 + \alpha_5\alpha_6], \\ &U_6 = 2G[1 - (1 - \alpha_1\alpha_2)(1 - \alpha_4\alpha_5)] - F[\alpha_6 + \alpha_1\alpha_6], \end{split}$$



Fig. 11.5 Example of a symmetric network.

where α_i is the probability that node *i* is willing to forward others' packets, *G* is the reward for successfully transmitting and receiving a packet, and *F* is the cost of forwarding others' packets. We assume that nodes are greedy and rational but not malicious; that is, every node chooses its forwarding probability to maximize its own utility function. If we consider the Nash equilibrium obtained non-cooperatively from (11.2), it is in each node's best interest to select zero forwarding probability (i.e., $\alpha_k = 0$, $\forall k$) to minimize its forwarding cost in the utility function. However, the overall network becomes disconnected as all of the nodes act in a non-cooperative manner.

Note that, because of the symmetry of the network in Fig. 11.5, the optimal forwarding probability and the corresponding utility for each node will also be symmetric. We omit subscripts for simplicity. If we consider the *system-wide* optimal solution to maximize everybody's utility, we can formulate the problem as

$$\max_{0 \le \alpha \le 1} U = 2G(2\alpha^2 - \alpha^4) - F(\alpha + \alpha^2).$$
(11.6)

Differentiating the above equation,

$$\frac{\partial U}{\partial \alpha} = 8G(\alpha - \alpha^3) - F(1 + 2\alpha) = 0, \qquad (11.7)$$

$$\alpha^3 - \left(1 - \frac{F}{4G}\right)\alpha + \frac{F}{8G} = 0, \qquad (11.8)$$

$$0 \le \alpha \le 1. \tag{11.9}$$

The optimal forwarding probability in the symmetrical network can be obtained by solving (11.8). Figure 11.6 shows the effects of forwarding probability α on the utility function for different normalized forwarding costs, F/G. We also show the optimal forwarding probabilities for different cases. It is obvious that as the cost F for forwarding is smaller than the transmitting/receiving reward G, the optimal forwarding probability will approach unity and the corresponding utility will also be high. On the other hand, when F/G is large, each node has a lower incentive to forward the others' packets, and utility is low. This is reasonable because when the cost for forwarding is very large, it is better for the node to save the energy for its own transmission. The goal of the cooperation mechanism design is to establish an incentive for the nodes to avoid the non-cooperative solution and to achieve a system-wide optimal forwarding probability. It is also worth mentioning that not every positive packet-forwarding probability will generate a larger utility than the full non-cooperation case, in which the packet-forwarding probability is zero. For example, when F/G = 1, the utility is higher than zero only when $\alpha \ge 0.37$. Thus, in the self-learning step, if α is less than 0.37, the system will have a worse performance than with non-cooperation. As a result, the new α will be discarded and the original α will be restored.

In Fig. 11.7, we show simulation results for utility and packet-forwarding probability over time, with F/G = 1 and N = 200. Initially, $\alpha = 0$ because of non-cooperative



Fig. 11.6 Effect of forwarding probability on utility.



Fig. 11.7 Utility function and forwarding probability over time.

transmission. Then the system tries to find a better packet transmission rate. When it finds a better solution, all users adopt their α to that value. However, because the punishment period T is not adjusted to an optimal value, deviation can have benefits, so there exists a period during which the utility and α switch from cooperation to



Fig. 11.8 Example of an asymmetrical network.

non-cooperation. During this period, T is increased until everybody realizes that there is no benefit from deviation because of the long period of punishment. If the system is stable for time N, a new α is attempted, to check if performance can be improved. If so, the new value is adopted; otherwise the original value is restored. Thus the packetforwarding probability is adjusted until an optimal solution is found, and the learned utility function is a non-decreasing function over time. Notice that users are less reluctant to deviate when α is close to the optimal solution. This is because the benefit of deviation becomes smaller and users already have the estimated punishment time according to (11.5).

Practical networks in nature are generally asymmetric. We now turn our attention to the performance of the framework in asymmetrical networks. An example of a synchronous asymmetrical network is shown in Fig. 11.8. Nodes 1 and 6 act as sinks for information. The right-pointing arrows indicate flow directions for which node 6 is the sink. In this case, nodes 1 to 5 want to transmit to node 6. Similarly, the left-pointing arrows indicate flow directions for which node 2 to 6 want to transmit to node 1. We formulate the utility functions for nodes 1 to 6 as follows:

$$\begin{split} &U_1 = 2G\alpha_2\alpha_3\alpha_4\alpha_5, \\ &U_2 = G[1 + \alpha_3\alpha_4\alpha_5] - F[2\alpha_2 + \alpha_2\alpha_3 + \alpha_2\alpha_3\alpha_4 + \alpha_2\alpha_3\alpha_4\alpha_5], \\ &U_3 = G[\alpha_2 + \alpha_4\alpha_5] - F[2\alpha_3 + \alpha_2\alpha_3 + \alpha_3\alpha_4 + \alpha_3\alpha_4\alpha_5], \\ &U_4 = G[\alpha_5 + \alpha_2\alpha_3] - F[2\alpha_4 + \alpha_3\alpha_4 + \alpha_4\alpha_5 + \alpha_2\alpha_3\alpha_4], \\ &U_5 = G[1 + \alpha_2\alpha_3\alpha_4] - F[2\alpha_5 + \alpha_4\alpha_5 + \alpha_3\alpha_4\alpha_5 + \alpha_2\alpha_3\alpha_4\alpha_5], \\ &U_6 = 2G\alpha_2\alpha_3\alpha_4\alpha_5. \end{split}$$

It is obvious that the non-cooperative solution for each node is to use zero forwarding probability. Notice that, because of the symmetry in the network flow, nodes 2 and 5, and nodes 3 and 4, have the same forwarding probabilities, respectively. Moreover, node 1's utility and node 6's utility are totally dependent on the other nodes' packet-forwarding probabilities. So the optimization parameters are α_2 ($\alpha_2 = \alpha_5$) and α_3 ($\alpha_3 = \alpha_4$) only. Since nodes 2 and 4 have their own optimization goals, from a system-optimization point of view this is a multiple-objective optimization. To quantify the optimality, we need to use the concept of Pareto optimality (a Pareto-optimal outcome cannot be improved upon without hurting at least one node).

In Fig. 11.9, we show the Pareto-optimal region and the simulated results for the framework. The x-axis and y-axis are α_2 and α_3 , respectively. Here, the system tries



Fig. 11.9 Pareto-optimal region and the simulated results.

to find a new packet-forwarding probability over the course of 250 trials. Any point within the shaded area is Pareto-optimal. Most of the simulated points are within this region, with very few points being located outside (a result of the failure to find the optimal packet-forwarding probability within 250 trials). We can see that the framework is effective in finding the Pareto optimum for asymmetrical networks.

Overall, we have studied a self-learning repeated-game framework for packetforwarding networks. Cooperation among users for packet forwarding is obtained by threat of future punishment, while the optimal packet-forwarding probability for each user can be studied distributively. From the simulation results for symmetrical and asymmetrical networks, we can see that the framework can effectively find solutions close to the optimal solutions in a distributed way. This framework may impact the design of future communication networks such as wireless networks, and wired networks, ad hoc networks, sensor networks.

11.3 Hierarchical routing using a network-formation game

In this section, we study a game-theoretic approach to the distributed formation of the hierarchical network architecture connecting the nodes in the uplink of a wireless multi-hop network. Existing literature focuses on the performance assessment of hierarchical multi-hop networks given an *existing topology*. Here we investigate the problem of the formation of this topology among nodes seeking to send data in the uplink to a central base

station through multi-hop. We model the problem as a hierarchical network-formation game, dividing the network into different hierarchy levels whereby the nodes belonging to the same level engage in a non-cooperative Nash game for selecting their next hop [416]. For a solution, we study the hierarchical Nash equilibrium for a sequence of multi-stage Nash games, which can be found analytically by backward induction. To find this equilibrium, we study a distributed myopic-dynamics algorithm based on fictitious play, in which each node computes the mixed strategies that maximize its utility, which represents the probability of successful transmission over the multi-hop communication path in the presence of interference.

In the literature, hierarchical multi-hop network architectures have become an essential aspect of emerging communication networks. For instance, while cellular-based communication has been the leading architecture in the past decade, recent advances such as distributed multi-hop communication have imposed a hierarchical architecture on many next-generation wireless networks. In fact, hierarchical structures have become ubiquitous in broadband networks [11], cognitive-radio networks [368], wireless local area networks (WLANs) [7], cellular networks [300], and sensor networks. In this regard, several IEEE workgroups have included hierarchical architectures in recent standards. For example, the IEEE 802.16j mobile multi-hop relay (MMR) task group introduced a hierarchical tree architecture as the base architecture in the next-generation IEEE 802.16 WiMAX family of broadband networks [11]. Moreover, IEEE 802.11s has standardized tree-based routing in mesh-based WLANs [7]. In [238], a theoretical framework for a hierarchical routing game is proposed. In [124], given a wireless tree network, the authors propose a low-complexity cooperative protocol that improves the average throughput of multi-hop upstream transmissions. The authors in [296] study the optimal deployment (that maximizes the throughput) of a single relay station for two-hop transmission in a hierarchical IEEE 802.16 network. In [297], the performance of multi-hop relaying is studied when dual relaying is performed. The authors in [453] study the resource-allocation problem for a multi-hop hierarchical cognitive network in the presence of an existing hierarchical topology. Other aspects of hierarchical wireless networks, such as routing and the optimal deployment of nodes, are discussed in [300, 524, 370, 395].

11.3.1 System model and game formulation

Consider a network of *N* users (or nodes) transmitting data to a central base station (BS) in the uplink direction. Let \mathcal{N} denote the set of nodes in the network. To improve their performance, the nodes can transmit through other nodes, i.e., multi-hop transmission. Hence, the final architecture governing the network is a hierarchical architecture, whereby each node $i \in \mathcal{N}$ is connected to one or more nodes in \mathcal{N} . For this purpose, we separate the nodes into *I* hierarchical levels (or stages¹) according to some criterion, for example, geographical distance to the BS (other criteria may also be used along with the studied algorithm). For multiple access at every hop, we consider a CDMA-based

¹ We use the terms "level" and "stage" interchangeably.

transmission. Let $\mathcal{L} = \{1, 2, ..., l\}$ denote the set of levels in the network, which are ordered in such a way that the nodes at level h_1 are farther away from the BS than nodes at level h_2 when $h_1 < h_2, \forall h_1, h_2 \in \mathcal{L}$. The nodes belonging to a hierarchy level $h \in \mathcal{L}$ form a set $\mathcal{N}_h \subset \mathcal{N}$. The different $\mathcal{N}_h, h \in \mathcal{L}$ are mutually exclusive sets, i.e., $\mathcal{N}_i \cap \mathcal{N}_j = \emptyset$ for $i \neq j, i, j \in \mathcal{L}$, and $\bigcup_{k=1}^l \mathcal{N}_k = \mathcal{N}$. We denote by $n_h = |\mathcal{N}_h|$ the number of nodes at level h.

The objective is an algorithm that allows the nodes at each level to select their next hops, in a distributed manner. To model these interactions among the nodes seeking to form the uplink tree structure, network-formation games provide a suitable framework [228, 190]. In such games, several independent decision-makers interact to form a network graph $G = (\mathcal{N}, \mathcal{E})$, where \mathcal{N} is the set of nodes or vertices and \mathcal{E} is the set of directed edges. The essence of network formation is to find the best set of directed edges among all possible configurations. Instead of an exhaustive search, we aim at a network-formation scheme that allows distributed decision-making at each node. Depending on the goals of each node, a final network graph forms as a result of individual nodes' decisions. Thus, we model the uplink tree-formation problem as a network-formation game for finding the directed uplink edges through which the nodes can transmit to the BS. An illustration of the studied model is shown in Fig. 11.10 for a network of nine users divided into three levels.

For each level $h \in \mathcal{L}$, a strategic game in normal form is defined by $\Xi_h = \langle \mathcal{N}_h, (\mathcal{A}_{i,h})_{i \in \mathcal{N}_h}, (U_{i,h})_{i \in \mathcal{N}_h} \rangle$, where \mathcal{N}_h is the set of players (nodes) of the game, $\mathcal{A}_{i,h}$ is



the set of actions of a player $i \in \mathcal{N}_h$, and $U_{i,h}$ is the utility function of a player $i \in \mathcal{N}_h$. In the uplink network-formation game, the action space $\mathcal{A}_{i,h}$ of a node $i \in \mathcal{N}_h$ is given by $\mathcal{A}_{i,h} = \mathcal{N}_{h+1}$, so $|\mathcal{A}_{i,h}| = |\mathcal{N}_{h+1}| = n_{h+1}$. At the last stage, h = l, \mathcal{A}_l is simply the singleton set, comprising the BSs. We denote by $(a_i, m) \in \mathcal{A}_m$ the node at level m + 1chosen by $i \in \mathcal{N}_h$ at level h. If $i \in \mathcal{N}_h$ and m = h, we call the choice (a_i, m) made by node i direct. If m > h, we call the choice (a_i, m) indirect as the choice of node i at a higher level is made via other intermediate nodes. For example, when m = h + 1, the choice (a_i, m) is indirectly made by (a_i, h) , which is directly chosen by i at level h. For m < h, we assume that $(a_i, m) = i$, the node itself. With this notation, the average SINR of any node $k \in \mathcal{N}_m$, denoted by $(a_i, m - 1)$ (k is any node selected by node $i \in \mathcal{N}_h$ at level m), received at node (a_i, m) , the direct choice of node i, is given by [519, 478]

$$\Gamma_{(a_i,m-1)}^{(a_i,m)} = S \cdot \frac{P_{(a_i,m-1)} \cdot g_{(a_i,m-1)}^{(a_i,m)}}{I_{(a_i,m-1)}^{(a_i,m)}},$$
(11.10)

where $P_{(a_i,m-1)}$ is the transmit power of node $(a_i, m-1)$, *S* is the spreading factor, and $g_{(a_i,m-1)}^{(a_i,m)} = \kappa \cdot \left(D_{(a_i,m-1)}^{(a_i,m)}\right)^{-\mu}$ is the channel gain between the node at level *m* and its selected node at level m+1, with $D_{(a_i,m-1)}^{(a_i,m)}$ being the distance between the nodes $(a_i, m-1)$ and (a_i, m) , μ the path-loss exponent, and κ the path-loss constant. $I_{(a_i,m-1)}^{(a_i,m)}$, the intra-level interference (and noise) perceived by $(a_i, m-1)$ at (a_i, m) from nodes at level m+1 that are connected to (a_i, m) , is given by

$$I_{(a_i,m-1)}^{(a_i,m)} = \sigma_m^2 + \sum_{k \in \mathcal{N}_m^{(a_i,m)} \setminus \{(a_i,m-1)\}} P_k \cdot g_k^{(a_i,m)}.$$
(11.11)

Here, σ_m^2 is the Gaussian noise variance at level *m*, the summation represents the interference from the other nodes at level *m* that are connected to node (a_i, m) $(\mathcal{N}_m^{(a_i,m)})$ is the set of players at level *m* connected to (a_i, m) , P_k is the transmit power of node *k*, and $g_k^{(a_i,m)} = \kappa \cdot (D_k^{(a_i,m)})^{-\mu}$ is the channel gain between *k* and (a_i, m) . We define the following utility function $U_{i,h}$ of a node $i \in \mathcal{N}_h$ from the starting stage *h* to the final stage *l*:

$$U_{i,h} = \prod_{m=h}^{l} \Pr_{(a_i,m-1)}^{(a_i,m)},$$
(11.12)

where $\Pr_{(a_i,m-1)}^{(a_i,m)}$, the probability of the average received SINR at node (a_i, m) from node $(a_i, m-1)$ being larger than a target $\nu_{(a_i,m-1)}$ (desired by $(a_i, m-1)$), i.e., the probability

of successful transmission for a single hop, is given by the following approximation (in a Rayleigh fading channel) [391]:

$$\Pr_{(a_i,m-1)}^{(a_i,m)} = e^{-\frac{\nu_{(a_i,m-1)}}{r_{(a_i,m-1)}^{(a_i,m)}}},$$
(11.13)

where $\nu_{(a_i,m-1)}$ is the target received SINR at (a_i, m) , and $\Gamma_{(a_i,m-1)}^{(a_i,m)}$ is given by (11.10). Hence, the utility function $U_{i,h} : \prod_{m=h}^{l} \left(\prod_{j=1}^{n_m} \mathcal{A}_m \right) \to [0, 1]$ of player $i \in \mathcal{N}_h$ corresponds to the multi-hop probability of successful transmission from the starting stage h to the final stage l. Note that the utility for a player i depends on its own strategies as well as the strategies of its parent nodes in the tree. In a nutshell, the action of any node at level m is to choose a node at level m + 1, and the corresponding utility is dependent on the actions of the nodes at level m + 1.

Moreover, we consider that each node requires a certain amount of money to be paid for offering its services, i.e., we consider a pricing scheme in the network. Let $c_{(a_i,m)}$ be the cost that node $i \in \mathcal{N}_h$ has to pay per unit of traffic if *i* transmits through $(a_i, m) \in \mathcal{A}_m$. If m = h, the cost is *direct*; otherwise, it is said to be *indirect*. We consider only direct costs, i.e., $c_{(a_i,m)} = 0$ if $m \neq h$. When m = h, a cost is incurred if a connection is made. Hence, the nodes pay only to the nodes at the next level. Thus, the utility function $\overline{U}_{i,h}$ to transmit from *h* to *l* for unit traffic becomes

$$\overline{U}_{i,h} = \prod_{m=h}^{l} \frac{e^{-\nu_{(a_i,m-1)}/\Gamma_{(a_i,m-1)}^{(a_i,m)}}}{c_{(a_i,h)}},$$
(11.14)

which corresponds to the success rate per unit of money. By taking the natural logarithm of both sides in (11.14), we get

$$\widetilde{U}_{i,h} = -\sum_{m=h}^{l} \frac{\nu_{(a_i,m-1)}}{\Gamma_{(a_i,m-1)}^{(a_i,m)}} - \widetilde{c}_{(a_i,h)} = \sum_{m=h}^{l} \widetilde{u}_{i,m} - \widetilde{c}_{(a_i,h)},$$
(11.15)

where $\tilde{u}_{i,m}$ is the stage utility, defined by $\tilde{u}_{i,m} := -\frac{\nu_{(a_i,m-1)}}{\Gamma_{(a_i,m)}^{(a_i,m)}}$, and $\tilde{c}_{(a_i,h)} = \ln c_{(a_i,h)}$. Therefore, in the transformed game $\tilde{\Xi}_h = \langle \mathcal{N}_h, (A_{i,h})_{i \in \mathcal{N}_h}, (\tilde{U}_{i,h})_{i \in \mathcal{N}_h} \rangle$, each node *i* at the level *h* attempts to optimize

$$\max_{(a_i,h)\in\mathcal{A}_h}\widetilde{U}_{i,h}.$$
(11.16)

Since all nodes in \mathcal{N} attempt to choose an action to optimize their payoffs at the level where they belong, $\tilde{U}_{i,h}$, the utility for $i \in \mathcal{N}_h$ at level h, is dependent on $\tilde{U}_{j,h+1}$, the utility obtained by $j \in \mathcal{N}_{h+1}$, at the next level h + 1, which is an outcome of the game

 Ξ_{h+1} , in which each player *j* at level h+1 solves

$$\max_{\substack{(a_j,h+1)\in\mathcal{A}_{h+1}}}\widetilde{U}_{j,h+1}.$$
(11.17)

This dependence on the next level is not present at the last stage, $\widetilde{\Xi}_l$, where each node $k \in \mathcal{N}_l$ optimizes

$$\max_{(a_k,l)\in\mathcal{A}_l} \widetilde{U}_{k,l} = \max_{(a_k,l)\in\mathcal{A}_l} [\widetilde{u}_{k,l} - \widetilde{c}_{(a_k,l)}].$$
(11.18)

As an example of this formulation, for the network in Fig. 11.10 we have $\mathcal{N}_1 = \{1, 2, 3\}, \mathcal{N}_2 = \{4, 5, 6\}, \mathcal{N}_3 = \{7, 8, 9\}, \mathcal{A}_1 = \{4, 5, 6\}, \mathcal{A}_2 = \{7, 8, 9\}, \text{ and } \mathcal{A}_3 = \{BS\}.$ From the chosen connections, we observe that $(a_1, 1) = 4, (a_2, 1) = 6, (a_3, 1) = 6, (a_4, 2) = 7, (a_5, 2) = 8, (a_6, 2) = 8, \text{ and } (a_7, 3) = (a_8, 3) = (a_9, 3) = BS.$ The utility for node 1 at level 1 is given by $\widetilde{U}_{1,1} = \widetilde{u}_{1,1} + \widetilde{u}_{1,2} + \widetilde{u}_{1,3} - \widetilde{c}_4$ with each $\widetilde{u}_{i,m}$ given by its definition as previously mentioned.

11.3.2 Hierarchical network-formation game solution

To solve the game presented, the network-formation game is defined by a sequence of non-cooperative Nash games $\{\widetilde{\Xi}_h\}_{h=1,...,l}$. We denote by $(a_i, h)^*$ the optimal action chosen at level *h* by a player $i \in \mathcal{N}_h$ directly. Let $(\mathbf{a}, h) \in \prod_{k=1}^{n_h} \mathcal{N}_h$ be an action profile at level *h*, i.e., $(\mathbf{a}, h) = [(a_i, h)]_{i \in \mathcal{N}_h}$. For convenience, we denote $(\mathbf{a}_{-i}, h) = [(a_j, h)]_{j \neq i, j \in \mathcal{N}_h}$, hence $(\mathbf{a}, h) = [(a_i, h), (\mathbf{a}_{-i}, h)]$. Given a level *l'* between the initial level *h* and the final level *l*, we define the Nash equilibrium of the hierarchical network-formation game $\{\widetilde{\Xi}_h\}_{h=1,...,l}$ as follows:

DEFINITION 11.2 Let $\{\widetilde{\Xi}_h\}_{h=1,...,l}$ be a hierarchical network-formation game. An action profile $(a_i, h)^* \in \mathcal{A}_{h+1}$, for $i \in \mathcal{N}_h$ and $h \in \mathcal{L}$ is a hierarchical Nash equilibrium if

$$U_{i,h}((a_i, h)^*, (\mathbf{a}_{-i}, h)^*) \ge U_{i,h}((a_i, h), (\mathbf{a}_{-i}, h)^*), \forall i \in \mathcal{N}_h, (a_i, h) \in \mathcal{A}_{h+1}, h \in \mathcal{L}.$$
(11.19)

The action profile $(\mathbf{a}, h)^*$ is said to be a stage-h Nash equilibrium, and the payoff at the equilibrium $\widetilde{U}_{i,h}^*((\mathbf{a}, h)^*)$ is called the stage-h optimal payoff to player i.

The hierarchical Nash equilibrium is thus defined as a solution to the multi-stage non-cooperative game, in which each node plays only once and no node, at any level of hierarchy, can unilaterally deviate and improve its utility given that the computed strategies of the other nodes at all stages remain fixed. Note that in (11.19) we shorthand the dependence of the payoff of $\tilde{U}_{i,h}$ into the actions at its own level, as the the choices of actions made by a node at its own level directly determine the utility.

To find the Nash equilibrium defined in (11.19), the game at a particular stage l' can be decomposed as follows:

$$\begin{aligned}
\tilde{U}_{i,l'}^{*} &= \max_{(a_{i},l')\in\mathcal{A}_{l'}} \tilde{U}_{i,l'} \\
&= \max_{(a_{i},l')\in\mathcal{A}_{l'}} \left[\sum_{m=l'+1}^{l} \tilde{u}_{(a_{i},l'),m}^{*} + \frac{\nu_{(a_{i},l'-1)}}{\Gamma_{(a_{i},l'-1)}^{(a_{i},l')}} - \tilde{c}_{(a_{i},l')} \right] \\
&= \max_{(a_{i},l')\in\mathcal{A}_{l'}} \left[\tilde{U}_{(a_{i},l'),l'+1}^{*} + \frac{\nu_{(a_{i},l'-1)}}{\Gamma_{(a_{i},l'-1)}^{(a_{i},l')}} - \tilde{c}_{(a_{i},l')} \right],
\end{aligned}$$
(11.20)

for all $i \in \mathcal{N}_{l'}$, where $\widetilde{U}^*_{(a_i,l'),l'+1}$ is the stage l'+1 optimal payoff to (a_i, l') . The payoff of the game at the last stage is

$$\widetilde{U}_{i,l}^* = \frac{\nu_{i,l-1}}{\Gamma_{(a_i,l)}^{(a_i,l)}} - \widetilde{c}_{(a_i,l)^*}, \qquad (11.21)$$

where $\{(a_i, l)^*, i \in \mathcal{N}_l\}$ is a hierarchical Nash equilibrium at level *l*. Note that $\Gamma_{(a_i, l-1)}^{(a_i, l)}$ depends (through the interference term) on the actions of all the nodes at level *l* at the hierarchical Nash equilibrium, i.e., $((a_1, l)^*, (a_2, l)^*, \dots, (a_{n_l}, l)^*)$.

Hence, by (11.20), we can obtain the Nash equilibrium of the game by starting with the final stage l' = l and solving the game iteratively by backward induction to stage l' = h. Such a decomposition of the game into stages is possible because the players at each level are different from each other, and the game at level l' is independent of the games at levels k < l' (in contrast, the game at l' is dependent on the higher levels k > l' through the utility). In a nutshell, we can proceed as follows for solving the hierarchical network-formation game:

PROPOSITION 11.1 Consider a network-formation game $\{\widetilde{\Xi}_h\}_{h=1,...,l}$. An action profile $(a_i, h)^* \in \mathcal{A}_{h+1}$, for $i \in \mathcal{N}_h$ and $h \in \mathcal{L}$, is a hierarchical Nash equilibrium if and only if it solves (11.20) recursively from stage h = l backwards to stage h = l'.

Thus, Proposition 11.1 provides an analytical way to solve for the hierarchical Nash equilibrium using backward induction.

As the Nash equilibrium in pure strategies may not exist, we generalize the above concepts to the case in which the nodes at every level use mixed strategies, since the mixed Nash equilibrium *always* exists [58]. Let $\mathbf{p}^{i,l'} \in \mathcal{P}^{i,l'}$ be the mixed strategies of the *i*th player at stage l', where

$$\mathcal{P}^{i,l'} = \left\{ \mathbf{p}^{i,l'} \in \mathbb{R}^{n_{l'}} \middle| \sum_{(a_i,l') \in \mathcal{A}_{l'}} p^{i,l'}_{(a_i,l')} = 1, \quad p^{i,l'}_{(a_i,l')} \ge 0, \forall (a_i,l') \in \mathcal{A}_{l'} \right\}, i \in \mathcal{A}_{l'}.$$

DEFINITION 11.3 Let $\{\widetilde{\Xi}_h\}_{h=1,...,l}$ be a hierarchical network-formation game. A mixed-strategy profile $p_{(a_i,h)}^{i,h*} \in \mathcal{P}^{i,h}$, for $i \in \mathcal{N}_h$ and $h \in \mathcal{L}$, is a hierarchical Nash

equilibrium in mixed strategies if the following n_h conditions are satisfied:

$$\begin{split} \sum_{(a_{1},h)\in\mathcal{A}_{h}} \cdots \sum_{(a_{n_{h}},h)\in\mathcal{A}_{h}} p_{(a_{1},h)}^{1,h*} p_{(a_{2},h)}^{2,h*} \cdots p_{(a_{n_{h}},h)}^{n_{h},h*} \widetilde{U}_{i,h} \\ &\geq \sum_{(a_{1},h)\in\mathcal{A}_{h}} \cdots \sum_{(a_{n_{h}},h)\in\mathcal{A}_{h}} p_{(a_{1},h)}^{1,h} p_{(a_{2},h)}^{2,h*} \cdots p_{(a_{n_{h}},h)}^{n_{h},h*} \widetilde{U}_{i,h} \\ &\cdots \\ \sum_{(a_{1},h)\in\mathcal{A}_{h}} \cdots \sum_{(a_{n_{h}},h)\in\mathcal{A}_{h}} p_{(a_{1},h)}^{1,h*} p_{(a_{2},h)}^{2,h*} \cdots p_{(a_{n_{h}},h)}^{n_{h},h*} \widetilde{U}_{i,h} \\ &\geq \sum_{(a_{1},h)\in\mathcal{A}_{h}} \cdots \sum_{(a_{n_{h}},h)\in\mathcal{A}_{h}} p_{(a_{1},h)}^{1,h*} p_{(a_{2},h)}^{2,h*} \cdots p_{(a_{n_{h}},h)}^{n_{h},h*} \widetilde{U}_{i,h} \\ &\geq \sum_{(a_{1},h)\in\mathcal{A}_{h}} \cdots \sum_{(a_{n_{h}},h)\in\mathcal{A}_{h}} p_{(a_{1},h)}^{1,h*} p_{(a_{2},h)}^{2,h*} \cdots p_{(a_{n_{h}},h)}^{n_{h},h} \widetilde{U}_{i,h} \\ &\forall i \in \mathcal{N}_{h}, h \in \mathcal{L}. \end{split}$$

The mixed-strategy profile $\mathbf{p}^{i,h*}$ is said to be a mixed stage-h Nash equilibrium, and the payoff at the equilibrium, i.e.,

$$\widehat{U}_{i,h}^* = \sum_{(a_1,h)\in\mathcal{A}_h} \cdots \sum_{(a_{n_h},h)\in\mathcal{A}_h} p_{(a_1,h)}^{1,h*} p_{(a_2,h)}^{2,h*} \cdots p_{(a_{n_h},h)}^{n_h,h*} \widetilde{U}_{i,h}$$

is called the stage-h optimal payoff to player i.

We can use a decomposition similar to that in (11.20) to show that the mixed Nash equilibrium can also be found by backward induction from the final stage to the initial stage *h*.

11.3.3 Hierarchical network-formation algorithm

To find the hierarchical Nash equilibrium, we study a dynamics algorithm that allows a distributed formation of the hierarchical network structure. The dynamics assume that each node is myopic, in the sense that the nodes aim to improve their payoffs considering only the current and previous states of the network. The network-formation algorithm consists of three phases: hierarchy formation, hierarchical fictitious play, and data transmission. In the first phase, the hierarchy in the network is formed. We consider a distance-based hierarchy whereby each hierarchy level corresponds to the area between two circles centered at the BS and with constant radii, with the uppermost level corresponding to the area within a circle centered at the BS with a specified radius. This hierarchy divide can be performed by the BS, and is assumed to be fixed throughout the network operation. Once the hierarchy is formed, Phase I continues by allowing each player belonging to a certain level *l* is allowed to select its nearest neighbor in level *l*+1. Hence, the initial network consists of a nearest-neighbor algorithm with predetermined hierarchy levels.

Once the network is initiated, the next phase of the algorithm, the actual networkformation process, starts. To form the network in Phase II, we use *fictitious play* [159] at each stage $l' \in \mathcal{L}$ to find the mixed Nash equilibrium at that level. Let $p_{(a_i,l')}^{i,l'}(k)$ be the empirical probability that a player $i \in \mathcal{N}_{l'}$, $i = 1, 2, ..., n_{l'}$ at a certain level l' chooses an action $(a_i, l') \in \mathcal{A}_{l'}$ at the *k*th iteration of the algorithm. Denote by $\mathbf{p}_{l'}^i(k) = [p_{(a_1,l')}^{1,l'}(k), ..., p_{(a_{n',l'})}^{n',l'}(k)]$ an $n_{l'}$ -dimensional vector of player *i*'s empirical mixed strategy at time *k*. At each iteration, the players update their strategy $\mathbf{p}_{l'}^i(k)$ as follows:

$$\mathbf{p}_{l'}^{i}(k+1) = \mathbf{p}_{l'}^{i}(k) + \frac{1}{k+1} \left(\mathbf{v}_{l'}^{i}(k) - \mathbf{p}_{l'}^{i}(k) \right), \qquad (11.22)$$

where $\mathbf{v}_{l'}^{i}(k) = [v_{(a_{i},l')}^{i}(k)]_{(a_{i},l') \in \mathcal{A}_{l'}}$ is an $n_{l'}$ -dimensional vector with $v_{a_{i},l'}(k) = 1$ if at time *k* the *i*th player chooses the action (a_{i}, l') , and $v_{a_{i},l'}(k) = 0$ otherwise. Since a player chooses only one action at each step, $\mathbf{v}_{l'}^{i}(k)$ is a vector with the entry that corresponds to the chosen action (a_{i}, l') being 1, while the remaining terms are equal to 0.

In the hierarchical fictitious-play phase, the action (a_i, l') of the *i*th node at time *k* is the *best response* to the observed empirical strategies of the opponents. Let $\mathbf{p}_{l'}^{-i}(k) = [\mathbf{p}_{l'}^{j}(k)]_{j \in \mathcal{N}_{l'}, j \neq i}$; let $q_{i,l'}(k)$ denote the action taken by the *i*th node at time *k*, given by $q_{i,l'}(k) = \arg \max_{(a_i, l') \in \mathcal{A}_{l'}} g_i((a_i, l'), \mathbf{p}^{-i})$; and let $\mathbf{q}_{l'}(k) = [q_{i,l'}(k)]_{i \in \mathcal{N}_h}$, where

$$g_{i}((a_{i}, l'), \mathbf{p}_{l'}^{-i}) = \mathbb{E}_{\mathbf{p}^{-i}}(\widetilde{U}_{(a_{i}, l'), l'})$$

$$= \sum_{(a_{1}, l') \in \mathcal{A}_{l'}} \cdots \sum_{(a_{i-1}, l') \in \mathcal{A}_{l'}} \sum_{(a_{i+1}, l') \in \mathcal{A}_{l'}} \cdots \sum_{(a_{n'_{l}}, l') \in \mathcal{A}_{l'}} p_{a_{1}, l'}^{1, l'}, \dots, p_{a_{i-1}, l'}^{i-1, l'}, p_{a_{i+1}, l'}^{i+1, l'}, \dots, p_{a_{n'_{l}}, l'}^{n_{l'}, l'} \widetilde{U}_{(a_{i}, l'), l'}(k), \quad (11.23)$$

and $U_{(a_i,l'),l'}(k)$ is the payoff at step k, which is dependent on the payoff matrix $U_{(a_i,l'),l'+1}(k)$ at step k, i.e.,

$$\widetilde{U}_{(a_{i},l'),l'}(k) = \widetilde{U}_{(a_{i},l'),l'+1}(k) + \frac{\nu_{i,l'-1}}{\Gamma^{(a_{i},l')}_{(a_{i},l'-1)}(\mathbf{q}_{l'}(k))} - \widetilde{c}_{(q_{i,l'}(k))}$$

and at the terminal stage

$$\widetilde{U}_{i,l}(k) = \frac{\nu_{i,l}}{\Gamma^{(a_i,l)}_{(a_i,l-1)}(\mathbf{q}_l(k))} - \widetilde{c}_{(q_{i,l}(k))}.$$
(11.24)

The optimal action $q_{i,l'}(k)$ taken at iteration k determines the vector $\mathbf{v}_{l'}^i$ at the following iteration and, hence, allows us to update the empirical frequency. This iterative process continues until the empirical frequencies converge to the hierarchical Nash equilibrium. Note that it is well known that whenever fictitious play converges, it converges to a Nash equilibrium [159, 445]. Hence, in our model, by using fictitious play at every level, we ensure that our algorithm reaches the mixed Nash equilibrium at every level when it reaches a steady state (consequently, the network converges to a hierarchical Nash equilibrium). In general, fictitious-play algorithms have been proven to converge

Algorithm 11.1 Hierarchical network-formation algorithm.

Phase I: Hierarchy formation

- a) The network is divided into *l* hierarchy levels, e.g., based on the distance to the BS.
- b) Each node at a level l' selects the nearest neighbor in level l' + 1 (initial network state).

Phase II: Hierarchical fictitious play

The nodes engage in a hierarchical network-formation game.

repeat (*iteration* k)

- a) Each node $i \in \mathcal{N}$ selects its best response based on the payoffs and empirical probabilities of the opponents in iteration k 1, as in (11.23).
- b) Node *i* updates its mixed strategies based on (11.22).
- until convergence to the hierarchical Nash equilibrium.

Phase III: Data transmission

The hierarchical Nash architecture is formed, and each node at level l' transmits to the BS using the nodes (strategies) at level l' + 1 with different probabilities over time (mixed strategies).

in almost all cases, and many modification schemes have also been proposed to ensure convergence [159, 445].

Upon convergence of Phase II to a hierarchical Nash equilibrium, the nodes are ready to start their transmission, in the last phase of the algorithm. In this phase, the nodes have already computed their mixed strategies, so they choose their next hop based on the probabilities that resulted from Phase II. Note that although the nodes mix between different actions with different probabilities, at any given time the network is structured into a tree architecture rooted at the BS. The algorithm is summarized as Algorithm 11.1.

The network-formation Algorithm 11.1 can be implemented in a distributed fashion. For hierarchy formation, the BS can broadcast this information to all the nodes at the beginning of all time, so the remaining phases of the algorithm can be performed with no further reliance on the BS (since the hierarchy is fixed, and based on distance). For instance, for the hierarchical fictitious-play algorithm, at every time *k*, each node *i* at a level *l'* needs only to know the payoffs to its parent nodes, i.e., the nodes at level *l'' > l'* that link node *i* to the BS, from the previous time instant k - 1, as well as the empirical probabilities of the nodes by observing the past actions of their opponents, as well as the payoffs to the players at the next level, in a distributed manner, without relying on the BS. The last phase of the algorithm is simply a transmission phase, in which each node sends its data to the BS through multi-hop, and uses the mixed strategies resulting from hierarchical fictitious play.

11.3.4 Simulation results and analysis

For simulations, the following network is set up: the BS is placed at the origin of a $2 \text{ km} \times 2 \text{ km}$ square with the nodes randomly deployed in the area around the BS. We



Fig. 11.11 Snapshot of a tree topology formed using the algorithm with ten randomly deployed nodes in three hierarchy levels. For each node, the solid arrows show the strategies with largest probabilities, while the dashed arrows indicates the remaining strategies.

consider three levels of hierarchy, as follows. The first level consists of nodes randomly deployed in the area between two circles centered at the BS and with radii 0.6 km and 1 km; the second level consists of nodes randomly deployed in the area between two circles centered at the BS and with radii 0.3 km and 0.6 km; and the third level consists of nodes randomly deployed within the area of a circle centered at the BS with radius 0.3 km. The nodes' transmit power is set to $P_i = 10 \text{ mW}$, $\forall i \in \mathcal{N}$, the target SINR is set to $\nu_i = 10 \text{ dB}$, $\forall i \in \mathcal{N}$, the noise level is set to $\sigma^2 = -90 \text{ dBm}$ for all levels, the path-loss constant is set to $\mathcal{K} = 1$, while the path-loss exponent is set to $\mu = 3$. The spreading factor is set to S = 64, which is typical for data services in the uplink of CDMA networks [478]. The pricing parameter is set to $c_i = 1$, $\forall i \in \mathcal{N}$.

In Fig. 11.11, we randomly deploy N = 10 nodes within the BS area, with $n_1 = 4$ nodes in the first hierarchy level, and $n_2 = n_3 = 3$ nodes in the second and third hierarchy levels. The network-formation game starts with the nearest-neighbor tree, whereby, within the same level, each node is connected to the nearest node in the next level with a probability of 1. Figure 11.11 shows the hierarchical Nash network resulting from our algorithm, with the solid arrows indicating the strategies of each node that have the highest probability of selection, and the dashed arrows indicating the remaining strategies with non-zero probability. On the one hand, as a result of the network-formation game, nodes 3, 5, and 6 choose to connect to their nearest neighbors in the next hierarchy level (respectively, nodes 6, 10, and 9) with a probability of 1. On the other hand, nodes 1, 2, and 7 choose between two different strategies with different probabilities, while node 4 mixes between three strategies. For instance, nodes 1 and 2 choose their mixed strategies in such a way



Fig. 11.12 Convergence of the mixed strategies of node 4 from Fig. 11.11.

that a high probability is given for connecting to the nearest node, i.e., node 7, while a small probability is given to a connection with node 5. Furthermore, although node 4 is at an almost equal distance from both node 5 and node 6 (node 6 is slightly closer to node 4 than node 5), it chooses to connect to node 6 with a probability of 0.5565, which is much higher than the probabilities of 0.1658 and 0.2783 with which node 4 selects nodes 5 and 7, respectively. This choice by node 4 is justified by the fact that this node attempts to minimize the interference with nodes 1 and 2 that may occur at nodes 5, and 7. Figure 11.11 summarizes how the different nodes in a wireless network select mixed strategies and self-organize into a hierarchical Nash network.

Figure 11.12 provides an insight into the convergence time of the algorithm. In this figure, we show the probabilities of all three strategies of node 4 from Fig. 11.11 as the number of iterations increases. Node 4 starts by being connected with its nearest neighbor, node 6, with a probability of 1. As the network-formation game evolves, Node 4 adapts its mixed strategies by decreasing the probability of selecting node 6 and increasing the probabilities of selecting nodes 5 and 7. The mixed strategies finally converge toward probabilities of 0.1658, 0.5565, and 0.2783 for connecting to nodes 5, 6, and 7, respectively. The convergence is ensured at around 25 iterations; hence, the convergence time of the algorithm is quite reasonable.

Overall, we have studied a distributed hierarchical network-formation game approach for the construction of the uplink network topology in a wireless multi-hop network. For this purpose, the network is divided into a number of hierarchy levels, in which the nodes belonging to the same level play a non-cooperative Nash game. To solve the game and find the network topology, we have studied an equilibrium concept, the hierarchical Nash equilibrium, which is the solution to a series of multi-stage Nash games using backward induction among the stages. To reach this equilibrium, a myopic fictitious-play-based algorithm is studied which allows the nodes to compute their mixed strategies in a distributed manner. With the studied algorithm, the nodes self-organize into a hierarchical multi-hop architecture, while improving their utility in terms of the probability of successful transmission in the presence of interference. The hierarchical network-formation game can be extended to other applications such as multi-hop cognitive radio and sensor networks.

11.4 Other typical approaches

Beyond the two examples discussed in the previous sections, here we study typical approaches described in the literature. First, in a price-based approach, each hop has a price and the game outcome is controlled between the source–destination pair and the intermediate hops. Second, an auction-based approach is studied to ensure that users reveal their information truthfully for network cooperation, because this will bring them the best payoffs. Finally, an evolutionary-game approach is applied to the dynamic behavior of distributed nodes.

11.4.1 Price-based solution

In [301], a price-based reliable routing game in a wireless network of selfish users is studied. Each node is characterized by the probability of reliably forwarding a packet, and each link is characterized by the cost of transmission. The objective is to form a stable and reliable routing path between a given source–destination pair. The pricing mechanism involved in this routing game is destination-driven and source-mediated: for each successfully delivered packet, the destination node pays the source, which in turn compensates all nodes that participated in routing the packet. We first formulate the problem, then solve it using polynomial-time algorithms to obtain an efficient Nash equilibrium routing path.

The wireless network is modeled as an undirected graph G(V, E), where V denotes all the nodes in the network and E represents the link set. Each node v_i in V has a reliability parameter R_i , which represents the probability that it will forward the packet. Each edge $e_{ij} = (v_i, v_j) \in E$ has a link cost $C_{i,j}$. We assume the source and destination nodes have a reliability of 1. The destination node gives the source a payment G for every successfully transmitted packet, and the source gives a payment p to any intermediate node along the selected route.

In the game, all nodes except the destination are players. For each node v_i , its strategy is an *n*-tuple $\mathbf{s}_i = (s_{i,1}, s_{i,2}, \dots, s_{i,n})$ where $s_{i,j} = 1$ if node v_j is v_i 's next hop in the route, and $s_{i,j} = 0$ otherwise. Each strategy–tuple has at most one 1, i.e., $\sum_{j=1}^{n} s_{i,j} = 1, \forall i$. If node v_i 's strategy–tuple contains all zeros, node v_i does not participate in the game. The system strategy profile $(\mathbf{s}_i)_{i \in I}$ contains all players' strategies in the network. There is either no

path from the source to the destination, or exactly one path $P = (src, v_1, v_2, ..., v_h, dst)$, where *h* denotes the number of hops in between. The utility for the source is

$$u_{src} = \begin{cases} 0, & \text{no route,} \\ (G - hp) \prod_{v_i \in P} R_i - C_{src, v_1}, & \text{otherwise.} \end{cases}$$
(11.25)

The source's utility is the expected income minus the link set-up cost for the first hop. The source's expected income is the destination payment minus the payments to all of the intermediate nodes, times the probability that the packet will be delivered over this route. For every other intermediate node v_i , the utility is

$$u_{v_i} = \begin{cases} 0, & \text{if no route through } v_i, \\ p \prod_{v_i}^{v_h} R_i - C_{v_i, v_{i+1}}, & \text{otherwise.} \end{cases}$$
(11.26)

The utility for each intermediate routing node equals the expected payment that it obtains from the source node, times the ongoing route reliability minus the transmission cost per packet to its next-hop neighbor. If the node does not participate in the routing, it gains (and loses) nothing.

For the game described above, the outcome is given by the following lemma [301]:

LEMMA 11.1 If a path exists and it is a Nash equilibrium, then every node on the path must have a non-negative payoff.

Having defined the game, we now solve it using some simple algorithm, first simplifying the utility functions. For each intermediate node to have a positive payoff,

$$\prod_{k=i}^{n} R_k \ge \frac{C_{i,i+1}}{p}.$$
(11.27)

Taking inverse and log operations,

$$\sum_{i=1}^{n} \log \frac{1}{R_k} \le \log \frac{p}{C_{i,i+1}}.$$
(11.28)

Define $r_k = \log \frac{1}{R_k}$, $c_{i,i+1} = \log \frac{p}{C_{i,i+1}}$, and $c_{src,nbr} = \log \frac{G-hp}{C_{src,v_1}}$. Then the new utility functions can be written as

$$\widetilde{u}_{v_i} = \sum_{k=i}^n r_k - c_{i,i+1}, \text{ and } \widetilde{u}_{src} = \sum_{k=1}^n r_k - c_{src,nbr}.$$
(11.29)

Because of the linear form of the above utility functions, we can employ some modified Dijkstra's algorithms [301] to efficiently solve the price-based routing-game problem.

11.4.2 Truthfulness and security using auction theory

In this subsection, we provide an example from [39, 137] demonstrating the use of auction theory (Chapter 8) for the routing problem in multi-hop networks. The source





(S) – destination (*D*) pair builds up the underlying graph N = (V, E, w) by some routing discovery algorithms, where *V* is the set of vertices, *E* is the set of edges, and *w* is the set of weights. Then, the shortest path *SP* is computed from *S* to *D*, say *S*, $v_{\delta(1)}, \ldots, v_{\delta(k)}, D$.² Let |SP| denote the total cost of the shortest path *SP*. In order to compute the VCG payments (Chapter 8) to the intermediate nodes $v_{\delta(1)}, \ldots, v_{\delta(k)}$, the destination also computes for each node $v_{\delta(i)}, 1 \le i \le k$, the shortest path $SP^{-v_{\delta}(i)}$ from *S* to *D* that does not contain node $v_{\delta(i)}$ as an intermediate node. The VCG payment $M_{\delta(i)}$ for intermediate node $v_{\delta(i)}$ is then defined as

$$M_{\delta(i)} \triangleq |SP^{-\nu_{\delta(i)}}| - |SP| + c_{\delta(i)} P_{\delta(i),\delta(i+1)}^{\min}.$$
(11.30)

In other words, $M_{\delta(i)}$ is the difference between the cost of the shortest path from *S* to *D* without node $v_{\delta(i)}$, and the cost of the shortest path from *S* to *D* without the cost incurred by $v_{\delta(i)}$. The term $c_{\delta(i)}P_{\delta(i),\delta(i+1)}^{min}$ in the payment corresponds to the cost incurred by node $v_{\delta(i)}$. The difference $|SP^{-v_{\delta(i)}}| - |SP|$ is the (always non-negative) premium paid to node $v_{\delta(i)}$.

Figure 11.13 shows an example of how to calculate the payments. In this small network, consisting of six nodes, the costs are written on the edges. The most cost-efficient path from source *S* to destination *D* is SP = S, v_2 , v_3 , *D*, with |SP| = 5 + 2 + 3 = 10. The shortest path without node v_2 is $SP^{-2} = S$, v_1 , v_4 , *D*, with $\cos |SP^{-2}| = 7 + 3 + 4 = 14$. The shortest path without node v_3 is $SP^{-3} = S$, v_2 , v_4 , *D*, with $\cos |SP^{-2}| = 5 + 3 + 4 = 12$. According to (11.30), the VCG payments are $M_2 = 14 - 10 + 2 = 6$, $M_3 = 12 - 10 + 3 = 5$.

Source node S has to send the data messages to node v_2 and incurs a cost of 5. In the source model, the source also needs to pay amounts M_2 and M_3 to nodes v_2 and v_3 , respectively, for their forwarding service, resulting in an overall cost of 5+6+5=16 for source S. On the other hand, if there exists a central-bank model, the source only pays the true cost to the intermediate nodes. This results in an overall cost of 5+2+3=10. Then the intermediate nodes receive their premiums from the central bank, which collects

 2 If there is more than one shortest path, then the destination randomly chooses one of them.

them evenly from all network nodes. This is the problem of over-payment in the VCG auction.

11.4.3 Evolutionary-game approach

In [37], an evolutionary-game model for traffic routing is presented for an IEEE 802.16 multi-hop wireless backhaul network. In this case, the channel quality between relay stations can fluctuate because of fading. Therefore, the users (i.e., players) at the source node have to be able to observe, learn, and change the routing strategy to achieve the most reliable path from source node to destination node (e.g., an Internet gateway). Therefore, the strategy of the player is the path to be selected. An example of a network is shown in Fig. 11.14. The objective of a user is to minimize the end-to-end packet error rate (PER) by sampling the quality of the path from the physical layer.

An IEEE 802.16 multi-hop wireless backhaul network is modeled as a graph, where \mathcal{K} is a set of paths and \mathcal{V} is a set of nodes. λ_k denotes the traffic rate of path k, and the normalized rate $x_k = \frac{\lambda_k}{\sum_{k \in \mathcal{K}} \lambda_k}$. This variable x_k can be considered as the proportion of the population of users to choose path k, and a vector can be defined as $\mathbf{x} = [x_1 \dots x_k \dots x_K]$, where $\mathcal{K} = |\mathcal{K}|$ is the total number of paths. Note that the capacity of path k is denoted C_k , where $0 \le \lambda_k \le C_k$.

The payoff of this routing evolutionary game is based on the reliability of the path. Since the system model considers IEEE 802.16 operating on the frequency spectrum above 10 GHz, rain attenuation, denoted by *a*, is an important factor in packet error. In this case, the PER can be defined as an increasing function of *a*: PER = f(a). As a result, the most reliable path is the path with the smallest rain attenuation. The users perform traffic routing by periodically and randomly sampling different paths, and the PER is measured and compared among these sampled paths. If a lower value of PER than the currently selected path is found, the user switches to the new path. In this case, the sampling rate (i.e., to measure PER) is denoted as $r_k(\mathbf{x})$, given the current state \mathbf{x} . The probability that the user switches from path *k* to path *k'* is denoted $p_{k\to k'}$, and the number of users switching from path *k* to *k'* is

$$\sum_{k'\in\mathcal{K},k'\neq k} x_k r_k(\mathbf{x}) p_{k\to k'} = x_k r_k(\mathbf{x}) \sum_{k'\in\mathcal{K},k'\neq k} p_{k\to k'} = x_k r_k(\mathbf{x}) (1-p_{k\to k}). \quad (11.31)$$





As the inflow to path k is $\sum_{k' \in \mathcal{K}, k' \neq k} x_{k'} r_{k'}(\mathbf{x})(1 - p_{k' \to k})$, the replicator dynamics of routing the evolutionary game can be expressed as follows:

$$\dot{x}_{k} = \sum_{k' \in \mathcal{K}, k' \neq k} x_{k'} r_{k'}(\mathbf{x}) p_{k' \to k} - x_{k} r_{k}(\mathbf{x}) (1 - p_{k \to k})$$
(11.32)

$$= \sum_{k'\in\mathcal{K}} x_{k'} r_{k'}(\mathbf{x}) p_{k'\to k} - x_k r_k(\mathbf{x}).$$
(11.33)

Given the sampling mechanism, the user switches from path k to path k' if the rain attenuation of this path is smaller, i.e., $a_k > a_{k'}$. Given that this rain attenuation is a random variable with a continuously differentiable cumulative distribution function $\phi(a)$, the conditional probability that the user will switch from path k to path k' can be expressed as

$$\phi(a_k - a_{k'}) = \Pr(a_k - a_{k'} > 0 | a_k > 0, a_{k'} > 0), \qquad (11.34)$$

where $\Pr(a_k - a_{k'} > 0 | a_k > 0, a_{k'} > 0)$ is the probability of event $a_k - a_{k'} > 0$ given $a_k > 0$ and $a_{k'} > 0$. The closed-form expression of $\phi(a_k - a_{k'})$ is provided in [37]. Then, the probability of switching paths is

$$p_{k \to k'} = \begin{cases} x_{k'} \phi(\mathbf{a}_k - \mathbf{a}_{k'}), & \text{if } k \neq k', \\ 1 - \sum_{l \neq k, l \in \mathcal{K}} x_l \phi(\mathbf{a}_k - \mathbf{a}_l), & \text{if } k = k'. \end{cases}$$
(11.35)

If $r_k(\mathbf{x}) = 1$, the replicator dynamics can be simply expressed as

$$\dot{x}_{k} = x_{k} \sum_{k' \in \mathcal{K}, k' \neq k} x_{k'} \left(\phi(a_{k'} - a_{k}) - \phi(a_{k} - a_{k'}) \right).$$
(11.36)

Given this replicator dynamics, the stability of the equilibrium is analyzed. With a rain fading condition, only one route will be an evolutionarily stable strategy, based on the fact that two paths cannot have exactly the same rain attenuation. In particular, the probability of $a_k = a_{k'}$ for $k \neq k'$ approaches zero. Then, Lyapunov's first method is applied to evaluate the stability of the strategy that corresponds to the path with the smallest rain attenuation. In addition, the convergence rate of the replicator dynamics is investigated. Although the investigation is performed for a specific network (i.e., with five paths between source and destination), a similar result is expected for the general case. That is, the algorithm to sample and switch the path converges faster for the higher frequency and for the longer distance between nodes in the path (e.g., as shown in Fig. 11.14). This is because rain attenuation is larger for higher frequency and longer distance. As a result, the users can observe this effect easily and switch to the better path.

11.5 Summary

In this chapter, we have studied the formulation of games for multi-hop wireless networks. The game is defined by the players (e.g., source–destination pairs), the strategies (e.g., packet-forwarding probability), and the utilities (representing not only each node's performance but also the interactions with other nodes). A simple Braess' paradox and its application to wireless networks were studied to show that network performance can be greatly degraded because of the greediness of distributed users. To overcome this problem, we studied two examples in detail. First, a repeated-game approach was investigated to maintain cooperation using the threat of future punishment. A learning aspect was also introduced, for finding better cooperation points. Second, a network-formation game approach was studied for hierarchical routing, fitting for multi-hop networks with a common sink (such as cellular relay networks or sensor networks). Finally, three other typical approaches – price-based, auction-based, and evolutionary-game-based – were briefly studied.

Cooperative communication has attracted significant recent attention as a transmission strategy for future wireless networks. It efficiently takes advantage of the broadcast nature of wireless networks to allow network nodes to share their messages and transmit cooperatively as a virtual antenna array, thus providing diversity that can significantly improve system performance. Cooperative communication can be applied in a variety of wireless systems and networks. In the research community, a considerable amount of work has been done in this area for networks such as cellular, WiFi, ad hoc/sensor networks, and ultra wideband (UWB). These ideas are also working their way into standards; e.g., the IEEE 802.16 (WiMAX) standards body for future broadband wireless access has established the 802.16 jrelay task group to incorporate cooperative relaying mechanisms into this technology. Most existing work on cooperative communication concentrates on the physical (PHY) and medium access control (MAC) layers of wireless networks, examining issues such as capacity improvement, power control, and relay selection. The impact on the higher layers, such as routing in the network layer, has not been fully investigated yet.

The merits of cooperative transmission at the physical layer have been well explored. However, the impact of cooperative transmission on the design of the higher layers is not well understood yet. Specifically, the issues for various layers are:

- **Physical layer.** Objectives include optimizing the capacity region, minimizing the bit error rate (BER), and improving the link quality by power control.
- MAC layer. Relay selection and channel allocation, i.e., among all possible relays and channels, can improve the source–destination link.
- **Routing layer.** The main problem is route selection. With cooperative communication, link quality can be improved using relays. So, the optimal route selection depends not only on the nodes of the links but also on the relays.
- **Application layer.** For multimedia transmission, relays can forward some coded and processed information instead of relaying only the original bits. The destination can improve the reconstructed voice/image/video quality because of the nature of multimedia data.

This chapter considers the impact of cooperative communication on these layers over a range of wireless-network scenarios, and offers insights into the vertical integration of wireless networks by cross-layer optimization. The goals of this chapter are to provide an understanding of the impact of cooperative communication and to provide new perspectives on system optimization based on the game-theoretic point of view.

Some basic cooperative-communication protocols are examined in Section 12.1, followed by a literature review and brief discussion of the impact of cooperative communication on the various layers of wireless networks. Four case studies are presented: non-cooperative games for relay selection and power control in Section 12.2; auction-theory-based resource allocation in Section 12.3; cooperative transmission using cooperative games in MANET in Section 12.4; and routing problems in general multi-hop networks in Section 12.5.

12.1 Basics of cooperative transmission

In this section, we classify the currently known cooperative-transmission (CT) protocols, and briefly review the state of the the art in cooperative communication, including its impact on different layers.

12.1.1 Cooperative-transmission protocols

To illustrate the basic idea of cooperative transmission, a highly simplified topology with one source node, two relay nodes, and one destination node is shown in Fig. 12.1. Cooperative transmission is conducted in two phases. In Phase 1, the source broadcasts a message to the destination and relay nodes. In Phase 2, relay nodes send information



Fig. 12.1 Cooperative-communication system model.

to the destination (in different time slots or on different orthogonal channels), and the destination combines the messages from the source and relays. It has been shown that the capacity region of this communication channel can be significantly increased by such techniques, and that the performance gain of cooperative transmission is proportional to the number of relays in the Rayleigh fading case [279]. Here, we denote the source node as *s*, the relay nodes as r_i , and the destination node as *d*. In Phase 1, the signals Y_d and Y_{r_i} received at the destination *d* and relay r_i , respectively, can be expressed as

$$Y_d = \sqrt{P_s G_{s,d}} X + n_d \tag{12.1}$$

and

$$Y_{r_i} = \sqrt{P_s G_{s,r_i}} X + n_{r_i}, \qquad (12.2)$$

where P_s is the transmit power from the source, X is the unit-energy information symbol transmitted by the source in Phase 1, $G_{s,d}$ and G_{s,r_i} are the channel gains from *s* to *d*, and from *s* to r_i , respectively, and n_d and n_{r_i} are samples from independent (discrete-time) additive white Gaussian noise (AWGN) processes, independent of X. Without loss of generality, we assume that the noise power, denoted by σ^2 , is the same for all the links.

A number of different types of cooperative-communication protocols have been developed in the literature. We describe several of the most relevant of these here.

Direct transmission

Without the relay nodes' help, the signal-to-noise ratio (SNR) from s to d can be expressed as

$$\Gamma_{s,d}^{DT} = \frac{P_s G_{s,d}}{\sigma^2},\tag{12.3}$$

and the capacity of the direct-transmission channel is

$$R_{s,d} = W \log_2\left(1 + \Gamma_{s,d}^{DT}\right), \qquad (12.4)$$

where W is the bandwidth used for information transmission.

Amplify-and-forward (AF) cooperative transmission

In Phase 2, relay *i* amplifies Y_{r_i} and forwards it to the destination with transmitted power P_{r_i} . The received signal at the destination is

$$Y_{r}^{i} = \sqrt{P_{r_{i}}G_{r_{i},d}}X_{r_{i}} + n_{d}^{\prime}, \qquad (12.5)$$

where

$$X_{r_i} = \frac{Y_{r_i}}{|Y_{r_i}|} \tag{12.6}$$

is the energy-normalized transmitted signal from the source to the destination in Phase 1, $G_{r_i,d}$ is the channel gain from relay *i* to the destination, and n'_d is the received noise in Phase 2. Substituting (12.2) into (12.6), we can rewrite (12.5) as

$$Y_{r}^{i} = \frac{\sqrt{P_{r_{i}}G_{r_{i,d}}}(\sqrt{P_{s}G_{s,r_{i}}}X_{s} + n_{r_{i}})}{\sqrt{P_{s}G_{s,r_{i}} + \sigma^{2}}} + n_{d}^{\prime}.$$
 (12.7)

Using (12.7), the relayed SNR at the destination for the source, assisted by relay node i, is given by

$$\Gamma_{s,r_i,d}^{AF} = \frac{P_{r_i} P_s G_{r_i,d} G_{s,r_i}}{\sigma^2 (P_{r_i} G_{r_i,d} + P_s G_{s,r_i} + \sigma^2)}.$$
(12.8)

Therefore, by (12.4) and (12.8), the channel capacity, assuming maximal ratio combining (MRC) at the destination, is given by

$$R_{s,r_{i},d}^{AF} = \frac{1}{2} W \log_2 \left(1 + \Gamma_{s,d}^{DT} + \Gamma_{s,r_{i},d}^{AF} \right).$$
(12.9)

If multiple relay nodes (say, $i \in L$, where |L| = N) are available to help the source, then we have

$$R_{s,L,d}^{AF} = \frac{1}{N+1} W \log_2 \left(1 + \Gamma_{s,d}^{DT} + \sum_{r_i \in L} \Gamma_{s,r_i,d}^{AF} \right).$$
(12.10)

Here, the $\frac{1}{2}$ in (12.9) and the $\frac{1}{N+1}$ in (12.10) arise because the relays need extra orthogonal channels for transmission.

Decode-and-forward (DF) cooperative transmission

Here, the relay decodes the source information in Phase 1 and relays it to the destination in Phase 2. The destination combines the direct-transmission information with the relayed information. The achievable rate in this case can be calculated by the following maximization:

$$R_{s,r_i,d}^{DF} = \max_{0 \le \rho \le 1} \min\{R_1, R_2\},$$
(12.11)

where

$$R_{1} = \log_{2} \left[1 + (1 - \rho^{2}) \frac{P_{s,d} G_{s,r_{i}}}{\sigma^{2}} \right]$$
(12.12)

and

$$R_{2} = \log\left(1 + \frac{P_{s}G_{s,d}}{\sigma^{2}} + \frac{P_{r_{i}}G_{r_{i},d}}{\sigma^{2}} + \frac{2\rho\sqrt{P_{s}G_{s,d}P_{r_{i}}G_{r_{i},d}}}{\sigma^{2}}\right).$$
 (12.13)
Estimate-and-forward (EF) cooperative transmission

Here, in Phase 2, the relay sends an estimate of the received signal of Phase 1. The destination uses the relay's information as side information to decode the direct transmission of Phase 1. From [115] and [249], the channel capacity resulting from this approach can be written as

$$R_{s,r_i,d}^{EF} = W \log_2 \left(1 + \Gamma_{s,d}^{DT} + \Gamma_{s,r_i,d}^{EF} \right), \qquad (12.14)$$

where

$$\Gamma_{s,r_i,d}^{EF} = \frac{P_s P_{r_i} G_{s,r_i} G_{r_i,d}}{\sigma^2 [P_{r_i} G_{r_i,d} + P_s (G_{s,d} + G_{s,r_i}) + \sigma^2]}.$$
(12.15)

Coded cooperation

This type of cooperative-transmission protocol integrates relay cooperation with channel coding [219]. Instead of exactly repeating the received information, the relay decodes the partner's transmission and transmits additional parity symbols (e.g., incremental redundancy) according to a certain overall coding scheme. The destination receiver conducts channel decoding by concatenating the data from the direct transmission and relay transmission, so that the channel gain can be obtained.

Distributed space-time coded cooperation

Space-time coding has been shown to significantly improve the link performance in multiple-input multiple-output (MIMO) systems. Distributed space-time cooperative diversity protocols [280] exploit the spatial diversity among a collection of distributed nodes that relay messages for one another, in such a manner that the destination terminal can combat the fading. Those relays can fully decode the transmission from the source and then encode the data using a space-time code to cooperatively relay to the destination. At the destination, the space-time code implemented by the source and relays can achieve full diversity.

Incremental relaying [279]

Here, the destination broadcasts ACK (acknowledgment) or NACK (negative acknowledgment) information after Phase 1. The relay retransmits only after receiving a NACK. In this way, the bandwidth efficiency can be greatly improved, since the only bandwidth increase occurs when the direct-transmission link fails. But the gain comes with additional implementation costs for the feedback mechanism.

Cognitive relaying [417, 220]

Cognitive radio is a revolutionary paradigm with high spectral efficiency involving wireless communication in an occupied spectrum without interfering with existing band occupants. Cooperative-communication protocols can help cognitive users reduce the detection time for a clear spectral band, and thus increase their agility. On the other hand, relays can monitor the spectrum cognitively so as to improve the source–destination link.

12.1.2 State of the art and impact on different layers

Cooperative transmission began as a physical-layer protocol [279, 432, 433], and most work in this area focuses on its merits in the physical layer. However, the impact of cooperative transmission on the design of the higher layers is obviously also of importance, although it is not yet well understood. We now briefly discuss the impact of cooperative communication on different layers. Specifically, we list only a limited number of sources, dividing the discussion as follows:

- *Capacity analysis and new cooperative communication protocols*: The major concerns here are to analyze how much gain cooperative transmission can bring to a link and to the overall network [466, 214, 213], and how to implement cooperative transmission under practical constraints [219, 221, 249].
- *Relay selection and power control*: When there are several relays, a question arises as to which one to select for a given retransmission. After relay selection, the next issue is how limited power resources should be distributed over sources and relays [306, 534]. These questions have also been addressed in systems using multi-user detection [192] and orthogonal frequency-division multiplexing (OFDM) [181].
- *Routing protocols*: Cooperative transmission can provide extra routes for network protocols so that network performance can be significantly improved. These can be traditional routes [518, 248] or cooperative routes [225]. It has been shown that network lifetime can be significantly improved via such considerations [190]. Multi-hop cooperative transmission can be considered as a special case of routing [82, 290, 418, 75].
- *Distributed resource allocation*: Game-theoretic approaches are natural for distributed cooperative resource-allocation problems, as the individual nodes can use only local information to optimize cooperative communication [494]. Moreover, as shown in [189], cooperative-game theory and cooperative transmission can be used to improve packet-forwarding networks with selfish nodes.
- *Others*: Cooperative transmission has been considered jointly with other problems such as source coding [177] and energy-efficient broadcasting [319].

12.2 Non-cooperative game for relay selection and power control

In this section, we propose a Stackelberg game-theoretic framework [494, 495] for distributive resource allocation over multi-user cooperative-communication networks to improve system performance and stimulate cooperation. Two main resource-allocation questions about cooperative multi-user wireless networks remain unanswered. First, among all the distributed nodes, which one can best help relay and improve the source's link quality? Second, for the selected relay nodes, how much power do they need to transmit? Both questions need to be answered in a distributed way.

To answer these questions, we employ a special kind of non-cooperative game, the Stackelberg game [459], to jointly consider the benefits of source nodes and relay nodes

in cooperative communication. The game is divided into two hierarchical levels: the source node plays a *buyer-level* game and the relay nodes play a *seller-level* game. Each player is selfish and wants to maximize its own benefit. Specifically, the source can be viewed as a buyer which aims to maximize its benefit at the least possible cost. Each relay can be seen as a seller who aims to earn the payment. The payment not only covers their forwarding cost but includes as much extra profit as possible. Then we derive expressions for the proposed game outcomes. We analyze how many relay nodes would be selected by the source to participate in the sale process, after the relay nodes announce their optimal prices. In addition, we optimize how much service the source should buy from each relay node. From the seller's point of view, the relay nodes set a corresponding optimal price per unit of service, such as relay power, so as to maximize its own benefit. From the simulations, because of competition by other relays and selections by the source, the relays have to set a price which will attract the source, so as to optimize their utility values. The source optimally selects the relays and their relaying power, while the relays set prices that can maximize their utilities.

12.2.1 Relay-selection and power-control problem

For the system model, we use the amplify-and-forward (AF) cooperative protocol as an example. Other cooperative protocols such as decode-and-forward (DF) can be applied in a similar way. The relay nodes help the source node by relaying the received information to the destination. The receiver at the destination combines the directly received signal from the source node and the relayed signals from the relay nodes, using techniques such as maximal ratio combining (MRC). This process can be described in two phases.

In Phase 1, without the relay nodes' help, the signal-to-noise ratio (SNR) that results from direct transmission from the source s to the destination d is

$$\Gamma_{s,d} = \frac{P_s G_{s,d}}{\sigma^2},\tag{12.16}$$

where P_s is the transmit power, $G_{s,d}$ is the channel gain, and σ^2 is the noise variance. The rate of non-cooperative transmission at the output is

$$R_{s,d}^{nc} = W \log_2 \left(1 + \frac{\frac{P_s G_{s,d}}{\sigma^2}}{\Gamma} \right), \qquad (12.17)$$

where Γ is a constant for the capacity gap. Without loss of generality, we assume that the noise variance is the same for all links. We also assume the channels are constant over each power-control interval.

In Phase 2, we consider the SNR at the destination that results from relay r_i relaying source *s*'s data to the destination. If X_{s,r_i} is the transmitted signal from source *s* to relay r_i , the received signal at relay r_i is

$$R_{s,r_i} = \sqrt{P_s G_{s,r_i}} X_{s,r_i} + \eta, \qquad (12.18)$$

where $\eta \sim N(0, \sigma^2)$ and σ^2 is the noise variance. Relay r_i amplifies R_{s,r_i} and relays it to the destination, where the received signal is

$$R_{r_{i},d} = \sqrt{P_{r_{i}}G_{r_{i},d}}X_{r_{i},d} + \eta, \qquad (12.19)$$

where

$$X_{r_{i},d} = \frac{R_{s,r_{i}}}{|R_{s,r_{i}}|}$$
(12.20)

is the transmitted signal from relay r_i to the destination, and the signal is normalized to have unit energy. Substituting (12.18) into (12.20), we can rewrite (12.19) as

$$R_{r_{i},d} = \frac{\sqrt{P_{r_{i}}G_{r_{i},d}}(\sqrt{P_{s}G_{s,r_{i}}}X_{s,r_{i}} + \eta)}{\sqrt{P_{s}G_{s,r_{i}} + \eta^{2}}} + \eta.$$
(12.21)

Using (12.21), the relayed SNR for the source s, which is helped by relay r_i , is

$$\Gamma_{s,r_i,d} = \frac{P_{r_i} P_s G_{r_i,d} G_{s,r_i}}{\sigma^2 (P_{r_i} G_{r_i,d} + P_s G_{s,r_i} + \sigma^2)}.$$
(12.22)

Therefore, by (12.16) and (12.22), the rate at the output of MRC via relay r_i in AF is

$$R_{s,r_{i},d}^{AF} = W \log_2 \left(1 + \frac{\frac{P_s G_{s,d}}{\sigma^2} + \frac{P_r P_s G_{r_{i},d} G_{s,r_{i}}}{\sigma^2 \left(P_r G_{r_{i},d} + P_s G_{s,r_{i}} + \sigma^2 \right)}}{\Gamma} \right).$$
(12.23)

If there are N relays helping the source, then

$$R_{s,r,d}^{AF} = W \log_2 \left(1 + \frac{\frac{P_s G_{s,d}}{\sigma^2} + \sum_{i=1}^{N} \frac{P_{r_i} P_s G_{r_i,d} G_{s,r_i}}{\sigma^2 \left(P_{r_i} G_{r_i,d} + P_s G_{s,r_i} + \sigma^2 \right)}}{\Gamma} \right).$$
(12.24)

12.2.2 Stackelberg-game approach

To explore the cooperative diversity for a multi-user system, from (12.24), the following fundamental questions need to be answered: (1) Which relay nodes should be included? (2) What is the optimal power P_{r_i} ? To answer these questions, we employ Stackelberg games for buyers and sellers, as follows.

The source can be modelled as a *buyer*, aiming to maximize its benefit at the least possible cost. The utility function of the source can be defined as

$$U_s = a\Delta R_{tot} - M, \qquad (12.25)$$

where

$$\Delta R_{tot} = R_{s,r,d} - R_{s,d} \tag{12.26}$$

denotes the total rate increment with the relay nodes helping transmission, *a* denotes the gain per unit of rate increment at the MRC output, and

$$M = p_1 P_{r_1} + p_2 P_{r_2} + \dots + p_N P_{r_N}$$
(12.27)

represents the total payment paid by the source to the relay nodes. In (12.27), p_i represents the price per unit of power sold by relay node *i* to the source *s*, and P_{r_i} denotes how much power the source would like to buy from relay r_i when the price from that relay is announced.

We assume the number of relay nodes is N. Without loss of generality the parameter a in (12.25) can be set to 1. The optimization problem, or the buyer's game, can be formulated as:

$$\max_{\{P_{r_i}\}} U_s = \Delta R_{tot} - M,$$
s.t. $\{P_{r_i}\} \ge 0.$
(12.28)

Each relay r_i can be seen as a *seller* aiming to earn a payment which not only covers its forwarding cost but as much extra profit as possible. We introduce the parameter c_i , the cost of power for relaying data, to correctly reflect the relays' judgements about whether they can actually profit by the sale. Then relay r_i 's utility function can be defined as

$$U_{r_i} = (p_i - c_i)P_{r_i}, (12.29)$$

where c_i is the cost per unit of power in relaying data, p_i has the same meaning as in (12.27), and P_{r_i} is the source's decision to optimize U_s as described in (12.28). It is obvious that the optimal p_i depends not only on each relay's own channel condition to the destination but also on its counterpart relays' prices. So in the sellers' competition, if one relay asks a higher price than the source expects after jointly considering all relays' prices, the source will buy less from that relay or even disregard it. On the other hand, if the price is too low, the profit obtained from (12.29) will be unnecessarily low. So there is a tradeoff for setting the price. Without loss of generality, set $c_i = c$ in (12.29) and the optimization for relay r_i , or the seller's game, is:

$$\max_{\{p_i\}\geq 0} U_{r_i} = (p_i - c)P_{r_i}, \quad \forall i.$$
(12.30)

Thus, the ultimate goals of the above two games are to decide the optimal pricing p_i to maximize the relays' profits U_{r_i} , the actual number of relays who will finally get selected by the source, and the corresponding optimal power consumption P_{r_i} to maximize U_s . Notice that the only signals required between the source and each relay are the price p_i and the information about how much power P_{r_i} to buy. Consequently, the proposed two-level game approach leads to distributed resource allocation for the cooperative-communication network. The outcome of the games will be shown in detail.

Let us make some observations about the U_s function with respect to $\{P_{r_i}\}$. When P_{r_i} is close to 0, less help is received from the relay, so U_s should be close to 0. As P_{r_i} increases, relays sell more power to the source, so a greater rate increment is obtained,

and U_s increases. If P_{r_i} increases further, the rate increment will saturate but the cost will continue to grow; hence the utility for the source U_s begins to decrease. If the selling prices p_i , i = 1, 2, ..., N have been announced, then from the first-order optimization condition, the following conditions must hold at the optimal point:

$$\frac{\partial U_s}{\partial P_{r_i}} = 0, \quad i = 1, 2, \dots, N.$$
(12.31)

For simplicity, define $C = 1 + \frac{P_s G_{s,d}}{\sigma^2 \Gamma}$ and $W' = \frac{W}{\ln 2}$. Then, by (12.24) and (12.17),

$$\begin{split} \Delta R_{tot} &= R_{s,r,d} - R_{s,d} \\ &= W \log_2 \left(C + \frac{\sum_{i=1}^{N} \frac{P_{r_i} P_s G_{r_i,d} G_{s,r_i}}{\sigma^2 (P_{r_i} G_{r_i,d} + P_s G_{s,r_i} + \sigma^2)}}{\Gamma} \right) - W \log_2 C \\ &= W \log_2 \left(1 + \frac{\sum_{i=1}^{N} \frac{P_{r_i} P_s G_{r_i,d} G_{s,r_i}}{\sigma^2 (P_{r_i} G_{r_i,d} + P_s G_{s,r_i} + \sigma^2)}}{\Gamma C} \right) \end{split}$$
(12.32)
$$&= W \log_2 \left(1 + \frac{\Delta SNR_{tot}}{\Gamma C} \right) = W' \ln [1 + \Delta SNR'_{tot}] \\ &= W' \ln \left[1 + \sum_{i=1}^{N} \Gamma'_{r_i,d} \right], \end{split}$$

where

$$\Delta SNR'_{tot} = \frac{\sum_{i=1}^{N} \frac{P_{r_i} P_s G_{r_i,d} G_{s,r_i}}{\sigma^2 (P_{r_i} G_{r_i,d} + P_s G_{s,r_i} + \sigma^2)}}{\Gamma C}$$
(12.33)

and

$$\Gamma_{r_{i},d}^{\prime} = \frac{\Gamma_{r_{i},d}}{\Gamma C} = \frac{\frac{P_{s}G_{s,r_{i}}}{(\Gamma \sigma^{2} + P_{s}G_{s,d})}}{1 + \left[\frac{\left(\frac{P_{s}G_{s,r_{i}} + \sigma^{2}}{G_{r_{i},d}}\right)}{P_{r_{i}}}\right]} = \frac{A_{i}}{1 + \frac{B_{i}}{P_{r_{i}}}} = \frac{A_{i}P_{r_{i}}}{P_{r_{i}} + B_{i}},$$
(12.34)

with $A_i = \frac{P_s G_{s,r_i}}{(\Gamma \sigma^2 + P_s G_{s,d})}$ and $B_i = \frac{P_s G_{s,r_i} + \sigma^2}{G_{r_i,d}}$. Substituting (12.27) and (12.32) into (12.31), we have

$$\frac{\partial U_s}{\partial P_{r_i}} = \frac{W'}{\left(1 + \sum_{k=1}^N \frac{A_k P_{r_k}}{P_{r_k} + B_k}\right)} \frac{A_i B_i}{\left(P_{r_i} + B_i\right)^2} - p_i = 0, \qquad (12.35)$$

i.e.,

$$\frac{W'}{\left(1 + \sum_{k=1}^{N} \frac{A_k P_{r_k}}{P_{r_k} + B_k}\right)} = \frac{p_i}{A_i B_i} \left(P_{r_i} + B_i\right)^2.$$
(12.36)

So

$$\frac{p_i}{A_i B_i} (P_{r_i} + B_i)^2 = \frac{p_j}{A_j B_j} (P_{r_j} + B_j)^2, \qquad (12.37)$$

$$P_{r_j} = \sqrt{\frac{p_i A_j B_j}{p_j A_i B_i} (P_{r_i} + B_i) - B_j.}$$
(12.38)

After some manipulation, we have

$$\Gamma'_{r_j,d} = \frac{A_j}{1 + \frac{B_j}{P_{r_j}}} = A_j - \sqrt{\frac{p_j A_i B_i}{p_i A_j B_j}} \frac{A_j B_j}{(P_{r_i} + B_i)},$$
(12.39)

so

$$\Delta SNR'_{tot} = \sum_{i=1}^{N} \Gamma'_{r_i,d}$$

$$= \left[A_1 - \sqrt{\frac{p_1 A_i B_i}{p_i A_1 B_1}} \frac{A_1 B_1}{(P_{r_i} + B_i)} \right] + \left[A_2 - \sqrt{\frac{p_2 A_i B_i}{p_i A_2 B_2}} \frac{A_2 B_2}{(P_{r_i} + B_i)} \right]$$

$$+ \dots + \left[A_i - \frac{A_i B_i}{P_{r_i} + B_i} \right] + \dots + \left[A_N - \sqrt{\frac{p_N A_i B_i}{p_i A_N B_N}} \frac{A_N B_N}{(P_{r_i} + B_i)} \right]$$

$$= \sum_{j=1}^{N} A_j - \sqrt{\frac{A_i B_i}{p_i}} \frac{1}{P_{r_i} + B_i} \sum_{j=1}^{N} \sqrt{p_j A_j B_j}.$$
(12.40)

Substituting (12.40) into (12.36), we have a quadratic equation in P_{r_i} :

$$\left(1+\sum_{j=1}^{N}A_{j}\right)\left[\sqrt{\frac{p_{i}}{A_{i}B_{i}}}\left(P_{r_{i}}+B_{i}\right)\right]^{2}$$

$$-\sum_{j=1}^{N}\sqrt{p_{j}A_{j}B_{j}}\left[\sqrt{\frac{p_{i}}{A_{i}B_{i}}}\left(P_{r_{i}}+B_{i}\right)\right]-W'=0.$$
(12.41)

The generalized solution for the optimal power consumption from each relay node is then

$$P_{r_i} = -B_i + \frac{\sqrt{\frac{A_i B_i}{p_i}}}{2\left(1 + \sum_{j=1}^N A_j\right)} \left[\sum_{j=1}^N \sqrt{p_j A_j B_j} + \sqrt{\left(\sum_{j=1}^N \sqrt{p_j A_j B_j}\right)^2 + 4\left(1 + \sum_{j=1}^N A_j\right) W'}\right].$$
(12.42)

This solution may be negative for some relay's high price or bad location, so the optimal price is modified as follows:

$$P_{r_i}^* = \max(P_{r_i}, 0) = (P_{r_i})^+, \qquad (12.43)$$

where P_{r_i} is the solution of (12.42).

Substituting (12.42) into (12.30), we have

$$\max_{\{p_i\}>0} U_{r_i} = (p_i - c) P_{r_i}(p_1, \dots, p_i, \dots, p_N).$$
(12.44)

Note that this is a non-cooperative game, and there exists a tradeoff between the price p_i and the relay's utility U_{r_i} . If the relay asks a relatively low price p_i at first, the source will buy more power from this cheaper seller, and U_{r_i} will increase as p_i grows. As p_i keeps growing, the source will come to think it is no longer profitable to buy power from this relay and P_{r_i} will shrink, resulting in a decrease in U_{r_i} . Thus, there is an optimal price for each relay to ask, and it is also affected by other relays' prices, since the source only chooses the most beneficial relays.

From this analysis and the necessary conditions, it follows that

$$P_{r_i}(p_1, \dots, p_i, \dots, p_N) + (p_i - c) \frac{\partial P_{r_i}(p_1, \dots, p_i, \dots, p_N)}{\partial p_i} = 0, \quad i = 1, 2, \dots, N.$$
(12.45)

Solving (12.45) for N unknowns p_i , we have

$$p_i^* = p_i^*(\sigma^2, \{G_{s,r_i}\}, \{G_{r_i,d}\}), \quad i = 1, 2, \dots, N.$$
(12.46)

The problem in (12.46) can be solved by a numerical method such as sequential quadratic programming (SQP) [353].

Substitute (12.46) into (12.42) to see whether P_{r_i} is positive. If it is negative, then the source will disregard this relay, and only the remaining relays constitute the actual relaying subset. Re-solve (12.35) by changing the set of relay nodes to this subset, resolve for the new p_i^* , and check P_{r_i} until all are non-negative. Then we can obtain the final optimal prices p_i^* to maximize the relays' utilities U_{r_i} , the number of relays selected by the source, and the corresponding optimal power consumption $P_{r_i}^*$ to maximize U_s .

To evaluate the performance of this scheme and decide what price each relay should ask and how much power the source should buy from each relay, we performed simulations for multiple relay systems. The simulation results for one-relay, two-relay, and multiple-relay cases are described below.

For the one-relay case, there is one source–destination pair and one relay in the network. The destination was located at (0,0), the source at (1,0), and the relays uniformly distributed over the 2D area: $(x, y) \in [-2, 3] \times [-1, 1]$. The propagation loss factor was set to 2, the noise level, $\sigma^2 = 10^{-4}$ the capacity gap $\Gamma = 1$, and the cost per unit of power c = 0.05.

For the two-relay case, the coordinates of the source and the destination were (1,0) and (0,0), respectively. Relay 1 was fixed at (0.5,0.25) and relay 2 moved along the line from (-2,0.25) to (3,0.25). Other settings were the same as for the one-relay case.



Fig. 12.2 Optimal relay prices when relay 2 moves.

In Fig. 12.2 we show the optimal price that each relay should ask to maximize its profit. We observe that even though only relay 2 moves, the prices of both relays change, because the two relays compete and influence each other in the Stackelberg game. When relay 2 is close to the destination at (0,0), it needs very little power to relay the source's information, so it sets a very high price, hoping to get more profit. When relay 2 is close to the source at (1,0), it is well suited to help the source transmit. Consequently, in order to attract the source to buy its service, relay 1 must reduce its price. When relay 2 is far away, its price will drop because it is less competitive compared to relay 1 at location (0.5, 0.25). When its utility is less than 0, relay 2 will quit the competition, but it cannot increase it too much, otherwise relay 2 will rejoin the competition.

As shown in Fig. 12.3, the source will cleverly buy different amounts of power from each relay. When relay 2 moves away from the source, $P_{r_2}^*$ gradually decreases. When relay 2 moves too far away from the source or the destination, the source will not choose relay 2. When relay 2 is close to the destination, its price as shown in Fig. 12.2 is too high, and the source will not buy power from it. When relay 2 quits the competition, relay 1 will increase its price, although the source will buy slightly less (this suppresses the incentive of relay 1 to ask for arbitrarily high price in the absence of competition). Note that when relay 2 moves to (0.5, 0.25), the same location as relay 1, the power consumptions and prices of both relays are the same, because the source now treats them equally.

In the multiple-relay simulations, the coordinates of the source and the destination were (1,0) and (0,0), respectively, and the relays were randomly located within the range [-2,3] on the *x*-axis and [-2,2] on the *y*-axis. In Fig. 12.4, we observe that as the total number of available relays increases, the source achieves a higher utility. However,



Fig. 12.3 Optimal power consumption by two relays when relay 2 moves.



Fig. 12.4 Optimal source utility and average money transfer vs. number of relay nodes.

the competition among relays becomes more severe, which leads to a lower average payment from the source.

In summary, we have applied a game-theory approach to distributive resource allocation over multi-user cooperative-communication networks, to determine which relays will be selected and how much power will be used for relaying in the AF cooperative scenario. We applied a Stackelberg game to jointly consider the benefits of different types of nodes. The proposed scheme not only helps the source choose relays at optimal locations but also helps the competing relays ask a reasonable price to maximize their utilities. From the simulation results, relays close to the source play a more important role in increasing source utility, so the source prefers to buy power from them. In order to attract more consumption from the source, a relay might adopt a "low-price, high-market" policy to further increase its utility value. It is also easy to use the current structures as building blocks in large-scale wireless ad hoc networks to stimulate cooperation among nodes.

12.3 Auction-theory-based resource allocation

In order to maximize the performance of the cooperative-transmission network, we need to consider the global channel information, including between source-destination, source-relay, and relay-destination. Most existing work in this area is based on central-ized control, which requires considerable overhead for signalling and measurement. As in the previous section, we focus on designing *distributed* resource-allocation algorithms for cooperative networks. In particular, we want to answer the following questions: (1) "When to relay," i.e., when is it beneficial to use the relay? (2) "How to relay," i.e., how should the relay allocate resources among multiple competing users?

We answer these two questions by designing an *auction-based* framework (see Chapter 8) for cooperative resource allocation. Auctions have recently been introduced into several areas of wireless communications (e.g., time-slot allocation [469] and power control [134, 217]). The idea is closely related to the auction mechanisms proposed in [217], where the authors considered distributed interference management in a cognitive-radio network without a relay.

We consider two network objectives here: *fairness* and *efficiency*. Either might be difficult to achieve even in a centralized fashion. This is because users' rate increases are non-smooth and non-concave in the relay's transmission power, and thus the corresponding optimization problems are non-convex. We study two auction mechanisms, the SNR auction and the power auction, which achieve the desired network objectives in a distributed fashion under suitable technical conditions. In both auctions, each user decides "when to relay" based on a simple threshold policy that is locally computable. The question of "how to relay" is answered by a simple weighted proportional allocation among users who use the relay. Simulation results show that the power auction achieves an average of 95 percent of the maximum rate increase in a two-user network over a wide range of relay locations. The SNR auction achieves a fair allocation among users but leads to a much lower total rate increase. This reflects a fairness efficiency tradeoff that can be exploited by a system designer.

12.3.1 Resource-allocation objectives

We focus our discussion on the amplify-and-forward (AF) cooperative protocol [279]. Other cooperation protocols can be analyzed in a similar fashion. There are one relay

node *r* and a set $\mathcal{I} = (1, ..., I)$ of source–destination pairs. We also refer to pair *i* as *user i*, which includes source node s_i and destination node d_i .

For each user *i*, the cooperative transmission consists of two phases. In Phase 1, source s_i broadcasts its information to both destination d_i and the relay *r*. The received signals Y_{s_i,d_i} and $Y_{s_i,r}$ at destination d_i and relay *r* are given by

$$Y_{s_i,d_i} = \sqrt{P_{s_i} G_{s_i,d_i}} X_{s_i} + n_{d_i}$$
(12.47)

and

$$Y_{s_{i},r} = \sqrt{P_{s_{i}}G_{s_{i},r}}X_{s_{i}} + n_{r}, \qquad (12.48)$$

where P_{s_i} represents the transmit power of source s_i , X_{s_i} is the transmitted information symbol with unit energy at Phase 1 at source s_i , G_{s_i,d_i} and $G_{s_i,r}$ are the channel gains from s_i to destination d_i and to relay r, respectively, and n_{d_i} and n_r are additive white Gaussian noises. Without loss of generality, we assume that the noise level σ^2 is the same for all of the links. We also assume that the channels are stable over each transmission frame.

The signal-to-noise ratio (SNR) at destination d_i in Phase 1 is

$$\Gamma_{s_i,d_i} = \frac{P_{s_i} G_{s_i,d_i}}{\sigma^2}.$$
(12.49)

For AF cooperative transmission, in Phase 2 relay *r* amplifies $Y_{s_i,r}$ and forwards it to destination d_i with transmitted power P_{r,d_i} . The received signal at destination d_i is

$$Y_{r,d_i} = \sqrt{P_{r,d_i} G_{r,d_i}} X_{r,d_i} + n'_{d_i}, \qquad (12.50)$$

where

$$X_{r,d_i} = \frac{Y_{s_i,r}}{|Y_{s_i,r}|}$$
(12.51)

is the unit-energy transmitted signal that relay *r* receives from source s_i in Phase 1, G_{r,d_i} is the channel gain from relay *r* to destination d_i , and n'_{d_i} is the received noise in Phase 2. Substituting (12.48) into (12.51), we can rewrite (12.50) as

$$Y_{r,d_i} = \frac{\sqrt{P_{r,d_i}G_{r,d_i}}(\sqrt{P_{s_i}G_{s_i,r}}X_{s_i,d_i} + n_r)}{\sqrt{P_{s_i}G_{s_i,r} + \sigma^2}} + n'_{d_i}.$$
 (12.52)

Using (12.52), the relayed SNR at destination d_i , with the help of the relay, is

$$\Gamma_{s_i,r,d_i} = \frac{P_{r,d_i} P_{s_i} G_{r,d_i} G_{s_i,r}}{\sigma^2 (P_{r,d_i} G_{r,d_i} + P_{s_i} G_{s_i,r} + \sigma^2)}.$$
(12.53)

If user *i* performs only the direct transmission in Phase 1 (i.e., not using the relay), it achieves a total information rate of

$$R_{s_{i},d_{i}} = W \log_{2} \left(1 + \Gamma_{s_{i},d_{i}} \right), \tag{12.54}$$

where W is the signal bandwidth. On the other hand, if user *i* performs the transmissions in both phases 1 and 2, it achieves a total information rate at the output, assuming maximal ratio combining, of

$$R_{s_i,r,d_i} = \frac{1}{2} W \log_2 \left(1 + \Gamma_{s_i,d_i} + \Gamma_{s_i,r,d_i} \right).$$
(12.55)

The coefficient $\frac{1}{2}$ is used to model the fact that cooperative transmission will occupy one out of two phases (e.g., time, bandwidth, code). Since Γ_{s_i,r,d_i} is the extra SNR increase compared with the direct transmission, we also denote

$$\Delta \text{SNR}_i \triangleq \Gamma_{\mathbf{s}_i, \mathbf{r}, \mathbf{d}_i}. \tag{12.56}$$

Based on (12.54) and (12.55), the rate increase that user i obtains by cooperative transmission is

$$\triangle R_i = \max\{R_{s_i, r, d_i} - R_{s_i, d_i}, 0\}, \qquad (12.57)$$

which is non-negative since the source can always choose not to use the relay and thereby obtain zero rate increase. $\triangle R_i$ is a function of the channel gains of the source–destination, source–relay, and relay–destination links, as well as of the transmission power of the source and the relay. In particular, $\triangle R_i$ is a non-decreasing, non-smooth, and non-concave function of the relay transmission power P_{r,d_i} , as illustrated in Fig. 12.5.

We assume that the source transmission power P_{s_i} is fixed for each user *i*, as is the relay's total power *P*. The relay determines the allocation of its transmission power among users, $\mathbf{P}_r \triangleq (P_{r,d_1}, ..., P_{r,d_l})$, such that the total power constraint is not violated, i.e.,

$$\mathbf{P}_{r} \in \mathcal{P}_{r} \triangleq \left\{ \mathbf{P}_{r} \left| \sum_{i} P_{r,d_{i}} \leq P, P_{r,d_{i}} \geq 0, \forall i \in \mathcal{I} \right. \right\}.$$
(12.58)

We consider two different network objectives: *efficiency* and *fairness*. An efficient power allocation $\mathbf{P}_r^{\text{efficient}}$ maximizes the total rate increase of all users by solving the



Fig. 12.5 Rate increase as a function of relay transmission power.

following problem:

$$\max_{\mathbf{P}_r \in \mathcal{P}_r} \sum_{i \in \mathcal{I}} \triangle R_i (P_{r,d_i}).$$
(12.59)

In many cases, an efficient allocation discriminates against users who are far away from the relay. To avoid this, we also consider a fair power allocation $\mathbf{P}_r^{\text{fair}}$, which solves the following problem:

$$\begin{array}{l} \min_{\mathbf{P}_{r}\in\mathcal{P}_{r}} c \\ \text{s.t.} \quad \frac{\partial \bigtriangleup R_{i}(\bigtriangleup SNR_{i})}{\partial(\bigtriangleup SNR_{i})} = c \cdot \mathbf{1}_{\{\bigtriangleup SNR_{i}>0\}}, \forall i \in \mathcal{I}. \end{array}$$
(12.60)

Here $\mathbf{1}_{\{\cdot\}}$ is the indicator function. The idea behind (12.60) is that, for all users that choose to use the relay, the corresponding \triangle SNR_{*i*} should be maximized subject to the same marginal utility among these users. This can be translated into the minimization of the common marginal utility, because of the concavity of $\triangle R_i$ in terms of \triangle SNR_{*i*} (within the appropriate region). As an example, when Γ_{s_i,d_i} , the direct-transmission SNR, is the same for all users *i*, the constraint in (12.60) means that \triangle SNR_{*i*} is the same for all users with positive rate increases.

We notice that a fair allocation needs to be Pareto-optimal, i.e., no user's rate can be increased without decreasing the rate of another user. However, an efficient or fair allocation need not fully utilize the resource at the relay, i.e., $\sum_{i \in \mathcal{I}} P_{r,d_i}$ can be less than *P*. This could happen, for example, when a relay is far away from all users so that allowing it to transmit half of the time only decreases the total achievable rate. This is very different from most previous network resource-allocation problems (including [217]), in which the network performance is maximized only if the resource is fully utilized.

Since $\triangle R_i (P_{r,d_i})$ is non-smooth and non-concave, it is well known that (12.59) and (12.60) are NP-hard to solve in a centralized fashion. Next, we propose two auction mechanisms that can (approximately) solve these problems in a distributed fashion under suitable technical conditions.

12.3.2 Share-auction approach

An auction discussed in more detail in Chapter 8, is a decentralized market mechanism for allocating resources in an economy. An auction consists of three key elements: (1) the *good*, or the resource to be allocated; (2) an *auctioneer*, who determines the allocation of the good according to the auction rules; (3) a group of *bidders*, who want to obtain the good from the auctioneer. The interactions and outcome of an auction are determined by the *rules*, which include four components: (1) the *information* the auctioneer and bidders know before the auction starts; (2) the *bids* submitted to the auctioneer by the bidders; (3) the *allocation* determined by the auctioneer based on the bids; (4) the *payments* by the bidders to the auctioneer as functions of bids and allocations.

In the cooperative network considered here, it is natural to design auction mechanisms in which the *good* is the relay's total transmit power P, the auctioneer is the relay, and the bidders are the users. One well-known auction mechanism that achieves an efficient allocation is the Vickrey–Clarke–Groves (VCG) auction [256]. However, the VCG auction requires the relay to gather global network information from the users, and solves l + 1 non-convex optimization problems. This might be too complicated for real-time implementation. To overcome this limitation, we propose two simpler share auctions, the *SNR auction* and the *power auction*. The main advantages of the two proposed auctions are the simplicity of bids and allocation. The rules of the two auctions are described below, with the only difference being in payment determination.

- *Information*: Besides the public and local information (i.e., $W, P, \sigma^2, P_{s_i}, G_{s_i,d_i}$), each user *i* also knows the channel gains $G_{s_i,r}$ and G_{r,d_i} , either through measurement or by explicit feedback from relay *r*. The relay announces a positive *reserve bid* $\beta > 0$ and a *price* $\pi > 0$ to all users before the auction starts.
- *Bids*: User *i* submits bid $b_i \ge 0$ to the relay.
- Allocation: The relay allocates transmit power according to

$$P_{r,d_i} = \frac{b_i}{\sum_{j \in \mathcal{I}} b_j + \beta} P.$$
(12.61)

• *Payments*: In an SNR auction, source *i* pays the relay $C_i = \pi \triangle SNR_i$. In a power auction, source *i* pays the relay $C_i = \pi P_{r,d_i}$.

A bidding profile is defined as the vector $\mathbf{b} = (b_1, ..., b_l)$, containing the users' bids. The bidding profile of user *i*'s opponents is defined as $b_{-i} = (b_1, ..., b_{i-1}, b_{i+1}, ..., b_l)$, so that $\mathbf{b} = (b_i; b_{-i})$. User *i* chooses b_i to maximize its payoff,

$$U_i(b_i; b_{-i}, \pi) = \triangle R_i(P_{r,d_i}(b_i; b_{-i})) - C_i(b_i; b_{-i}, \pi).$$
(12.62)

For notational simplicity, we omit the dependence on β and other system parameters.

If the reserve bid $\beta = 0$, then the resource allocation in (12.61) depends only on the ratio of the bids. A bidding profile $k\mathbf{b}$ (for any k > 0) leads to the same resource allocation as **b**, which is not desirable in practice. That is why we need a positive reserve bid. However, the value of β is not important as long as it is positive. For example, if we increase β to $k'\beta$, then users can just scale **b** to $k'\mathbf{b}$, which leads to the same resource allocation. For simplicity, we will choose $\beta = 1$ in all the simulations.

The desirable outcome of an auction is a *Nash equilibrium* (NE), which is a bidding profile \mathbf{b}^* such that no user wants to deviate unilaterally, i.e.,

$$U_{i}(b_{i}^{*}; b_{-i}^{*}, \pi) \ge U_{i}(b_{i}; b_{-i}^{*}, \pi), \forall i \in \mathcal{I}, \forall b_{i} \ge 0.$$
(12.63)

Define user *i*'s best response (for fixed b_{-i} and price π) as

$$\mathcal{B}_i(b_{-i},\pi) = \left\{ b_i \left| b_i = \arg\max_{\tilde{b}_i \ge 0} U_i\left(\tilde{b}_i; b_{-i},\pi\right) \right. \right\},\tag{12.64}$$

which in general could be a set. An NE is also a fixed-point solution of all users' best responses. We would like to answer the following four questions for both auctions: (1) When does an NE exist? (2) When is the NE unique? (3) What are the properties of the NE? (4) How can the NE be reached in a distributed fashion?

SNR auction

Let us first determine the users' best responses (e.g., (12.64)) in the SNR auction, which clearly depend on the price π . For each user *i*, there are two critical price values, $\underline{\pi}_i^s$ and $\hat{\pi}_i^s$, where

$$\underline{\pi}_{i}^{s} \triangleq \frac{W}{2\ln 2\left(1 + \Gamma_{s_{i},d_{i}} + \frac{PG_{r,d_{i}}P_{s_{i}}G_{s_{i},r}}{\left(P_{s_{i}}G_{s_{i},r} + PG_{r,d_{i}} + \sigma^{2}\right)\sigma^{2}}\right)},$$
(12.65)

and $\hat{\pi}_i^s$ is the smallest positive root of

$$g_i^s(\pi) \triangleq \pi \left(1 + \Gamma_{s_i, d_i}\right) - \frac{W}{2} \left(\log_2 \left(\frac{2\pi \ln 2}{W} \left(1 + \Gamma_{s_i, d_i}\right)^2\right) + \frac{1}{\ln 2}\right). \quad (12.66)$$

Both $\underline{\pi}_i^s$ and $\hat{\pi}_i^s$ can be calculated locally by user *i*.

THEOREM 12.1 In an SNR auction, user i's unique best-response function is

$$\mathcal{B}_{i}(b_{-i},\pi) = f_{i}^{s}(\pi)(b_{-i}+\beta).$$
(12.67)

If $\hat{\pi}_i^s > \underline{\pi}_i^s$, then

$$f_{i}^{s}(\pi) = \begin{cases} \infty, & \pi \leq \underline{\pi}_{i}^{s}, \\ \frac{\left(P_{s_{i}}G_{s_{i},r} + \sigma^{2}\right)\sigma^{2}}{\frac{PG_{r,d_{i}}P_{s_{i}}G_{s_{i},r}}{\frac{W}{2\pi \ln 2} - 1 - \Gamma_{s_{i},d_{i}}} - \left(P_{s_{i}}G_{s_{i},r} + PG_{r,d_{i}} + \sigma^{2}\right)\sigma^{2}, & \pi \in \left(\underline{\pi}_{i}^{s}, \hat{\pi}_{i}^{s}\right), \\ 0, & \pi \geq \hat{\pi}_{i}^{s}. \end{cases}$$

$$(12.68)$$

If $\hat{\pi}_i^s < \underline{\pi}_i^s$, then $f_i^s(\pi) = \infty$ for $\pi < \hat{\pi}_i^s$ and $f_i^s(\pi) = 0$ for $\pi \ge \hat{\pi}_i^s$.

First consider the case in which $\hat{\pi}_i^s > \underline{\pi}_i^s$, where $\mathcal{B}_i(b_{-i}, \pi)$ is illustrated in Fig. 12.6. The price $\hat{\pi}_i^s$ determines when it is beneficial for user *i* to use the relay. With any price larger than $\hat{\pi}_i^s$, user *i* cannot obtain a positive payoff from the auction no matter what bid it submits, and thus it should simply use direct transmission and achieve a rate of R_{s_i,d_i} . As a result, $\mathcal{B}_i(b_{-i},\pi)$ is discontinuous at $\hat{\pi}_i^s$. When $\pi \in (\underline{\pi}_i^s, \hat{\pi}_i^s)$, user *i* wants to participate in the auction, and its best response depends on how much other users bid (b_{-i}) . When the price is smaller than $\underline{\pi}_i^s$, user *i* becomes so aggressive that it demands a large SNR increase that could not be achieved even of all the resource were allocated to it. This is reflected in an infinite bid in (12.68). Now consider the case in which $\hat{\pi}_i^s < \underline{\pi}_i^s$. User *i* either cannot obtain a positive payoff or cannot achieve the desired SNR increase, and thus the best response is either 0 or ∞ .

Combining (12.61) and (12.68), we know that if an NE exists, the relay power allocated for user i is

$$P_{r,d_i}(\pi) = \frac{f_i^s(\pi)}{f_i^s(\pi) + 1} P,$$
(12.69)

and $\sum_{i \in \mathcal{I}} \frac{f_i^s(\pi)}{f_i^s(\pi)+1} < 1$. The strict inequality is due to the positive reserve bid β .

Next we need to find the fixed point of all users' best responses, i.e., the NE. A trivial case would be $\hat{\pi}_i^s \le \underline{\pi}_i^s$ for all users *i*, in which case there exists a unique all-zero NE, $\mathbf{b}^* = \mathbf{0}$. The more interesting case would be the following:

DEFINITION 12.1 A network is SNR-regular if there exists at least one user i such that $\hat{\pi}_i^s > \underline{\pi}_i^s$.

THEOREM 12.2 Consider an SNR auction in an SNR-regular network. There exists a threshold price π_{th}^s such that a unique NE exists if $\pi > \pi_{th}^s$; otherwise no NE exists.

Unlike the result in [217], the unique NE in Theorem 12.2 might not be a continuous function of π , because of the discontinuity of the best-response function, as shown in Fig. 12.6. This observed in the simulation results to be discussed later. In particular, the unique NE could be all-zero for any price $\pi > \pi_{th}^s$, even if the network is SNR-regular.

It can be seen that the "marginal utility equalization" property of a fair allocation (i.e., the constraint in (12.60)) is satisfied at the NE of the SNR auction. However, there always exists some "resource waste" since some power will never be allocated to any user because of the positive reserve bid β . However, by choosing a price π larger than, but very close to, π_{th}^{s} , we could reduce the resource waste to a minimum and approximate the fair allocation. Formally, we define a reduced feasible set parameterized by δ as

$$\mathcal{P}_{r}^{\delta} \triangleq \left\{ \mathbf{P}_{r} \left| \sum_{i} P_{r,d_{i}} \leq P(1-\delta), P_{r,d_{i}} \geq 0, \forall i \in \mathcal{I} \right. \right\}.$$
(12.70)





Then we can show the following:

THEOREM 12.3 Consider an SNR auction in an SNR-regular network, where $f_i^s(\pi)$ is continuous at π_{th}^s for each user *i*, and greater than zero for at least one user. For any sufficiently small δ , there exists a price $\pi^{s,\delta}$ under which the unique NE achieves the fair allocation \mathbf{P}_r^{fair} with a reduced feasible set \mathcal{P}_r^{δ} .

A sufficiently small δ makes sure that we deal with a regime in which $f_i^s(\pi)$ is continuous for any user *i*. This is also desirable in practice since we want to minimize the amount of resource wasted.

Power auction

The best-response function in a power auction is nonlinear and complicated in general. However, in the special case of low SNR where Γ_{s_i,d_i} and \triangle SNR_{*i*} (b_i, b_{-i}) are small for all *i*, i.e.,

$$W\log_2\left(1+\Gamma_{s_i,d_i}+\bigtriangleup\operatorname{SNR}_i\left(b_i,b_{-i}\right)\right) \approx \frac{W}{\ln 2}\left(\Gamma_{s_i,d_i}+\bigtriangleup\operatorname{SNR}_i\left(b_i,b_{-i}\right)\right), \quad (12.71)$$

 $\mathcal{B}_i(b_{-i},\pi)$ has a linear form similar to that in (12.68). For each user, we can similarly define $f_i^p(\pi), \underline{\pi}_i^p, \hat{\pi}_i^p$, and $g_i^p(\pi)$ as in the SNR-auction case. One key difference here is that the value of $\hat{\pi}_i^p$ depends on the relationship between G_{s_i,d_i} and $G_{s_i,r}$. If $G_{s_i,d_i} > G_{s_i,r}$, then $\hat{\pi}_i^p = 0$ and user *i* never uses the relay. If $G_{s_i,d_i} < G_{s_i,r}$, then $\hat{\pi}_i^p$ is the smallest positive root of $g_i^p(\pi)$. Details are omitted because of space limitations.

In terms of the existence, uniqueness, and properties of the NE, we have the following:

DEFINITION 12.2 A network is power-regular if $\hat{\pi}_i^p > \underline{\pi}_i^p$ for at least one user *i*.

THEOREM 12.4 Consider a power auction in a power-regular network with low SNR. There exists a threshold price $\pi_{th}^{p} > 0$ such that a unique NE exists if $\pi > \pi_{th}^{p}$; otherwise no NE exists.

THEOREM 12.5 Consider a power auction in a power-regular network with low SNR, where $f_i^{p}(\pi)$ is continuous at π_{th}^{p} for each user *i*, and greater than zero for at least one user. For any sufficiently small δ , there exists a price $\pi^{p,\delta}$ under which the unique NE achieves the efficient allocation $\mathbf{P}_r^{efficient}$ with a reduced feasible set \mathcal{P}_r^{δ} .

Distributed iterative best-response updates

The final question we want to answer is how the NE can be reached in a distributed fashion. Consider the SNR auction as an example. It is clear that the best-response function in (12.68) can be calculated in a distributed fashion with limited information feedback from the relay. However, each user does not have enough information to calculate the best response of other users, which prevents it from directly calculating the NE. Nevertheless, the NE can be achieved in a distributed fashion if we allow the users to *iteratively* submit their bids based on best-response functions. Suppose users update their bids $\mathbf{b}(t)$ at time t according to the best-response functions as in (12.67), based on other users' bids $\mathbf{b}(t-1)$ in the previous time t-1, i.e.,

$$\mathbf{b}(t) = \mathbf{F}^{s}(\pi)\mathbf{b}(t-1) + \mathbf{f}^{s}(\pi)\beta, \qquad (12.72)$$

where both $\mathbf{b}(t)$ and $\mathbf{b}(t-1)$ are column vectors, $\mathbf{F}^{s}(\pi)$ is an *l*-by-*l* matrix whose (i, j)th component equals $f_{i}^{s}(\pi)$, and $\mathbf{f}^{s}(\pi) = [f_{1}^{s}(\pi), ..., f_{l}^{s}(\pi)]'$.

THEOREM 12.6 If there exists a unique non-zero NE in the SNR auction, the bestresponse updates in (12.72) globally and geometrically converge to the NE from any positive $\mathbf{b}(0)$.

Similar convergence results can be proved for the power auction.

Simulations

We first simulate various auction mechanisms for a two-user network. As shown in Fig. 12.7, the locations of the two sources (s_1 and s_2) and two destinations (d_1 and d_2) are fixed at (200m, -25m), (0m, 25m), (0m, -25m), and (200m, 25m), respectively. We fix the x coordinate of the relay node r at 80m and its y coordinate varies within the range [-200m, 200m]. In the simulation, the relay moves along a line. The propagation loss factor is set to 4, and the channel gains are distance-based (i.e., time-varying fading is not considered here). The source transmit power is $P_{s_i} = 0.01$ W for all user *i*, the noise level is $\sigma^2 = 10^{-11}$ W, and the bandwidth is W = 1 MHz. The total power of the relay node is set to P = 0.1 W.

In Fig. 12.8, we show the total rate increases achieved by two users in three auctions. The VCG auction achieves the efficient allocation by solving three non-convex optimization problems by the relay. For both the SNR auction and the power auction, the resource allocation depends on the choice of price π (but is independent of the reserve bid β). Every point on the curve represents an allocation in which the price is adjusted so that the total resource allocated to both users is more than 0.99*P* (unless this is not







Fig. 12.8 Total rate increase vs. relay location (*y*-axis) for three auctions.

possible). The power auction achieves a performance very close to that of the VCG auction. At those locations where the VCG auction achieves a positive rate increase, the power auction achieves a rate increase with an average of 95 percent of that achieved by the VCG auction. The SNR auction achieves smaller total rate increases but leads to fair resource allocations when both users use the relay (as can be seen in Fig. 12.9).

In Fig. 12.9, we show the individual rate increases of both users in the SNR auction and the power auction. The individual rate increases in the VCG auction are similar to those in the power auction and thus are not shown here. First consider the power auction. Since the relay-movement trajectory is relatively closer to source s_2 than to source s_1 , user 2 achieves an overall better performance compared with user 1. In particular, user 2 achieves a peak rate increase of 1.35 bits Hz⁻¹ when the relay is at location 25m (*y*-axis), compared with a peak rate increase of 0.56 bits Hz⁻¹ achieved by user 1 when the relay is at location -25m. Things are very different in an SNR auction, where the resource allocation is fair. In particular, since the distance between source and destination is the same for both users in our simulation, both users achieve the same positive rate increases when they both use the relay. This is the case when the relay is between locations -60mand 10m. At other locations, users just choose not to use the relay since they cannot get equal rate increases while also obtaining a positive payoff. This shows the tradeoff between efficiency and fairness.

Summary

Cooperative transmission can greatly improve communication system performance by taking advantage of the broadcast nature of wireless channels and cooperation among



Fig. 12.9 Individual rate increase vs. relay location (*y*-axis) for the SNR auction and the power auction.

users. We have studied two share-auction mechanisms, the SNR auction and the power auction, to distributively coordinate the relay power allocation among users. The existence and uniqueness of the Nash equilibrium are proven in both auctions. Under suitable conditions, the SNR auction achieves a fair allocation, while the power auction achieves a more efficient allocation. Simulations for both two-user and multiple-user networks have been used to demonstrate the effectiveness of the auction mechanisms. In particular, the power auction achieves an average of 95 percent of the maximum rate in the two-user case under a wide range of relay locations, and the SNR auction leads to a performance improvement having small variation among users.

12.4 Cooperative transmission using a cooperative game in MANET

In wireless packet-forwarding networks with selfish nodes, applications of a repeated game can induce the nodes to forward each other's packets, so that the network performance can be improved. This cooperation arises because nodes depend on each other for packet forwarding. However, the nodes on the boundary of such networks cannot benefit from this strategy, as the other nodes do not depend on them. This problem is sometimes known as the *curse of the boundary nodes*. In this section, following [189], an approach to this problem based on coalitional games is discussed, in which the boundary nodes can use cooperative transmission to help the backbone nodes in the middle of

the network. In return, the backbone nodes are willing to forward the boundary nodes' packets. We will discuss the stability of such coalitions using the concept of a core. Then two types of fairness, namely the min-max fairness using nucleolus and the average fairness using the Shapley function, are described. Finally, we discuss a protocol that uses both repeated games and coalitional games. Simulation results show how boundary nodes and backbone nodes form coalitions according to different fairness criteria. As we will see, the proposed protocol can improve network connectivity by about 50 percent, compared with pure repeated-game schemes.

12.4.1 Selfishness in packet-forwarding networks

In wireless networks with selfish nodes, such as ad hoc networks, the nodes may not be willing to fully cooperate to accomplish the overall network goals. Specifically for the packet-forwarding problem, forwarding of other nodes' packets consumes a node's limited battery energy. Therefore, it may not be in a node's best interest to forward others' arriving packets. However, refusal to forward others' packets non-cooperatively will severely affect network functionality and thereby impair a node's own performance. Hence, it is crucial to design a mechanism to enforce cooperation for packet forwarding among greedy and distributed nodes.

The packet-forwarding problem in ad hoc networks has been extensively studied in the literature. The fact that nodes act selfishly to optimize their own performance has motivated many researchers to apply game theory [377, 347] in solving this problem. Broadly speaking, the approaches used to encourage packet forwarding can be categorized into two general types. The first type makes use of virtual payments. Pricing [117] and credit-based methods [536] fall into this first type. The second type of approach is related to personal and community enforcement to maintain long-term relationships among nodes. Cooperation is sustained because defection against one node causes personal retaliation or sanction by others. Watchdog and pathrater are proposed in [320] to identify misbehaving nodes and deflect traffic around them. Reputation-based protocols are proposed in [86] and [337]. In [33], a model is considered to show cooperation among participating nodes. Packet-forwarding schemes using Tit-for-Tat schemes are proposed in [464]. In [182], a cartel-maintenance framework is constructed for distributed rate control for wireless networks. In [188], self-learning repeated-game approaches are constructed to enforce and improve cooperation. Some recent work using game theory to enhance energy-efficient behavior in infrastructure networks can be found in [334, 331, 332, 333].

A wireless packet-forwarding network can be modeled as a directed graph G(L, A), where *L* is the set of all nodes and *A* is the set of all directed links $(i, l), i, l \in L$. Each node *i* has several transmission destinations which are included in set D_i . To reach destination *j* in D_i , the available routes form a *depending graph* G_i^j whose nodes represents the potential packet forwarding nodes. The transmission from node *i* to node *j* depends on a subset of the nodes in G_i^j for packet forwarding. Notice that this dependency between nodes can be mutual. In general, this mutual dependency is common, especially for backbone nodes at the center of the network. Next, we will discuss how to make use of this mutual dependency for packet forwarding using a repeated game, and then examine the curse of the boundary nodes.

Repeated games for mutually dependent nodes

A repeated game is a special type of dynamic game, one that is played multiple times. When nodes interact by playing a static game numerous times, the game is called a repeated game. Unlike a static game, a repeated game allows a strategy to be contingent on past moves, thus allowing reputation effects and retribution, which give possibilities for cooperation. The game is defined as follows:

DEFINITION 12.3 A *T*-period repeated game is a dynamic game in which, at each period t, the moves during periods 1, ..., t - 1 are known to every node. In such a game, the total discounted payoff for each node is $\sum_{t=1}^{T} \beta^{t-1} u_i(t)$, where $u_i(t)$ denotes the payoff to node i at period t, and β is a discount factor. β is a measure of $0 \le \beta \le 1$ the node's patience or on the other hand how important the past affects the current payoff. If $T = \infty$, the game is referred as an infinitely repeated game. The average payoff u_i to node i is then

$$u_{i} = (1 - \beta) \sum_{t=1}^{\infty} \beta^{t-1} u_{i}(t).$$
(12.73)

It is known that repeated games can be used to induce greedy nodes in communication networks to show cooperation. In packet-forwarding networks, if a greedy node does not forward the packets of other nodes, it can enjoy benefits such as power saving. However, this node will be punished by the other nodes in the future if it depends on the other nodes to forward its own packets. The benefit of greediness in the short term is offset by the loss associated with punishment in the future. So the nodes will prefer to act cooperatively if they are sufficiently patient. From the Folk Theorem, below, we infer that in an infinitely repeated game, any feasible outcome that gives each node a better payoff than the Nash equilibrium [377, 347] can be obtained.

THEOREM 12.7 (Folk Theorem [377, 347]) Let $(\hat{u}_1, ..., \hat{u}_L)$ be the set of payoffs from a Nash equilibrium and let $(u_1, ..., u_L)$ be any feasible set of payoffs. There exists an equilibrium of the infinitely repeated game that attains any feasible solution $(u_1, ..., u_L)$ with $u_i \ge \hat{u}_i, \forall i$ as the average payoff, provided that β is sufficiently close to 1.

In the literature of packet-forwarding wireless networks, the Folk Theorem is demonstrated in several approaches. Tit-for-Tat [33, 464] is proposed so that all mutually dependent nodes have the same set of actions. A cartel-maintenance scheme [182] has closed-form optimal solutions for both cooperation and non-cooperation. A self-learning repeated-game approach is proposed in [188] for individual distributed nodes to study the cooperation points and to develop protocols for maintaining them.

Curse of the boundary nodes

However, packet-forwarding networks display the so-called *curse of the boundary nodes*. The nodes at the boundary of the network depend on the backbone nodes in the middle



Fig. 12.10 Example of the curse of the boundary nodes.

of the network to forward their packets. On the other hand, the backbone nodes do not depend on the boundary nodes. As a result, the backbone nodes do not worry about retaliation or lost reputation as a result of not forwarding the packets of the boundary nodes. An example is shown in Fig. 12.10. Suppose node 1 needs to send data to node 3, and node 2 needs to send data to node 0. Because node 1 and node 2 depend on each other for packet forwarding, each is obliged to do so because of the possible threat of retaliation from the other. However, if node 0 wants to transmit to node 2 and node 3, or node 3 tries to communicate with node 0 and node 1, the nodes in the middle have no incentive to forward the packets because of their greediness. Moreover, this greediness cannot be punished in the future since the dependency is not mutual. This problem is especially severe for the nodes on the boundary of the network.

On the other hand, if node 0 can form a coalition with node 1 and help node 1's transmission (for example by reducing the transmit power of node 1), then node 1 has an incentive to help node 0 transmit. A similar incentive arises for node 3 to form a coalition with node 2. We call nodes like 1 and 2 *backbone nodes*, while nodes like 0 and 3 are *boundary nodes*. In the following subsection, we examine how coalitions can be formed to address the above issue using cooperative transmission.

12.4.2 Cooperative transmission using a coalitional game

In this subsection, we first study a cooperative-transmission technique that allows nodes to participate in coalitions. Then, we formulate a coalitional game with cooperative transmission. Furthermore, we investigate the fairness issue and consider two types of fairness definitions. Finally, a protocol for packet-forwarding using repeated games and coalitional games is constructed.

Coalitional game formation for boundary nodes

Here we study possible coalitions between the boundary nodes and the backbone nodes, in situations where the boundary nodes can help relay information of the backbone nodes using cooperative transmission. We first define some basic concepts.

DEFINITION 12.4 A coalition S is defined as a subset of the total set of nodes $\mathbb{N} = \{0, ..., N\}$. The nodes in a coalition want to cooperate with each other. The coalition

form of a game is given by the pair (\mathbb{N}, v) , where v is a real-valued function called the characteristic function. v(S) is the value of the cooperation for coalition S, with the following properties:

- $v(\emptyset) = 0.$
- Super-additivity: if S and Z are disjoint coalitions $(S \cap Z = \emptyset)$, then $v(S) + v(Z) \le v(S \cup Z)$.

A coalition states the benefit obtained from cooperation agreements. However, we still need to examine whether the nodes are willing to participate in the coalition. A coalition is called *stable* if no other coalition will have the incentive and power to upset the cooperative agreement. Such a division of v is called a point in the *core*, which is defined as follows:

DEFINITION 12.5 A payoff vector $U = (U_0, ..., U_N)$ is said to be group-rational (or efficient) if $\sum_{i=0}^{N} U_i = v(\mathbb{N})$. A payoff vector U is said to be individually rational if the node can obtain a benefit which is not less than it would obtain if acting alone, i.e., $U_i \ge v(\{i\}), \forall i$. An imputation is a payoff vector satisfying the above two conditions. (Note that this definition is a restatement of Definitions 7.4–7.6 but in a different context.)

DEFINITION 12.6 An imputation U is said to be unstable through a coalition S if $v(S) > \sum_{i \in S} U_i$, i.e., the nodes have an incentive for coalition S and upset the proposed U. The set C of a stable imputation is called the core:

$$C = \{ U : \sum_{i \in \mathbb{N}} U_i = v(\mathbb{N}) \text{ and } \sum_{i \in S} U_i \ge v(S), \forall S \in \mathbb{N} \}.$$
(12.74)

In the economics literature, the core gives a reasonable set of possible shares. A combination of shares is in the core if there is no subcoalition in which its members may gain a higher total outcome than the combination of shares of concern. If a share is not in the core, some members may be frustrated and may think of leaving the whole group with other members to form a smaller group.

In the packet-forwarding network shown in Fig. 12.11, we first assume one backbone node to be the source node (node 0) and the nearby boundary nodes (node 1 to node N) to be relay nodes. (We will discuss the case of multiple source nodes later.) If no



Fig. 12.11 Cure for the curse of the boundary nodes.

cooperative transmission is employed, the utilities for the source node and the relay nodes are

$$v(\{0\}) = -P_d$$
, and $v(\{i\}) = -\infty, \forall i = 1, ..., N.$ (12.75)

With cooperative transmission and a grand coalition that includes all nodes, the utilities for the source node and the relay nodes are

$$U_0 = -P_0 - \sum_{i=1}^{N} \alpha_i P_d, \qquad (12.76)$$

$$U_i = -\frac{P_i}{\alpha_i},\tag{12.77}$$

where $\alpha_i \ge 0$ is the ratio of the number of packets that the backbone node is willing to forward for boundary node *i*, to the number of packets that the boundary node *i* relays for the backbone node using cooperative transmission. Here we use negative power as the utility so as to be consistent with conventions in game-theory literature. Smaller α_i means the boundary nodes have to relay more packets before realizing the rewards of packet forwarding. The other interpretation of the utility is as the average power per transmission for the boundary nodes.¹ The following theorem gives conditions under which the core is not empty, i.e, in which the grand coalition is stable:

THEOREM 12.8 The core is not empty if $\alpha_i \ge 0$, i = 1, ..., N, and α_i are such that $U_0 \ge v(\{0\})$, *i.e*,

$$\sum_{i=1}^{N} \alpha_i \le \frac{P_d - P_0}{P_d}.$$
(12.78)

Proof First, any relay node will get $-\infty$ utility if it leaves the coalition with the source node, so no node has an incentive to leave the coalition with node 0. The inclusion of relay nodes will increase the received SNR monotonically, so P₀ will decrease monotonically with the addition of any relay node. As a result, the source node has an incentive to include all the relay nodes, as long as the source power can be reduced, i.e., $U_0 \ge v(\{0\})$. A grand coalition is formed and the core is not empty if (12.78) holds.

The concept of the core defines the stability of a utility allocation. However, it does not define how to allocate the utility. For the proposed game, each relay node can obtain different utilities by using different values of α_i . Next, we study how to achieve min-max fairness and average fairness.

Min-max fairness of a game coalition using the nucleolus

We introduce the concepts of *excess*, *kernel*, and *nucleolus* [377, 347]. For a fixed characteristic function v, an imputation **U** is found such that, for each coalition S and its

¹ Notice that we omit the transmitted power needed to send the boundary node's own packet to the backbone node, since it is irrelevant to the coalition.

associated dissatisfaction, an optimal imputation is calculated to minimize the maximum dissatisfaction. The dissatisfaction is quantified as follows:

DEFINITION 12.7 The measure of dissatisfaction of an imputation U for a coalition S is defined as the excess:

$$e(U, S) = v(S) - \sum_{j \in S} U_j.$$
 (12.79)

Obviously, any imputation U is in the core if and only if all its excesses are negative or zero.

DEFINITION 12.8 A kernel of v is the set of all allocations U such that

$$\max_{S\subseteq\mathbb{N}-j,i\in S} e(U,S) = \max_{T\subseteq\mathbb{N}-i,j\in T} e(U,T).$$
(12.80)

If nodes i and j are in the same coalition, then the highest excess that i can make in a coalition without j is equal to the highest excess that j can make in a coalition without i.

DEFINITION 12.9 The nucleolus of a game is the allocation U that minimizes the maximum excess:

$$\boldsymbol{U} = \arg\min_{\boldsymbol{U}}(\max \ \boldsymbol{e}(\boldsymbol{U}, \boldsymbol{S}), \ \forall \boldsymbol{S}). \tag{12.81}$$

The nucleolus of a game has the following property: in coalitional form, it exists and is unique. The nucleolus is group-rational and individually rational. If the core is not empty, the nucleolus is in the core and kernel. In other words, the nucleolus is the best allocation under the min-max criterion.

Using the above concepts, we prove the following theorem to show that the optimal α_i in (12.76) have min-max fairness.

THEOREM 12.9 The maximal α_i to yield the nucleolus of the proposed coalitional game is given by

$$\alpha_i = \frac{P_d - P_0(\mathbb{N})}{NP_d},\tag{12.82}$$

where $P_0(\mathbb{N})$ is the required transmitted power of the source when all relays transmit with transmitted power P_{max} .

Proof Since for any coalition other than the grand coalition, the excess will be $-\infty$, we need only consider the grand coalition. Suppose the min-max utility is μ for all nodes, i.e.,

$$\mu = -\frac{P_i}{\alpha_i}.\tag{12.83}$$

From (12.78) and since U_i is monotonically increasing with α_i in (12.77), we have

$$\alpha_i = \frac{P_i}{\sum_{i=1}^N P_i} \cdot \frac{P_d - P_0}{P_d}.$$
(12.84)

Since P_0 for SNR is a monotonically increasing function of P_i , to achieve the maximal α_i and μ , each relay transmits with the largest possible power P_{max} . Notice here that we assume the backbone node can accept an arbitrarily small power gain to join the coalition.

Average fairness of a game coalition using the Shapley function

The core concept defines the stability of an allocation of payoff, and the nucleolus concept quantifies the min-max fairness of a game coalition. Next, we study another average measure of fairness for each individual, the Shapley function [377, 347].

DEFINITION 12.10 A Shapley function ϕ is a function that assigns to each possible characteristic function v a vector of real numbers, i.e.,

$$\phi(\mathbf{v}) = (\phi_0(\mathbf{v}), \phi_1(\mathbf{v}), \phi_2(\mathbf{v}), \dots, \phi_N(\mathbf{v})), \qquad (12.85)$$

where $\phi_i(v)$ represents the worth or value of node *i* in the game. There are four Shapley axioms that $\phi(v)$ must satisfy:

- 1. Efficiency axiom: $\sum_{i \in \mathbb{N}} \Phi_i(v) = v(\mathbb{N})$.
- 2. Symmetry axiom: If node *i* and node *j* are such that $v(S \bigcup \{i\}) = v(S \bigcup \{j\})$ for every coalition *S* not containing node *i* and node *j*, then $\phi_i(v) = \phi_j(v)$.
- 3. Dummy axiom: If node *i* is such that $v(S) = v(S \bigcup \{i\})$ for every coalition S not containing *i*, then $\phi_i(v) = 0$.
- 4. Additivity axiom: If u and v are characteristic functions, then $\phi(u+v) = \phi(v+u) = \phi(u) + \phi(v)$.

There exists a unique function ϕ_i , the Shapley function, satisfying the Shapley axioms. Moreover, the Shapley function can be calculated as

$$\phi_i(\mathbf{v}) = \sum_{S \subset \mathbb{N} - i} \frac{(|S|)!(N - |S|)!}{(N+1)!} [\mathbf{v}(S \cup \{i\}) - \mathbf{v}(S)].$$
(12.86)

Here |S| *denotes the cardinality of set* S *and* $\mathbb{N} = \{0, 1, ..., N\}$ *.*

The Shapley function can be interpreted as follows. Suppose one backbone node plus N boundary nodes form a coalition. The nodes join the coalition in random order, so there are (N+1)! different ways that the nodes might be ordered in joining the coalition. For any set S that does not contain node i, there are |S|!(N - |S|)! different ways to order the nodes so that S is the set of nodes who enter the coalition before node i. Thus, if the various orderings are equally likely, |S|!(N - |S|)!/(N + 1)! is the probability

that, when node *i* enters the coalition, the coalition of *S* is already formed. When node *i* finds *S* ahead of it as it joins the coalition, then its marginal contribution to the worth of the coalition is $v(S \cup \{i\}) - v(S)$. Thus, under the assumption of randomly ordered joining, the Shapley function of each node is its expected marginal contribution when it joins the coalition.

Here, we consider the case in which the backbone node is always in the coalition and the boundary nodes randomly join the coalition. We have $v(\{0\}) = -P_d$ and

$$\mathbf{v}(\mathbb{N}) = P_d - P_0(\mathbb{N}) - \sum_{i \in \mathbb{N}} \alpha_i P_d, \qquad (12.87)$$

which is the overall power saving. The problem here is to find a given node's α_i that satisfies the average fairness. This is addressed in the following theorem:

THEOREM 12.10 The maximal α_i that satisfies the average fairness with the physical meaning of the Shapley function is given by

$$\alpha_i = \frac{P_i^s}{P_d},\tag{12.88}$$

where P_i^s is the average power saving with random entering orders, given by

$$P_i^s = \frac{1}{N} [P_d - P_0(\{i\})] + \frac{\sum_{j=1, j \neq i}^{N} [P_0(\{j\}) - P_0(\{i, j\})]}{N(N-1)} + \dots$$
(12.89)

Proof The maximal α_i is determined by the following conditions:

$$\begin{cases} \frac{\alpha_i}{\alpha_j} = \frac{\phi_i}{\phi_j}, \\ \mathbf{v}(\mathbb{N}) \ge \mathbf{0}. \end{cases}$$
(12.90)

The first equation in (12.90) is the average fairness according to the Shapley function, and the second relation in (12.90) is the condition for a non-empty core. As with min-max fairness, we assume that the backbone node can accept arbitrarily small power gain to join the coalition.

If boundary node *i* is the first to join the coalition, the marginal contribution for power saving is $\frac{1}{N}[P_d - P_0(\{i\}) - \alpha_i P_d]$, where $\frac{1}{N}$ is the probability. If boundary node *i* is the second to join the coalition, the marginal contribution is $\frac{\sum_{j=1,j\neq i}^{N}[P_0(\{j\}) + \alpha_j P_d - P_0(\{i,j\}) - (\alpha_i + \alpha_j) P_d]}{N(N-1)}$. By means of some simple derivations, we can obtain the Shapley function ϕ_i as

$$\phi_i = -\alpha_i P_d + \frac{1}{N} [P_d - P_0(\{i\})] + \frac{\sum_{j=1, j \neq i}^N [P_0(\{j\}) - P_0(\{i, j\})]}{N(N-1)} + \cdots, (12.91)$$

Table 12.1 Packet-forwarding protocol using repeated games and coalitional games.

- 1. Route discovery for all nodes
- 2. Packet-forwarding enforcement by the backbone nodes, using threat of future punishment in repeated games
- 3. Neighbor discovery by the boundary nodes
- 4. Coalitional-game formation
- 5. Packet relay for the backbone nodes with cooperative transmission
- 6. Transmission of the boundary nodes' own packets to the backbone nodes for forwarding.

and then we can obtain

$$\alpha_{i} = \frac{[P_{d} - P_{0}(\mathbb{N})]P_{i}^{s}}{P_{d}\sum_{i=1}^{N}P_{i}^{s}}.$$
(12.92)

Since

$$P_d - P_0(\mathbb{N}) = \sum_{j=1}^N P_j^s,$$
 (12.93)

we prove (12.88).

Notice that different nodes have different values of P_i^s , owing to their channel conditions and their ability to reduce the backbone node's power. Compared with min-max fairness in the previous subsection, average fairness using the Shapley function gives different nodes different values of α_i according to their locations.

Using the above analysis, we now develop a packet-forwarding protocol based on repeated games and coalitional games, based on Table 12.1.

First, all nodes in the network undergo route discovery. Then each node knows who depends on it and on whom it depends for transmission. Using this route information, the repeated games can be formulated for the backbone nodes. The backbone nodes forward the other nodes' information, because of the threat of future punishment if these packets are not forwarded. Because of the network topology, some nodes' transmissions depend on the others while the others do not depend on these nodes. These nodes are most often located at the boundary of the network. In the next step, these boundary nodes try to find their neighboring backbone nodes, and try to form coalitions with the backbone nodes, so that the boundary nodes can be rewarded for transmitting their own packets. Cooperative transmission gives an opportunity for the boundary nodes to pay some "credits" first to the backbone nodes for the rewards of packet forwarding in return. On the other hand, competition among the backbone nodes prevents the boundary nodes from being forced to accept the minimal payoffs.

In the numerical study, we model all channels as additive white Gaussian noise channels having a propagation factor of 3; that is, power falls off spatially according to an inverse-cubic law. The maximal transmitted power is 10 dBm and the thermal noise



Fig. 12.12 (a) α for different channels and numbers of nodes, min-max fairness; (b) P_0 for different channels and numbers of nodes, min-max fairness.

level is -60 dBm. The minimal SNR γ is 10 dB. In the first setup, we assume the backbone node is located at (0m, 0m), and the destination is located at either (100m, 0m) or (50m, 0m). The boundary nodes are located on an arc with angles randomly distributed from 90° to 270° and with distances varying from 5 m to 100 m.

In Fig. 12.12(a), we study min-max fairness and show the average α_i over 1000 iterations as a function of distance from the relays to the source node. Because of

the min-max nature, all boundary nodes have the same α_i . When the distance is small, i.e., when the relays are located close to the source, α_i approaches $\frac{1}{N}$. This is because the relays can serve as virtual antennas for the source, and the source needs very low power for transmission to the relays. When the distance is large, the relays are less effective and α_i decreases, which means that the relays must transmit more packets for the source to earn the rewards of packet forwarding. When the destination is 50 m, the source–destination channel is better than that at 100 m. When N = 1 and the source–destination distance is 50 m, the relays close to the source have larger α_i and the relays farther away have lower α_i than in the 100 m case. In Fig. 12.12(b), we show the corresponding P_0 for the backbone node. We can see that P_0 increases when the distances between the boundary nodes and the backbone node increase.

If we consider the multiple-backbone (multiple-core) case with min-max fairness, Figs. 12.12(a) and 12.12(b) provide guidelines for the boundary nodes in selecting a backbone node with which to form a coalition. First, a less crowded coalition is preferred. Second, the nearest backbone node is preferred. Third, for N = 1, if the source-destination channel is good, the closer backbone node is preferred; otherwise, the farther one can provide larger α_i .

Next, we investigate average fairness using the Shapley function. The simulation setup is as follows. The backbone node is located at (0m,0m) and the destination is located at (-50 m, 0 m). Boundary node 1 is located at (20 m, 0 m) and (50 m, 0 m), respectively. Boundary node 2 moves from (5m,0m) to (100m,0m). The remaining simulation parameters are the same. In Fig. 12.13(a), we show maximal α_i for two boundary nodes. We can see that when boundary node 2 is closer than boundary node 1 to the backbone node, $\alpha_2 > \alpha_1$, i.e., boundary node 2 can help relay fewer packets for backbone node 1 before being rewarded. The two curves for α_1 and α_2 for the same boundary node 1 location cross at the boundary node 1 location. The figure shows that the average fairness using the Shapley function gives greater rewards to the boundary node whose channel is better and who can help the backbone node more. When boundary node 2 moves from (20 m, 0 m) to (50 m, 0 m), α_1 becomes smaller, but α_2 becomes larger. This is because the backbone node must depend more on boundary node 2 for relaying. However, the backbone node will pay less for the boundary nodes. Notice that α_i at the crossover point is lower. This is because the overall power for the backbone node is high when boundary node 2 is far away, as shown in Fig. 12.13(b).

We have studied a coalitional-game approach to providing benefits to selfish nodes in wireless packet-forwarding networks using cooperative transmission, so that the boundary nodes can transmit their packets effectively. We have used the concepts of coalitional games to maintain stable and fair game coalitions. Specifically, we have studied two fairness concepts: min-max fairness and average fairness. A protocol has been constructed using repeated games and coalitional games. From simulation results, we have seen how boundary nodes and backbone nodes form coalitions according to different fairness criteria. We can also see that network connectivity can be improved by about 50 percent, compared to the pure repeated game approach.



Fig. 12.13 (a) α_i of average fairness for different user locations; (b) P_0 of average fairness for different user locations.

12.5 Cooperative routing

In this section, we study the impact of cooperative communication on the network layer for general wireless networks. Then, we briefly discuss the current standard development for IEEE 802.16j WiMAX relay networks.

12.5.1 Cooperative-routing algorithms

In the network layer, routing algorithms select multi-hop links between a source and destination with minimal cost in terms such as overall power, or with maximal gain in terms such as throughput. For each hop, metrics can be defined to cover such information as bandwidth, delay, hop count, path cost, load, reliability, and communication cost. A routing metric is a value used by a routing algorithm to determine whether one route performs better than another. Then a routing algorithm, such as the shortest-path algorithm (e.g., Dijkstra's algorithm), can find the optimal route among all possible connections from the source to the destination.

With cooperative transmission, the cooperative-routing problem has been recently considered in the literature [518, 248, 290, 457, 306]. Most of the current cooperative-based routing algorithms, such as cooperation along the minimum-energy non-cooperative path (CAN) [248], progressive cooperation (PC) [248], and cooperative routing along truncated non-cooperative route (CTNCR) [518], are implemented in *two consecutive steps*. First, a non-cooperative route is constructed using any shortest-path routing algorithm. Second, cooperative-communication protocols are applied on some or all of the nodes along the established route. In fact, these routing algorithms do not fully exploit the merits of cooperative transmission, since the optimal cooperative routes might not be along non-cooperative routes.

One simple example is shown in Fig. 12.14, in which a regular grid topology is studied. To illustrate the routes selected by different routing schemes, we assume that the source is node 0 and the destination is node 7. The shortest-path routing algorithm chooses one of the possible shortest routes. For instance, the chosen shortest route is $\{(0,1), (1,5), (5,6), (6,7)\}$, where (i,j) denotes the direct-transmission mode from node *i* to node *j*. Figure 12.14(a) shows the route chosen by the shortest-path routing algorithm, where the solid line between each two nodes indicates the direct-transmission mode. The cooperative route based on the shortest-path algorithm applies cooperation among each three consecutive nodes on the shortest route, and the resulting route is $\{(0,1,5), (5,6,7)\}$, where (x, y, z) denotes the cooperative-transmission mode between sender *x*, relay *y*, and destination *z*. Figure 12.14(b) shows the route chosen by this routing algorithm.



Fig. 12.14 (a) Shortest-path route; (b) cooperative route based on shortest path; (c) optimal cooperative route.

dashed lines indicate the sender–relay and relay–receiver transmissions. Finally, we find that the optimal cooperative route is given by $\{(0,5,1), (1,2,6), (6,11,7)\}$, as shown in Fig. 12.14(c). In this example, we can see visually the difference between the routes chosen by the optimal cooperative-routing algorithm and by the cooperative-routing algorithm based on the shortest path.

The above example provides a motivation for proposing a one-step cooperativerouting algorithm, where the routing decision is based directly on cooperative transmission. The design process can be summarized as follows:

- The one-hop cost function is to be designed while considering cooperative transmission. The cost function can be the cost for traditional direct transmission, or the cost to the source and relays together for cooperative transmission. A particular quality of service (QoS) needs to be guaranteed. The major challenge for the cooperative cost function is the relay selection, which is a complicated integer-optimization problem. Some heuristics need to be designed to reduce computational complexity and still achieve near-optimal performance.
- 2. The optimal route is then defined as the route which requires the minimum overall cost. Any routing algorithms can be utilized to calculate the optimal route.

In [225], the one-hop cost is defined as the power while a certain bit error rate (BER) is ensured. The heuristic is to select only one relay. It is shown that this one-step algorithm outperforms the two-step algorithms [518, 248] by about 10 percent.

12.5.2 WIMAX IEEE 802.16j

Here, we discuss the impact of cooperative transmission on wireless-network standards. WiMAX, based on the IEEE 802.16 standard for wireless metropolitan area networks (WMANs), is expected to enable broadband speeds over wireless networks at a cost that enables mass-market adoption, and thereby make the vision of pervasive connectivity a reality. WMANs are designed for relatively large-scale networks such as a large corporate or university campus, or an entire city. The IEEE 802.16 standard has helped to pave the way for WMAN technology globally and, since its first inception, has been expanded considerably. Next, we discuss one of these expansions related to cooperative transmission.

Current deployments of IEEE 802.16 standards suffer from problems such as limited spectrum, low signal-to-interference-plus-noise ratio (SINR) at cell edges, coverage holes due to shadowing, and non-uniformly distributed traffic loads. To address these issues, the IEEE instituted work on the standard 802.16j mobile multi-hop relay (MMR) in 2006 [8, 4]. The basic idea behind MMR is to allow WiMAX base stations to impose a demanding performance requirement on relay stations (RSs). These relays will serve functionally as aggregating points on behalf of the base station (BS) for traffic collection from, and distribution to, multiple mobile stations (MSs) associated with the relays, and thus naturally incorporate a notion of "traffic aggregation." On the one hand, this approach will of course reduce the bandwidth available to users in the cells involved in relaying packets. On the other hand, it is an elegant way to reduce costs and extend



Fig. 12.15 Three types of IEEE 802.16j relay stations. Shaded areas indicate enhanced zones.

network coverage into areas where connecting a base station directly to the network via a fixed-line connection is economically or technically unfeasible. The goal is to enhance the coverage, throughput, and system capacity of existing IEEE 802.16 networks by specifying multi-hop relay capability and functionality of the relay stations and base stations. Some of the design requirements of IEEE 802.16 are as follows:

- · Backwards compatibility with the existing structure
- Definition of relay frequency and channel bandwidth
- Support from relays for network entry of mobile stations
- Support of QoS and hybrid automatic repeat request (ARQ)
- Support of handover and mobility
- · Deployment of multiple antennas for the relay link
- Support of multiple hops between base station and mobile station
- Enhancement of link reliability.

According to various network scenarios, there are three types of WiMAX relay stations, as follows (see Fig. 12.15):

- Fixed relay stations (FRSs): permanently installed at fixed locations
- Nomadic relay stations (NRSs): location fixed for periods of time, but can be moved around; used for situations such as special events
- Mobile relay stations (MRSs): for use in mobile environments.

It is anticipated that there will be no change to WiMAX subscriber devices used with the above WiMAX relay stations. However, 802.16 BS is being updated to support MMR functions and to be backward-compatible with the current version of WiMAX subscriber services. Figure. 12.16 shows the frame structure for IEEE 802.16 ji n order for the frames to transmit between the BS and the destination through the relay stations. In this example, a message from the BS can be forwarded to the destination through three relay stations. However, this brings the following challenges for the network design.


Fig. 12.16 IEEE 802.16j frame structure with relay stations.

- System configuration/management. To optimally deploy the relay stations, the network topology needs to be known. Moreover, for dynamic scenarios with mobile relay stations, neighbor detection is necessary. A relay path management process (such as a path-selection algorithm and path recovery) can eliminate coverage holes. For areas with high subscriber densities, congestion control at the network level needs to be implemented. Also, connection management determines how the BS can be connected to the destination nodes. Finally, QoS provisioning needs to be considered, especially for multimedia payloads.
- Network entry, bandwidth management, and scheduling. When a node enters the network, admission-control synchronization, ranging, and authorization need to be implemented. When nodes request bandwidth, the BS and RS need to perform bandwidth allocation. The scheduling of packets from the BS, RSs, and mobile users can be difficult to implement centrally if the number of relays is large. In this case, distributed scheduling is preferred.
- Data delivery. For MAC protocol data unit (PDU) processing, information is delivered as a unit among peer entities of a network and may contain control information, address information, or data. The transmitted data can be classified as unicast/multicast/broadcast data. To ensure successful delivery of data, ARQ or hybrid ARQ processes are needed. Finally, if the receiver can combine the data in different frames, cooperative-communication techniques can also be used to improve the communication links.

	Conventional repeater	MMR	Base station
Concept	Dummy repeater	Smart repeater	Radio access station
Function	Amplify and forward	Decode and forward	Encode and decode
Cost	Low	Reasonable	High
Coverage	Narrow-wide	Narrow-wide	Wide
Performance	Severe degradation	Mild degradation	No degradation
Resource management	Controlled by BS	Collaborative with BS	Self-controlled
Interference	Unmanaged	Managed by BS	Self-managed

• **Mobility management.** Algorithms need to be constructed for MS handover and MRS handover to support mobility. This can be intra-MMR-cell handover, in which only one MMR-BS participates, or inter-MMR-cell handover, which involves multiple MMR-BS.

A comparison of MMS with conventional repeaters and base stations is shown in Table 12.2.

In summary, 802.16j is under development for coverage extension and throughput enhancement for existing WiMAX. There are many open issues related to MMR systems, such as system configuration and management, network admission, bandwidth management, scheduling, data delivery, and mobility management.

12.6 Summary

Overall, cooperative communication has attracted considerable attention as a transmit technology for future wireless networks, as it efficiently takes advantage of the broadcast nature of wireless networks, and exploits inherent spatial and multi-user diversities by treating cooperative relays as virtual antenna arrays. Most work in this area focuses on how to improve link quality in the physical layer, while cross-layer issues have been less well studied.

The cross-layer impact of cooperative communication lies in the fact that it offers a new degree of freedom for traditional communication problems. For example, for power control, the relay's power can determine the performance at the destination. With different power-control strategies, rate adaptation can be adjusted to fully use spectrum. With limited spectrum and multiple users, relay selection and channel allocation address the problems of multiple access and spectrum access. Moreover, in the network layer, the routing metrics can be significantly different from traditional ones under cooperative transmission, since routes with cooperative users can greatly improve performance. The impact of cooperative communication can also be seen in the application layer, with applications such as cooperative video transmission [177, 271].

In this chapter, we have concentrated on exploring the impact of cooperative transmission using game theory. Specifically, we have examined the cases of ad hoc networks, sensor networks, and general networks. The key point to be made is that, by using cooperative transmission, new methods can emerge to improve network performance. Game theory provides a natural means for the distributed implementation of such network improvement. We have provided several examples of how to formulate the game for cooperative transmission over different layers to improve wireless network design. Cognitive radio [201] is a new paradigm for designing wireless communication systems which aims at enhancing the utilization of the radio-frequency spectrum. The motivation for cognitive radio arises from the scarcity of the available frequency spectrum. Emerging wireless applications will increase spectrum demand from mobile users. However, most of the available radio spectrum has been allocated to existing wireless systems, and only small portions of the radio spectrum can be licensed to new wireless applications. Nonetheless, a study in [143] by the spectrum policy task force (SPTF) of the Federal Communications Commission (FCC) shows that there are also many frequency bands which are only partly occupied or largely unoccupied. For example, spectrum bands allocated to cellular networks in the USA [325] reach their highest utilization during working hours, while they remain unoccupied from midnight until early morning.

The major factor that leads to inefficient use of radio spectrum is the spectrum licensing scheme itself. In traditional spectrum allocation, based on a command-and-control model, where radio spectrum allocated to licensed users (i.e., primary users) is not used, it cannot be utilized by unlicensed users (i.e., secondary users) or applications [87]. Because of this static and inflexible allocation, legacy wireless systems operate only on a dedicated spectrum band, and cannot adapt the transmission band according to the changing environment. For example, if one spectrum band is heavily used, they cannot change to other bands which are lightly used. Because of the current static spectrum licensing scheme, *spectrum holes* or *spectrum opportunities* arise (Fig. 13.1). Spectrum opportunities are defined as frequency bands, allocated to primary users, which at some locations and times are not utilized by these primary users, and could therefore be accessed by secondary users [201].

To improve the efficiency and utilization of the available spectrum, new spectrum licensing models have been introduced. The idea is to make spectrum access more flexible by allowing secondary users to access the radio spectrum under certain conditions. There are three major spectrum-licensing models [533]: common-use, exclusive-use, and shared-use models. In the common-use model, the spectrum is open to all users. This model is already in use in the ISM band [287], e.g., through the IEEE 802.11 standard. In the exclusive-use model, a primary user can grant access to a particular frequency band to a secondary user for a certain period of time [199]. This model is more flexible than the traditional command-and-control spectrum-licensing model, since the type of use and the licensee can be dynamically changed. In the



Fig. 13.1 Spectrum opportunities.

shared-use model, primary users are allocated frequency bands which are opportunistically accessed by secondary users. Spectrum access by secondary users is transparent to the primary users. There are two approaches to spectrum access in the shared-use model: *spectrum underlay* and *spectrum overlay*. Spectrum underlay constrains the transmission power of secondary users so that they operate below the interference limit for primary users. Spectrum overlay (or opportunistic spectrum access) allows secondary users to identify and exploit spectrum opportunities defined in space, time, and frequency.

However, since legacy wireless systems were designed to operate on dedicated frequency bands, they cannot utilize the improved flexibility of this shared-use spectrum licensing scheme. For this reason the concept of *cognitive radio* has been introduced. The main goal of cognitive radio is to provide adaptability to wireless transmission through dynamic spectrum access, implemented using software-defined radio (SDR) [340]. In this way, the performance of wireless transmission can be optimized, and the utilization of the frequency spectrum can be enhanced. The major functionalities of a cognitiveradio device include spectrum sensing, spectrum management, and spectrum mobility. Through spectrum sensing, information on the target radio spectrum (e.g., type and current activity of the primary user) is obtained. Different spectrum-sensing techniques have been developed, e.g., energy detection, matched-filter detection, cyclostationary detection, and wavelet detection [525]. This information is then exploited by the spectrum-management function to determine spectrum opportunities and make decisions on spectrum access. If the status of the target spectrum changes, the spectrum-mobility function can change the operational parameters (e.g., frequency bands).

Since with cognitive radio primary and secondary users will be more adaptive, they can adjust the transmission parameters for their benefit. Therefore, interaction among users becomes more important and affects transmission performance significantly. For example, in the spectrum-underlay approach, power control is important for secondary users, not only for maximizing the transmission rate but for maintaining interference below target levels. In this chapter, the following game models, developed to analyze the performance of cognitive-radio networks with rational primary and secondary users, are reviewed.

- **Cooperative spectrum sensing.** The performance of spectrum sensing can be improved by cooperation among secondary users who can form coalitions (i.e., groups) and share individual sensing results. With more information from the coalition, decisions on spectrum sensing can be made more accurate. A coalitional-game model is formulated to obtain stable coalitional structures (Section 13.1).
- **Power allocation/control.** As with the power-control problem in cellular (e.g., CDMA) systems, spectrum access based on the underlay approach requires secondary users to choose the proper transmit power to maximize spectrum utilization and transmission rate. Unlike that in the traditional cellular system, the power-control problem in cognitive radio imposes strict constraints on interference to the primary users. When secondary users are non-cooperative in choosing transmit power, a non-cooperative game can be modeled to obtain the Nash equilibrium, given the interference constraints (Section 13.2).
- Medium access control. For the spectrum-overlay approach based on the shareduse model, medium access control is important to secondary users for detecting and accessing spectrum opportunities. Especially in multi-channel environments, channel selection/allocation to avoid congestion among secondary users can be formulated as a non-cooperative game (Section 13.3). If secondary users are cooperative, channel selection/allocation can be formulated as a bargaining game to achieve an efficient and fair Nash bargaining solution.
- Decentralized dynamic spectrum access. Since a centralized controller may not exist in the cognitive-radio environment, rational secondary users have to implement a decentralized dynamic spectrum-access algorithm which takes account of historical information on the primary users and other secondary users. A decision to access the spectrum is then made accordingly (Section 13.4). With the ability to learn, this decentralized algorithm can converge to a correlated equilibrium.
- Cheat-proof strategies open spectrum sharing. Secondary users can maintain cooperation by reporting true private information for spectrum access to optimize performance. However, some secondary users may deviate from cooperation to gain a higher benefit. If secondary users interact among each other (e.g., to choose transmit power) repeatedly, a punishment mechanism can be implemented to avoid deviation. A repeated game can be applied to analyze the cooperation and deviation behavior of secondary users.
- **Spectrum leasing and cooperation.** Secondary users can help the primary user to relay data in order to improve performance (i.e., using cooperative diversity techniques). In return, the primary user may allow secondary users to transmit on its licensed spectrum. A Stackelberg game model is formulated for this situation, in which the primary user as leader chooses the size of the time-share for secondary users. The secondary users as followers choose the transmit power (from the primary user) so that benefit is maximized for the allocated time-share. Also, with multiple secondary users, a non-cooperative game model can be used to obtain the Nash equilibrium for transmit power (Section 13.7).
- Service provider competition for dynamic spectrum allocation. A service provider can buy spectrum from a regulator and use this spectrum to provide service to users.

A Stackelberg game model is formulated for this situation, in which the service provider as leader chooses the price to maximize profit. The users as followers then determine the demand which maximizes their utility, given the quality of service and price from the service provider (Section 13.8).

13.1 Cooperative spectrum sensing

In cognitive radio, cooperative spectrum sensing is developed to improve the performance of spectrum-opportunity detection [484]. In cooperative sensing, multiple secondary users/nodes collaborate by exchanging information (e.g., sensing results) on the spectrum owned by primary users/nodes. These results are combined to identify spectrum opportunities. Cooperative spectrum sensing can lead to more accurate decisions by secondary users than independent sensing, especially when the signal from a primary user shows fading or shadowing. One approach to cooperative spectrum sensing is based on a centralized fusion center, which collects individual sensing results from each secondary user in the network. The fusion center then processes these results, e.g., using the OR rule [491], to obtain a final sensing decision (e.g., on whether primary users are accessing the target spectrum). In addition, on the basis of spatial diversity in cooperative sensing, the error probability, due to fading on the control channel used to exchange sensing results between secondary users and the central fusion center, can be improved [530].

Group formation by secondary users for spectrum sensing and communication with a fusion center play an important role in the performance of cooperative sensing. For example, with the OR rule, if a group contains few secondary users, the detection probability (i.e., the probability that a primary user is reported to be accessing the channel) will be low. However, if a group contains many secondary users, although the detection probability is high, so is the false-alarm probability (i.e., the probability that the fusion center reports the detection of primary users when in fact no primary users are accessing the spectrum). This false-alarm event results in low utilization of the spectrum, since secondary users will defer spectrum access as the result of a wrong sensing decision. A coalitional-game model is formulated in [413] to obtain a stable group formation of secondary users for cooperative spectrum sensing. Also, a distributed algorithm based on split-and-merge is introduced and its stability analyzed.

In the rest of this chapter we explain in detail how game theory is applied to cognitiveradio networks. Some notation is summarized in Table 13.1.

13.1.1 System model

In a cognitive-radio network, there are *N* secondary users (i.e., transmitters) and a primary user (Fig. 13.2). The set of all secondary users is denoted by $\mathcal{N} = \{1, ..., N\}$. Each secondary user senses the target spectrum in order to detect the presence of the primary user. An energy detector is used [164]. For secondary user *i*, given Rayleigh channel fading, the individual detection probability P_i^{det} and individual false alarm probability

Notation	Description	
С	Set of channels	
δ	Discount factor	
F _m	Profit to service provider <i>m</i>	
h	Channel gain	
$\overline{\gamma}$	Average SNR	
\mathcal{M}	Set of primary users	
Ν	Total number of secondary users	
\mathcal{N}	Set of secondary users	
p_m	Spectrum price from service provider m	
Pi	Transmit power of secondary user <i>i</i>	
$P^{\sf suc}, P^{\sf col}$	Probabilities of successful transmission and collision	
φ	Learning rate	
σ^2	Noise power	
x	Channel allocation or channel-access decision	
у	Channel status (idle or occupied by primary user)	

Table 13.1 Notation in game models for cognitive-radio networks.





 P_i^{fal} are, from [413],

$$P_{i}^{\text{det}} = \exp\left(-\frac{\gamma_{\text{thr}}}{2}\right) \sum_{n=0}^{b-2} \frac{1}{n!} \left(\frac{\gamma_{\text{thr}}}{2}\right)^{n} + \left(\frac{1+\overline{\gamma}_{\mathsf{P},i}}{\overline{\gamma}_{\mathsf{P},i}}\right)^{b-1}$$
(13.1)

$$\times \left(\exp\left(-\frac{\gamma_{\text{thr}}}{2\left(1+\overline{\gamma}_{\mathsf{P},i}\right)}\right) - \exp\left(-\frac{\gamma_{\text{thr}}}{2}\right) \sum_{n=0}^{b-2} \frac{1}{n!} \left(\frac{\gamma_{\text{thr}}\overline{\gamma}_{\mathsf{P},i}}{2\left(1+\overline{\gamma}_{\mathsf{P},i}\right)}\right)^{2}\right),$$
(13.2)
$$P_{i}^{\text{fal}} = \frac{\Gamma(b,\gamma_{\text{thr}}/2)}{\Gamma(b)},$$
(13.2)

where *b* is the time-bandwidth product, γ_{thr} is the energy-detection threshold, $\Gamma(\cdot, \cdot)$ is the incomplete gamma function, and $\Gamma(\cdot)$ is the gamma function. $\overline{\gamma}_{P,i}$ indicates average SNR of the received signal from the primary user to secondary user. In this case, the

probability of misdetection by secondary user *i* is simply $P_i^{\text{mis}} = 1 - P_i^{\text{det}}$. In general, given a particular parameter of spectrum sensing (e.g., the threshold γ_{thr} of the energy detector), there is a tradeoff between detection probability and false-alarm probability. In particular, as the detection probability increases (e.g., through a decrease in the value of γ_{thr}), the false-alarm probability will increase. Although interference for the primary user decreases, spectrum utilization will be low. This issue has been investigated, e.g., in [491].

13.1.2 Coalitional-game formulation

To optimize the performance of cooperative spectrum sensing, secondary users (i.e., players) can form coalitions (Fig. 13.2). The decision by a secondary user to join or leave any coalition is the strategy. In each coalition $S \subset N$, there is a coalition head (i.e., the local fusion center), which collects sensing results from the secondary users in the same coalition head is the secondary user with the lowest misdetection probability P_i^{mis} . The error probability for the transmission of a sensing result from secondary user $i \in S$ to the coalition head $k \in S$ is, from [413],

$$P_{i,k}^{\text{err}} = \frac{1}{2} - \frac{1}{2} \sqrt{\frac{\overline{\gamma}_{i,k}}{1 + \overline{\gamma}_{i,k}}},$$
(13.3)

where $\overline{\gamma}_{i,k}$ is the average SNR in transmitting the sensing result to the coalition head. With cooperative spectrum sensing based on the OR rule, the coalition misdetection probability Q_S^{fal} for coalition S with coalition head k are

$$Q_{S}^{\text{mis}} = \prod_{i \in S} \left(P_{i}^{\text{mis}} (1 - P_{i,k}^{\text{err}}) + (1 - P_{i}^{\text{mis}}) P_{i,k}^{\text{err}} \right),$$
(13.4)

$$Q_{S}^{fal} = 1 - \prod_{i \in S} \left((1 - P_{i}^{fal})(1 - P_{i,k}^{err}) + P_{i}^{fal} P_{i,k}^{err} \right),$$
(13.5)

where the sensing result is assumed to be one bit of data. It is observed that as the number of secondary users in the coalition increases, the misdetection probability will decrease while the false-alarm probability will increase. Also, if the secondary users are independent, the distributed algorithm for forming a coalition would have to minimize the misdetection probability (or equivalently to maximize the detection probability) while the false-alarm probability is maintained below a threshold α .

The utility $U_i(S)$ of secondary user *i* in coalition *S* is assumed to be identical to the value v(S) of a coalition $S \subset N$. The utility and value are a function of the detection and false-alarm probabilities:

$$v(S) = Q_S^{det} - \mathscr{C}(Q_S^{fal}), \qquad (13.6)$$



Fig. 13.3 Example of logarithmic barrier-penalty cost function.

where $\mathscr{C}(Q_S^{fal})$ is a cost function of the false-alarm probability. This cost function is defined as the logarithmic barrier-penalty function:

$$\mathscr{C}(Q_{\mathcal{S}}^{\mathsf{fal}}) = \begin{cases} -\alpha^2 \log\left(1 - \left(\frac{Q_{\mathcal{S}}^{\mathsf{fal}}}{\alpha}\right)^2\right), & Q_{\mathcal{S}}^{\mathsf{fal}} < \alpha, \\ \infty, & Q_{\mathcal{S}}^{\mathsf{fal}} \ge \alpha. \end{cases}$$
(13.7)

Note that this cost function depends on both the channel quality between secondary and primary users and the number of secondary users in the coalition. An example of this cost function with $\alpha = 0.1$ is shown in Fig. 13.3. It is found in [413] that this coalitional game of cooperative spectrum sensing has a *non-transferable* utility.

To compare various coalition settings, a comparison relation is defined. Let $S = \{S_1, ..., S_l\}$ and $\mathcal{R} = \{R_1, ..., R_m\}$ be two collections (i.e., partitions) of the coalition. For $\mathcal{R} \triangleright S$, collection \mathcal{R} is preferred to collection S, or $\sum_{i=1}^{m} v(R_i) > \sum_{i=1}^{l} v(S_i)$. In this case, the utility for secondary user *j* in coalition R_j for $R_j \in \mathcal{R}$ is $U_j(\mathcal{R}) = U_j(R_j) = v(R_j)$. In particular, all secondary users in the same coalition will have the same misdetection and false-alarm probabilities. In addition, the Pareto order is considered. Pareto order can be defined for $\mathcal{R} \triangleright S$ as:

$$v_i(\mathcal{R}) \ge v_i(\mathcal{S}), \quad \forall j \in \mathcal{R}, \mathcal{S}.$$
 (13.8)

Pareto order is used in the distributed coalition-formation algorithm, which is based on the merge-and-split rule for secondary users. Partition $S' = \{S_1, ..., S_l\}$ will merge and become $\left\{\bigcup_{j=1}^l S_j\right\}$ if $\left\{\bigcup_{j=1}^l S_j\right\} \triangleright S'$. Conversely, any coalition of $\left\{\bigcup_{j=1}^l S_j\right\}$ will split and become S' if $S' = \{S_1, ..., S_l\} \triangleright \left\{\bigcup_{j=1}^l S_j\right\}$.

The coalition-formation Algorithm 13.1 starts by allowing the secondary users to merge the existing coalition. In this case, coalition $T_i \in T$ of initial partition T searches and tries to merge with a nearby coalition. If merge occurs, the new coalition will again search and try to merge. Once every coalition T_i performs the merging step, the resulting partition becomes \mathcal{F} . Then the split step is performed by all coalitions in this partition \mathcal{F} .

Algorithm 13.1 Distributed coalition-formation algorithm.

- Secondary users are partitioned into T = {T₁,..., T_k} for T = N.
 Phase 1: Local sensing
- 2: Each secondary user performs spectrum sensing. *Phase 2: Coalition formation*
- 3: repeat
- 4: $\mathcal{F} = \mathsf{Merge}(\mathcal{T})$ (Coalition in \mathcal{T} decides to merge)
- 5: $T = \text{Split}(\mathcal{F})$ (Coalition in \mathcal{F} decides to split based on the Pareto order)
- 6: **until** merge-and-split terminates *Phase 3: Standard cooperative sensing*
- 7: Secondary user sends sensing result to coalition head.
- 8: Coalition head of each coalition makes final decision on the presence of primary user using OR rule.
- 9: Secondary user in coalition receives final decision from coalition head.



Fig. 13.4 Maximum coalition size vs. false-alarm probability.

Once this is done, the secondary users perform standard cooperative spectrum sensing based on the available partition, and observe the outcome of the coalition.

Using this coalition-formation algorithm, it is proved in [413] that the maximum size of the coalition is

$$M_{\max} = \frac{\log(1-\alpha)}{\log(1-P_i^{\text{fal}})},\tag{13.9}$$

where α is the target false-alarm probability. The maximum coalition size for various false-alarm probabilities is shown in Fig. 13.4. The stability of the algorithm is studied in [413] using the concept of a *defection function*.

13.1.3 Centralized approach and performance comparison

The centralized approach to coalition formation is used as a benchmark for Algorithm 13.1. In this case, a central fusion center is assumed to exist in the network. An optimization problem is formulated for the central fusion center as follows:

$$\min \frac{\sum_{S \in \mathcal{P}} |S| Q_S^{\text{mis}}}{N}, \text{s.t. } Q_S^{\text{fal}} \le \alpha,$$
(13.10)

where \mathcal{P} is a decision variable of the partition and α is the threshold for the maximum false-alarm probability. It is found that obtaining the optimal partition \mathcal{P}^* is an NP-complete problem [420]. The optimal solution can be obtained by enumeration.

Simulation is performed to evaluate the performance of the distributed coalitionformation algorithm. First, it is found that the performance (i.e., average misdetection probability) of distributed coalition formation is much better than without cooperation. The performance of the centralized approach is better than that of the distributed algorithm. However, the centralized approach requires much more computational resources (e.g., time and memory) to obtain optimal partition. Also, the centralized approach can obtain the solution only for a small number of secondary users, owing to its great complexity. For the distributed coalition-formation algorithm, it is also found that as the target false-alarm probability increases, the size of the coalition becomes smaller, in agreement with (13.9).

13.2 Power allocation as a non-cooperative game

In cognitive radio, the spectrum licensed to a primary user can be opportunistically accessed by secondary users using an overlay or underlay approach. In overlay dynamic spectrum access, secondary users can access the spectrum only when it is not occupied by the primary user. In underlay dynamic spectrum access, secondary and primary users can access the same spectrum simultaneously. In this case, interference to the primary user can be limited by controlling the transmit power of the secondary users. This is similar to the concept of the CDMA cellular system. Power control/allocation is crucial in order for the secondary user to achieve the best performance, while interference to the primary users becomes more challenging when the secondary users are non-cooperative. All secondary users need equilibrium strategies (i.e., for transmit power) to ensure not only that none of them deviates, but also that the interference requirement is not violated. This problem of power allocation can be formulated as a non-cooperative game [510].

13.2.1 Underlay spectrum access and power allocation

The general system model for underlay spectrum access by secondary users is shown in Fig. 13.5. In this multi-channel network, there are N secondary users (i.e., pairs of



Fig. 13.5 Secondary users' transmitters and receivers and primary users in power allocation game.

transmitters and receivers) whose set is denoted by \mathcal{N} . Secondary user *i* has a target data rate (i.e., QoS requirement) denoted by R_i^{\min} for transmission over all channels. Each secondary user is rational to minimize its power consumption while meeting the target transmission rate. There are *C* channels in total, whose set is denoted by $\mathcal{C} = \{1, 2, ..., C\}$. The bandwidth of each channel is denoted by *B*. There are *M* primary users (i.e., monitoring stations to observe interference) in total, whose set is denoted by $\mathcal{M} = \{1, 2, ..., M\}$. For primary user *m* with channel *c*, the target level for interference from all secondary users is $\kappa \gamma_{c,m}^{\max} B$, where κ is Boltzmann's constant (1.38 × 10⁻²³ J K⁻¹) and $\gamma_{c,m}^{\max}$ is the interference temperature limit.

A non-cooperative game is formulated to solve the power-allocation problem [510]. The players are the secondary users. The strategy of player *i* is the transmit power $P_{c,i}$ on channel *c*. Its cost function (i.e., negative payoff) is

$$\mathscr{C}_{i}(\mathbf{p}_{i}) = \omega_{i} \sum_{c \in \mathcal{C}} P_{c,i}, \qquad (13.11)$$

where $\mathbf{p}_i = [P_{1,i}, \dots, P_{c,i}, \dots, P_{C,i}]$ is a 1 × C vector of strategies of player *i*, and ω_i is a weighting factor. In this game model, the strategy space of player *i* is denoted by ξ_i . Because of the constraint on interference to primary users and on the target transmission rate of each secondary user, this strategy space ξ_i is coupled with the strategies of all other secondary users. It can be defined as

$$\xi_{i}(\mathbf{p}_{-i}) = \{\mathbf{p}_{i} \in \Omega_{\mathrm{P},i} | \sum_{c \in \mathcal{C}} \mathcal{R}_{c,i}(P_{c,i}, \mathbf{p}_{c,-i}) \ge R_{i}^{\mathrm{min}},$$
$$P_{c,i}h_{c,i,m} + \sum_{j \neq i} P_{c,j}h_{c,j,m} \le \kappa \gamma_{c,m}^{\mathrm{max}} \mathcal{B}, m \in \mathcal{M}, c \in \mathcal{C}\}, \qquad (13.12)$$

where the general strategy space is defined as

$$\Omega_{\mathsf{P},i} = \{\mathbf{p}_i | \sum_{c \in \mathcal{C}} P_{c,i} \le P_i^{\max}, P_{c,i}^{\min} \le P_{c,i} \le P_{c,i}^{\max}\}.$$
(13.13)

 P_i^{\max} is the total transmit budget on all channels for secondary user *i*. $[P_{c,i}^{\min}, P_{c,i}^{\max}]$ is the interval of transmit power on channel *c*. \mathbf{p}_{-i} and $\mathbf{p}_{c,-i}$ are the vectors of transmit power of all secondary users except user *i* on all channels and on channel *c*, respectively. $h_{c,i,m}$ is the channel gain on channel *c* between secondary user *i* and primary user *m*. The transmission rate is

$$\mathcal{R}_{c,i}(P_{c,i}, \mathbf{p}_{c,-i}) = B \log \left(1 + \frac{P_{c,i}h_{c,i}}{\eta_{c,i}}\right), \qquad (13.14)$$

where $h_{c,i}$ is the channel gain between transmitter and receiver of secondary user *i*, and $\eta_{c,i}$ is the total interference plus noise.

Given the coupled strategy space, the best response of the secondary user is

$$\mathscr{B}_{i}(\mathbf{p}_{-i}) = \min_{\mathbf{p}_{i} \in \xi_{i}(\mathbf{p}_{-i})} \mathscr{C}_{i}(\mathbf{p}_{i}).$$
(13.15)

Based on best response, the Nash equilibrium $(\mathbf{p}_1^* \cdots \mathbf{p}_i^* \cdots \mathbf{p}_N^*)$ satisfies the following condition:

$$\mathbf{p}_{i}^{*} = \arg\min_{\mathbf{p}_{i} \in \xi_{i}(\mathbf{p}_{-i}^{*})} \mathscr{C}_{i}(\mathbf{p}_{i}).$$
(13.16)

13.2.2 Properties of the Nash equilibrium for power allocation

The existence and uniqueness properties of the Nash equilibrium in the above powerallocation non-cooperative game are studied in [510]. First, it is observed that, given the strategies \mathbf{p}_{-i} , of other secondary users, secondary user *i* can achieve a unique optimal power allocation. This is the direct result of the convexity of $\sum_{c \in C} \mathscr{R}_{c,i}(P_{c,i}, \mathbf{p}_{c,-i})$. Also, the optimal achievable transmission rate is a continuous, monotonic increasing function of power capacity P_i^{\max} if the set of interference temperature constraints is satisfied. From these observation, if the target rate $(R_1^{\min}, \ldots, R_i^{\min}, \ldots, R_N^{\min})$ is feasible, then the Nash equilibrium can be assumed to exist. In addition, at the Nash equilibrium, none of the secondary users will receive data at a rate which is greater than the target rate, so the minimum transmit power is used. This is shown by contradiction: for any strategy profile which yields a transmission rate higher than the target rate, the secondary user will have an incentive to reduce the transmission rate so that the transmit power will decrease. As a result, the transmit power decreases until the transmission rate reaches the minimum target threshold R_i^{\min} . Based on above result, the existence of the Nash equilibrium can be demonstrated. This Nash equilibrium is obtained using a modified water-filling algorithm, as follows:

$$P_{c,i}^{*} = \begin{cases} \left[w_{i} - \frac{\eta_{c,i}^{*}}{h_{c,i}} \right]^{+}, & \text{if } c \in \mathcal{C} \setminus \mathcal{I}_{i}, \\ \\ \min_{m \in \mathcal{M}} \frac{\kappa \gamma_{c,m}^{\max} B - \sum_{j \in \mathcal{N}, j \neq i} P_{c,j}^{*} h_{c,j,m}}{h_{c,i,m}}, & \text{if } c \in \mathcal{I}_{i}, \end{cases}$$
(13.17)

where the water-filling level w_i is chosen such that

$$\sum_{c \in \mathcal{C}} \mathscr{R}_{c,i}(P_{c,i}^*, \mathbf{p}_{c,-i}) = R_i^{\min}, \qquad (13.18)$$

and \mathcal{I}_i is a set of channels of secondary user *i* whose interference is saturated, i.e., $\left\{ c \in \mathcal{C} \left| \left[w_i - \frac{\eta_{c,i}^*}{h_{c,i}} \right]^+ > \min_{m \in \mathcal{M}} \frac{\kappa \gamma_{c,m}^{\max} B - \sum_{j \in \mathcal{N}, j \neq i} P_{c,j}^* h_{c,j,m}}{h_{c,i,m}} \right\}$. η^* is the total interference plus noise at the Nash equilibrium. However, the uniqueness of the Nash equilibrium depends largely on the structure of (13.17) and (13.18). It is stated in [510] that if the conditions in (13.17) and (13.18) give a single solution, then the Nash equilibrium is unique.

13.2.3 Distributed algorithm

Given the constraints on interference to primary users and on the target transmission rate for secondary users, a distributed algorithm based on dual decomposition and layered structure is now studied. In particular, the partial dual decomposition is used to relax both constraints. The Lagrangian function can be expressed as

$$\mathscr{L}_{i}(\mathbf{p}_{i}, \mathbf{p}_{-i}, z_{i}, \boldsymbol{\mu}) = \omega_{i} \sum_{c \in \mathcal{C}} P_{c,i} + z_{i} \left(R_{i}^{\min} - \sum_{c \in \mathcal{C}} \mathscr{R}_{c,i}(P_{c,i}, \mathbf{p}_{c,-i}) \right) + \sum_{m \in \mathcal{M}} \sum_{c \in \mathcal{C}} \mu_{c,m} \left(\sum_{i \in \mathcal{N}} P_{c,i} h_{c,i,m} - \kappa \gamma_{c,m}^{\max} B \right), \quad (13.19)$$

where z_i and $\mu = [\mu_{1,1}, ..., \mu_{c,m}, ..., \mu_{C,M}]$ denote the dual price and the vector of dual prices for the constraints on target transmission rate and interference to primary users, respectively. Then, the new cost function is

$$\Theta_{i}(\mathbf{p}_{i},\mathbf{p}_{-i},z_{i},\boldsymbol{\mu}) = \sum_{c\in\mathcal{C}} \left(\omega_{i} + \sum_{m\in\mathcal{M}} \mu_{c,m} h_{c,i,m} \right) P_{c,i} - z_{i} \sum_{c\in\mathcal{C}} \mathscr{R}_{c,i}(P_{c,i},\mathbf{p}_{c,-i}).$$
(13.20)

The objective is to minimize $\Theta_i(\mathbf{p}_i, \mathbf{p}_{-i}, z_i, \mu)$. A subgradient algorithm to update dual prices for secondary user *i*, given the Nash equilibrium, is defined as follows:

$$z_i = z_i + \phi_1 \left(R_i^{\min} - \sum_{c \in \mathcal{C}} \mathscr{R}_{c,i} (P_{c,i}, \mathbf{p}_{c,-i}) \right), \qquad (13.21)$$

$$\mu_{c,m} = \mu_{c,m} + \phi_2 \left(\sum_{i \in \mathcal{N}} P_{c,i} h_{c,i,m} - \kappa \gamma_{c,m}^{\max} B \right), \qquad (13.22)$$

where ϕ_1 and ϕ_2 are the step sizes.

Algorithm 13.2 Distributed power allocation, given constraints on target transmission rate of secondary user and interference to primary users.

1: Each secondary user *i* initializes $\mathbf{p}_i(t_T)$ and $z_i(T)$ and primary user initializes $\mu_{c,m}(T)$ at T=0.

2: repeat

 $t_T \leftarrow 0$ 3:

• Jacobian iteration for power allocation:

- 4: repeat
- Given dual prices $z_i(T)$ and $\mu_{c,m}(T)$, the transmit power of each secondary 5: user is updated as follows:

$$P_{c,i}(t_T) \leftarrow \left[\frac{z_i(T)}{\omega_i + \sum_{m \in \mathcal{M}} \mu_{c,m}(T) h_{c,i,m} - \frac{\eta_{c,i}(t_T - 1)}{h_{c,i}}}\right]_{P_{c,i}^{\min}}^{P_{c,i}^{\max}}, \quad (13.23)$$

where $[\cdot]_{P_{c,i}^{\min}}^{P_{c,i}^{\max}}$ takes a value within $[P_{c,i}^{\max}, P_{c,i}^{\min}]$. $t_T \leftarrow t_T + 1$.

- 6:
- **until** $P_{c,i}(t_T)$ stops changing. 7:
 - Subgradient updating for dual prices:
- Each secondary user updates z_i as follows: 8:

$$z_i(T) \leftarrow \left[z_i(T-1) + \phi_1 \left(R_i^{\min} - \sum_{c \in \mathcal{C}} \mathscr{R}_{c,i}(P_{c,i}, \mathbf{p}_{c,-i}) \right) \right]^+.$$
(13.24)

9: Each primary user updates $\mu_{c,m}$ as follows:

$$\mu_{c,m} \leftarrow \left[\mu_{c,m}(T-1) + \phi_2 \left(\sum_{i \in \mathcal{N}} P_{c,i} h_{c,i,m} - \kappa \gamma_{c,m}^{\max} B \right) \right]^+.$$
(13.25)

 $T \leftarrow T + 1$ 10:

11: **until** $z_i(T)$ and $\mu_{c,m}$ stop changing.



Fig. 13.6 Iteration for updating dual prices and subiteration for updating transmit power.

To iteratively solve for the Nash equilibrium in a distributed setting, a layered structure (i.e., a Stackelberg game) is introduced. The first equilibrium power allocation \mathbf{p}^* is, obtained, based on Jacobian iteration, given dual prices. Then, the dual prices z_i and $\mu_{c,m}$ are updated. Algorithm 13.2 has three major steps. The transmit power is updated on a finer time scale, i.e., every Δt . The dual prices are updated on a longer time scale, i.e., every Δt . The dual prices is denoted by T, while the subiteration for power update is denoted by t_T . The relationship between iteration and subiteration is indicated by $\frac{\Delta T}{\Delta t} = Y$, where Y is a large integer (Fig. 13.6). The convergence of Algorithm 13.2 is also studied in [510].

13.2.4 Pigouvian taxation and social optimality

The socially optimal solution of

$$\min_{(\mathbf{p}_1,\dots,\mathbf{p}_N)} \sum_{i \in \mathcal{N}} \omega_i \sum_{c \in \mathcal{C}} P_{c,i},$$
(13.26)

is also considered in [510]. Pigouvian taxation [321] is applied to achieve the socially optimal solution. An externality cost is applied to each secondary user, so the Nash equilibrium is shifted to the socially optimal solution. The taxation $\pi_i = [\pi_{1,1}, \ldots, \pi_{c,i}, \ldots, \pi_{C,N}]$ is added into the cost function. Based on the Karush–Kuhn–Tucker necessary condition for a socially optimal solution, the objective function can be defined as follows:

$$\hat{\Theta}_{i}(\mathbf{p}_{i}, \mathbf{p}_{-i}, \boldsymbol{\pi}_{i}, z_{i}, \boldsymbol{\mu}) = \sum_{c \in \mathcal{C}} \left(\omega_{i} + \sum_{m \in \mathcal{M}} \mu_{c,m} h_{c,i,m} \right) P_{c,i} + \sum_{c \in \mathcal{C}} \pi_{c,i} P_{c,i} - z_{i} \sum_{c \in \mathcal{C}} \mathscr{R}_{c,i} (P_{c,i}, \mathbf{p}_{c,-i}),$$
(13.27)

where $\pi_{c,i} = -\sum_{j \in \mathcal{N}, j \neq i} \frac{\partial z_j \mathscr{R}_{c,j}(P_{c,j}, \mathbf{p}_j)}{\partial P_{c,i}}$. The distributed implementation to achieve the socially optimal solution is also proposed in [510].

For the proposed power allocation algorithm, based on a non-cooperative-game model, numerical results in [510] show that the Nash equilibrium can be achieved while all constraints on target transmission rate and interference to the primary users are met. Also, with the Pigouvian-taxation mechanism, a higher transmission rate can be obtained.

13.2.5 Related work

The power-allocation problem for underlay dynamic spectrum access is studied from a variety of perspectives [500, 163, 515, 514]. In [515], a potential game is formulated for power allocation to secondary users with an interference-temperature constraint. A distributed algorithm to determine the secondary link for spectrum access and power control is developed. A constraint on minimum signal-to-interference ratio (SIR) is used to guarantee QoS performance for secondary users.

13.3 Medium access control

For overlay spectrum access in cognitive-radio networks, medium access control (MAC) must be optimized for the best performance. In the commons-use spectrum-licensing model, joint channel allocation and access based on game theory is proposed in [148]. Channel allocation refers to the selection of a channel for transmission by the secondary user, while channel access refers to the obtaining of transmission parameters (e.g., size of contention window in CSMA/CA protocol). Given non-cooperative behavior by secondary users, two game models for channel allocation and channel access are developed to obtain an equilibrium strategy for the secondary users.

In the cognitive-radio network under consideration in [148], there are multiple channels, whose set is denoted by C. A secondary user (i.e., player) consists of a transmitter and a receiver (i.e., transmission pair), and the set of secondary users is denoted by \mathcal{N} . Each secondary user has x radio interfaces which can access a maximum of x channels simultaneously (e.g., as in [528, 16]). For each interface, the CSMA/CA protocol is used. It is assumed that the number of radio interfaces is smaller than or equal to the number of available channels, i.e., $x \leq |\mathcal{C}|$. A secondary user can observe the transmissions by all other users. The strategies of a secondary user are the set of channels to access and the size of the contention window in the CSMA/CA protocol. The payoff is the throughput (i.e., effective transmission rate), denoted by $r_i = \sum_{c \in C} r_{c,i}$ for secondary user *i*, and $r_{c,i}$ is the throughput of user *i* on channel *c*. It is assumed that the throughput on channel *c* (i.e., the sum of throughputs for all users accessing channel c) is a decreasing function of the number of users (i.e., radio interfaces) accessing this channel. Since there are two strategies for each secondary user (i.e., the set of channels and the size of the contention window), two game models are proposed, i.e., channel allocation and access. In this case, the channel-allocation game is solved first, and the channel-access game is solved given the strategy for channel allocation.

13.3.1 Channel allocation

For the channel-allocation game, secondary user chooses the number of radios $x_{c,i}$ to allocate to channel *c*. Therefore, the strategy of user *i* becomes $s_i = (x_{1,i}, ..., x_{c,i}, ..., x_{C,i})$ where C = |C| is the total number of channels. In this game, the secondary user assumes that the total throughput of the channel will be fairly shared (without adapting the channel-access parameter). With all identical channels, the achievable throughput is denoted by $r(x_c)$ where $x_c = \sum_{i \in \mathcal{N}} x_{c,i}$ is the total number of radios allocated to channel *c*. The payoff can be defined as

$$r_{i} = \sum_{c \in \mathcal{C}} \frac{x_{c,i}}{\sum_{i' \in \mathcal{N}} x_{c,i'}} r(x_{c}).$$
(13.28)

It is found in [148] that in this channel-allocation game, there can be multiple Nash equilibria. It is noted that the candidate strategy for Nash equilibria should have the following properties:

- Secondary users allocate all radios to the available channels. To maximize throughput, all radios should be allocated for transmission.
- The maximum difference between the number of radios allocated to any channel is one. To avoid congestion on the channel, the radios should be equally allocated to all channels.

The difference between the number of radios allocated to channels *c* and *c'* is $\Delta x_{c,c'} = x_c - x_{c'}$. The necessary and sufficient conditions for Nash equilibria, based on the load-balancing concept, are

- $\Delta x_{c,c'} \leq 1$ for any $c, c' \in C$
- $x_{c,i} \leq 1$ for any $c \in C$.

The latter condition ensures that a secondary user does not allocate many radios to the same channel. However, there is another set of Nash equilibria in which some secondary users allocate multiple radios to the same channel. The conditions for these Nash equilibria are

- $\Delta x_{c,c'} \leq 1$ for any $c, c' \in C$
- For secondary user *i* with $x_{c,i} \ge 2$, $x_{c,i} \le \frac{r(x_c-1)/(x_c-1) r(x_c+1)/(x_c+1)}{r(x_c-1)/(x_c-1) r(x_c)/x_c}$
- For secondary user *i* with x_{c',i} ≥ 2 and c' ∈ C_{max}, x_{c,i} ≥ x_{c',i} for c ∈ C_{min}, where C_{min} and C_{max} are the sets of channels with minimum and maximum number of radios, respectively.

Note that there could be also the other sets of Nash equilibria, subject to specific conditions on the throughput function $r(\cdot)$.

The efficiency of the Nash equilibria for this channel-allocation model is also analyzed in [148]. The price of anarchy is derived when the throughput function $r(\cdot)$ is a decreasing

function of x_c , and can be express as

$$\rho = \frac{r(1)}{\left(x_c + 1 - \frac{|\mathcal{N}|x}{|\mathcal{C}|}\right)\left(r(x_c) - r(x_c + 1)\right) + r(x_c + 1)},$$
(13.29)

where $x_c = \left\lfloor \frac{|\mathcal{N}|x}{|\mathcal{C}|} \right\rfloor$ and $\lfloor \cdot \rfloor$ is floor function. It is observed that as the throughput function becomes more independent of the number of allocated radios, the price of anarchy converges to 1, which indicates that the Nash equilibria can achieve almost the maximum total throughput.

13.3.2 Channel access

In this channel-access model, CSMA/CA protocol with distributed coordination function (DCF) is assumed. The objective of this game is to optimize the bandwidth utilization of all channels, and also to influence secondary users to play the channel-allocation game optimally. In this game, secondary users $i \in \mathcal{N}_{cht}$ can cheat by making the contention window smaller than in the standard. The cheating user (i.e., the cheater) is a player in the channel-access game. $\mathcal{N}_{cht} \subset \mathcal{N}_c \subseteq \mathcal{N}$, where \mathcal{N}_{cht} is the set of cheaters with \mathcal{N}_{cht} cheaters in total, and \mathcal{N}_c is the set of secondary users accessing channel *c*. Note that the analysis is performed for a single channel *c* to which is allocated multiple radios of two or more secondary users. In general, cheaters will decrease the size of their contention window to achieve a higher channel-access probability, and thus a higher throughput. In this channel-access game, the strategy of the secondary user (i.e., cheater) is the set of contention-window sizes $s_i = (W_{i,1}, \ldots, W_{i,l}, \ldots, W_{i,x_{c,i}})$, where $W_{i,l}$ is the size of the contention window for radio *l* of user *i*. The channel-access probability is $\tau_{i,l} = \frac{1}{W_{i,l+1}}$. The average throughput $r_{c,i}$ of cheater *i* is

$$r_{c,i} = \sum_{l=1}^{x_{c,i}} r_{c,i,l},$$
(13.30)

where $r_{c,i,l}$ is the average throughput achieved from the *l*th radio of secondary user *i*.

As shown in [88], there are two sets of Nash equilibria for this channel-access game. In the first set, only a single player adopt $W_{i,l} = 1$, while the other users adopt $W_{j,l} > 1$ for $j \neq i$. This Nash equilibrium achieves the highest throughput for player *i* since this user *i* always has a chance to transmit. However, the other users receive zero throughput, so these Nash equilibria are efficient but totally unfair. In the second set of Nash equilibria, more than one player adopts $W_{i,l} = 1$, so all users receive zero throughput. These Nash equilibria are referred to as *the tragedy of the commons*. Because of the undesirable outcome of these Nash equilibria, an alternative solution based on Nash bargaining is introduced [376]. This Nash bargaining solution can be obtained from the following optimization problem:

$$(s_{1}^{\star}, \dots, s_{|\mathcal{N}_{cht}|}^{\star}) = \arg \max_{(s_{1}, \dots, s_{|\mathcal{N}_{cht}|})} \prod_{i \in \mathcal{N}_{cht}} \prod_{l=1}^{s_{c,i}} r_{c,i,l}(s_{1}, \dots, s_{|\mathcal{N}_{cht}|}).$$
(13.31)

This solution is Pareto-optimal, and fair. To make the optimization problem tractable, it is assumed that the size of the contention window is continuous. With this assumption, it can be shown analytically that the Nash bargaining solution is unique. Where the formulation in (13.31) considers continuous values of $W_{i,l}$, the case for discrete values of $W_{i,l}$ is shown by simulation to also have a solution.

To force all players to adopt the Nash bargaining solution, a penalizing mechanism is introduced in [148]. In particular, if there is a player deviating from the Nash bargaining solution, other players will punish the deviating player by jamming the transmissions of that player. The penalty that player i with radio l imposes on deviating player j with radio l' is

$$z_{c,j,l'} = \begin{cases} r_{c,j,l'} - r_{c,i,l}, & \text{if } r_{c,j,l'} > r_{c,i,l}, \\ 0, & \text{otherwise.} \end{cases}$$
(13.32)

According to the penalty $z_{c,j,l'}$, the jamming period can be determined. Details of the jamming mechanism can be found in [88].

13.3.3 Distributed algorithms

To achieve efficient Nash equilibria, distributed algorithms are proposed for the channelallocation and channel-access game models.

Distributed channel-allocation algorithm

For channel allocation, the player (i.e., the secondary user with a modified contention window) chooses a strategy knowing only the number of radios allocated to each channel. The distributed algorithm is shown as Algorithm 13.3. This algorithm can achieve the Nash equilibrium in which all players allocate only one radio to each channel.

Algorithm 13.3 Distributed channel-allocation algorithm.

- 1: player initializes channel allocation.
- 2: player observes the number of radios of each channel in which the average number of radios on the channel in C_i is \overline{x}_i . C_i is the set of channels with radios allocated by player *i*.
- 3: For channel $c' \in C_i$ with $x_{c'} \overline{x}_i \ge 1$, player *i* moves radio to channel $c \notin C_i$ with probability $\frac{1}{|C| \setminus C_i|}$.
- 4: To avoid reaching an inefficient stable channel allocation, with small probability ϵ player randomly moves a radio to a different channel even though $0 < x_{c'} \overline{x}_i < 1$.

From the performance evaluation, Algorithm 13.3 quickly converges to one of the Nash equilibria which is efficient. The algorithm can fluctuate as a result of randomness in line 13.3, but this fluctuation prevents players being trapped in an inefficient strategy.

Distributed channel-access algorithm

Given a penalty mechanism to punish the cheater, channel-access algorithm to achieve a unique Pareto-optimal solution is introduced in [148]. In this case, one of the players, denoted by *i*, is selected to be a coordinator. This coordinator observes the channel and performs punishment if necessary. The coordinator uses Algorithm 13.4, where ε is a small value used in the termination condition of the algorithm. In this case, the coordinator forces the other players to reach the Pareto-optimal point.

Algorithm 13.4 Channel-access algorithm.

- 1: Coordinator (i.e., player *i*) chooses the size of the contention window $W_{i,l} = W_0 > 1$ for radio *l*.
- 2: repeat
- 3: Coordinator punishes any deviating player *j* if $r_{c,j,l'}(W_{j,l'}) > r_{c,i,l}(W_{i,l})$ for $j \neq i$.
- 4: Punished player maximizes the throughput which can be achieved by adjusting the size of the contention window to $W_{j,l'} = W_{i,l}$.
- 5: Coordinator updates the size of the contention window to $W_{i,l} = W_{i,l} + 1$.
- 6: **until** $|r_{c,i,l}(W_{i,l}) r_{c,i,l}(W_{i,l}-1)| < \varepsilon$.

From the performance evaluation, Algorithm 13.4 can maximize the total throughput. In summary, it is found in [148] that if a player wants to achieve a fair allocation on each channel, it will not be beneficial for any player to be fully selfish by setting the smallest contention window size.

13.4 Decentralized dynamic spectrum access

In cognitive-radio networks, in most cases there is no centralized controller to govern channel access. Therefore, a rational secondary user requires a decentralized algorithm to reach an equilibrium strategy. This decentralized algorithm needs to rely only on local information and has to adapt to the environment quickly. In [322], a decentralized algorithm for overlay spectrum access is developed based on the stochastic approximation technique. It is shown that this algorithm can converge to a *correlated equilibrium* if the spectrum-access activity of the primary user varies slowly [380, 423, 499].

13.4.1 Overlay dynamic spectrum access

Cognitive radio with *N* secondary users accessing the channels on a time-slot basis is considered. The length of time slot is *Y*. There are *C* channels in total, each of which is licensed to a primary user. The channel quality (i.e., transmission rate) of channel *c* at time slot *t* is denoted by $r_c[t]$. The channel status (i.e., activity of corresponding primary user) is denoted by $y_c[t]$, such that channel *c* is occupied by the primary user if $y_c[t] = 0$. The decision of secondary user *i* is denoted by $x_{c,i}[t] = 0$ otherwise.



Fig. 13.7 Example of a channel-access decision by secondary users, in decentralized dynamic spectrum access. Channel-access decision of secondary users.

The channel-access decision vector is $\mathbf{x}_i[t] = [x_{1,i}[t], \dots, x_{c,i}[t], \dots, x_{C,i}[t]]$. For secondary user *i*, the channel-access demand is $D_i[t]$, and the maximum number of channels that this secondary user can access simultaneously is X_i . $D_i[t]$ and X_i are private information which is not known by other secondary users. An example of channel access with C = 3 channels and N = 3 secondary users is shown in Fig. 13.7.

The decision space (i.e., action space) of the secondary user is denoted by

$$\mathcal{A}_{i}[t] = \left\{ \mathbf{x}_{i}[t] \left| \sum_{c=1}^{C} x_{c,i}[t] y_{c}[t] = 1, \sum_{c=1}^{C} x_{c,i} \leq X_{i}[t] \right\}.$$
 (13.33)

That is, the secondary user will access only an idle channel, and the total number of access channels has to be smaller than $X_i[t]$. The vector of decisions of all secondary users is denoted by $\mathbf{x}[t] \in \mathcal{A}[t] = \mathcal{A}_1[t] \times \cdots \times \mathcal{A}_i[t] \times \cdots \times \mathcal{A}_N[t]$, where \times is the Cartesian product. Given decision $x_{c,i}[t]$, secondary user *i* accesses a channel based on the CSMA scheme. In this case, the time slot is divided into W minislots. The minislot in time slot t is denoted by $t_1, \ldots, t_w, \ldots, t_W$. If secondary user i decides to access channel c in time slot t, i.e., $x_{c,i}[t] = 1$, a backoff mechanism will be performed. In particular, the secondary user generates a backoff time $b_{c,i}[t]$ with uniform distribution in the interval $[0, b_{max}]$, where b_{max} is the maximum backoff size. Then the timer counts down from $b_{c,i}[t]$, and before it reaches zero, the secondary user transmits data if channel c is sensed to be idle. However, a collision can occur if more than one secondary user has the same $b_{c,i}[t] = b_{c,i}[t]$ for $i \neq j$. Specifically, a collision can occur in two cases. In the first case, secondary user *i* has the lowest backoff (i.e., $b_{c,i}[t] = \min_j b_{c,j}[t]$), but there is another user with backoff in the range $[b_{c,i}[t], b_{c,i}[t] + \delta]$, where δ is the time required to sense the channel and to switch from receive to transmit mode. In the second case, secondary user *i* does not have the lowest backoff. However, its backoff is within $[b_{c,i}[t], b_{c,i}[t] + \delta]$, where secondary user *i* has the lowest backoff. The success of data transmission can be observed at the end of the time slot. Given this backoff mechanism, the expected number of other secondary users $\tilde{N}_{c,i}[t]$ contending with secondary user *i* for the idle time slot on channel *c* at time slot *t* can be estimated. Also, the probability of channel capture $P_{c,i}^{cap}[t]$ and probability of collision $P_{c,i}^{col}[t]$ for secondary user *i* can be obtained. These probabilities will be used in the decentralized algorithm of the secondary user to decide on channel access.

13.4.2 Utility function

First, the global system utility function is formulated. This function is based on a maxmin fairness criterion in which the utility of the secondary user with the minimum ratio of received transmission rate to demand will be maximized, i.e.,

$$\mathscr{U}^{\mathsf{glo}}(\mathbf{x}[t]) = \min_{i} \left(\min\left(1, \frac{\sum_{c=1}^{C} r_c[t] P_{c,i}^{\mathsf{cap}}[t]}{D_i[t]}\right) \right),$$
(13.34)

where $\frac{\sum_{c=1}^{C} r_c[t] P_{ci}^{cap}[t]}{D_i[t]}$ represents the satisfaction level of secondary user *i*. The optimal solution of (13.34) can be obtained if all complete and perfect information is available to a centralized controller. However, in a decentralized setting, there is no such controller. Also, the secondary users do not know one another's private information (e.g., demand). Therefore, based on the global utility function, a local utility function is defined, composed of three components. The first component accounts for the self-interest to maximize the secondary user's satisfaction:

$$\mathscr{U}_{1}^{\mathsf{loc}}(x_{c,i}) = \min\left(\sum_{c=1}^{C} \frac{r_{c}[t]}{D_{i}} \frac{x_{c,i}[t]}{1 + N_{c,i}[t]} \left(1 - \frac{\delta}{b_{\mathsf{max}}}\right)^{1 + N_{c,i}[t]}\right).$$
(13.35)

This first component requires information about the number of secondary users contending for the same channel, which can be approximated locally. Furthermore, the secondary user should maximize the satisfaction of other users, so cooperation among secondary users is introduced by requiring that no user may have a transmission rate higher than the demand, so that other users have a chance for channel access. Consequently, the second component of the local utility function is defined as follows:

$$\mathscr{U}_{2}^{\text{loc}}(x_{c,i}) = -\frac{1}{D_{i}} \max\left(0, \left(\sum_{c=1}^{C} r_{c}[t] P_{c,i}^{\text{cap}}[t] - (D_{i} + \kappa)\right)\right).$$
(13.36)

This second component is the penalty if the data rate of secondary user *i* is higher than the demand plus a tolerance margin κ . This margin is introduced because of the randomness of the system parameters and because of estimation errors. The secondary user should minimize collisions, which will degrade the performance of all users. Therefore, the third component of the local utility function accounts for the penalty of collision:

$$\mathscr{U}_{3}^{\text{loc}}(x_{c,i}) = -\frac{1}{\sum_{c=1}^{C} r_{c}[t]} \sum_{\{c \mid \hat{N}_{c,i} > 0\}} r_{c}[t] \frac{P_{c,i}^{\text{col}}[t]}{\hat{N}_{c,i}}, \qquad (13.37)$$

where $\hat{N}_{c,i}$ is the estimated number of contending secondary users. Given these three components, the local utility function of secondary user *i* can be expressed as

$$\mathscr{U}_i^{\mathsf{loc}}(\mathbf{x}_{c,i}) = \max\left(0, \omega_1 \mathscr{U}_1^{\mathsf{loc}}(\mathbf{x}_{c,i}) + \omega_2 \mathscr{U}_2^{\mathsf{loc}}(\mathbf{x}_{c,i}) + \omega_3 \mathscr{U}_3^{\mathsf{loc}}(\mathbf{x}_{c,i})\right), \qquad (13.38)$$

where ω_1 , ω_2 , and ω_3 are non-negative weights. This local utility function is used by secondary user *i* to adjust its channel-access decision $\mathbf{x}_i[t]$. This utility is also a function of the estimated number of contending secondary users $\hat{N}_{c,i}$, which is a function of actions adopted by the other secondary users. Therefore, the utility function may be expressed as $\mathcal{U}_i^{\text{loc}}(\mathbf{x}_i[t], \mathbf{x}_{-i}[t])$, where $\mathbf{x}_{-i}[t]$ is the vector of actions of all secondary users except user *i*.

13.4.3 Decentralized algorithm for channel access

A non-cooperative game can be formulated whose players are the secondary users. The strategy is the channel-access action $\mathbf{x}_i[t]$ and the payoff is the local utility function $\mathcal{U}_i^{\text{loc}}(\mathbf{x}_i[t])$. A decentralized stochastic approximation algorithm, used to obtain the channel-access action of the secondary user using local and private information, is developed based on the regret-matching procedure [196]. The adaptive mechanism integrated into this procedure is referred to as *regret tracking*. In this algorithm, the secondary user takes a sequence of actions defined as follows:

$$\{\mathbf{x}_i[t] \in \mathcal{A}_i[t] \mid t = 0, 1, 2, ...\}.$$
(13.39)

The secondary user observes a sequence of rewards, denoted by $u_i[t]$ for t = 0, 1, 2, ..., and makes a decision on channel access at time slot t + 1.

The decentralized algorithm is presented as Algorithm 13.5. $\epsilon[t]$ is the step size. $\theta_i[t]$ is the average regret matrix, with elements $\theta_{j,d}^i[t]$. This average regret indicates the gain that secondary user *i* would receive if decision *d* had been made during time 0 to *t*, instead of decision *j*. μ satisfies the condition $\mu > (S_i - 1)(U_i^{\text{max}} - U_i^{\text{min}})$, where U_i^{max} and U_i^{min} are, the respectively, maximum and minimum bounds of the local utility (13.38). $H_{i,d}^i(\mathbf{x}[t])$ is the instantaneous regret, defined as

$$H_{j,d}^{i}(\mathbf{x}[t]) = \mathscr{U}_{i}^{\mathsf{loc}}(d, \mathbf{x}_{-i}[t]) - \mathscr{U}_{i}^{\mathsf{loc}}(j, \mathbf{x}_{-i}[t]), \qquad (13.40)$$

i.e., the regret at making decision *j* instead of *d*. This regret $H_{j,d}^i$ is an element at row *j* and column *d* of the regret matrix H_i .

In Algorithm 13.5, the step size $\epsilon[t]$ can be fixed ($\epsilon[t] = \epsilon$) or decreasing ($\epsilon[t] = \frac{1}{t+1}$). With decreasing step size, the algorithm will converge with probability 1 to the correlated equilibria [322]. However, the algorithm may not adapt to a change of system parameter. Alternatively, with a fixed step size, the algorithm can adapt to the slowly changing activity of primary users and converge to the correlated equilibria. Algorithm 13.5 is proved to converge to the correlated equilibria using stochastic averaging theory. For a decreasing step size, the proof is based on [67], while for a fixed step size, the proof

Algorithm 13.5 Adaptive learning for channel access.

- 1: *Initialization:* Set t = 0, take action $\mathbf{x}_i[0]$, and initialize $\theta_i[0] = \mathbf{H}_i(\mathbf{x}[0])$.
- 2: Set n = 1, take action $\mathbf{x}_i[1] = \arg \max_d H_{j,d}^i(\mathbf{x}[0])$ where $j = \mathbf{x}_i[0]$ and set $\theta_i[1] = \mathbf{H}_i(\mathbf{x}[1])$.
- 3: **loop**
- 4: *Action update:* Choose $\mathbf{x}_i[t+1] = d$ with probability

$$\Pr(\mathbf{x}_{i}[t+1] = d | \mathbf{x}_{i}[t] = j, \boldsymbol{\theta}_{i}[t] = \boldsymbol{\theta}_{i}) = \begin{cases} \max\left(0, \theta_{j,d}^{i}\right) / \mu, & \text{if } d \neq j, \\ 1 - \sum_{\hat{d} \neq j} \max\left(0, \theta_{j,\hat{d}}^{i}\right), & \text{if } d = j, \end{cases}$$
(13.41)

where $\theta_{i,d}^{i}$ is an element of θ_{i} .

5: Average regret update: Given $H_i(x[t+1])$, update $\theta_i[t+1]$ according to the following stochastic approximation algorithm with step size $\epsilon[t]$:

$$\boldsymbol{\theta}_{i}[t+1] = \boldsymbol{\theta}_{i}[t] + \boldsymbol{\epsilon}[t] (\mathbf{H}_{i}(\mathbf{x}[t+1]) - \boldsymbol{\theta}_{i}[t]).$$
(13.42)

6: t = t + 1. 7: **end loop**

is based on [270]. In short, with various step sizes, the decision of the secondary user converges to the trajectory of the given differential inclusion.

13.4.4 Alternative algorithms

Best-response, fictitious-play, and modified regret-tracking algorithms are also considered in [322]. In best-response algorithm, the secondary user decides to access the channel to maximize its utility, on the assumption that the decisions of other users remain the same, i.e., $\mathbf{x}_i[t+1] = \arg \max_d H_{j,d}^i(\mathbf{x}[t])$, where $j = \mathbf{x}_i[t]$. For fictitious play, which is a special case of regret-based algorithms [198], the update of the decision in line 5 of Algorithm 13.5 becomes $\mathbf{x}_i[t+1] = \arg \max_d (\theta_{j,d}^i[t](\mathbf{x}_i[t]))$ for decreasing step size. However, the best-response and fictitious-play algorithms assume that secondary users know all possible decisions, not just the available actions. To overcome this limitation, the modified regret-tracking algorithm [197] estimates the decisions which have not been made, so the secondary user can make an approximately optimal decision.

From the performance evaluation, it is shown that the studied regret-tracking algorithm outperforms all of these alternative algorithms (i.e., best-response, fictitious-play, and modified regret-tracking) by most closely converging to the correlated equilibria of the spectrum-access game. With the proposed regret-tracking algorithm, it is found that the average channel utilization is higher for slower changes in the primary user's activity. This is because the secondary user has more time to observe and adapt to the changes. Also, the performance with spectrum-sensing error is investigated in [322]. As expected, network performance is degraded when the probability of a sensing error (i.e., an occupied channel is sensed to be idle or an idle channel is sensed to be occupied) increases. Therefore, Algorithm 13.5 requires some level of channel-sensing accuracy to achieve the best performance.

13.5 Radio resource competition based on a stochastic learning game

To obtain spectrum access, a secondary user can bid competitively for the spectrum from a central spectrum moderator (i.e., a spectrum broker or regulator). To bid for the spectrum, not only the channel state but also the local state (e.g., buffer occupancy) of the secondary user will impact the strategy selection. In a dynamic environment, to obtain a competitive strategy for spectrum bidding, a stochastic-game model [159] can be formulated [158, 485]. Also, if information about the other secondary users is not publicly available, each user has to learn and adapt its strategy dynamically to achieve the highest reward or, equivalently, lowest cost.

13.5.1 System model of radio resource competition

In the system, there are *C* channels. Each channel is allocated to one primary user whose channel occupancy is modeled as a two-state Markov chain [513, 446] (Fig. 13.8). In this channel model, the state of channel *c* at time slot *t* is denoted by $y_c[t]$, where $y_c[t] = 0$ if the channel is in the On state (secondary user cannot transmit) and $y_c[t] = 1$ if the channel is in the Off state (secondary user can transmit). The probability of a state transition is denoted by α_c and β_c from On to Off and from Off to On, respectively. There are *N* secondary users in total, with $N \ge C$. One channel can be accessed by one secondary user in one time slot, and the transmit power is fixed. Therefore, there is no interference in the system. Each secondary user has a buffer to store an incoming packet, whose arrival is assumed to be a Possion process. The secondary user has to determine external and internal actions, denoted, respectively, by $\mathbf{a}_i[t]$ and $d_i[t]$, for time slot *t*. The external action corresponds to the bidding strategy to acquire the channel from the spectrum broker. The internal action corresponds to the transmission strategy on the allocated channel and time slot. Secondary users compete with each other to obtain







Fig. 13.9 System model of radio resource competition.

channel access by optimizing their external actions. Then, once the channel is allocated, each secondary user optimizes its internal action to maximize the reward.

To obtain the action of a secondary user, a stochastic-game model is defined. The players are the secondary users. The state of a player is composed of buffer occupancy (i.e., the number of packets in the queue), status, and quality of the channel. The strategy is composed of an external action to bid for the channel and an internal action for packet transmission. The payoff is computed from the buffer overflow and cost of channel bidding. An example of a system model with two secondary users is shown in Fig. 13.9.

13.5.2 Auction mechanism

It is assumed that the spectrum broker maintains the status of all channels (i.e., On–Off) in every time slot. The spectrum broker can allow secondary users to access channels with Off state in a certain slot. In this case, The spectrum broker informs secondary users about the channel status (i.e., $\mathbf{y}[t] = [y_1[t], ..., y_c[t], ..., y_c[t]]$). Each secondary user submits a bid to the broker (i.e., $\mathbf{a}_i[t] = [a_{1,i}[t], ..., a_{c,i}[t], ..., a_{C,i}[t]]$). Given the bids from all competing secondary users, the broker determines the winner and sends channel allocations back to all secondary users. The channel allocation vector is denoted by $\mathbf{z}_i[t] = [z_{1,i}[t], ..., z_{C,i}[t]]$, where $z_{c,i} = 1$ if channel *c* is allocated to secondary user *i* and $z_{c,i} = 0$ otherwise. Also, the spectrum broker computes the payment p_i which secondary user *i* has to pay to access the channel. In this case, feedback information from the broker to secondary user *i* can be defined as $f_i[t] = (\mathbf{z}_i[t], p_i[t])$. The message





exchange, in this auction, between the spectrum broker and secondary users is shown in Fig. 13.10.

Let Z[t] denote the channel-allocation matrix whose elements are $z_{c,i}$. The auction rule is to maximize total social welfare, i.e.,

$$\mathbf{Z}^{*}[t] = \arg \max_{\mathbf{Z}[t]} \sum_{i=1}^{N} \sum_{c=1}^{C} z_{c,i}[t] a_{c,i}[t].$$
(13.43)

In a second-price auction mechanism [328], the price to be paid by the secondary user i is

$$p_{i}[t] = \sum_{j=1, j \neq i}^{N} \sum_{c=1}^{C} z_{c,j}^{*}[t] a_{c,j}[t] - \max_{\mathbf{Z}_{-i}[t]} \sum_{j=1, j \neq i}^{N} \sum_{c=1}^{C} z_{c,j}[t] a_{c,j}[t], \quad (13.44)$$

where $z_{c,i}^*[t]$ is an element of $Z^*[t]$ and $Z_{-i}[t]$ is the matrix of channel allocation for all secondary users except user *i*. The optimization defined in (13.43) and (13.44) can be efficiently solved using linear programming.

With this auction mechanism, the spectrum broker does not need to know the private information (e.g., utility functions) of the secondary users. Also, the outcome of the auction can be obtained efficiently using a standard algorithm. Therefore, it is suitable for online resource management.

13.5.3 Secondary-user strategy

The state of user *i* is composed of the buffer occupancy $b_i[t]$ and the channel status $y_c[t]$ and channel quality. For channel quality, adaptive modulation and coding is considered. In this case, channel quality (i.e., SNR) can characterize the channel state according to the thresholds. A finite-state Markov-chain (FSMC) model is adopted in this case [363]. Channel status and channel quality can be combined. Let the FSMC state of channel *c* allocated to secondary user *i* be $e_{c,i}[t]$. At state $e_{c,i}[t] = 0$, the transmission rate is zero, which could be due to the low SNR or because the channel is occupied by the primary user (i.e., $y_c[t] = 0$). At state $e_{c,i}[t] = 1, ..., E$, where *E* is the highest state of the FSMC model, the transmission rate is positive and the channel is idle (i.e., $y_c[t] = 1$).

In this way, the state of secondary user *i* can be expressed as $s_i[t] = (b_i[t], e_{c,i}[t])$, i.e., a composite of buffer occupancy and channel state. The transition of state can be determined from the packet arrival, channel allocation, channel state, and internal action for packet transmission. In this case, the internal action is chosen to maximize the number of transmitted packets. Let $\mathcal{N}_i(s_i[t], \mathbf{z}_i[t], d_i[t])$ denote the function for the number of transmitted packets of secondary user *i* given state $s_i[t]$, channel allocation $\mathbf{z}_i[t]$, and internal action $d_i[t]$. The number of successfully transmitted packets is

$$n_i(s_i[t], \mathbf{z}_i[t]) = \max_{d_i[t]} \mathcal{N}_i(s_i[t], \mathbf{z}_i[t], d_i[t]).$$
(13.45)

To obtain the optimal action, the method of cross-layer optimization [157, 486] can be adopted in this case. The change of buffer occupancy can be expressed as

$$b_{o}[t+1] = \min(\max(0, b_{i}[t] - n_{i}(s_{i}[t], \mathbf{z}_{i}[t])) + \lambda_{i}[t], b_{i}^{\max}), \quad (13.46)$$

where $\lambda_i[t]$ is the number of arriving packets in time slot t and b_i^{max} is the maximum buffer size. In this case, some arriving packets can be lost if the buffer is full (i.e., $b_i[t] = b_i^{\text{max}}$); this loss is considered to be the cost for the secondary user. Therefore, the immediate reward for the secondary user is

$$r_i(s_i[t], f_i[t]) = -(p_i + \max(0, \max(0, b_i[t] - n_i(s_i[t], \mathbf{z}_i[t])) + \lambda_i[t] - b_i^{\max})),$$
(13.47)

where $f_i[t] = (\mathbf{z}_i[t], p_i[t])$ is feedback information from the spectrum broker. This reward can be fully characterized by the state, channel status, and action of all secondary users, i.e., $r_i(\mathbf{s}[t], \mathbf{y}[t], \mathbf{a}[t])$, where $\mathbf{s}[t] = [s_1[t], \dots, s_i[t], \dots, s_N[t]]$ is a vector of states, $\mathbf{y}[t]$ is a vector of channel status, and $\mathbf{a}[t]$ is a vector of actions of all secondary users on all channels.

To obtain the optimal external action, the secondary user observes the state and action in the system. The observation by secondary user *i*, with private information, can be denoted by $o_i[t]$. The action can be determined given the observation, and this mapping is referred to as the policy, i.e., $\pi_{i,t}$ where $\mathbf{a}_i[t] = \pi_{i,t}(o_i[t])$. Since the secondary user can observe only its own state and the channel status as provided by the spectrum broker, the policy can be defined as $\pi_{i,t}(o_i[t]) = \pi_{i,t}(s_i[t], \mathbf{y}[t])$. Given a discount factor δ_i with $0 \le \delta_i < 1$, the total discounted sum of rewards to secondary user *i* is

$$Q_{i,t}(\mathbf{s}[t], \mathbf{y}[t], \pi) = \sum_{t'=t}^{\infty} (\delta_i)^{t'-t} r_i(\mathbf{s}[t'], \mathbf{y}[t'], \pi(\mathbf{s}[t'], \mathbf{y}[t'])), \quad (13.48)$$

where π is the set of stationary Markov policies of all users. Note that $\pi_{i,t}$ is a Markov policy if $\pi_{i,t}$ is independent of the past, given the current state $s_i[t]$ and channel status $\mathbf{y}[t]$.

The best-response policy can be defined as:

$$\mathscr{B}_{i}(\boldsymbol{\pi}) = \arg\max_{\pi_{i}} Q_{i,t}(\mathbf{s}[t], \mathbf{y}[t], (\pi_{i}, \boldsymbol{\pi}_{-i})), \qquad (13.49)$$

where π_{-i} is the set of policies of all secondary users except user *i*. Note that the bestresponse policy defined in (13.49) can be obtained only when the state and policy of other secondary users are available. To overcome this limitation, the concept of *preference* is introduced. Preference $u_{c,i}[t]$ is the benefit for secondary user *i* in accessing channel *c* in time slot *t*. This preference can be considered the optimal bid submitted to the spectrum broker. Since secondary user *i* will bid to maximize its immediate reward, this preference can be defined as function of its own state and channel status. This is referred to as the myopic strategy. However, if the impact of future reward is taken into account, the secondary user needs to predict future reward with discount factor δ_i . This prediction can be performed based on past observation.

13.5.4 Learning algorithm

A learning algorithm similar to reinforcement learning [503, 471] is developed to predict the impact of the strategy adopted by other secondary users. This learning is based on observations $o_i[t] = \{s_i[0], \mathbf{y}[0], \mathbf{a}_i[0], p_i[0], \dots, s_i[t-1], \mathbf{y}[t-1], \mathbf{a}_i[t-1], p_i[t-1], s_i[t], \mathbf{y}[t]\}$ of all local and observable information. The secondary user is characterized by its class. In this case, K_i classes are defined and the state space of other users is mapped to each class. Given this class, approximated preference, optimal bidding policy, and approximated transition probability can be obtained. The average future reward is denoted by $V_{i,t}((s_i, s_{-i}), \mathbf{y}[t])$. The learning algorithm is presented as Algorithm 13.6.

Algorithm 13.6 Learning algorithm for radio resource competition.

- Initialize average future reward V_{i,0}((s_i, š_{-i}), y) for all possible states s_i of user i and approximated states š_{-i} of other users.
- 2: **loop**
- 3: Observe current state $s_i[t]$ and channel status $\mathbf{y}[t]$.
- 4: Choose action (i.e., bid) $\mathbf{a}_i[t] = [u_{1,i}[t], \dots, u_{c,i}[t], \dots, u_{C,i}[t]]$ according to approximated reward $V_{i,t+1}((s_i[t+1], \tilde{s}_{-i}[t+1]), \mathbf{y}[t+1])$.
- 5: Submit action $\mathbf{a}_i[t]$ to spectrum broker.
- 6: Receive channel allocation $\mathbf{z}_i[t]$ and payment $p_i[t]$.
- 7: Compute the approximated states $\tilde{s}_{-i}[t]$ of other secondary users.
- 8: Compute the expected total discounted sum of rewards $Q_{i,t}((s_i[t], \tilde{s}_{-i}[t]), \mathbf{y}[t], \pi).$
- 9: Update future reward $V_{i,t}((s_i, \tilde{s}_{-i}), \mathbf{y})$ using learning rate $\phi_i[t]$.
- 10: t = t + 1.
- 11: end loop

In the performance evaluation, different approaches to the game of spectrum bidding for secondary users are considered, i.e., fixed bidding, source-award bidding, myopic bidding, and best-response learning bidding strategies. In the fixed-bidding strategy, a constant bid vector is adopted by the secondary user. In the source-award bidding strategy, the bidding action is obtained optimally considering only the buffer occupancy. In the myopic strategy, the bidding action is chosen to maximize immediate reward. It is found that as the secondary user has more information to choose an action, the reward is higher (i.e., packet loss due to lack of buffer space and payment are smaller). In particular, the source-award bidding strategy is observed to outperform the fixed-bidding strategy. The myopic bidding strategy is better than the source-aware bidding strategy, and the bestresponse learning bidding strategy yields the best performance because of the ability to learn and adapt to both environment and opponent changes.

13.6 Cheat-proof strategies for open spectrum sharing

With non-cooperative secondary users, it is mostly found that the performance of all users is not maximized because of selfishness (e.g., all secondary users transmit at the highest power). This effect is known as the *tragedy of the commons* [194, 145]. The efficient or optimal strategy (e.g., secondary users transmit at a lower power level to avoid severe interference) is more desirable if secondary users can cooperate. However, because of the rationale to maximizing their own benefit, some secondary users may deviate from cooperation and even cheat, to reporting false private information to gain a higher payoff. Therefore, an enforcement method/strategy will be required to ensure that secondary users always cooperate and adopt an efficient strategy. A cheat-proof strategy for secondary users in open spectrum sharing (i.e., the common-use model) is proposed in [511].

13.6.1 One-shot non-cooperative game

There are *N* secondary users (i.e., *N* pairs of transmitter and receiver) whose set is denoted by \mathcal{N} . These secondary users share a single channel with the primary user. With CDMA channel access, the received signal of secondary user *i* at time slot *t* is

$$\psi_i[t] = \sum_{j=1}^{N} h_{j,i}[t] \chi_j[t] + w_i[t], \qquad (13.50)$$

where $\chi_j[t]$ is the transmitted information, $h_{j,i}$ is the instantaneous channel gain from transmitter *j* to receiver *i*, and $w_i[t]$ is the white noise at receiver *i*. The secondary user $i \in \mathcal{N}$ can adjust transmit power P_i (i.e., strategy). The payoff is defined as the transmission rate $R_i(\cdot)$ achieved by secondary user *i*, which can be approximated by

$$R_i(P_1, \dots, P_i, \dots, P_N) = \log\left(1 + \frac{P_i |h_{i,i}|^2}{\sigma^2 + \sum_{j \neq i} P_j |h_{j,i}|^2}\right),$$
(13.51)

where σ^2 is the noise power. In this case, the mutual interference is treated as noise with Gaussian random variable. Given the maximum transmit power P_i^{max} for secondary user *i*, the Nash equilibrium is found to be $\mathbf{p}^* = (P_1^{\text{max}}, \dots, P_i^{\text{max}}, \dots, P_N^{\text{max}})$. It is shown by contradiction that if secondary user *i* decreases transmit power to $P_i < P_i^{\text{max}}$, the transmission

rate $R_i(\cdot)$ will not be maximized. With Nash equilibrium, the transmission rate of selfish secondary user *i* with channel gain $h_{j,i}$ in one time slot is denoted by $R_i^{\text{slf}}(h_{1,i}, \dots, h_{N,i})$. The expected payoff is denoted by $\overline{R}_i^{\text{slf}} = E_{h_{i,i};j \in \mathcal{N}} (R_i^{\text{slf}}(h_{1,i}, \dots, h_{N,i}))$.

13.6.2 Cooperative strategy

It is proposed that the optimal transmission rate for all secondary users can be achieved when a single user is allowed to transmit in the time slot [511]. The main reason is that there is no interference and it is simple to implement this transmission policy. Also, it is easy to detect deviating secondary users if they start transmission in the same time slot. Since only one secondary user can transmit data in one time slot (i.e., time-division multiple-access or TDMA), a transmission policy (i.e., who should transmit in the current time slot) must be determined. Two criteria to evaluate candidate transmission policy are considered: maximum total throughput [187] and proportional fairness [242]. The policy which maximizes total throughput is defined as $L_1 = \arg \max_{P_1,...,P_i,...,P_N} \sum_{i=1}^N R_i(P_1,...,P_i,...,P_N)$ and that which achieves proportional fairness is defined as $L_2 = \arg \max_{P_1,...,P_i,...,P_N} \prod_{i=1}^N R_i(P_1,...,P_i,...,P_N)$. For policy L_1 to maximize total throughput, the secondary user with the highest channel quality $\gamma_i = P_i^{\max} |h_{i,i}[t]|^2$ will be selected for transmission in each time slot. In this case, the transmission rate of secondary user *i* under cooperation to maximize total throughput is given by

$$\hat{R}_{L_{1},i}^{\text{cop}} = \int_{0}^{\infty} \log\left(1 + \frac{\gamma_{i}}{\sigma_{i}^{2}}\right) \Pr(\gamma_{i} \ge \max_{j \neq i} \gamma_{j}) f(\gamma_{i}) \mathsf{d}\gamma_{i}, \qquad (13.52)$$

where $f(\gamma_i)$ is the probability density function of random variable γ_i . Note that γ_i is an i.i.d. exponentially distributed random variable with mean $P_i^{\max}\nu_i$, where ν_i is the parameter of Rayleigh fading. However, the shortcoming of this transmission policy in maximizing the total transmission rate is its lack of fairness. A secondary user with good channel quality (e.g., close to the receiver) will have a much higher transmission rate than those with poor channel quality (e.g., far from the receiver). To solve this problem, a policy to achieve proportional fairness is studied. In transmission policy L_2 , the secondary user with the highest normalized channel quality $\check{\gamma}_i = \gamma_i / E(\gamma_i)$ is selected for transmission in a time slot. In this case, the normalized channel gain is exponentially distributed with mean 1. Therefore, every secondary user will have an equal chance of 1/N to transmit data. The transmission rate of this policy L_2 is

$$\hat{R}_{L_{2},i}^{\text{cop}} = \int_{0}^{\infty} \log\left(1 + \frac{P_{i}^{\max}\nu_{i}\check{\gamma}}{\sigma_{i}^{2}}\right) \exp(-\check{\gamma})(1 - \exp(-\check{\gamma}))^{N-1} \mathsf{d}\check{\gamma}.$$
(13.53)

Given the above transmission policies, in each time slot a secondary user has to detect whether other users are transmitting in the same time slot. Since users cannot detect and transmit data simultaneously in one time slot, the secondary user will be in a detection period with length αT to listen to the channel and observe the transmission signals from other users. T is the length of the time slot and α is the proportion of time for detection. However, detection is not perfect, and false alarms can occur (i.e., a transmission signal by another user is falsely detected by the transmitting secondary user). This false-alarm event reduces the performance of all secondary users. In this case, increasing the length of the detection period can decrease the false-alarm probability, but at the cost of shorter transmission times. Given from this tradeoff, a method to obtain the optimal length $\alpha^* T$ of the detection period is also proposed in [185] given that an energy-detection algorithm is used by the secondary user. With optimal α^* , the transmission rate for the secondary user *i* with a cooperative strategy is denoted by $R_i^{cop} = (1 - \alpha^*) \hat{R}_{l,i}^{cop}$.

13.6.3 Repeated games

Secondary users can be selfish by transmitting using Nash equilibrium (i.e., maximum power), and experience high interference. The expected transmission rate of selfish secondary user *i* is denoted by $\overline{R}_i^{\text{slf}}$. Alternatively, secondary users can cooperate by transmitting using a cooperative policy. The corresponding expected transmission rate is denoted by $\overline{R}_i^{\text{cop}}$. In general, a cooperative strategy will yield a higher transmission rate (i.e., $\overline{R}_i^{\text{slf}} < \overline{R}_i^{\text{cop}}$). However, any secondary user can deviate to gain a higher transmission rate when all other users adopt a cooperative strategy. The transmission rate of the deviating user *i* in one time slot is denoted by R_i^{dev} . To prevent any secondary user from deviating, a punishment-based repeated game is formulated so that all secondary users have an incentive to maintain cooperation. In this repeated game, the payoff is defined as the sum of discounted transmissions

$$U_{i} = (1 - \delta) \sum_{t=0}^{\infty} \delta^{t} R_{i}[t], \qquad (13.54)$$

where $0 < \delta < 1$ is the discount factor. The first punishment policy considered in [511] is a *trigger* punishment, where all secondary users initially cooperate. However, if any secondary user deviates from cooperation, all users will punish by playing the Nash equilibrium strategy forever. It is proved in [511] that as $\delta \to 1^-$ (i.e., the transmission rate in the future is weighted close to the current transmission rate), the utility of deviation U_i^{dev} almost surely converges to $\overline{R}_i^{\text{slf}}$, while the utility of cooperation U_i^{cop} almost surely converges to $\overline{R}_i^{\text{cop}}$. Therefore, a rational secondary user will maintain cooperation if $U_i^{\text{cop}} > U_i^{\text{dev}}$. This is the self-enforced policy.

However, this *trigger* policy has the shortcoming that the deviating user will be punished forever, and that other cooperative users will be affected by receiving low transmission rates. Therefore, a policy to punish for a certain period of time and then forgive the deviating user is introduced (i.e., *punish-and-forgive*). This policy aims to prevent rather than avenge deviation. With this policy, if a deviation is detected, all secondary users punish the deviating user by adopting Nash equilibrium for the next T - 1 time slots. Then a cooperative strategy is applied again, after T - 1 time slots. T is referred to as the duration of punishment. With this policy, the condition for subgame-perfect equilibrium can be obtained as follows. Given $\overline{R}_i^{cop} > \overline{R}_i^{slf}$ for all i = 1, ..., N, there is a minimum value δ_{min} of the discount factor such that if $\delta > \delta_{min}$, then the

repeated game has a subgame-perfect equilibrium. The proof is based on the Folk Theorem [159]. The parameter T can be obtained such that subgame-perfect equilibrium is achieved (i.e., given the discount factor δ). If the maximum transmission rate gained from the deviation is \hat{R}_i^{dev} , the necessary condition for T is

$$T > \max_{i} \frac{\log\left(\delta - \frac{(1-\delta)\hat{R}_{i}^{dev}}{\overline{R}_{i}^{cop} - \overline{R}_{i}^{sff}}\right)}{\log \delta}.$$
(13.55)

This condition is obtained from the fact that the expected utility of cooperation must be greater than that of deviation, i.e., $\overline{U}_i^{cop} > \overline{U}_i^{dev}$. These expected utilities are

$$\overline{U}_{i}^{\text{cop}} = E[U_{i}^{\text{cop}}] \ge (1 - \delta) \left(\sum_{t=1}^{t_{0}-1} \delta^{t} \overline{R}_{i}^{\text{cop}} + \sum_{t=t_{0}+1}^{\infty} \delta^{t} \overline{R}_{i}^{\text{cop}} \right),$$
(13.56)

$$\overline{U}_{i}^{\mathsf{dev}} = E[U_{i}^{\mathsf{dev}}] \leq (1-\delta) \left(\sum_{t=0}^{t_{0}-1} \delta^{t} \overline{R}_{i}^{\mathsf{cop}} + \delta^{t_{0}} \hat{R}_{i}^{\mathsf{dev}} + \sum_{t=t_{0}+1}^{t_{0}+U-1} \delta^{t} \overline{R}_{i}^{\mathsf{slf}} + \sum_{t=t_{0}+U}^{\infty} \delta^{t} \overline{R}_{i}^{\mathsf{cop}} \right).$$
(13.57)

13.6.4 Cheat-proof strategy

Given a cooperative strategy, the secondary user with the highest channel quality is selected for transmission in one time slot in order for the policy to maximize the total transmission rate. Secondary users have an incentive to report false values of channel quality γ_i and they will do so since there is no centralized controller in the system. Therefore, a cheat-proof strategy is introduced so that secondary users will report true information. This cheat-proof strategy is based on the concept of *transfer*. When secondary users report a high value of channel quality, they will be asked to pay a tax which will increase as the value of reported channel quality increases. The tax will be paid to the secondary users reporting low value of channel quality. This is referred to as the transfer in Bayesian mechanism design theory [160]. In this case, the utility for the secondary user includes both that from transmission rate and the transfer.

With cooperation, private information on channel quality $\{\gamma_1, ..., \gamma_i, ..., \gamma_N\}$ is exchanged among all secondary users. Let $\{\tilde{\gamma}_1, ..., \tilde{\gamma}_i, ..., \tilde{\gamma}_N\}$ denote the actual channel quality at one time slot. Secondary user *i* reports $\hat{\gamma}_i$ which may not be the same as $\tilde{\gamma}_i$. In this case, $\{\hat{\gamma}_1, ..., \hat{\gamma}_i, ..., \hat{\gamma}_N\}$ is common knowledge for all secondary users, but $\{\tilde{\gamma}_1, ..., \tilde{\gamma}_i, ..., \tilde{\gamma}_N\}$ is private information. The transfer of secondary user *i* in the cheat-proof strategy is defined as

$$\phi_i(\hat{\gamma}_1, \dots, \hat{\gamma}_i, \dots, \hat{\gamma}_N) = \Phi_i(\hat{\gamma}_i) - \frac{1}{N-1} \sum_{j=1, j \neq i}^N \Phi_j(\hat{\gamma}_j), \quad (13.58)$$

where

$$\Phi_i(\hat{\gamma}_i) = E\left(\sum_{j=1, j\neq i}^N R_j(\gamma_j, \mathscr{L}_1(\gamma_1, \dots, \gamma_j, \dots, \gamma_N))\right| \gamma_i = \hat{\gamma}_i\right), \quad (13.59)$$

and $\mathscr{L}_1(\gamma_1, ..., \gamma_i, ..., \gamma_N) = \arg \max_i \gamma_i$ to maximize the total transmission rate. Expectation $E(\cdot)$ is taken over all values of $\{\gamma_1, ..., \gamma_i, ..., \gamma_N\}$. In this case, $\Phi_i(\hat{\gamma}_i)$ is the sum of all expected data rates from all other users given that secondary user *i* reports $\hat{\gamma}_i$. If secondary user *i* reports $\hat{\gamma}_i > \tilde{\gamma}_i$, this user will have a higher chance to transmit data, but the higher transfer has to be paid to other users. It is proved that with this transfer, all secondary users will report true private information, i.e., $\hat{\gamma}_i = \tilde{\gamma}_i$. The proof is based on the fact that it is the best response for the secondary user to report true private information given the transfer. That is, the utility will be maximized only when the secondary user reports $\tilde{\gamma}_i$.

For the policy to achieve proportional fair performance, secondary user *i* reports normalized channel quality $\check{\gamma}_i$. In this case, if the secondary user reports true information, in the long term each user will receive a 1/N time-share of the transmission. If any secondary user receives a time-share greater than $1/(N + \kappa)$, then this user will be identified as the deviating user and will be punished. κ is the tolerance margin. With this statistics-based strategy, it is shown that the gain from cheating is bounded.

From the performance evaluation, it is shown that as the interference level increases, the transmission rate of a selfish secondary user (i.e., the Nash equilibrium of a non-cooperative game) decreases. However, for a cooperative secondary user, the transmission rate remains constant since only one user will transmit at a time. For the punishment, it is found that as the discount factor increases, the punishment duration becomes shorter. Since the user is more concerned about the payoff in the future, the punishment duration can be decreased so that perfect Nash equilibrium is achieved. Also, it is shown that the expected overall payoff (i.e., transmission rate and transfer) is maximized only when the secondary user reports true information about channel quality.

13.7 Spectrum leasing and cooperation

In cognitive-radio networks, especially with the exclusive-use model, the primary user needs an incentive to share spectrum with secondary users. This incentive could be through pricing. Alternatively, secondary users can help the primary user to transmit data so that transmission rate and reliability are improved. In [460], an incentive is considered in which the cooperative diversity technique (e.g., decode-and-forward) is used so that secondary users can relay the transmitted data of the primary user. In return, the primary user allows secondary users to access the spectrum. This exchange of resources can be formulated as a hierarchical game.

There is one primary user with one channel, and a group of secondary users, whose set is denoted by \mathcal{N}_{all} . A subset of secondary users denoted by $\mathcal{N} \subseteq \mathcal{N}_{all}$ participates in relaying the transmitted data of the primary user, so these secondary users share the


Fig. 13.11 (a) Primary transmitter transmits to primary receiver and secondary users; (b) secondary users relay received data to primary receiver; (c) secondary users transmit their own data.

spectrum allocated by the primary user. As shown in Fig. 13.11, with a TDMA system between primary user and secondary users, a portion $1 - \alpha$ ($0 \le \alpha \le 1$) of the time slot is used by the primary transmitter to transmit the data. This transmitted data is received by both the primary receiver and the secondary users. A portion $\alpha\beta$ (i.e., $0 \le \beta \le 1$) of the time slot is used by the secondary users to transmit (i.e., relay) the received data to the primary receiver. Finally, a portion $\alpha(1 - \beta)$ of the time slot is used by the secondary users to transmit among themselves. The transmission by secondary users is based on the CDMA system.

13.7.1 Game formulation with instantaneous CSI

Given the above system model of a cognitive-radio network, two game models are formulated. In the first, players are the primary user and a group of secondary users. The strategy of the primary user is a time-share to allow the secondary users to access the spectrum (i.e., α and β) and a set of secondary users \mathcal{N} to relay the transmitted data. The strategy of a secondary user is the transmit power to relay the transmitted data of the primary user. The payoff for both primary user and secondary users is the transmission rate. Since the primary user can control the time-share and select the set of secondary users, the primary user is considered to be the leader while the secondary users are



Fig. 13.12 Interaction among primary user and secondary users.

considered to be followers. With this structure, the first game is based on the Stackelberg game. The primary user will choose an optimal strategy under the assumption that the secondary users will perform their best-response strategies as well. However, in this case, the primary user can choose a strategy before the secondary users. In other words, once the primary user determines the strategy, the secondary users in set \mathcal{N} compete with each other to select a transmit power to maximize their transmission rates. Note that the transmit power to help the primary user is assumed to be the same as that used for communication among the secondary users. This is the second game model, based on a non-cooperative game. The interaction of these game models is shown in Fig. 13.12.

Game of primary user

For the primary transmitter and primary receiver, the objective is to maximize the transmission rate $R_P(\alpha, \beta, \mathcal{N})$ whose decision parameters (i.e., strategy) are α , β , and \mathcal{N} . If $\alpha = 0$, i.e., the primary user does not allow secondary users to relay the transmitted data and there is no time-share for secondary users, the transmission rate of the primary user, given perfect instantaneous channel-state information (CSI), is

$$R_{\rm dir} = \log\left(1 + \frac{|h_{\rm P}|^2 P_{\rm P}}{\sigma^2}\right),\tag{13.60}$$

where h_P is the instantaneous channel gain between primary transmitter and primary receiver, P_P is the transmit power of the primary transmitter, and σ^2 is the noise power. However, if $\alpha > 0$, the decode-and-forward multi-hop and space-time coding is used for the transmission between primary transmitter and secondary users [280, 428]. In this case, the transmission rate of the primary user is

$$R_{\rm cop}(\alpha,\beta,\mathcal{N}) = \min\left((1-\alpha)R_{\rm PS}(\mathcal{N}),\alpha\beta R_{\rm SP}(\alpha,\beta,\mathcal{N})\right),\qquad(13.61)$$

where $R_{PS}(\mathcal{N})$ and $R_{SP}(\alpha, \beta, \mathcal{N})$ are the transmission rates from the primary transmitter to the set of secondary users \mathcal{N} and from the secondary users in the same set to the primary receiver, respectively. These transmission rates are

$$R_{\mathsf{PS}}(\mathcal{N}) = \log\left(1 + \frac{\min_{i \in \mathcal{N}} |h_{\mathcal{PS},i}|^2 P_{\mathcal{P}}}{\sigma^2}\right),\tag{13.62}$$

$$R_{\rm SP}(\alpha,\beta,\mathcal{N}) = \log\left(1 + \sum_{i\in\mathcal{N}} \frac{|h_{\rm SP,i}|^2 P_i^*(\alpha,\beta,\mathcal{N})}{\sigma^2}\right),\tag{13.63}$$

where $h_{PS,i}$ and $h_{SP,i}$ are the instantaneous channel gains from primary transmitter to secondary user *i* and from secondary user *i* to primary receiver, respectively. $P_i^*(\alpha, \beta, \mathcal{N})$ is the transmit power of secondary user *i* to relay the transmitted data of the primary user. This transmit power will be the outcome of a competition among the secondary users, i.e., Nash equilibrium of the corresponding non-cooperative game (Fig. 13.12). The transmission rate in (13.63) is obtained as the information-theoretic bound of orthogonal space-time codes (STCs). Given the strategy α and β , the transmission rate of the primary user can be expressed as

$$R_{\mathsf{P}}(\alpha,\beta,\mathcal{N}) = \begin{cases} R_{\mathsf{dir}}, & \text{if } \alpha = 0, \\ R_{\mathsf{cop}}(\alpha,\beta,\mathcal{N}), & \text{if } \alpha > 0. \end{cases}$$
(13.64)

The objective of the primary user is to maximize this rate, i.e.,

$$\max_{\alpha,\beta,\mathcal{N}} R_{\mathsf{P}}(\alpha,\beta,\mathcal{N}), \text{ s.t. } \mathcal{N} \subseteq \mathcal{N}_{\mathsf{all}}, \quad 0 \le \alpha,\beta \le 1.$$
(13.65)

Game of secondary users

Given the strategy of the primary user, the non-cooperative game among the secondary users can be formulated as follows. The players are the secondary users in set N, chosen by the primary transmitter. The strategy of each secondary user is the transmit power P_i . The secondary users use the same transmit power P_i for relaying the transmitted data of the primary user and for data transmission to secondary receivers. Therefore, P_i as defined in (13.63) becomes the strategy of the secondary users. The payoff is the transmission rate minus the cost due to the transmit power. The transmission rate between secondary transmitters and secondary receivers, based on CDMA, $\alpha(1-\beta)R_i(P_i, \mathbf{p}_{-i})$, where

$$R_{i}(P_{i}, \mathbf{p}_{-i}) = \log\left(1 + \frac{|h_{\mathsf{S},i}|^{2}P_{i}}{\sigma^{2} + \sum_{j \neq i} |h_{\mathsf{S},i,j}|^{2}P_{j}}\right),$$
(13.66)

 $h_{S,i}$ is the channel gain between secondary transmitter and secondary receiver *i*, and \mathbf{p}_{-i} is a vector of transmit power of all secondary users in \mathcal{N} except user *i*. In this case, the payoff function of secondary user $i \in \mathcal{N}$ is

$$u_i(P_i, \mathbf{p}_{-i}) = \alpha \left((1 - \beta) R_i(P_i, \mathbf{p}_{-i}) - \omega P_i \right), \qquad (13.67)$$

where ω is the cost constant. In this case, α can be omitted, and the transmission rate of a secondary user can be expressed as $R_{SP}(\beta, \mathcal{N})$. It is proved in [460] that, with $\alpha > 0$,

a Nash equilibrium P_i^* exists and is unique if $\sum_{j \in \mathcal{N}} \frac{|h_{\mathsf{S},i,j}|^2}{|h_{\mathsf{S},i}|^2} < 1$. The proof is based on the KKT condition [431, 139].

Then the optimal strategy of the primary user can be obtained. In particular, the optimal strategy α^* (i.e., time-share for secondary users) will be positive if and only if there exists a set of secondary users $\mathcal{N} \subseteq \mathcal{N}_{all}$ such that

$$\frac{\beta^* R_{\mathsf{SP}}(\beta^*, \mathcal{N}) R_{\mathsf{PS}}(\mathcal{N})}{\beta^* R_{\mathsf{SP}}(\beta^*, \mathcal{N}) + R_{\mathsf{PS}}(\mathcal{N})} > R_{\mathsf{dir}},$$
(13.68)

where the optimal strategy β^* (i.e., time-share for secondary users to relay the transmitted data from the primary transmitter) is

$$\beta^* = \arg\max_{\beta} \beta R_{\mathsf{SP}}(\beta, \mathcal{N}). \tag{13.69}$$

In particular, the optimal strategy α^* is

$$\alpha^* = \frac{1}{1 + \frac{\beta^* R_{\text{SP}}(\beta^*, \mathcal{N})}{R_{\text{RS}}(\mathcal{N})}}.$$
(13.70)

The optimal set \mathcal{N}^* of secondary users to participate in cooperative transmission can be obtained by enumeration.

From the performance evaluation in [460], it is observed that the primary user should select the secondary users with high channel gain for relay data transmission. An approximate algorithm to select only the optimal set \mathcal{N} of secondary users with the highest channel gains is introduced. The transmission rate obtained from the approximate algorithm is observed to be close to that from enumeration, which is optimal [460]. Also, it is found that as the interference among secondary users decreases, they can reduce the transmission rate to maximize their payoff, so smaller transmit power will be used for relay transmission for the primary user.

13.7.2 Game formulation with long-term CSI

Without instantaneous CSI, the game model can be modified for probabilistic parameters. For example, the primary user cannot select a subset of secondary users \mathcal{N} which can decode and relay data for a given channel realization. In this case, long-term CSI is used in a game where the primary user and secondary users optimize their strategies. Randomized distributed space-time coding (DSTC) is used in this case. Each secondary user selected by the primary user will select a codeword within the orthogonal space-time codebook, randomly and independently. The payoff to the primary user is changed to be the outage probability to be minimized, which is more suitable to represent performance in the long term. For transmission, there is a minimum SNR γ_{th} or threshold to guarantee a target BER for a given transmission rate. The outage probability $P_{\text{out}}(\alpha, \beta, \mathcal{N})$ is defined as the probability that the SNR of the relay transmission from the secondary users to the primary receiver is smaller than the threshold, i.e.,

$$P_{\text{out}}(\alpha,\beta,\mathcal{N}) = \Pr\left(\gamma_{\text{SP}}(\beta,\mathcal{N}) < \gamma_{\text{th}}\left(\frac{\overline{R}_{\text{P}}}{\alpha\beta R_{\text{STC},i}}\right)\right), \quad (13.71)$$

where \overline{R}_{P} is an overall transmission rate and $R_{STC,i}$ is the reduced rate of orthogonal STCs of secondary user *i*. For secondary users, long-term channel gain applies, so the same game formulation can be formulated and solved directly. Thus the optimal strategy for the primary user is obtained in a similar way to the case of instantaneous CSI.

13.8 Service-provider competition for dynamic spectrum allocation

In the exclusive-use model, service providers (i.e., primary users) bid and buy spectrum from a regional broker [87, 161], e.g., a government agency, and then resell the spectrum as a service to users (i.e., secondary users). This dynamic spectrum sharing is considered in [15]. In the system, there are M service providers and N users. The size of the nonoverlapping spectrum used for communication between service provider m and user i is denoted by $x_{m,i}$. This is similar to an OFDMA network. If $x_{m,i} = 0$, user *i* does not buy service from provider m. Let the channel gain from the base station of service provider m to user i be denoted by $h_{m,i}$, and the spectral efficiency (i.e., channel quality) for the transmission between the base station and user be denoted by $k_{m,i} = \log \left(1 + \frac{\nu_m h_{m,i}}{\sigma^2}\right)$, where ν_m is the spectral density and σ^2 is the noise power. The transmission rate of user i to service provider m is $r_{m,i} = k_{m,i} x_{m,i}$. The user is characterized by a utility function $\mathcal{U}_i(r_i)$ of the transmission rate, where r_i is the total transmission rate, i.e., $r_i = \sum_{m=1}^{M} r_{m,i} x_{m,i}$. This utility function is assumed to be increasing and concave in r_i . In this system, users want to maximize their utility while the service provider wants to maximize profit. This situation is formulated as a Stackelberg game [15]. The players of this game are the service provider (i.e., leader) and the users (i.e., followers). The strategy of the service provider is the price p_m , and the strategy of the users is the size of spectrum $x_{m,i}$ that each will buy from the service provider. The payoff to the service provider is the profit, while that to the user is the net utility. The two main components of this game model are the profit for users of the service provider and utility maximizations (Fig. 13.13).

13.8.1 User demand

The users have to determine the size of the spectrum to buy from the service provider so that their payoffs are maximized. Given a price p_m charged by the service provider per unit of spectrum, the payoff (i.e., net utility) to the user is

$$u_{i} = \mathscr{U}_{i}(r_{i}) - \sum_{m=1}^{M} p_{m} x_{m,i}.$$
(13.72)



Fig. 13.13 Spectrum broker, service provider, users, and their interactions.



Fig. 13.14 Example of an exponential utility function.

To maximize the payoff for user *i*, the Lagrangian multiplier $\alpha_{m,i}$ for the constraint $x_{m,i} \ge 0$ can be expressed as

$$\mathscr{L}_{i} = \mathscr{U}_{i}(r_{i}) - \sum_{m=1}^{M} p_{m} x_{m,i} + \sum_{m=1}^{M} \alpha_{m,i} x_{m,i}.$$
 (13.73)

It is found that the user will buy service from provider *m* rather than *m'* (i.e., $x_{m,i} > 0$ and $x_{m',i} = 0$) if $\frac{p_m}{k_{m,i}} < \frac{p_{m'}}{k_{m',i}}$. The effective price charged by service provider *m* to user *i* can be defined as $\rho_{m,i} = \frac{p_m}{k_{m,i}}$ and, in the general case, a user will buy service from the provider with the lowest effective price.

An exponential utility function is defined by

$$\mathscr{U}_{i}(r_{i}) = \Gamma_{i}\left(1 - e^{-r_{i}/\Gamma_{i}}\right), \qquad (13.74)$$

where Γ_i is the constant of the utility function. An example of an exponential utility function is shown in Fig. 13.14. The demand function (i.e., the size of spectrum given



Fig. 13.15 Example of demand of spectrum from user with $k_{m,i} = 3$.

the price) can be obtained by maximizing the payoff u_i in (13.72). This demand function is

$$x_{m,i}(p_m) = \max\left(0, \frac{\Gamma_i}{k_{m,i}}\log\left(\frac{k_{m,i}}{p_m}\right)\right) = \max\left(0, \frac{\Gamma_i}{k_{m,i}}\log\left(\frac{1}{\rho_{m,i}}\right)\right).$$
(13.75)

An example of a demand function is shown in Fig. 13.15.

13.8.2 Optimal price

Given the demand function (13.75), a service provider can maximize profit by adjusting the price p_m . The special case of a single service provider is considered here. The revenue of service provider *m* is the price paid by all users, i.e., $p_m \sum_{i=1}^{N} x_{m,i}$. The cost for the service provider is a result of buying spectrum from the broker, and of the transmit power. The former is denoted by $\omega_{bro} \sum_{i=1}^{N} x_{m,i}$ and the latter by $\omega_{pow} \nu_m \sum_{i=1}^{N} x_{m,i}$, where ω_{bro} and ω_{pow} are the cost constants arising from bidding for spectrum from the broker and from transmit power, respectively. Therefore, the profit to service provider *m* is

$$F_m = \left(p_m - \omega_{\text{bro}} - \omega_{\text{pow}}\right) \sum_{i=1}^N x_{m,i}.$$
(13.76)

Examples of profit functions for various prices p_m from a single user are shown in Fig. 13.16. However, if there are multiple users whose parameters (e.g., Γ_i and $k_{m,i}$) are different, the profit function is not concave. Only numerical methods can be applied to obtain optimal prices. Naturally, for the service provider, the price to charge a user has to satisfy the condition $p_m > \omega_{bro} + \omega_{pow}$, which ensures that the service provider gains a positive profit. Moreover, the user will buy service only if $k_{m,i} > p_m$. Since $k_{m,i}$ can be considered as the marginal utility, the price has to be higher than this marginal utility so that the user has an incentive to buy service from provider *m*. From these conditions, the minimum effective price to service provider *m* is $\rho_{m,i}^{\min} = \frac{\omega_{bro} + \omega_{pow}}{k_{m,i}}$.

For multiple service providers, it is found that the users will be partitioned into different groups, each buying service from a different provider. For example, with two service



Fig. 13.16 Examples of profit to a service provider with $\Gamma_i = 10$.

providers, if *N* users have channel gain $h_{m,i}$ (i.e., between provider *m* and user *i*) which can be arranged as follows:

$$h_{1,1} > h_{1,2} > \dots > h_{1,N}$$
, and $h_{2,1} < h_{2,2} < \dots < h_{2,N}$, (13.77)

then there is a user i^* such that users $1, ..., i^*$ will buy service from provider 1 and the rest, i.e., users $i^* + 1, ..., N$, will buy service from provider 2.

13.8.3 Related work

In [239], a hierarchical spatial game model is formulated for two service providers and multiple users. The first part of this game model is the interaction among users (i.e., mobile subscribers) to choose a service provider. It is assumed that the users are spatially distributed between the base stations of two service providers (e.g., Hotelling's model [215]). The users can choose to use high-or low-priority services from either provider. This game is analyzed based on the size of spectrum and price charged by each service provider. It is found that a Wardrop equilibrium exists in this game. The second part of this game models the interaction between service providers to acquire spectrum from a regulator (i.e., broker) and set the price so that profit is maximized. The Nash equilibrium is considered to be the solution of this game. In [34], a similar system model is considered. In [34], the problem of base station placement is considered. Base station selection by users depends on SINR. A Stackelberg equilibrium is identified for this game formulation.

13.9 Summary

With cognitive radio, transmission by wireless users can be adapted to a changing environment. Rational users can adjust transmission parameters to achieve their benefits. Game theory has been applied to obtain optimal solutions for primary and secondary users in cognitive-radio networks with different types of spectrum sensing and access. First, secondary users can cooperate in channel sensing. Coalitions can be formed among secondary users to achieve optimal performance. Spectrum access in a shared-use model can be based on an underlay or overlay approach. For underlay spectrum access, power control is important for secondary users not only to achieve the highest transmission rate but also to maintain interference to the primary user below a target level. A noncooperative game model can be formulated to obtain the Nash equilibrium. For overlay spectrum access, medium access control is important to identify and access spectrum. Without complete information and a centralized controller, secondary users can learn from experience and make decisions to achieve the optimal solution. In spectrum access with the exclusive-use model, incentive for the primary user to allow secondary users to access the spectrum is an important issue. In this case, secondary users can relay the transmitted data of the primary user to improve performance. In exchange, the primary user shares the spectrum with secondary users. A Stackelberg game model has been formulated to obtain the optimal strategy for the primary user, given that the secondary users will maximize their own benefit. The incentive for spectrum sharing and access can be the price paid by secondary users to the primary user. In this case, the pricing scheme is important. A Stackelberg game model can be formulated in which the primary service provider chooses an optimal price while the secondary users determine their spectrum demand according to channel quality and price.

Communication networks such as the Internet are becoming more and more dependent on the interactions of intelligent devices that are capable of autonomously operating within a highly dynamic and rapidly changing environment. The dynamism and complexity of Internet networks is a consequence of their size, heterogeneity, traffic diversity, and decentralized nature. Next-generation communication networks such as the future Internet are envisioned to be self-organizing, self-configuring, selfprotecting, and self-optimizing. The applications and services that make use of these networks will also grow in complexity and impose stringent constraints and demands on network design: increased quality-of-service requirements for routing data, content distribution based on peer-to-peer networks, advanced pricing, and congestion control mechanisms, etc.

While these challenges were initially perceived with the emergence of the Internet, they are now essential in the design of every current and future network. To efficiently analyze and study such Internet-like networks, there is a need for a rich analytical framework such as game theory, whose models and algorithms can capture the numerous challenges arising in current and emerging communication networks. The challenges in designing Internet networks differ from those of their wireless counterparts in several aspects. In general, one does not need to worry about the reliability of the communication channel, as in the wireless case. But because Internet networks are generally composed of heterogeneous nodes having different capabilities and communicating over long paths and routes, the network size and as well as heterogeneity the nodes' capabilities, for example, play a role that is more critical in the design of Internet networks than in the wireless case. Moreover, issues such as pricing and service providers' competition or cooperation arise more frequently with Internet networks, owing to their wellestablished infrastructure and varied services. Therefore, to complement the development of game-theoretic models for the wireless case in the previous chapters, this chapter presents the use of game theory to tackle important challenges in Internet networks, in particular:

• **Routing and flow control:** Routing and flow control lie at the heart of the challenges facing any communication network. As the number of nodes and their autonomy increase, devising efficient algorithms for routing and flow control becomes more and more challenging. In Section 14.1, we tackle the problem of combined routing and

flow control in an Internet-like communication network, and show how, using game theory, nodes can individually control their transmission rates and select their network path, while taking into account the tradeoff between their throughput and their average delay.

- **Congestion control and pricing:** In a communication network, the users and the network service provider have conflicting objectives. While the provider has an incentive to optimize its prices so as to maximize its revenues from selling bandwidth, network users have an incentive to efficiently utilize the available network bandwidth by performing congestion control, i.e., controlling their transmission rates while minimizing the cost they pay for the used bandwidth. From the operator's perspective, this gives rise to an important tradeoff. On the one hand, if prices are kept very low, this will attract many users to the service; however, it will reduce the revenues of the provider. On the other hand, raising prices can increase the revenues, but it may drive off the demand for the service, which will potentially reduce revenues. From the users' perspective, while each user has an incentive to utilize the entire bandwidth for its own communication, this comes at the expense of an increased cost to use the bandwidth and a loss of performance (e.g., increased delay) due to the increasing traffic (from this user as well as other users) in the network. In Section 14.2, we discuss a gametheoretic model to tackle these two challenging problems for a network with a single provider.
- Revenue sharing between Internet service providers: The pricing of Internet services is a challenging issue, specifically when an end-to-end service needs to go through a number of service providers owned by different entities. In such a scenario, the self-interest of each provider in maximizing its own profit leads to an increase in the service price, which in turn will discourage users from utilizing the service. Therefore, providing incentives for multiple service providers to collaborate in pricing and fair revenue-sharing is an important challenge in the development of pricing mechanisms for Internet services. In Section 14.3, we analyze, using non-cooperative games, the inefficient pricing scheme that arises when providers act in complete non-cooperation. Then, examine see how the result can be improved by allowing some form of collaboration among the providers, even when these providers are still making independent and non-cooperative decisions.
- **Cooperative peer-to-peer file sharing:** Using a combination of advanced search and communication techniques as well as large-scale distributed file-storage systems, peer-to-peer file sharing networks allow a number of users to download and share content in a decentralized, scalable, and fault-tolerant manner. One important challenge in such networks is to improve download delays for peers when they are competing to download, concurrently, multiple resources from the same seed at the time the availability of the resources is announced. In Section 14.4, we discuss cooperative strategies, using coalitional-game theory, that enable peers to alleviate download delays during the phase of simultaneous download of resources at the time they become available. We show that, with a game-theoretic framework for this scenario, peers

can significantly improve their average download delay compared to the traditional non-cooperative case.

14.1 Combined flow control and routing in communication networks

With the emergence of large-scale communication networks such as the Internet and ATM networks, the need for robust resource and traffic management has significantly increased. In this context, algorithms for controlling the throughput and transmission rates of network nodes such as routers, service providers, and computing devices are critical for the efficient operation of the network. In particular, because of bandwidth constraints on the communication links and the traffic delay incurred on the communicating nodes, the design of efficient *flow-control* schemes is of central interest. Flow control is a scheme by which nodes having best-effort traffic adjust their instantaneous transmission rates depending on available bandwidth and delay requirements.

Routing is another important problem in the analysis of communication networks such as the Internet. Routing in such networks often involves the selection of a communication path that can optimize a desired quality-of-service parameter such as delay or throughput. In this regard, communication networks often use advanced routing algorithms by which nodes with conflicting objectives can agree on the path each must select in the network.

It is often the case that flow-control and routing decisions must be taken by the network nodes, i.e., the users, rather than by a centralized entity because every decision is made in such a way as to optimize each user's own objectives or performance criteria. Consequently, game-theoretic approaches for flow control and routing have been proposed in the literature, e.g., [200, 400, 379, 274, 27, 21, 28, 235] and references therein. First, when *jointly* considering flow control and routing, it is well known that, if the objective functions of network users are functions of a reward in terms of throughput and the sum of link costs, then the problem can be reduced to a routing game [200]. Furthermore, performing joint flow control and routing when the utilities are a function of a delay defined as the sum of all link capacities minus all link flows multiplied by an entropy yields a computable Nash equilibrium in the case of parallel links [400]. Other aspects of flow control and routing, such as the presence of heterogeneous services and robustness requirements, are studied in [379, 274, 27, 21, 235]. Additional information on the use of control and game theory in flow control and routing problems can be found in [55].

In this section, we are interested in studying, based on [28], the problem of combined flow control and routing in a large-scale network when the total throughput of the users is not fixed and when each user's utility is a function of the perceived tradeoff between delay and throughput. First, we consider the simpler case of a single user accessing multiple parallel links, and study the optimal solution. Using the results of the singleuser case, we study, using non-cooperative games and the concept of a Nash equilibrium, the case when the number of users is large. We determine all symmetric Nash equilibria, show that multiple equilibria exist in some cases, and discuss the features of the Nash equilibria in the asymptotic case.

14.1.1 Single user with multiple links

Consider a single user, i.e., a source node, that needs to send infinitesimally divisible traffic to a certain destination, over M possible communication links. Let \mathcal{M} denote the set of links in the network, and let c_m and λ_m denote, respectively, the capacity and throughput (transmission rate) of a link $m \in \mathcal{M}$. The links are labeled in such a way that

$$c_1 \geq \ldots \geq c_M$$
.

Given M/M/1 queues at each link, $\lambda_m > 0$ for at least one $m \in \mathcal{M}$, and the vector of transmission rates $\lambda = [\lambda_1, ..., \lambda_M]$, the average delay experienced by a given user will be

$$\tau(\boldsymbol{\lambda}) = \frac{\sum_{m=1}^{M} \frac{\lambda_m}{c_m - \lambda_m}}{\sum_{m=1}^{M} \lambda_m}.$$
(14.1)

The objective of each user is to optimize a metric that captures the tradeoff between the perceived delay and its achieved throughput. Although every user has an incentive to increase its throughput, this increase can yield an increased delay on the links, due to capacity constraints, as is clearly seen in (14.1). In any application where there is a need to optimize a throughput/delay tradeoff, it is of interest to utilize the concept of *power*, defined as the ratio between a power of the expected throughput and the expected delay. This concept is quite popular in communication networks whenever one needs to capture incentives for higher throughput while penalizing increasing delays. Formally, for the considered user, we define the following utility function:

$$u(\boldsymbol{\lambda}) = \frac{\left(\sum_{m=1}^{M} \lambda_m\right)^{\beta}}{\tau(\boldsymbol{\lambda})},$$
(14.2)

where $\beta \in (0, 1)$ is a parameter highlighting the tradeoff between throughput and delay. The utility in (14.2) represents the power perceived by the considered user in a communication network with multiple parallel links. For convenience, we define the function $l(\lambda)$ as the logarithm of (14.2), i.e.,

$$l(\boldsymbol{\lambda}) = (\beta + 1) \log\left(\sum_{m=1}^{M} \lambda_m\right) - \log\left(\sum_{m=1}^{M} \frac{\lambda_m}{c_m - \lambda_m}\right).$$
(14.3)

The objective of the user is to find the transmission rates vector λ in the set $C = [0, c_1] \times \cdots \times [0, c_m]$ that maximizes (14.3). For mathematical convenience, we

assume that $u(\mathbf{0}) = 0$ when $\lambda_m = 0$, $\forall m \in \mathcal{M}$, so although $l(\lambda)$ is not well-defined at the point **0** (the origin), this point cannot be considered as a solution. We note that the single-user case is an optimization problem that does not require any game-theoretic concepts for a solution, owing to the lack of any competitive or cooperative environment. However, the single-user case gives interesting insights and foundations for the case of multiple users. Thus, the use of game theory for the joint flow-control and routing problem will become clearer when we start dealing with multiple users in the next subsection.

First, we remark that the existence of an optimal solution to $l(\lambda)$ is guaranteed by the fact that C is compact and $l(\lambda)$ is continuous on C [28]. However, since $l(\lambda)$ is not concave, finding the optimal solution requires examining all stationary points as well as the values on the boundary of C. By performing this analysis, one can obtain an interesting result on the optimal point, as shown in [28]. Before presenting the result, we define a few necessary mathematical expressions. Define $w_m(\mu_m)$ as

$$w_m(\mu_m) = \frac{(\bar{c}_m - \mu_m \tilde{c}_m)^{(\beta+1)} \mu_m}{\tilde{c}_m - m \mu_m},$$
(14.4)

with $\bar{c}_m \triangleq \sum_{j=1}^m c_j$, $\tilde{c}_m \triangleq \sum_{j=1}^m \sqrt{c_j}$, and

$$\mu_{m} = \frac{(\beta+2)\tilde{c}_{m} - \sqrt{(\beta+2)^{2}\tilde{c}_{m}^{2} - 4m(\beta+1)\bar{c}_{m}}}{2(\beta+1)m}.$$
(14.5)

Furthermore, let \mathcal{M}_f denote the set of all integers in \mathcal{M} with the property that $m \in \mathcal{M}_f$ implies that $\mu_m < \sqrt{c_m}$, and

$$\frac{\tilde{c}_m^2}{m\bar{c}_m} \ge \frac{4(\beta+1)}{(\beta+2)^2} \tag{14.6}$$

is satisfied.

Given these definitions, we present the following interesting result on the flow control and routing problem for the single-user case.

THEOREM 14.1 For the single-user multiple-link combined flow control and routing problem, we have the following:

- 1. A transmission-rate vector that maximizes the power of the user, i.e., (14.2), exists.
- 2. The optimal solution dictates positive flows on links $1, ..., \hat{m}$ and zero flows on the remaining links $\hat{m} + 1, ..., M$ (if $\hat{m} < M$), where

$$\hat{m} = \underset{m \in \mathcal{M}_f}{\arg \max w_m(\mu_m)}.$$
(14.7)

3. The optimal flows are given by

$$\lambda_{j} = \begin{cases} c_{j} - \mu_{\hat{m}} \sqrt{c_{j}}, & j \in \mathcal{S}_{\hat{m}}, \\ 0, & j \in \mathcal{M} \setminus \mathcal{S}_{\hat{m}}, \end{cases}$$
(14.8)

where $\mu_{\hat{m}}$ is given by (14.5) and $S_{\hat{m}} = \{1, ..., \hat{m}\}$ is a subset of \mathcal{M} .

The proof of this theorem is found in [28] and is the result of developing candidate optimal solutions of the problem. Furthermore, in the case of equal link capacities, i.e., when $c_m = c$, $\forall m \in \mathcal{M}$, the optimal flows will be given by [28]

$$\lambda_m^* = \frac{\beta}{\beta + 1} c, \ \forall m \in \mathcal{M}.$$
(14.9)

The equal-links case exhibits two important robustness properties (detailed proofs are in [28]):

THEOREM 14.2 For any communication system with one user and M links, with the capacity of the mth link being $c + \delta_m$, there exits $\delta > 0$ such that, for $|\delta_m| < \delta$, $\forall m \in \mathcal{M}$, the utility-maximizing solution is unique and is an inner solution.

THEOREM 14.3 Consider a network with one user, M links, and the first \tilde{m} links having equal capacity c, with the remaining $M - \tilde{m}$ having capacities smaller than c. Then there exists a $\beta^* > 0$ that depends on c and $c_{\tilde{m}+1}$ such that, for $\beta \in (0, \beta^*)$, the unique utility-maximizing solution dictates zero flow over the links $\tilde{m} + 1, ..., M$; i.e.,

$$\lambda_{j}^{*} = \begin{cases} \frac{\beta}{\beta+1}c, & j \in \mathcal{S}_{\tilde{m}}, \\ 0, & j \in \mathcal{M} \setminus \mathcal{S}_{\tilde{m}}. \end{cases}$$
(14.10)

Theorem 14.3 implies that, starting with an original network with equal-capacity links, the solution remains intact even when links of lower capacity are added (i.e., the optimal solution dictates zero flow over these additional links), provided that β is sufficiently small.

The above theorems for the single-user case will be used in proving many of the properties and results of the multiple-user case, which is tackled in the following subsection.

14.1.2 Multiple users with multiple parallel links

While the single-user case provides interesting insights into how a user can divide its network flows over multiple links, the case of multiple users is more complex. The single-user case can, as seen in the previous subsection, be tackled using classical optimization techniques; however, the multiple-user case requires more advanced tools because of the conflicting interests of the users, i.e., the mutual delays incurred over the routes, which in turn impact the flow-control and routing decisions of the users.

In this regard, we denote $\mathcal{N} = \{1, ..., N\}$ as the set of N network users, $\lambda_{ij} \ge 0$ as the flow of player *i* over link *j*, and $\lambda_j = \sum_{i=1}^N \lambda_{ij}$ as the total flow on link *j*. We can define a non-cooperative continuous-kernel game with the players being the network users, i.e., the set \mathcal{N} . In this game, the strategy of each player *i* is to choose the transmission rates $\{\lambda_{ij}\}_{j\in\mathcal{N}}$ that maximize its utility function u_i , which, as in (14.2), is defined as the power

achieved by the user over the different links; i.e.,

$$u_{i}(\mathbf{\Lambda}) = \begin{cases} \left(\sum_{j \in \mathcal{M}} \lambda_{ij}\right)^{\beta+1} \\ \sum_{j \in \mathcal{M}} \frac{\lambda_{ij}}{c_{j} - \lambda_{j}}, & \sum_{j \in \mathcal{M}} \lambda_{ij} > 0, \\ 0, & \sum_{j \in \mathcal{M}} \lambda_{ij} = 0, \end{cases}$$
(14.11)

where $\mathbf{\Lambda} = {\lambda_{ij}}_{i \in \mathcal{N}, j \in \mathcal{M}}$ and β is assumed to be user-independent. Nonetheless, we see later that β can depend on N, so we denote it by $\beta(N)$. We show the dependence of the utility on N by denoting it u_i^N . Without loss of generality, we consider that the utility that each user $i \in \mathcal{N}$ aims to optimize is l_i^N , which is the logarithm of (14.11). In this game, we are not restricted to packet-level utilities, i.e., cases in which the users have a single connection. Instead, a user can be a service provider that generates a flow of traffic through its subscribers.

Naturally, given this formulated strategic game, the first key concept to investigate is the Nash equilibrium, as discussed in Chapter 3. In this game, the Nash equilibrium is an *N*-tuple $\{\lambda_{ii}^*\}_{i \in \mathcal{N}, j \in \mathcal{M}}$ that satisfies

$$I_{i}^{N}\left(\{\lambda_{ij}^{*}\}_{j\in\mathcal{M}},\{\lambda_{kj}^{*}\}_{k\in\mathcal{N}\setminus\{i\},j\in\mathcal{M}}\right)\geq I_{i}^{N}\left(\{\lambda_{ij}\}_{j\in\mathcal{M}},\{\lambda_{kj}^{*}\}_{k\in\mathcal{N}\setminus\{i\},j\in\mathcal{M}}\right),\qquad(14.12)$$

for all $\{\lambda_{ij}\}_{j \in \mathcal{M}}$ and for all $i \in \mathcal{N}$. Certainly these inequalities at the Nash equilibrium as well as the results presented hereinafter would also be satisfied for the actual utility u_i^N .

As discussed in Chapter 3, finding the Nash equilibria in a general case is quite complex, especially when the utilities are not concave as in the considered case. In order to provide a suitable analysis, we consider the case of a large number of users. In the context of combined flow control and routing, this consideration is meaningful given the large-scale nature of many communication networks such as the Internet. Moreover, the use of a large number of users for games involving routing is common, as discussed in the context of the Wardrop equilibrium in Chapter 3. For this purpose, we consider the concept of an *asymptotic Nash equilibrium*, defined as follows [28]:

DEFINITION 14.1 For the considered combined flow-control and routing game, assuming that N is arbitrarily large, let $\lambda_{ij}^*(N)$, $i \in \mathcal{N}$, $j \in \mathcal{M}$ be a set of flow rates for the users, dependent on N and defined for all positive integers N. These flow rates are said to constitute an asymptotic Nash equilibrium *if*, for all $i \in \mathcal{N}$,

$$\lim_{N \to \infty} I_i^N(\{\lambda_{ij}^*\}_{j \in \mathcal{M}}, \{\lambda_{kj}^*\}_{k \in \mathcal{N} \setminus \{i\}, j \in \mathcal{M}})$$
$$= \lim_{N \to \infty} \max_{\{\lambda_{ij}\}_{j \in \mathcal{M}}} I_i^N(\{\lambda_{ij}\}_{j \in \mathcal{M}}, \{\lambda_{kj}^*\}_{k \in \mathcal{N} \setminus \{i\}, j \in \mathcal{M}}).$$
(14.13)

In other words, the policies (flow rates) of a given player i at an asymptotic Nash equilibrium (i.e., when N is large) forbid any unilateral deviations at the asymptotic case

since they are equal to the policies that maximize I_i^N given that the policies of the other players are fixed. By specifying how close the two expressions in (14.13) are, we can define the following refinement on the defined equilibrium concept [28]:

DEFINITION 14.2 For the defined N-user non-cooperative game with an arbitrarily large number of users N, let $\lambda_{ij}^*(N)$, $i \in \mathcal{N}$, $j \in \mathcal{M}$ be a set of asymptotic equilibrium flow rates for the users. These flow rates are said to constitute an O(1/N) Nash equilibrium with exponent κ if there exists a non-positive scalar κ , independent of N, such that, for all $i \in \mathcal{N}$,

$$I_{i}^{N}(\{\lambda_{ij}^{*}\}_{j\in\mathcal{M}},\{\lambda_{kj}^{*}\}_{k\in\mathcal{N}\setminus\{i\},j\in\mathcal{M}})$$

$$=\max_{\{\lambda_{ij}\}_{j\in\mathcal{M}}}\left(I_{i}^{N}(\{\lambda_{ij}\}_{j\in\mathcal{M}},\{\lambda_{kj}^{*}\}_{k\in\mathcal{N}\setminus\{i\},j\in\mathcal{M}})\right)+\frac{\kappa}{N}+o(1/N).$$
(14.14)

In order to find asymptotic equilibrium solutions, assume that for each N the Nash equilibrium exists and is an inner solution, i.e., $\lambda_{ik}^*(N) \neq 0$, $\forall i \in \mathcal{N}$, $k \in \mathcal{M}$, and consider the first-order necessary conditions given by

$$\frac{\partial l_i^N}{\partial \lambda_{ik}} = 0, \; \forall i \in \mathcal{N}, \; k \in \mathcal{M}, \tag{14.15}$$

which yields

$$\frac{\beta(N)+1}{\sum_{k\in\mathcal{M}}\lambda_{ij}(N)} - \frac{c_k - \lambda_k(N) + \lambda_{ik}(N)}{(c_k - \lambda_k(N))^2 \sum_{j\in\mathcal{M}}\frac{\lambda_{ij}(N)}{c_j - \lambda_j(N)}} = 0,$$
(14.16)

where $\beta(N)$ is the parameter β redefined as a function of N, as follows:

$$\beta = \beta(N) = \frac{\alpha}{N}, \forall N, \qquad (14.17)$$

where α is a positive constant. This definition stems from the fact that, in the limit $N \to \infty$, it turns out that non-trivial solutions for (14.16) exist only if β is scaled appropriately as a function of N, such as in (14.17). This need for non-trivial solutions to (14.16) justifies the dependence of the utilities l_i^N on N, introduced earlier. Consider the single-link case (i.e., M = 1). For this case, one can solve (14.16) and find that there exists a unique and symmetric Nash equilibrium solution given by

$$\lambda^*(N) = \lambda_{i1}(N) = \frac{\beta(N)}{N\beta(N) + 1} c_1, \ \forall i \in \mathcal{N}.$$
(14.18)

From (14.18), we note that the corresponding total flow over the single link is $\lambda_1 = N\lambda^*(N)$, so to ensure a finite delay as $N \to \infty$ we must require that $\beta(N)$ be of the order of $\frac{1}{N}$.

Given this scaling of β , and in order to derive an asymptotic equilibrium solution that is also an O(1/N) Nash equilibrium for the studied joint flow-control and routing model, it is useful to derive the limiting value of the solution that satisfies the first-order

conditions in (14.16). Given the fact that, in the considered model, the users enter the game symmetrically, we can restrict our attention to symmetric solutions where

$$\lambda_{ij}(N) = \frac{\lambda_j(N)}{N}, \ \forall i \in \mathcal{N}, \text{ and each } j \in \mathcal{M}.$$
 (14.19)

Given this symmetric assumption, this exact derivation of this limiting value is found in [28] and is a result of algebraic manipulation of (14.16) using (14.19) as well as sequence limits and convergence. In essence, the limiting value as N goes to infinity turns out to be the solution of the quadratic equation

$$(\alpha+1)\bar{\lambda}^2 - \alpha\bar{c}\bar{\lambda} + M\bar{c^2} - \bar{c}^2 = 0, \qquad (14.20)$$

where $\bar{\lambda} = \sum_{j=1}^{M} \lambda_j$, $\bar{c} = \sum_{j=1}^{M} c_j$, $\bar{c^2} = \sum_{j=1}^{M} c_j^2$. The solution of (14.20) is given by

$$\bar{\lambda} = \frac{\alpha \pm \sqrt{(\alpha+2)^2 - 4(\alpha+1)M\nu}}{2(\alpha+1)}\bar{c},$$
(14.21)

where $\nu = \frac{c^2}{c^2}$. This result is valid only if at least one of the solutions in (14.21) satisfies the bounds

$$0 \le \bar{\lambda} \le \bar{c},\tag{14.22}$$

in which case the individual link flow λ_i becomes

$$\lambda_j = c_j - \frac{\bar{c} - \bar{\lambda}}{M}, \ j \in \mathcal{M},$$
(14.23)

given that

$$\lambda_i \ge 0, \ \forall j \in \mathcal{M}. \tag{14.24}$$

With these results, one can prove, as per [28], the following theorem regarding the asymptotic Nash equilibrium of the considered joint flow-control and routing problem:

THEOREM 14.4 Suppose that there exists $\{\lambda_j\}_{j=1}^M$ satisfying (14.21)–(14.24). Then

$$\lambda_{ij} = \frac{\lambda_j}{N}, \ i \in \mathcal{N}, \ j \in \mathcal{M}$$
(14.25)

constitute an O(1/N) Nash equilibrium with exponent

$$\kappa = \alpha \log \frac{2\bar{\lambda}}{\sqrt{\gamma \bar{Q}} + \alpha M \gamma} < 0, \qquad (14.26)$$

where

$$\bar{Q} \triangleq \frac{\bar{\lambda}^2 - M\bar{\lambda}^2}{\gamma} + \alpha^2 M^2 \gamma > 0, \qquad (14.27)$$

with $\bar{\lambda^2} = \sum_{j=1}^M \lambda_j^2$ and

468

$$\gamma \triangleq \frac{\bar{\lambda}^2}{\alpha \bar{\lambda}}.$$
 (14.28)

Proof The proof of this theorem is based on [28]. Fix the flows of all users except player *i*, as given by (14.21)–(14.24). Denote the flows of player *i*, which are arbitrary, by $\eta_{i,m}$, for $m \in \mathcal{M}$. Given this consideration, the situation faced by player *i* is similar to the single-user multiple-links case of Section 14.1.1, with the capacity of link $m \in \mathcal{M}$ as seen by player *i* being

$$c_m^i = c_m - \frac{N-1}{N}\lambda_m = \frac{1}{\alpha\bar{\lambda}}\bar{\lambda}^2 + \frac{1}{N}\lambda_m, \ m \in \mathcal{M}.$$
(14.29)

Since λ_m , $\bar{\lambda}^2$, and $\bar{\lambda}$ are independent of N, c_m^i constitutes a (1/N)-perturbation around a nominal constant capacity $\frac{\bar{\lambda}^2}{\alpha \lambda}$ that is also independent of the user. Thus, from the perspective of player *i*, the network has equal link capacities. By Theorem 14.2, there exists a sufficiently large N* such that, for all $N > N^*$, player *i*'s response to (14.21)-(14.24) is an inner solution. For each such N, the solution is also the unique stationary point of u_i^N with all other users' policies fixed as given. Thus, the unique solution is the stationary point of the function

$$v_i^N(\{\eta_{ij}\}_{j\in\mathcal{M}}) = \frac{\left(\sum_{m=1}^M \eta_{im}\right)^{1+\frac{N}{M}}}{\sum_{m=1}^M \frac{\eta_{im}}{c_m^i - \eta_{im}}},$$
(14.30)

where c_m^i is given by (14.29). From Theorem 14.1, it follows that the unique inner maximizing solution for v_i^N yields the value

$$(v_i^N)^* = \max_{\{\eta_{ij}\}_{j \in \mathcal{M}}} v_i^N = \frac{\left(\bar{c^i} - \mu \tilde{c^i}\right)^{1+\frac{\alpha}{N}}}{\tilde{c^i} - M\mu}\mu,$$
 (14.31)

where $\bar{c}^i = \sum_{m=1}^M c_m^i = M\gamma + \frac{\bar{\lambda}}{N}$ and $\tilde{c}^i = \sum_{m=1}^M \sqrt{c_m^i} = \sum_{m=1}^M \sqrt{\gamma + \frac{\lambda_m}{N}}$, and γ is given by (14.28). The value μ is given by

$$\mu = \frac{\left(1 + \frac{\alpha}{N}\right)\tilde{c}^{i} - \sqrt{Q}}{2M(1 + \frac{\alpha}{N})},$$
(14.32)

with

$$Q = \frac{\alpha^2}{N^2} \tilde{c^i}^2 + 4(1 + \frac{\alpha}{N}) (\tilde{c^i}^2 - M \bar{c^i}).$$
(14.33)

For computing $(v_i^N)^*$ for large N, we perform the expansion

$$\sqrt{\gamma + \frac{\lambda_k}{N}} = \sqrt{\gamma} \left(1 + \frac{\lambda_k}{2\gamma N} - \frac{1}{8} \left(\frac{\lambda_k}{\gamma N} \right)^2 + O\left(\frac{1}{N^3} \right) \right), \quad (14.34)$$

using the fact that, for any real positive number x,

$$\sqrt{1+x} = 1 + \frac{x}{2} - \frac{x^2}{8} + O(x^3).$$

Using the expansion (14.34) in $\bar{c^i}$ and $\tilde{c^i}$, we obtain

$$Q = \bar{Q} \cdot \frac{1}{N^2} + O\left(\frac{1}{N^3}\right),$$

where \bar{Q} is given by (14.27), which can be rewritten as

$$\bar{Q} = \frac{\bar{c}^2 - M\bar{c}^2 + \alpha^2(\bar{c} - \bar{\lambda})^2}{\bar{c} - \bar{\lambda}}M$$

We note that

$$\mu = \frac{\tilde{c}^{i}}{M} - \frac{1}{2MN} (\alpha \tilde{c}^{i} + \sqrt{\bar{Q}}) + O\left(\frac{1}{N}\right).$$

For the above expression to be valid, it is necessary that \overline{Q} be positive, which readily follows from the fact that, as in Theorem 14.2, Q is positive for all sufficiently large N. Now let us expand the numerator and denominator of (14.31):

$$\bar{c}^{i} - \mu \tilde{c}^{i} = \left(\frac{\gamma \alpha M}{2} + \frac{\sqrt{\gamma \bar{Q}}}{2}\right) \frac{1}{N} + O\left(\frac{1}{N}\right),$$
$$\tilde{c}^{i} - M\mu = \frac{M\alpha \sqrt{\gamma} + \sqrt{\bar{Q}}}{2N} + O\left(\frac{1}{N}\right).$$

Hence,

$$\lim_{N \to \infty} (v_i^N)^* = \lim_{N \to \infty} \left[\left(\sqrt{\bar{Q} + \alpha M \sqrt{\gamma}} \right)^{\frac{\alpha}{N}} \left(\frac{1}{2N} \right)^{\frac{\alpha}{N}} \cdot \gamma^{1 + \frac{\alpha}{2N}} \right]$$
$$= \gamma = \frac{\bar{c} - \bar{\lambda}}{M}.$$
(14.35)

Furthermore, with $\eta_{ij} = \lambda_{ij}$ substituted into v_i^N given by (14.30), we obtain

$$v_i^N(\{\frac{\lambda_j}{N}\}) = \gamma \left(\frac{\bar{\lambda}}{N}\right)^{\frac{\alpha}{N}} = \frac{\bar{c} - \bar{\lambda}}{M} \left(\frac{\bar{\lambda}}{N}\right)^{\frac{\alpha}{N}}.$$
 (14.36)

As N goes to infinity, (14.36) tends to $\frac{\bar{c}-\bar{\lambda}}{M}$, which is identical to (14.35). Thus, the corresponding flow rates of Theorem 14.4 constitute an asymptotic Nash equilibrium.

We need to further show that these flow rates are also in O(1/N) equilibrium, which can be computed using the already-computed expansion

$$\log v_i^N(\frac{\lambda_j}{N}) - \log (v_i^N)^* = \frac{\alpha}{N} \log \frac{2\bar{\lambda}}{\sqrt{\gamma \bar{q}} + \alpha M \gamma} + O\left(\frac{1}{N}\right), \qquad (14.37)$$

which is exact to the $\frac{1}{N}$ term. The exponent κ is easily computed from (14.37), and its negativity is easily shown.

The symmetric O(1/N) Nash quilibrium provided in Theorem 14.4, which exists when (14.21)–(14.24) admit a solution, requires that the flows on *all* M links be positive. However, there might exist a symmetric O(1/N) Nash equilibrium that uses only a subset of the links, i.e., $S_m = \{1, ..., m\}$ for some m with zero flows on the remaining links in $\mathcal{M} \setminus S_m$. The following theorem provides necessary and sufficient conditions to characterize when a set of positive flows on a subset of links can constitute a symmetric O(1/N) Nash equilibrium for the considered M-link game:

THEOREM 14.5 For some subset of links S_m , consider that a solution to (14.21)–(14.24) exists, with the corresponding total flows on link j denoted by $\lambda_j^{(m)}$. Then, for every $j \in \mathcal{M}$, $i \in \mathcal{N}$, the set of flows

$$\lambda_{ij}(N) = \begin{cases} \frac{\lambda_j^{(m)}}{N}, & j \le m, \\ 0, & j > m, \end{cases}$$
(14.38)

provides an O(1/N) Nash equilibrium if

$$c_{m+1} < c_1 - \lambda_1^{(m)}. \tag{14.39}$$

In this case, the exponent κ is given by (14.34) with M replaced by m. Conversely, if $c_{m+1} \ge c_1 - \lambda_1^{(m)}$, then the set of flows in (14.38) is not in O(1/N) Nash equilibrium, nor does it constitute an asymptotic Nash equilibrium.

The proof of this theorem can be found in [28].

14.1.3 Sample Nash equilibria

The two theorems previously discussed constitute a complete solution to the continuouskernel game formulation of the combined flow-control and routing problem in a large-scale communication network with multiple links. For instance, the two theorems provide conditions that can be tested to characterize the entire set of symmetric O(1/N) Nash equilibria, and the number of tests is equal to M, i.e., the number of links. Each test involves checking the existence of a solution to (14.21)-(14.24) for a general m and, if m < M, checking condition (14.39). When the conditions are not satisfied for all $m \in \mathcal{M}$, then the considered game has no symmetric O(1/N) Nash equilibrium. However, this does not rule out the existence of an asymmetric equilibrium which is an O(1/N) Nash equilibrium. Furthermore, the devised equilibrium concept is applicable even when the total throughput of the users is not fixed, and it is also characterized by the fact that it is possible for only a strict subset of the available network links to carry positive flow. We also highlight that the theorems show that the symmetric O(1/N)Nash equilibrium exhibits an equal-delay property such as in the Wardrop equilibrium, discussed in Chapter 3.

Moreover, in the previous subsection we showed that if, for a given non-cooperative game, computing the Nash equilibrium for the generic case is difficult, one can use some assumptions, such as a large number of users, to ease the mathematical treatment. For instance, as Internet-like communication networks are often characterized by a large number of nodes, assuming a large number of users proves to be both mathematically appealing and practically desirable. By assuming a large number of users, we had to extend the concept of the Nash equilibrium to the asymptotic case, using Definition 14.1. This new concept is along the same lines as the key idea of a Nash equilibrium, i.e., no unilateral deviations, but applied at the limit of the utility. We further refined the concept using Definition 14.2 to show how close the utility at the equilibrium is to its maximum when N is large. Such refinements of the Nash equilibrium are also useful in other applications and with non-asymptotic analysis; we refer the reader to [58] for further information. In brief, when dealing with non-cooperative games in which no immediate characterization for the Nash equilibrium exists, in the generic case, one can use some practically motivated assumptions on the number of players, the strategies, and even the utilities in order to characterize equilibrium concepts.

We highlight the results of the previous subsection in two numerical examples taken from [28].

Example 14.1 Consider a network of ten links with the capacities of the mth link being 100 - 10(m - 1) with $\alpha = 0.9$. There is no solution to (14.21)–(14.24) if we consider the subsets S_m for $m \ge 7$. For m = 6, we obtain a feasible solution through the positive square root in (14.21), with the corresponding value being $\overline{\lambda}^+ = 182.95$. Thus, we can find the flows corresponding to $\overline{\lambda}^+$ as follows:

 $\lambda_1 = 55.49, \ \lambda_2 = 45.49, \ \lambda_3 = 35.49, \ \lambda_4 = 25.49, \ \lambda_5 = 15.49, \ \lambda_6 = 5.49.$ (14.40)

Note that the negative square root, $\overline{\lambda}^- = 30.2$, leads to a negative value for λ_6 and hence is not a feasible solution. Using Theorem 14.5, we can easily verify that we have a symmetric O(1/N) Nash equilibrium and that the exponent is $\kappa = -0.0875$. For m < 6, although (14.21)–(14.24) admit a solution, the condition of Theorem 14.5 is not verified, so, no symmetric asymptotic Nash equilibria exist in which the users transmit positive flows on only the first five (or fewer) links.

Example 14.2 Suppose we have a network with ten links with

$$c_1 = 100 \text{ and } c_m = 500, m = 2, 3, \dots, 10,$$

with $\alpha = 0.9$. For this case, Theorem 14.4 directly applies, as it turns out that all ten links can be used with both roots of (14.21), yielding feasible solutions. Thus, this example admits two symmetric O(1/N) Nash equilibria. For the positive root $\bar{\lambda}^+ = 201.9$, the equilibrium flow rates are

$$\lambda_1 = 65.19, \ \lambda_m = 15.19, \ m = 2, 3, \dots, 10,$$

and the one corresponding to the negative square root $\bar{\lambda}^- = 58.7$ is

$$\lambda_1 = 50.87, \ \lambda_m = 0.87, \ m = 2, 3, \dots, 10.$$

This example highlights the delay-equalizing property of the considered equilibria for both cases. The exponents for the two equilibria are $\kappa^+ = -0.1472$ and $\kappa^- = -0.777$.

We can further explore the possibility of other Nash equilibria with fewer links having positive flows. We compute $\bar{\lambda}$ for all m < 10. For the subsets S_m with m = 6, 7, 8, 9, both the roots provide $\{\lambda_i\}$ that satisfy (14.21)–(14.24), but these solutions do not satisfy the condition in Theorem 14.5, so they are not O(1/N) Nash equilibria. For m = 2, 3, 4, 5, there exists no solution for (14.21)–(14.24). For m = 1 (i.e., S_1), the negative square root of (14.21) yields a solution, given by $\lambda_1 = 47.37$. Since $c_2 = 50 < 52.63 = c_1 - \lambda_1^{(1)}$, the condition of Theorem 14.5 is satisfied, so , there is an O(1/N) Nash equilibrium in which all users use only one link, i.e., link 1. It must be noted that the total flow for this case is 47.37, which is less than those under the two other equilibria, which are 201.9 and 58.7, respectively.

The last example shows that some O(1/N) Nash equilibria can use *all* available links or *only one* link for all users.

The analysis of this section has focused on symmetric Nash equilibria when the network's users optimize their utilities, in terms of power, when the number of users is large. One can further study the possibility of asymmetric equilibria, the characterization of equilibria for a finite number of users, the formation of a network topology (i.e., the interconnection among the links), and many other extensions.

14.2 Congestion control in networks with a single service provider

In many communication networks, users such as routers or computing devices need to control flow of traffic based on their quality-of-service requirements, the perceived congestion, and the availability and cost of bandwidth. Given these incentives, there has been much research devoted to maintaining small queue sizes at the routers of such communication networks (e.g., see [265, 243, 60] and references therein). The motivation for such research stems from the well-known result that, in a network with a large capacity and a large number of users, the probability that the arrival rate will exceed the available capacity is small, i.e., the probability of queue build-up is small [78].

While much work in this area focuses on flow control from the perspective of the users and on the aggregate efficiency of the network as noted in Section 14.1, it is also of interest to investigate the problem from the perspective of the service provider. In other words, it is of interest to analyze how the service provider can price its bandwidth while optimizing its profit and given the incentives of the users, which control their flow in such a way as to avoid congestion and reduce their costs. In [59], this problem was investigated in a single-link network, focusing on the economics of providing a large capacity to reduce the queue build-up at the routers, from the perspective of the

service provider. However, beyond this it is important to study the pricing scheme of the service provider, given the flow-control actions of the users as well as the presence of a multiple-link network.

In this section, based on the work in [60], we first present the network model of interest and its underlying assumptions. Then, we formulate the problem using a Stackelberg game in which the leader is the service provider and the followers are the users. We investigate the users' flow-control problem, and for devise the optimal policy for the service provider. Finally, we discuss the results and their implications for pricing and flow control in communication networks.

14.2.1 Pricing and congestion control

Suppose that we have a communication network with a single service provider and M tandem links accessed by N users belonging to M + 1 different classes. The users belonging to class 0 are able to use *all* M links in the network, while users belonging to class 1 can only use a single link l with l = 1, ..., M. We let n_k denote the number of users belonging to class k = 0, ..., M. Here, c_l denotes the capacity (bandwidth) of a link l, l = 1, ..., M and p denotes the price per unit bandwidth charged by the service provider. The transmission rate of the *j*th user in class k is denoted by x_{kj} . Thus, the total flow \bar{x}_k of users of class k is given by

$$\bar{x}_k = \sum_{j=1}^{n_k} x_{kj}, \ k = 0, \dots, M.$$
 (14.41)

Hereinafter, we use the term "user kj" to refer to the *j*th user in class *k*. Furthermore, we let x_{-k} denote the collection of flow rates of all users except those in class *k* and x_{-jk} the collection of flow rates of all users in class *k*, except that of the *j*th user in class *k*.

The objective of any user *i* in class 0, i.e., a user that utilizes all *M* links, is to maximize a function that reflects the achieved benefit, i.e., utility, from the flow x_{0i} , the price paid for the used bandwidth, and the cost due to congestion on a particular link *l*. Thus, user 0*i* aims to maximize the following objective function:

$$u_{0i}(x_{0i}, x_{-0i}, x_{-0}, p) = w_0 \log(1 + x_{0i}) - Mp x_{0i} - \sum_{l=1}^M \frac{1}{c_l - \bar{x}_0 - \bar{x}_l}, \quad (14.42)$$

where $w_0 \log(1 + x_{0i})$ is the benefit or utility of flow x_{0i} to user 0i, with $w_0 > 0$ a preference parameter, and the term $\frac{1}{c_l - \bar{x}_0 - \bar{x}_l}$ is the congestion cost on a link *l*. The maximization in (14.42) is with respect to x_{0i} over the range $[0, \min_l \{c_l - \bar{x}_0 + x_{0i} - \bar{x}_l\})$. For the users in other classes, which use only a single link, we can define a similar objective function for user kj (with $k \ge 1$):

$$u_{kj}(x_{kj}, x_{-kj}, x_{-k}, p) = w_k \log(1 + x_{kj}) - p x_{kj} - \frac{1}{c_k - \bar{x}_0 - \bar{x}_k}, \quad (14.43)$$

with the optimization being for x_{kj} over $[0, c_k - \bar{x}_0 + x_{kj} - \bar{x}_k)$. The motivation for the congestion used cost in (14.42) and (14.43) is that, assuming the queueing at the *k*th link is an M/M/1 process, this cost is simply the delay on the link.

From the perspective of the service provider, the main objective is to maximize its profits received from the bandwidth used by the users of all classes. With this objective, the provider would aim to set its price p in a way that maximizes the following function:

$$l(p, \bar{\mathbf{x}}(p)) = Mp\bar{x}_0 + p\sum_{k=1}^{M} \bar{x}_k, \qquad (14.44)$$

where $\bar{\mathbf{x}}(p) = [\bar{x}_0(p), \dots, \bar{x}_M(p)]$ is the vector of the total flows of all M + 1 user classes. The dependence of this vector on p will be clarified shortly.

By closely inspecting (14.42), (14.43), and (14.44), we can see that these objectives are interdependent and manifest a clear conflict of interest between the provider and the users. The network provider needs to set its price so as to ensure that the tradeoff between price and bandwidth utilization is optimized. If the provider sets its price too high, few users would use the bandwidth, and the profits would go down. If the provider sets its price too low, although many users are encouraged to use the bandwidth, the overall profit might remain low. Furthermore, given a set price as in (14.42) and (14.43), the users themselves would engage in a non-cooperative situation, since the objective function of a user depends on the amount of bandwidth used by the others over the desired link, i.e., the congestion cost. Clearly, we have a two-level competitive situation: the users attempt to utilize as much bandwidth as possible while maintaining reasonable congestion and bandwidth costs, while the provider attempts to set its price so as to maximize its profit.

This competition invites the use of non-cooperative game theory. The presence of a hierarchy, between the provider that controls the prices and the users, implies that a Stackelberg framework, as discussed in Chapter 3, is quite suitable. In this context, we can formulate a Stackelberg game having the following components:

- A single leader, which is the network provider aiming to set its price so as to maximize its profit, as in (14.44)
- A total of $N = \sum_{k=0}^{M} n_k$ non-cooperative followers (i.e., network users), with each follower aiming to maximize its objective function in (14.42) for class 0 and (14.43) for the other classes.

In this context, the Stackelberg game can be split into two goals. First, given a set price p, we can see the followers as a single group. For this group, the reaction set $\mathcal{R}^{F}(p)$ is a function of the price set by the leader, and can be defined as follows:

$$\mathcal{R}^{\mathrm{F}}(p) = \{ \{ \{x_{kj} \ge 0\}_{j=1}^{n_{k}} \}_{k=0}^{M} : u_{kj}(x_{kj}, x_{-kj}, x_{-k}, p) \ge u_{kj}(x_{kj}', x_{-kj}, x_{-k}, p), \\ \forall x_{kj}' \; \forall j, 1 \le j \le n_{k} \text{ and } k, 0 \le k \le M \},$$
(14.45)

where the interval for x'_{kj} is $[0, \min_{l} \{c_l - \bar{x}_0 + x_{0i} - \bar{x}_l\})$ for k = 0 and $[0, c_k - \bar{x}_0 + x_{kj} - \bar{x}_k)$ for $k \ge 1$.

By closely inspecting the reaction set of the followers group, we can see that, for a fixed p > 0, there exists a non-cooperative game among the followers whereby each follower, depending on its class, attempts to find the transmission rate that maximizes (14.42) or (14.43). By closely looking at (14.42) and (14.43), we can see that these functions depend not only on a follower's own transmission rate but also on the transmission rate of its competitors (and the price). Thus, the reaction set of the followers group can be investigated using the set of strategies that constitute a *Nash equilibrium* for the non-cooperative continuous-kernel game defined with the *N* users being the players, the transmission rates being the strategies, and the objective functions being given by (14.42) and (14.43), depending on the followers' class. In other words, the reaction strategies of the followers to a fixed price p > 0 of the leader is the tuple $\{\{x_{kj}^*(p) \ge 0\}_{j=1}^{n_k}\}_{k=0}^M$ satisfying, for all j, $1 \le j \le n_k$ and k, $0 \le k \le M$,

$$\max_{x_{ki}} u_{kj}(x_{kj}, x_{-kj}^*, x_{-k}^*, p) = u_{kj}(x_{kj}^*, x_{-kj}^*, x_{-k}^*, p),$$
(14.46)

where the constraint interval is $[0, \min_{l} \{c_{l} - \bar{x}_{0}^{*} + x_{0i}^{*} - \bar{x}_{l}^{*}\})$ for k = 0 and $[0, c_{k} - \bar{x}_{0}^{*} + x_{kj}^{*} - \bar{x}_{k}^{*}]$ for $k \ge 1$. (14.46) clearly shows the dependence of any strategy, i.e., transmission rate, for the followers on the price p, which was shown earlier.

From the leader's perspective, assuming that the *N*-player followers' game admits a unique Nash equilibrium (which will be shown in the next subsection), then the Stackelberg solution of the game would be to determine the optimum price that can maximize the revenue given the followers' Nash strategy, i.e.,

$$\max_{p \ge 0} I(p, \bar{\mathbf{x}}^*(p)).$$
(14.47)

Note that the pricing function considered in (14.44) assumes that a user is charged a price in proportion to the product of its bandwidth usage and the number of hops/links on its route. This is because, if a user utilizes r links on its route while transmitting with a rate x, then the total consumption is rx units of network resources.

14.2.2 Non-cooperative Nash game between followers

For the followers' non-cooperative game, it can be shown that a unique Nash equilibrium exists, as shown by the following lemma:

LEMMA 14.1 For each fixed price p > 0, the N-player non-cooperative game among the followers with objective functions given in (14.42) and (14.43) admits a unique Nash equilibrium in the strategies $\{x_{kj}^*(p) \ge 0; 1 \le j \le n_k, 0 \le k \le M\}$, with $\bar{x}_0^* + \bar{x}_l^* < c_l$, $1 \le l \le M$.

Proof *The proof is from [60]. We note that by adding the quantity*

$$w_0 \sum_{j \neq i} \log (1 + x_{0j}) + \sum_{k=1}^{M} w_k \sum_{j=1}^{n_k} \log (1 + x_{kj}) - Mp \sum_{j=1, \neq i}^{n_0} x_{0j} - p \sum_{k=1}^{M} \bar{x}_k$$

to u_{0i} , the resulting function can be used as a new objective function for user 0*i* without affecting the Nash equilibrium. Similarly, adding

$$w_{m} \sum_{j \neq i} \log (1 + x_{m,j}) + \sum_{k=0, k \neq m}^{M} w_{k} \sum_{j=1}^{n_{k}} \log (1 + x_{kj})$$
$$- \sum_{l=1, \neq m}^{M} \frac{1}{c_{l} - \bar{x}_{0} - \bar{x}_{l}} - Mp\bar{x}_{0} - p \sum_{l=1}^{M} \bar{x}_{l} + px_{mi}$$

to u_{mi} for each $i, m, 1 \le i \le n_m, 1 \le m \le M$, will not change the Nash equilibrium because the quantity added does not depend on the decision variable of user mi. One can easily show that all the modified objective functions become identical and are given by

$$u(x_0, \dots, x_M, p) = \sum_{k=0}^{M} w_k \sum_{j=1}^{n_k} \log(1 + x_{kj}) - \sum_{l=1}^{M} \frac{1}{c_l - \bar{x}_0 - \bar{x}_l} - Mp\bar{x}_0 - p \sum_{l=1}^{M} \bar{x}_l.$$
(14.48)

In consequence, the Nash equilibrium of the original game is also a Nash equilibrium of the game with the common objective function (14.48). The function $u(\cdot)$ of (14.48), is strictly concave in $(x_{01}, ..., x_{Mn_M})$, which is restricted to the non-negative orthant bounded by the hyperplanes $\bar{x}_0 + \bar{x}_l = c_l$ on which u is unbounded from below. From standard results in finite-dimensional optimization (e.g., see [81]), we have that u has a unique maximum in this bounded region and every person-by-person optimal solution is also globally optimal. Thus, the Nash equilibrium of the original game exists and is unique. The maximizing solution cannot lie on the hyperplane, which leads to the strict inequality on $\bar{x}_0^* + \bar{x}_l^*$ for all $l, 0 \le l \le M$.

The unique Nash equilibrium can, depending on the value of the price p, yield some transmission rates (i.e., x_{kj}) that are zero. Whenever this does not occur, we refer to the Nash equilibrium as *inner* or *positive*. By setting the partial derivatives of u with respect to x_{kj} to zero for all admissible j and k, we obtain necessary and sufficient conditions for the Nash equilibrium to be positive. To find a tractable condition for this positive Nash equilibrium to exist, we make the following assumptions:

- All users that use single links have the same preference parameter w_k, so w_k is independent of k for k ≥ 1.
- The number of users in each class that uses a single link is also independent of k, i.e., n_k is independent of k for all $k \ge 1$.
- The capacity of each link is proportional to the total number of users using that link.

Given the above assumptions, the capacity of a link *l* can be written as $c_l = (n_0 + n_1)c = nc$. Furthermore, we let

$$\bar{y}_0 \triangleq n_0 + \bar{x}_0, \ \bar{y}_1 \triangleq n_1 + \bar{x}_1, \ \bar{y} \triangleq = \bar{y}_0 + \bar{y}_1,$$

and

$$\bar{w} \triangleq n_0 w_0 + M n_1 w_1, \ w_{av} = \frac{\bar{w}}{n}.$$

Given these quantities, it is shown [60] through algebraic manipulation of the first-order conditions that a positive Nash equilibrium exists if and only if there exists a $\bar{y}(p)$ solving

$$g(\bar{y}) \triangleq \frac{\bar{w}}{\bar{y}} - \frac{M}{(nc+n-\bar{y})^2} - Mp = 0$$
(14.49)

and satisfying the positivity constraint

$$\min(w_0, Mw_1)\frac{\bar{y}(p)}{\bar{w}} > 1.$$
(14.50)

As demonstrated in [60], it turns out that there exists a unique solution to (14.49) in the open interval (n, (c+1)n) if and only if g(n) > 0; that is,

$$p < \hat{p} \triangleq \frac{w_{\mathrm{av}}}{M} - \frac{1}{(nc)^2}.$$
(14.51)

Thus, a range of values for the price p exists for which the Nash equilibrium of the followers is positive, which, as will be seen in the next subsection, implies that the leader's problem is feasible.

14.2.3 Optimal pricing policy for the service provider

In order to inspect the optimal pricing policy for the service provider, i.e., the leader, we maintain the assumptions made in the previous subsection regarding having, for each class of users using a single link, a number of users and preference parameters independent of the class, while having a capacity proportional to the number of users using a link.

In (14.49), we observe that there exists a one-to-one correspondence between \bar{y} and the price *p*. As a result, the leader's problem can be turned into a maximization, with respect to $\bar{y} > n$, of the following objective function:

$$\tilde{l}(\bar{y}) = \bar{w}(1 - \frac{n}{\bar{y}}) - \frac{M(\bar{y} - n)}{(n(c+1) - \bar{y})^2},$$
(14.52)

which is obtained by substituting p from (14.49) as a function of \bar{y} . For notational convenience, we let $\tilde{l}_{\bar{y}} = \frac{\partial \tilde{i}}{\partial \bar{y}}$ and $\tilde{l}_{\bar{y}\bar{y}} = \frac{\partial^2 \tilde{i}}{\partial \bar{y}\partial \bar{y}}$. Over the desired interval (n, (c+1)n), \tilde{i} is an analytic function (i.e., a function that is locally given by a convergent power series), so

$$\tilde{l}_{\bar{y}} = \frac{n\bar{w}}{\bar{y}^2} - \frac{M[n(c-1)+\bar{y}]}{(n(c+1)-\bar{y})^3}, \ \tilde{l}_{\bar{y}\bar{y}} < 0.$$
(14.53)

Therefore, \tilde{l} is strictly concave, and since it becomes unbounded negative at the upper end of the interval, it follows that it has a unique maximum in the interval [n, (c+1)n). To avoid the situation at $\bar{y} = n$, we require that $\tilde{l}_{\bar{y}} > 0$, which translates into

$$n^2 c^2 w_{\rm av} > M.$$
 (14.54)

Under this condition, there exists a unique solution to $\tilde{I}_{\bar{y}} = 0$, which is $\bar{y}^* \in (n, (c+1)n)$. For this solution, the corresponding price value which maximizes $I(p, \bar{x}^*(p))$ is obtained directly from (14.49):

$$p^* = \frac{\bar{w}}{M\bar{y}^*} - \frac{1}{(nc+n-\bar{y}^*)^2},$$
(14.55)

which satisfies (14.51). Furthermore, we need the price p^* to satisfy (14.50), i.e.,

$$\min(w_0, Mw_1)\bar{y}^* > \bar{w}.$$
(14.56)

However, we note that \bar{y}^* is a solution to $\tilde{l}_{\bar{y}} = 0$, which is a third-order polynomial. Thus, as discussed in [60], a closed-form expression for \bar{y}^* can be obtained only for some special cases, such as when c = 1. In this case, we have

$$\bar{y}^* = \frac{2n(n\bar{w})^{\frac{1}{3}}}{M^{\frac{1}{3}} + (n\bar{w})^{\frac{1}{3}}},$$
(14.57)

provided that $n\bar{w} > M$, which ensures that the total throughput is positive, i.e., $\bar{y}^* > n$. Furthermore, satisfying (14.56) in this case maps to

$$\left(\frac{2\min(w_0, Mw_1)}{w_{av}} - 1\right) \left(n^2 w_{av}\right)^{\frac{1}{3}} > 1,$$
(14.58)

which is a condition that is more restrictive than $n^2 w_{av} > M$, which can now be dropped.

The Stackelberg solution, i.e., the revenue-maximizing price for the service provider, can be obtained for the case c = 1 using (14.55) and (14.57):

$$p^* = \frac{w_{\rm av}}{2M} \left(1 + M^{\frac{1}{3}} (n^2 w_{\rm av})^{-\frac{1}{3}}\right) - \frac{1}{4n^2} \left(1 + M^{\frac{1}{3}} (n^2 w_{\rm av})^{\frac{1}{3}}\right)^2, \tag{14.59}$$

which is seen to be positive whenever $n^2 w_{av} > M$, which already holds. Furthermore, one can clearly see that $p^* < \hat{p}$.

14.2.4 Network with a large number of followers

In the previous subsection, we saw that finding the Stackelberg solution for the considered communication network game is difficult unless a simplifying assumption such as c = 1 is used. The difficulty stems mainly from the fact that a closed-form expression for \bar{y}^* is difficult to find.

For more insight into the problem, we will consider the case with a large number of followers. In particular, we study the behavior of the system when *n* is large. When dealing with large *n*, we will use the convention that as $n \to \infty$ the sequence $\{w_{av}\}$ has a well-defined limit, $w_{av} > 0$. For example, this case is applicable when there exists $\alpha_0 \in (0, 1)$ such that $n_0 = \alpha_0 n$, which means that there will be infinitely many users of both classes as $n \to \infty$. In this case, we would have $w_{av} = \alpha_0 w_0 + (1 - \alpha_0) M w_1$. This convention immediately implies that (14.54) is satisfied.

To study the case of a large number of followers, it is convenient to work with the arithmetic mean of the x_i (or the y_i) rather than their sum. We denote this arithmetic mean, for the x_i by

$$x_{\mathrm{av}}(n) = \frac{1}{n}(\bar{x}_0 + \bar{x}_1).$$

In this case, $\tilde{l}_{\bar{y}} = 0$ translates into

$$\frac{w_{\rm av}(n)}{M(x_{\rm av}(n)+1)^2} = \frac{c+x_{\rm av}(n)}{n^2(c-x_{\rm av}(n))^3}$$

Given that $w_{av} \rightarrow w_{av}$ as $n \rightarrow \infty$, a positive solution to x_{av} exists for large *n* if and only if

$$\lim_{n\to\infty}n^2(c-x_{\rm av}(n))^3=\alpha,$$

for some $\alpha > 0$. By substituting into (14.49), we obtain

$$p \sim \frac{w_{\rm av}}{M(c+1)} + \frac{2c-1}{\alpha^{\frac{2}{3}}n^{\frac{2}{3}}},$$
 (14.60)

where $f(n) \sim h(n)$ if $\lim_{n \to \infty} \frac{f(n)}{h(n)} = 1$. By using (14.60) in $\tilde{l}_{\bar{y}} = 0$, and letting $n \to \infty$, we obtain

$$\alpha = \frac{2c(c+1)^2M}{w_{\rm av}},$$

so

$$x_{\rm av}(n) \sim c - n^{-\frac{2}{3}} \alpha^{\frac{1}{3}}$$
,

with the positivity condition

$$x_{av}(n) > \max(\frac{w_{av}}{w_0} - 1, \frac{w_{av}}{Mw_1} - 1).$$
 (14.61)

By letting $\alpha_0 = \frac{n_0}{n}$ and assuming that $\alpha_0 < 1$ for all *n*, as $n \to \infty$, we have that (14.61) is equivalent to

$$\frac{\alpha_0}{c + \alpha_0} < \frac{Mw_1}{w_0} < \frac{c}{1 - \alpha_0} + 1, \tag{14.62}$$

which is a necessary and sufficient condition for the existence of an inner solution. This also places an upper bound on the number of links M:

$$M < \frac{w_0}{w_1} \left(\frac{c}{1-\alpha_0} + 1\right),\tag{14.63}$$

considering that w_0 is not a function of M. However, if we pick $w_0 = \tilde{w}_0 M$ for some constant $\tilde{w}_0 > 0$, then no upper bound on the number of links M would exist.

We remark that the optimal price, i.e., the Stackelberg solution, for large *n* is positive; however, depending on the capacity *c*, it can be an increasing or a decreasing function of *n*. For instance, when $c > \frac{1}{2}$ the price decreases with *n*, whereas when $c < \frac{1}{2}$ it increases

with *n*. Nonetheless, when *n* is sufficiently large, the revenue *B* per unit bandwidth per link increases with *n* independently from c:

$$B = \frac{px_{\rm av}}{c} \sim \frac{w_{\rm av}}{M(c+1)} - 3\alpha^{-\frac{2}{3}}n^{-\frac{2}{3}}.$$
 (14.64)

Whenever $\frac{w_{av}}{(c+1)}$ is larger than the cost of adding one unit of bandwidth, we can show that the service provider's profit increases with the number of users. Hence, in this case, the provider has an incentive to increase the link capacity as this increase would drive the congestion cost to zero. For instance, the congestion cost decreases with *n* when *n* is large:

$$\frac{1}{n(c - x_{\rm av}(n))} \sim \alpha^{-\frac{1}{3}} n^{-\frac{1}{3}}.$$
(14.65)

The utilities of the users would decrease, as can be easily seen from (14.42) and (14.43).

In brief, under the studied model, for the asymptotic case the service provider has an incentive to increase the link capacity, and this increase is somewhat detrimental to the users as it decreases their net utilities, as in (14.42), and (14.43), as the system becomes more crowded.

In conclusion, using the Stackelberg framework, we have studied the problem of bandwidth pricing and flow control in communication networks. With this formulation, we have seen that the network provider, i.e., the leader in the Stackelberg game, has an incentive to increase the available capacity in proportion to the number of users in the network. It turns out that increasing this capacity decreases the congestion delay seen by the users (although it also decreases their net utility), so it is beneficial for the network provider to give better quality of service, in terms of delay, to the users.

While this work has considered two specific classes of users, it can be extended to multiple classes, as well as to the case in which multiple providers exist. In such a case, the use of a multi-leader multi-follower Stackelberg solution can constitute a good model. Moreover, in the multiple-provider case one can also study, using cooperative games, the possibility of cooperation among the providers in setting a joint pricing strategy and among the users to obtain better transmission rates. While in this formulation we have assumed that we have a game with complete information, the case of incomplete information is also of interest, and the interested reader is referred to [451] for further information. The reader is also referred to [451] for a discussion of improvement using non-linear pricing policies.

14.3 Pricing and revenue sharing for Internet service providers

The pricing of Internet services is a complex issue because of the interdependency among the various service providers and network operators. In this context, traffic circulating over the Internet needs to pass through many networks, owned by different providers, which requires that these providers work together in order to deliver an end-to-end service coverage. This gives an incentive for providers to collaborate in setting the price so as to maintain a respectable profit and demand. However, in the collaborative case, devising a scheme for sharing revenues among the providers is a challenging task. The issue of Internet service provider pricing has been studied in [244, 266, 303, 452] and references therein. However, these studies mainly focus on the flow-control and queueing aspects of the problem, with price only being an input. Moreover, most of this work assumes a single service provider that is aiming to maximize the social welfare. However, in this section, we are interested in studying the interactions of a number of Internet service providers that need to strategically set their prices so as to maximize their own profits, while having an efficient revenue-sharing mechanism. In such a design, two key issues arise: (i) modeling and analyzing the competitive pricing strategies among the providers, and (ii) devising a revenue-sharing scheme that can ensure better performance for the providers. Clearly, both objectives highlight the need for game theory. While (i) can easily be tackled with a non-cooperative game, objective (ii) highlights the idea of equilibrium selection and cooperation enforcement through some incentive, i.e., a revenue-sharing scheme. To tackle these objectives, we adopt the analysis carried out in [203].

First, we present a model for Internet pricing and formulate a non-cooperative game. Then we investigate non-cooperative strategies and the efficiency of the resulting equilibria. To improve the equilibrium efficiency, we study a fair revenue-sharing scheme and its implications for the non-cooperative game. Finally, we discuss a possible distributed algorithm that the providers can use to reach an equilibrium point.

14.3.1 Pricing game among Internet service providers

Consider a number of interconnected service providers that deliver an Internet service over a set \mathcal{R} of routes, i.e., end-to-end paths that traverse a sequence of service providers. The price charged for the service is the sum of the prices charged by all providers on a route. For every route $r \in \mathcal{R}$, we define a price p_r that controls the number of users on that route. This assumption stems from the fact that, given a certain price, only a certain number of users are willing to pay this price for the service. This relationship between the number of users and the price on a route is captured by a demand function $d(p_r)$. Note that, in this section, we do not deal with the rates or routing decisions of the users, as these aspects were analyzed in Section 14.1 and Section 14.2. Furthermore, we consider that the provided service has certain quality-of-service requirements in terms of delay, rate, or other factors, and that these requirements map into a local capacity constraint.

In such a model, the objective of every provider is to maximize its profit, i.e., the difference between the revenues and the costs, while maintaining its quality-of-service commitment to the served users. Before mathematically formulating the model and objectives, we make some assumptions:

• The demand function is assumed to be relatively inelastic, i.e., it does not change much when the price is low, with its elasticity increasing as the price becomes higher. This assumption reflects the nature of Internet services: at low prices, demand is dominated by the users' need to communicate, and so is close to being saturated; however, once the price goes beyond a certain threshold, it becomes a major deciding factor in whether to use the service. A mathematical motivation for this assumption is also found in [203].

- We assume that the links between the providers constitute the capacity bottlenecks. Thus, we can see the providers as network nodes connected by capacity-constrained links, and each route in \mathcal{R} is a sequence of inter-provider links.
- We assume that a fluctuation or change in prices does not yield a change of route between a source-destination pair.

With these assumptions, we formulate a strategic non-cooperative continuous-kernel game among the providers. In this game, the strategy of each provider *i* is the vector \mathbf{p}_i that maximizes its profit function. Each element p_{lr} of the vector \mathbf{p}_i represents the price charged by provider *i* for its service over link $l \in \mathcal{E}_i$ going through route $r \in \mathcal{R}_l$, where \mathcal{E}_i is the set of egress links owned by provider *i* and \mathcal{R}_l is the set of routes going through a link *l*. The profit function u_i of any provider *i* can be written as

$$u_i(\mathbf{p}_i, \mathbf{p}_{-i}) = \sum_{l \in \mathcal{E}_i} \sum_{r \in \mathcal{R}_l} (p_{lr} - c_l) d_r(p_r), \qquad (14.66)$$

where c_l is a cost per unit bandwidth and $p_r = \sum_{j \in \mathcal{L}_r} p_{jr}$ is the end-to-end price for route r, with \mathcal{L}_r being the set of all links that route r traverses. The non-cooperative nature of this game is indicated by the dependence of (14.66) not only on the own price of the provider but also on the prices set by the other providers over common routes, as seen in the dependence of the demand function on the term p_r . For instance, each element of the vector of strategies of the opponents of i (i.e., \mathbf{p}_{-i}) is defined for every link $l \in \mathcal{E}_i$ and route $r \in \mathcal{R}_l$ as

$$p_{-lr} \triangleq \sum_{k \in \mathcal{L}_r \setminus l} p_{kr} = p_r - p_{lr}.$$
(14.67)

Nonetheless, when finding its optimal strategy, each provider i attempts to maximize its profit in (14.66) given a capacity constraint. In the formulated strategic game, each provider i attempts to solve the following optimization problem:

$$\max_{p_{lr} \ge 0} \sum_{l \in \mathcal{E}_i} \sum_{r \in \mathcal{R}_l} (p_{lr} - c_l) d_r(p_r),$$

s.t.
$$\sum_{r \in \mathcal{R}_l} d_r(p_r) \le C_l, \forall l \in \mathcal{E}_i,$$
 (14.68)

where C_l is the capacity constraint on link l. When the providers, each acting *independently*, simultaneously solve their corresponding optimization problems in (14.68), they would be able to find their Nash equilibrium pricing strategies for every service provided on every route. Equivalently, denoting by \mathcal{L} the set of all links in the network, the Nash equilibrium is the set of prices $\{p_{lr}, l \in \mathcal{L}, r \in \mathcal{R}\}$ that solve the following system of equations:

$$\begin{cases} p_{lr} = c_l + \mu_l + g_r(p_{-lr} + p_{lr}), \ \forall l \in \mathcal{L}, \ r \in \mathcal{R}, \\ \mu_l(\sum_{r \in \mathcal{R}_l} d_r(p_r) - C_l) = 0, \ \forall l \in \mathcal{L}, \end{cases}$$
(14.69)

where $\{\mu_l, l \in \mathcal{L}\}$ are Lagrangian multipliers and $g_r(p) \triangleq -\frac{d_r(p)}{d'_r(p)} (d'_r(p))$ is the first-order derivative).

As discussed in Chapter 3, the Nash equilibrium is often inefficient or undesirable. In this Internet service providers' game, it is shown in [203], using an example with two providers, that a Nash equilibrium can lead to an unfair distribution of revenues among service providers. For instance, for the considered two-provider example in [203], the provider with smaller capacity can obtain a larger profit. In reality, a provider that has a larger capacity should, in principle, be able to obtain a larger profit, which highlights the unfairness of the Nash equilibrium in this game. Furthermore, at the Nash equilibrium, it can turn out that a provider having a bottleneck link has no incentive to upgrade its capacity since, by not doing so, it can maintain an advantage in obtaining more profits than the other providers.

14.3.2 Revenue-sharing strategies

In order to overcome the undesirable outcomes of the Nash equilibrium of the noncooperative pricing game discussed in the previous subsection, one can naturally design improved pricing schemes that can maintain each provider's self-interest while avoiding the drawbacks of the non-cooperative equilibrium.

The providers can easily realize, upon reaching the undesired Nash equilibrium, that some form of collaboration can make them better off. This gives rise to a situation in which a number of players (i.e., providers) have an incentive to collaborate, but need to agree on *how* to reach an agreement on distributing the revenues fairly. This scenario corresponds to a bargaining situation, as discussed in Chapter 7. The key question to be answered is which agreement, among all the feasible ways of allocating revenues, the providers can agree upon. This agreement should, as in any bargaining problem, be Pareto-efficient, and, for the providers' game, it must not depend on the scale by which the profits are measured nor on the order of the providers' indices.

Certainly, in such a bargaining situation, one can revert to using well-known concepts such as the Nash bargaining solution and its generalization. However, in [203] it is shown that such concepts can yield non-sensible solutions. For example, for a scenario in which M access providers are connected to a backbone provider through M links, it is shown that the generalized Nash bargaining solution with a zero disagreement point, i.e., the weighted proportional fair solution, yields M times more profit for the access providers, which is practically unfair and unacceptable to the backbone provider.

For this model, this unfairness stems from the fact that "negotiation" is dictated by two kinds of compromise. First, on a route traversing a sequence of providers, these providers negotiate how to share the revenues collected from this route, according to their respective contributions. Second, a provider carrying traffic on multiple routes needs, because of its capacity constraint, to negotiate with others on how to allocate its capacity among different routes, or, equivalently, the end-to-end price of the routes it serves.

As an alternative, [203] proposes a fair allocation that takes into account both negotiations, i.e., intra-route and inter-route. In this scheme, on each route $r \in \mathcal{R}$, the providers agree to share the revenues according to the following rule:

$$\frac{p_{lr}-c_l}{c_l} = \frac{p_{mr}-c_m}{c_m}, \ \forall l, m \in \mathcal{L}_r.$$
(14.70)

This rule, as computed in [203], is a result of the generalized Nash bargaining solution applied over a single route r with zero disagreement point and the bargaining power of every provider i set to be equal to the provider's cost, i.e., c_i . The allocations in (14.70) can be seen as a return on investment rate and are not influenced by the end-to-end prices. Thus each provider can, independently, choose its local price (i.e., p_{lr}) in such a way that, when it is combined with those of others, the resulting end-to-end price would maximize its own total profit. Hence, for end-to-end prices, this leads to the use of a Nash game in which each provider's allocated profit is the payoff and the local prices are the strategies.

This game is different from the one discussed in the previous subsection, since a provider's revenue and profit are no longer solely determined by its own prices and the demand function. They are also dependent on the other providers' prices through the allocation rule in (14.70). For instance, for provider i, given (14.70), the total generated profit can be rewritten as

$$v_{i}(\mathbf{p}_{i},\mathbf{p}_{-i}) = \sum_{l \in \mathcal{E}_{r}} \sum_{r \in \mathcal{R}_{l}} \left(\frac{c_{l}(p_{lr} + p_{-lr})}{\sum_{m \in \mathcal{L}_{r}} c_{m}} - c_{l} \right) d_{r}(p_{lr} + p_{-lr}).$$
(14.71)

In this new game, each provider *i*'s objective is to find the prices that solve the following optimization problem:

$$\max_{p_{lr}>0} v_i(\mathbf{p}_i, \mathbf{p}_{-i}),$$

s.t.
$$\sum_{r\in\mathcal{R}_l} d_r(p_{lr} + p_{-lr}) \le C_l, \forall l \in \mathcal{E}_i.$$
 (14.72)

It is shown in [203] that, for the game defined by the simultaneous optimizations in (14.72), a Nash equilibrium exists. The proof is done in [203] by showing that, for any opponents' strategy profile, the problem in (14.72) admits a unique maximizer, and by using the Brouwer Fixed-Point Theorem to show the existence of a solution to a system of fixed-point equations. From the proof of existence it is also demonstrated that, under this new Nash game with fair revenue-sharing, the Internet providers always have an incentive to upgrade their links, as long as there exists unserved demand. As a direct result, it turns out that the Nash equilibrium of the game produces an allocation that Pareto-dominates the non-cooperative pricing strategy of the previous subsection.

14.3.3 Distributed algorithm for finding a Nash equilibrium

In the previous subsection, we determined that using revenue-sharing concepts can lead to an equilibrium that is fair and more efficient than the pure non-cooperative case. One key issue that remains to be tackled is to devise an algorithm for reaching this equilibrium. In [203], a distributed algorithm is devised for finding the equilibrium. This algorithm, shown as Algorithm 14.1, is generally composed of five phases. The gist of this algorithm is based on two key observations made in [203]. First, for a given route *r*, the end-to-end price is determined by the link with the largest scaled Lagrangian multiplier $\frac{\mu}{c} = \arg \max_{l \in \mathcal{L}_r} \frac{\mu_l}{c_l}$. Second, the Lagrangian multipliers can be computed iteratively based on the traffic loads on the links. In the first phase, the provider maintains and updates a state variable μ_l (which is a Lagrangian multiplier) for each link *l*, based on the following rule:

$$\mu_{l} = \max\left\{0, \mu_{l} + w_{l} \left(\sum_{r \in \mathcal{R}_{l}} d_{r}(p_{r}) - C_{l}\right)\right\},$$
(14.73)

with $w_l > 0$ a small constant.

The next three steps of the algorithm rely mainly on stamping control packets so as to allow the first-hop provider to obtain, from the destination, in Phase 4, the values for the largest-scaled Lagrangian multiplier on the route and the sum of link costs. Once this information is available, the first-hop provider can easily compute the optimal price p_r^* by solving [203]

$$p_r^* = \sum_{m \in \mathcal{L}_r} c_m + \frac{\mu}{c} + g_r(p_r^*).$$
(14.74)

In the final phase, each provider on the route records its share of the price $\frac{p_r^* c_l}{\sum_{m \in \mathcal{L}_r} c_m}$ on the link / it is providing for forwarding the packet. In this phase, it is assumed that a system is established for the providers to collect or distribute revenues, presumably on a time scale much longer than that of the traffic dynamics. The five phases in Algorithm 14.1 are operated on a route-by-route basis.

Note that, since the cost of a link c_l is private information for each provider, the providers can instead use a virtual value to avoid any cheating (the algorithm would remain unchanged). Practically, the algorithm, as discussed in [203], can be relatively easily implemented since it does not require a lot of information to be maintained by the providers. More importantly, it is shown in [203] that, when applied to all routes, Algorithm 14.1 converges to the Nash equilibrium of (14.72). This is shown through analytical proof and using a numerical example.

In this section, we have illustrated the use of non-cooperative game theory for setting service prices in a multi–service provider communication network such as the Internet. We identified a Nash equilibrium of the non-cooperative solution and showed that it is inefficient and undesirable. In consequence, we showed how, using concepts from bargaining theory, one can improve the efficiency of the Nash equilibrium and reach a desired agreement among the providers, even when the providers are acting independently and
Algorithm 14.1 Distributed algorithm for finding the equilibrium.

Phase 1: State variable

Each provider maintains a state variable μ_l for each link *l*.

This state variable is updated periodically using the well-defined rule of (14.73).

Phase 2: Control/pricing packets update

A number of packets (e.g., control packets) are assigned for carrying pricing information.

Each pricing packet carries two fields:

- (i) A first field that contains information on a scaled Lagrangian multiplier
- (ii) A second field that contains information on the sum of costs over all links in a route.

The router on each link in a packet's route to the destination updates the first field if the local link has a larger-scaled Lagrangian multiplier.

The router on each link in a packet's route to the destination always updates the second field.

Phase 3: Feedback

The destination returns the two fields to the sending host using a control packet (e.g., an ACK packet).

Phase 4: First-hop provider optimal pricing

The first-hop provider is assumed to maintain an estimate of the demands initiating from its network on each route.

The first-hop provider receives the two fields from the third phase and uses them to compute and update the optimal price along this route.

Phase 5: Revenue-sharing

For the established connection, all packets are stamped with the optimal price and the sum of costs.

Each one in the sequence of providers over this route can compute its share of the price using the stamped information in the packets.

are self-interested. Furthermore, we explained how a practical algorithm can be built to reach the equilibrium point. We note that the main focus of the study has been on profit maximization of the providers. Nonetheless, this study can be extended to account for the preferences of the users and their behavior. To do so, one can combine different game-theoretic concepts such as Nash equilibria, Stackelberg equilibria, and repeated games. Other extensions can include the integration of routing as part of the providers' strategies as well as the study of the price of anarchy so as to further inspect the efficiency of the reached suboptimal Nash equilibrium.

14.4 Cooperative file sharing in peer-to-peer networks

Peer-to-peer (P2P) communication networks such as Napster [349], Gnutella [168], KaZaa [241], and BitTorrent [74] are self-organizing and distributed systems with no

central authority that manages the network. Instead, each peer in a P2P system is independent and can dynamically self-optimize and take communication decisions depending on its perceived environment. For instance, P2P systems allow the pooling of the resources of many autonomous nodes to provide an accessible platform for distributed data exchange, storage, or file sharing. The interest in P2P systems stems from their nature as enablers for communication services and applications. Consequently, numerous research activities [304, 394, 385, 112, 80, 36, 426, 311, 408] have focused on studying and analyzing peerto-peer systems from various perspectives, ranging from data exchange [394], resource reciprocation [385], and cooperative download [36] to legal issues [408].

Because of the heterogeneous and limited capabilities of the peers (e.g., in terms of upload and download rates) as well as the decentralized nature of P2P file sharing, there is a strong need for new architectures and protocols for data exchange and packet scheduling among the peers. In [385], the authors propose a study that analyzes the problem of resource reciprocation of multimedia content among peers using a stochastic game. Furthermore, packet scheduling schemes and peer contributions are studied in [96] for improving the performance of video streaming of P2P content-distribution networks. Different architectures, including tree-based [73] and mesh-based approaches [304, 295], are analyzed for content dissemination in P2P networks. Given these different architectures, a number of peers can efficiently share different content, resources, or fragments of various files.

Most of the aforementioned work in P2P networks focuses on the P2P sharing of resources after these resources are entirely acquired by the peers, independently, from different seeds. However, a prominent challenge of P2P systems is the scenario in which a number of peers compete to download, concurrently, a number of resources, e.g., files or file fragments that complement their already-owned resources, from the same seed at the time the availability of the resources is announced. Whenever a new file, fragment, or group of files/file fragments becomes available at a certain seed, i.e., a server or another peer, a large number of interested peers, e.g., a *flash crowd* of nodes interested in these resources, will, concurrently, attempt to download these resources from the seed. Eventually, because of the limited upload capabilities of the seed and the heterogeneous characteristics of the peers, this scenario yields increased download delays for the competing peers (this issue is highlighted in [295] from a protocol and topology point of view for asymmetric P2P networks). Hence, one important challenge is to propose cooperative schemes for P2P networks that can alleviate the download delays for the peers during this phase of simultaneous download of resources at the time they become available.

In this section, based on the work in [408], we study cooperative strategies, using cooperative game theory, notably coalitional games, to enable peers to improve their download delays during the concurrent download of resources from a common seed. First, we highlight the problem and formulate a cooperative model using coalitional games. Then, we develop an algorithm for coalition formation, and show some results on the optimal division of the peers' download requests between the main seed and other cooperative peers. Finally, we assess the performance of cooperative file sharing in P2P systems for the cases of two peers and N peers.

14.4.1 Cooperative vs. non-cooperative file sharing

Consider a P2P network consisting of N peers, and let \mathcal{N} denote the set of all peers. In this network, each peer already owns a small number of resources (e.g., files or fragments) related to a particular content and is seeking to download the remainder of this content. Whenever a seed, e.g., a server or another independent peer (not in the set \mathcal{N}) that owns the entire content of interest to the peers, announces the availability of a number of resources, all N peers will concurrently attempt to download these resources from the seed. This model assumes that the seed owns *all* the data related to the content of common interest to the peers, while each peer in \mathcal{N} owns different chunks of this data.

Each peer $i \in \mathcal{N}$ has an upload rate μ_i , and, in a non-cooperative manner, attempts to download the remainder of its data from the common seed with a download rate λ_i . For any peer *i*, the arrival process for the download requests is considered exponentially distributed with parameter λ_i^{-1} . In such a scenario, given that the seed has an upload rate of μ^s , and considering that the file service process is an M/D/1 queue, the average delay τ_{nc} for any peer $i \in \mathcal{N}$ non-cooperatively downloading data from the seed is

$$\tau_{\rm nc} = \frac{\lambda^s}{2\mu^s (\mu^s - \lambda^s)},\tag{14.75}$$

where $\lambda^{s} = \sum_{i \in \mathcal{N}} \lambda_{i}$ is the total request rate at the seed.

As the seed has a limited upload rate μ^s , one can see from (14.75) that, whenever the number of peers interested in the seed's content is large, the download delay increases significantly. In particular, the total download requests that the seed can handle must satisfy the stability condition of the queueing system in (14.75), i.e., $\lambda^s < \mu^s$. As a direct result of this condition, the number of peers that can concurrently download the data from the seed, non-cooperatively, is limited and strongly depends on the heterogeneous download rates of the peers. For example, a single peer with large download rate can congest the seed and forbid other peers from utilizing the resource.

To reduce their delay, instead of solely downloading the remaining content directly from the seed, the peers can cooperate, downloading the content of interest from the seed as well as from a subset of other peers. By doing so, the peers can potentially reduce the load on the seed and possibly improve their delay. Within each group of cooperating peers, i.e., a coalition $S \subseteq N$, every peer $i \in S$ can direct its download requests to the seed as well as to other cooperative peers $j \in S \setminus \{i\}$ with a certain fraction p_{ij} such that

$$0 \le p_{ij} \le \frac{\mu_j}{\lambda_i}, \forall j \in S \setminus \{i\} \text{ and } \sum_{j \in S \setminus \{i\}} p_{ij} + p_{i0} = 1,$$
(14.76)

where the upper bound in the first condition ensures that no peer $i \in S$ will download data from another peer $j \in S$ with a rate that exceeds the upload rate μ_j of peer j, and $0 \le p_{i0} \le \frac{\mu^s}{\lambda_i}$ is the fraction of requests directed to the seed from peer i. For every peer $i \in S$, we define the $(|S| + 1) \times 1$ ($|\cdot|$ is the set cardinality operator) column vector $\mathbf{p}_i = [p_{i0}, \dots, p_{i|S|}]^T$ of all the fractions of requests directed by by peer i to its partners



Fig. 14.1 Illustration of peer coalitions for content download.

in *S* and the seed. Note that $p_{ii} = 0, \forall i \in S$. Furthermore, we let $\mathbf{P}_S = [\mathbf{p}_1, \dots, \mathbf{p}_{|S|}]$ be the $(|S|+1) \times |S|$ matrix of all the fractions of the peers in coalition *S*. Note that any peer outside a coalition *S*, i.e., in $\mathcal{N} \setminus S$, will not allow peers inside *S* to use its upload bandwidth.

Subsequently, for any cooperative peer *i*, member of a coalition $S \subseteq \mathcal{N}$, that is downloading data from the seed and the peers in $\mathcal{K} \subseteq S$ with $|\mathcal{K}| = K$, given that each of the K queues is independent, and by using Little's law [71], the average download delay will be

$$\tau_i(\mathbf{P}_{\mathcal{S}}) = \frac{1}{\mathcal{K}+1} \left(\sum_{j \in \mathcal{S}, \ j \neq i} \frac{\Lambda_j}{2\mu_j(\mu_j - \Lambda_j)} + \tau_s \right), \tag{14.77}$$

where the first term is the delay from the data downloaded by peer *i* from *K* partners in coalition *S*, with μ_j the service rate of peer $j \in S$, and τ_s the delay from the data downloaded by peer *i* from the seed, which is the same for all $i \in S$ and is given using (14.75) with $\lambda^s = \sum_{i \in S} p_{i0}\lambda_i$. Furthermore, $\Lambda_j = (\mathbf{P}_S)_j\lambda$ in (14.77) represents the total load at peer $j \in S$, with $(\mathbf{P}_S)_j$ the *j*th line of \mathbf{P}_S and λ an $|S| \times 1$ vector, with each element λ_k corresponding to the download rate of a peer $k \in S$. Note that we assume that, for any coalition *S* that forms, $\exists i \in S$ such that $p_{i0} \neq 0$, i.e., *at least one peer downloads from the seed*, otherwise the download delay in (14.77) is considered to be infinite. By adequately selecting the distribution of their download requests, i.e., the matrix \mathbf{P}_S , the cooperative peers might be able to improve their average. An illustration of the considered model is shown in Fig. 14.1. Although the peers can divide their download requests in any way between the cooperative peers and the seed, one scheme that the peers can adopt is to distribute their requests in such a way as to minimize the total social cost for a coalition, i.e., *total* average delay experienced by the coalition as a whole. Thus, given a coalition $S \subseteq N$ of cooperative peers, the peers in the coalition distribute their download requests, i.e., compute the vectors \mathbf{p}_i , $\forall i \in S$ by jointly solving the following optimization problem:

$$\min_{\mathbf{P}_{S}} \sum_{i \in S} \tau_{i}(\mathbf{P}_{S}),$$
s.t. $0 \leq p_{i0} < \frac{\mu^{s}}{\lambda_{i}} \text{ and } 0 \leq p_{ij} < \frac{\mu_{j}}{\lambda_{i}}, \forall j \in S \setminus \{i\}, i = 1 \dots |S|,$

$$\sum_{j \in S \setminus \{i\}} p_{ij} + p_{i0} = 1, i = 1 \dots |S|,$$
(14.78)

where τ_i is given by (14.77) and the two constraints describe previously mentioned properties of the download-request fractions. Note that at least one peer in the coalition must be connected to the seed, i.e., $\exists i \in S$ with $p_{i0} \neq 0$, otherwise the delay is infinite. Subsequently, whenever a group of peers cooperates within a coalition *S*, they can compute the download request distribution that minimizes the total average delay of their coalition based on (14.78). As a result of this optimization, the delay of every peer *i* in coalition *S* is given by $\tau_i(\mathbf{P}_S^*)$ in (14.77), with \mathbf{P}_S^* denoting the solution of (14.78).

Nonetheless, although minimizing the social cost is an attractive approach, in several scenarios the social optimum and the individual incentives of the peers might not be aligned, i.e., optimizing the social cost for a coalition does not guarantee a better individual delay for every peer involved in the coalition. Hence, there is a need to devise cooperative strategies that allow each peer to autonomously form coalitions such as in Fig. 14.1 while taking into account two conflicting objectives: (i) improving its individual delay through cooperation, and (ii) distributing its download requests inside the coalition so as to minimize the overall social cost for the network, based on (14.78).

14.4.2 File sharing as a coalitional game in partition form

By inspecting Fig. 14.1, we can see that the objective of the problem is to study cooperative strategies and the formation of coalitions. For this, we can use the framework of coalitional games that was developed in Chapter 7.

For this purpose, denoting by \mathfrak{B} the set of all partitions of \mathcal{N} and by $\phi_i(S, \Pi)$ the payoff to any peer *i* in coalition $S \in \Pi$ within a partition $\Pi \in \mathfrak{B}$ of \mathcal{N} , we define the coalitional value as follows:

$$V(S,\Pi) = \{\phi(S,\Pi) \in \mathbb{R}^S | \forall i \in S, \phi_i(S,\Pi) = -\tau_i(\mathbf{P}_S^*)\},$$
(14.79)

where $\tau_i(\mathbf{P}_S^*)$ is given by (14.77), with \mathbf{P}_S^* the solution of (14.78) for coalition S and the minus sign being inserted for convenience, in order to turn the problem into a maximization problem.

The delay $\tau_i(\mathbf{P}_S)$ in (14.77) depends, through its second term, not only on the download request distribution inside *S*, but also on the distribution outside *S* through the seed, so it depends on the partition Π (for notational convenience, this dependence is dropped from $\tau_i(\mathbf{P}_S^*)$). Therefore, we can see that the value in (14.79) is in *partition form* because of the dependence on the entire partition and not only on the considered coalition. Moreover, we note that in (14.79) we have a singleton set that assigns to every coalition a *vector* of payoffs and not a single value. Therefore, we have a game with *non-transferable utility*. The fact that the utility for a peer depends on its delay, which is a non-transferable metric, implies that one cannot define a utility that is transferable, i.e., that can be divided in any way among the peers.

Hence, the peer-to-peer cooperative file-sharing problem is formulated as an (\mathcal{N}, V) coalitional game in partition form with non-transferable utility, where the mapping V is given in (14.79). As mentioned in Chapter 7, the partition-form class of coalitional games is a framework that is more complex than the characteristic form, because of the dependence of the value on the network partition. The main challenge of the formulated P2P coalitional game in partition form is to construct algorithms for forming coalitions such as those in Fig. 14.1. In essence, coalitional games in partition form are classified as coalition-formation games as discussed in Chapter 7.

Prior to developing an algorithm to solve the studied P2P coalition-formation game, we define the following concept:

DEFINITION 14.3 Given any peer $i \in \mathcal{N}$, a preference relation, denoted by \succeq_i , is defined as a complete, reflexive, and transitive binary relation over the set of all coalition/partition pairs that peer i can be a member of, i.e., the set $\{(S_k, \Pi) | S_k \subseteq \mathcal{N}, i \in S_k, S_k \in \Pi, \Pi \in \mathfrak{B}\}$.

The concept of preference relation that we define here enables us to provide, for any peer $i \in \mathcal{N}$, a quantification of its benefit for every potential coalition that it can form. Thus, the preference relation \succeq_i can be used to compare the peer's preference between any two coalitions $S_1 \subseteq \mathcal{N}$, $S_1 \in \Pi$, and $S_2 \subseteq \mathcal{N}$, $S_2 \in \Pi'$ such that $i \in S_1$ and $i \in S_2$ and their respective partitions. This notion differs slightly from the concept of a preference relation defined in Chapter 3 in the sense that the latter was defined for comparing *collections of coalitions* (in characteristic form, although its extension to partition form is straightforward) and not only single coalitions.

Thus, using the preference relation defined above, $(S_1, \Pi) \succeq_i (S_2, \Pi')$ implies that peer *i* is better off working cooperatively in coalition S_1 when Π is in place than being a member of coalition S_2 when Π' is in place, or at least *i* prefers both coalition/partition pairs equally (when the preference is strict, it is denoted by \succ_i). Note that this preference relation can be used to compare two coalitions in the same partition, or the same coalition in two different partitions. For the peer-to-peer coalition-formation game, we propose the following preference relation for any peer $i \in \mathcal{N}$:

$$(S_1, \Pi) \succeq_i (S_2, \Pi') \Leftrightarrow w_i(S_1, \Pi) \ge w_i(S_2, \Pi'), \tag{14.80}$$

where $S_1 \in \Pi$, $S_2 \in \Pi'$, with $\Pi, \Pi' \in \mathfrak{B}$, are any two coalitions that contain peer *i*, i.e., $i \in S_1$ and $i \in S_2$, and w_i is a preference function defined for a peer $i \in \mathcal{N}$ as follows:

$$w_i(S,\Pi) = \begin{cases} \phi_i(S,\Pi), & \text{if } \phi_j(S,\Pi) \ge \phi_j(S \setminus \{i\},\Pi), \\ & \forall j \in S \setminus \{i\} \& S \notin h(i) \text{ or } (|S| = 1), \\ 0, & \text{otherwise}, \end{cases}$$
(14.81)

where $\phi_i(S, \Pi)$ is given by (14.77)–(14.79) and represents the delay perceived by peer $i \in S$ when partition Π is in place, and h(i) is the history set of peer *i* that holds the coalitions of size larger than 1 that *i* was a member of in the past, and had left.

The main idea behind the function w_i is that a peer *i* assigns a preference equal to its achieved payoff for any coalition/partition pair (S, Π) as long as coalition *S* is either a singleton coalition (i.e., peer *i* is acting non-cooperatively) or *S* is a coalition of size larger than 1 which was not previously visited by *i* (not in h(i)), and where the joining of peer *i* to coalition *S* is not detrimental to any of the peers already in $S \setminus \{i\}$. Otherwise, the peer assigns a zero preference value to any coalition whose members' payoffs decrease because of the presence of *i*, since the members of such a coalition will refuse to have peer *i* join the coalition. Also, peer *i* assigns a zero preference to any coalition it has already visited and *left* because peer *i* has no incentive to revisit such a coalition (this can be seen as a basic learning rule by which a peer has no incentive to revisit a coalition that, eventually, ended up being detrimental).

14.4.3 Distributed algorithm for coalition formation

In forming coalitions, we can utilize a rule that can be followed individually by each peer in the network based on the previously defined preference relation:

DEFINITION 14.4 (*Change Rule*) For a given partition $\Pi = \{S_1, ..., S_M\}$ of the set of peers \mathcal{N} , a peer *i* can decide to change its coalition, i.e., leave its current coalition S_m , for some $m \in \{1, ..., M\}$, and join another coalition $S_k \in \Pi \cup \{\emptyset\}$, with $S_k \neq S_m$, if and only if $(S_k \cup \{i\}, \Pi') \succ_i (S_m, \Pi)$, where $\Pi' = \{\Pi \setminus \{S_m, S_k\}\} \cup \{S_m \setminus \{i\}, S_k \cup \{i\}\}$ is the partition resulting from the change. This change rule is represented by $\{S_m, S_k\} \rightarrow \{S_m \setminus \{i\}, S_k \cup \{i\}\}$, and $\Pi \rightarrow \Pi'$.

Using this change rule, any peer can decide to leave its present coalition $S_m \in \Pi$ and join a new coalition $S_k \in \Pi$, forming a new partition Π' , as long as the new pair $(S_k \cup \{i\}, \Pi')$ is strictly preferred over (S_m, Π) through the preference relation defined by (14.80) and (14.81). That is, a peer can move to a new coalition if it can strictly improve its payoff *without* decreasing the payoff to any member of the new coalition, i.e., given the consent of the new members as in (14.80). Furthermore, each time a peer executes a change rule from its current coalition $S_m \in \Pi$, coalition S_m is stored in its history set h(i) (if $|S_m| > 1$).

Consequently, as in [408], for coalition formation one can use Algorithm 14.2 which consists of three main stages: peer discovery, coalition formation, and cooperative

Algorithm 14.2 Peer-to-peer coalition-formation algorithm.

Initial state

The network is partitioned by $\Pi_0 = \{S_1, ..., S_M\}$. At the beginning of all time, the network is non-cooperative; hence, $\Pi_0 = \mathcal{N}$.

Stage 1: Peer discovery

The seed announces the availability of resources sought by all peers. Each peer in \mathcal{N} attempts to download the content from the seed.

Using peer-discovery algorithms such as trackers [168, 74, 304], the

peers discover the presence of other peers at the seed.

Stage 2: Coalition formation

repeat

Each peer $i \in \mathcal{N}$ investigates potential change operations using the preference in (14.80) by engaging in pairwise negotiations with existing coalitions in partition Π (initially $\Pi = \Pi_0$).

Once a change operation is found:

a) Peer *i* leaves its current coalition.

b) Peer *i* updates its history h(i), if needed.

c) Peer *i* joins the new coalition with the consent of its members.

until convergence to a Nash-stable partition

Stage 3: Cooperative download

The formed coalitions perform cooperative download from the seed and their partners as discussed in Section 14.4.1.

download. During the first stage, as the seed announces the availability of the resources, all interested peers attempt to download the content from the seed. Meanwhile, the peers can use a tracker or other well-known peer-discovery algorithm such as in [74, 304, 168] to learn of the presence of other peers downloading from the seed. Once peer discovery is done, the peers engage in the coalition-formation stage. In this stage, each peer attempts to estimate its payoff from changing its current coalition and joining another coalition (or peer). Once a peer finds a potential change possibility (satisfying (14.80) and (14.81)), it can make a *distributed* decision to break from its current coalition and join a new coalition (the change rule guarantees that the new coalition accepts the joining of this peer). In this stage, we consider that the peers make decisions in a certain random order (dictated by who first requests to cooperate). The peers make their change decisions based on an assessment, using (14.77) and (14.78), of the payoff given the current partition, and not on the long-term payoff. Such a strategy can be seen as a *myopic* strategy. Following the convergence of the coalition-formation process, a partition Π_f is in place in the network for downloading the resources from the seed, which occurs in the last stage of the algorithm.

The stability of the partition Π_f can be studied using the following stability concept, which is, in essence, related to the idea of a Nash equilibrium from non-cooperative games:

DEFINITION 14.5 A partition $\Pi = \{S_1, ..., S_M\}$ is Nash-stable if $\forall i \in \mathcal{N}$ such that $i \in S_m, S_m \in \Pi, (S_m, \Pi) \succeq_i (S_k \cup \{i\}, \Pi')$ for all $S_k \in \Pi \cup \{\emptyset\}$ with $\Pi' = (\Pi \setminus \{S_m, S_k\} \cup \{S_m \setminus \{i\}, S_k \cup \{i\}\})$.

Hence, a partition Π is Nash-stable if no peer has an incentive to move from its current coalition to another coalition in Π or to deviate and act alone. Given this notion, the convergence of the peer-to-peer coalition-formation algorithm during the coalition-formation stage is guaranteed, as shown in [408]:

THEOREM 14.6 Starting from any initial network partition Π_0 , the coalition-formation stage of Algorithm 14.2 always converges to a final Nash-stable network partition $\Pi_f \in \mathfrak{B}$.

The proof is, as shown in [408], mainly a result of two facts: (i) coalition formation consists of a sequence of change rules based on (14.80) and (14.81), and (ii) the number of partitions of the set \mathcal{N} is *finite* (given by the Bell number).

14.4.4 Coalition formation in two-peer and *N*-peer networks

In order to gain further insight into the properties of the formulated coalition-formation game, we start by studying the case of a network with two peers that seek to download resources from a common seed at the time the resources are announced by the seed. For this case, under certain conditions on the upload rate of the seed μ^s , the optimal social cost–minimizing solution for the coalition $S = \{1, 2\}$, if this coalition forms, can be given by the following theorem:

THEOREM 14.7 Consider two peers seeking to download data from a common seed and having respective download rates λ_1 , λ_2 , and upload rates μ_1 , μ_2 . In this scenario, given $\lambda_T = \lambda_1 + \lambda_2$, whenever the seed's upload rate $\mu^s > \lambda_T$ it satisfies

$$\mu^{s} \leq \begin{cases} (\sqrt{2}+1)\mu_{1} - \mu_{2} + \lambda_{T}, & \text{if } \mu_{1} \leq \mu_{2} < (\sqrt{2}+1)\mu_{1}, \\ (\sqrt{2}+1)\mu_{2} - \mu_{1} + \lambda_{T}, & \text{if } \mu_{2} < \mu_{1} \leq (\sqrt{2}+1)\mu_{2}. \end{cases}$$
(14.82)

The optimal solution that minimizes the social cost for coalition $S = \{1, 2\}$, if S forms, is $\mathbf{P}_{S}^{*} = \begin{bmatrix} p_{10}^{*} & p_{20}^{*} \\ p_{12}^{*} & p_{21}^{*} \end{bmatrix}$, with

$$p_{12}^* = 1 - p_{10}^* = \frac{(\sqrt{2} - 2)(\mu_1 + \mu^s - \lambda_T) + \sqrt{2}\mu_2}{2\lambda_1},$$
(14.83)

$$p_{21}^* = 1 - p_{20}^* = \frac{(\sqrt{2} - 2)(\mu_2 + \mu^s - \lambda_T) + \sqrt{2}\mu_1}{2\lambda_2}.$$
 (14.84)

Proof The proof of this theorem follows from [408]. Consider the scenario in which two peers, with respective download rates λ_1 , λ_2 , and load rates μ_1 , μ_2 , wish to download resources from a common seed having an upload rate $\mu^s > \lambda_T$, $\lambda_T = \lambda_1 + \lambda_2$. To

analyze whether it is to their benefit to form coalition $S = \{1, 2\}$, the peers need to find the social cost-minimizing request rate distribution matrix \mathbf{P}_{5}^{*} , as in (14.78). Denote by $x_{12} = p_{12}\lambda_1$ the number of download requests directed by peer 1 to peer 2 and $x_{21} = p_{21}\lambda_2$ its counterpart from peer 2 on peer 1. Thus, given the download delays $\tau_1(\mathbf{P}_5)$ and $\tau_2(\mathbf{P}_5)$ of peers 1 and 2, respectively, as in (14.77), the total delay of the coalition S (the dependence on \mathbf{P}_5 dropped for simplicity) is

$$\tau_{total}^{S}(\mathbf{P}_{S}) = \tau_{1} + \tau_{2} = \frac{x_{12}}{4\mu_{2}(\mu_{2} - x_{12})} + \frac{x_{21}}{4\mu_{1}(\mu_{1} - x_{21})} + \tau_{s}, \quad (14.85)$$

with $\tau_s = \frac{\lambda_T - x_{12} - x_{21}}{2\mu^s(\mu^s - (\lambda_T - x_{12} - x_{21}))}$ as in (14.75), with $\lambda^s = (\lambda_1 - x_{12}) + (\lambda_2 - x_{21})$.

To find the optimal \mathbf{P}_{S}^{*} using (14.78), the peers need to minimize $\tau_{total}^{S}(\mathbf{P}_{S})$ given the constraints

$$0 \le p_{12} < \frac{\mu_2}{\lambda_1}, \ 0 \le p_{21} < \frac{\mu_1}{\lambda_2}, \quad p_{12} + p_{10} = 1, \ p_{21} + p_{20} = 1.$$
 (14.86)

Note that, since $\mu^{s} > \lambda_{T}$, p_{10} and p_{20} always satisfy the constraints of (14.78). By solving $\frac{\partial \tau_{lotal}^{s}(\mathbf{P}_{5})}{\partial x_{12}} = 0$ and manipulating the resulting quadratic equation, we obtain, using $\mu^{s} > \lambda_{T}$, two roots for the equation

$$x_{12}' = (\alpha + 1)x_{21} + (\alpha + 1)(\mu^{s} - \lambda_{T}) + (\alpha + 2)\mu_{2}, \qquad (14.87)$$

where $\alpha = \pm \sqrt{2}$. One can verify that with $\alpha = \sqrt{2}$, and given $\mu^s > \lambda_T$, (14.87) yields $x'_{12} > \mu_2$, which is an infeasible solution since we must have $x_{12} \le \mu_2$. Hence, the only possible solution is

$$x_{12} = (1 - \sqrt{2})x_{21} + (1 - \sqrt{2})(\mu^{s} - \lambda_{T}) + (2 - \sqrt{2})\mu_{2}.$$
(14.88)

In a symmetric manner, by setting $\frac{\partial \tau_{\text{total}}^{S}(\mathbf{P}_{S})}{\partial x_{21}} = 0$, we find

$$x_{21} = (1 - \sqrt{2})x_{12} + (1 - \sqrt{2})(\mu^s - \lambda_T) + (2 - \sqrt{2})\mu_1, \quad (14.89)$$

so the optimal solution is the solution of the system (14.88) and (14.89), which satisfies the constraints, and is given by

$$x_{12}^* = \frac{(\sqrt{2}-2)}{2}\mu_1 + \frac{\sqrt{2}}{2}\mu_2 + \frac{(\sqrt{2}-2)}{2}(\mu^s - \lambda_T), \qquad (14.90)$$

$$x_{21}^{*} = \frac{(\sqrt{2}-2)}{2}\mu_{2} + \frac{\sqrt{2}}{2}\mu_{1} + \frac{(\sqrt{2}-2)}{2}(\mu^{s} - \lambda_{T}).$$
(14.91)

By substituting (14.90) and (14.91) in the constraints of (14.86) and through algebraic manipulation while maintaining $\mu^{s} > \lambda_{T}$, we find that the derived solution is feasible whenever (14.82) is satisfied. Finally, given $x_{12}^{*} = p_{12}^{*}\lambda_{1}$, $x_{21}^{*} = p_{21}^{*}\lambda_{2}$, $p_{10}^{*} = 1 - p_{12}^{*}$, and $p_{20}^{*} = 1 - p_{21}^{*}$, the optimal matrix \mathbf{P}_{5}^{*} can be computed. Theorem 14.7 provides an analytical solution that the peers can use to compute their optimal policy for download-request distribution, given that the seed's upload rate satisfies (14.82).¹ Once the two peers compute the optimal divisions for the coalition $S = \{1, 2\}$, using (14.80) and (14.81) (as in the change rule), they would agree to join into a single coalition S if at least one peer is better off in the cooperative case without decreasing the payoff to the other peer, i.e., $\tau_1(\mathbf{P}_S^*) \ge \tau_{nc}$ with $\tau_2(\mathbf{P}_S^*) \ge \tau_{nc}$ or $\tau_1(\mathbf{P}_S^*) \ge \tau_{nc}$ with $\tau_2(\mathbf{P}_S^*) \ge \tau_{nc}$, where τ_{nc} is the non-cooperative delay as in (14.75).

As seen by (14.77), (14.83), and (14.84), for a given seed load $\mu^s > \lambda_1 + \lambda_2$, the decision of the two peers to join inherently depends on the characteristics of the peers, i.e., their download and upload rates. For instance, from Theorem 14.7, it is interesting to note that the optimal divisions p_{12}^* and p_{21}^* are linear functions of the upload rates $\mu_1, \mu_2, \text{ and } \mu^s$. Furthermore, given a fixed upload rate for a peer, e.g., μ_1 , as the upload rate μ_2 of the other peer increases, the social cost–minimizing solution dictates that peer 1 leeches less and less on peer 2. In order to emphasize this aspect and show how the upload rate of the peers affects their cooperative file-sharing policies (14.83) and (14.84), we consider the following example:

Example 14.3 Consider two peers with, respectively, upload rates $\lambda_1 = 1.4$ Mbps, $\lambda_2 =$ 1 Mbps, and download rates $\mu_1 = 512$ kbps, $\mu_2 \in [256, 850]$ kbps. We consider the case when a server, acting as a seed, with an upload rate of $\mu^{s} = 2.5$ Mbps, announces the availability of resources that peers 1 and 2 seek to download. These parameters are chosen such that, at all μ_2 , (14.82) is verified and both peers find it always beneficial to form a single coalition as per the change rule and (14.80). For this scenario, we are interested in the variation of the optimal policies of the peers, as in (14.83) and (14.84), when coalition $S = \{1, 2\}$ forms and as μ_2 changes. By inspecting (14.83), we can see that, as the upload capability of peer 2 increases, peer 1 tends to put more download requests on peer 2, while peer 2 decreases its download requests on peer 1. For instance, while at $\mu_2 = 256$ kbps, peer 1 does not download any data from peer 2 as $p_{12}^* \approx 0$ as in (14.83) (but still benefits from cooperation arising from the reduced load on the server from the cooperation of peer 2); at $\mu_2 = 850$ kbps, peer 1 downloads up to $p_{12}^* = 30$ percent of its data from peer 2. In contrast, the fraction of download of peer 2 from peer 1 decreases from around $p_{21}^* = 25.8$ percent at $\mu_2 = 256$ kbps down to $p_{21}^* = 8.4$ percent at $\mu_2 = 850$ kbps. Hence, as one of the peers leeches more on the other, the optimal policy of the other peer dictates that it download more from the server (which is less congested because of the cooperative behavior of the other peer) and less from the other peer. Finally, by solving the system of equalities in (14.83) and (14.84), we find that, at $\mu_2 = 577.5$ kbps, the optimal policy of both peers is to equally download from each other, i.e., $p_{12}^* = p_{21}^* = 16.3$ percent.

¹ Alternatively, in the case where (14.82) is not satisfied, the peers can always find their optimal download distributions using classical optimization methods.



Fig. 14.2 Average download delay per peer achieved by the coalition-formation algorithm and the non-cooperative case as the number of peers *N* increases.

As *N* becomes larger than 2, finding closed-form analytical results becomes complicated. However, in [408], simulations were used to assess the performance of coalition formation in a peer-to-peer system. In Fig. 14.2, we show the average download delay per peer as the number of peers downloading from the server *N* increases. We can see that, as the number of peers *N* increases, the average delay for the non-cooperative scheme increases because of the increased load on the server. In contrast, for coalition formation, although the average delay starts by increasing slowly, at N = 10 peers this average delay starts to slightly decrease with the number of peers *N* since the benefit from cooperation grows because of (i) the presence of more peers to cooperate with, and (ii) the growing need for cooperation with the increasing server load. Figure 14.2 demonstrates that the performance advantage of coalition formation increases with the size *N* and can reach up to 99.6 percent (two orders of magnitude) improvement in the average download delay per peer, relative to the non-cooperative case at N = 15 peers.

Figure 14.3 shows further statistics on the average and average maximum coalition size as *N* increases. In particular, this figure shows that when the upload rate μ^s of the server is large relative to the total download rate of the peers, then no cooperation occurs, as shown for the case of N = 2 peers. Furthermore, it shows that, using coalition formation, the resulting partition is mainly composed of an average number of medium-sized coalitions with the occasional emergence of large coalitions.



Fig. 14.3 Average and average maximum coalition size achieved by the coalition formation algorithm as the number of peers *N* increases.

Finally, the simulations in [408] also showed that, for a given peer-to-peer network, coalition formation would maintain a performance advantage as the upload rate of the server, μ^{s} , varies.

In summary, coalition-formation games in partition form constitute a suitable tool for cooperation among a number of peers seeking to download content of interest from a common seed at the time this seed announces the availability of the resources. By engaging in coalition formation, the peers can take individual decisions to join or break from a coalition while minimizing their average download delay. This approach admits numerous extensions such as considering a cost for uploading, accounting for contribution levels, integrating non-cooperative solutions at the level of a coalition, and the use of advanced solutions and algorithms for partition-form games.

14.5 Summary

In future communication networks such as Internet networks or peer-to-peer networks, the presence of autonomous and independent nodes is ubiquitous. Ultimately, networks having a centralized authority that controls the entire network will give way to distributed and self-optimizing networks. To enable the nodes of such distributed networks to communicate efficiently, route data, or interact, there is a need for game-theoretic tools and frameworks. In particular, the use of game theory is strongly indicated by the increasing demand for fair and robust algorithms to govern the operation of large-scale and decentralized networks. In fact, as shown in this chapter, numerous challenging problems such as flow control, routing, pricing, and file sharing admit interesting and insightful solutions through game-theoretic techniques ranging from classical Nash games to advanced Stackelberg and cooperative games. One can also envision emerging applications for game theory within Internet or communication networks such as in the areas of security for large-scale networks, autonomous agent deployment in communication networks, "Internet of things," communication in smart-grid networks, and even military networks.

References

- [1] http://www.bluetooth.com/bluetooth/.
- [2] http://www.cdg.org.
- [3] http://www.eecs.harvard.edu/~parkes/.
- [4] http://www.ieee802.org/16/relay/index.html.
- [5] http://www.umtsworld.com.
- [6] http://w3.antd.nist.gov/wahn_ssn.shtml.
- [7] IEEE 802.11. Draft amendment to standard for information technology IEEE 802.11:ESS mesh networking. IEEE Technical report P802.11s/D1.08, January 2008.
- [8] IEEE 802.16j Mobile Multi-hop Relay Project Authorization Request (PAR). Official IEEE 802.16j website, March 2006. http://standards.ieee.org/board/nes/ projects/802-16j.pdf.
- [9] IEEE P802.21/D14.0. Media independent handover services. September 2008.
- [10] IEEE Standard for Local and Metropolitan Area Networks. Part 16: Air Interface for Fixed Broadband Wireless Access Systems. IEEE Standard 802.16-2004 (Revision of IEEE Standard 802.16-2001). IEEE Standards Association, 2004.
- [11] Relay Task Group of IEEE 802.16. Baseline Document for Draft Standard for Local and Metropolitan Area Networks. Technical report 802.16j-06/026r4, IEEE 802 LAN/MAN Standards Committee, June 2007. http://www.qoscom.de/fileadmin/data/ ieee802/802.16j-06_026r4.pdf.
- [12] Revision of Part 15 of the Commission's Rules Regarding Ultra-Wideband Transmission Systems, First Report and Order. Federal Communications Commission, February 2002. http://www.naic.edu/~astro/RXstatus/Lnarrow/fcc_UWB.pdf.
- [13] 3GPP TR 36. 814 Technical Specification Group Radio Access Network. Further Advancements for E-UTRA Physical Layer Aspects. Technical report.3GPP, 2010. http://www. 3gpp.org/ftp/Specs/html-info/36814.htm.
- [14] Wireless LAN Medium Access Control (MAC) and Physical Layer (PHY) specifications. ANSI/IEEE Standard 802.11. IEEE Standards Association, 1999.
- [15] J. Acharya and R. D. Yates. Service provider competition and pricing for dynamic spectrum allocation. In *Proc. Int. Conf. on Game Theory for Networks (GameNets 2009)*, pages 190–198, Istanbul, Turkey, May 2009.
- [16] A. Adya, P. Bahl, J. Padhye, A. Wolman, and L. Zhou. A multi-radio unification protocol for IEEE 802.11 wireless networks. In *Proc. 1st Int. Conf. on Broadband Networks (BroadNets* 2004), pages 344–354, San Jose, CA, October 2004.
- [17] A. Akella, J. Pang, B. Maggs, S. Seshan, and A. Shaikh. A comparison of overlay routing and multihoming route control. ACM SIGCOMM Computer Communications Rev., 34(4): 93–106, 2004.

- [18] K. Akkarajitsakul and E. Hossain. An auction mechanism for channel access in vehicleto-roadside Commun. In Proc. IEEE Int. Conf. on Communications Workshops (ICC Workshops 2009), pages 1–5, Dresden, Germany, June 2009.
- [19] S. M. Alamouti. A simple transmit diversity technique for wireless communications. *IEEE J. Sel. Areas Commun.*, 16(8):1451–1458, 1998.
- [20] T. Alpcan and T. Başar. A hybrid systems model for power control in multicell wireless networks. *Performance Evaluation*, 57(4):477–495, 2004.
- [21] T. Alpcan and T. Başar. Distributed algorithms for Nash equilibria of flow control games. Advances in Dynamic Games: Applications to Economics, Finance, Optimization, and Stochastic Control. Annals of the International Society of Dynamic Games, 7:473–498, 2005.
- [22] T. Alpcan and T. Başar. *Network Security: A Decision and Game Theoretic Approach*. Cambridge University Press, 2011.
- [23] T. Alpcan, T. Başar, R. Srikant, and E. Altman. CDMA uplink power control as a noncooperative game. *Wireless Networks*, 8(6):659–670, 2002.
- [24] T. Alpcan, X. Fan, T. Başar, M. Arcak, and J. T. Wen. Power control for multicell CDMA wireless networks: A team optimization approach. *Wireless Networks*, 14(5):647–657, 2008.
- [25] E. Altman. Applications of dynamic games in queues. *Advances in Dynamic Games*, 7: 309–342, 2005.
- [26] E. Altman and Z. Altman. S-modular games and power control in wireless networks. *IEEE Trans. Autom. Control*, 48:839–842, 2003.
- [27] E. Altman, T. Başar, T. Jimenez, and N. Shimkin. Routing into two parallel links: Game-theoretic distributed algorithms. *Journal of Parallel and Distributed Computing*, 61(9):1367–1381, 2001.
- [28] E. Altman, T. Başar, and R. Srikant. Nash equilibria for combined flow control and routing in networks: Asymptotic behavior for a large number of users. *IEEE Trans. Autom. Control*, 47(6):917–930, 2002.
- [29] E. Altman, R. ElAzouzi, Y. Hayel, and H. Tembine. An evolutionary game approach for the design of congestion control protocols in wireless networks. In *Proc. 6th Int. Symp.* on Modeling and Optimization in Mobile, Ad Hoc, and Wireless Networks and Workshops (WiOpt 2008), pages 547–552, Berlin, Germany, April 2008.
- [30] E. Altman, Y. Hayel, and H. Kameda. Evolutionary dynamics and potential games in noncooperative routing. In Proc. of the Workshop on Wireless Networks: Communication, Cooperation and Competition (WNC3 2007), Limassol, Cyprus, April 2007.
- [31] E. Altman, T. Jimenez, R. Nunez Queija, and U. Yechiali. Optimal routing among ./M/1 queues with partial information. *Stochastic models*, 20(2):149–172, 2004.
- [32] E. Altman, A. A. Kherani, P. Michiardi, and R. Molva. Non-cooperative forwarding in ad-hoc networks. In Proc. 15th IEEE Int. Symp. Personal, Indoor and Mobile Radio Communications (PIMRC 2004), pages 486–498, Barcelona, Spain, September 2004.
- [33] E. Altman, A. A. Kherani, P. Michiardi, and R. Molva. Non-Cooperative Forwarding in Ad Hoc Networks. Springer, May 2005.
- [34] E. Altman, A. Kumar, C. Singh, and R. Sundaresan. Spatial SINR games combining base station placement and mobile association. In *Proc. IEEE Conf. on Computer Communications* (*INFOCOM 2009*), pages 1629–1637, Rio de Janeiro, Brazil, April 2009.
- [35] E. Altman, A. Silva, and P. Bernhard. The mathematics of routing in massively dense ad-hoc networks. In *Proc. Int. Conf. on Ad-Hoc Networks and Wireless*, pages 10–13, Sophia Antipolis, France, September 2008.

- [36] E. Altman, S. Wong, and J. Rojas-Mora. P2P business and legal models for increasing accessibility to popular culture. In *Proc. Int. Conf. on Digital Business*, pages 130–138, London, UK, February 2009.
- [37] M. P. Anastasopoulos, P. D. M. Arapoglou, R. Kannan, and P. G. Cottis. Adaptive routing strategies in IEEE 802.16 multi-hop wireless backhaul networks based on evolutionary game theory. *IEEE J. Sel. Areas Commun.*, 26(7):1218–1225, 2008.
- [38] M. P. Anastasopoulos, D. K. Petraki, R. Kannan, and A. V. Vasilakos. TCP throughput adaptation in WiMAX networks using replicator dynamics. *IEEE Trans. Syst., Man, Cybern. B, Cybern.*, 40(3):647–655, 2010.
- [39] L. Anderegg and S. Eidenbenz. Ad hoc-VCG: A truthful and cost-efficient routing protocol for mobile ad hoc networks with selfish agents. In *Proc. 9th Int. Conf. on Mobile Computing and Networking (MobiCom 2003)*, pages 245–259, San Diego, CA, USA, September 2003.
- [40] J. G. Andrews, A. Ghosh, and R. Muhamed. Fundamentals of WiMAX: Understanding Broadband Wireless Networking. Prentice Hall, 2007.
- [41] K. Apt and A. Witzel. A generic approach to coalition formation. In *Proc. Int. Workshop on Computational Social Choice (COMSOC)*, pages 347–367, Amsterdam, Netherlands, December 2006.
- [42] A. Aram, C. Singh, and S. Sarkar. Cooperative profit sharing in coalition based resource allocation in wireless networks. In *Proc. IEEE Conf. on Computer Communications (INFOCOM 2009)*, pages 2123–2131, Rio de Janeiro, Brazil, April 2009.
- [43] E. Arcaute, R. Johari, and S. Mannor. Network formation: Bilateral contracting and myopic dynamics. *Lecture Notes in Computer Science*, 4858:191–207, 2007.
- [44] T. Arnold and U. Schwalbe. Dynamic coalition formation and the core. J. Econ. Behav. Organ., 49:363–380, 2002.
- [45] K. J. Arrow. Economics and Human Welfare: The Property Rights Doctrine and Demand Revelation Under Incomplete Information. Academic Press, 1979.
- [46] A. Attar, M. Nakhai, and A. H. Aghvami. Cognitive radio game for secondary spectrum access problem. *IEEE Trans. Wireless Commun.*, 8:2121–2131, 2009.
- [47] R. J. Aumann. Subjectivity and correlation in randomized strategy. J. Math. Econ., 1(1):67– 96, 1974.
- [48] R. J. Aumann. Correlated equilibrium as an expression of Bayesian rationality. *Econometrica*, 55(1):1–18, 1987.
- [49] R. J. Aumann and J. H. Drèze. Cooperative games with coalition structures. Int. J. Game Theory, 3:317–237, 1974.
- [50] R. J. Aumann and B. Peleg. von Neumann–Morgenstern solutions to cooperative games without side payments. *Bull. Am. Math. Soc.*, 6:173–179, 1960.
- [51] L. Ausubel and P. Cramton. Demand reduction and inefficiency in multi-unit auctions. Technical report, University of Maryland, 1998. http://www.cramton.umd.edu/ papers1995-1999/98wp-demand-reduction.pdf.
- [52] A. P. Azad, E. Altman, and R. El-Azo. Routing games: From egoism to altruism. In Proc. 6th Int. Workshop on Wireless Network Measurements, pages 528–537, Avignon, France, May 2010.
- [53] K. Back and J. F. Zender. Auctions of divisible goods: On the rationale for the treasury experiment. *Rev. Finan. Stud.*, 6:733–764, 1993.
- [54] T. Başar. The Gaussian test channel with and intelligent jammer. *IEEE Trans. on Information Theory*, IT-29(1): 152–157, 1983.

- [55] T. Başar. Control and game theoretic tools for communication networks (overview). App. Comput. Math., 6:104–125, 2007.
- [56] T. Başar and A. Haurie. Feedback equilibria in differential games with structural and modal uncertainties. In J. B. Cruz, Jr., editor, *Advances in Large Scale Systems*, pages 163–201. JAI Press, 1984.
- [57] T. Başar and Y. C. Ho. Informational properties of the Nash solutions of two stochastic nonzero-sum games. J. Econ. Theory, 7(4):370–387, 1974.
- [58] T. Başar and G. J. Olsder. *Dynamic Non-cooperative Game Theory*. SIAM Series in Classics in Applied Mathematics, Society for Industrial and Applied Mathematics, 1999.
- [59] T. Başar and R. Srikant. Revenue-maximizing pricing and capacity expansion in a manyusers regime. In *Proc. IEEE Conf. on Computer Communications (INFOCOM 2002)*, pages 294–301, New York, NY, USA, June 2002.
- [60] T. Başar and R. Srikant. A Stackelberg network game with a large number of followers. J. Optimiz. Theory App., 115:479–490, 2002.
- [61] V. Bala and S. Goyal. Non-cooperative model of network formation. *Econometrica*, 68:1181–1230, 2000.
- [62] C. Barakat and E. Altman. Bandwidth tradeoff between TCP and link-level FEC. Computer Networks, 39(2):133–150, 2002.
- [63] F. Bari and V. C. M. Leung. Automated network selection in a heterogeneous wireless network environment. *IEEE Network*, 21(1):34–40, 2007.
- [64] A. Batra, J. Balakrishnan, A. Dabak, *et al.* Multi-band OFDM physical layer proposal for IEEE 802.15 task group 3a. Technical report, IEEE P802.15-03/268r3, March 2004.
- [65] M. Beckman, C. B. McGuire, and C. B. Winsten. Studies in the Economics of Transportation. Yale University Press, January 1956.
- [66] E. V. Belmega, S. Lasaulce, and M. Debbah. Power allocation games for MIMO multiple access channels with coordination. *IEEE Trans. Wireless Commun.*, 8:3182–3192, 2009.
- [67] Michel Benaïm, Josef Hofbauer, and Sylvain Sorin. Stochastic approximations and differential inclusions, part II: Applications. *Math. Oper. Res.*, 31(4):673–695, 2006.
- [68] R. Ben Ali and S. Pierre. On the impact of soft vertical handoff on optimal voice admission control in PCF-based WLANs loosely coupled to 3G networks. *IEEE Trans. Wireless Commun.*, 8(3):1356–1365, 2009.
- [69] V. Berinde. Iterative Approximation of Fixed Points, 2nd edition. Springer, 2007.
- [70] D. P. Bertsekas. Parallel and Distributed Computation: Numerical Methods. Athena Scientific, 1991.
- [71] D. P. Bertsekas and R. Gallager. Data Networks. Prentice Hall, 1992.
- [72] G. Bianchi. Performance analysis of the IEEE 802.11 distributed coordination function. *IEEE J. Sel. Areas Commun.*, 18(3):535–547, 2000.
- [73] E. W. Biersack, D. Carra, R. L. Cigno, P. Rodriguez, and P. Felber. Overlay architectures for file distribution: Fundamental performance analysis for homogeneous and heterogeneous cases. *Computer Networks*, 51(3):901–917, 2007.
- [74] BitTorrent. http://www.bitconjurer.org/BitTorrent/.
- [75] A. Bletsas, A. Lippman, and D. P. Reed. A simple distributed method for relay selection in cooperative diversity wireless networks, based on reciprocity and channel measurements. In *Proc. IEEE Vehicular Technology Conf. (VTC 2005 Spring)*, pages 1484–1488, May 2005.
- [76] H. Boche and M. Schubert. Nash bargaining and proportional fairness for wireless systems. *IEEE/ACM Trans. Networking*, 17:1453–1466, 2009.

- [77] A. Bogomonlaia and M. Jackson. The stability of hedonic coalition structures. *Games and Economic Behavior*, 38:201–230, 2002.
- [78] D. D. Botvich and N. G. Duffield. Large deviations, economies of scale, and the shape of the loss curve in large multiplexers. *Queueing Systems*, 20:293–320, 1995.
- [79] J. Y. Le Boudec and M. Vojnovic. Perfect simulation and stationarity of a class of mobility models. In *Proc. IEEE Conf. on Computer Communications (INFOCOM 2005)*, pages 2743–2754, Miami, FL, USA, August 2005.
- [80] C. Bouroghain, D. Agrawal, and S. Suri. A game theoretic framework for incentives in P2P systems. In *Proc. 3rd Int. Conf. on Peer-to-Peer Computing*, pages 48–56, Linköping, Sweden, September 2003.
- [81] S. Boyd and L. Vandenberghe. *Convex Optimization*. Cambridge University Press, September 2004.
- [82] J. Boyer, D. D. Falconer, and H. Yanikomeroglu. Cooperative connectivity models for wireless relay networks. *IEEE Trans. Wireless Commun.*, 6(6):1992–2000, 2007.
- [83] M. H. Brady, M. Mohseni, and J. M. Cioffi. Spatially-correlated jamming in Gaussian multiple access and broadcast channels. In *Proc. 40th Ann. Conf. on Information Sciences* and Systems (CISS 2006), pages 1635–1639, Princeton, NJ, March 2006.
- [84] G. W. Brown and J. von Neumann. Solutions of games by differential equations. *Contributions to the Theory of Games I.* Annals of Mathematical Studies, volume 24. Princeton University Press, 1950.
- [85] R. Bruno, M. Conti, and E. Gregori. Analytical modeling of TCP clients in Wi-Fi hot spot networks. In *Proc. Networking* 2004, pages 626–637, Athens, Greece, May 2004.
- [86] S. Buchegger and J. Y. Le Boudec. Performance analysis of the CONFIDANT protocol (cooperation of nodes-fairness in dynamic ad-hoc networks). In *Proc. ACM Int. Symp. on Mobile Ad Hoc Networking and Computing (MobiHoc 2002)*, pages 80–91, June 2002.
- [87] M. M. Buddhikot. Understanding dynamic spectrum access: Models, taxonomy and challenges. In Proc. 2nd IEEE Int. Symp. on New Frontiers in Dynamic Spectrum Access Networks (DySPAN 2007), pages 649–663, Dublin, Ireland, April 2007.
- [88] M. Cagalj, S. Ganeriwal, I. Aad, and J. P. Hubaux. On selfish behavior in CSMA/CA networks. In *Proc. IEEE Conf. on Computer Communications (INFOCOM 2005)*, pages 2513–2524, Miami, FL, USA, March 2005.
- [89] J. Cai and U. Pooch. Allocate fair payoff for cooperation in wireless ad hoc networks using Shapley value. In *Proc. Int. Parallel and Distributed Processing Symp.*, pages 219–227, Santa Fe, NM, USA, April 2004.
- [90] T. Camp, J. Boleng, and V. Davies. A survey of mobility models for ad hoc network research. Wireless Commun. and Mobile Computing (WCMC): Special issue on Mobile Ad Hoc Networking: Research, Trends and Applications, 2(5):483–502, 2002.
- [91] L. Cao and H. Zheng. Distributed spectrum allocation via local bargaining. In Proc. 2nd Ann. IEEE Communications Society Conf. on Sensor and Ad Hoc Communications and Networks (SECON 2005), pages 475–486, Santa Clara, CA, USA, December 2005.
- [92] J. Castro, D. Gomez, and J. Tejada. Polynomial calculation of the Shapley value based on sampling. *Computers and Operations Research*, 36:1726–1730, 2009.
- [93] R. Cendrillon, J. Huang, M. Chiang, and M. Moonen. Autonomous spectrum balancing for digital subscriber lines. *IEEE Trans. Signal Process.*, 55(6):4241–4257, 2005.
- [94] R. Cendrillon and M. Moonen. Iterative spectrum balancing for digital subscriber lines. IEEE Trans. Commun., 54(7):1937–1941, 2005.

- [95] R. Cendrillon, M. Moonen, J. Verlinden, T. Bostoen, and W. Yu. Optimal multiuser spectrum management for digital subscriber lines. In *Proc. IEEE Int. Conf. on Communications (ICC 2004)*, pages 1–5, Paris, France, June 2004.
- [96] J. Chakareski and P. Frossard. Delay-based overlay construction in P2P video broadcast. In Proc. IEEE Int. Conf. on Acoustics, Speech, and Signal Processing, pages 1985–1988, Taiwan, April 2009.
- [97] A. Chan and S. C. Liew. Performance of VoIP over multiple co-located IEEE 802.11 wireless LANs. *IEEE Trans. Mobile Comput.*, 8(8):1063–1076, 2009.
- [98] V. Chandrasekhar and J. G. Andrews. Uplink capacity and interference avoidance for two-tier femtocell networks. *IEEE Trans. Commun.*, 8(7):3498–3509, 2009.
- [99] V. Chandrasekhar, J. G. Andrews, and A. Gatherer. Femtocell networks: A survey. *IEEE Comm. Mag.*, 46:59–67, 2008.
- [100] V. Chandrasekhar, J. G. Andrews, Z. Shen, T. Muharemovic, and A. Gatherer. Distributed power control in femtocell-underlay cellular networks. In *Proc. IEEE Global Communications Conf. (GLOBECOM 2009)*, pages 1–6, Honolulu, HI, USA, November– December 2009.
- [101] B. Chen and P. K. Varshney. A Bayesian sampling approach to decision fusion using hierarchical models. *IEEE Trans. Signal Process.*, 50(8):1809–1818, 2002.
- [102] L. Chen and J. Leneutre. Selfishness, not always a nightmare: Modeling selfish MAC behaviors in wireless mobile ad hoc networks. In *Proc. 27th IEEE Int. Conf. on Distributed Computing Systems (ICDCS)*, pages 1809–1818, Toronto, Ontario, Canada, June 2007.
- [103] L. Chen, S. H. Low, and J. C. Doyle. Contention control: A game-theoretic approach. In Proc. 46th IEEE Conf. on Decision and Control (CDC 2007), pages 3428–3434, New Orleans, LA, USA, December 2007.
- [104] Z. Chen, A. Vorobyov, C.-X. Wang, and J. S. Thompson. Nash bargaining over MIMO interference systems. In *Proc. IEEE Int. Conf. on Communications (ICC 2009)*, pages 1–5, Dresden, Germany, June 2009.
- [105] C.-C. Chiang, H.-K. Wu, W. Liu, and M. Gerla. Routing in clustered multi-hop, mobile wireless networks with fading channel. In *Proc. IEEE Singapore Int. Conf. on Networks* (SICON 1997), pages 197–211, Singapore, April 1997.
- [106] M. Chiang, P. Hande, T. Lan, and C. W. Tan. Power control in wireless cellular networks. *Foundations and Trends in Networking*, 2(4):381–533, 2008.
- [107] Y. Cho, C. Hwang, and F. A. Tobagi. Design of robust random access protocols for wireless networks using game theoretic models. In *Proc. IEEE Conf. on Computer Communications* (*INFOCOM 2008*), pages 1750–1758, Phoenix, AZ, USA, April 2008.
- [108] Y. Cho and F. A. Tobagi. Cooperative and non-cooperative aloha games with channel capture. In *Proc. IEEE Global Communications Conf. (GLOBECOM 2008)*, pages 1–6, New Orleans, LA, USA, November–December 2008.
- [109] H. H. Choi, O. Song, Y. K. Park, and J. R. Lee. Performance evaluation of opportunistic vertical handover consider on-off characteristics of VoIP traffic. *IEEE Trans. Veh. Technol.*, 59(6):3115–3121, 2010.
- [110] J. M. Cioffi, W. Rhee, M. Mohseni, and M. H. Brady. Band preference in dynamic spectrum management. In *Proc. 12th Eur. Signal Process. Conf. (Eusipco 2004)*, Vienna, Austria, September 2004.
- [111] R. Clayton. Fundamentals of Electric Circuit Analysis. John Wiley and Sons, 2001.
- [112] B. Cohen. Incentives build robustness in BitTorrent. In Proc. 1st Workshop on Economics of Peer-to-Peer Systems, Berkeley, CA, USA, June 2003.

- [113] V. Conitzer and T. Sandholm. Complexity of determining nonemptiness of the core. Technical Report CS-02-137, CMU, 2002. http://arxiv.org/abs/cs/0307016.
- [114] P. Coucheney, C. Touati, and B. Gaujal. Fair and efficient user-network association algorithm for multi-technology wireless networks. In *Proc. IEEE Conf. on Computer Communications* (*INFOCOM 2009*), pages 2811–2815, Rio de Janeiro, Brazil, April 2009.
- [115] T. M. Cover and El Gamal. Capacity theorems for the relay channel. *IEEE Trans. Inf. Theory*, 25(5):572–584, 1979.
- [116] T. M. Cover and J. Thomas. Elements of Information Theory. John Wiley and Sons, 1991.
- [117] R. J. Crowcroft. Modelling incentives for collaboration in mobile ad hoc networks. *Performance Evaluation*, 57(4):427–439, 2004.
- [118] I. Csiszár and J. Körner. Broadcast channels with confidential messages. *IEEE Trans. Inf. Theory*, 24(3):339–348, 1978.
- [119] T. Cui, L. Chen, and S. Low. A game-theoretic framework for medium access control. *IEEE J. Sel. Areas in Commun.*, 26(7):1116–1127, 2008.
- [120] Z. Dai, R. Fracchia, J. Gosteau, P. Pellati, and G. Vivier. Vertical handover criteria and algorithm in IEEE 802.11 and 802.16 hybrid networks. In *Proc. IEEE Int. Conf. on Communications (ICC 2008)*, pages 2480–2484, Beijing, China, May 2008.
- [121] A. Das and R. Srikant. Diffusion approximations for a single node accessed by congestion controlled sources. *IEEE Trans. Autom. Control*, 45:1783–1799, 2000.
- [122] A. P. C. da Silva, M. Varela, E. de Souza e Silva, R. M. M. Leão, and G. Rubino. Quality assessment of interactive voice applications. *Comput. Netw.*, 52(6):1179–1192, 2008.
- [123] C. d'Aspremont and L. Gerard-Varet. Incentives and incomplete information. J. Public Econ., 29(45):11–25, 1979.
- [124] A. de Baynast, O. Gurewitz, and E. W. Knightly. Cooperative strategies and optimal scheduling for tree networks. In *Proc. IEEE Conf. on Computer Communications (INFOCOM* 2007), pages 1857–1865, Anchorage, AK, USA, May 2007.
- [125] G. Debreu. A social equilibrium theorem. Proc. Nat. Acad. Sci. USA, 38:386–393, 1952.
- [126] S. Dehnie and N. Memon. A dynamic game model for amplify-and-forward cooperative Communications. In Proc. IEEE Int. Conf. on Acoustics, Speech, and Signal Processing (ICASSP 2009), pages 3485–3488, Taipei, Taiwan, April 2009.
- [127] G. Demange and M. Wooders, eds. Group Formation in Economics: Networks, Clubs, and Coalitions. Cambridge University Press, 2006.
- [128] J. Derks and J. Kuipers. Implementing the simplex method for computing the prenucleolus of transferable utility games. Technical report, University of Maastricht, Netherlands, 1997. http://www.personeel.unimaas.nl/Jean-Derks/ downlds/derimp.pdf.
- [129] J. Derks, J. Kuipers, M. Tennekes, and F. Thuijsman. Local dynamics in network formation. In Proc. 3rd World Congress of the Game Theory Society, Evanston, IL, USA, July 2008.
- [130] E. de Souza e Silva, D. R. Figueiredo, and R. M. M. Leão. The TANGRAMII integrated modeling environment for computer systems and networks. ACM SIGMETRICS Performance Evaluation Review, 36(4):64–69, 2009.
- [131] N. Dimitriou, P. Mertikopoulos, and A. L. Moustakas. Vertical handover between wireless standards. In *Proc. IEEE Int. Conf. on Communications (ICC 2008)*, pages 3269–3273, Beijing, China, May 2008.
- [132] E. J. Dockner, S. Jörgensen, N. V. Long, and G. Sorger. *Differential Games in Economics and Management Science*. Cambridge University Press, 2000.

- [133] L. Dong, Z. Han, A. P. Petropulu, and H. V. Poor. Secure collaborative beamforming. In Proc. 46th Ann. Allerton Conf. on Communication, Control, and Computing, Monticello, IL, USA, September 2008.
- [134] M. Dramitinos, G. Stamoulis, and C. Courcoubetis. Auction-based resource reservation in 2.5/3G networks. *Mobile Networks and Applications*, 9(6):557–566, 2004.
- [135] R. Dube, C. D. Rais, K. Y. Wang, and S. K. Tripathi. Signal stability-based adaptive routing (SSA) for ad-hoc mobile networks. *IEEE Personal Communications Magazine*, 4(2):36–45, 1997.
- [136] D. Easley and J. Kleinberg. Networks. Cornell Store Press, 2009.
- [137] S. Eidenbenz, G. Resta, and P. Santi. The commit protocol for truthful and cost-efficient routing in ad hoc networks with selfish nodes. *IEEE Trans. Mobile Comput.*, 7(1):19–33, 2008.
- [138] O. Ercetin. Association games in IEEE 802.11 wireless local area networks. *IEEE Trans. Wireless Commun.*, 7(12):5136–5143, 2008.
- [139] R. Etkin, A. Parekh, and D. Tse. Spectrum sharing for unlicensed bands. *IEEE J. Sel. Areas Commun.*, 25(3):517–528, 2007.
- [140] B. G. Evans and K. Baughan. Visions of 4G. J. of Electronics and Commun. Eng., 12(6): 293–303, 2000.
- [141] K. Fan. Fixed-point and minimax theorems in locally convex topological linear spaces. *Proc. Nat. Acad. Sci. USA*, 38:121–126, 1952.
- [142] A. Fattahi and F. Paganini. New economic perspectives for resource allocation in wireless networks. In Proc. Am. Control Conf., Portland, OR, USA, June 2005.
- [143] FCC. Report of the spectrum efficiency group. Technical report, FCC Spectrum Policy Task Force, Federal Communications Commission, November 2002. http://www.fcc. gov/sptf/files/SEWGFinalReport_1.pdf.
- [144] G. Federico and D. Rahman. Bidding in an electricity pay-as-bid auction. J. Regul. Econ., 24(2):175–211, 2003.
- [145] D. Feeny, F. Berkes, B. J. McCay, and J. M. Acheson. The tragedy of the commons: Twenty-two years later. *Human Ecology*, 18(1):1–19, 1990.
- [146] M. Feldman, K. Lai, I. Stoica, and J. Chuang. Robust incentive techniques for Peer-to-Peer networks. In Proc. 5th ACM Conf. on Electronic Commerce (EC 2004), pages 102–111, New York, NY, USA, 2004.
- [147] M. Felegyhazi, L. Buttyan, and J. P. Hubaux. Equilibrium analysis of packet forwarding stratiegies in wireless ad-hoc networks: The static case. In *Personal Wireless Communications 2003*, pages 23–25, Venice, Italy, 2003.
- [148] M. Felegyhazi, M. Cagalj, and J. P. Hubaux. Efficient MAC in cognitive radio systems: A game-theoretic approach. *IEEE Trans. Wireless Commun.*, 8(4):1984–1995, 2009.
- [149] T. Feng and Y. Zhen. A new algorithm for weighted proportional fairness based spectrum allocation of cognitive radios. In Proc. 8th ACIS Int. Conf. on Software Engineering, Artificial Intelligence, Networking, and Parallel/Distributed Computing (SNPD 2007), pages 531–536, Qingdao, China, July 2007.
- [150] J. Filar and K. Vrieze. Competitive Markov Decision Processes. Springer 1997.
- [151] S. Fischer, L. Olbrich, and B. Vocking. Approximating Wardrop equilibria with finitely many agents. *Springer Distributed Computing*, 21(2):129–139, 2008.
- [152] R. A. Fisher. The Genetical Theory of Natural Selection. The Clarendon Press, 1930.
- [153] S. D. Flam. Equilibrium, evolutionary stability and gradient dynamics. Int. Game Theory Rev., 4(4):357–370, 2002.

- [154] F. Forgo, J. Szep, and F. Szidarovszky. Introduction to the Theory of Games: Concepts, Methods, Applications. Springer, 1999.
- [155] S. Frattasi, H. Fathi, F. H. P. Prasad, and R. Katz. Defining 4G technology from the users perspective. *IEEE Network*, 20(1):35–41, 2006.
- [156] D. Friedman, D. P. Friedman, and J. Rust, eds. *The Double Auction Market: Institutions, Theories, and Evidence.* Westview Press, 1993.
- [157] F. Fu, A. R. Fattahi, and M. van der Schaar. Game-theoretic paradigm for resource management in spectrum agile wireless networks. In *Proc. IEEE Int. Conf. on Multimedia and Exposition*, pages 873–876, Toronto, Ontario, Canada, July 2006.
- [158] F. Fu and M. van der Schaar. Learning to compete for resources in wireless stochastic games. *IEEE Trans. Veh. Technol.*, 58(4):1904–1919, 2009.
- [159] D. Fudenberg and D. Levine. The Theory of Learning in Games. MIT Press, 1998.
- [160] D. Fudenberg and J. Tirole. Game Theory. MIT Press, 1991.
- [161] S. Gandhi, C. Buragohain, L. Cao, H. Zheng, and S. Suri. A general framework for wireless spectrum auctions. In *Proc. 2nd IEEE Int. Symp. on New Frontiers in Dynamic Spectrum Access Networks (DySPAN 2007)*, pages 22–33, Dublin, Ireland, April 2007.
- [162] R. Garg, A. Kamra, and V. Khurana. A game-theoretic approach towards congestion control in communication networks. ACM SIGCOMM Comput. Commun. Rev., 32(3):47–61, 2002.
- [163] N. Gatsis, A. G. Marques, and G. B. Giannakis. Utility-based power control for peer-to-peer cognitive radio networks with heterogeneous QoS constraints. In *Proc. IEEE Int. Conf. on Acoustics, Speech, and Signal Processing (ICASSP 2008)*, pages 2805–2808, Las Vegas, NV, USA, March–April 2008.
- [164] A. Ghasemi and E. S. Sousa. Collaborative spectrum sensing for opportunistic access in fading environments. In Proc. 1st IEEE Int. Symp. on New Frontiers in Dynamic Spectrum Access Networks (DySPAN 2005), pages 131–136, Baltimore, MD, USA, November 2005.
- [165] A. Gibbard. Manipulation of voting schemes: A general result. *Econometrica*, 41(4):587–601, 1973.
- [166] R. Gibbons. *Game Theory for Applied Economists*. Princeton University Press, July 1992.
- [167] I. L. Glicksberg. A further generalization of the Kakutani fixed point theorem, with application to Nash equilibrium points. *Proc. Am. Math. Soc.*, 3(1):170–174, 1952.
- [168] Gnutella Protocol Specification. 2000. http://cryptnet.net/fsp/cpcd/ gnutella_protocol_0.4.pdf.
- [169] K. Goebel and W. A. Kirk. *Topics in Metric Fixed Point Theory*. Cambridge University Press, 1990.
- [170] D. E. Goldberg. Genetic Algorithms in Search, Optimization, and Machine Learning. Addison-Wesley Professional, 1989.
- [171] D. Goodman and N. Mandayam. Power control for wireless data. IEEE J. Sel. Areas Commun., 7:48–54, 2000.
- [172] P. K. Gopala, L. Lai, and H. El Gamal. On the secrecy capacity of fading channels. *IEEE Trans. Inf. Theory*, 54(10):4687–4698, 2008.
- [173] D. Grandblaise, C. Kloeck, T. Renk, et al. Microeconomics inspired mechanisms to manage dynamic spectrum allocation. In Proc. 2nd IEEE Int. Symp. on New Frontiers in Dynamic Spectrum Access Networks (DySPAN 2007), pages 452–461, Dublin, Ireland, April 2007.
- [174] J. Green and J. J. Laffont. On coalition incentive compatibility. *Rev. Econ. Stud.*, 46(2): 243–254, 1979.
- [175] T. Groves. Incentives in teams. *Econometrica*, 45:617–631, 1973.

- [176] D. Gu and J. Zhang. QoS enhancement in IEEE 802.11 wireless local area networks. *IEEE Comm. Mag.*, 41(6):120–124, 2003.
- [177] D. Gunduz and E. Erkip. Joint source-channel cooperation: Diversity versus spectral efficiency. In Proc. IEEE Int. Symp. on Information Theory, Chicago, IL, USA, June–July 2004.
- [178] S. Guruacharya, D. Niyato, E. Hossain, and D. I. Kim. Hierarchical competition in femtocellbased cellular networks. In *Proc. IEEE Global Communications Conf. (GLOBECOM 2010)*, Miami, FL, USA, December 2010.
- [179] K. Han and S. Choi. Performance analysis of sleep mode operation in IEEE 802.16e mobile broadband wireless access systems. In *Proc. IEEE Vehicular Technology Conf. (VTC 2006 Spring)*, pages 1141–1145, Melbourne, Australia, May 2006.
- [180] T. Han and K. Kobayashi. A new achievable rate region for the interference channel. *IEEE J. Select. Areas Commun.*, 27:49–60, 1981.
- [181] Z. Han, T. Himsoon, W. Siriwongpairat, and K. J. R. Liu. Energy efficient cooperative transmission over multiuser OFDM networks: Who helps whom and how to cooperate. In *Proc. IEEE Wireless Commun. and Networking Conf. (WCNC 2005)*, pages 1030–1035, New Orleans, LA, USA, March 2005.
- [182] Z. Han, Z. Ji, and K. J. R. Liu. Dynamic distributed rate control for wireless networks by optimal cartel maintenance strategy. In *Proc. IEEE Global Communications Conf.* (*GLOBECOM 2004*), Dallas, TX, USA, November 2004.
- [183] Z. Han, Z. Ji, and K. J. R. Liu. Fair multiuser channel allocation for OFDMA networks using Nash bargaining and coalitions. *IEEE Trans. Commun.*, 53(8):1366–1376, 2005.
- [184] Z. Han, Z. Ji, and K. J. R. Liu. Non-cooperative resource competition game by virtual referece in multi-cell OFDMA networks. *IEEE J. Select. Areas Commun.*, 53(10):1079–1090, 2007.
- [185] Z. Han, Z. Ji, and K. J. R. Liu. A cartel maintenance framework to enforce cooperation in wireless networks with selfish users. *IEEE Trans. Wireless Commun.*, 7(5):1889–1899, 2008.
- [186] Z. Han and K. J. R. Liu. Non-cooperative power-control game and throughput game over wireless networks. *IEEE Trans. Commun.*, 53(10):1625–1629, 2005.
- [187] Z. Han and K. J. R. Liu. Resource Allocation for Wireless Networks: Basics, Techniques, and Applications. Cambridge University Press, 2008.
- [188] Z. Han, C. Pandana, and K. J. R. Liu. A self-learning repeated game framework for optimizing packet forwarding networks. In *Proc. IEEE Wireless Communications and Networking Conf. (WCNC 2005)*, pages 2131–2136, New Orleans, LA, USA, March 2005.
- [189] Z. Han and H. V. Poor. Coalition game with cooperative transmission: A cure for the curse of boundary nodes in selfish packet-forwarding wireless networks. In *Proc. 5th Int. Symp. on Modeling and Optimization in Mobile, Ad Hoc, and Wireless Networks (WiOpt 2007)*, Limassol, Cyprus, April 2007.
- [190] Z. Han and H. V. Poor. Lifetime improvement in wireless sensor networks via collaborative beamforming and cooperative transmission. *IEE Proceedings: Microwaves, Antennas* and Propagation: Special Issue on Antenna Systems and Propagation for Future Wireless Commun., December 2007.
- [191] Z. Han and H. V. Poor. Coalition games with cooperative transmission: A cure for the curse of boundary nodes in selfish packet-forwarding wireless networks. *IEEE Trans. Commun.*, 57:203–213, 2009.
- [192] Z. Han, X. Zhang, and H. V. Poor. Cooperative transmission protocols with high spectral efficiency and high diversity order using multiuser detection and network coding. In *Proc.*

IEEE Int. Conf. on Communications (ICC 2007), pages 4232–4236, Glasgow, UK, June 2007.

- [193] P. Hande, S. Rangan, M. Chiang, and X. Wu. Distributed uplink power control for optimal SIR assignment in cellular data networks. *IEEE/ACM Trans. Networking*, 16(6):1420–1433, 2008.
- [194] G. Hardin. The tragedy of the commons. Science, 162(3859):1243–1248, 1968.
- [195] J. C. Harsanyi. Games with incomplete information played by "Bayesian" players, part ii. Bayesian equilibrium points. *Management Science*, 14(5):320–334, 1968.
- [196] S. Hart and A. Mas-Colell. A simple adaptive procedure leading to correlated equilibrium. *Econometrica*, 68(5):1127–1150, 2000.
- [197] S. Hart and A. Mas-Colell. A reinforcement procedure leading to correlated equilibrium. In *Economic Essays*, pages 181–200. Hebrew University of Jerusalem, 2001.
- [198] S. Hart and A. Mas-Colell. Uncoupled dynamics do not lead to Nash equilibrium. Am. Econ. Rev., 93(5):1830–1836, 2003.
- [199] D. N. Hatfield and P. J. Weiser. Property rights in spectrum: Taking the next step. In Proc. Ist IEEE Int. Symp. on New Frontiers in Dynamic Spectrum Access Networks (DySPAN 2005), pages 43–55, Baltimore, MD, USA, November 2005.
- [200] A. Haurie and P. Marcotte. On the relationship between Nash-Cournot and Wardrop equilibria. *Networks*, 15:295–308, 1985.
- [201] S. Haykin. Cognitive radio: Brain-empowered wireless communications. IEEE J. Sel. Areas Commun., 23(2):201–220, 2005.
- [202] G. He, M. Debbah, and E. Altman. K-player Bayesian waterfilling game for fading multiple access channels. In *IEEE 3rd Int. Workshop on Computational Advances in Multi-Sensor Adaptive Processing*, pages 17–20, Aruba, Dutch Antilles, 2009.
- [203] L. He and J. Walrand. Pricing and revenue sharing strategies for Internet service providers. *IEEE J. Select. Areas Commun.*, 24(5):942–951, 2006.
- [204] X. He, A. Prasad, S. P. Sethi, and G. J. Gutierrez. A survey of Stackelberg differential game models in supply and marketing channels. J. Syst. Sci. Syst. Eng., 16(4):385–413, 2007.
- [205] P. Herings, G. van der Laan, and D. Talman. *Cooperative Games in Graph Structure*. Research Memoranda, Maastricht Research School in Economics of Technology and Organization, 11, August 2002.
- [206] A. O. Hero. Secure space-time communication. *IEEE Trans. Inf. Theory*, 49(12):3235–3249, 2003.
- [207] M. Heusse, F. Rousseau, G. Berger-Sabbatel, and A. Duda. Performance anomaly of 802.11b. In Proc. IEEE Conf. on Computer Communications (INFOCOM 2003), pages 836–843, San Francisco, CA, USA, March–April 2003.
- [208] M. Heusse, F. Rousseau, R. Guillier, and A. Duda. Idle sense: An optimal access method for high throughput and fairness in rate diverse wireless LANs. In ACM SIGCOMM Comput. Commun. Rev., 35:121–132, 2005.
- [209] J. Hicks and A. B. MacKenzie. A convergence result for potential games. In Proc. 11th Int. Symp. on Dynamic Games and Applications, Tucson, AZ, USA, December 2004.
- [210] F. S. Hillier and G. J. Lieberman. Introduction to Operations Research. McGraw-Hill Science/Engineering/Math, 2002.
- [211] J. Hofbauer and K. Sigmund. *Evolutionary Games and Population Dynamics*. Cambridge University Press, 1998.
- [212] A. Hortacsu. Mechanism Choice and Strategic Bidding in Divisible Good Auctions: An Empirical Analysis of the Turkish Treasury Auction Market. Technical report, Stanford University, 2000. http://home.uchicago.edu/~hortacsu/ttreas.pdf.

- [213] A. Høst-Madsen. A new achievable rate for cooperative diversity based on generalized writing on dirty paper. In *Proc. IEEE Int. Symp. Information Theory*, Yokohama, Japan, June 2003.
- [214] A. Høst-Madsen. Upper and lower bounds for channel capacity of asynchronous cooperative diversity networks. *IEEE Trans. Inf. Theory*, 50(4):3062–3080, 2004.
- [215] H. Hotelling. Stability in competition. Econ. J., 39(153):41-57, 1929.
- [216] J. Hou, J. Yang, and S. Papavassiliou. Integration of pricing with call admission control to meet QoS requirements in cellular networks. *IEEE Trans. Parallel Distrib. Syst.*, 13(9):898–910, 2002.
- [217] J. Huang, R. Berry, and M. L. Honig. Auction-based spectrum sharing. *Mobile Networks and Applications*, 11(3):405–418, 2006.
- [218] J. Huang, Z. Han, M. Chiang, and H. V. Poor. Auction-based resource allocation for multirelay asynchronous cooperative networks. In *Proc. IEEE Int. Conf. on Acoustics, Speech, and Signal Processing (ICASSP 2008)*, pages 5356–5359, Las Vegas, NV, USA., March 2008.
- [219] T. E. Hunter and A. Nosratinia. Performance analysis of coded cooperation diversity. In Proc. IEEE Int. Conf. on Communications (ICC 2003), pages 2688–2692, Anchorage, AK, USA, May 2003.
- [220] T. E. Hunter and A. Nosratinia. Outage performance of cognitive wireless relay networks. In Proc. IEEE Global Communications Conf. (GLOBECOM 2006), pages 1–5, San Francisco, CA, USA, November–December 2006.
- [221] T. E. Hunter, S. Sanayei, and A. Nosratinia. Outage analysis of coded cooperation. *IEEE Trans. Inf. Theory*, 52(2):375–391, 2006.
- [222] L. Hurwicz. Decision and Organization: On Informationally Decentralized Systems, 2nd edition. University of Minnesota Press, 1972.
- [223] C. Hwang, K. Seong, and J. M. Cioffi. Opportunistic p-persistent CSMA in wireless networks. In *Proc. IEEE Int. Conf. on Communications (ICC 2006)*, pages 183–188, Istanbul, Turkey, June 2006.
- [224] A. Ibing and H. Boche. Fairness vs. efficiency: Comparison of game theoretic criteria for OFDMA scheduling. In Proc. 41st Asilomar Conf. on Signals, Systems, and Computers, pages 275–279, Pacific Grove, CA, USA, November 2007.
- [225] A. S. Ibrahim, Z. Han, and K. J. R. Liu. Distributed energy-efficient cooperative routing in wireless networks. In *Proc. IEEE Global Communications Conf. (GLOBECOM 2007)*, pages 4413–4418, Washington, DC, USA, November 2007.
- [226] O. Ileri, D. Samardzija, and N. B. Mandayam. Demand responsive pricing and competitive spectrum allocation via a spectrum server. In *Proc. 1st IEEE Int. Symp. on New Frontiers in Dynamic Spectrum Access Networks (DySPAN 2005)*, pages 194–202, Baltimore, MD, USA, November 2005.
- [227] H. Inaltekin and S. B. Wicker. The analysis of Nash equilibria of the one-shot randomaccess game for wireless networks and the behavior of selfish nodes. *IEEE/ACM Trans. Networking*, 16(5):1094–1107, 2008.
- [228] M. O. Jackson. A Survey of Models of Network Formation: Stability and Efficiency. Working paper 1161, California Institute of Technology, Division of the Humanities and Social Sciences, November 2003.
- [229] M. O. Jackson and A. Wolinsky. A strategic model of social and economic networks. J. Econ. Theory, 71:44–74, 1996.

- [230] K. Jain, J. Padhye, V. Padmanabhan, and L. Qiu. Impact of interference on multi-hop wireless network performance. In *Proc. 9th Ann. Int. Conf. on Mobile Computing and Networking* (*MobiCom 2003*), pages 66–80, San Diego, CA, USA, September 2003.
- [231] R. Jain and J. Walrand. An efficient mechanism for network bandwidth auction. In Proc. IEEE Network Operations and Management Symp. Workshops (NOMS Workshops 2008), pages 227–234, Salvador da Bahia, Brazil, April 2008.
- [232] Z. Ji and K. J. R. Liu. Belief-assisted pricing for dynamic spectrum allocation in wireless networks with selfish users. In Proc. 3rd Ann. IEEE Communications Society Conf. on Sensor, Mash and Ad Hoc Commun. and Networks (SECON 2006), pages 119–127, Reston, VA, USA, September 2006.
- [233] N. Jindal, J. Andrews, and S. Weber. Bandwidth partitioning in decentralized wireless networks. *IEEE Trans. Wireless Commun.*, 7(12):5408–5419, 2008.
- [234] R. Johari, S. Mannor, and J. Tsitsiklis. A contract based model for directed network formation. *Games and Economic Behavior*, 56:201–224, 2006.
- [235] R. Johari and J. N. Tsitsiklis. Efficiency loss in a network resource allocation game. Math. Oper. Res., 29(3):407–435, 2004.
- [236] D. B. Johnson. Routing in ad hoc networks of mobile hosts. In Proc. 1st IEEE Workshop on Mobile Computing Systems and Applications, pages 158–163, Santa Cruz, CA, USA, December 1994.
- [237] E. Kalai and M. Smorodinsky. Other solutions to Nash's bargaining problem. *Econometrica*, 43:513–518, 1975.
- [238] V. Kamble, E. Altman, R. El-Azouzi, and V. Sharma. A theoretical framework for hierarchical routing games. In *Proc. IEEE Conf. on Computer Communications (INFOCOM 2010)*, pages 1–5, San Diego, CA, USA, March 2010.
- [239] G. S. Kasbekar, E. Altman, and S. Sarkar. A hierarchical spatial game over licenced resources. In Proc. Int. Conf. on Game Theory for Networks (GameNets 2009), pages 70–79, Istanbul, Turkey, May 2009.
- [240] A. Kashyap, T. Başar, and R. Srikant. Correlated jamming on MIMO Gaussian fading channels. *IEEE Trans. Inf. Theory*, 50(9):2119–2123, 2004.
- [241] KaZaa media desktop. http://www.kazaa.com.
- [242] F. P. Kelly. Charging and rate control for elastic traffic. *Euro. Trans. Telecommun.*, 8:33–37, 1997.
- [243] F. P. Kelly. Models for a self-managed Internet. *Phil. Trans. Roy. Soc. A*, 358:2335–2348, 2000.
- [244] F. P. Kelly, A. K. Maulloo, and D. K. H. Tan. Rate control in communication networks: Shadow prices, proportional fairness and stability. J. Oper. Res. Soc., 49:237–252, 1998.
- [245] K. J. Kerpez, D. L. Waring, S. Galli, J. Dixon, and P. Madon. Advanced DSL management. *IEEE Comm. Mag.*, 41(9):116–123, 2003.
- [246] A. H. Khan, M. A. Qadeer, J. A. Ansari, and S. Waheed. 4G as next generation wireless network. In *Proc. Int. Conf. on Future Computer and Communication*, pages 334–338, Kuala Lumpur, Malaysia, April 2009.
- [247] M. A. Khan, A. C. Toker, C. Troung, F. Sivrikaya, and S. Albayrak. Cooperative game theoretic approach to integrated bandwidth sharing and allocation. In *Proc. Int. Conf. on Game Theory for Networks (GameNets 2009)*, pages 1–9, Istanbul, Turkey, May 2009.
- [248] A. E. Khandani, E. Modiano, L. Zheng, and J. Abounadi, eds. *Advances in Pervasive Computing and Networking*. Kluwer Academic Publishers, 2004.

- [249] M. A. Khojastepour, A. Sabharwal, and B. Aazhang. On the capacity of "cheap" relay networks. In *Proc. Conf. on Information Sciences and Systems (CISS 2003)*, pages 12–14, Baltimore, MD, USA, 2003.
- [250] R. Y. Kim, J. S. Kwak, and K. Etemad. WiMAX femtocell: Requirements, challenges, and solutions. *IEEE Comm. Mag.*, 47(9):84–91, 2009.
- [251] C. Kloeck, H. Jaekel, and F. K. Jondral. Dynamic and local combined pricing, allocation and billing system with cognitive radios. In *Proc. 1st IEEE Int. Symp. on New Frontiers in Dynamic Spectrum Access Networks (DySPAN 2005)*, pages 73–81, Baltimore, MD, USA, November 2005.
- [252] Z. Kong and Y. Kwok. Relay auction algorithms for downlink bandwidth allocation in IEEE 802.16-based OFDM/TDMA wireless mesh networks. In Proc. 4th Int. Conf. on Communications and Networking in China (ChinaCom 2009), pages 1–5, Xi'an, China, August 2009.
- [253] J. Konorski. A game-theoretic study of CSMA/CA under a backoff attack. *IEEE/ACM Trans. Networking*, 14(6):1167–1178, 2006.
- [254] S. Koskie and Z. Gajic. A Nash game algorithm for SIR-based power control in 3G wireless CDMA networks. *IEEE/ACM Trans. Networking*, 13(5):1017–1026, 2005.
- [255] L. Kovács and A. Vidács. One-shot multi-bid auction method in dynamic spectrum allocation networks. In *Proc. Ist Mobile and Wireless Communications Summit*, pages 1–5, Budapest, Hungary, July 2007.
- [256] V. Krishna. Auction Theory, 2nd edition. Academic Press, 2010.
- [257] B. Krishnamachari. Networking Wireless Sensors. Cambridge University Press, 2005.
- [258] H. W. Kuhn. Extensive games and the problem of information. *Contributions to the Theory of Games*. Annals of Mathematical Studies, volume 28. Princeton University Press, 1953.
- [259] H. W. Kuhn. The Hungarian method for the assignment problem. Naval Research Logistics Quarterly, 2:83–97, 1955.
- [260] A. Kuhne and A. Klein. Throughput analysis of multi-user OFDMA systems using imperfect CQI feedback and diversity techniques. *IEEE J. Select. Areas Commun.*, 26(8):1440–1450, 2008.
- [261] A. Kumar, E. Altman, D. Miorandi, and M. Goyal. New insights from a fixed point analysis of single cell IEEE 802.11 WLANs. In *Proc. IEEE Conf. on Computer Communications* (*INFOCOM 2005*), pages 1550–1561, Miami, FL, USA, March 2005.
- [262] D. Kumar, E. Altman, and J. M. Kelif. User-network association in an 802.11 WLAN and 3G UMTS hybrid cell: Individual optimality. In *Proc. IEEE Sarnoff Symp.*, pages 1–6, Princeton, NJ, USA, April–May 2007.
- [263] Z. Kun, D. Niyato, and P. Wang. Network selection in heterogeneous wireless networks: Evolution with incomplete information. In *Proc. IEEE Wireless Communications and Networking Conf. (WCNC 2010)*, pages 1–6, Sydney, Australia, April 2010.
- [264] Z. Kun, D. Niyato, and P. Wang. Optimal bandwidth allocation with dynamic service selection in heterogeneous wireless networks. In *Proc. IEEE Global Communications Conf.* (GLOBECOM 2010), Miami, FL, USA, December 2010.
- [265] S. Kunniyur and R. Srikant. Analysis and design of an adaptive virtual queue algorithm for active queue management. In *Proc. ACM Conference on Applications, Technologies, Architectures, and Protocols for Computer Communication (SIGCOMM 2001)*, pages 123–134, San Diego, CA, USA, August 2001.
- [266] S. Kunniyur and R. Srikant. An adaptive virtual queue algorithm for active queue management. *IEEE/ACM Trans. Netw.*, 12(2):286–299, 2004.

- [267] Y. Kuo, C. Lu, E. H. K. Wu, and G. Chen. An admission control strategy for differentiated services in IEEE 802.11. In *Proc. IEEE Global Communications Conf. (GLOBECOM* 2003), pages 707–712, San Francisco, CA, USA, December 2003.
- [268] Y. Kuo, E. H. K. Wu, and G. Chen. Non-cooperative admission control for differentiated services in IEEE 802.11 WLANs. In *Proc. IEEE Global Communications Conf. (GLOBECOM 2004)*, pages 2981–2986, Dallas, TX, USA, November–December 2004.
- [269] G. Kuriakose, S. Harsha, A. Kumar, and V. Sharma. Analytical models for capacity estimation of IEEE 802.11 WLANs using DCF for Internet applications. *Wireless Networks*, 15(2):259–277, 2009.
- [270] H. J. Kushner and G. G. Yin. *Stochastic Approximation and Recursive Algorithms and Applications*. Springer, July 2003.
- [271] A. Kwasinski, Z. Han, and K. J. R. Liu. Cooperative multimedia Communications: Joint source coding and collaboration. In *Proc. IEEE Global Communications Conf.* (*GLOBECOM 2005*), page 5, St. Louis, MO, USA, November 2005.
- [272] P. Kyasanur and N. H. Vaidya. Selfish MAC layer misbehavior in wireless networks. *IEEE Trans. Mobile Comput.*, 4(5):502–516, 2005.
- [273] R. J. La and V. Anantharam. Utility-based rate control in the Internet for elastic traffic. *IEEE/ACM Trans. Networking*, 10(2):272–286, 2002.
- [274] R. J. La and V. Anantharam. Optimal routing control: Repeated game approach. *IEEE Trans. Autom. Control*, 47(3):437–450, 2002.
- [275] R. J. La and V. Anantharam. A game-theoretic look at the Gaussian multiaccess channel. In Proc. DIMACS Workshop on Network Information Theory, pages 87–106, Piscataway, NJ, USA, March 2003.
- [276] L. Lai and H. El Gamal. Fading multiple access channels: A game theoretic perspective. In Proc. IEEE Int. Symp. on Information Theory, pages 1334–1338, July 2006.
- [277] L. Lai and H. El Gamal. The water-filling game in fading multiple-access channel. *IEEE Trans. Inf. Theory*, 54(5):2110–2122, 2008.
- [278] L. Lai and H. El Gamal. The relay-eavesdropper channel: Cooperation for secrecy. *IEEE Trans. Inf. Theory*, 54(9):4005–4019, 2008.
- [279] J. Laneman, D. Tse, and G. Wornell. Cooperative diversity in wireless networks: Efficient protocols and outage behavior. *IEEE Trans. Inf. Theory*, 50(12):3062–3080, 2004.
- [280] J. N. Laneman and G. W. Wornell. Distributed space-time-coded protocols for exploiting cooperative diversity in wireless networks. *IEEE Trans. Inf. Theory*, 49(10):2415–2425, 2003.
- [281] E. Larsson and E. Jorswieck. Competition versus cooperation on the MISO interference channel. *IEEE J. Select. Areas Commun.*, 26:1059–1069, 2008.
- [282] E. Larsson, E. Jorswieck, J. Lindblom, and R. Mochaourab. Game theory and the flat-fading Gaussian interference channel. *IEEE Signal Processing Magazine: Special issue on Game Theory in Signal Processing and Communications*, 26:18–27, 2009.
- [283] S. Lasaulce, M. Debbah, and E. Altman. Methodologies for analyzing equilibria in wireless games. *IEEE Signal Processing Magazine: Special issue on Game Theory in Signal Processing and Communications*, 26:41–52, 2009.
- [284] H. Lee, H. Kwon, A. Motskin, and L. Guibas. Interference-aware MAC protocol for wireless networks by a game-theoretic approach. In *Proc. IEEE Conf. on Computer Communications* (*INFOCOM 2009*), pages 1854–1862, Rio de Janeiro, Brazil, April 2009.
- [285] H. Lee, H. Park, Y. Choi, Y. Chung, and S. Rhee. Link adaptive multi-hop path management for IEEE 802.16j. In *IEEE C802/16j-07/1053*, January 2007.

- [286] J. Lee, W. Liao, J. Chen, and H. Lee. A practical QoS solution to voice over IP in IEEE 802.11 WLANs: Topics in design and implentation. *IEEE Comm. Mag.*, 47(4):111–117, 2009.
- [287] W. Lehr and J. Crowcroft. Managing shared access to a spectrum commons. In Proc. 1st IEEE Int. Symp. on New Frontiers in Dynamic Spectrum Access Networks (DySPAN 2005), pages 420–444, Baltimore, MD, USA, November 2005.
- [288] A. Leshem and E. Zehavi. Cooperative game theory and the Gaussian interference channel. *IEEE J. Select. Areas Commun.*, 26:1078–1088, 2008.
- [289] S. K. Leung-Yan-Cheong and M. E. Hellman. The Gaussian wiretap channel. *IEEE Trans. Inf. Theory*, 24(4):451–456, 1978.
- [290] F. Li, K. Wu, and A. Lippman. Energy-efficient cooperative routing in multi-hop wireless ad hoc networks. In *Proc. IEEE Int. Performance, Computing, and Communications Conf.*, pages 215–222, Phoenix, AZ, USA, April 2006.
- [291] S. Li and T. Başar. Distributed learning algorithms for the computation of non-cooperative equilibria. *Automatica*, 23(4):523–533, 1987.
- [292] X. Li, L. Qian, and D. Kataria. Downlink power control in co-channel macrocell femtocell overlay. *Proc. 43rd Ann. Conf. on Information Sciences and Systems (CISS 2009)*, pages 383–388, Baltimore, MD, USA, March 2009.
- [293] Z. Li, W. Trappe, and R. Yates. Secret communication via multi-antenna transmission. In Proc. 41st Conf. on Information Sciences and Systems (CISS 2007), pages 905–910, Baltimore, MD, USA, March 2007.
- [294] Y. Liang, H. V. Poor, and S. Shamai (Shitz). Secure communication over fading channels. *IEEE Trans. Inf. Theory*, 54(6):2470–2492, 2008.
- [295] Y. Lien. Performance issues of P2P file sharing over asymmetric and wireless networks. In Proc. IEEE Int. Conf. on Distributed Computing Systems, pages 850–855, Columbus, OH, USA, June 2005.
- [296] B. Lin, P. Ho, L. Xie, and X. Shen. Optimal relay station placement in IEEE 802.16j networks. In *Proc. Int. Conf. on Communications and Mobile Computing*, Honolulu, HI, USA, August 2007.
- [297] B. Lin, P. Ho, L. Xie, and X. Shen. Relay station placement in IEEE 802.16j dual-relay MMR networks. In *Proc. IEEE Int. Conf. on Communications (ICC 2008)*, pages 3437–3441, Beijing, China, May 2008.
- [298] H. Lin, M. Chatterjee, S. K. Das, and K. Basu. ARC: An integrated admission and rate control framework for competitive wireless CDMA data networks using non-cooperative games. *IEEE Trans. Mobile Comput.*, 4(3):243–258, 2005.
- [299] L. Lin, X. Zhou, L. Du, and X. Miao. Differential game model with coupling constraint for routing in ad hoc networks. In *Proc. 5th Int. Conf. on Wireless Communications, Networking and Mobile Computing (WiCOM 2009)*, pages 3042–3045, Beijing, China, September 2009.
- [300] Y. D. Lin and Y. C. Hsu. Multi-hop cellular: A new architecture for wireless communications. In Proc. IEEE Conf. on Computer Communications (INFOCOM 2000), pages 1273–1282, Tel Aviv, Israel, March 2000.
- [301] H. Liu and B. Krishnamachari. A price-based reliable routing game in wireless networks. In Workshop on Game Theory for Networks (Gamenets 2006), Pisa, Italy, 2006.
- [302] L. López, G. del R. Almansa, S. Paquelet, and A. Fernández. A mathematical model for the TCP tragedy of the commons. *Theor. Comput. Sci.*, 343(1–2):4–26, 2005.

- [303] S. H. Low and D. E. Lapsley. Optimization flow control, I: Basic algorithm and convergence. *IEEE/ACM Trans. Netw.*, 7(6):871–875, 1999.
- [304] E. K. Lua, J. Crowcroft, M. Pias, R. Sharma, and S. Lim. A survey and comparison of peerto-peer overlay network schemes. *IEEE Commun. Survey and Tutorials*, 7:72–93, 2005.
- [305] R. D. Luce and H. Raiffa. Games and Decisions. John Wiley and Sons, 1957.
- [306] J. Luo, R. S. Blum, L. J. Greenstein, L. J. Cimini, and A. M. Haimovich. New approaches for cooperative use of multiple antennas in ad hoc wireless networks. In *Proc. IEEE 60th Vehicular Technology Conf. (VTC 2004 Fall)*, pages 2769–2773, Milan, Italy, September 2004.
- [307] R. T. B. Ma, V. Misra, and D. Rubenstein. An analysis of generalized slotted-aloha protocols. *IEEE/ACM Trans. Networking*, 17(3):936–949, 2009.
- [308] A. MacKenzie, L. DaSilva, and W. Tranter. *Game Theory for Wireless Engineers*. Morgan and Claypool, 2006.
- [309] A. B. MacKenzie and S. B. Wicker. Stability of multipacket slotted aloha with selfish users and perfect information. In *Proc. IEEE Conf. on Computer Communications (INFOCOM* 2003), pages 1583–1590, San Francisco, CA, USA, March 2003.
- [310] M. Madiman. Cores of cooperative games in information theory. *EURASIP J. Wirel. Commun.*, 2008, March 2008.
- [311] N. Magharei and R. Rejaie. PRIME: Peer-to-peer receiver-driven mesh-based streaming. *IEEE/ACM Trans. Networking*, 17(4):1052–1065, 2009.
- [312] R. T. Maheswaran and T. Başar. Decentralized network resource allocation as a repeated non-cooperative market game. In *Proc. 40th IEEE Conf. on Decision and Control (CDC 2001)*, pages 4565–4570, Orlando, FL, USA, December 2001.
- [313] R. T. Maheswaran and T. Başar. Coalition formation in proportionally fair divisible auctions. Proc. 2nd Int. Joint Conf. on Autonomous Agents and Multi-Agent Systems, pages 25–32, Melbourne, Australia, July 2003.
- [314] R. T. Maheswaran and T. Başar. Nash equilibrium and decentralized negotiation in auctioning divisible resources. *Group Decision and Negotiation*, 12(5):361–395, 2003.
- [315] R. T. Maheswaran and T. Başar. Equilibrium and negotiation in multiple resource auctions. *IEEE Conf. on Decision and Control* (CDC 2003), pages 5939–5944, Maui, Hawaii, USA, December 2003.
- [316] R. T. Maheswaran and T. Başar. Social welfare of selfish agents: Motivating efficiency for divisible resources. *Proc. IEEE Conf. on Decision and Control (CDC 2004)*, pages 1550–1555, Nassau, Bahamas, December 2004.
- [317] R. T. Maheswaran and T. Başar. On revenue generation when auctioning network resources. Proc. Joint 44th IEEE Conf. on Decision and Control and European Control Conference (CDC-ECC 2005), pages 7466–7471, Seville, Spain, December 2005.
- [318] R. T. Maheswaran and T. Başar. Efficient signal proportional allocation (ESPA) mechanisms: Decentralized social welfare maximization for divisible resources. *IEEE J. Sel. Areas Commun.*, 24(5): 1000–1009, 2006.
- [319] I. Maric and R. D. Yates. Cooperative multi-hop broadcast for wireless networks. *IEEE J. Sel. Areas Commun.*, 22(6):1080–1088, 2004.
- [320] S. Marti, T. J. Giuli, K. Lai, and M. Baker. Mitigating routing misbehaviour in mobile ad hoc networks. In Proc. ACM/IEEE Ann. Int. Conf. on Mobile Computing and Networking (MobiCom 2000), pages 255–265, Boston, MA, USA, August 2000.
- [321] A. Mas-Colell, M. D. Whinston, and Jerry R. Green. *Microeconomic Theory*. Oxford University Press, 1995.

- [322] M. Maskery, V. Krishnamurthy, and Q. Zhao. Decentralized dynamic spectrum access for cognitive radios: Cooperative design of a non-cooperative game. *IEEE Trans. Commun.*, 57(2):459–469, 2009.
- [323] S. Mathur, L. Sankaranarayanan, and N. Mandayam. Coalitions in cooperative wireless networks. *IEEE J. Select. Areas Commun.*, 26:1104–1115, 2008.
- [324] C. Maxwell. *Auctioning Divisible Commodities: A Study of Price Determination*. Technical report, Harvard University, 1983.
- [325] M. A. McHenry, P. A. Tenhula, D. McCloskey, D. A. Roberson, and C. S. Hood. Chicago spectrum occupancy measurements and analysis and a long-term studies proposal. In *Proc. 1st Int. Workshop on Technology and Policy for Accessing Spectrum (TAPAS 2006)*, page 1, Boston, MA, USA, 2006.
- [326] S. Meguerdichian, F. Koushanfar, M. Potkonjak, and M. B. Srivastava. Coverage problems in wireless ad hoc sensor networks. In *Proc. IEEE Conf. on Computer Communications* (*INFOCOM 2001*), pages 1380–1387, Anchorage, AK, USA, April 2001.
- [327] D. S. Menasche, D. R. Figueiredo, and E. de S. Silva. An evolutionary game-theoretic approach to congestion control. *Performance Evaluation*, 62(1–4):295–312, 2005.
- [328] F. M. Menezes and P. K. Monteiro. An Introduction to Auction Theory. Oxford University Press, 2008.
- [329] K. Menger. Zur allgemeinen kurventheorie. Fund. Math., 10:96–115, 1927.
- [330] R. Menon, A. MacKenzie, M. R. Buehrer, and J. Reed. Game theory and interference avoidance in decentralized networks. In *Proc. SDR Forum Tech. Conf.*, Phoenix, AZ, USA, November 2004.
- [331] F. Meshkati, M. Chiang, H. V. Poor, and Stuart C. Schwartz. A game-theoretic approach to energy-efficient power control in multi-carrier CDMA systems. *IEEE J. Sel. Areas Commun.: Special Issue on Advances in Multicarrier CDMA*, 24(6):1115–1129, 2006.
- [332] F. Meshkati, D. Guo, H. V. Poor, and S. C. Schwartz. A unified approach to energy-efficient power control in large CDMA systems. *IEEE Trans. Wireless Commun.*, 7(4):1208–1216, 2008.
- [333] F. Meshkati, H. V. Poor, S. C. Schwartz, and R. Balan. Energy-efficient resource allocation in wireless networks with quality-of-service constraints. *IEEE Trans. Commun.*, 57(11):3406–3414, 2009.
- [334] F. Meshkati, H. V. Poor, S. C. Schwartz, and N. B. Mandayam. An energy-efficient approach to power control and receiver design in wireless data networks. *IEEE Trans. Commun.*, 53(11):1885–1894, 2005.
- [335] S. Micali and R. L. Rivest. Micropayments revisited. *Lecture Notes in Computer Science*, 2271/2002:171–203, 2002.
- [336] P. Michiardi and R. Molva. CORE: A collaborative reputation mechanism to enforce node cooperation in mobile ad hoc networks. In *Proc. IFIP Communication and Multimedia Security Conf. (CMS 2002)*, volume 228, pages 107–121, Montréal, Québec, Canada, 2002.
- [337] P. Michiardi and R. Molva. A game theoretical approach to evaluate cooperation enforcement mechanisms in mobile ad hoc networks. In *Proc. IEEE/ACM Int. Symp. on Modeling and Optimization in Mobile, Ad Hoc, and Wireless Networks (WiOpt 2003)*, pages 80–91, Sophia Antipolis, France, March 2003.
- [338] F. Milan, J. J. Jaramillo, and R. Srikant. Achieving cooperation in multi-hop wireless networks of selfish nodes. In *Workshop on Game Theory for Networks (GameNets 2006)*, Pisa, Italy, October 2006.
- [339] P. Milgrom. Putting Auction Theory to Work. Cambridge University Press, 2004.

- [340] J. Mitola.Cognitive radio for flexible mobile multimedia communications. In Proc. IEEE Int. Workshop on Mobile Multimedia Communications (MoMuC 1999), pages 3–10, San Diego, CA, USA, November 1999.
- [341] S. Mohamed, G. Rubino, and M. Varela. Performance evaluation of real-time speech through a packet network: A random neural networks-based approach. *Performance Evaluation*, 57(2):141–161, 2004.
- [342] D. Monderer and L. S. Shapley. Potential games. Games and Economic Behavior, 14:124–143, 1996.
- [343] S. Murthy and J. J. Garcia-Luna-Aceves. An efficient routing protocol for wireless networks. ACM Mobile Networks and App. J.: Special Issue on Routing in Mobile Communication Networks, 1(2):183–197, 1996.
- [344] J. Musacchio and J. Walrand. WiFi access point pricing as a dynamic game. IEEE/ACM Trans. Networking, 14(2):289–301, 2006.
- [345] A. Muthoo. Bargaining Theory with Applications. Cambridge University Press, 1999.
- [346] R. B. Myerson. Graphs and cooperation in games. Math. Oper. Res., 2:225–229, 1977.
- [347] R. B. Myerson. Game Theory: Analysis of Conflict. Harvard University Press, 1991.
- [348] R. B. Myerson and M. A. Satterthwaite. Efficient mechanisms for bilateral trading. J. Econ. Theory, 29:265–281, 1983.
- [349] Napster. http://www.napster.com.
- [350] J. Nash. The bargaining problem. Econometrica, 18:155–162, 1950.
- [351] J. Nash. Equilibrium points in n-person games. Proc. Natl. Acad. Sci. USA, 36(1):48–49, 1950.
- [352] J. Nash. Two-person cooperative game. *Econometrica*, 21:128–140, 1953.
- [353] S. G. Nash and A. Sofer. Linear and Nonlinear Programming. McGraw-Hill, 1995.
- [354] J. Neel, J. Reed, and R. Gilles. The role of game theory in the analysis of software radio network. In *Proc. SDR Forum Tech. Conf.*, San Diego, CA, USA, November 2002.
- [355] J. Neel, J. Reed, and R. Gilles. Convergence of cognitive-radio networks. In *Proc. IEEE Wireless Communications and Networking Conf. (WCNC 2004)*, pages 2250–2255, Atlanta, GA, USA, March 2004.
- [356] J. Neel, J. Reed, and R. Gilles. Game models for cognitive radio algorithm analysis. In Proc. SDR Forum Technical Conf., Phoenix, AZ, USA, November 2004.
- [357] R. Negi and S. Goelm. Secret communication using artificial noise. In *Proc. IEEE Vehic-ular Technology Conf. (VTC 2005 Fall)*, pages 1906–1910, Dallas, TX, USA, September 2005.
- [358] S. Ng and W. K. G. Seah. Game-theoretic model for collaborative protocols in selfish, tariff-free, multi-hop wireless networks. In *Proc. IEEE Conf. on Computer Communications* (*INFOCOM 2008*), pages 216–220, Phoenix, AZ, USA, April 2008.
- [359] S. Ng and W. K. G. Seah. Game-theoretic approach for improving cooperation in wireless multi-hop networks. *IEEE Trans. Syst., Man, Cybern. B, Cybern.*, 40(3):559–574, 2010.
- [360] N. Nisan, T. Roughgarden, E. Tardos, and V. Vazirani, eds. Algorithmic Game Theory. Cambridge University Press, 2007.
- [361] D. Niyato and E. Hossain. Queue-aware uplink bandwidth allocation and rate control for polling service in IEEE 802.16 broadband wireless networks. *IEEE Trans. Mobile Comput.*, 5(6):668–679, 2006.
- [362] D. Niyato and E. Hossain. A queuing-theoretic and optimization-based model for radio resource management in IEEE 802.16 broadband wireless networks. *IEEE Trans. Comput.*, 55(11):1473–1488, 2006.

- [363] D. Niyato and E. Hossain. A novel analytical framework for integrated cross-layer study of call-level and packet-level QoS in wireless mobile multimedia networks. *IEEE Trans. Mobile Comput.*, 6(3):322–335, 2007.
- [364] D. Niyato and E. Hossain. QoS-aware bandwidth allocation and admission control in IEEE 802.16 broadband wireless access networks: A non-cooperative game theoretic approach. *Elsevier Computer Networks*, 51(11):3305–3321, 2007.
- [365] D. Niyato and E. Hossain. Dynamics of networks selection in heterogeneous wireless networks: An evolutionary game approach. *IEEE Trans. Veh. Technol.*, 57(6):3778–3788, 2008.
- [366] D. Niyato and E. Hossain. Modeling user churning behavior in wireless networks using evolutionary game theory. In *Proc. IEEE Wireless Communications and Networking Conf.* (WCNC 2008), pages 2793–2797, Las Vegas, NV, USA, March–April 2008.
- [367] D. Niyato and E. Hossain. Dynamics of network selection in heterogeneous wireless networks: An evolutionary game approach. *IEEE Trans. Veh. Technol.*, 58(4):2008–2017, 2009.
- [368] D. Niyato, E. Hossain, and Z. Han. Dynamic Spectrum Access in Cognitive Radio Networks. Cambridge University Press, 2009.
- [369] D. Niyato, E. Hossain, D. I. Kim, and Z. Han. Relay-centric radio resource management and network planning in IEEE 802.16j mobile multi-hop relay networks. *IEEE Trans. Wireless Commun.*, 8(2):6115–6125, 2009.
- [370] D. Niyato, E. Hossein, D. Kim, and Z. Han. Joint optimization of placement and bandwidth reservation for relays in IEEE 802.16j mobile multi-hop networks. In *Proc. IEEE Int. Conf.* on Communications (ICC 2009), pages 1–5, Dresden, Germany, June 2009.
- [371] D. Niyato, P. Wang, W. Saad, and A. Hjørungnes. Coalition formation games for bandwidth sharing in vehicle-to-roadside communications. In *Proc. IEEE Wireless Communications* and Networking Conf. (WCNC 2010), pages 1–5, Sydney, Australia, April 2010.
- [372] P. Nuggehalli, M. Sarkar, and R. R. Rao. QoS and selfish users: A MAC layer perspective. In *Proc. IEEE Global Communications Conf. (GLOBECOM 2007)*, pages 4719–4723, Washington, DC, USA, November 2007.
- [373] P. Nurmi. Modelling routing in wireless ad hoc networks with dynamic Bayesian games. In Proc. 1st IEEE Int. Communications Society Conf. on Sensor and Ad Hoc Communications and Networks, pages 63–70, Santa Clara, CA, USA, October 2004.
- [374] S. Olariu and M. C. Weigle, eds. Vehicular Networks: From Theory to Practice. Chapman & Hall/CRC Computer and Information Sciences Series. CRC Press, London, UK, March 2009.
- [375] F. Ordonez and N. E. Stier-Moses. Robust Wardrop equilibrium. In Proc. 1st EuroFGI Int. Conf. on Network Control and Optimization, pages 247–256, Avignon, France, June 2007.
- [376] M. J. Osborne and A. Rubinstein. A Course in Game Theory. MIT Press, 1994.
- [377] G. Owen. Game Theory, 3rd edition. Academic Press, 1995.
- [378] M. Ozay and F. T. Y. Vural. A theoretical analysis of feature fusion in stacked generalization. In Proc. 17th IEEE Signal Processing and Communications Applications Conf., Antalya, Turkey, April 2009.
- [379] Z. Pan, E. Altman, and T. Başar. Robust adaptive flow control in high speed telecommunication networks. J. Appl. and Comp. Math., 1(2):108–130, 2002.
- [380] P. Papadimitratos, S. Sankaranarayanan, and A. Mishra. A bandwidth sharing approach to improve licensed spectrum utilization. *IEEE Comm. Mag.*, 43(12):10–14, 2005.

- [381] J. Papandriopoulos and J. S. Evans. Low-complexity distributed algorithms for spectrum balancing in multi-user DSL networks. In *Proc. IEEE Int. Conf. on Communications (ICC 2006)*, pages 3270–3275, Istanbul, Turkey, June 2006.
- [382] C. H. Papdimitriou. Algorithms, games, and the Internet. In Proc. 33rd Ann. ACM Symp. on Theory of Computing, pages 749–753, Hersonissos, Greece, July 2001.
- [383] P. Parada and R. Blahut. Secrecy capacity of SIMO and slow fading channels. In Proc. IEEE Int. Symp. on Information Theory, pages 2152–2155, Adelaide, Australia, September 2005.
- [384] H. Park and M. van der Schaar. Bargaining strategies for networked multimedia resource management. *IEEE Trans. Signal Process.*, 55:3496–3511, 2007.
- [385] H. Park and M. van der Schaar. A framework for foresighted resource reciprocation in P2P networks. *IEEE Trans. Multimedia*, 11(1):101–116, 2009.
- [386] V. D. Park and M. S. Corson. A highly adaptive distributed routing algorithm for mobile wireless networks. In *Proc. IEEE Conf. on Computer Communications (INFOCOM 1997)*, pages 1405–1413, Kobe, Japan, April 1997.
- [387] C. Perkins. Highly dynamic destination-sequenced distance-vector routing (DSDV) for mobile computers. In Proc. ACM Conf. on Communications Architectures, Protocols, and Applications (SIGCOMM 1994), London, UK, September 1994.
- [388] C. Perkins and E. Royer. Ad-hoc on-demand distance vector routing. In Proc. 2nd IEEE Workshop on Mobile Computing Systems and Applications, pages 90–100, New Orleans, LA, USA, February 1999.
- [389] S. W. Peters, A. Panah, K. Truong, and R. W. Heath. Relay architectures for 3GPP LTE-Advanced. *EURASIP J. Wirel. Comm.*, May 2009.
- [390] R. H. Porter. Optimal cartel trigger price strategies. J. Econ. Theory, 29:313–318, 1983.
- [391] J. Proakis and M. Salehi. *Digital Communications*, 5th edition. McGraw-Hill Science/Engineering/Math, 2007.
- [392] D. Qiao and K. G. Shin. Achieving efficient channel utilization and weighted fairness for data Communications in IEEE 802.11 WLAN under the DCF. In *Proc. 10th IEEE Int. Workshop on Quality of Service (IWQOS 1997)*, pages 227–236, New York, NY, USA, May 2002.
- [393] X. Qin and R. Berry. Exploiting multiuser diversity for medium access control in wireless networks. In Proc. IEEE Conf. on Computer Communications (INFOCOM 2003), pages 1084–1094, San Francisco, CA, USA, March–April 2003.
- [394] D. Qiu and R. Srikant. Modeling and performance analysis of BitTorrent-like peer-to-peer networks. In Proc. ACM Conference on Applications, Technologies, Architectures, and protocols for Computer Communication (SIGCOMM 2004), pages 367–378, Portland, OR, USA, August 2004.
- [395] W. Qiu, E. Skafidas, and P. Hao. Enhanced tree routing for wireless sensor networks. Ad Hoc Networks, 7(3):638–650, 2009.
- [396] V. Raghunathan and P. R. Kumar. Wardrop routing in wireless networks. *IEEE Trans. Mobile Comput.*, 8(5):636–652, 2009.
- [397] T. S. Rappaport. *Wireless Communications: Principles and Practice*, 2nd edition. Prentice-Hall, 2002.
- [398] D. Ray. A Game-Theoretic Perspective on Coalition Formation. Oxford University Press, January 2007.
- [399] M. Raya, J. Hubaux, and I. Aad. DOMINO: A system to detect greedy behavior in IEEE 802.11 hotspots. In Proc. 2nd Int. Conf. on Mobile Systems, Applications, and Services (MobiSys 2004), pages 84–97, Boston, MA, USA, June 2004.

- [400] S. H. Ree and T. Konstantopoulos. Optimal flow control and capacity allocation in multiservice networks. In *Proc. IEEE Conf. on Decision and Control* (CDC 1998), pages 1662–1667, Tampa, FL, USA, December 1998.
- [401] W. Rhee and J. M. Cioffi. Increase in capacity of multiuser OFDM system using dynamic subchannel allocation. In *Proc. IEEE Vehicular Technology Conf. (VTC 2000 Spring)*, pages 1085–1089, Tokyo, Japan, May 2000.
- [402] J. B. Rosen. Existence and uniqueness of equilibrium points for concave n-person games. *Econometrica*, 33:520–534, 1965.
- [403] T. Roughgarden. Routing games. In N. Nisan, T. Roughgarden, E. Tardos, and V. V. Vazirani, eds., Algorithmic Game Theory. Cambridge University Press, 2007. www-cs.stanford.edu/~tim/papers/rg.pdf.
- [404] A. N. Rouskas, A. A. Kikilis, and S. S. Ratsiatos. Admission control and pricing in competitive wireless networks based on non-cooperative game theory. In *Proc. IEEE Wireless Communications and Networking Conf. (WCNC 2006)*, pages 205–210, Las Vegas, NV, USA, April 2006.
- [405] A. Rubinstein. Perfect equilibrium in a bargaining model. *Econometrica*, 50:97–110, 1982.
- [406] W. Saad, Z. Han, T. Başar, M. Debbah, and A. Hjørungnes. A selfish approach to coalition formation among unmanned aerial vehices in wireless networks. In *Proc. Int. Conf. on Game Theory for Networks (GameNets 2009)*, pages 259–267, Istanbul, Turkey, May 2009.
- [407] W. Saad, Z. Han, T. Başar, M. Debbah, and A. Hjørungnes. Physical layer security: Coalitional games for distributed cooperation. In *Proc. 7th Int. Symp. on Modeling and Optimization in Mobile, Ad Hoc, and Wireless Networks (WiOpt 2009)*, Seoul, South Korea, June 2009.
- [408] W. Saad, Z. Han, T. Başar, M. Debbah, and A. Hjørungnes. A coalition-formation game in partition form for peer-to-peer file sharing networks. In *Proc. IEEE Global Communications Conf. (GLOBECOM 2010)*, Miami, FL, USA, December 2010.
- [409] W. Saad, Z. Han, T. Başar, A. Hjørungnes, and J. B. Song. Hedonic coalition-formation games for secondary base station cooperation in cognitive-radio networks. In *Proc. IEEE Wireless Commun. and Networking Conf. (WCNC 2010)*, pages 1–6, Sydney, Australia, April 2010.
- [410] W. Saad, Z. Han, M. Debbah, and A. Hjørungnes. Network formation games for distributed uplink tree construction in IEEE 802.16j networks. In *Proc. IEEE Global Communications Conf. (GLOBECOM 2008)*, pages 1–5, New Orleans, LA, USA, November–December 2008.
- [411] W. Saad, Z. Han, M. Debbah, and A. Hjørungnes. A distributed coalition formation framework for fair user cooperation in wireless networks. *IEEE Trans. Wireless Commun.*, 8:4580–4593, 2009.
- [412] W. Saad, Z. Han, M. Debbah, A. Hjørungnes, and T. Başar. Coalitional game theory for communication networks: A tutorial. *IEEE Signal Processing Magazine: Special issue on Game Theory in Signal Processing and Communications*, 26:77–97, 2009. 413
- [413] W. Saad, Z. Han, M. Debbah, A. Hjørungnes, and T. Başar. Coalitional games for distributed collaborative spectrum sensing in cognitive-radio networks. In *Proc. IEEE Conf. on Computer Communications (INFOCOM 2009)*, pages 2114–2122, Rio de Janeiro, Brazil, April 2009.
- [414] W. Saad, Z. Han, M. Debbah, A. Hjørungnes, and T. Başar. A game-based self-organizing uplink tree for VoIP services in IEEE 802.16j networks. In *Proc. IEEE Int. Conf. on Communications (ICC 2009)*, pages 1–5, Dresden, Germany, June 2009.
- [415] W. Saad, Z. Han, A. Hjørungnes, D. Niyato, and E. Hossain. Coalition formation games for distributed cooperation among roadside units in vehicular networks. *IEEE J. Select. Areas Commun., Special Issue on Vehicular Communications and Networks*, 29(1):48–60, 2011.
- [416] W. Saad, Q. Zhu, T. Başar, Z. Han, and A. Hjørungnes. Hierarchical network-formation games in the uplink of multi-hop wireless networks. In *Proc. IEEE Global Communications Conf. (GLOBECOM 2009)*, pages 1–6, Honolulu, HI, USA, December 2009.
- [417] A. K. Sadek, K. J. R. Liu, and A. Ephremides. Cognitive multiple access via cooperation: Protocol design and performance analysis. *IEEE Trans. Inf. Theory*, 53(10):3677–3696, 2007.
- [418] A. K. Sadek, W. Su, and K. J. R. Liu. A class of cooperative communication protocols for multi-node wireless networks. In *Proc. IEEE Int. Workshop on Signal Processing Advances in Wireless Commun. (SPAWC 2005)*, pages 560–564, New York, NY, USA, June 2005.
- [419] L. Samuelson. Evolutionary Games and Equilibrium Selection. MIT Press, January 1997.
- [420] T. Sandholm, K. Larson, M. Andersson, O. Shehory, and F. Tohme. Coalition structure generation with worst case guarantees. *Artificial Intelligence*, 111(1):209–238, 1999.
- [421] W. H. Sandholm. Potential games with continuous player sets. J. Econ. Theory, 97(1): 81–108, 2001.
- [422] S. Sanghavi and B. Hajek. Optimal Allocation of a Divisible Good to Strategic Buyers. Technical report, University of Illinois at Urbana-Champaign, 2004. http://ima.umn.edu/talks/workshops/3-8-13.2004/sanghavi/ sanghavi_hajek_divisible_good.pdf.
- [423] S. Sankaranarayanan, P. Papadimitratos, A. Mishra, and S. Hershey. A bandwidth sharing approach to improve licensed spectrum utilization. In *Proc. 1st IEEE Int. Symp. on New Frontiers in Dynamic Spectrum Access Networks (DySPAN 2005)*, pages 279–288, Baltimore, MD, USA, November 2005.
- [424] C. U. Saraydar, N. B. Mandayam, and D. J. Goodman. Pricing and power control in a multicell wireless data network. *IEEE J. Sel. Areas Commun.*, 19(10):1883–1892, 2001.
- [425] C. U. Saraydar, N. B. Mandayam, and D. J. Goodman. Efficient power control via pricing in wireless data networks. *IEEE Trans. Commun.*, 50(2):291–303, 2002.
- [426] S. Saroiu, P. K. Gummadi, and S. D. Gribble. A measurement study of peer-to-peer file sharing systems. In *Proc. SPIE Multimedia Computing and Networking*, San Jose, CA, USA, January 2002.
- [427] M. A. Satterthwaite. Strategy-proofness and arrow's conditions: Existence and correspondence theorems for voting procedures and social welfare functions. J. Econ. Theory, 10:187–217, 1975.
- [428] A. Scaglione, D. L. Goeckel, and J. N. Laneman. Cooperative communications in mobile ad hoc networks. *IEEE Signal Processing Magazine*, 23(5):18–29, 2006.
- [429] R. Schoenen, R. Halfmann, and B. H. Walke. MAC performance of a 3GPP-LTE multihop cellular network. In *Proc. IEEE Int. Conf. on Communications (ICC 2008)*, pages 4819–4824, Beijing, China, May 2008.

- [430] G. Scutari, S. Barbarossa, and D. P. Palomar. Potential games: A framework for vector power control problems with coupled constraints. In *Proc. IEEE Intl. Conf. on Acoustics, Speech, and Signal Processing*, pages 241–244, Toulouse, France, May 2006.
- [431] G. Scutari, D. P. Palomar, and S. Barbarossa. Optimal linear precoding strategies for wideband non-cooperative systems based on game theory and art I: Nash equilibria. *IEEE Trans. Signal Process.*, 56(3):1230–1249, 2008.
- [432] A. Sendonaris, E. Erkip, and B. Aazhang. User cooperation diversity, part I: System description. *IEEE Trans. Commun.*, 51(11):1927–1938, 2003.
- [433] A. Sendonaris, E. Erkip, and B. Aazhang. User cooperation diversity, part II: Implementation aspects and performance analysis. *IEEE Trans. on Commun.*, 51(11):1939–1948, 2003.
- [434] S. Sengupta, S. Anand, K. Hong, and R. Chandramouli. On adversarial games in dynamic spectrum access networking based covert timing channels. ACM Mobile Computing and Commun. Review (MC2R): Special Issue on Cognitive Radio Technologies and Systems, pages 96–107, 2009.
- [435] S. Sengupta, R. Chandramouli, S. Brahma, and M. Chatterjee. A game theoretic framework for distributed self-coexistence among IEEE 802.22 networks. In *Proc. IEEE Global Communications Conf. (GLOBECOM 2008)*, pages 1–6, New Orleans, LA, USA, November–December 2008.
- [436] S. Sengupta and M. Chatterjee. Synchronous and Asynchronous Auction Models for Dynamic Spectrum Access. In *Distributed Computing and Networking*. Proceedings of the 8th International Conference (ICDCN 2006), Guwahati, India, December 2006, pages 558–569. Springer, 2006.
- [437] S. Sengupta and M. Chatterjee. Sequential and concurrent auction mechanisms for dynamic spectrum access. In Proc. Int. Conf. on Cognitive Radio Oriented Wireless Networks and Commun. (CrownCom 2007), pages 448–455, Orlando, FL, USA, August 2007.
- [438] S. Sengupta and M. Chatterjee. Designing auction mechanisms for dynamic spectrum access. ACM/Springer Mobile Networks and Applications (MONET): Special issue on Cognitive Radio Oriented Wireless Networks and Communications, 13(5):498–515, 2008.
- [439] A. Sfairopoulou, B. Bellalta, and C. Macian. How to tune VoIP codec selection in WLANs? IEEE Comm. Lett., 12(8):551–553, 2008.
- [440] S. Shafiee and S. Ulukus. Achievable rates in Gaussian MISO channels with secrecy constraints. In *Proc. IEEE Int. Symp. on Information Theory*, pages 2466–2470, Nice, France, June 2007.
- [441] S. Shafiee and S. Ulukus. Mutual information games in multi-user channels with correlated jamming. *IEEE Trans. Inf. Theory*, 55(10):4598–4607, 2009.
- [442] S. Shakkottai, E. Altman, and A. Kumar. Multihoming of users to access points in WLANs: A population game perspective. *IEEE J. Sel. Areas Commun.*, 25(6):1207–1215, 2007.
- [443] S. Shakkottai and R. Srikant. Economics of network pricing with multiple isps. *IEEE/ACM Trans. Networking*, 14(6):1233–1245, 2006.
- [444] S. Shakkottai, R. Srikant, A. Ozdaglar, and D. Acemoglu. The price of simplicity. *IEEE J. Sel. Areas Commun.*, 26(7):1269–1276, 2008.
- [445] J. Shamma and G. Arslan. Unified convergence proofs of continuous-time fictitious play. *IEEE Trans. Autom. Control*, 49(7):1137–1142, 2004.
- [446] S. N. Shankar, C. Chou, K. Challapali, and S. Mangold. Spectrum agile radio: Capacity and QoS implications of dynamic spectrum assignment. In *Proc. IEEE Global Communications Conf. (GLOBECOM 2005)*, pages 2510–2516, St. Louis, MO, USA, December 2005.

- [447] L. Shapley and M. Shubik. The assignment game, I: The core. Int. J. Game Theory, 1:111–130, 1972.
- [448] L. S. Shapley. Stochastic games. Proc. Natl. Acad. Sci., USA, 39:1095-1100, 1953.
- [449] A. Shawish, X. Jiang, P. Ho, and S. Horiguchi. Wireless access point voice capacity analysis and enhancement based on clients' spatial distribution. *IEEE Trans. Veh. Technol.*, 58(5):2597–2603, 2009.
- [450] H. Shen and T. Başar. Network game with a probabilistic description of user types. In *Proc.* 43rd IEEE Conf. on Decision and Control (CDC 2004), pages 4225–4230, Paradise Island, Bahamas, December 2004.
- [451] H. Shen and T. Başar. Optimal nonlinear pricing for a monopolistic network service provider with complete and incomplete information. *IEEE J. Sel. Areas Commun.*, 25(6):1216–1223, 2007.
- [452] H. Shen and T. Başar. Pricing under information asymmetry for a large population of users. *Telecommunication Systems*, 47(1–2):123–136, 2011.
- [453] H. Shiang and M. van der Schaar. Distributed resource management in multi-hop cognitiveradio networks for delay sensitive transmission. *IEEE Trans. Veh. Technol.*, 58(2):941–953, 2009.
- [454] T. Shono, H. Shiba, Y. Shirato, *et al.* Performance of IEEE 802.11 wireless LAN implemented on software defined radio with hybrid programmable architecture. In *Proc. IEEE Int. Conf. on Communications (ICC 2003)*, pages 2035–2040, Anchorage, AK, USA, May 2003.
- [455] B. Shrestha, D. Niyato, Z. Han, and E. Hossain. Wireless access in vehicular environments using BitTorrent and bargaining. In *Proc. IEEE Global Communications Conf.* (GLOBECOM 2008), pages 1–5, New Orleans, LA, USA, November 2008.
- [456] K. W. Shum, K. Leung, and C. W. Sung. Convergence of iterative waterfilling algorithm for Gaussian interference channels. *IEEE J. Select. Areas Commun.*, 25(6):1091–1100, 2007.
- [457] M. Sikora, J. N. Laneman, M. Haenggi, D. J. Costello, and T. E. Fuja. Bandwidth- and powerefficient routing in linear wireless networks. *IEEE Trans. Inf. Theory*, 53(11):2624–2633, 2006.
- [458] A. Silva, P. Bernhard, and E. Altman. Numerical solutions of continuum equilibria for routing in dense ad-hoc networks. In Proc. Workshop on Interdisciplinary Systems Approach in Performance Evaluation and Design of Computer and Communication Systems (Inter-Perf), Athens, Greece, October 2008.
- [459] M. Simaan and J. B. Cruz Jr. On the Stackelberg strategy in nonzero-sum games. J. Optimiz. Theory App., 11(5):533–555, 1973.
- [460] O. Simeone, I. Stanojev, S. Savazzi, et al. Spectrum leasing to cooperating secondary ad hoc networks. *IEEE J. Sel. Areas Commun.*, 26(1):203–213, 2008.
- [461] H. Singh and S. Singh. Energy consumption of TCP Reno, Newreno, and SACK in multihop wireless networks. In Proc. 2002 ACM Int. Conf. on Measurement and Modeling of Computer Systems (SIGMETRICS 2002), pages 206–216, New York, NY, USA, 2002.
- [462] J. Slotine and W. Li. Applied Nonlinear Control. Prentice-Hall, 1991.
- [463] J. M. Smith and G. R. Price. The logic of animal conflict. *Nature*, 246:15–18, 1973.
- [464] V. Srinivasan, P. Nuggehalli, C. F. Chiasserini, and R. R. Rao. Cooperative in wireless ad hoc networks. In *Proc. IEEE Conf. on Computer Communications (INFOCOM 2003)*, pages 808–817, San Francisco, CA, USA, March 2003.

- [465] C. A. St Jean and B. Jabbari. Bayesian game-theoretic modeling of transmit power determination in a self-organizing CDMA wireless network. In *Proc. IEEE 60th Vehicular Technology Conf. (VTC 2004 Fall)*, pages 3496–3500, Milan, Italy, September 2004.
- [466] W. Su, A. K. Sadek, and K. J. R. Liu. SER performance analysis and optimum power allocation for decode-and-forward cooperation protocol in wireless networks. In *Proc. IEEE Wireless Communications and Networking Conf. (WCNC 2005)*, pages 984–989, New Orleans, LA, USA, March 2005.
- [467] Y. Su and M. van der Schaar. A new perspective on multi-user power control games in interference channels. *IEEE Trans. Wireless Commun.*, 8:2910–2919, 2009.
- [468] Y. H. Suk and H. Y. Kai. Challenges in the migration to 4G mobile systems. *IEEE Comm. Mag.*, 41(12):54–59, 2003.
- [469] J. Sun, E. Modiano, and L. Zheng. Wireless channel allocation using an auction algorithm. *IEEE J. Sel. Areas Commun.*, 24(5):1085–1096, 2006. 470
- [470] K. J. Sunnevåg. Auction Design for the Allocation of Emission Permits. Working Paper in Economics, University of California at Santa Barbara, June 2001.
- [471] R. S. Sutton and A. G. Barto. Reinforcement Learning: An Introduction. MIT Press, 1998.
- [472] K. Taniuchi, Y. Ohba, V. Fajardo, *et al.* IEEE 802.21: Media independent handover: Features, applicability, and realization. *IEEE Comm. Mag.*, 47(1):112–120, 2009.
- [473] H. Tembine, E. Altman, R. El-Azouzi, and Y. Hayel. Evolutionary games in wireless networks. *IEEE Trans. Syst., Man, Cybern. B, Cybern.*, PP(99):1–13, 2009.
- [474] O. Teyeb, V. V. Phan, B. Raaf, and S. Redana. Dynamic relaying in 3GPP LTE-Advanced networks. *EURASIP J. Wirel. Comm.*, 2009, July 2009.
- [475] R. Thrall and W. Lucas. N-person games in partition function form. *Naval Research Logistics Quarterly*, 10:281–298, 1963.
- [476] C. K. Toh. Associativity-based routing for ad hoc mobile networks. Wireless Personal Communications, 4(2):103–139, 1997.
- [477] J. M. Torrance and L. Hanzo. Adaptive modulation in a slow Rayleigh fading channel. In Proc. IEEE Int. Symp. on Personal, Indoor and Mobile Radio Commun. (PIMRC), pages 497–501, Taipei, Taiwan, October 1996.
- [478] A. Toskala and H. Holma, eds. HSDPA/HSUPA for UMTS. John Wiley and Sons, June 2006.
- [479] D. Tse and P. Viswanath. Fundamentals of Wireless Communication. Cambridge University Press, 2005.
- [480] S. Tsutsui and K. Mino. Nonlinear strategies in dynamic duopolistic competition with sticky prices. J. of Econ. Theory, 52(1):136–161, 1990.
- [481] K. Tuyls and A. Nowé. Evolutionary game theory and multi-agent reinforcement learning. *Knowledge Engineering Review*, 20(1):63–90, 2005.
- [482] S. Ulukus and R. D. Yates. Stochastic power control for cellular radio systems. *IEEE Trans. Commun.*, 46(6):784–798, 1998.
- [483] UN Agency for Information and Communication Technologies. Recommendation G107: The E-Model, a Computational Model for Use in Transmission Planning. ITU-T Technical report, June 2002. http://www.itu.int/itudoc/itu-t/aap/ sg12aap/history/g107/g107.html.
- [484] J. Unnikrishnan and V. V. Veeravalli. Cooperative sensing for primary detection in cognitive radio. *IEEE J. Selected Topics Signal Process.*, 2(1):18–27, 2008.
- [485] M. van der Schaar and F. Fu. Spectrum access games and strategic learning in cognitive radio networks for delay-critical applications. *Proc. IEEE*, 97(4):720–740, 2009.

- [486] M. van der Schaar and D. S. Turaga. Cross-layer packetization and retransmission strategies for delay-sensitive wireless multimedia transmission. *IEEE Trans. Multimedia*, 9(1):185–197, 2007.
- [487] S. J. Vaughan-Nichols. Achieving wireless broadband with WiMAX. *IEEE Computer*, 37(6):10–13, 2004.
- [488] N. Vieille. Stochastic games: Recent results. In *Handbook of Game Theory*, pages 1833–1850. Elsevier Science, 2002.
- [489] T. L. Vincent and J. S. Brow. Evolutionary Game Theory, Natural Selection, and Darwinian Dynamics. Cambridge University Press, 2005.
- [490] E. Visotsky, J. Bae, R. Peterson, R. Berryl, and M. L. Honig. On the uplink capacity of an 802.16j system. In *Proc. IEEE Wireless Communications and Networking Conf. (WCNC* 2008), pages 2657–2622, Las Vegas, NV, USA, April 2008.
- [491] E. Visotsky, S. Kuffner, and R. Peterson. On collaborative detection of TV transmissions in support of dynamic spectrum sharing. In *Proc. 1st IEEE Int. Symp. on New Frontiers in Dynamic Spectrum Access Networks (DySPAN 2005)*, pages 338–345, Baltimore, MD, USA, November 2005.
- [492] J. von Neumann and O. Morgenstern. *Theory of Games and Economic Behavior*. Princeton University Press, 1944.
- [493] J. P. Walters, Z. Liang, W. Shi, and V. Chaudhary. Wireless Sensor Network Security: A Survey–Security in Distributed, Grid, and Pervasive Computing. Auerbach Publications, 2006.
- [494] B. Wang, Z. Han, and K. J. R. Liu. Distributed relay selection and power control for multiuser cooperative communication networks using buyer/seller game. In *Proc. IEEE Conf.* on Computer Communications (INFOCOM 2007), pages 544–552, Anchorage, AK, USA, May 2007.
- [495] B. Wang, Z. Han, and K. J. R. Liu. Stackelberg game for distributed resource allocation over multiuser cooperative communication networks. *IEEE Trans. Mobile Comput.*, 8(7):975–990, 2009.
- [496] B. Wang, K. J. R. Liu, and T. C. Clancy. Evolutionary cooperative spectrum sensing game: How to collaborate? *IEEE Trans. Commun.*, 58(3):890–900, 2010.
- [497] J. J. D. Wang and J. F. Zender. Auctioning divisible goods. *Econ. Theory*, 19(4):673–705, 2002.
- [498] W. Wang, M. Chatterjee, and K. Kwiat. Coexistence with malicious nodes: A game theoretic approach. In *Proc. Int. Conf. on Game Theory for Networks (GameNets 2009)*, pages 277–286, Istanbul, Turkey, May 2009.
- [499] W. Wang and X. Liu. List-coloring based channel allocation for open-spectrum wireless networks. In *Proc. IEEE 62nd Vehicular Technology Conf. (VTC 2005 Fall)*, pages 690–694, Dallas, TX, USA, September 2005.
- [500] W. Wang, T. Peng, and W. Wang. Optimal power control under interference temperature constraints in cognitive-radio network. In *Proc. IEEE Wireless Communications and Networking Conf. (WCNC 2007)*, pages 116–120, Hong Kong, China, March 2007.
- [501] J. G. Wardrop. Some theoretical aspects of road traffic research. Proc. Inst. Civ. Eng., Part 2, 1:325–378, 1952.
- [502] E. H. Watanabe, D. S. Menasche, E. de Souza e Silva, and R. M. M. Leão. Modeling resource sharing dynamics of VoIP users over a WLAN using a game-theoretic approach. In *Proc. IEEE Conf. on Computer Communications (INFOCOM 2008)*, pages 915–923, Phoenix, AZ, USA, April 2008.

- [503] C. J. C. H. Watkins and P. Dayan. Technical note: Q-learning. *Machine Learning*, 8(3–4):279–292, 1992.
- [504] J. W. Weibull. Evolutionary Game Theory. MIT Press, 1997.
- [505] R. Wilson. Auctions of shares. Quart. J. Econ., 93:675-698, 1979.
- [506] T. Wirth, V. Venkatkumar, T. Haustein, E. Schulz, and R. Halfmann. LTE-advanced relaying for outdoor range extension. In *Proc. Vehicular Technology Conf. (VTC 2009 Fall)*, pages 1–4, Anchorage, AK, USA, September 2009.
- [507] C. Y. Wong, R. S. Cheng, K. B. Letaief, and R. D. Murch. Multiuser OFDM with adaptive subcarrier, bit, and power allocation. *IEEE J. Select. Areas Commun.*, 17(10):1747–1758, 1999.
- [508] C. Y. Wong, C. Y. Tsui, R. S. Cheng, and K. B. Letaief. A real-time sub-carrier allocation scheme for multiple access downlink OFDM transmission. In *Proc. IEEE Vehicular Technology Conf. (VTC 1999 Fall)*, pages 1124–1128, Amsterdam, Netherlands, September 1999.
- [509] L. Wooyul, K. Youngjae, M. H. Brady, and J. M. Cioffi. Band-preference dynamic spectrum management in a DSL environment. In *Proc. IEEE Global Communications Conf.* (*GLOBECOM 2006*), pages 1–5, San Francisco, CA, USA, November–December 2006.
- [510] Y. Wu and D. H. K. Tsang. Distributed power allocation algorithm for spectrum sharing cognitive radio networks with QoS guarantee. In *Proc. IEEE Conf. on Computer Communications (INFOCOM 2009)*, pages 981–989, Rio de Janeiro, Brazil, April 2009.
- [511] Y. Wu, B. Wang, K. J. R. Liu, and T. C. Clancy. Repeated open spectrum sharing game with cheat-proof strategies. *IEEE Trans. Wireless Commun.*, 8(4):1922–1933, 2009.
- [512] A. D. Wyner. The wire-tap channel. Bell System Technical Journal, 54(8):1355–1387, 1975.
- [513] Y. Xing, R. Chandramouli, and C. Cordeiro. Price dynamics in competitive agile spectrum access markets. *IEEE J. Sel. Areas Commun.*, 25(3):613–621, 2007.
- [514] Y. Xing, C. N. Mathur, M. A. Haleem, R. Chandramouli, and K. P. Subbalakshmi. Priority based dynamic spectrum access with QoS and interference temperature constraints. In *Proc. IEEE Int. Conf. on Communications (ICC 2006)*, pages 4420–4425, Istanbul, Turkey, June 2006.
- [515] Y. Xing, C. N. Mathur, M. A. Haleem, R. Chandramouli, and K. P. Subbalakshmi. Dynamic spectrum access with QoS and interference temperature constraints. *IEEE Trans. Mobile Comput.*, 6(4):423–433, 2007.
- [516] C. Yang, J. Li, and Z. Tian. Optimal power control for cognitive-radio networks under coupled interference constraints: A cooperative game-theoretic perspective. *IEEE Trans. Veh. Technol.*, 59:1696–1706, 2010.
- [517] S. Yang and B. Hajek. An Efficient Mechanism for Allocation of a Divisible Good. Technical report, University of Illinois at Champaign Urbana, 2005. http://ftp.ima.umn.edu/talks/ workshops/3-8-13.2004/yang/YangHajek.pdf.
- [518] Z. Yang, J. Liu, and A. Høst-Madsen. Cooperative routing and power allocation in adhoc networks. In *Proc. IEEE Global Communications Conf. (GLOBECOM 2005)*, pages 2730–2734, St. Louis, MO, USA, November 2005.
- [519] R. D. Yates. A framework for uplink power control in cellular radio systems. *IEEE J. Select. Areas Commun.*, 13(7):1341–1347, 1995.
- [520] R. D. Yates and C. Y. Huang. Integrated power control and base-station assignment. *IEEE Trans. Veh. Technol.*, 44:638–644, 1995.
- [521] S.-P. Yeh, S. Talwar, S.-C. Lee, and H. Kim. WiMAX femtocells: A perspective on network architecture, capacity, and coverage. *IEEE Comm. Mag.*, 46(10):58–65, 2008.

- [522] H. Yin and H. Liu. An efficient multiuser loading algorithm for OFDM-based broadband wireless systems. In *Proc. IEEE Global Communications Conf. (GLOBECOM 2000)*, pages 103–107, San Francisco, CA, USA, December 2000.
- [523] W. Yu, W. Rhee, S. Boyd, and J. M. Cioffi. Iterative water-filling for Gaussian vector multiple access channels. *IEEE Trans. Inf. Theory*, 50(1):145–152, 2004.
- [524] Y. Yu, S. Murphy, and L. Murphy. A clustering approach to planning base station and relay station locations in IEEE 802.16j multi-hop relay networks. In *Proc. IEEE Int. Conf. on Communications (ICCC 2008)*, pages 2586–2591, Beijing, China, May 2008.
- [525] T. Yucek and H. Arslan. A survey of spectrum sensing algorithms for cognitive radio applications. *IEEE Commun. Surveys Tuts*, 11(1):116–130, 2009.
- [526] A. H. Zahran, B. Liang, and A. Saleh. Signal threshold adaptation for vertical handoff in heterogeneous wireless networks. ACM/Springer Mobile Networks and Applications, 11(4):625–640, 2006.
- [527] E. Zeydan, D. K. Tureli, C. Comaniciu, and U. Tureli. Bottleneck throughput maximization for correlated data routing: A game theoretic approach. In *Proc. 44th Ann. Conf. on Information Sciences and Systems (CISS 2010)*, pages 1–6, Princeton, NJ, USA, March 2010.
- [528] L. Zhang, X. Wang, and C. Liu. Channel assignment in multi-radio multi-channel wireless mesh network by topology approach. In *Proc. WRI Int. Conf. on Communications and Mobile Computing (CMC 2009)*, pages 358–362, Kunming, China, January 2009.
- [529] Q. Zhang, C. Guo, Z. Guo, and W. Zhu. Efficient mobility management for vertical handoff between WWAN and WLAN. *IEEE Comm. Mag.*, 41(11):102–108, 2003.
- [530] W. Zhang and K. Letaief. Cooperative spectrum sensing with transmit and relay diversity in cognitive-radio networks. *IEEE Trans. Wireless Commun.*, 7(12):4761–4766, 2008.
- [531] Y. Zhang, W. Zhuang, and A. Saleh. Vertical handoff between 802.11 and 802.16 wireless access networks. In *Proc. IEEE Global Communications Conf. (GLOBECOM 2008)*, pages 1–6, New Orleans, LA, USA, November–December 2008.
- [532] B. Q. Zhao, J. C. S. Lui, and D. Chiu. Analysis of adaptive incentive protocols for P2P networks. In *Proc. IEEE Conf. on Computer Communications (INFOCOM 2009)*, pages 325–333, Rio de Janeiro, Brazil, April 2009.
- [533] Q. Zhao and B. M. Sadler. A survey of dynamic spectrum access. *IEEE Signal Processing Magazine*, 24(3):79–89, 2007.
- [534] Y. Zhao, R. S. Adve, and T. J. Lim. Improving amplify-and-forward relay networks: Optimal power allocation versus selection. In *Proc. IEEE Int. Symp. on Information Theory*, pages 3114–3123, Seattle, WA, USA, July 2006.
- [535] L. Zheng and D. N. C. Tse. Diversity and multiplexing: A fundamental tradeoff in multiple-antenna channels. *IEEE Trans. Inf. Theory*, 49:1073–1096, 2003.
- [536] S. Zhong, J. Chen, and Y. R. Yang. Sprite: A simple, cheat-proof, credit-based system for mobile ad-hoc networks. In *Proc. IEEE Conf. on Computer Communications (INFOCOM* 2003), pages 1987–1997, San Francisco, CA, USA, March 2003.

Index

802.11, 14, 26–30, 37, 47, 158, 307, 308, 321, 322, 332, 337, 418 802.15, 14, 32, 35 802.16, 14, 25–27, 256, 257, 287–289, 293, 307, 308, 317, 320 802.16j, 219, 287, 299, 300, 302, 307, 413 802.20, 27 802.21, 257, 307 802.22, 51, 52

access point, 116, 119, 321, 332, 337, 343 ad hoc network, 37-40, 389 adaptive modulation and coding, 154, 443 additive white Gaussian noise, 151 admission control, 322, 337 admission-control, 339 aggregator, 339 Aloha, 37, 146, 329 attenuation factor, 18 auction theory, 221 applications, 370 double auction, 233 share auction, 232 VCG, 230 auction-theory applications, 389 audio on demand, 30 backoff, 116, 321, 326, 437 bargaining bargaining outcome, 172, 183 bargaining power, 177, 184, 485 bargaining problem, 173, 182, 183, 246, 271, 272, 278, 484 bargaining process, 172 bargaining situation, 171, 271, 484

bargaining theory, 171, 172, 205, 220, 270 base-station assignment, 263–267 baseband, 19, 32 battle of sexes, 59, 71 Bayesian game, 57, 101–103, 105, 107, 109, 123, 312, 315 applications, 257, 311–314, 316, 317

Bayesian Nash equilibrium, 102, 104, 107, 108, 111 Bayesian strategy, 315-317 equilibrium, 314, 315, 317 perfect Bayesian equilibrium, 104 beamforming, 48, 179 belief, 102 Bell number, 205 BER. 454 best-response, 66-68, 94, 106, 108, 111, 116, 118, 120, 131, 135, 138, 162, 259, 260, 265, 268, 283, 285, 297, 299, 303-305, 311, 314-317, 328, 440, 444, 450, 452 best-response dynamics, 95, 315-317 best-response, 66 bidder, 107 bimatrix, 296 binary exponential backoff, 322 bluetooth, 14, 31-35 Bondareva-Shapley Theorem, 192, 193 BPSK, 28 broadband wireless access, 255-257, 287, 336 broadband wireless network, 299 Brouwer-Kakutani Fixed-Point Theorem, 72, 485 Brown-von Neumann-Nash, 334 canonical coalitional game, 185, 189, 191, 194, 195, 197, 216, 217 applications, 198-200, 202 balanced game, 192, 193 convex game, 193, 194, 196 envy-free allocation, 200 goals, 190 properties, 189 solutions, 190 carrier-sense multiple-access with collision avoidance (CSMA/CA), 321, 322 CCK, 28, 29 CDMA network, 87, 115, 256-258, 263 CDMA2000, 14, 21, 24 IS-95, 24 TD-SCDMA, 24 WCDMA, 14, 21, 24

cellular network, 14, 255, 256, 258, 263, 279, 280, 287, 317, 336, 418 3G. 21-24 4G, 24 channel capacity, 51 channel model, 14, 15, 441 channel-state information, 452 characteristic form, 186, 187, 189, 199-201, 204, 215 characteristic function, 186, 187, 199, 203 cheat-proof strategy, 446 chicken game, 60, 65 chip rate, 36 churning, 158, 160, 336 closed-loop feedback control, 124 closed-loop information, 129 co-channel interference, 47 co-state variable, 129, 167, 169 coalition-formation game, 185, 203, 208, 211, 215, 216 algorithms, 205-208, 211, 214, 488, 493, 494, 498, 499 applications, 209, 210, 212-215, 492, 495, 499 cooperative file sharing, 497 defection function, 207, 208 dynamic, 203-205, 207, 208, 210 goals, 203 hedonic, 208 merge rule, 206-208, 211 Nash-stable partition, 214, 495 partition stability, 207, 208, 211 properties, 203 split rule, 207, 208, 211 stable coalitional structure, 420 static, 203 switch operation, 214 coalitional game, 171, 212, 220, 420, 421, 424, 488, 491, 492 basics, 185 classes, 185 coalition value, 185 coalitional-game theory, 185, 461 definition, 186 grand coalition, 189, 190, 196, 199-201, 209 introduction, 185 payoff, 186 payoff allocation, 186, 190, 196 restricted game, 204, 211 simple games, 193, 194 coalitional graph game, 185, 215-217, 219, 301, 307 applications, 219, 257, 287, 300-304, 307 balanced core, 219 coalitional stability, 218 goals, 216 Myerson value, 216, 219 Nash network, 218, 304, 305

network-formation game, 217-219, 301-304, 357 pairwise stability, 218 properties, 215 code-division multiple-access (CDMA), 24, 148 coalitional-game, 402 cognitive radio, 52, 215, 236, 418, 419, 421, 426, 432, 436, 450, 455, 458 coherent bandwidth, 18, 19, 35 combination, 35, 44, 52 common-use model, 418, 446 complete information, 57, 58, 61, 74, 309, 311, 316 complexity, 24, 42, 52, 139, 426 conditional probability mass function, 103 congestion control, 144, 330, 460, 461, 473, 474 contention window, 321-323, 325, 344, 432, 435, 436 continuous-kernel game, 65, 66, 69, 87 applications, 259, 264, 265, 268, 465, 471, 476, 483 Nash equilibrium, 67-69 relationship game, 67 cooperative game theory, 171, 172, 220, 488 cooperative spectrum sensing, 152, 420, 421 cooperative transmission, 300, 302, 303, 454 amplify-and-forward, 377 coded cooperation, 379 decode-and-forward, 200, 300, 303, 378, 450, 452 estimate-and-forward, 379 cooperative-transmission, 375 core, 190-194, 196, 197, 199, 203, 204, 208, 403 definition, 190, 191 drawbacks, 195 examples, 191, 199, 201, 202 existence, 191-194 graphical approach, 192, 193 properties, 191 correlated equilibrium, 91-95, 100, 420, 436, 439 correlated optimal, 93 correlated strategy, 91, 94 definition, 91 correlation, 18, 44 Cournot, 105, 107 CSMA, 29, 322, 323, 325, 329, 432, 434, 437 data fusion, 42-44 **DECT. 24** Dedicated Short Range Communications (DSRC), 30 Defense Advanced Research Projects Agency (DARPA), 37 delay delay profile, 18 delay spread, 18 propagation delay, 18 demand function, 159, 456 diagonal strict concavity, 69, 285 differential games, 124, 130, 131

diffraction, 16 **DIFS**, 29 direct-payment, 340 disagreement point, 173, 176, 183, 272, 273, 324, 484 dissatisfaction. 338 distributed coordination function (DCF), 29, 158, 321, 323, 434 distributed space-time coding, 454 diversity cooperative diversity, 382 diversity gain, 24 spatial diversity, 28, 47 dominant strategy, 61, 62 dominant-strategy equilibrium, 61 dominated strategy, 62 Doppler, 18, 19 Doppler shift, 18, 19 Doppler spread, 18, 19 duopoly, 105, 107, 136 dynamic bargaining, 178 dynamic game, 56, 74, 75, 77-80, 84, 104, 105, 134, 321, 322, 324, 340, 343 backward induction, 77, 82 multi-act, 77 repeated game, 80-84, 348 cartel-maintenance, 83, 84 finite horizon, 81, 82 Folk Theorem, 82, 83 imperfect monitoring, 84 infinite horizon, 81-83 observable actions, 80, 82 perfect monitoring, 80, 82 Tit-for-Tat, 83, 84, 326 trigger strategy, 83 sequential game, 74, 75 single-act, 77 stochastic game, 84, 85 dynamic programming, 124, 125, 130 dynamic spectrum access, 419, 420, 426, 432

EDGE, 24 eigenvalue, 142, 329 embedded operating system, 41 enhanced distributed coordination function (EDCF), 30, 337 equal fair, 194 error correction, 330 Ethernet, 40 evolutionary game, 139, 143–145, 148, 150, 152, 154, 155, 159–161, 164, 310, 330, 331, 336, 372 evolutionary equilibrium, 138, 142, 146, 161 evolutionarily stable, 140 evolutionarily stable strategies, 138, 139, 153 evolutionary-game, 138 exclusive-use model, 418, 455 fading fast fading, 19 flat fading, 19 frequency selective fading, 19 large-scale model, 15 multi-path fading, 18 shadowing, 17, 42 slow fading, 19 small-scale fading, 19 false alarm probability, 421, 423, 425, 448 FEC, 25, 330, 332 federal communications commission (FCC), 28, 36, 49 418 feedback information, 129 feedback Nash equilibrium, 130, 136 feedback Stackelberg equilibrium differential game, 134 feedback Stackelberg solution, 131 femtocell access point, 255, 257, 279-282 femtocell network, 279, 280, 320 **FFT. 28** flow control, 73, 85, 99, 460-462, 464-468, 471, 473, 474, 481, 482, 500 Folk Theorem, 449 free-space model, 15 frequency hopping, 28, 33 Gaussian frequency shift keying, 32 Gaussian random variable, 17, 446 GPS, 35, 42, 43, 51 gradient-play strategy, 328 graph form, 187, 216 group-rational, 190, 197 GSM, 24 guard guard band, 51, 52 guard subcarrier, 28 Hamilton-Jacobi-Bellman, 126 Hamiltonian, 129, 132, 166, 168 high occupancy vehicles (HOV), 50 HIPERMAN, 25 Hungarian method, 276-278 imperfect information, 74, 75, 77, 105 impulse response, 19 imputation, 190, 192, 199 IMT-2000, 24 incomplete information, 57, 74, 102-105, 114, 116, 120, 121, 311, 332 incomplete-information, 341 independent basic service set (IBSS), 40 individually rational, 83, 190, 197 indoor-propagation model, 18

infinite horizon, 136 intelligent transportation systems, 212 interference channel, 180–183, 200, 201, 281 interference management, 55 Internet, 97, 119, 288, 313, 332, 339, 460–462, 466, 472, 481, 482, 484–486, 499, 500 ISM, 28, 32, 33, 35, 418 iterated strict dominance, 62, 63, 95 iterated weak dominance, 63 ITU, 24, 35

Jacobi-play strategy, 329 Jacobian, 329, 431 jitter, 31, 331

k-connectivity, 38 Kalai–Smorodinsky solution, 178 Kleinrock independence approximation, 300

LADAR, 43 Lagrangian, 114, 150, 429, 456 line of sight, 16, 26, 27, 32 linear-state differential game, 167 localization, 42 log distance path-loss model, 18 logical link control, 27 LTE, 219, 269, 299, 300, 302, 307

malicious node, 110 marginal utility, 457 Markov chain, 160, 161, 288, 330, 331, 336, 441, 443 Markov decision process, 85 Markov modulated Poisson process, 288, 290, 291 matching pennies, 60, 65 matrix game, 58, 63, 75, 77, 90, 140 max-min fair, 438 maximum principle, 124-126, 129 mean opinion score, 331 mechanism design, 221 medium access control, 27, 51, 158, 321, 432 MIMO, 28, 49, 209-211 mixed strategy, 58, 69, 70, 111, 112, 139, 310 mixed-strategy, 111 mobile ad hoc network (MANET), 38, 39, 136 mobility, 14, 25, 30-31, 38-40, 52, 336 multi-user diversity, 269 multihoming, 332, 335 multi-hop, 109, 111, 220, 299, 300, 302, 303, 320, 326, 345, 452 multimedia resource management, 183 multi-path component, 20 multiple access, 94, 198-200 myopic strategy, 341-343, 445, 446

Nakagami distribution, 20 Nakagami-m channel, 289 Nash bargaining solution, 172, 176, 201, 434, 484 applications, 179, 182-184, 272-275, 277, 278, 485 definition, 174 examples, 175 generalized Nash bargaining solution, 177, 484 N-player, 176 Nash bargaining axioms, 173 Nash equilibrium, 58, 63-66, 72, 73, 76-80, 82-85, 90, 91, 95, 99, 102, 107, 113, 125, 130, 138, 144, 153, 162, 180, 181, 258-262, 265-268, 285, 297-299, 311, 314, 320, 324, 328, 336, 339, 420, 428, 431, 434, 446, 448, 450, 458, 462, 466-468, 471-473, 476-478, 483-486 asymptotic, 466-468, 470, 471 efficiency, 72 examples, 64 iterative-update algorithm, 260 mixed-strategy, 70-72, 92, 93, 310, 311 proper, 70, 71 pure-strategy, 64-70, 77, 86, 94, 95 selection, 72, 73, 485 Nash product, 174-177, 184, 272, 273 network-formation algorithm, 304, 305 network interface card, 40 neural network, 331 non-transferable utility, 187, 190, 191, 193-196, 200, 201, 205, 206, 216, 492 non-cooperative game, 55, 57, 88, 90, 98, 100, 138, 161, 257, 258, 260, 263-266, 293, 295, 297, 304, 308, 309, 311, 329, 339, 420, 427, 432, 452, 453, 461, 467, 471, 476, 482-484 action, 56, 57, 74, 75 basics, 55, 56 dynamic, 56, 74 extensive-form, 74, 75, 77, 78 finite game, 57, 89 game tree, 75 information set, 75 non-zero-sum game, 58, 85 non-cooperative game theory, 287 normal form, 56 static, 56, 57, 65, 74, 75, 86 strategic form, 56, 58, 59, 75 strategy, 56, 57, 74, 75 zero-sum game, 57, 61 non-cooperative game theory, 55, 217, 475 nucleolus, 196-198, 203, 204, 211 computation, 197 definition, 197 examples, 197 excess, 196 kernel, 196, 197

OFDMA, 270 OFDMA network, 28, 45, 46, 256, 257, 269, 270, 278, 455 open-loop information, 129 open-loop Nash equilibrium, 166 open-loop Stackelberg differential game, 131 optimal control, 126, 128, 130, 165, 167 optimal open-loop control, 128 optimal-control, 124, 125, 132 outage probability, 454 outdoor-propagation model, 17 overhead, 340 packet radio networks (PRNET), 37 parallel update, 260, 261 Pareto Pareto boundary, 176, 182 Pareto efficiency, 173 Pareto-optimal, 58, 72, 73, 84, 182, 184, 208, 220, 324, 325, 435, 436 Pareto order, 206-208, 212, 424 Pareto-superior, 72 Paris Metro Pricing model, 164 partial differential equation, 126 partition form, 187, 203, 215, 491, 492 path-loss factor, 17, 115, 116 PCS, 17, 49 peer-to-peer, 94, 215, 220, 488, 489, 492, 494 cooperative file sharing, 461, 487, 492 penalization mechanism, 325 perfect Bayesian equilibrium, 341 perfect information, 74-77, 282, 438 physical-layer security, 248 piconet, 31, 33, 34, 40 Pigouvian taxation, 431, 432 point coordination function (PCF), 29, 321 Poisson, 147, 159 Pontryagin, 126, 127, 129, 166 population, 138-140, 146, 148, 154, 159, 164, 333, 335 population state, 142, 165 potential game, 85-88, 99, 149, 150, 432 exact, 86 Nash equilibrium, 86 ordinal, 86 potential function, 85-88, 99 power control, 51, 58, 69, 87, 88, 95, 103, 112, 256-260, 262, 263, 265-268, 280, 282, 380, 381, 419, 426, 432, 459 preference, 102 preference relation, 206, 492 prepaid scheme, 340 price of anarchy, 38, 73, 335, 433 price of stability, 73 pricing, 27, 161, 259, 261-263, 266-269, 321, 335, 339, 369, 383, 386, 450, 461, 474, 478, 482, 483, 485

primary users, 151, 418, 424, 429, 439, 450, 455 Prisoner's Dilemma, 59, 62, 65, 66, 72, 80, 82-84, 142 private information, 109, 119, 122, 312, 420, 437-439, 443, 444, 446, 449 propagation delay, 329 proportional fair, 179, 194, 273, 274, 447, 450 protocol data unit (PDU), 26 pseudo-random, 33 pseudo-subjective quality assessment, 332 punishment mechanism, 323 pure strategy, 57, 69, 70, 77, 111, 122, 141, 147.310 **OAM**, 26 **QPSK**, 28 quality of service, 46, 421 R-factor, 302, 303, 305 random update, 260, 261 random walk, 39 Rayleigh, 19, 20, 117, 118, 421, 447 reflection model. 15 relay station, 219, 299-305 replicator dynamics, 141, 144, 146, 148-150, 153, 155, 156, 158, 310, 334 reseller, 341 resource allocation, 256, 257, 269, 287 Ricean distribution, 19 RMS, 18, 19 roaming, 26, 30 routing, 73, 85, 99, 149, 219, 220, 345, 460, 462, 464-468, 471, 487, 500 RTS/CTS, 29 Rubinstein process, 178 satisfaction, 121, 159, 164, 337, 338, 438 scalability, 43 scatter, 16, 20 Schauder Fixed-Point Theorem, 285 scheduling, 25, 26, 37, 40, 43 secondary users, 151, 154, 418, 421, 424, 428, 432, 434, 436, 440, 443, 445, 448, 452, 454 service provider, 121, 158, 160, 336, 339, 341, 343, 421, 457, 459-462, 466, 473-475, 478, 479, 481, 482, 484, 486 Shapley value, 195, 196, 203, 204, 211, 216 axioms, 195 complexity, 196 definition, 195 examples, 199 shared-use model, 419 sigmoid, 338 signal-to-interference-plus-noise ratio, 148 signal-to-noise ratio, 37, 381

sleeping mechanism. 43 social welfare, 168, 169, 443 software-defined radio, 419 space-division multiple-access (SDMA), 49 space-time coding, 49, 452 spectral efficiency, 52, 455 spectrum mobility, 419 spectrum sensing, 151, 419, 421, 423, 425, 441 spread spectrum, 24, 28, 31, 33 stability, 38, 124, 142, 148, 149, 322 Stackelberg game, 88, 90, 131, 329, 342, 380, 382, 387, 389, 420, 431, 452, 455, 459 applications, 257, 280, 282, 382, 474-476, 479-481 follower, 89 leader, 89 multi-follower, 90, 91, 475 multi-leader, 91 multi-leader multi-follower, 282, 284, 285 reaction set. 89-91 Stackelberg equilibrium, 90, 122, 124, 125, 133, 134, 284-286, 320, 458, 479 Stackelberg strategy, 89-91 Stackelberg utility, 89, 90 standard function, 68, 266 strategic game, 56, 57, 69, 70, 72, 80-82, 86, 87, 91, 96, 258, 259, 466, 483 subgame-perfect equilibrium, 78, 79, 82, 448 superadditive, 189-191, 199, 201-203, 210 supermodular game, 94-96, 267, 328 applications, 266, 267 definition, 95 increasing differences, 94 positive spillovers, 95 properties, 95

TCP, 144, 154, 155, 333, 334

time-division duplexing, 33 time-division multiple-access (TDMA), 447 time synchronization, 43 transferable utility, 186, 187, 190-196, 200, 203, 205, 206, 211, 214, 216 tree architecture, 299, 300 turbo code, 25 two-ray model, 15 UDP, 333, 334 Ultra Wideband, 14, 35-37 UMTS, 24, 337 upper semi-continuous, 86, 95 value function, 126, 127, 130 vehicular network, 119, 212 vertical handover, 257, 307-311, 314-317 video on demand, 30 voice over IP, 30, 302, 303, 330 Wardrop equilibrium, 96-100, 335, 458, 466, 471 water-filling, 113, 269, 283, 286, 429 Web browsing utility, 341 WiFi, 14, 25, 27-31, 257, 279, 307, 309, 339 WiMAX, 14, 24-27, 154, 164, 219, 256, 269, 299, 307, 309, 311 wired equivalent privacy, 30 wireless personal area network, 14, 31-37, 40 WLAN, 14, 23, 27, 31, 116, 158, 311, 321, 322, 324, 326, 332, 336 WLAN authentication and privacy infrastructure (WAPI), 30 WMAN, 24-27 WSN, 40-44

ZigBee, 14, 35